Copyright is owned by the Author of the thesis. Permission is given for a copy to be downloaded by an individual for the purpose of research and private study only. The thesis may not be reproduced elsewhere without the permission of the Author. Exploring the Data Needs and Sources for Severe Weather Impact Forecasts and Warnings

A thesis presented in partial fulfilment of the requirements for the degree of

Doctor of Philosophy

in

Emergency Management at Massey University, Wellington, New Zealand

Sara Ebony Zylka Harrison

2021

Abstract

Early warning systems offer an essential, timely, and cost-effective approach for mitigating the impacts of severe weather hazards. Yet, notable historic severe weather events have exposed major communication gaps between warning services and target audiences, resulting in widespread losses. The World Meteorological Organization (WMO) has proposed Impact Forecasts and Warnings (IFW) to address these communication gaps by bringing in knowledge of exposure, vulnerability, and impacts; thus, leading to warnings that may better align with the position, needs, and capabilities of target audiences.

A gap was identified in the literature around implementing IFWs: that of accessing the required knowledge and data around impacts, vulnerability, and exposure. This research aims to address this gap by exploring the data needs of IFWs and identifying existing and potential data sources to support those needs.

Using Grounded Theory (GT), 39 interviews were conducted with users and creators of hazard, impact, vulnerability, and exposure (HIVE) data within and outside of Aotearoa New Zealand. Additionally, three virtual workshops provided triangulation with practitioners. In total, 59 people participated in this research. Resulting qualitative data were analysed using GT coding techniques, memo-writing, and diagramming.

Findings indicate a growing need for gathering and using impact, vulnerability, and exposure data for IFWs. New insight highlights a growing need to model and warn for social and health impacts. Findings further show that plenty of sources for HIVE data are collected for emergency response and other uses with relevant application to IFWs. Partnerships and collaboration lie at the heart of using HIVE data both for IFWs and for disaster risk reduction.

This thesis contributes to the global understanding of how hydrometeorological and emergency management services can implement IFWs, by advancing the discussion around implementing IFWs as per the WMO's guidelines, and around building up disaster risk data in accordance with the Sendai Framework Priorities. An important outcome of this research is the provision of a pathway for stakeholders to identify data sources and partnerships required for implementing a hydrometeorological IFW system.

I hear hurricanes a-blowing ... I fear rivers overflowing ... Hope you got your things together Hope you are quite prepared ... Lyrics adapted from "Bad Moon Rising," written by John C. Fogerty of *Creedence Clearwater Revival*

Acknowledgements

"It takes a village to support a PhD student." - unknown

First and foremost, I would like to express my deepest appreciation to my supervisors, Doctor Sally Potter from GNS Science, Doctor Raj Prasanna, Doctor Emma Hudson-Doyle, and Professor David Johnston from the Joint Centre for Disaster Research (JCDR), Massey University. I would not have been able to complete this thesis without you all on my team. Thank you for giving me the independence to explore my PhD topic, while also being readily available to offer guidance when needed. Sally, thank you for providing me with my scholarship and for this opportunity to come to New Zealand for my PhD. Your mentorship shaped me into the researcher that I am today; you provided invaluable opportunities throughout my PhD journey and beyond, and you shared in my enthusiasm for my PhD topic. For all of this and more, you have my deepest gratitude. Raj, thank you for always being available for a chat. I could always trust your invaluable advice on navigating academic processes and respective relationships. Emma, thank you for providing an empathetic ear and support during all the highs and lows of this journey; you always knew the right thing to say when it was needed. David, thank you for the opportunities that you gave me to get my research out there, to meet people in the field, and to enrich my experience as a PhD student at the JCDR.

Completing my PhD at the JCDR was a privilege and a joy that I will never forget. I am grateful to the JCDR staff and students who welcomed me with open arms upon my arrival from Canada. I also wish to thank the Social Science team at GNS Science for welcoming me as a member of the team while I came in and out to help Sally and the team with supporting work throughout my PhD journey and for opening opportunities to me at GNS Science.

I am extremely grateful to GNS Science and the Resilience to Nature's Challenges programme for providing me with the scholarship for this PhD study. I am also appreciative of the grants I received from the World Meteorological Organization (WMO), Massey University's School of Psychology, the National Center for Atmospheric Research (NCAR), and the Meteorological Society of New Zealand that allowed me to travel both nationally and internationally to share my research and have invaluable networking experiences that directly aided my PhD data collection.

I am deeply indebted to all my participants who agreed to participate in my study; this research would not have been possible without you. Thank you for providing your time, expertise, and experiences to inform and shape my study and my knowledge pool. I thank the MetService, in particular, for your insightful conversations and advice throughout this research.

Completing a PhD is a unique experience. I am grateful for having a PhD cohort to share this experience with and to learn from each other. Doctor Marion Tan, Doctor Ashleigh Rushton, Lisa McLaren, Doctor Lauren Vinnell, Neeraj Shankar, Lesley Gray, Doctor Syed Yasir Imtiaz, Bruce Pepperell, Doctor Martín García Cartagena, Doctor Nancy Brown, and Rangika Nilani, thank you all for your companionship and wisdom. I wish you all the best in your endeavours and I look forward to watching your careers progress.

My PhD journey was enriched by the people in my life, both near and far, to whom I would like to extend my heartfelt thanks. My dearest friends, Mingsze, Jenny, Alex, Shirley, and Nathan, thank you all for cheering me on from Canada, for (virtually) showing up for me, and for constantly reminding me that I have you in my corner. I look forward to the next time we can reunite in person. Mingsze, my kindred spirit, you never let me down. Thank you for always being available for a rant, for celebrating each other's successes, and for being an overall amazing friend. Kelsea and Alicia, my 'badass women' here in New Zealand, you inspire me and motivate me to be the best version of myself. We developed an unbreakable bond over the last four years that will only become stronger. Amy and Jeremy, you both took me under your wings and brought me along on your adventures. If this doesn't work out, I would love to be your goat nanny. Chris and Sandy, you always seemed to know when it was time to check in on me. Thank you both for sharing your vibrant energy and nurturing my soul. Catalina, we found a sister in each other when you moved to Wellington, and I am grateful for your endless love and support. Rosie, I cherish your friendship over the last four years. I appreciate you always coming up with fun adventures like hunting down pierogis in Wellington. Michael, thank you for being my PhD/climbing companion, and for helping me stay focused on achieving my goals in both respects.

I must also express my sincerest thanks to my counsellor and therapist, Russell, who has walked alongside me during my PhD journey. You were instrumental in helping me find the courage within me to take ownership of my PhD and thrive in life.

I never expected to gain a second family when I moved to New Zealand, but that is exactly what happened. I count my lucky stars every day for meeting Margaret, John, Brian, and Tracy during my first week in Wellington. I am eternally grateful to my 'kiwi parents', Margaret and John, for welcoming me into your home for Sunday dinners and into your wonderful family as your surrogate daughter. You gave me a 'home away from home' and you were there to pick me up and dust me off in the hard times. Brian, thank you for enlightening me on the quirky ways of New Zealand, for being my writing buddy in Queenstown while everyone else enjoyed the amazing sights, and for proofreading my thesis. Tracy, thank you for being my lovely host when I first landed in New Zealand and for connecting me to this beautiful whānau.

Most importantly, thank you to my family for your unwavering support and belief in me. To my sisters, Asia and Santana, you grounded me to my roots whenever I felt lost. Auntie Lydia, you continue to provide a safe space for me to confide in you. Shane, thank you for your continuous support. And finally, my wonderful parents, Caroline and Peter, I cannot begin to express my thanks to you for encouraging me to pursue my dream of completing a PhD in New Zealand. You continue to give me the strength and courage to fulfil my goals and dreams. Thank you for everything. Please give George several cookies from me (perhaps one for every chapter of my thesis).

Table of Contents

ABSTRACT	I
ACKNOWLEDGEMENTS	V
TABLE OF CONTENTS	VII
LIST OF FIGURES	XI
LIST OF TABLES	XIII
LIST OF ACRONYMS AND ABBREVIATIONS	XIV
CHAPTER ONE: INTRODUCTION	2
	2
1.2 Severe Weather and Society	
1.2.1 Severe Weather Warning System Challenges	
1.2.2 Impact Forecasting and Warning Systems	7
1.3 Research Problem and Aims	8
1.4 Thesis Outline	9
1.5 CHAPTER SUMMARY	10
CHAPTER TWO: BACKGROUND AND CONTEXT	12
2.1 Early Warning Systems	12
2.1.1 Severe Weather Early Warning Systems	13
2.1.2 Early Warning Systems Performance	16
2.2 IMPACT FORECASTS AND WARNING SYSTEMS	17
2.2.1 Impact Forecasting and Warning System Decision-Making and Communication	n 10
Processes and Tools	19 22
2.2.2 Evaluation of impact of leasts and warnings	20
Systems	24
2.2.4 Crowdsourcing, Social Media, and Volunteered Geographic Information	25
2.3 Sendai Framework for Disaster Risk Reduction	27
2.4 New Zealand Weather Warning System	29
2.4.1 New Zealand's Severe Weather Hazards	29
2.4.2 Severe Weather Warnings	36
2.4.4 The New Zealand Emergency Management Sector	41
2.5 CHAPTER SUMMARY	44
CHAPTER THREE: RESEARCH DESIGN	47
	18
3.1.1 A Pragmatic Choice	40
3.2 Methodology	52
3.2.1 Research Approach	52
3.2.2 Research Strategy	53
3.3 DATA COLLECTION METHODS	61
3.3.1 Interviews	62
	63
3.4 AMALT FICAL TECHNIQUES AND TOOLS	04 77
3.4.2 Coding	64

3.5 ETHICAL CONSIDERATIONS	67
3.6 EVALUATION OF RIGOUR AND CREDIBILITY	/0/ 72
S.7 CHAPTER SUMIMARY	
CHAPTER FOUR: DATA COLLECTION, ANALYSIS, AND OVERARCHING F	FINDINGS
	/4
4.1 DATA COLLECTION: INTERVIEWS	74
4.2 DATA COLLECTION: WORKSHOPS	76
4.2.1 Auckland Region Workshop	
4.2.2 Southland Region Workshop	
4.2.3 Community of Research Workshop	
4.3 DATA ANALYSIS AND REY FINDINGS	∠0
4.3.1 Memo writing and diagramming	
4.5.2 County Λ Λ Chadted Slimmady	
CHAPTER FIVE: VOLUNTEERED GEOGRAPHIC INFORMATION FOR PEO CENTRED SEVERE WEATHER EARLY WARNING: A LITERATURE REVIEW .	PLE- 117
5 1 Δρετραστ	117
5.3 People-Centred Farly Warning Systems	110
5.5 METHOD	120 123
5.6 PECHITS	123
5.6 1 Hazard Type	12J 126
5.6.2 Farly Warning System Components	1.32
5.6.3 VGI Platforms and Data Types	
5.7 Discussion	
5.7.1 Volunteered Geographic Information in Severe Weather Early Warning	g Systems
5.7.2 Implications for the different types of VGI	137
5.8 CONCLUSION	
CHAPTER SIX' IDENTIFYING THE IMPACT RELATED DATA LISES AND GA	
HYDROMETEOROLOGICAL IMPACT FORECASTS AND WARNINGS	
6.1 Abstract	
6.2 Introduction	
6.2.1 Elements of Impact-Based Forecasting and Warning Systems	
6.2.2 The Impact Forecasting and Warning Value Chain	
6.3 Research Method	152
6.4 Findings and Discussion	156
6.4.1 Data Uses in the Impact Forecasting and Warning Value Chain	156
6.4.2 Need for more understanding on different types of impacts	171
6.5 Conclusions	174
CHAPTER SEVEN: 'WHERE OH WHERE IS THE DATA?': IDENTIFYING DA SOURCES FOR HYDROMETEOROLOGICAL IMPACT FORECASTS AND W	ΓA VARNINGS
IN AOTEAROA NEW ZEALAND	177
7.1 Abstract	178
7.2 Introduction	
7.3 IMPACT FORECASTING AND WARNING DATA GAPS AND SOURCES	179
7.3.1 Hazards	
7.3.2 Impacts	
7.3.3 Vulnerability	

7.3.4 Exposure	183
7.4 Data Characteristics for Early Warning Systems	183
7.5 Research Methods	184
7.6 Findings and Discussion	188
7.6.1 Data Sources	188
7.6.2 Data Creation, Collection, and Use	209
7.7 Conclusions and Limitations	214
CHAPTER EIGHT: 'SHARING IS CARING': A SOCIO-TECHNICAL ANALYSIS OF TH SHARING AND GOVERNING OF HYDROMETEOROLOGICAL IMPACT DATA IN AOTEAROA NEW ZEALAND	НЕ 217
8.1 ABSTRACT	218
8.2 INTRODUCTION	219
8.2.1 Data Governance	219
8.2.2 Data Access and Sharing	221
8.3 RESEARCH METHODS	
8.4 FINDINGS AND DISCUSSION	227
8.4.1 The Roles and Responsibilities of Data Custodianship	227
8.4.2 Data Access and Sharing	233
8.5 Conclusions and Limitations	246
CHAPTER NINE: INTEGRATION OF PARTNERSHIPS AND COLLABORATION TO	
SUPPORT IFWS AND HIVE DATA COLLECTION, ACCESS, AND USE	248
	240
9.1 FARTNERSHIPS AND COLLABORATION CORE CATEGORY DEFINED	
9.2 FROPERTIES AND DIMENSIONS OF THE CORE CATEGORY. FARTNERSHIPS AND	າເວ
9.2.1 Types of partnerships	
9.2.3 Strategies for building and strongthening partnerships and collaboration	254
9 3 PARTNERSHIPS AND COLLARORATION IN THE IMPACT FORECASTING AND WARNING	255
Systems Phenomenon	257
9 3 1 Sharing Data and Knowledge	257
9.3.2 Re-defining Warning Thresholds	237
9.3.3 Consistent Warning Messages	260
9.4 Partnerships and coll aboration in the HIVE Data Phenomenon	
9.4.1 Data Collection	
9.4.2 Data Governance, Access, and Sharing	263
9.5 DISCUSSION AND INTEGRATION	264
9.6 Chapter Summary	265
	260
	200
10.1 Research Overview	268
10.1.1 Research Question 1: What are the current and potential uses of Volunteered	1
Geographic Information (VGI) for severe weather warnings?	271
10.1.2 Research Question 2: What are the data uses and gaps for impact forecasts a	nd 272
10.1.3 Research Question 3: What are the sources for HIVE data?	272
10.1.4 Research Question 4: What is the governance and acquisition process for HIV	275 /F
data for severe weather hazards in New Zealand?	
10.1.5 Research Question 5: How can partnerships and collaboration facilitate bette	er er
collection, creation, and access to HIVE data for IFWs?	274
10.2 Research Implications and Contributions	274
10.2.1 Data Uses and Gaps for Impact Forecasts and Warnings	275
10.2.2 Sources of hazard, impact, vulnerability, and exposure (HIVE) data in New	
Zealand	278

10.2.3 Management and Governance of HIVE Data 10.2.4 The importance of Partnerships and Collaboration for implementing IFWs 10.3 LIMITATIONS AND FUTURE RESEARCH DIRECTIONS 10.4 CONCLUSION	279 280 281 283
APPENDIX A: STATEMENTS OF CONTRIBUTION	286
APPENDIX B: INFORMATION SYSTEMS RESEARCH PARADIGMS	290
APPENDIX C: RESEARCH STRATEGIES	293
APPENDIX D: INTERVIEW GUIDES	295
APPENDIX E: WORKSHOP AGENDA AND ACTIVITIES	309
APPENDIX F: HUMAN ETHICS NOTIFICATIONS, INFORMATION SHEETS, AND CONSENT FORMS	311
APPENDIX G: AUCKLAND AND SOUTHLAND HAZARD PROFILES	320
APPENDIX H: CODEBOOK	323
APPENDIX I: AXIAL CODING RESULTS	346
APPENDIX J: CHAPTER FIVE PUBLICATION	356
APPENDIX K: CHAPTER SIX PUBLICATION © AMERICAN METEOROLOGICAL SOCIETY. USED WITH PERMISSION.	375
APPENDIX L: CHAPTER SEVEN PUBLICATION	397
APPENDIX M: CHAPTER EIGHT PUBLICATION	414
REFERENCES	427

List of Figures

FIGURE 2.1. FOUR OPERATIONAL COMPONENTS OF AN EARLY WARNING SYSTEM.	13		
FIGURE 2.2. GLOBAL POSITIONING OF NEW ZEALAND	30		
FIGURE 2.3. NEW ZEALAND REGIONAL MAP	31		
FIGURE 2.4. THE METSERVICE'S COLOUR-CODED SEVERE WEATHER WATCHES AND WARNINGS	40		
FIGURE 2.5. IMAGE RELEASE OF THE FIRST RED WARNING ISSUED BY THE METSERVICE ON 3 FEBRUARY 2020	41		
FIGURE 2.6 MAP OF THE 16 CDEM GROUPS IN NEW ZEALAND WITH THEIR RESPECTIVE GROUP MEMBERS	44		
FIGURE 3.1. CONCEPTUAL FRAMEWORK OF THE RESEARCH DESIGN	48		
FIGURE 3.2. CONCEPTUAL FRAMEWORK OF THE ES-GT ANALYTICAL PROCESS EMPLOYED IN THIS RESEARCH	65		
FIGURE 4.1. FINAL IMAGE OF THE POPULATED MURAL WHITEBOARD FROM THE AUCKLAND WORKSHOP	77		
FIGURE 4.2. FINAL IMAGE OF THE POPULATED MURAL WHITEBOARD FROM THE SOUTHLAND WORKSHOP	79		
FIGURE 4.3. DIAGRAM OF WEST COAST WARNING PROCESS FOLLOWING AN INTERVIEW WITH THE WEST COAST CDEM	Λ		
GROUP	85		
FIGURE 4.4. DIAGRAM OF THE BAY OF PLENTY SEVERE WEATHER WARNING PROCESS	85		
FIGURE 4.5. DIAGRAM OF SEVERE WEATHER WARNING CHAIN BASED ON INTERVIEWS	86		
FIGURE 4.6. FIRST ITERATION OF THE IMPACT DATA FRAMEWORK	87		
FIGURE 4.7. SECOND ITERATION OF THE IMPACT DATA FRAMEWORK FOLLOWING THE EMERGENCE OF THE EXPOSURE TH	IEME88		
FIGURE 4.8. DATA PROCESS FRAMEWORK	89		
FIGURE 4.9. IMPACT DATA PROCESS FRAMEWORK PRESENTED TO GNS WORKSHOP PARTICIPANTS IN FEBRUARY 2021.	91		
FIGURE 4.10. AXIAL CODING PROCESS	94		
Figure 4.11. The open codes and categories that make up the Impact Forecasting and Warning Systems			
Phenomenon	95		
FIGURE 4.12. THE OPEN CODES AND CATEGORIES THAT MAKE UP THE HIVE DATA PHENOMENON.	96		
FIGURE 4.13. CONTEXTUAL CONDITIONS IDENTIFIED FROM AXIAL CODING.	98		
FIGURE 4.14. THE CAUSAL CONDITIONS IDENTIFIED IN THE AXIAL CODING PROCESS.	99		
FIGURE 4.15. INTERVENING CONDITIONS IDENTIFIED RELATING TO THE IFW SYSTEMS PHENOMENON.	100		
FIGURE 4.16. INTERVENING CONDITIONS IDENTIFIED RELATING TO THE HIVE DATA PHENOMENON	101		
FIGURE 4.17. INTERVENING CONDITIONS RELATING TO GOVERNANCE	102		
FIGURE 4.18. ACTION/INTERACTION STRATEGIES IDENTIFIED TO ADDRESS INTERVENING CONDITIONS AND OTHER			
CHARACTERISTICS OF THE IFW SYSTEMS PHENOMENON.	103		
FIGURE 4.19. ACTION/INTERACTION STRATEGIES IDENTIFIED TO ADDRESS DATA COLLECTION INTERVENING CONDITIONS	105		
FIGURE 4.20. ACTION/INTERACTION STRATEGIES IDENTIFIED FOR ADDRESSING DATA SHARING, MANAGEMENT, AND			
INTEGRATION INTERVENING CONDITIONS.	105		
FIGURE 4.21. ACTION/INTERACTION STRATEGIES FOR ADDRESSING GOVERNANCE CONTEXTUAL CONDITIONS AND			
TECHNOLOGICAL ADVANCEMENT CAUSAL CONDITIONS.	106		
FIGURE 4.22. CONSEQUENCES AND OUTCOMES OF IMPLEMENTING IFW SYSTEMS.	107		
FIGURE 4.23. CONSEQUENCES AND OUTCOMES OF COLLECTING HIVE DATA.	108		
FIGURE 4.24. CONSEQUENCES AND OUTCOMES OF DATA SHARING, MANAGEMENT, AND INTEGRATION.	109		
FIGURE 4.25. CONSEQUENCES AND OUTCOMES RELATED TO GOVERNANCE.	110		
FIGURE 4.26. RELATIONSHIPS BETWEEN THE TWO PHENOMENA UNDER STUDY AND THE CORE CATEGORY.	113		
FIGURE 4.27. LINKS BETWEEN THE TWO PHENOMENA AND THE RESULTING CORE CATEGORY FOR THIS PHD STUDY	115		
FIGURE 5.1. FOUR OPERATIONAL COMPONENTS OF AN EARLY WARNING SYSTEM	118		
FIGURE 5.2. DISTRIBUTION OF VGI PLATFORMS USED FOR EACH EARLY WARNING SYSTEM (EWS) FRAMEWORK COMP	ONENT		
	135		
FIGURE 5.5. VOLUNTEERED GEOGRAPHIC INFORMATION FOR PEOPLE-CENTRED SEVERE WEATHER EARLY WARNING SYS	1EMS.		
	361 1 10		
FIGURE 0.1. A VISUAL REPRESENTATION OF THE SEVERE WEATHER WARNING VALUE CHAIN	149		
GURE 6.2. DATA INPUTS AND USES FOR IMPACT FORECASTS AND THE TWO MAIN APPROACHES			

FIGURE 6.3. DATA INPUTS AND USES FOR DEFINING IMPACT THRESHOLDS	.163
FIGURE 6.4. DATA INPUTS AND USES FOR UPDATING WARNINGS.	.166
FIGURE 6.5. VERIFICATION OF IMPACT WARNINGS	.167
FIGURE 6.6. THIS FIGURE REPRESENTS THE WARNING VALUE CHAIN AND THE ASSOCIATED DATA INPUTS AND OUTPUTS FO	R
EACH COMPONENT	.170
FIGURE 7.1. RESULTS OF APPLYING THE CODING PARADIGM	.209
FIGURE 8.1. SUMMARY OF FINDINGS RELATING TO THE CAUSAL CONDITIONS, INTERVENING CONDITIONS, AND	
ACTION/INTERACTION STRATEGIES RELATING TO DATA ACCESS AND SHARING, IDENTIFIED USING THE ES-GTM CODIN	NG
PARADIGM. BLUE BOXES REPRESENT ORGANISATION ASPECTS AND GREEN BOXES REPRESENT TECHNICAL ASPECTS	.233
FIGURE 9.1. INTEGRATIVE DIAGRAM OF THE MICRO-THEORY RESULTING FROM THIS DOCTORAL RESEARCH	.251
FIGURE 9.2. A VENN DIAGRAM DISPLAYING AGENCIES THAT POSSESS HIVE DATA IN NEW ZEALAND	.258
FIGURE 9.3. A CONCEPTUAL EXAMPLE OF A SEVERE WEATHER WARNING CHAIN IN NZ AND THE PARTNERS INVOLVED IN TH	IS
CHAIN	.261

List of Tables

TABLE 1.1. SUMMARY OF NOTABLE, HISTORIC EVENTS IN WHICH SOME MEMBERS OF THE PUBLIC FAILED	TO
UNDERSTAND AND RESPOND TO WARNINGS.	6
TABLE 1.2. THE FOUR TYPES OF DATA NEEDED FOR IFWS.	7
TABLE 1.3. CHAPTER AND RESEARCH ALIGNMENT.	9
TABLE 2.1. SUMMARY OF WARNING VALUE CHAIN COMPONENTS	14
TABLE 2.2. DEFINITIONS AND EXAMPLES OF HAZARD-BASED WARNINGS, IMPACT-ORIENTED WARNINGS	1
IMPACT-BASED WARNINGS, AND IMPACT WARNINGS	18
TABLE 2.3. EXAMPLES OF SECTORS AND AGENCIES NEEDED FOR BUILDING PARTNERSHIPS TO DEVELOP	IFW
SYSTEMS	21
TABLE 2.4. METEOROLOGICAL FORECAST AND WARNING SERVICES PROVIDED BY THE METSERVICE	37
TABLE 2.5. THE THRESHOLDS AS DEFINED BY THE METSERVICE	39
TABLE 3.1. SUMMARY OF PARADIGMS USED IN IS RESEARCH	50
TABLE 3.2. SUMMARY OF COMMON RESEARCH STRATEGIES USED IN IS RESEARCH	54
TABLE 3.3. SUMMARY OF THE THREE MAIN STRANDS OF GROUNDED THEORY METHODOLOGY	59
Table 3.4. Summary of recommendations for IS researchers using Evolved Straussian	
GROUNDED THEORY	62
TABLE 3.5. CODING PARADIGM SUMMARY	66
Table 3.6. Interviewee codes	68
TABLE 3.7. CRITERIA FOR EVALUATING QUALITATIVE RESEARCH	70
TABLE 4.1. SUMMARY OF TARGETED AND RECRUITED INTERVIEW PARTICIPANTS	75
TABLE 4.2. OPEN CODES ABOUT PARTNERSHIPS AND COLLABORATION	.112
TABLE 5.1. SUMMARY OF VGI TYPES	.121
TABLE 5.2. SEARCH STRING EMPLOYED IN EBSCO DISCOVERY AND SCOPUS DATABASES	.124
TABLE 5.3. CATEGORIES FOR LITERATURE REVIEW.	.125
TABLE 5.4. SUMMARY OF SELECTED STUDIES COVERING FLOOD HAZARDS	.127
TABLE 5.5. SUMMARY OF SELECTED STUDIES COVERING OTHER SEVERE WEATHER HAZARDS	.130
TABLE 6.1. DEFINITION AND EXAMPLES OF THE WARNING TERMS USED IN THIS PAPER	.147
TABLE 6.2. THE FOUR TYPES OF DATA NEEDED FOR IFWS.	.148
TABLE 6.3. SUMMARY CLASSIFICATION OF INTERVIEW PARTICIPANTS	.153
Table 6.4. Participant Codes	.154
TABLE 7.1. SUMMARY OF DATA SOURCES AND DATASETS FOR HAZARD, IMPACT, VULNERABILITY, AND	
EXPOSURE DATA	.181
Table 7.2. Participant Codes	.185
TABLE 7.3. SUMMARY OF THE CODING PARADIGM DIMENSIONS	.187
TABLE 7.4. SUMMARY OF HAZARD DATA SOURCES IN NEW ZEALAND	.190
TABLE 7.5. SUMMARY OF IMPACT DATA SOURCES	.194
TABLE 7.6. SUMMARY OF VULNERABILITY DATA SOURCES	.203
TABLE 7.7. SUMMARY OF EXPOSURE DATA SOURCES	.207
Table 8.1. Interviewee Codes	.224
TABLE 8.2. CODING PARADIGM SUMMARY	.226
TABLE 8.3. SUGGESTED (S), PROPOSED (P), OR APPOINTED (A) STEWARDS AND CUSTODIANS OF THE	HIVE
DATA SETS	.229
Table 9.1. Examples of themes relating to the Partnerships and Collaboration core	
CATEGORY FROM THE MAIN RESULTS OF EACH CHAPTER	.250
TABLE 9.2. SUMMARY OF THE PROPERTIES AND DIMENSIONS OF THE PARTNERSHIPS AND COLLABORAT	ION
CORE CATEGORY	.253
TABLE 10.1. ALIGNMENT OF CHAPTERS WITH RESEARCH QUESTIONS AND OBJECTIVES.	.269

List of Acronyms	and Abbreviations
------------------	-------------------

ACRONYM /	FULL TERM
ACC	Accident Compensation Corporation
ACC	Action Research
	Application Programming Interface
CAD	Common Alerting Protocol
CAP	Civil Defence and Emergency Management
CINAS	Coordinated Incident Management System
CIIVIS	
COP	
	District Legith Deard
DHR	District Health Board
DUC	Department of Conservation
DRR	Disaster Risk Reduction
DSS	Decision Support System
DSR	Design Science Research
EF SCALE	Enhanced-Fujita Scale
EM	Emergency Management
EOC	Emergency Operations Centre
EQC	Earthquake Commission
ES-GT	Evolved-Straussian Grounded Theory
EWS	Early Warning Services
FENZ	Fire and Emergency New Zealand
FEWS	Flood Early Warning System
GIS	Geographical Information System
GIS4EM	Geographical Information System for Emergency Management
GNS SCIENCE	Te Pū Ao Institute of Geological and Nuclear Sciences Limited
GOV.	Government
GT	Grounded Theory
GWE	Global Weather Enterprise
HIVE	Hazard, Impact, Vulnerability, and Exposure
HYD.	Hydrological
HIWEATHER	High Impact Weather
IBFW	Impact-Based Forecasts and Warnings
IDSS	Impact-based Decision Support System
IFRC	International Federation of Red Cross and Red Crescent Societies
IFW	Impact Forecast and Warning
IFWS	Impact Forecasting and Warning System
IS	Information Systems
LGGA	Local Government Geospatial Alliance
LGNZ	Local Government New Zealand
LINZ	Toitū Te Whenua Land Information New Zealand
MBIE	Ministry of Business, Innovation, and Employment
MET.	Meteorological
METOFFICE	United Kingdom Meteorological Office

ACRONYM / ABBREVIATION	FULL TERM
METSERVICE	Meteorological Service of New Zealand Limited – Te Ratonga Tirorangi
MCDEM	Ministry of Civil Defence and Emergency Management
MFE	Ministry for the Environment
МОН	Ministry of Health
MPI	Ministry of Primary Industries
NAT.	National
NEMA	National Emergency Management Agency
NHP	Natural Hazards Partnership
NIWA	National Institute of Water and Atmospheric Research
NMHS	National Meteorological and Hydrological Services
NMS	National Meteorological Services
NWP	Numerical Weather Prediction
NWS	National Weather Service
NZ	Aotearoa New Zealand
NZD	New Zealand Dollars
NZDF	New Zealand Defence Force
NZTA	Waka Kotahi NZ Transport Agency
PACRIS	Pacific Risk Information System
PIMS	Project Information Memoranda
RC	Regional Council
REG.	Regional
RIACT	Real-time Individual Asset Attribute Collection Tool
RNC	Resilience to Nature's Challenges Kia manawaroa – Ngā Ākina o Te Ao
	Turoa Statistics New Zealand
	Statistics New Zedianu
SUE	State-Owned Enterprise
	Surface water Flooding Model
(EX-)IC	(ex-) i ropical Cyclone
UK	United Kingdom
USA	United States of America
	United Nations
	United Nations Office for Disaster Risk Reduction (formerly UNISDR)
UNISDK	UNDRR)
VGI	Volunteered Geographic Information
VOT	Vehicle OverTurning Model
WMO	World Meteorological Organization
WWRP	World Weather Research Programme

This illustration will be provided at the beginning of each chapter to position the reader in the thesis.

	Chapter One		
Part 1	Introduction		
Fait I	Chapter Two		
	Background and Context		
	Chapter Three		
Part 2	Research Design		
T all 2	Chapter Four		
	Methods Results and Overarching Findings		
	Chapter Five		
	Volunteered Geographic Information for people-		
	centred severe weather early warning: A		
	literature review		
	Chapter Six		
	Identifying the data uses and gaps for severe		
	weather impact forecasts and warnings		
	Chapter Seven		
	Identifying data sources for severe weather		
Part 3	impact forecasts and warnings in Aotearoa New Zealand		
	Chapter Eight		
	A socio-technical analysis of sharing and		
	governing hydrometeorological impact data in		
	Aotearoa New Zealand		
	Chapter Nine		
	Integration of Partnerships and Collaboration to		
	Support IFWs and HIVE Data Collection, Access,		
	and Use		
Part 1	Chapter Ten		
rait 4	Discussion and Conclusion		

Chapter One: Introduction

The World Meteorological Organization (WMO) recommends that Impact Forecasts and Warnings (IFWs) are implemented to reduce the impacts from hydrometeorological hazards (WMO, 2015). Yet there is a lack of knowledge about where the supporting data are and how they can be connected (e.g., Potter et al., 2021). This thesis sets out to resolve this. This chapter sets the scene for this doctoral thesis by introducing the research problem. The chapter begins with a review of the researcher's role (Section 1.1) and the significance of the research in the context of hydrometeorological hazards and their impacts on society (Section 1.2), including the role of hydrometeorological warnings and the need for Impact Forecasting and Warning (IFW) systems (see a list of acronyms and abbreviations on page xiii). The research problem and aims are introduced in Section 1.3, along with the research question and objectives. A thesis outline is provided in Section 1.4, and a summary concludes this chapter in Section 1.5.

1.1 My Role as Researcher

My interest in studying severe weather and hydrometeorological hazards came from my experiences with severe weather as I grew up in Midhurst, Ontario, Canada. Midhurst is a small village outside of the City of Barrie, situated in central Ontario between the Great Lakes. This location sets Midhurst and Barrie up for a wide range of active weather in all four seasons. In the winter months, Midhurst falls in the 'snowbelt', where the lake-effect results in heavy snowfall on the leeward side of the Great Lakes (Goldfinger, 2020). Midhurst can receive heavy snowfall from the lakeeffect snow coming off Lake Huron and Georgian Bay, accumulating 10-25cm of snow in one event (Current Results, 2021). During my childhood, these snowfall events resulted in many 'snow-days', where school buses were cancelled due to hazardous road conditions and schools were closed due to extremely cold temperatures.

During the summer months, Midhurst and the surrounding area experiences many moderate to severe thunderstorms including tornadoes, dubbing Southern and Central Ontario the 'Tornado Alley' of Canada (Harrison et al., 2015). In 1985, before I was born, an F5¹ tornado struck the City of Barrie, damaging much of the downtown core, and unfortunately killing 16 people. During my childhood, many of my outdoor soccer games were cancelled due to lightning and the occasional tornado watch or warning. In one instance, a tornado warning was issued for my area when a funnel cloud was observed, and I sheltered with my parents, sisters, and pets in our basement. Fortunately, no tornado resulted from that storm. Recently, an EF2 tornado struck the southern end of Barrie on 15 July 2021, damaging several homes and resulting in minor injuries. Almost every summer a tornado threat presents itself

¹ The Fujita Scale (represented by the letter F) was used to classify tornado intensity based on damage prior to use of the Enhanced Fujita (EF) Scale, which phased out the Fujita Scale in Canada in 2013 (Government of Canada, 2018).

Chapter 1: Introduction

around the Barrie and Midhurst area, which led me to fear strong gusts of wind (so it is a good thing I chose to move to Wellington, New Zealand, for my PhD!).

My experiences with these severe weather hazards sparked a desire to study and understand them. When it came time to decide on a university program, I chose to pursue a bachelor's degree in Geography for the opportunity to study natural hazards. For my honour's thesis, I investigated the process of conducting post-storm damage surveys following potentially tornadic events in Canada to understand the strengths and barriers of the damage surveys (Harrison et al., 2015). I did this by interviewing officials from Environment Canada (the National Meteorological Service of Canada) and wind engineers who conduct these damage surveys. Towards the end of my thesis, a tornado struck the town of Angus, Ontario, just 20 km away from Barrie; I was invited by my interviewees to accompany them on the damage survey to observe their process. One key challenge identified in my honours thesis research was difficulty accessing damaged sites, collecting the data from remote locations, and collecting it quickly enough before clean-up-efforts began (Harrison et al., 2015). This led me to question how crowdsourcing and social media could potentially fill this data gap.

I then pursued my master's degree in Geography to explore the opportunities and challenges of using social media and crowdsourcing tools in the Canadian emergency management context. I interviewed emergency management officials across Canada to understand the role of social media and crowdsourcing in emergency management (Harrison & Johnson, 2016), and to identify the challenges of and barriers to using it (Harrison & Johnson, 2019). This research concluded with some practical suggestions for emergency management officials to ease their adoption of social media and crowdsourcing.

Upon completing my master's degree, I remained curious about severe weather and its impacts on society, how severe weather risks are communicated, and how social media and crowdsourcing have been used to aid response efforts in large-scale disasters like Hurricane Sandy in 2012, Hurricane Harvey 2017, and the Fort McMurray wildfire in 2017. I decided to pursue my PhD on the initial topic of exploring how social media and crowdsourcing can be used to collect impact data to inform Impact Forecasts and Warnings (IFW) for severe weather hazards in New Zealand. This is a type of warning recommended by the WMO with the aim of increasing understanding of the potential impacts of hydrometeorological hazards (see Sections 1.2.2 and 2.2).

After exploring the literature and conducting preliminary interviews with meteorologists, emergency managers, and other stakeholders both within and outside of New Zealand, I began to realise that social media, crowdsourcing, and impact data are one small part of the larger picture of the kind of data that we need for IFW systems and how to access it. I decided to shift the focus of my PhD topic from investigating social media and crowdsourcing for collecting impact data, to exploring the overall data needs of IFWs and how we can access this data. As I found out from talking to my participants, the data exists in many formats, it is just not stored in such a way that it can be easily accessed and used. This sparked my new, and final research

Chapter 1: Introduction

question that I aimed to answer in this doctoral thesis (discussed further in Section 1.3):

Which data are needed to support Impact Forecasting and Warning (IFW) systems for hydrometeorological hazards, and how are they currently collected, stored, and shared in New Zealand?

To begin answering this question, we first need to know why we want to implement IFWs. This justification will be summarised in the following sections, with further elaboration provided in Chapter Two.

Before we begin this journey, I would like to acknowledge New Zealand's response to the COVID-19 pandemic. In the last 18 months, I, and 'the team of 5 million', have witnessed the importance of communicating clear, understandable, relevant, and relatable information to the public to incite the desired responses. This is done, in part, by focusing on the impact on New Zealanders' daily lives, being empathetic, and providing clear and easy steps to protect each other (Hunt, 2021). There is much to be learned from this communication approach, which can inform communication of other risks and hazards, such as hydrometeorological hazards.

1.2 Severe Weather and Society

Weather influences everyday life, from deciding how to dress and how to commute to planning weekend activities. Severe weather conditions can also produce hazards that present a risk to people, property, and infrastructure. When these hazards present themselves, people are faced with more serious decisions to protect life and property.

Hydrometeorological hazards continue to cause significant impacts to society worldwide. A hydrometeorological event becomes a disaster when the hazard affects the lives, wealth, or livelihoods of people such that community functions are disrupted to the point where external help and support are needed for recovery (UNDRR, 2016).

Meteorological phenomena that have caused the above impacts include extreme temperatures such as cold waves, heat waves, and extreme winter conditions; and storms, such as tropical cyclones, convective storms, and extra-tropical cyclones (UNDRR, 2021a). These storm events produce further hazards including lightning and thunderstorms, tornadoes, winter storms/blizzards, hail, severe storms, derechos (wind storms), heavy rainfall, sand/dust storms, and storm surge (UNDRR, 2021a). Meteorological phenomena can also produce hydrological hazards, consisting of floods (e.g., flash floods, riverine floods, and coast floods), and landslides (debris flow, avalanche, mudslide, rockfall, and subsidence) (UNDRR, 2021a). Wildfires and droughts are additional meteorological/climatological hazards that continue to impact society (UNDRR, 2021a; WMO, 2021).

Chapter 1: Introduction

From 2000 to 2021, hydrometeorological hazards have affected over 24.9 million people and resulted in over 9.1 million deaths, and 33.2 million injuries (UNDRR, 2021a). Furthermore, these events have caused over 15.8 trillion USD in damages globally (UNDRR, 2021a). These impacts are attributed to the interaction of the hydrometeorological phenomenon with the underlying exposure and vulnerabilities of people, properties, and infrastructure (WMO, 2021). Hazards and disasters caused by this interaction have been well documented (e.g., Ching et al., 2015; Wagenmaker et al., 2011).

Prevention and mitigation of hydrometeorological-induced impacts, and thus disasters, may be achieved by reducing the hazard (e.g., reducing emissions to improve air quality by reducing NOx and O₃ (smog)/PM₁₀ particulates), reducing the exposure of the people and assets (e.g., land-use planning and/or evacuations), and/or reducing vulnerability (e.g., strengthening infrastructure; Golding et al., 2019). These actions require extensive knowledge of hazards, vulnerability, and exposure to inform policy decisions (Golding et al., 2019).

As well as policy changes for long-term mitigation, better systems are needed to improve immediate safety decisions for life and property. Such systems need to not only adequately inform individuals of the hazard, but also of the likely impact(s) and consequence(s), such that they can make appropriate decisions (WMO, 2015). Early Warning Systems (EWSs) are used worldwide to monitor, forecast, assess, and communicate hazard and risk information to enable individuals, communities, governments, businesses, and other stakeholders to take timely action to reduce the risks of disaster (UNDRR, 2021b). However, EWSs for hydrometeorological hazards continue to face critical challenges around the world, resulting in devastating impacts.

1.2.1 Severe Weather Warning System Challenges

Notable historic high-impact weather events exposed major communication gaps between meteorologists, warning services, and target audiences, resulting in widespread losses (Basher, 2006; Ching et al., 2015; WMO, 2015; Zhang et al., 2021). This communication gap is a result of both technical failings and human behaviour (Wagenmaker et al., 2011). Research has shown that meteorologists and warning services do not typically consider the warning audiences' current state of vulnerability and exposure at the time of the warning or at the expected time of impact (WMO, 2015). For example, a recent tragedy occurred on 22 May 2021 in Baiyin, Gansu Province, China where 21 ultramarathon race runners died due to exposure to high winds combined with low temperatures and precipitation (Zhang et al., 2021). The forecasts for this weather phenomenon were timely and accurate, yet the race continued (Zhang et al., 2021). Zhang et al. (2021) argued that while this weather phenomenon would not normally have been a high-impact weather event, the vulnerability of runners in shorts and vests to the low temperatures resulted in unfortunate deaths, making this a high-impact event. Thus, there is a need for warning systems to consider the current exposure and vulnerability of people at the time and location of the hazard, and to translate the hazard-based warning into an impact warning such that appropriate decisions can be made around protective actions (Zhang et al., 2021).

Furthermore, some warning recipients may fail to understand and respond to the warnings effectively due to ambiguous terminology (Ching et al., 2015), lack of trust in the warning system and service provider (Taylor et al., 2018), and warning fatigue (Mackie, 2013). As such, EWSs have continued to evolve in attempts to reduce the effects of these factors. The examples listed in Table 1.1 exemplify some of these factors such as warning fatigue (e.g., the Joplin tornado) and poor terminology and understanding (e.g., Typhoon Haiyan and the Uttarakhand floods).

Event	Warnings	Impacts	Causes	Recommendations
Joplin Missouri EF- 5 Tornado USA May 2011	Issued by National Weather Services in days and minutes lead-up.	159 deaths, over 1,000 injuries.	Warnings and sirens lacked sense of urgency; residents desensitised to tornado warnings in 'Tornado Alley' (e.g. warning fatigue) (Wagenmaker et al., 2011).	Tornado warnings include simplified and localised information to convey the severity of the storm (Wagenmaker et al., 2011).
Tropical Cyclone Haiyan (Yolanda) Category 5 Philippines November 2013	Issued by Philippine Atmospheric, Geophysical and Astronomical Services Administration (PAGASA) Heavy rain and wind warnings during peak rush hour.	6,201 deaths, 28,626 injuries, 1,785 missing people, Storm surge the leading cause of death.	88% of warning recipients did not understand message about 'storm surge'; 95% of warning recipients did not evacuate because they did not expect the storm to be so catastrophic (Ching et al., 2015).	"Warning messages should be conveyed in terms understood by the population at risk" (Ching et al., 2015, p. 34).
Tropical Cyclone Fitow China October 2013	Issued by China Meteorological Administration/Shanghai Meteorological Service to over 18 million people during peak rush hour.	97 roads and 900 communities were flooded, 1 death, 1.2 million people directly affected.	Public perception was that the warnings were too late; many people were in the middle of their morning commute on the first day back to school and work after a national holiday (WMO, 2015).	Meteorologists need to consider vulnerability and exposure of population at risk when issuing warnings (WMO, 2015).
Uttarakhand Floods India June 2013	Floods were accurately forecasted, and warnings issued by the Indian Meteorological Department.	5,700 deaths, over 900,000 people affected.	Majority of deaths were tourists on a pilgrimage; the wording in the warning messages was ambiguous, did not communicate the impacts of "very heavy rain" (WMO, 2015).	Meteorologists need to consider vulnerability and exposure of population at risk when issuing warnings, and convey the expected impacts in clear and understandable terms (WMO, 2015).

Table 1.1. Summary of notable, historic events in which some members of the public failed to understand and respond to warnings.

These examples demonstrate the communication gap that exists between meteorologists and other agencies that distribute warnings, and the public. While great advances have been made in improving the timing and accuracy of high impact weather warnings, these efforts are fruitless unless the target audiences understand the warnings and understand how best to respond or prepare (Paton & Johnston, 2006). It is argued that Impact Forecasting and Warning (IFW) Systems may help reduce the effects of these factors, and thus increase warning compliance (e.g., Morss et al., 2018; Otto et al., 2017; Potter et al., 2018; Weyrich et al., 2018).

1.2.2 Impact Forecasting and Warning Systems

Impact Forecasting and Warning (IFW) systems play a key role in resilience to hydrometeorological hazards (Rogers & Tsirkunov, 2013). Identifying the specific social, economic, and environmental impacts of a hazard allows communities to adjust their plans and actions to better adapt to, and cope with, the outcomes of the hazard (WMO, 2015). Rather than preparing for a hazard with no specific knowledge of how they could be affected, people can plan for impacts that are specific to their situation.

A key challenge with implementing IFWs is the lack of supporting data. IFW systems require more than hazard information as they introduce the human element to the warnings (Potter et al., 2021). Impact, vulnerability, and exposure knowledge are also needed to convey the risk accurately and effectively to warning audiences (Poolman, 2014; Sai et al., 2018). Table 1.2 provides the definitions of hazard, impact, vulnerability, and exposure data that are used for this thesis, while Chapter 6 provides further explanation and examples of these four data types. Meteorologists and forecasters have indicated that they do not possess knowledge of impacts, exposure, or vulnerability to effectively incorporate accurate impact information within their messages to suit all warning recipients (Potter et al., 2021).

Hazard	A hazard is "a hydrometeorological-based, geophysical or human-induced element that poses a level of threat to life, property or the environment", such as rainfall, snowfall, high winds, tornado, hail, flood (WMO, 2015, p. 4). In weather forecasting, likelihood is usually combined with hazard (WMO, 2015).
Impact	Impacts are the negative outcomes of an event (Casteel, 2016), and are defined as "a loss of life and injuries, damage to the environment, infrastructure, and private property, often followed by secondary effects like psychological trauma, or disruption of workflow and traffic" (Kox, Lüder, et al., 2018, p. 116). Direct and indirect impacts may be a function of vulnerability and exposure to the hazard (WMO, 2015). Impact data refers to observed post-event impacts and outputs and results from risk/impact models.
Vulnerability	Vulnerability is "the susceptibility of exposed elements, such as human beings and their livelihoods and property, to suffer adverse effects when affected by a hazard" (WMO, 2015, p. 6). Predispositions, sensitivities, fragilities, weaknesses, deficiencies, or lack of capacities that favour adverse effects on exposed elements can increase vulnerability (WMO, 2015). Vulnerability may also be time- and space-dependent.
Exposure	Exposure refers to people, infrastructure, housing, and other tangible human assets that may be affected in an area where hazards may occur (WMO, 2015). Exposure is time- and space-dependent, and it is possible to be exposed to a hazard but not vulnerable.

Table 1.2. The four types of data needed for IFWs.

Beyond IFWs, there is a growing need for effectively collecting and documenting hazard, impact, vulnerability, and exposure data to build more understanding of people's and other assets' risk to natural hazards (i.e., risk assessment) for Disaster Risk Reduction (DRR). This was identified as a priority of the Sendai Framework for Disaster Risk Reduction (UNDRR, 2015b). More details on the Sendai Framework are presented in Chapter Two, Section 2.3. In Aotearoa New

Zealand (NZ), Crawford et al. (2018) found that Civil Defence and Emergency Management (CDEM) agencies and city, district, and regional councils in NZ were not clear on who was responsible for collecting disaster risk-related data, such as impact, vulnerability, and exposure, and thus lacked the data needed for model-based risk assessments.

1.3 Research Problem and Aims

IFW systems are the focus of this study as the World Meteorological Organization (WMO) has encouraged and pushed for National Meteorological and Hydrological Services (NMHSs) to implement IFW systems. A key challenge to implementing an IFW system is the lack of knowledge and data around impact, vulnerability, and exposure from the perspective of the NMHSs (Potter et al., 2021). Knowledge of impacts, vulnerability, and exposure would inform the crafting of more effective warning messages, as called for in the Sendai Framework for Disaster Risk Reduction (UNDRR, 2015b) and the WMO Guidelines (WMO, 2015).

This research aims to support the implementation of an IFW system in Aotearoa New Zealand (NZ) by first exploring the data needs of IFWs and then identifying existing and potential data sources and the processes for collecting, creating, accessing, and using these data (i.e., hazard, impact, vulnerability, and exposure data). This study is based in NZ to support the country's efforts towards improving warnings for hydrometeorological hazards and implementing an IFW system as a membernation of the WMO. More of the NZ context is provided in Section 2.4.1. The NZ-focus provides a case analysis for international discussions and learnings around IFW system implementation.

This dissertation is written under the 'PhD thesis with publications' format in the Massey University Guidelines (Massey University, n.d.). This study aims to build an understanding of the data gaps and needs for implementing IFWs in a New Zealand context, and to offer avenues for accessing the required data to support IFW implementation.

This doctoral research thus seeks to answer the following research question:

Which data are needed to support Impact Forecasting and Warning (IFW) systems for hydrometeorological hazards, and how are they currently collected, stored, and shared in New Zealand?

This research question was broken down into five sub-questions and associated objectives in Table 1.3. These research questions and objectives are each addressed by a manuscript or chapter of this thesis. Each manuscript is presented as a thesis chapter. Table 1.3 illustrates the alignment of each manuscript or chapter to the research questions and objectives. Each manuscript can stand alone as an independent research article intended for publication. Due to this format, there will be some level of repetition of ideas, particularly in the introduction and research method sections of the manuscripts.

Research Questions		Objectives	Chapter
1.	What are the current and potential uses of Volunteered Geographic Information (VGI) for severe weather warnings?	1.1 To establish VGI as a potential source of impact data for IFWs.	Chapter Five
2.	What are the data uses and gaps for impact forecasts and warnings?	 2.1 Identify the actors involved in an IFW system and their associated roles. 2.2 Determine how hazard, impact, vulnerability and exposure (HIVE) data are used in an IFW system. 2.3 Identify further data gaps/needs for implementing IFWs. 	Chapter Six
3.	What are the sources for HIVE data?	 3.1 Identify the creators, collectors, and users of HIVE data that is relevant to severe weather IFWs. 3.2 Identify the inhibitors of and facilitators to collecting and using HIVE data to support the implementation of an IFW system in New Zealand for severe weather hazards. 	Chapter Seven
4.	How can HIVE data governance, access, and sharing be improved for hydrometeorological hazards in New Zealand?	 4.1 Identify and understand the governance and access and sharing processes of HIVE data for severe weather hazards in New Zealand. 4.2 To support efforts to fulfil the Sendai Framework priorities around disaster data access. 4.3 To support the implementation of a severe weather IFW system. 	Chapter Eight
5.	How can partnerships and collaboration better facilitate the collection, creation, and access to HIVE data for IFWs?	 5.1 To identify the required partnerships and collaborations required for IFW systems. 5.2 To identify existing partnerships and collaboration in NZ that can support IFW systems. 5.3 To outline a path forward for nurturing partnerships and collaboration for IFW systems. 	Chapter Nine

Table 1.3. Chapter and research alignment.

1.4 Thesis Outline

This thesis consists of ten chapters. Several of the chapters have been submitted as journal articles and are formatted as such in this thesis (see Appendix A for Statements of Contribution for each manuscript). In addition, this thesis contains an introductory chapter, a background and context chapter, a research design chapter, a methods results and overarching findings chapter, and finally a discussion and conclusion chapter.

Chapter One introduces the role of the researcher, and the research problem and justification. The intersection of weather and society is provided to 'set the scene' for the rest of this doctoral thesis.

Chapter Two provides a background on EWSs and IFW systems and the associated data needs for IFWs. Further background of the NZ hydrometeorological hazardscape, hydrometeorological warning system, and emergency management sector is provided for context.

Chapter Three presents the research design, which covers the overarching research philosophy and chosen methods used for primary data collection and analysis.

Chapter Four presents the initial results from the data collection methods, and overarching findings from the Grounded Theory analysis, where the key themes and phenomena of this doctoral study are identified.

Chapter Five, the first paper (i.e., journal article), presents a literature review to explore the role of Volunteered Geographic Information (VGI) for EWSs.

Chapter Six, the second paper, investigates the data uses and gaps for IFWs to determine how hazard, impact, vulnerability, and exposure (HIVE) data are used in an IFW system, and to identify gaps in the data that need to be addressed in order to effectively implement an IFW system.

Chapter Seven, the third paper, identifies existing and potential sources of HIVE data in New Zealand that can support IFW systems. Inhibitors and facilitators to collecting these data are also discussed.

Chapter Eight, the fourth paper, delves into issues of data governance and access for HIVE data and identifies opportunities for improving data governance and acquisition.

Chapter Nine integrates the results of each manuscript together around the core category that was identified in the Grounded Theory analysis.

Chapter Ten concludes the thesis by returning to the research questions and objectives of the research and identifying the significance, impacts, limitations of this doctoral research, and areas for future research.

1.5 Chapter Summary

This chapter outlined the research problem and the structure of this doctoral thesis research. It began by introducing the role of the researcher in this study, motivations for the broader body of work, and the research background. From there, the research problem describing the need for Impact Forecasts and Warnings and their supporting data was outlined. The research question was presented, along with objectives for answering the research question. The outline of the thesis was then presented with a summary of each chapter.

Position	of the	reader

	Chapter One
Part 1	
	Chapter Two
	Background and Context
	Chapter Three
Part 2	Research Design
	Chapter Four
	Methods Results and Overarching Findings
	Chapter Five
	Volunteered Geographic Information for people-
	centred severe weather early warning: A
	literature review
	Chapter Six
	Identifying the data uses and gaps for severe
	weather impact forecasts and warnings
	Chapter Seven
	Identifying data sources for severe weather
Part 3	impact forecasts and warnings in Aotearoa New Zealand
	Chapter Eight
	A socio-technical analysis of sharing and
	governing hydrometeorological impact data in
	Aotearoa New Zealand
	Chapter Nine
	Integration of Partnerships and Collaboration to
	Support IFWs and HIVE Data Collection, Access,
	and Use
Part 4	Chapter Ten
	Discussion and Conclusion

Chapter Two: Background and Context

This chapter provides an overview of severe weather EWSs (Section 2.1), including key concepts of warning thresholds and the inherent uncertainty of forecasting and warning for weather hazards. IFW Systems are then presented in Section 2.2 as a suggested approach by the WMO to address the aforementioned challenges of traditional severe weather EWSs. Section 2.3 provides an overview of the Sendai Framework and its relevance to this doctoral research. This chapter then provides contextual background on the NZ hydrometeorological hazardscape (Section 2.4.1), severe weather warning system (Section 2.4.2), flood warning system (Section 2.4.3), and Emergency Management (EM) sector (Section 2.4.4). The chapter concludes by bringing together the literature gap that this research aims to fill within the NZ context.

2.1 Early Warning Systems

There has been a paradigm shift in Disaster Risk Reduction (DRR) in the last two decades from reactive post-disaster response and recovery to proactive preparedness and mitigation. A key component of better preparedness lies in EWSs (UNDRR, 2015a). Starting in the late 1990s EWSs became the focus of international attention, generating international conferences and experts' symposia, and documents such as the United Nations Global EWS Survey Report (2006), and the World Disaster Report (2009) (Golnaraghi, 2012). In response to this apparent paradigm shift, the second priority of the Hyogo Framework for Action 2005-2015 was made to "identify, assess, and monitor disaster risks and enhance early warning" (UNISDR, 2005, p. 7). EWSs have continued to be a priority for the Sendai Framework for Disaster Risk Reduction 2015-2030 (UNDRR, 2015b).

EWSs have four key operational components: disaster risk knowledge; detection, monitoring, and warning services; communication and dissemination, and response capacity (Figure 2.1) (Basher, 2006). These components are further explained in the literature review within Chapter Five (published as Harrison et al., 2020).

It has been argued that poor linkages between these components have been major causes for EWS failures resulting in disasters (Garcia & Fearnley, 2012). As such, linking these four components into an integrated EWS is critical for their effective performance (Garcia & Fearnley, 2012). Doing so requires coordination and collaboration across agencies and governance levels that have relevant information about the hazards and impacts that threaten people and assets (Garcia & Fearnley, 2012; Golnaraghi, 2012). Such collaborative approaches allow for effective communication networks to be established for integrating research into practice (Garcia & Fearnley, 2012). This can enable the development of effective decision-making processes that include local contexts and define roles and responsibilities (Garcia & Fearnley, 2012).



Figure 2.1. Four operational components of an Early Warning System(Harrison et al., 2020 adapted from Basher, 2006; Golnaraghi, 2012; WMO, 2018).

2.1.1 Severe Weather Early Warning Systems

In the context of EWSs for hydrometeorological hazards, National Meteorological and Hydrological Services (NMHSs) are often responsible for monitoring, detecting, and forecasting the hazards (WMO, 2015). In many countries, hydrological services and meteorological services are provided by separate national or regional agencies (Golnaraghi, 2012). Simultaneously, Emergency Management (EM) agencies, or Civil Defence and Emergency Management (CDEM) Groups in NZ, are typically responsible for managing risks and emergency preparedness and response by conducting risk assessments and building mitigation, preparedness and response plans, etc. (Golnaraghi, 2012; WMO, 2015). These agencies and other sector-specific agencies, often monitor and document the impacts of hazards across sectors, such as health, agriculture, infrastructure, water resource management, and transportation (Golnaraghi, 2012).

Golding et al. (2019) outlined the Warning Value Chain concept for representing the information flow amongst actors in the warning chain for designing and operating hydrometeorological warnings. This Warning Value Chain provides a conceptual basis for understanding the flow of information from detection through to warning delivery. The Warning Value Chain consists of six components: (1) Observation, Monitoring, and Detection; (2) Weather Forecasting; (3) Hazard Forecasting; (4) Impact Forecasting; (5) Warning; and (6) Decision/Action. These components are summarised in Table 2.1, and a more detailed description and analysis of the Warning Value Chain is provided in Chapter Six (Harrison et al., 2022).

Table 2.1. Summary of Warning	Value	Chain	components	based	on	the	literature	review
provided in (Harrison et al., 2022).								

Warning Value Chain Component	Description
Observation, Monitoring, and Detection	Meteorological and hydrological services routinely monitor and observe atmospheric, meteorological, and hydrological conditions using specialised tools and collect regular observational data. Satellite and radar imagery, weather stations, and rivers gauges are some examples of monitoring and observational tools.
Weather Forecasting	Meteorological services issue frequent weather forecasts of varying time ranges by running Numerical Weather Prediction (NWP) models. Probabilistic forecasts have become the new standard and allow for longer lead times.
Hazard Forecasting	Hazard forecasting occurs when meteorologists and forecasters identify the potential for hazards based on observed and forecasted atmospheric, meteorological, and hydrological conditions. Floods and flash floods, storm surge, droughts, wildfires, snowfall, ice, extreme temperatures, and damaging winds are examples of weather-induced hazards.
Impact Forecasting	Impact forecasting involves identifying the impacts of the hazard, often by combining the hazard forecast with underlying vulnerability and exposure of people and assets.
Warning	Warnings are the end-product of a warning system. Warnings are intended to deliver information about the oncoming hazard such that warning audiences can prepare themselves.
Decision/Action	Recipients of warnings face many decisions with regards to preparing and taking protective action. A warning is considered successful when it is timely, accurate, and incites the appropriate protective actions by warning audiences.

Warning thresholds determine the sensitivity of an EWS and are used to assign warning levels for various hazards (Feinberg, 2009; Stepek et al., 2012). Thresholds are typically pre-defined (Cools et al., 2012), based on 'trigger points,' 'critical amounts,' or a set of 'criterion' by which a given hazard may occur (Frugis & Wasula, 2011; Tiranti & Rabuffetti, 2010). For example, thresholds for landslide warnings can be based on "critical rainfall amounts" (p. 472) that trigger landslides for certain soil types and slopes (Tiranti & Rabuffetti, 2010).

The sensitivity of an EWS can affect its performance (Feinberg, 2009). Thresholds that are too sensitive may result in over-warning while thresholds that are not sensitive enough may result in under-warning (Feinberg, 2009). The consequences of the imbalanced sensitivity can result in false alarms and warning fatigue, or inaction due to a lack of warnings (see Section 2.1.2 for more). Thus, defining warning thresholds, and the sensitivity of the EWS requires finding the right balance between scientific methods of forecasting and assigning thresholds, and the needs of the audiences for which the EWS is designed (Gaztelumendi et al., 2012).

Severe weather warning thresholds are typically based on the physical characteristics of the hazard(s) (WMO, 2015), such as 'x' amount of rainfall over 'y' amount of time. Thresholds can be defined by experiences of damage resulting from hazards of a measured severity (Frugis & Wasula, 2011; Stepek et al., 2012), and through modelling by incorporating multiple physical factors contributing to the hazard trigger point (e.g., Tiranti & Rabuffetti, 2010). National meteorological services (NMSs) typically define the warning thresholds (WMO, 2015), and may also seek consultation with other warning stakeholders to calibrate the thresholds/sensitivity (Frugis & Wasula, 2011). For example, the USA National Weather Service changed the criterion for severe hail based on empirical research and feedback from EM agencies (Frugis & Wasula, 2011).

The likelihood or probability of a hazard occurring at a certain place and time also factors into threshold definition (Gaztelumendi et al., 2012; Tiranti & Rabuffetti, 2010), and in how the forecasts and warnings are communicated (WMO, 2015). Thresholds can also be based on the risk of impact according to the underlying vulnerability and exposure of the people and infrastructure at the place and time of the expected hazard (Gaztelumendi et al., 2012; Potter et al., 2014). This approach is becoming more important with the advent of IFWs (further explained in Section 2.2). The level of uncertainty in a hazard occurring, its severity, and how that uncertainty is communicated to warning audiences, can further affect the warning performance.

Forecasting the weather is prone to uncertainty. Uncertainty in weather forecasts and warnings is introduced from the outset due to the chaotic nature of the atmosphere (National Research Council, 2006). Uncertainty increases when attempts are made to predict or forecast the extreme conditions of the atmospheric system that produce hydrometeorological hazards (Golding et al., 2019). Furthermore, uncertainty is higher in forecasts with longer lead times, and decreases respectively with lead time (Hirschberg et al., 2011). To complicate things further, smaller-scale events such as localised thunderstorms, tornadoes, and flash floods are more difficult to forecast than larger systems such as tropical cyclones due to the spatial resolution of the forecasting models (National Research Council, 2006).

Advancements in technology have allowed meteorologists and forecasters to greatly reduce and quantify uncertainty in their forecast and warning products (Golding et al., 2019). The meteorological forecasting field has shifted from deterministic forecasting to probabilistic forecasting, using ensemble prediction systems (GFDRR, 2016; WMO, 2015). Deterministic forecasts produce just one estimate or outcome of a forecasted phenomenon, while probabilistic forecasts produce a probability distribution of the forecasted hazard from across a range of potential futures (Chen & Yu, 2007). Ensemble prediction systems are one method of supporting probabilistic forecasts by producing multiple modelled outcomes which then provide relevant probability statements (Chen & Yu, 2007). As higher resolution ensemble NWPs become increasingly available, meteorologists can access the 'best available' probabilistic weather prediction guidance (GFDRR, 2016). With probabilistic forecasting, warnings can be issued earlier based upon low probabilities

that can increase or decrease as confidence increases (Mason & Senkbeil, 2015; Neal et al., 2014; Rogers & Tsirkunov, 2013).

While probabilistic forecasts are considered an effective tool in uncertain situations, challenges and limitations exist in communicating them. It has been suggested that disaster risk and emergency managers still prefer deterministic information to inform their decisions (WMO, 2017). Kox et al. (2015) found that emergency services tended to exclude low-probability forecasts from their decision-making. However, other work has shown that probabilistic forecasts have improved the decision-making for warning issuance. For example, Pak et al. (2007) concluded that the provision of probabilistic forecasts resulted in participants issuing wind advisories when the probabilistic information, they tended to issue too many advisories with low probabilities, and too few advisories with high probabilities. Similarly, LeClerc and Joslyn (2015) suggest, based on their findings, that adding probabilistic uncertainty estimates in forecasts can improve both compliance and decision quality.

This highlights the importance of improving the communication of likelihood, forecasts, or warning messages in uncertain contexts such that the information can be used effectively (Losee et al., 2017). When uncertainty is communicated in a way that is "compatible with both the decision task and cognitive processes of the user" (p. 308), uncertainty information can improve warning recipients' trust and expectations (Joslyn & LeClerc, 2013).

2.1.2 Early Warning Systems Performance

As highlighted in Chapter One, the performance of hydrometeorological EWSs is influenced by factors such as warning fatigue, cry-wolf syndrome, and perceived trustworthiness. Until recently, warning fatigue was dismissed by disaster researchers as a myth (Mileti, 1999; Mileti & Sorensen, 1990), whilst emergency management practitioners perceived it as a "very real problem" (Mackie, 2014, p. 1). Mackie (2014) argued that trust in warnings and warning services, helplessness, false alarms, overwarning, and scepticism in combination can affect warning fatigue of slow-onset bushfire hazards. While Mackie's research is specific to Australian slow-onset bushfire hazards, evidence shows that cry-wolf syndrome and warning fatigue may delay or prevent appropriate response actions to warnings for other events, such as tornadoes (Khan et al., 2017; Ripberger, Silva, Jenkins-Smith, Carlson, et al., 2015). This was found to be a contributing factor in inadequate responses to warnings in the case of the 2011 Joplin tornado described in Table 1.1. Thus, it was suggested that tornado warnings should include simplified and localised information to convey the severity of the storm (Wagenmaker et al., 2011).

Risk communication literature often brings up the concept of trust and how to build trust with the warning audiences (Losee & Joslyn, 2018; Poortinga & Pidgeon, 2003; Sheppard et al., 2012). The broad literature body on trust in DRR, risk communication, and warnings highlights factors that influence trust, which then influences risk perceptions and protective behaviours (Losee & Joslyn, 2018; Poortinga & Pidgeon, 2003). The level of trust is influenced by false alarms and over warning, lack of dialogue between the warning services and the warning recipients, and inconsistent messages and terminology (Losee & Joslyn, 2018; Ripberger, Silva, Jenkins-Smith, Carlson, et al., 2015).

Another contributing factor to the communication gap of EWSs is the terminology used in warning messages and difficulty understanding that terminology. A key factor to an effective EWS is providing warnings in clear, concise, easy-tounderstand language (International Network for Multi-Hazard Early Warning Systems, 2017). Paton and Johnston (2006) and Garcia and Fearnley (2012) argue that target audiences must be able to understand the warnings and prepare accordingly, or else the EWS is futile. The cases of Typhoon Haiyan (Yolanda) and the floods in Uttarakhand, India as shown in Table 1.1 highlight the lack of understanding amongst warning recipients of the terminology used in the warnings. In the case of Typhoon Haiyan, a post-event assessment determined that the people in areas at high risk of storm surge hazards who received the warnings did not understand the terminology of the warning, namely the term 'storm surge' (Ching et al., 2015). Supporting literature indicates that many people living in areas where tropical cyclones frequently occur are not aware of, or do not fully understand the risk of storm surge (Morss et al., 2018; Morss, Demuth, et al., 2016; Morss, Mulder, et al., 2016). This may be due to misunderstanding the terminology, as found by Ching et al. (2015), and/or the forecasters and warning services may be unaware of the lack of understanding of technical hazard-related terms. IFWs have been suggested to improve these communication challenges.

2.2 Impact Forecasts and Warning Systems

In the hydrometeorological hazards space, IFW systems are an advancement of traditional EWSs that use thresholds based on physical characteristics of the hazards (i.e., hazard-based thresholds) such as the Saffir-Simpson Hurricane Scale used in the USA to categorise tropical systems based on windspeeds. This scale was designed to predict the expected intensity of wind damage to structures but falls short in accounting for other tropical storm hazards, such as storm surge and impacts (Kantha, 2006).

By nature, these hazard-based warning systems overlook the dynamic exposure and vulnerability of warning recipients (WMO, 2015) such as the low elevation of a coastal area and the capacities of coastal residents to evacuate, as observed in the landfall of Hurricane Katrina (Eisenman et al., 2007). The IFW system was suggested by the WMO to bring in knowledge and understanding of exposure, vulnerability, and historic impacts and build new warning thresholds that better align with the position, needs, and capabilities of target audiences.
The terminology of IFWs varies across the literature. In 2015, the WMO published "Guidelines on Multi-Hazard Impact-Based Forecast and Warning Services" (see WMO, 2015) that provided definitions of 'impact warnings', 'impact-based warnings', exposure, vulnerability, and risk. Additionally, Kaltenberger et al. (2020) proposed 'impact-oriented warnings' to refer to warnings that contain impact information and clear protective action advice. The four types of warnings (hazard-based, impact-oriented, impact-based, and impact warning) are summarised in Table 2.2.

Warning Type	Definition	Example	Data Considered
Hazard- based warning	These are traditional warnings, with thresholds based on the physical characteristics of the hazard (WMO, 2015).	"Bora winds are expected tonight with wind speeds of 20 metres per second" (WMO, 2015, p. 6).	Hazard only
Impact- oriented warning	These warnings are independent from the criteria of a warning, which may be hazard-based or impact-based and contain information on impacts and clear protective action advice (Kaltenberger et al., 2020).	"Power and phone distribution networks may be disrupted for relatively long periods. Roofs and chimneys can be damaged" (Météo-France via Kaltenberger et al., 2020, p. 30).	Any
Impact- based warning	These warnings include hazard and vulnerability and are "designed to express the expected impacts as a result of the weather" (WMO, 2015, p. 6).	"Bora winds are expected tonight which may result in delays or cancellation to ferry services" (WMO, 2015, p. 6).	Hazard and vulnerability
Impact warning	These warnings include hazard, vulnerability, and exposure, and are designed to provide detailed messages down to the individual, activity, or community level (WMO, 2015).	"Ferry services for the island of Brač will most likely be cancelled tonight due to Bora winds" (WMO, 2015, p. 6).	Hazard, vulnerability, and exposure

Table 2.2. Definitions and examples of hazard-based warnings, impact-oriented warnings, impact-based warnings, and impact warnings (Chapter Six; Harrison et al., Under Review).

'Impact-oriented warnings', 'impact-based warnings' and 'impact warnings' broadly differ from traditional hazard-based warnings by communicating what the hazard(s) will do, in addition to what they will be (WMO, 2015). Traditionally, warnings have relied solely on weather-based factors (e.g. wind speeds of a minimum velocity, snowfall of a minimum depth), but IFWs also consider the timing and location of the hazards to identify vulnerable populations and infrastructure (WMO, 2015).

Amongst the three types of impact warnings, 'impact-oriented warnings' may still be based on hazard thresholds as with hazard-based warnings, but include a tangible and understandable description of an expected damage scenario (i.e., impact information) and clear advice on what to do (i.e., instructions/advisories) (Kaltenberger et al., 2020). Thus, impact-oriented warnings are independent of the production process (i.e., the warning thresholds or criteria) used to issue warnings, which may be hazard-based or impact-based. As such, the term impact-oriented warnings may encapsulate impact-based and impact warnings. Impact-based warnings and impact warnings differ based on the production process used to issue them.

For 'impact-based warnings', vulnerability is added to the formula to provide slightly more specific impact information (WMO, 2015). Impact-based warnings are triggered by the likely impacts and their severity from a hazard, and describe the likely impacts based on hazard and vulnerability information (WMO, 2015). 'Impact warnings' are more detailed, and communicate impacts on the individual, activity, or community level (Potter et al., 2018; WMO, 2015). Impact warnings require information on local exposure, which is more difficult to obtain (WMO, 2015).

The goal of the WMO Guidelines is to evolve severe weather warning services into the final 'Impact Forecasting and Warning' form, where vulnerability and exposure are integrated into the thresholds. Herein, I refer to the warning system for which this thesis is intended to support as an IFW system to align with the objectives of the WMO. I use the term 'impact-oriented warnings' to refer to other warnings when it is unclear whether they are impact-based warnings or impact warnings.

2.2.1 Impact Forecasting and Warning System Decision-Making and Communication Processes and Tools

An IFW system requires several streams of information and data to be considered in the decision-making process, beyond hydrometeorological data. Managing all of these data streams that often come from different sources, and formulating effective and efficient decisions, is a key challenge with implementing IFWs (Harrison et al., 2014; Kox & Lüder, 2021). As such, several tools and processes have been discussed in the literature for their potential to support this decisionmaking process within an IFW system. These tools and processes can broadly be categorised into subjective and objective approaches (WMO, 2015).

Subjective approaches to decision-making typically involve discussions with partners and stakeholders and tools to support those discussions (WMO, 2015). A risk matrix is an example of a tool that can support the decision-making for determining the level of an impact warning. It is a visual tool for estimating risk by combining likelihood and level of impact or severity of the hazard (Hofmann & Schüttrumpf, 2019; Neal et al., 2014). This approach allows for early expressions of likely impacts to be communicated consistently, and for the expressions to progress based on changing expectations of risk (Neal et al., 2014). For example, in the days leading up to an event, the likely impacts are not as well known (i.e., the exact location and timing are uncertain). Thus, early expressions may communicate lower impacts and likelihood (i.e., green or yellow), and as confidence in the likelihood increases as the event nears, the messages can be changed to match the more certain impacts and likelihood (Neal et al., 2014).

Risk matrices have some shortcomings that can limit their effectiveness. Their low resolution (i.e., the small number of colours and rating levels) can result in assigning identical risk ratings to different risks (Cox, 2008, p. 506). For example, a low impact-medium likelihood event may be assigned the same level of risk as a medium impact-low likelihood event using the risk matrix presented by Neal et al. (2014). This can potentially lead audiences to take inadequate protective actions against medium impacts or take overly protective actions against low impacts. Furthermore, the risk matrix is usually built on predefined impact thresholds resulting from a hazard (Neal et al., 2014; Sai et al., 2018), and thus does not include dynamic exposure and vulnerability and may not apply to all audiences (WMO, 2015). It is important to account for dynamic exposure and vulnerability and to design the matrices to suit various audiences to convey the level of risk more accurately. For example, different parts of a country might have different impact thresholds to the same hazard due to underlying vulnerability and exposure (Neal et al., 2014). It is important to consider these underlying conditions and ensure that the risk matrix reflects these different thresholds. Discussions with partners and stakeholders can inform the design of risk matrices and the overall design of an IFW system, to account for these variances in exposure, vulnerability, and warning audiences (WMO, 2015).

As previously described in Section 2.1, interagency communication is critical for integrated EWSs and decision-making in disaster response (Doyle & Paton, 2017; Garcia & Fearnley, 2012; Golnaraghi, 2012). Agencies involved in disaster response require both access to scientific data and a sufficient understanding of these data for rapid decision-making (Paton et al., 1999). Inter-agency networks have proven to fulfil this need (Paton et al., 1999). For example, during the 2012 Tongariro volcanic eruption crisis in NZ, interagency communication was essential for supporting a coordinated response to, and communication process of, the associated hazards and impacts (Leonard et al., 2014). Furthermore, the event revealed a critical communication gap between the health sector and the response and science sectors and provided an opportunity to bridge this gap (Leonard et al., 2014). Learning from events such as this indicate that interagency communication and collaboration supported through developing partnerships are key for hydrometeorological IFW systems such that knowledge of impacts, vulnerability, and exposure can be shared across agencies to inform the warning thresholds, warning messages, and responses (Hemingway & Gunawan, 2018; Kox, Kempf, et al., 2018).

Examples of agencies that possess the information and knowledge needed for IFWs and thus needed to build partnerships and collaborations for IFWs were consolidated from the literature and are summarised in Table 2.3. As mentioned earlier, when meteorologists create forecasts and warnings, they may not possess knowledge of the vulnerable or exposed populations and infrastructure (Hemingway & Gunawan, 2018; Kox, Kempf, et al., 2018). However, insurance/re-insurance companies often systematically collect this information, and EM agencies often conduct hazard and risk assessments to identify areas at risk or exposed to different hazards (Crawford et al., 2018). Working with local communities and economic sectors can help identify the appropriate impact thresholds for the various populations and

locations (Neal et al., 2014; Sai et al., 2018). Governments and NGOs can also assist in conducting vulnerability and exposure assessments, as well as identify potential impacts and adequate mitigation actions (GFDRR, 2016). Scientific institutions can help develop and improve observation, forecasting, and modelling tools and processes, as well as share appropriate data sources/datasets (Leonard et al., 2014).

	·
Stakeholders	Information/data sharing
Meteorological, Hydrological, and/or hydrometeorological services	Meteorological and hydrological hazard forecasts and observations
Governments, EM agencies (agencies from all levels that are directly responsible for the safety and security of populations)	Evaluate vulnerabilities, identify potential impacts and proper mitigation actions, share expertise
Scientific institutions	Improving and developing technical processes/equipment (e.g., modelling), sharing data sources/datasets, share expertise
Local communities	Identify thresholds, vulnerability, and exposure
Economic sectors (Agriculture, energy, health, transportation, natural resources, tourism, etc.)	Identify thresholds, vulnerability, and exposure
Insurance/Re-insurance	Vulnerability of physical infrastructure
Media	Identify potential impacts, communicate information to and from public

Table 2.3. Examples of sectors and agencies needed for building partnerships to develop IFW systems. Created by the author based on the presented literature review.

Quantitative approaches to assessing risk and supporting decision-making for IFWs are also available. Risk modelling is an example of a quantitative tool for assessing risk and associated impacts with hazard intensity. Risk modelling uses fragility functions and vulnerability curves to associate hazard intensity with levels of damage and losses (Martins & Silva, 2020; Schmidt et al., 2011). Fragility curves and vulnerability functions combine hazard intensity with vulnerability (determined by the characteristics of assets such as buildings, road networks, vehicles, etc.) to determine degrees of damage or impact indicators (Martins & Silva, 2020; Schmidt et al., 2011). RiskScape is an example of software under development in NZ to support multi-hazard risk modelling, that is, the

mitigation actions

Non-governmental organisations

(NGOs)

Vulnerability and exposure assessments,

quantitative estimation of the spatial distributions of potential losses for an area ... multiple ... natural hazards, multiple ... event probabilities ..., multiple ... human assets, and multiple potential loss components (for each of the assets, e.g. buildings, streets, people, etc.) (Schmidt et al., 2011, p. 1170).

RiskScape operates on a modular framework, where inputs are exchangeable under three modules: the hazard module, the asset module, and the vulnerability module (Schmidt et al., 2011). The hazard module consists of hazard specifications such as hazard type, the hazard model (such as a hydraulic inundation model), and hazard exposures (such as flood inundation depth and velocity), typically in the form of a geospatial layer (Schmidt et al., 2011).

Inputs for the asset module consist of asset types such as buildings, trees, people; asset attributes such as building height, demographics; and asset attribute types (Schmidt et al., 2011). These are all contained in one asset dataset, which can also be geospatial (Schmidt et al., 2011).

The vulnerability module uses a fragility curve or vulnerability function to relate the hazard exposure from the hazard module and the asset data to the resulting potential damage (Schmidt et al., 2011). The output of a model like RiskScape is the quantification of affected assets and losses, which can be provided in tabular and/or geospatial forms for visualisation (Deligne et al., 2017; Schmidt et al., 2011).

Visualisations aid the communication of hydrometeorological warnings, including IFWs. Apart from risk modelling, maps can also be created to convey exposure and vulnerability. Exposure maps are created by overlaying spatial asset layers (e.g., building footprints) with hazard layers (e.g., flood inundation levels), as was done by Bell et al. (2016) and Paulik et al. (2020) to map exposure to floods and sea level rise in NZ. Vulnerability maps can also be created using frameworks to develop social vulnerability indicators using demographic data (e.g., Census population data; Mason et al., 2021). These indicators can then be visualised on maps for further context and planning, as was done by Mason et al. (2021) for Porirua City in New Zealand.

Maps are also the preferred visualisation method using colours to represent warning levels over defined geographic areas (GFDRR, 2016). Amongst others, the UK MetOffice; MeteoAlarm in Europe (Neal et al., 2014); Meteo Shanghai and the Meteorological Administration of Guangdong Province in China; the Philippine Atmospheric, Geophysical and Astronomical Services Administration in the Philippines (Otto et al., 2017); and Instant Weather in Ontario, Canada use this approach to communicate IFWs on maps. These maps can easily communicate impacts to the public, as well as sector-specific impacts to stakeholders (GFDRR, 2016; WMO, 2015)..

2.2.2 Evaluation of Impact Forecasts and Warnings

Warning evaluation has a large body of research that has ultimately led the suggestions for communicating hazards *and* their impacts, as demonstrated by the Joplin Tornado and Typhoon Haiyan examples previously presented in Table 1.1. Since the introduction of IFW systems in the past decade, much of the research has thus focused on evaluating and assessing the performance, effectiveness of, and public response and perceptions to the new warning messages.

Research shows that the public may deem impact-oriented warnings or their sources as less credible. In the USA, Perreault and colleagues (2014) found that 'scary messages' were perceived as less credible than traditional warnings. Similarly, Morss, Demuth, et al. (2016) found that their participants felt that extreme impact messages were 'overblown' and thus rated the source as less reliable. In New Zealand, Potter et al. (2018) found no significant difference in the perceived credibility of impactoriented warning messages and phenomenon-based messages. The impact-oriented messages increased recipients' understanding of the impacts but did not increase intention to take protective action. In Europe, survey results indicate that the perceived credibility of impact-based warnings that included prescribed behavioural responses or protective actions was higher than that of either impact-based warnings on their own, or simple hazard-based warnings (Weyrich et al., 2018). In a post-event survey following Storm Doris in the UK during which impact-based warnings were issued by the MetOffice, results showed that trust and perceived trustworthiness in the forecasts and warnings increased as the understandability of the warning message increased (Taylor et al., 2019). Findings from most of these studies suggest that when formulating impact-oriented warnings, the language used to communicate the impacts must be carefully considered to ensure accurate understanding of the message and incite desired action, and to avoid or prevent undesired action from being taken by warning audiences. Including protective action guidance in the message in understandable terms can improve warning perception and understanding as well as intended behavioural responses (Weyrich et al., 2018).

Assessments of public response to and perceptions of impact-oriented warnings showed mixed results, and the IFW system seems to work better for some hazards than for others. An overview of these results is presented in the literature review of Chapter Six.

2.2.3 Benefits and Challenges of Implementing Impact Forecasting and Warning Systems

Implementing IFW systems can be costly and difficult, as they require a system overhaul starting with warning thresholds and changing or adding roles and responsibilities (Potter et al., 2021). Existing research around the perceived benefits and challenges of IFWs from the operational perspective is summarised next.

Forecasters and practitioners in Europe perceive the following added value to introducing an impact-focus for communicating hydrometeorological hazards, for building situation awareness, and for planning mitigation and response efforts (Kox, Kempf, Lüder, Hagedorn, & Gerhold, 2018; Kox, Lüder, et al., 2018; Terti et al., 2019). In Germany, emergency managers (EMs), responders, forecasters, and infrastructure managers indicated that receiving IFWs would allow them to better prepare for and adapt to severe weather hazards (Kox, Kempf, et al., 2018; Kox, Lüder, et al., 2018). Furthermore, in a role-playing game study with researchers and practitioners involved in weather communication in Finland and France, Terti et al. (2019) reported that the participants found the additional impact-based information helpful for building situation awareness and thus increased their confidence in decision-making (Terti et al., 2019). However, Terti et al. (2019) further noted that in-game, practitioners use the impact-based information less than appeared to traditional hydrometeorological data. Terti et al. (2019) speculated that this was due to unfamiliarity with the new information, limited understanding of how to use the new impact information, lack of trust in the new developments, and incapability of the new visualisations to convey useful information.

Verification is a standard practice that allows for warning services to measure the quality of their warnings for future improvement. The verification process of traditional hazard-based warnings typically involves measuring the timeliness (e.g., lead time) and accuracy (e.g., hits, misses, false alarms) of the forecast or warning (Wilson & Giles, 2013). These measurements are typically based on observations and reports (or lack thereof) of the physical hazard that was forecasted or warned for. With the advent of IFW systems, there is now a need to verify how well the forecasted or warned event related to the impacts felt on the ground (Robbins & Titley, 2018). As such, forecasters must now develop new verification methodologies to account for the observed impacts in addition to the observed hazard (Robbins & Titley, 2018). This has introduced new challenges. For example, Hemingway and Robbins (2019) guestioned how to verify forecasted impacts when impacts stop happening as a result of IFWs. In a NZ-based study involving meteorologists and hydrologists, the participants echoed this concern around verifying warnings and suggested verifying warnings using "dynamic, predetermined hazard-based criteria, even if no impacts occur" (Potter et al., 2021, p. 310).

IFW systems require more than hydrometeorological information. Forecasters and warning services need additional information around historic and potential impacts, and current vulnerability and exposure (Kox, Lüder, et al., 2018; Obermeier & Anderson, 2014; Potter et al., 2021). In developing an impact-based warning system for typhoon gales for the Chinese Meteorological Agency, Wei et al. (2018) noted that accessing sufficient information on hazards, exposure, and vulnerability, and sharing information across disciplines are major challenges to current and future application. Similarly, in developing an impact-forecasting model for vehicles overturning due to high winds in the UK, Hemingway and Robbins (2019) questioned how sufficient and appropriate impact data could be collected. Potter et al. (2021) confirmed these concerns amongst a global sample of meteorological service officials, in addition to NZ-based meteorologists and hydrologists. The findings of Potter et al. (2021) further support the need for collecting and storing impact-related data for hydrometeorological events and point to the need for research into identifying sources of this data.

Researchers and practitioners have indicated the need for more understanding around the decision-making process behind issuing IFWs, and selecting the appropriate warning thresholds for the appropriate audiences (Harrison et al., 2014; Obermeier & Anderson, 2014). This decision-making process requires input from various stakeholders with knowledge from their specialised areas (Harrison et al., 2014; Kox, Kempf, et al., 2018; Potter et al., 2021). For example, forecasters' expertise in weather and atmospheric systems remains foundational to the system (Obermeier & Anderson, 2014), yet linking this information to impacts requires integrating the knowledge and experience of local EM practitioners and responders with the forecasters' expertise (Kox, Kempf, et al., 2018). Furthermore, community involvement and local knowledge of past impacts can help determine impact thresholds for various groups (Kox, Kempf, et al., 2018; Sai et al., 2018).

2.2.4 Crowdsourcing, Social Media, and Volunteered Geographic Information

Crowdsourcing is a potential approach to accessing data for IFW systems. Crowdsourcing has been used by some NMHSs to detect weather hazards and to monitor them (Harrison & Johnson, 2016; Krennert, Kaltenberger, et al., 2018; Krennert, Pistotnik, et al., 2018). The WMO has recognised the need to explore the use of crowdsourcing for severe weather, and researchers in the weather warning space have alluded to its potential for collecting impact-related data (Henriksen et al., 2018; Kox, Kempf, et al., 2018; Marchezini et al., 2018; WMO, 2017).

Crowdsourcing refers to "volunteer-generated decentralised information" contributed online, or by telephone and mail (Harrison & Johnson, 2016, p. 17). Social media, that is, online blogs, micro-blogs, online social networking, and forums that allow sharing of text, audio, photographs, and videos (Alexander, 2014), enables both active and passive crowdsourcing and facilitates multidirectional communication amongst various actors and stakeholders (Alexander, 2014; Harrison & Johnson, 2016; Hong et al., 2018). Active crowdsourcing refers to people proactively contributing to a call for information or tasks through a specialised platform (Loukis &

Charalabidis, 2015). Alternatively, passive crowdsourcing refers to the collection of volunteer-generated information without an active call for contributions; the users may be unaware that they are contributing (Harrison & Johnson, 2016; Loukis & Charalabidis, 2015). Social media platforms are the most accessible resources for passive crowdsourcing from which the information generated by users is harvested by an entity (Loukis & Charalabidis, 2015).

The WMO identified crowdsourcing techniques as an area in which expertise and knowledge is needed for IFW systems (WMO, 2015). In support of this, Krennert, Kaltenberger, et al. (2018) and Krennert, Pistotnik, et al. (2018) surveyed European NMHSs to understand the current state of crowdsourcing practices for severe weather in Europe. Findings showed that nearly all European NMHSs have used crowdsourcing in some way, for updating severe weather warnings in real-time, for verification, and for establishing a climatology of severe weather events (Krennert, Pistotnik, et al., 2018).

Crowdsourcing and social media have the potential to become modern tools for supporting the Sendai Framework objective of making EWSs more people-centred (UNDRR, 2015b). Meissen and Fuchs-Kittowski (2014a, 2014b) developed a conceptual framework which demonstrated that crowdsourced data can theoretically be fully integrated into an existing EWS for severe weather and flood hazards as another dataset to augment or enhance the existing data collection methods. This could be taken a step further by incorporating observed impacts in the data and augmenting warnings with that impact data (Meissen & Fuchs-Kittowski, 2014a, 2014b). Furthermore, Marchezini et al. (2018) completed a literature review to determine the elements involved in bridging crowdsourcing with EWSs and found that more research is needed to examine how crowdsourcing can be mainstreamed in EWSs.

Some agencies have started collecting crowdsourced or social media data for event detection, and some of these agencies also keep track of the impacts recorded. The British Geological Survey began collecting landslide impact data from Twitter and uses text description, photos, and video footage of the resulting impacts (Pennington et al., 2015). This data is integrated into the National Landslide Database and is used to create a Hazard Impact Model as part of the UK's Natural Hazards Partnership (Pennington et al., 2015). In Canada, the National Meteorological Service (NMS) detects weather events such as tornadoes on Twitter and verifies and updates current weather watches and warnings based on verified Twitter posts (Harrison & Johnson, 2016). This is done by using reports posted on Twitter from one location to alert populations further down the storm track to the severe weather that is heading their way (Harrison & Johnson, 2016). These reports are also used to evaluate the service's performance (Harrison & Johnson, 2016). Volunteered Geographic Information (VGI) offers potential as a collaborative process for collecting vulnerability and exposure data and for creating impact-based thresholds (Fdez-Arroyabe et al., 2018; Harrison et al., 2020; Lavers & Charlesworth, 2018; Sharma et al., 2016). Broadly, VGI is crowdsourced geographic information, or information contributed by volunteers with geographic information associated with it (Harrison et al., 2020). In Australia, Haworth et al. (2018) found that VGI offers a contextual, social aspect of DRR for bushfire hazards. VGI brings members of the public into the DRR process and offers the opportunity for them to share their knowledge about community assets, exposure, vulnerability, impacts, and even 'safer' places for evacuation or sheltering in their local areas (Haworth et al., 2012).

These virtues of VGI make it a sound model for crowdsourcing impact-related data for IFWs as it has the potential to collect the required data whilst also supporting the shift to people-centred EWSs (Marchezini et al., 2018). It is important to collect and create geographically referenced data because disasters are inherently location- and time-dependent: if humans did not reside close to a natural hazard at the time of occurrence, there would be no disaster. Furthermore, visually displaying the impact information for IFWs on maps requires geographic information, thus it is beneficial to begin with collecting geographic information.

A literature review paper was published in the *Australasian Journal of Disaster and Trauma Studies* (Harrison et al., 2020) as part of this PhD study, investigating the role of VGI in severe weather mitigation and preparedness. This is presented in Chapter Five.

2.3 Sendai Framework for Disaster Risk Reduction

The Sendai Framework is built upon decades of international efforts for disaster risk reduction, starting in the 1990s when the UN set-up the International Decade for Natural Disaster Reduction to promote international collaboration towards reducing the effects of disasters (UN Secretary General, 1989). At the end of this decade, the Hyogo Framework for Action 2005-2015 was adopted at the World Conference on Disaster Reduction to further strengthen international cooperation for DRR (UNISDR, 2005). The Sendai Framework for Disaster Risk Reduction 2015-2030 (herein referred to as the Sendai Framework) is the successor of the Hyogo Framework and was developed based on learnings from the Hyogo Framework (UNDRR, 2015b).

The Sendai Framework for Disaster Risk Reduction 2015-2030 is an international agreement between the Member States of the United Nations (UN) to achieve

the substantial reduction of disaster risk and losses in lives, livelihoods and health and in the economic, physical, social, cultural and environmental assets of persons, businesses, communities and countries (UNDRR, 2015b, p. 12).

The Sendai Framework is the result of stakeholder consultations and intergovernmental negotiations supported by the UN Office for Disaster Risk Reduction (UNDRR) as requested by the UN General Assembly (UNDRR, 2015b).

The Sendai Framework provides the Member States with measurable actions to achieve the intended outcome cited above. These actions are grouped into four priority areas: (1) Understanding disaster risk; (2) Strengthening disaster risk governance to manage disaster risk; (3) Investing in disaster risk reduction for resilience; and (4) Enhancing disaster preparedness for effective response and to 'Build Back Better' in recovery, rehabilitation, and reconstruction (UNDRR, 2015b). The priorities most relevant to this doctoral research are (1) and (4). This section will focus on these two priorities and their relevance to this research.

The first priority of the Sendai Framework stipulates the need for policies and practices for disaster risk management to be based on understanding all dimensions of disaster risk, including vulnerability, capacity, exposure, hazard characteristics, and the environment (UNDRR, 2015b). This knowledge can be used for risk assessments, prevention and mitigation, and for developing and implementing appropriate preparedness and effective disaster response (UNDRR, 2015b). Actions listed to fulfil this priority centre around collecting, analysing, managing, using, and sharing relevant data, including hazard, impacts/loss, vulnerability, and exposure data (UNDRR, 2015b).

The fourth priority of the Sendai Framework recognises the need for further strengthening of disaster preparedness, taking anticipatory action for events, and integrating DRR with response and preparedness (UNDRR, 2015b). This involves, in part, "investing in, developing, maintaining and strengthening people-centred multi-hazard, multi-sectoral forecasting and Early Warning Systems" (UNDRR, 2015b, p. 21), which are tailored to the needs of the users. IFWs contribute to both of these priorities based on addressing warning audiences' needs for relevant, clear, understandable, and actionable information, and on the basis that IFWs require and make use of the data identified in the Sendai Framework for understanding risk (Harrowsmith et al., 2020).

The Sendai Framework recognises that the State has the primary role for reducing disaster risk and further suggests that responsibilities should be shared with local government, the private sector, and other stakeholders (UNDRR, 2015b). As a Member State, New Zealand is responsible for implementing the Sendai Framework, working towards the priorities and targets, and reporting its progress to the UNDRR (see UNDRR, 2020). To fulfil these tasks, the NZ Government, through the National Emergency Management Agency (NEMA), developed the National Disaster Resilience Strategy 2019 (herein referred to as the Resilience Strategy) (NEMA, 2019). The priorities and activities of the Sendai Framework are mirrored in the Resilience Strategy, particularly for understanding and managing risk, and continuing to increase the availability of and access to multi-hazard EWSs (NEMA, 2019). However, following the nation's 2019 Sendai Framework reporting, key challenges and needs were identified that require more work and attention (UNDRR, 2020). These challenges are

around increasing the availability of information and assessments to people and planners across government and improving the availability and quality of information (UNDRR, 2020). As such, there is a primary need in NZ to

> establish a comprehensive Disaster Information Management System [DIMS], combining various existing platforms and databases to harmonize and synergize available information in terms of loss data (for estimating disaster trends), and to support decisionmaking, prioritization and funding needs projection. This should consider the dimensions of potential climate change impacts as well and be made publicly available to stakeholders and communities. Current, existing systems are maintained by a plethora of actors, and a comprehensive DIMS remains to be established (UNDRR, 2020, p. 28).

This doctoral research is thus focused on New Zealand to support the country's efforts in fulfilling the Sendai Framework priorities by supporting the implementation of hydrometeorological IFWs and by developing more understanding around the data challenges both for IFWs and for general DRR in New Zealand.

2.4 New Zealand Weather Warning System

It is argued that National Meteorological and Hydrological Services (NMHSs) do not carry the full responsibility of obtaining the required data and of fully implementing an IFW system (WMO, 2015). Instead, it is a joint effort between the NMHSs and other stakeholders, particularly EM agencies (WMO, 2015). This varies country by country, as many countries have defined the roles and responsibilities for issuing warnings for different hazards through policies and legislation. Since this study is focused on the implementation of IFWs in New Zealand it is important to understand the hydrometeorological hazards in NZ, and the existing roles and responsibilities for severe weather warnings in New Zealand.

2.4.1 New Zealand's Severe Weather Hazards

Sometimes it does us a power of good to remind ourselves that we live on two volcanic rocks where two tectonic plates meet, in a somewhat lonely stretch of windswept ocean just above the Roaring Forties. If you want drama - you've come to the right place. - Sir Geoffrey Palmer, Former Prime Minister of New Zealand.

The geographic setting of New Zealand means the country experiences a variety of natural hazards, from earthquakes, volcanic eruptions, to ex-tropical cyclones and floods. As an archipelago located in the mid-latitude zone (Figure 2.2) of westerly winds (i.e., the 'Roaring Forties') and approximately 1,600km away from the nearest land mass (Australia's west coast; Walrond, n.d.), New Zealand is particularly exposed to hazardous weather (DPMC, 2007). Unless otherwise stated the information in the following severe weather hazards profile is sourced from the Natural Hazardscape Report from the Department of the Prime Minister and Cabinet (DPMC, 2007).



Figure 2.2. Global positioning of New Zealand (Maps Open Source, n.d.).

New Zealand experiences a wide range of weather-related hazards. These hazards can produce minor to significant impacts on NZ society. These hazards are organised into six categories: (1) Coastal Hazards, (2) Geophysical Hazards, (3) Floods, (4) Severe Winds and Thunderstorms, (5) Snow, and (6) Droughts and Wildfire. The weather-related hazards that relate to each of these are now discussed. A map of New Zealand is presented in Figure 2.3 to provide context to the locations described in the following text.

Weather-related coastal hazards that NZ experiences are swells and storm surges, with related coastal erosion and inundation. Exposure to these hazards will increase with climate change, thus leading to greater impacts (Spalding et al., 2014). Swells are wind-generated waves with long distances between wave crests that do not lose much energy as they travel thousands of kilometres across the ocean. Southwest swells over 2 metres high regularly affect much of the west coast of the North and South Islands, while the northeast of the North Island can experience occasional northeast 4+ meter-high swells accompanied by storm surge from ex-tropical cyclones. The east of the North and South Islands experiences swells that are more variable in height and direction, with an average of 0.5-2 metres, rising to 4-6 metres on occasion. Significant swell events have caused notable damage and evacuations. For example, on 15 April 2020 large swells recorded at 5.5 metres unexpectedly impacted Wellington's South Coast, resulting in the evacuation of five properties, and property damage and road closures (WREMO, 2020). These swells were driven by a low-pressure system passing near Chatham Island, approximately 770km away. The impacts of this event resulted in the development of a Wave Warning Program with adjusted warning thresholds for swell hazards on the Wellington South Coast due to several recent impactful events on coastal properties and the exposure of these properties to significant swells (WREMO, 2021). This warning system was used a year later on 28 June 2021 where the New Zealand Meteorological Service - Te Ratonga Tirorangi (MetService) issued a heavy swell warning for parts of the Wellington South Coast (Morton, 2021). In preparation for these swells, some residents were told to evacuate and cordons were set up to disallow entry of residents and non-residents (Campbell, 2021).



Figure 2.3. New Zealand Regional Map (Maps Open Source, n.d.).

Storm surge is a "temporary elevation in sea level during storm conditions created by ... low barometric pressure and wind" (DPMC, 2007, p. 56). A common cause for storm surge is ex-tropical cyclones that approach from the northeast of New Zealand. The most vulnerable areas to storm surge are low-lying coastal areas. Storm surges combined with high tide are likely to cause more damage than storm surge during low tide. As of 2017, 22 billion NZD worth of infrastructure, 4,000 homes, and 13,000 people are situated along the NZ coastline (Bell et al., 2017). Storm surge can cause coastal inundation and erosion, risking damage to infrastructure and further impacts to the communities. For example, in 2018, ex-Tropical Cyclone Fehi made landfall with NZ; the storm's large storm surge, observed at 2.06m (unverified, see Adam et al., 2020) flooded many homes in the Nelson region (Noll, 2018). Rising sea levels and intensification of storm events due to climate change are expected to worsen the impacts of storm surge on the NZ coastline (Adam et al., 2020; Bell et al., 2017).

Land slips and debris flows are **geophysical hazards** but can be triggered by heavy rainfall. Rainfall-induced land slips and debris flow are influenced by slope angle, vegetation cover, soil depth, drainage patterns, geology/rock type, and the frequency of intense rainstorms. These land slips can damage homes, roads, rail, and farm infrastructure. Past significant rainfall-induced landslides have caused farreaching impacts including decreases in agricultural productivity due to pasture loss, and degradation of water quality due to silt washing into rivers and streams.

An example of this occurred on 18 May 2005 when heavy rainfall triggered several land slips and debris flows in Matatā, Bay of Plenty (Figure 2.3), destroying 27 homes and damaging 87 properties (McSaveney et al., 2005). One such debris flow occurred at Awatarariki Stream (Massey et al., 2019). Fortunately, no injuries or deaths occurred as most people were inside sheltering from the heavy rainfall that contributed to the debris flow (Massey et al., 2019). However, the results of subsequent risk assessments and the calculated annual individual fatality risk to dwelling occupants (Tonkin & Taylor Ltd, 2013, 2015) led the Whakatāne District Council to deem the risk assessment as "intolerable" (Massey et al., 2019, p. 1), thus requiring mitigative action. In response, an evaluation of a potential debris flow EWS was launched and concluded that a multi-staged debris flow EWS would be unlikely to allow all potential people in the hazard zone to evacuate safely and thus would not be an effective mitigative approach (Massey et al., 2019).

Floods are the most frequent and costly natural hazard in NZ, aside from earthquakes, causing millions of dollars' worth of damage (Rouse, 2011). *Floods* can be a result of widespread heavy rain and/or localised rain. *Heavy rainfall* can occur anytime and anywhere in NZ but is most common in the west and south of the South Island where moist air arriving from the west is forced upwards over the Southern Alps, and the air cooling produces precipitation in the form of heavy rainfall. Heavy rainfall in the North Island and northeast of the South Island is typically caused by ex-tropical cyclones moving over or past NZ from the north. In the southeast of the South Island, mid-latitude *cyclones* to the east can also produce heavy rainfall. In early April 2017, ex-Tropical Cyclone Debbie brought heavy rainfall to New Zealand (Cullen et al.,

2017). This rainfall, combined with saturated soil from a wet month in March resulted in very high river levels in both the Whakatāne and Rangitāiki Rivers in the Bay of Plenty (Cullen et al., 2017). These high water levels resulted in stop bank breaches, causing widespread, devastating flooding in the town of Edgecumbe (Cullen et al., 2017). Consequently, Edgecumbe was entirely evacuated due to flooding, leaving some residents unable to return to their homes (Cullen et al., 2017).

Localised rain is typically associated with **thunderstorms**, which can produce intense rain, damaging winds, hail, and lightning. They are difficult to forecast accurately due to the rapid speed at which they develop. Thunderstorms can last from minutes to several hours producing intense rain which can lead to flash flooding and sometimes hail. Flash floods pose a threat to life and can cause significant damage due to fast-moving debris flows. Over six weeks in March and April 2017, Auckland (Figure 2.3) was hit by six localised storms that brought heavy rainfall leading to flooding and landslips (Golubiewski, 2019). The suburb of New Lynn was particularly affected by the third storm, during which heavy rainfall exceeded the capacity of the stormwater network, leading to widespread road flooding that was exacerbated by a partial culvert collapse (Golubiewski, 2019). Fortunately, no injuries or deaths occurred from this event, but 69 buildings were affected, roads were closed for up to four weeks, 23 people were rescued, and 12 residents were evacuated from a multidwelling residential block (Golubiewski, 2019).

Lightning and hail are additional **thunderstorm**-related hazards that threaten New Zealanders. NZ experiences more than 50,000 *lightning* strikes per year, with one death reported every five to ten years (MCDEM, 2010). *Hailstorms* pose the greatest meteorological risk to the agricultural sector and buildings and infrastructure, by damaging crops and buildings respectively. Hailstorms account for some of the costliest weather events in NZ; the Timaru Hailstorm of 20 November 2019 ranks as the costliest meteorological event in NZ at over 170 million NZD (ICNZ, 2021), where golf ball-sized hail stones heavily damaged buildings, caused business closures and hazardous driving conditions (stuff, 2019). This is followed by the Marlborough-Nelson Hailstorm which occurred a year later, on 26 December 2020, costing over 50 million NZD, and causing widespread damage to fruit crops, agricultural infrastructure, and buildings (Neal, 2020).

Considering the **severe winds** category, New Zealand lies in the path of the 'Roaring Forties' (mid-latitude westerly winds) which produce strong and damaging winds. *Windstorms, tornadoes,* and *ex-tropical cyclones* pose the most significant wind hazards. In September and October 2013, windstorms heavily impacted the agriculture sector in Canterbury, costing millions of NZ dollars in direct and indirect impacts to farmers (Safa & Birendra, 2015). On 10 April 2018, Auckland was impacted by a windstorm that caused widespread power outages lasting several days due to downed trees and power distribution networks (Smol, 2018). The extended power outage in conjunction with cold weather led to increased welfare impacts and increased vulnerabilities of people depending on power supplies for medical support (Smol, 2018). As such, a significant welfare response ensued to distribute portable toilets, potable water, and power generators (Smol, 2018).

Tornadoes produced by **thunderstorms** are relatively small and short-lived in New Zealand but still threaten life and property. Between 20-30 tornadoes are observed in NZ annually, leaving 10-20-meter-wide and up to 5-kilometre-long damage paths. Tornadoes can happen anywhere in the country, and several have resulted in unfortunate fatalities. On 6 December 2021, an EF1 tornado struck the Whenuapai and Hobsonville suburbs in Auckland, resulting in the deaths of three construction workers on-site, seven injuries, and 10 million NZD worth of damage (Turner et al., 2013). More recently, another EF1 (1 News, 2021) tornado struck the Auckland suburb of Papatoetoe on 18 June 2021, unfortunately killing one person, injuring two people, and damaging 60 houses (Franks, 2021)

In addition to meteorological hazards presented by thunderstorms such as extreme wind, tornadoes, heavy rainfall, and lightning, thunderstorms have also recently been found to pose *health risks* in NZ relating to asthma attacks and allergens. On 2 December 2017, thunderstorm-induced asthma attacks were detected by an increase in asthma presentations at the Waikato Hospital in Hamilton (Sabih et al., 2020), (Figure 2.3). This event highlighted a need to develop response plans for paramedics and emergency facilities in NZ in the event of a large-scale thunderstorm, like the outbreak that occurred in Melbourne, Australia in 2016 (Thien et al., 2018).

Ex-tropical cyclones are the most common source of widespread high wind in Northland, Auckland, and the Bay of Plenty. They are also usually accompanied by heavy rainfall causing flooding. Tropical cyclones form over the warm tropical waters north of New Zealand. Typically, they tend to weaken as they move south to cooler waters. However, some ex-tropical cyclones heading south towards New Zealand can form into large mid-latitude storms infused with colder air; these storms can create widespread impacts on New Zealand's coastal communities. The ex-tropical cyclone cases previously presented (e.g., ex-Tropical Cyclones Fehi and Debbie) exemplify the risks posed by these systems to New Zealand. Both storms contributed to power outages and travel disruptions (i.e., road closures, flight cancellations, rail service cancellations) due to high winds (Noll, 2018).

The higher altitude areas of New Zealand commonly experience winter **snowfall**, while altitudes below 1000 metres experience winter snowfall less frequently. *Snowstorms* caused by cold southerlies or warm advection can disrupt electricity, telecommunications, and transportation, and can damage buildings and infrastructure. On 18 September 2010, heavy snowfall resulted in the collapse of a sports stadium in Southland (Department of Building and Housing, 2012). Snowstorms can also result in significant agricultural losses through lack of access to water supplies and lack of feed. In June 2006, heavy snowfall damaged power and telecommunication networks and closed roads for several days in South Canterbury (Kelly & Smith, 2012). Farmers experienced both *physical* and *mental health impacts* due to physical exhaustion from clearing snow and the added stress of the threat of another snowstorm (Kelly & Smith, 2012). Additionally, *snow avalanches* in New Zealand's alpine areas threaten the lives of people working in the mining industry and of people involved in mountaineering, skiing, or snowboarding. From 1999 to 2018, there were 742 reported avalanche incidents and 27 fatalities; averaging out to 37

annual reported incidents, and 1.35 annual fatalities (NZ Mountain Safety Council, 2021).

Droughts are caused by prolonged periods of below-average rainfall resulting in reduced soil moisture levels for extended periods of time. Droughts can cause widespread *economic, environmental, and public health* and *safety impacts* (e.g., food supply, water supply) (Centre for Public Health Research, 2014; Hendy et al., 2018), as well as *acute mental health impacts* on farmers (Goffin & ACC Policy Team, 2014). This makes droughts one of the costliest hazards in New Zealand. The estimated economic losses from droughts between mid-2007 and mid-2017 were estimated at 720 million NZD (Frame et al., 2018). These costs are expected to increase due to climate change (Frame et al., 2018; Mullan et al., 2005); eastern regions of NZ were found to experience more dry periods from 2004-2013 (Centre for Public Health Research, 2014).

Each year, New Zealand experiences 4,100 wildfires that burn around 4,170 hectares of forest and rural land annually (Langer & Wegner, 2018). Meteorological conditions such as strong winds, low humidity, and seasonal drought can lead to favourable conditions for dangerous fire weather (Pearce & Clifford, 2008). These conditions, combined with human activity such as arson, escaped burns, forestry operations, vehicle operation, and campfires (Cameron et al., 2007), produce most of the wildfire hazards in NZ (Hart & Langer, 2011; Pearce et al., 2008). Furthermore, with human development into rural lands and climate change, the risk of wildfires impacting humans is increasing and has already resulted in some of NZ's worst wildfire events at the rural-urban interface (Huggins et al., 2020; Pearce, 2018). In February 2017, the Port Hills wildfire impacted the southern boundary of Christchurch, burning 1,661 hectares, destroying nine houses, and damaging five other houses (Langer & Wegner, 2018). The fire led to the evacuation of over 450 homes, between 1,400 and 2,800 residents for three to nine days (Langer & Wegner, 2018), and cost nearly 18 million NZD (ICNZ, 2021). Two years later the Pigeon Valley fire impacted Nelson and became NZ's costliest wildfire disaster and the most destructive wildfire in 60 years (Dudfield et al., 2020). In February 2019 the Pigeon Valley fire burned more than 2,300 hectares of land southwest of Nelson and caused the evacuation of more than 2,500 residents (Dudfield et al., 2020). An investigation concluded the cause of the Pigeon Valley fire to be an accidental spark from agricultural equipment (Cowan, 2019).

New Zealand is prone to, and experiences, a wide range of hydrometeorological hazards, many of which require an EWS to alert people to their dangers. This thesis will focus on the severe weather warning and flood warning systems in New Zealand as opposed to EWSs for droughts and wildfires, as these were determined to be the most relevant warning systems for IFW systems based on discussions with experts and stakeholders both within and outside of New Zealand. For example, informal discussions with wildfire communications experts in NZ and Australia indicated that IFWs were not fully conceptualised for wildfires at the outset of this thesis, and it was unclear whether IFWs would be effective for these hazards. Alternatively, warnings for droughts require a different context from severe weather and flood warnings due to

the longer onset of droughts as opposed to shorter lead times for severe weather and floods.

2.4.2 Severe Weather Warnings

The Meteorological Services Act 1990 of New Zealand ensures the provision of meteorological services in the country and holds the Minister of Transport responsible for ensuring the provision of meteorological warning services (Ministry of Transport, 1990). As such, the Minister of Transport is responsible for arranging for the making and issuing of forecasts of the weather, and the collection and recording of meteorological information (Ministry of Transport, 1990). In line with this mandate, and under the State Owned Enterprises Act 1986, the Minister of Transport entered into a commercial contract with the Meteorological Service of New Zealand Limited – Te Ratonga Tirorangi (MetService) (a State Owned Enterprise or SOE) in 1992 (referred to as the Meteorological Services Contract) for the provision of these meteorological services (Williamson, 1998). Thus, the MetService is the appointed National Meteorological Service (NMS) of New Zealand. This is stipulated in the National Civil Defence Emergency Management Plan Order 2015, where:

- 1) Monitoring, identification, and analysis of geological and meteorological hazards and threats and subsequent issuing of hazard information is to be undertaken at all times by the following agencies:
 - a. the Meteorological Service of New Zealand Limited (severe weather); and
 - b. GNS Science (earthquake, volcanic activity, and landslides); and
 - c. the NEMA (tsunamis).
- 2) Relevant government agencies, CDEM Groups, local authorities, and lifeline utilities are to maintain arrangements to receive and respond to hazard information. (New Zealand Government, 2015b, cl 119)

The Meteorological Services Contract specifies the services that the MetService must provide. These services consist of three primary tasks: making and issuing warnings and forecasts, acquiring meteorological data, and providing emergency advice (Williamson, 1998). The MetService is responsible for providing basic public forecasts and issuing warnings of hazardous weather affecting land areas, issuing marine forecasts and warnings (Williamson, 1998). The explicit requirements for each of these services are provided in Table 2.4 based on the Meteorological Services Contract (Williamson, 1998). For land-based hazards, the MetService is responsible for issuing warnings where the hazard meets a set of predefined physical thresholds: widespread heavy rainfall, widespread heavy snowfalls, widespread severe gales, heavy coastal swells, and severe thunderstorms. The MetService must also provide regular forecasts covering the following three days, and two brief forecasts for the mountainous regions. Marine forecasts and warnings issued by the MetService include warnings of gales, storms, and hurricanes for the Tasman Sea and a large part

of the South Pacific Ocean, as well as a series of regular forecasts described in Table 2.4.

Table 2.4. Meteorological Forecast and Warning Services provided by the MetService , taken directly from the Meteorological Services Contract (Williamson, 1998. paras. 5-7)².

Service	Requirements & Responsibilities
Warnings of hazardous weather affecting land areas	 Widespread heavy rain, exceeding 50 mm in 6 hours or 100 mm in 24 hours Widespread heavy snowfalls below 1000 metres [above sea level] on the North Island and 500 metres [above sea level] on the South Island, exceeding a depth of 10 cm in 6 hours or 25 cm in 24 hours Widespread severe gales with a sustained wind speed of 90 km/hr or more, or frequent gusts of 110 km/hr or more Heavy coastal swells, according to agreed criteria for certain regional authorities
Basic public forecasts	 Four short forecasts are issued daily, covering the next two days, for all of New Zealand Two extended short forecasts are issued daily, covering the following three days Two brief forecasts are issued daily for the mountain areas of New Zealand
Marine forecasts and warnings	 Warnings of gales, storms, and hurricanes are issued as required for the Tasman Sea and a large part of the South Pacific Ocean, extending halfway to South America, and from latitude 25°S to 55°S (roughly 6% of the world's oceans) Twice daily synopses and forecasts are issued for the same area Four times daily detailed marine warnings and forecasts are issued for the coastal waters of New Zealand (up to 100 km from the coast) and the Chatham Islands (the precise areas covered for these coastal services are as specified in the New Zealand Nautical Almanac) Warnings of near gales (25 to 33 knots) are issued as required for the Auckland marine area - Manukau and Waitemata harbours and the Hauraki Gulf south of a line from Cape Colville to Bream Head. Four times daily, marine forecasts are issued for the Auckland marine area, Wellington Harbour and south coast, for inshore waters from Waitarere to Pukerua Bay, and for Pegasus Bay from the mouth of the Waimakariri River to Lyttelton Harbour MetService operates a radio-facsimile service broadcasting marine weather charts over the Pacific Ocean south of the equator

² Over the last 20 years since signing the Meteorological Services Contract, the requirements in this contract have been expanded. However, the current Commercial Contract cannot be shared publicly. Examples of such expansions are the introduction of thunderstorm warnings, and an increase in the number of inshore marine forecasts.

In times of emergency, the contract ensures that the MetService will make urgent meteorological advice available. The contract defines emergencies as

the need for weather information for national and international search and rescue operations, a fruit fly or a foot and mouth disease outbreak, volcanic eruptions and marine pollution incidents (Williamson, 1998, para. 8).

The creation and issuance of forecasts and warnings requires extensive meteorological observations to ensure the services are as accurate as possible.

In addition to creating and issuing meteorological forecasts and warnings, the MetService is responsible for acquiring the appropriate data to support these services, as outlined in the Contract (Williamson, 1998). To meet this requirement, the MetService maintains a data network in compliance with standards prescribed by the WMO. The data collected comes from observations across New Zealand, upper air observations, weather surveillance radar, ships, a buoy network in the Tasman Sea, and aircraft (e.g., Aircraft Meteorological Data Relay or AMDAR) (Williamson, 1998). The MetService makes a selection of its data available nationally and internationally in accordance with WMO regulations for the exchange of data. The MetService provides free public access to a particular set of observational data through its official website, and distributes observational data from the Regional Basic Synoptic Network through the WMO communications for unrestricted use (Williamson, 1998). The MetService passes all observational data in direct support of forecast services to the National Institute of Water and Atmospheric Research (NIWA) for archiving in the National Climate Database (Williamson, 1998). Additional tasks outlined in the Meteorological Services Contract include providing support services to the Pacific, representing New Zealand to the WMO, and maintaining quality assurance and performance standards (see Williamson, 1998).

The severe weather warning service provided by the NZ MetService includes Outlooks, Watches, and Warnings. An Outlook is defined as a "heads up" that bad weather may occur in the next 3-5 days (MetService, 2019b). By issuing an Outlook, the MetService aims to increase recipients' awareness of potential severe weather (MetService, 2019b). Following an outlook, if hazardous weather is expected, the MetService can issue a Watch or Warning, as listed in Table 2.5 with their respective thresholds.

A Severe Weather Watch is typically issued within 1-3 days of potential severe weather with additional details around timing, location, and intensity (MetService, 2019b). The MetService issues a Watch either when they expect severe weather will occur but has not reached warning criteria, or when uncertainty is significant (MetService, 2019b). The MetService instructs Watch recipients to remain alert and prepared (MetService, 2019b).

Table 2.5. The thresholds as defined by the MetService for its Severe Weather Watches, Severe
Weather Warnings, Severe Thunderstorm Warnings, Road Snowfall Warnings, and Heavy Swell
Warnings (taken directly from MetService, 2021c).

Watch or Warning	Thresholds
Severe Weather	 After the next 24 hours but within 48-72 hours, or
Watch	• If there is a high level of uncertainty within the next 24 hours.
Severe Weather	• Widespread ³ rainfall greater than 50 mm within 6 hours or
Warning	100 mm within 24 hours;
	 Widespread snowfall below 1000 m [above sea level]on the
	North Island, South Canterbury or Otago - or below 500 m [above
	sea level] elsewhere on the South Island with a snow depth of 10
	cm within 6 hours or 25 cm within 24 hours;
	 Widespread severe gales with a minimum mean speed of
	90 km/h or frequent gusts exceeding 110km/h.
Severe	 Heavy rain from thunderstorms of 25mm/h or more.
Thunderstorm	 Large hail of 20mm diameter or more.
Warning	 Strong wind gusts from thunderstorms of 110km/h (60 knots) or
	more.
	 Damaging tornadoes with wind speeds more than 116 km/h
	(63 knots) (i.e., Fujita F1 or stronger).
Road Snowfall	• For Napier-Taupo Road (SH5), Desert Road (SH1), Remutaka Hill
Warning	Road (SH2), Lewis Pass (SH7), Arthur's Pass (SH73), Porters Pass
	(SH/3), Haast Pass (SH6), Lindis Pass (SH8), Crown Range Road,
	Milford Road (SH94) or Dunedin to Waitati Highway (SH1).
	Whenever there is a likelihood of snow settling on one or more of
	those roads within the next 24 hours.
Heavy Swell	The Wellington South Coast:
vvarning	• Swell height of 4 m or more from the S or SE with a long period of
	14 seconds or more
	• Combined waves of 6m or more from the S or SE with a long
	period of 14 seconds or more

A Warning is issued when MetService forecasters are confident about what will happen and where, and that any of the thresholds listed in Table 2.5 will be met by the weather conditions in the next 24 hours (MetService, 2021c). Warnings are usually issued within 1-2 days of the potential severe weather (MetService, 2019b). When a Warning is issued, the MetService instructs recipients to act, prepare for disruptions, and follow the advice of local authorities if the impacts are expected to be significant.

The MetService issues a Severe Thunderstorm Warning when one or more of the Severe Thunderstorm Warning criteria listed in Table 2.5 are met (MetService, 2021c) (MetService, 2021c). The MetService further notes that tornadic systems such as funnel clouds, waterspouts, and land-based tornadoes are possible with thunderstorms that they have not classified as severe. The MetService does not currently issue any specific tornado warnings due to difficulties in detecting and forecasting for them (e.g., Franks, 2021), however tornadoes may be signalled as a threat in a Thunderstorm Warning issued for other phenomena including heavy rain, damaging winds or large hail.

³ "Widespread" means over an area of 1,000 square kilometres or more (MetService, 2021c).

Road Snowfall Warnings are issued for specific state highways listed in Table 2.5 when snowfall is expected (MetService, 2021c). An additional disclaimer stipulates that the MetService does not provide information about road conditions or the amount of snow currently on the roads (MetService, 2021c). Heavy Swell Warnings are issued publicly for the Wellington South Coast whenever it is expected that the conditions listed in Table 2.5 will occur in the next 24-48 hours (MetService, 2021c). The MetService notes that arrangements exist with Regional Councils along other parts of the NZ coast for Swell Warnings for those areas, but those are not currently published on the MetService website.

In 2019 the MetService instituted changes to their warnings such that a differentiation can be made for events that are expected to produce significant impacts (MetService, 2019a). The new warnings are categorised by colour to correlate with the level of expected severity and impacts, as shown in Figure 2.4. Severe Weather Watches are Yellow (Figure 2.4a) for possible severe weather (MetService, 2019a). Severe Weather Warnings can be either Orange (Figure 2.4b) or Red (Figure 2.4c). An Orange Warning is issued when the forecast indicates incoming severe weather as per the thresholds in Table 2.5. A Red Warning is reserved for the "most extreme" weather events, where they can expect significant impacts and disruptions, and immediate action is required (MetService, 2019a).



Figure 2.4. The MetService's colour-coded Severe Weather Watches and Warnings. a) Yellow Watch for recipients to "Stay Alert". b) Orange Warning for people to "Take Action". c) Red Warning reserved for events to produce to significant impacts, instructing recipients to "Take immediate action, act now!" (MetService, 2019a).

At the time of thesis submission (October 2021), the MetService has issued four Red Warnings since the implementation of the new warning system, three for heavy rainfall (e.g., Forrester & Daly, 2021; Gabel, 2021; SMC, 2020), and one for severe gales (e.g., Tan, 2021). The first Red Warning was issued on 3 February 2020 (Figure 2.5) for heavy rainfall in parts of Southland and West Coast Regions on the South Island of NZ after receiving 350 mm of rain in 24 hours, and surface flooding and land slips were observed, with more rain forecasted for the next two days (SMC, 2020). A state of emergency was declared after floods occurred in Southland, resulting in road closures due to land slips and the evacuation of over 5,000 people (Emergency Management Southland, 2020). The second Red Warning was issued over a year later on 28 May 2021 for heavy rain in Canterbury. In response to the event, a state of emergency was declared for the entire region of Canterbury; impacts included 50 school closures, over 300 homes evacuated, 4 bridges washed away, and 20 road closures (Quinlivan et al., 2021). The third Red Warning was issued on 15 July 2021 for heavy rain, upon expectation of flooding and slips in Westland and Buller (Gabel, 2021). Buller District Council pre-emptively declared a state of emergency stating an expectation that houses were likely to flood overnight; the state of emergency declaration allowed emergency services to activate and plan for evacuations (Gabel, 2021). The flood impacts resulted in evacuations of more than 800 people in Westport (RNZ, 2021).

As shown in these three Red Warning examples, severe weather hazards such as rainfall can introduce secondary hazards, such as floods, which can result in their own impacts. In New Zealand, floods resulting from severe weather hazards such as ex-tropical cyclones and heavy rainfall have resulted in significant impacts. As such, an overview of the flood forecasting and warning system is provided next.



Figure 2.5. Image release of the first Red Warning issued by the MetService on 3 February 2020for parts of West Coast and Southland regions on New Zealand's South Island (source: MetService).

2.4.3 Flood Warnings

Flood warnings do not fall under the jurisdiction of the MetService in New Zealand. Flood risk communication (specifically forecasts and warnings) in New Zealand is more complex and layered than severe weather communication as it is tied into flood risk management practices, which are multi-layered and 'hierarchical' (Rouse, 2011). Three pieces of legislation outline the responsibilities for natural hazard management, including floods: the Resource Management Act 1991, the Civil Defence and Emergency Management Act 2002, and the Local Government

Amendment Act 2012. Under the Resource Management Act 1991 (the RMA), sixteen regional councils or unitary authorities hold the responsibility for water management (Ministry for the Environment, 1991; Rouse, 2011). Furthermore, the RMA assigns natural hazard mitigation responsibilities to territorial authorities, such as district and city councils (Ministry for the Environment, 1991; Rouse, 2011). This includes flood risk management (Ministry for the Environment, 1991; Rouse, 2011).

The Civil Defence and Emergency Management Act 2002 (the CDEM Act) is another piece of legislation which clarifies more specific roles and responsibilities for hazard management and mitigation (Rouse, 2011). The CDEM Act stipulates local authorities must create and become members of the Regional Civil Defence and Emergency Management Groups, who then write Civil Defence and Emergency Management Plans (MCDEM, 2016; Rouse, 2011). These plans support the coordination and planning for reduction, readiness, response, and recovery for all hazards, including floods (MCDEM, 2016; Rouse, 2011).

Finally, the Local Government Amendment Act 2012 (the LGA) ensures that local authorities take a sustainable development approach to promote social, economic, environmental, and cultural wellbeing (Department of Internal Affairs, 2012; Rouse, 2011). The LGA empowers local authorities to decide which activities to undertake and how it will be done (MCDEM, 2016; Rouse, 2011). As such, if flood hazard management is deemed key to the communities' well-being, the council is responsible for arranging funding and planning of these management efforts (MCDEM, 2016; Rouse, 2011).

Local and Regional Councils hold the bulk of responsibility for flood monitoring, forecasting, and warning, with support from NIWA, the MetService, and CDEM Groups (Ministry for the Environment & The Flood Risk Management and River Control Review Steering Group, 2008). The Local and Regional Councils carry out flood forecasts and create flood warnings which they disseminate through their own networks (such as their official websites) and via CDEM Groups (Ministry for the Environment & The Flood Risk Management and River Control Review Steering Group, 2008).

Reports on flood risk management have noted the need for greater collaboration between organisations and increased data and information sharing not only for improving flood warning systems, but also for improving greater flood risk management efforts (e.g., Cullen et al., 2017; Ministry for the Environment & The Flood Risk Management and River Control Review Steering Group, 2008; Rouse, 2011). One suggestion is for NIWA to take on the role of aggregating and sharing flood data (Ministry for the Environment & The Flood Risk Management and River Control Review Steering Group, 2008; Rouse, 2011).

Severe weather warnings and flood warnings require actions from the New Zealand Emergency Management (EM) sector for dissemination and response. Thus, the NZ EM structure is presented next.

2.4.4 The New Zealand Emergency Management Sector

When a severe weather warning and/or flood warning is issued, the Emergency Management (EM) sector in New Zealand has a role to play in the dissemination of and response to the warning messages. The EM sector in New Zealand is governed by the Civil Defence Emergency Management (CDEM) Act 2002.

The National Emergency Management Agency (NEMA) is the lead agency under the CDEM Act, as described in the *Guide to the National CDEM Plan 2015* (New Zealand Government, 2015a). The objective of NEMA is to provide leadership, strategic guidance, national co-ordination, and facilitate and promote key activities for reduction, readiness, response, and recovery (New Zealand Government, 2015a). The recipients of NEMA's leadership and guidance are primarily CDEM Groups (New Zealand Government, 2015a).

Under the CDEM Act, local authorities must form and maintain CDEM Groups, usually at the regional level (NEMA, 2021a). CDEM Groups are responsible for implementing local CDEM plans efficiently and effectively (New Zealand Government, 2015a). CDEM Groups provide leadership and co-ordination amongst member local authorities, partner agencies, clusters, and communities within their jurisdiction (New Zealand Government, 2015a).

Sixteen CDEM Groups exist in New Zealand, as shown in Figure 2.6. CDEM Group members can consist of regional councils and territorial authorities (i.e., city and district councils). For example, the members of the CDEM Group for the Southland Region of NZ (known as Emergency Management Southland), consist of Gore District Council, Invercargill City Council, Southland District Council, Southland Regional Council. Each CDEM Group outlines their own tasks and associated roles and responsibilities in their Group Plan, which they make publicly available through their websites⁴.

Amongst their designated responsibilities under the CDEM Act, CDEM Groups are responsible for two tasks relevant to this research: emergency information dissemination, and response. While the MetService is the designated authority for issuing severe weather warnings and regional councils are responsible for issuing local flood warnings, CDEM Groups act to respond to and disseminate these warnings across their jurisdictions. For example, within the Southland Regional Group Plan, it is stipulated that in addition to responding to relevant warnings, the Southland EM Group "operates a 24-hour, 365-day-a-year, point of contact to receive and disseminate all warnings to the appropriate key stakeholders" (Emergency Management Southland, 2017, p. 34).

⁴ For example (accessed 4 July 2021): Southland EM Group Plan, <u>https://www.es.govt.nz/about-us/meetings?item=id:28zrwy7e21cxbynl4xbc</u>



Figure 2.6 Map of the 16 CDEM Groups in New Zealand with their respective group members, which consist of regional, city, and district councils. Source: NEMA.

During emergency response, the CDEM Group is primarily responsible for coordinating response activities. One key task to support this role is collecting, analysing, and disseminating information to the public and stakeholders (e.g., Auckland Emergency Management, 2016; Emergency Management Southland, 2017). This facilitates effective leadership and decision-making, efficient and accurate communication with the public and stakeholders, and situational awareness for response personnel (e.g., Auckland Emergency Management, 2017).

2.5 Chapter Summary

This chapter provided (1) a literature review of hydrometeorological EWSs and their associated challenges and shortcomings, IFW systems and how they have been proposed to address the shortcomings of traditional hydrometeorological EWSs, (2) the relevance of this research to the Sendai Framework, and (3) the context of NZ hydrometeorological hazards, EWSs, and emergency management.

The literature review indicates that IFWs may improve the communication of hydrometeorological hazards and their impacts, and thus may allow warning audiences to make informed decisions and take appropriate protective action. IFWs differ from traditional hydrometeorological warnings by incorporating knowledge of impacts, vulnerability, and exposure into the decision-making process for issuing warnings. However, a key challenge was identified in the literature review around implementing IFWs: meteorological and hydrological services that issue warnings do not have the required knowledge and data relating to impacts, vulnerability, and exposure. Thus, there is a need to explore how these services can obtain and use this information to support the development and design of IFWs.

It was then argued that as a member nation of the UN and the WMO, New Zealand is responsible for implementing effective and user-friendly EWSs for its various natural hazards and is duty-bound to report its progress towards achieving the priorities and objectives of the Sendai Framework. Two such priorities were identified to be relevant to this doctoral thesis: those of understanding disaster risk; and of enhancing disaster preparedness for effective response. IFWs align with these Sendai Framework priorities, and thus have been identified as an opportunity for this doctoral research which aims to support NZ's goals of meeting these priorities. This will be achieved by exploring the data needs of IFWs in the NZ context and investigating how these data can be accessed and used by stakeholders involved in an IFW system for hydrometeorological hazards in NZ.

	Chapter One		
Part 1	Introduction		
	Chapter Two		
	Background and Context		
	Chapter Three		
Part 2	Research Design		
T GIT Z	Chapter Four		
	Methods Results and Overarching Findings		
	Chapter Five		
	Volunteered Geographic Information for people-		
	centred severe weather early warning: A		
	literature review		
	Chapter Six		
	Identifying the data uses and gaps for severe		
	weather impact forecasts and warnings		
	Chapter Seven		
	Identifying data sources for severe weather		
Part 3	impact forecasts and warnings in Aotearoa New Zealand		
	Chapter Eight		
	A socio-technical analysis of sharing and		
	governing hydrometeorological impact data in		
	Aotearoa New Zealand		
	Chapter Nine		
	Integration of Partnerships and Collaboration to		
	Support IFWs and HIVE Data Collection, Access,		
	and Use		
Part 4	Chapter Ten		

Position of the reader

Chapter Three: Research Design

The research question guiding this PhD study is

Which data are needed to support Impact Forecasting and Warning (IFW) systems for hydrometeorological hazards, and how are they currently collected, stored, and shared in New Zealand?

Based on the overview and definition of an EWS, provided in Chapter Two, it can be argued that an EWS/IFW system is a "central component and output" of a larger information system involving risk assessments, communication, and decision support (Pulwarty & Sivakumar, 2014, p. 15). Since data sources and information needed to support severe weather IFW systems are the focus of this study, it is logical to conduct this research through an Information Systems lens. Information Systems (IS) research investigates the use and impacts of information technology development through understanding "communication and collaboration between people and organisations, inter-organisational systems, electronic commerce and the Internet" (Myers & Avison, 2002, p. 2).

The IS lens applied to this research informed the research design, which consists of two major components, the research paradigm and the research methodology (Al Kilani & Kobziev, 2016), as shown in Figure 3.1. The research paradigm consists of the researcher's ontological and epistemological views (i.e., how the researcher views and interprets the world) (Morgan, 2007; Scotland, 2012). The research methodology is comprised of the research approach (e.g., qualitative, quantitative, or mixed methods approach), the research strategy (e.g., Grounded Theory, Case Study research, Design Science Research), the data collection methods (e.g., interviews, workshops, surveys), and the analytical tools and techniques (e.g., constant comparison, statistical analysis techniques) (Al Kilani & Kobziev, 2016). These components and sub-components will be defined in the following sections, and the overall design of the research will be provided.

This chapter presents the chosen design for this doctoral research, consisting of the research paradigm (Section 3.1), the methodology (Section 3.2), the data collection methods (Section 3.3), the analytical techniques and tools (Section 3.4), and the ethical considerations (Section 3.5). The chapter then concludes with a chapter summary (Section 3.6). Results of these methods, including the number of participants, and themes and codes identified from the analytical techniques and tools are provided in Chapter Four.



Figure 3.1. Conceptual framework of the research design (based on Al Kilani & Kobziev, 2016).

3.1 The Research Paradigm

There is the ontological question, 'What is the form and nature of reality and, therefore, what is there that can be known about it?'; the epistemological question, 'What is the relationship between the knower or would-be knower and what can be known'; and the methodological question, 'How can the inquirer ... go about finding out whatever he or she believes can be known about?' (Heron & Reason, 1997, p. 276).

Within social science research, the research design is guided by an overarching paradigm, which encapsulates a researcher's beliefs and view of the world (Morgan, 2007). Ontology (Figure 3.1) is "the study of being" (Crotty, 1998, p. 10), and refers to the 'nature of reality'. In conducting research, ontology refers to the researcher's beliefs of the world and reality (Lee, 2004). Epistemology (Figure 3.1) refers to the theory of knowledge (Cardinal et al., 2004), and is "concerned with how knowledge can be created, acquired and communicated" (Scotland, 2012, p. 9). The epistemological approach of a researcher describes how they understand, interpret, and develop their knowledge of the world, guiding their methodology (Gray, 2014). An understanding of a researcher's chosen methodology (Gray, 2014).

Chapter 3: Research Design

Based on the IS literature, there are five commonly used research paradigms within IS research: Positivism/Post-positivism, Interpretivism, Critical Theory, Pragmatism, and Participatory Inquiry. A deeper understanding of each paradigm supported by references to the literature is presented in Appendix B, and a summary is presented in Table 3.1.

As summarised in Table 3.1, *Positivist/Post-positivist researchers* see one reality, and gain knowledge of this reality through experimentation and measurements to investigate, determine, and predict the causal relationships (i.e., the 'how' and 'why') and theories developed from the data deductively (Creswell, 2014; Scotland, 2012). Deduction involves using a pre-existing theory or set of rules to examine the data and determine how the data support or disconfirm the theory or rules (Kennedy, 2018).

The Interpretivist research paradigm proposes that reality is relative, formed by the social actors and their social structure and culture (Guba & Lincoln, 1994; Lee, 2004; Scotland, 2012). Knowledge is gained through transactions and dialogue with social actors, such as interviews, and is based on personal experience (Guba & Lincoln, 1994; Mertens, 2015; Scotland, 2012). The researcher employs methods that require interaction with the participant(s) to understand their perspective and build context (Scotland, 2012). Theories are drawn from the data through induction, rather than deduction (Morgan, 2007). Induction occurs when the researcher identifies patterns through empirical cases and creates a general statement based on this evidence (Kennedy, 2018).

The Participatory Inquiry paradigm proposes that a world exists outside of the observer, but the observer perceives and understands the world based on their own experiences and interactions with the world (Breu & Peppard, 2001; Heron & Reason, 1997). Participatory Inquiry research is designed for intervention and action and emphasises the importance of conducting research with the participants, rather than on the participants (Breu & Peppard, 2001; Heron & Reason, 1997). Participatory Inquiry is a cyclical process in which all the people involved in the research continuously engage in defining the problem, applying the methodology, and communicating the results. This cyclical approach ensures the practice of critical subjectivity (Heron & Reason, 1997).

Critical Theory holds that multiple realities exist and are shaped over time and history, with power dynamics at play (Guba & Lincoln, 1994; Mackenzie & Knipe, 2006; Mertens, 2015). The epistemological approach under this paradigm is similar to that of the Interpretivist, with the added dimension of applying a cultural lens and maintaining awareness of power issues (Mertens, 2015). The Critical Theory researcher gains knowledge through similar methods used in Interpretivist research but maintains awareness of power dynamics and ensures that diverse, marginalised groups are included (Mertens, 2015).

Table 3.1. Summary of paradigms used in IS research.

	Positivism/Post-positivism	Interpretivism	Critical Theory	Pragmatism	Participatory Inquiry
Ontological approach	Realist. Reality is real and independent of the knower.	Relativist. Reality is formed by the social actors and their social structure and culture.	Historical realism. Human experiences and cultures shape reality over time and there are multiple versions of reality based on social positioning.	Less emphasis is placed on ontology. Pragmatism recognises the value of the former paradigms, namely Positivism/Post-positivism and Interpretivism, in conducting social research.	Subjective-objective and participative. A world exists outside of the observer, but the observer perceives and understands the world based on their own experiences and interactions with the world.
Epistemological Approach	Objectivist. In positivist research, the researcher and subject are independent. Under post- positivism the researcher approaches the subject objectively to prevent the researcher's biases from influencing the outcomes.	Transactional and subjectivist. Knowledge is gained through interactions; knowledge is subjective because it is based on personal experience.	Transaction and subjectivist with a cultural lens applied to maintain awareness of power dynamics.	Experiential - Pragmatists gain knowledge from inquiries and actions.	Four forms of 'knowing': experiential, presentational, propositional, and practical. The researcher practices critical subjectivity to maintain awareness of the four ways of knowing and of how they are interacting.
Research Approach	Experimental and manipulative. Typically, quantitative, guided by hypotheses that are empirically tested and verified; theories are developed deductively.	Hermeneutical and dialectical. Research is built on discussions and interpretation; typically, qualitative; theories are developed inductively.	Dialogic and dialectical. The researcher and subject must develop a transactional dialogue to exchange and develop ideas, and to transform ignorance and misapprehensions.	Mixed methods to use both quantitative and qualitative research to develop and test theories through abductive reasoning.	Participatory and collaborative - research is done with the people, rather than on them.
Methods	Experiments, quasi- experiments, tests, surveys, scales.	Interviews, observations, document reviews, visual data analysis.	Similar to methods used in Interpretivist research, but especially designed to include diverse, marginalised groups and allow them to participate in the research.	The researcher chooses the most appropriate methods in answering the research question(s).	Similar to mixed methods with emphasis placed on participation.

Pragmatic researchers recognise the importance and value of the Positivist/Postpositivist and Interpretivist epistemologies, and believe that there is a single and measurable 'real world,' but human experiences also help to build knowledge and understanding of the world in which we live (Goldkuhl, 2004). Under the Pragmatism paradigm, knowledge is gained through experience and action (Goldkuhl, 2004; Morgan, 2014). The Pragmatic paradigm offers flexibility for the researcher to draw from other paradigms and methodologies for obtaining 'useful answers' (Johnson & Onwuegbuzie, 2004; Mertens, 2015).

3.1.1 A Pragmatic Choice

Severe weather IFWs bring together the measurable world of meteorology and its associated hazards (e.g., floods) and the various experiences (or realities) of people exposed to the hydrometeorological hazards (i.e., impacts). As such, it is not adequate to approach this research problem from just a Positivist/Post-positivist or Interpretivist perspective. Doing so risks excluding either the objective nature of meteorology or the subjective experiences of the impacts.

The participatory nature of Participatory Inquiry in every stage of the research introduces practical challenges to a PhD study. The participation may be limited to interviews, workshops, focus groups, and discussions at limited times that are best-suited for the 'co-researchers' and 'co-subjects' (Heron & Reason, 1997). This can introduce delays in the various stages of the research, which were deemed less than ideal for the time-limited duration of a PhD study.

The Critical Theory paradigm introduces challenges like those of the Participatory Inquiry paradigm. The need for a transactional dialogue to exchange and develop ideas may also be limited to interviews, workshops, focus groups, and discussions. Furthermore, Critical Theory researchers direct their efforts towards politics and social oppression in research (Mertens, 2015). While these issues are relevant to this research, for example for marginalised groups, it is not the sole objective to address them in this study. Thus, the Critical Theory paradigm was also not fully suited for this study.

Under a Pragmatic paradigm, the researcher recognises the importance and value of the Positivist/Post-positivist and Interpretive epistemologies and believes that there is a single and measurable 'real world,' but is aware that human experiences also help to build knowledge and understanding of the world in which we live (Goldkuhl, 2004). This is an appropriate approach for severe weather impact forecasting and warning research as it recognises both the measurable world of meteorology and the less measurable, more subjective impacts that various people experience from those hazards. Furthermore, this research is looking at what information is available and how it can be used for IFWs. Pragmatist researchers recognise the value and importance of information and go beyond this by exploring how the information is used for decision-making (Ansell & Boin, 2017). This aligns well with the scope of this research.

Furthermore, a pragmatic (i.e., 'what works') approach is appropriate for meteorological and emergency/crisis/disaster management operations and practice, where it is important to find the most effective, efficient, and usually affordable approach to achieving the immediate task (Ansell & Boin, 2017; Theis et al., 2005). For this reason, the Pragmatist research paradigm has been argued for and employed in relevant research. For example, Ansell and Boin (2017) argued that the analytical and prescriptive approach of Pragmatism provides "a way of thinking that will help practitioners prepare for, and deal with, emerging risks, crises, and disasters" (p. 1079). Additionally, Theis et al. (2005) presented a "pragmatic, low-budget postprocessing procedure" (p. 266) for probabilistic precipitation forecasts to overcome cost barriers. Thus, Pragmatism was the chosen paradigm to guide this doctoral research.

3.2 Methodology

Methodology refers to the combination of the chosen research approach (e.g., qualitative, quantitative, or mixed methods), research strategy, data collection methods, and analytical tools and techniques used to create and build knowledge and understanding of that knowledge, as shown in Figure 3.1 (Al Kilani & Kobziev, 2016; Kothari, 2004). When designing the methodology, the researcher must evaluate which research approach, strategy, and methods will be most effective in answering the research questions.

3.2.1 Research Approach

The Pragmatist research paradigm is typically associated with a mixed methods approach (Mertens, 2015). However, it is also argued that mixed methods is not the only suitable approach for Pragmatist research (Feilzer, 2010; Mertens, 2015). The Pragmatist researcher can decide on the most appropriate and effective approach to answer their research question (Feilzer, 2010; Mertens, 2015). Along with assessing the effectiveness, the researcher also seeks the most appropriate methods based on consensus from the research community (Mertens, 2015).

A qualitative research approach was deemed as the most suitable research approach for this study due to its exploratory nature of investigating an area (i.e., IFW system implementation) that has had very little research conducted on New Zealand⁵. Exploratory research is "a flexible procedure in which the scholar shifts from one to another line of inquiry, adopts new points of observation as [their] study progresses, moves in new directions previously unthought of, and changes [their] recognition of what are relevant data as [they acquire] more information and better understanding" (Blumer, 1969, p. 40). The objectives of this research reiterated from Chapter One are:

- Objective 1.1: To establish VGI as a potential source of impact data for IFWs.
- Objective 2.1: Identify the actors involved in an IFW system and their associated roles.
- Objective 2.2: Determine how HIVE data are used in an IFW system.

⁵ See Potter et al., 2018; 2021 for New Zealand-specific IFW research.

- Objective 2.3: Identify further data gaps/needs for implementing IFWs.
- Objective 3.1: Identify the creators, collectors, and users of HIVE data that are relevant to severe weather IFWs.
- Objective 3.2: Identify the inhibitors and facilitators to collecting and using these data, to support the implementation of an IFW system in New Zealand for severe weather hazards.
- Objective 4.1: Identify and understand the governance and acquisition process for HIVE data for severe weather hazards in New Zealand.
- Objective 4.2: To support efforts to fulfil the Sendai Framework priorities around disaster data access.
- Objective 4.3: To support the implementation of a severe weather IFW system.
- Objective 5.1: To identify the required partnerships and collaborations required for IFW systems.
- Objective 5.2: To identify existing partnerships and collaboration in NZ that can support IFW systems.
- Objective 5.3: To outline a path forward for nurturing partnerships and collaboration for IFW systems.

Achieving these objectives involved identifying and interacting with the creators, users, sharers, and custodians of these data for the purpose of understanding current challenges with first collecting, using, sharing, and maintaining the data, and second, using it for an IFW system in New Zealand. A qualitative research approach was chosen for this study as it met the needs of this research by allowing for the development of a deeper understanding of the reasons behind processes leading to data collection, use, sharing, maintenance, and its applicability for IFW systems (Creswell, 2003). Qualitative research in IS employs several research strategies which will be described next. The most suitable research strategy for this study is also selected.

3.2.2 Research Strategy

Meeting the research objectives required engaging with the issuers of severe weather warnings, and the users of hazard, impact, vulnerability, and exposure data. The most suitable research strategies for this kind of engagement in Pragmatic IS research for this study were Action Research, Design Science Research, Practical Inquiry, Case Study, and Grounded Theory (Darke, Shanks, Broadbent, Gartner, & Pacific, 1998; Goldkuhl, 2008a, 2008b; livari & Venable, 2009; Peffers, Tuunanen, Rothenberger, & Chatterjee, 2007). These research strategies are summarised in Table 3.2. For a more in-depth description of the research strategies with references to the supporting literature, please see Appendix C. The following section provides an analysis of the benefits and limitations of each strategy in the context of this research, from which the most suitable strategy is selected.
	Action Research (AR)	Practical Inquiry	Design Science Research (DSR)	Case Study	Grounded Theory Methodology (GT)
Description	AR is "grounded in practical action, aimed at solving an immediate problem situation while carefully informing theory" (Baskerville, 1999, p. 3).	Practical Inquiry is rooted in the Pragmatic paradigm and is focused on generating knowledge for general practice.	DSR "creates and evaluates [information technology] artefacts intended to solve identified organisational problems" (Hevner et al., 2004, p. 77).	In-depth investigation into a unit of analysis (i.e., the case).	GT is the process of building a theory by systematically gathering and analysing data through the lens of the researcher (Chun Tie et al., 2019).
Key Objective(s)	To simultaneously inform practice and research to create change.	To generate knowledge for general practice, rather than local practice as is done through AR.	To produce an artefact to address a problem. Artifacts include constructs, models, methods, instantiation, social innovations, or new resource properties.	To develop an understanding of the interactions between information technology innovations and organisational contexts.	To generate conceptual insights from empirical evidence in a systematic and rigorous way (Bryant, 2009; Chun Tie et al., 2019).
Key Characteristics	Context-dependent, heavy involvement of participants. An iterative process involving collaboration between the researcher and participants through five phases. True AR research requires that phases be completed, but variations in AR exist for varying degrees of intervention and phase completion.	Possess similar characteristics as AR, but is concerned with creating knowledge for general, practical use outside of the local practice, which also contributes to the scientific body of knowledge.	The artefact is built by drawing from existing theories and knowledge. There is no specific client or collaboration between the researcher and client, but the artefact is developed and designed for a generalised class of clients (people or organisations).	Most associated with qualitative research but is not tied to one fixed paradigm, and is flexible in ontology, epistemology, and methodology. No specific requirements guide it, which allows for tailored research design. The Case Study can be single- or multi-case (Darke et al., 1998).	Several methodological genres exist: classic- Glaserian GT, evolved- Straussian GT, and constructivist GT (Chun Tie et al., 2019). The constant comparative method offers a process for organising and analysis qualitative data (Chun Tie et al., 2019). Theoretical sampling guides data collection (Chun Tie et al., 2019). Memo writing facilitates interpretation of data.

Table 3.2. Summary of common research strategies used in IS research. These were identified from a review of the literature.

Action Research (AR) is an applicable research strategy for this study. However, true AR with the completion of all phases was not feasible due to time and funding constraints that come with a PhD study. These limitations also apply to *Practical Inquiry Research*. It may also be difficult to apply the knowledge generated from Practical Inquiry to general practice across NZ, and outside of NZ. This is because, as was discovered in the preliminary interviews with EM officials across NZ, the warning system is not consistent across the country, and each CDEM Group has their own approach to issuing warnings (Ministry of Civil Defence and Emergency Management, n.d.). Thus, the knowledge contribution will likely be to local practice, rather than general practice. As such, it was determined that Action Research and Practical Inquiry were not suitable for this research.

The key objective of Design Science Research (DSR) is to develop an artefact to solve organisational problems (Hevner et al., 2004). Artifacts include constructs, models, methods, instantiation, social innovations, or new resource properties (Aken, 2004; Hevner et al., 2004). The objective of this study was not to design an artefact, thus DSR was deemed not suitable for this study.

The Case Study research strategy offers flexibility whilst also enabling in-depth understanding of the study case. Traditional case studies have been limited by geography and time, such as an event that occurred, or an agency/agencies (e.g., Ehnis, 2018; Yates & Paquette, 2011) The Case Study research strategy was not the most appropriate strategy for this PhD research because of the wide range of actors and organisations involved in hydrometeorological warnings and HIVE data collection in New Zealand. HIVE data are collected and used by various stakeholders, such as meteorological services, emergency management groups, risk modellers, researchers, etc. Case studies would not have captured a wide enough range of perspectives for this study.

Grounded Theory (GT), the process of building a theory by systematically gathering and analysing data through the lens of the researcher (Chun Tie et al., 2019), has become a widely used research strategy in IS research (Urguhart & Fernández, 2016; Wiesche et al., 2017). As an exploratory strategy, GT is particularly useful for research on areas with little to no prior research and where theory building is needed or desired (Fernandez, 2005; Lehmann, 2010). The field of IS research has been criticised for lacking theories developed within the IS context (Weber, 1987, 2003), due to the rapid development and evolution of information technologies (Weber, 1987, 2003). As such, GT has been identified as a viable approach for developing ISbased theories (Urguhart et al., 2010). Yet, while many IS research studies have claimed to use GT, they have been criticised for incorrectly labelling the study as a GT study when the GT was not fully utilised, and for failing to develop any theories (Urquhart & Fernández, 2016; Wiesche et al., 2017). Therefore, when GT is said to be applied for IS research, the researcher must ensure that they explicitly state how they used GT, whether for data analysis (in which case, the full methodology is not applied), or for data planning, collection, and analysis (i.e., the full GT is used) (Urquhart et al., 2010).

Chapter 3: Research Design

There is little empirical research on the data needs for IFWs. As such, GT was chosen as the research approach for this study to enable an exploration of the data needs and opportunities for IFWs, with a particular focus on New Zealand while simultaneously integrating international perspectives and experiences. Furthermore, the sampling technique commonly employed in GT (i.e. Theoretical Sampling) allows for, and encourages, a wide range of perspectives and participants to be included in the study.

The next section provides a brief history of GT and the three common branches: classic (or 'Glaserian') Grounded Theory, evolved (or 'Straussian') Grounded Theory, and Constructivist Grounded Theory. Rationale for choosing the Evolved-Straussian GT for this doctoral research is provided, followed by the research methods and tools used to conduct the research.

3.2.2.1 Grounded Theory Research Strategy

In 1967, sociologists Barney Glaser and Anselm Strauss published their seminal book titled *"The Discovery of Grounded Theory: Strategies for Qualitative Research"* (Glaser & Strauss, 1967) in an attempt to address a need for more systematic data collection and analysis methods for generating theory in social science and qualitative research. In their book, Glaser and Strauss identified concerns with an overemphasis on traditional theory testing and verification through fact-gathering (1967). They indicated that until their "discovery" of Grounded Theory, the field of sociology lacked theory development from empirical data which could then provide "relevant predictions, explanations, interpretations, and applications" (Glaser & Strauss, 1967, p. 1).

The novelties of GT lie in the fact that the concepts constructing the theory are derived directly from the data collected during the research, rather than being chosen before the research starts; and in the interrelatedness of data collection and analysis (Corbin & Strauss, 2015). Three basic principles form the foundation of the GT: (1) *emergence*, (2) *comparative analysis*, and 3) *theoretical sampling*.

The term "*emergence*" has been described as a "metaphor" (Seidel & Urquhart, 2016, p. 158) for the key principle of GT whereby categories and concepts 'emerge' (or are 'discovered') from the data through comparative analysis (Glaser & Strauss, 1967; Kelle, 2005; Seidel & Urquhart, 2016). Glaser and Strauss argued that, while GT researchers can borrow categories from existing theory (as long as the researcher continuously studies the data to ensure the categories fit), emergent conceptualisation of categories is the preferred approach to generating theory (Glaser & Strauss, 1967). This is because selecting data for a pre-existing category hinders the generation of new categories, and pre-existing categories are less relevant to the data in question, thus not the "best fitted" to the data (Glaser & Strauss, 1967, p. 37). Furthermore, trying to fit a category of another theory to the current data introduces the risk of 'forcing' the data into inappropriate categories (Glaser & Strauss, 1967). This reduces the credibility of the research (Glaser & Strauss, 1967). To facilitate the emergence of categories, Glaser and Strauss suggested the GT researcher "at first, literally to ignore the literature of theory and focus on the area under study" to ensure

that emerging categories are not "contaminated" by external concepts (Glaser & Strauss, 1967, p. 37). This suggestion has been widely criticised and debated in the literature and is discussed further in Section 3.2.2.2.

Comparative analysis, the second basic principle of GT, is a commonly used term in sociology and anthropology with several different meanings (Glaser & Strauss, 1967). Comparative analysis is useful for fact checking the evidence gathered; for establishing boundaries for where to apply categories and concepts derived from the evidence (i.e., making the theory generalisable); for specifying the unit of analysis for a one-case study by comparing other possible cases; for verifying theory; and for generating theory (Glaser & Strauss, 1967). Glaser and Strauss coined the term 'Constant Comparative Method,' which encapsulates four stages: "(1) Comparing incidents applicable to each category, (2) Integrating categories and their properties, (3) Delimiting the theory, and (4) Writing the theory" (Glaser & Strauss, 1967, p. 105).

Constant comparison is an iterative process whereby the researcher breaks the data down into pieces and compares each piece against the other. For example, the researcher may start by comparing incident to incident while coding. Through this process, the researcher will start thinking about the category, its dimensions, the conditions leading to it, the consequences of it, and how it relates to other categories (Glaser & Strauss, 1967). Once codes have been categorised they can be compared to other codes in other categories, categories compared to each other, and new data compared to earlier data (Glaser & Strauss, 1967). The stages do not have to be done consecutively; the researcher can transition between stages as the research and theory development progresses (Glaser & Strauss, 1967). Through this process, the researcher will identify clues, gaps, and uncertainties that future data collection can target (Chun Tie et al., 2019). This process is termed 'theoretical sampling' (Glaser & Strauss, 1967).

Theoretical sampling is the third basic principle of GT. Glaser and Strauss defined theoretical sampling as "the process of data collection for generating theory whereby the analyst jointly collects, codes, and analyses [their] data and decides what data to collect next, and where to find them, in order to develop [their] theory as it emerges" (1967, p. 45). Initial data collection may start with a "general sociological perspective" (Glaser & Strauss, 1967, p. 45) on a subject or problem area, but these decisions are not based on a preconceived theoretical framework (Glaser & Strauss, 1967). The researcher should employ 'theoretical sensitivity' to conceptualise the theory as it 'emerges' from the data during the concurrent data collection and analysis process (Glaser & Strauss, 1967).

Theoretical sensitivity is developed by asking questions of the data and the theory, such as "What does the theory do? How is it conceived? What is its general position? What kinds of models does it use?" (Glaser & Strauss, 1967, p. 46). Theoretical sensitivity is also developed from the researcher's experiences, and their theoretical insight into their field of research (Glaser & Strauss, 1967). Through the iterative process of constant comparison through writing memos and remaining

theoretically sensitive, theoretical sampling is used for data collection until 'theoretical saturation' is reached (Glaser & Strauss, 1967).

GT researchers complete the data collection phase of their research once they have reached 'theoretical saturation'; when no new properties or insights are obtained from new data (Glaser & Strauss, 1967). An indication of theoretical saturation is when the researcher observes repetitive instances in the data and "becomes empirically confident that a category is saturated" (Glaser & Strauss, 1967, p. 61). The researcher should collect a diverse range of data to ensure that they have reached saturation; this means gathering as many differences in the data as possible and comparing them (Glaser & Strauss, 1967). Saturation is met with the combined process of data collection and analysis. Furthermore, saturation is reached category by category. As such, when saturation is met in one category, the researcher continues data collection until saturation is met for all categories. At this point, the researcher may find that gaps in their theory have been filled (Glaser & Strauss, 1967).

3.2.2.2 The Three Main Strands of Grounded Theory

Since the release of Glaser and Strauss' seminal book, *"The Discovery of Grounded Theory"* (1967), the methodology has evolved to account for apparent contradictions in the book and different interpretations of the book. A key contradiction in the book is the concepts of emergence and theoretical sensitivity (Kelle, 2005). The GT literature has documented questions, debates, and criticisms of how researchers can allow for emergence, while also practicing theoretical sensitivity. In other words, how can a researcher know what to look for in the emerging categories and concepts without drawing from their own existing theoretical knowledge or practical experience? And how can researchers ensure that they are enabling emergence, rather than forcing, if they are to be theoretically sensitive?

In their book, Glaser and Strauss indicated that they were aware of this conflict, by writing "Of course, the researcher does not approach reality as a tabula rasa. [They] must have a perspective that will help [them] see relevant data and abstract significant categories from [their] scrutiny of the data" (Glaser & Strauss, 1967, p. 3). Yet, it was still identified as a conflict or contradiction by the wider research community. In response, Glaser and Strauss each developed their own approaches to attempt to reconcile this conflict, which consequently resulted in two different strands of GT. The first strand is considered closest to the initial idea of GT and is often referred to as 'Classic Grounded Theory' or 'Glaserian Grounded Theory'. Strauss developed his own approach, referred to as 'Evolved Grounded Theory' or 'Straussian Grounded Theory'. Finally, 'Constructivist Grounded Theory' is a third strand of GT are summarised in Table 3.3.

	Classic-Glaserian GT	Evolved-Straussian GT	Constructivist GT
Philosophical perspective	Positivist realist ontology or postpositivist critical realist ontology, objectivist epistemology (Rieger, 2019).	Pragmatism and symbolic interactionism (Rieger, 2019).	Constructivist stance with relativist ontology and subjective epistemology (Rieger, 2019).
Coding	Substantive coding	Open coding	Initial coding
procedures	Open coding	Axial coding	Focused coding
	Selective coding	Selective coding	Axial coding
	Theoretical coding		Theoretical coding
Analytical	13-23 Coding Families.	The Coding Paradigm (optional).	The researcher can employ analytical tools
Tools	Memo writing.	Memo writing.	developed by other grounded theorists if
	Constant Comparison.	Constant Comparison.	they are appropriate for the emerging analysis (Rieger, 2019). Memo writing.
Apriori	Doos not oncourago a priori litoraturo	Allows for the potential of a priori literature	Allows for a priori literature review, which
literature	review	review	helps to orient the research(er) (Charmaz
review	Compromises the researcher's ability to	Provides context and justification for the	2014)
	keep the tenets of classic GT (Rieger	study (Rieger 2019)	2011).
	2019).	Informs research questions (Rieger, 2019).	
	May waste researcher's time reviewing	Provides basis for demonstrating the	
	literature that is not significant to the core	appropriateness of GT for the study	
	variable that eventually emerges from the	(Matavire & Brown, 2013).	
	data (Matavire & Brown, 2013).	Increases theoretical sensitivity (Rieger,	
	Danger of researcher using literature to	2019).	
	deductively verify reviewed theoretical	Satisfies institutional requirements for	
	concepts (Matavire & Brown, 2013).	literature review (Matavire & Brown, 2013).	
Research	Does not start with a research question or	Researcher starts with a research question	A research question is developed before
question or	problem, but investigates area (Matavire &	as a statement about the phenomenon	beginning the research (Charmaz, 2014).
problem	Brown, 2013).	(Matavire & Brown, 2013).	

Table 3.3. Summary of the three main strands of Grounded Theory Methodology: Classic-Glaserian, Evolved-Straussian, and Constructivist.

3.2.2.3 Evolved-Straussian Grounded Theory Methodology for this study

Evolved-Straussian GT (ES-GT) was chosen for this PhD study. The primary reasons for this choice are: (1) ease of use for novice researchers (Charmaz, 2006; Kelle, 2005), (2) the allowance for a priori literature review (Hughes & Jones, 2003; Matavire & Brown, 2013), and (3) its wide use in IS research (Matavire & Brown, 2013; Urquhart et al., 2010).

ES-GT is considered the more appropriate strand of GT for novice researchers to learn as it provides clear guidelines and frameworks for data collection and analysis via the coding paradigm (Kelle, 2005). The coding paradigm used in ES-GT is said to provide novice researchers with structure and a starting point in generating categories (Charmaz, 2006; Seidel & Urquhart, 2016; Strauss, 1987). See Section 3.4.2 for more on the coding paradigm.

Despite these perceived benefits of the coding paradigm for novice researchers, the coding paradigm has received much criticism. The most notable concern with the coding paradigm is around the risks of 'forcing' versus emergence (Glaser, 1992). As Glaser and Strauss described, "to preconceive relevance is to force data, not to discover from data what really works as a relevant explanation" (1967, pp. 142-143).

Concerns of forcing by using the coding paradigm are valid because the coding paradigm stems from Pragmatist social theory (Kelle, 2005, para. 21). Thus, if the coding paradigm is applied in the wrong context, a risk of forcing exists. However, Kelle argued that upon a closer look of the conceptual design of the coding paradigm, the underlying theory of action is compatible with various sociological theories (Kelle, 2005). Furthermore, the coding paradigm appears to represent everyday human action (Kelle, 2005). While Strauss and Corbin initially viewed the coding paradigm as a mandatory tool in their methodology (1990), they have since indicated in more recent works that the paradigm is not mandatory but can be used as a 'springboard' to guide the analysis (Corbin & Strauss, 2008). The 1990 version (see Strauss & Corbin, 1990) of the coding paradigm was deemed a good fit for the data in this doctoral research, and thus was used in the analysis for this research. More on the analysis process and results for this doctoral thesis are presented in Chapter Four.

ES-GT allows for a literature review to be completed before the research is undertaken. The benefits of completing an a priori literature review, as listed in Table 3.3, are that a literature review provides context and justification for the study (Rieger, 2019); it can inform research questions (Rieger, 2019); it provides a basis for demonstrating the appropriateness of GT for the study (Rieger, 2019); it increases theoretical sensitivity (Rieger, 2019); and it satisfies institutional requirements for a literature review (Matavire & Brown, 2013). Furthermore, a literature review is expected for the completion of a PhD study, which is presented in Chapter Two and Chapter Five.

ES-GT has become a widely accepted research method in the field of IS research (Matavire & Brown, 2013; Urquhart et al., 2010). The coding paradigm is particularly useful for identifying and explaining causal relationships (Seidel & Urguhart, 2016). Axial coding and the coding paradigm from the ES-GT may provide "potential devices" for understanding causal and intervening conditions (2016, p. 191). In terms of the quality of the theories developed from ES-GT, the resultant theories may be classed as 'explaining' theory (i.e., explaining how and why things happen), as defined by Gregor (2006), because studies that applied the coding paradigm tended to produce causal relationships (Seidel & Urquhart, 2016). For example, Day et al. (2009) used ES-GT and applied the Strauss and Corbin coding paradigm to identify impediments to the flow of information in disaster relief supply chains. They found that the paradigm 'fit the data' well, and only had to alter one category, that of intervening conditions, as it was "too broadly defined" for their purposes (p. 642). Through this paradigm, they identified the causes of information flow impediments during disaster response and proposed recommendations to overcome these impediments based on the experiences of their interviewees.

Urquhart and colleagues (2010) and Seidel and Urquhart (2016) developed recommendations for researchers looking to apply ES-GT, and by extension, the coding paradigm. The recommendations from Seidel and Urquhart are summarised in Table 3.4. These recommendations were kept in mind throughout the data collection and analysis for this doctoral thesis.

3.3 Data Collection Methods

This study used a qualitative research approach. Qualitative data collection methods have been commonly used in both IS and warnings research (e.g., Potter, 2014; Prasanna, 2010). In the first phase of this doctoral research, interviews were the primary data collection method. The purpose of the interviews was to develop an understanding of the current perspectives on IFW systems in New Zealand and the associated data needs for the warning system. From these interviews, following ES-GT, a conceptual framework was developed around the data needs, sources, and uses of impact, vulnerability, and exposure data for IFW systems in New Zealand. More details on the conceptualisation of the framework are provided in Chapter Four.

The second phase of the research involved running workshops whereby the resulting framework developed from the interviews was presented to further New Zealand participants for verification and triangulation. More detail around the design and process of conducting the interviews and workshops is provided in the following sections.

Table 3.4. Summary of recommendations for IS researchers using Evolved Straussian Grounded Theory presented by Seidel and Urquhart (2016).

Recommendation	Summary
Flexible use of axial coding.	IS researchers should flexibly use axial coding "as a stage, where relationships between categories are identified" (Seidel & Urquhart, 2016, p. 190). If IS researchers apply axial coding as suggested in Strauss and Corbin (1990) they may limit themselves and risk hindering theoretical sensitivity and result in forcing of data.
A rationale for adaptations.	IS researchers should provide clear rationale for any adaptations they make to using the coding paradigm to ensure the integrity of the method.
Awareness of 'forcing' issues.	Researchers can use the paradigm as a "jumping-off point" (Seidel & Urquhart, 2016, p. 185) as opposed to rigidly using it, to avoid 'forcing.' Jointly using the coding paradigm with other coding families can enhance theoretical sensitivity. Many IS researchers that have used ES-GT opted out of using the coding paradigm, thus treating it as an option as suggested by Corbin and Strauss (2008).
Theoretical sensitivity towards causality.	IS researchers can use the coding paradigm as a "sensitizing device" (Klein & Myers, 1999; Seidel & Urquhart, 2016, p. 190). This can help the researcher think about causal relationships, which can lead to the development of a 'theory to explain' (Gregor, 2006).
Contextualisation.	Context is a feature of the paradigm. As such, IS researchers should consider using the paradigm specifically in studies seeking to uncover the context leading to the occurrence of a phenomenon. It is helpful for IS researchers to view "technology as being enmeshed with the conditions of its development and use" (Seidel & Urquhart, 2016, p. 190).

3.3.1 Interviews

A purposive sampling method (Chun Tie et al., 2019) was initially used to target and recruit interviewees based on their roles in severe weather risk communication, response, and use of impact, vulnerability, and exposure data. Recruitment was targeted towards individuals or organisations that issue hydrometeorological warnings, and/or collect, create, share, manage, maintain, and/or use hazard, impact, vulnerability, and exposure data for severe weather hazard and risk management. Potential participants were identified through networks and contacted directly via email. After the initial interviews were conducted, theoretical sampling guided the rest of the data collection process for this research whereby recruitment targeted participants who were knowledgeable or experienced with themes that emerged from previous interviews (see Section 3.2.2.1).

Data collection took place from November 2018 to May 2021. The purpose of the interviews was to supplement the gaps in the literature around the data needs and potential sources for IFW systems. The interview script was semi-structured to allow for follow-up questions to be asked based on interviewees' responses, and to follow-up on emerging themes in the interview (Patton, 2015). Questions were asked regarding the participants' thoughts on IFWs (e.g. what they know about it, perceived

challenges and benefits, requirements for implementation); what kind of impact, vulnerability, and/or exposure data they use, why, and how; the life path of the data (e.g. how it is obtained, used, stored, what happens to it after its intended use), experienced and/or perceived challenges with obtaining the required data for IFWs and other uses; thoughts on collecting and using alternative data sources (e.g. social media and crowdsourcing) to address their challenges (if they identified any). See Appendix D for the interview questions. The interviews were transcribed verbatim for subsequent analysis.

3.3.2 Workshops

Workshops were conducted in the second stage of data collection to diversify the qualitative dataset and to triangulate the research findings (Corbin & Strauss, 2015; Gibson, 2007). Workshops were chosen to capture a wider coverage of participants with diverse perspective for facilitating co-creation of knowledge (e.g., Belanger, 2012; Henriksen et al., 2018; Potter et al., 2021). Furthermore, nurturing partnerships and engaging with stakeholders is an important process for the successful implementation of IFWs (Hemingway & Gunawan, 2018; WMO, 2015), and workshops were deemed an effective method for stakeholder engagement and nurturing partnerships.

3.3.2.1 Virtual Workshop Methods

The workshops were held virtually due to COVID-19 concerns. From 12 to 30 August 2020, Auckland went into Alert Level 3⁶ (partial lockdown), and the rest of the country moved from Alert Level 1 to Alert Level 2, introducing physical distancing measures. From 30 August to 7 October 2020 Auckland was in Alert Level 2, while the rest of the country moved back to Alert Level 1 on 21 September 2020. On 7 October 2020, Auckland moved back to Alert Level 1 with the rest of the country. Due to the uncertainty around the changing COVID-19 Alert Levels and concerns with traveling across the country, the workshops were held virtually over Microsoft Teams.

The web-based whiteboarding platform, Mural⁷, was used to facilitate the workshop. A virtual whiteboard was mapped out with specific questions and activities for the participants to navigate and interact with. Prior to the day of the workshop, participants were provided with training resources to become familiar with the Mural platform. Workshop questions and activities are available in Appendix E.

Participants were asked to provide feedback on the impact, vulnerability, and exposure data framework that resulted from the interviews. They were then asked to identify data requirements and sources for IFWs from the perspective of their roles and agencies. The third activity requested the participants to identify 1-2 datasets/data sources that are important and outline the life track of the data, and to

⁶ The New Zealand government implemented a 4-level COVID-19 alert level system to inform the population on what measures are needed to reduce the spread of the virus

⁽https://covid19.govt.nz/about-our-covid-19-response/history-of-the-covid-19-alert-system/). ⁷ https://www.mural.co/

describe how they understand what impacts are occurring or could occur from a severe weather hazard. The fourth activity looked at identifying alternative uses for the datasets/data sources that they identified, beyond IFWs. Finally, the last activity investigated the application of the framework: whether the participants would use it, why or why not, and how. The workshops were audio recorded through Microsoft Teams, and data from the virtual sticky notes were entered into a spreadsheet for further analysis. More detail on the workshop results and the number of participants is provided in the next chapter, Chapter Four.

3.4 Analytical Techniques and Tools

Data collection and analysis occur concurrently in GT to facilitate theoretical sampling. Data collection and analysis are complete when saturation has been met and the core category/ies has/have been identified (Strauss & Corbin, 1990). Analysis for this doctoral thesis occurred in the form of constant comparison, memo writing and diagramming, and coding. Nvivo 12, a qualitative data analysis software package, was used to support the transcribing, coding, and memo writing. Nvivo 12 is an industry-standard and facilitates data management, idea management, data querying, graphical modelling of ideas and concepts, and data reporting (Bergin, 2011).

3.4.1 Memo Writing and Diagramming

Memos are "written records of analysis" (Corbin & Strauss, 2015, p. 57). Memo writing allows the researcher to document the constant comparison analysis, and to formulate and deepen their thoughts (Corbin & Strauss, 2015). The GT researcher should continuously write memos and draw diagrams to track their analytical process and draw out conceptual connections between categories (Strauss & Corbin, 1990). As such, GT researchers are encouraged to "stop coding and record a memo on your ideas" (Glaser & Strauss, 1967, p. 113). Memo writing and diagramming were carried out in this study throughout the data collection and analysis process. Evidence of this is presented in Chapter Four.

3.4.2 Coding

Coding is described as "denoting concepts to stand for meaning" (Corbin & Strauss, 2015, p. 57). Coding in ES-GT takes place in three stages: open coding, axial coding, and selective coding. Figure 3.2 provides a visualisation of the ES-GT data collection and analysis process employed in this doctoral research.

When open coding the researcher codes the data by line, sentence, or paragraph (Charmaz, 2006). Axial coding investigates the relationships between concepts and categories resulting from the open coding process (Strauss & Corbin, 1990). When axial coding, the researcher can employ Strauss and Corbin's coding paradigm for guidance (Strauss & Corbin, 1990). Selective coding, also referred to as theoretical integration in Corbin and Strauss' 2008 and 2015 texts, involves relating all categories to the core category/ies and to each other to develop the Grounded Theory. A core category is one which is central, where all major categories relate to it (Matavire & Brown, 2013). Multiple core categories may be identified (Strauss &



Corbin, 1990). Open coding, axial coding, and selective were conducted in this research until the core category/ies were identified and integrated.

Figure 3.2. Conceptual framework of the ES-GT analytical process employed in this research. Figure created by S. Harrison based on interpretations of Corbin and Strauss (2015), Corbin and Strauss (2008), Charmaz (2006), and Glaser and Strauss (1967).

Open coding, typically the first step in analysing the data in ES-GT, was carried out by conceptualising and categorising the data piece by piece and constantly comparing the data (Vollstedt & Rezat, 2019). Open codes were created to describe the core idea of concepts (Vollstedt & Rezat, 2019). Data that are categorised and conceptualised into one code is called a 'concept', and concepts that are related to each other to form a concept of a higher order are called categories (Strauss & Corbin, 1990). The goal of open coding is to develop rich analytical codes to describe the data (Vollstedt & Rezat, 2019). Theoretical sensitivity was applied throughout the coding process, which is developed from the researcher's expertise, experience, and knowledge of the literature (Vollstedt & Rezat, 2019).

Axial coding, the second phase of coding in ES-GT, was used in this research to investigate the relationships between concepts and categories resulting from the open coding process (Strauss & Corbin, 1990). After the data were broken up into pieces (concepts and categories), axial coding was used to piece the data back together to form higher conceptual codes (Vollstedt & Rezat, 2019).

Chapter 3: Research Design

The coding paradigm, a tool introduced by Strauss and Corbin (1990) to aid in the axial coding process, was used in this research as it was found to fit well with the data. The coding paradigm focuses on phenomena, causal conditions, contextual conditions, intervening conditions, action/interaction strategies, and consequences (see Table 3.5). By using the coding paradigm, open codes were linked back to these five dimensions for increased density and precision in the resulting theory (Strauss & Corbin, 1990). Axial coding was completed by relating each code and category to the coding paradigm dimensions presented in Table 3.5. Upon completing the axial coding phase selective coding was conducted as the final stage of analysis.

Coding Paradigm Dimension	Description
Causal Conditions	A set of events that influence the phenomena or result in the appearance or development of a phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Phenomenon	The subject or object under study (Strauss & Corbin, 1990).
Contextual Conditions	The specific set of conditions and characteristics surrounding the phenomena and resulting in action/interaction strategies taken to address the phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Intervening Conditions	Unexpected events or factors leading to action/interaction strategies (e.g., time, space, culture, socioeconomic status, technological status, history) (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Action/Interaction Strategies	Purposeful and deliberate acts taken to address the phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Consequences	Predictable or unpredictable, intended or unintended outcomes of the action/interaction strategies (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).

Table 3.5. Coding paradigm summary (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).

Selective coding in this doctoral thesis involved relating all categories to the core category/ies to develop the Grounded Theory. The core category/ies are categories that "incorporates or supersedes other categories in explanatory importance and hence is 'elevated' to the status of an important concept" (Timonen et al., 2018, p. 7). The goal of selective coding is to integrate all of the different categories resulting from axial coding into one cohesive theory (Vollstedt & Rezat, 2019). Selective coding aims to answer the questions "*What is the research all about?*" and "*What seems to be going on here?*" by relating the core category to all of the other categories (Corbin & Strauss, 2008, p. 14). In this stage, relations were validated, and categories were refined and further elaborated (Vollstedt & Rezat, 2019).

Once the relationships between codes were established through axial coding, supported by the coding paradigm, selective coding commenced to unify the resulting categories around a core category (Corbin & Strauss, 2008). Through continuous reflection, constant comparison between codes and categories, reassignment of codes and categories, memo-writing, and diagramming, the core

category was established. The last three interviews were conducted to explore any unclear aspects or outstanding questions of the core category to confirm or negate propositions made around the core category based on previous coding analysis.

3.5 Ethical considerations

A 'low risk' ethics notification was lodged with the Massey University Human Ethics Committee prior to data collection in 2018 (see Appendix F). Participants received information sheets and consent forms which they signed and returned. All interviewees remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by area of expertise and/or practice (e.g., Meteorology, Emergency Management, Data Management), industry (e.g., Private, Governance), location (e.g., NZ or International), or governance level (e.g., National, Regional, Local) (Table 3.6). The acronyms and abbreviations in Table 3.6 are as follow: Meteorology (Met.), Emergency Management (EM), New Zealand (NZ), Geographic Information Systems (GIS), Regional (Reg), Government (Gov.), Early Warning System (EWS), International (Int.). Table 3.6. Interviewee codes.

Interviewee Code	Position	Classification	Location	Government Level
Agriculture/Rural NZ. A	Agriculture policy coordinator	Agriculture/Rural	NZ	National
Data Management Gov. NZ. Nat. A	Senior Resilience Advisor	Data Management	NZ	National
Data Management Private NZ. B	Geospatial Specialist	Data Management	NZ	
Data Management Research NZ. C	GIS Specialist	Data Management	NZ	
Data Management Private NZ. D	GIS Specialist	Data Management	NZ	
Data Management Gov. NZ. Nat. E	Head of Data	Data Management; Governance	NZ	National
EM. NZ. Reg. A	Director	Emergency Management	NZ	Regional
EM. NZ. Reg. B	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. C	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. D	Principal Science Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. E	Principal Advisor Strategy and Partnerships	Emergency Management	NZ	Regional
EM. NZ. Reg. F	GIS Lead	Emergency Management; Data Management	NZ	Regional
EM. Gov. NZ. Nat. G	Senior Hazard Risk Management Advisor	Emergency Management; Governance	NZ	National
EM. NZ. Reg. H	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Nat. I	First Responder	Emergency Management	NZ	National
EM. Gov. NZ. Nat. J	National Operations Manager	Emergency Management; Governance	NZ	National
EM. NZ. Reg. K	Regional Manager	Emergency Management	NZ	Regional
EM. NZ. Reg. L	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. M	Group Controller	Emergency Management	NZ	Regional
Health NZ. Reg. A	Respiratory Doctor	Public Health	NZ	Regional

Interviewee Code	Position	Classification	Location	Government Level
Hyd. Gov. NZ. Reg. A	Flood EWS Programme manager	Hydrology; Governance	NZ	Regional
Lifelines NZ. Reg. A	Civil Engineer	Lifelines	NZ	Regional
Loss Modelling Research NZ. A	Economist	Loss Modelling; Research	NZ	
Met. Int. A	Science Manager	Meteorology	International	National
Met. Int. B	National Manager Disaster Mitigation Policy	Meteorology	International	National
Met. Int. C	Senior Policy Officer	Meteorology	International	National
Met. Int. D	Senior Social Scientist	Meteorology	International	National
Met. Int. E	Consultant Meteorologist	Meteorology	International	National
Met. NZ. F	Senior Meteorologist	Meteorology	NZ	National
Met. NZ. G	Communications	Meteorology	NZ	National
Met. NZ. H	Public Relations	Meteorology	NZ	National
Met. Int. I	Division Chief/Meteorologist	Meteorology	International	National
Met. Research NZ. J	Meteorologist	Meteorology; Research	NZ	National
Met. NZ. K	Senior Meteorologist	Meteorology	NZ	National
Met. Private NZ. L	Head Weather Analyst	Meteorology	NZ	
Risk Modelling NZ. A	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. B	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. C	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. D	Risk Modeller	Risk Modelling	NZ	National

3.6 Evaluation of Rigour and Credibility

Ensuring credibility and rigour in Grounded Theory research involves several measures of evaluation similar to those of other qualitative research methods. Corbin and Strauss (2008) provided a set of evaluation criteria which were used to ensure the credibility and rigour of this research, presented in Table 3.7.

Criteria	Description
	Do the findings resonate/fit with the experience of both the
Fit	professionals for whom the research was intended and the
	participants who took part in the study?
	The usefulness of the findings. Do the findings offer new explanations
Applicability	or insights? Can they be used to develop policy, change practice,
	and add to the knowledge base of a profession?
	Is there a logical flow of ideas? Do the findings "make sense"? Or are
	there gaps or missing links in the logic that leave the reader confused
Logic	and with a sense that something is not quite right? Are
	methodological decisions made clear so that the reader can judge
	their appropriateness for gathering data and doing analysis?
Denth	Are the descriptive details rich and varied, and do they lift the
	findings out of the realm of the ordinary?
	Has variation been built into the findings, meaning are there
Variation	examples of cases that don't fit the pattern or that show differences
	along certain dimensions or properties?
Creativity	Are the findings presented in an innovative manner? Does it say
	something new or put a twist on an old idea?
	Did the researcher demonstrate sensitivity towards the participants
Sensitivity	and to the data? Did the analysis drive the research or was the
	research driven by preconceived ideas?
	Since a researcher cannot possibly recall all the insights that go on
Evidence of	during analysis, memos are among the most necessary of all
Memos	procedures. Thus, there should be some evidence or discussion of
	memos in the final report.
	Concepts are necessary for developing common understandings and
Concepts	for professionals to talk among themselves, therefore one would
	expect that findings would be organized around concepts/themes.
Contextualisation	Without the context of the concepts, the reader cannot fully
of Concepts	understand why events occurred and ascertain the meaning of the
	experiences being described

Table 3.7. Criteria for evaluating qualitative research (Corbin & Strauss, 2008).

Triangulation and member-checking were carried out to ensure the fit and applicability of the findings. Triangulation is the use of multiple approaches to increase the credibility and validity of a study. Interviews, workshops, and document analysis were employed in this study (Creswell, 2003; Patton, 2002). First, three workshops were conducted after most interviews were conducted to collect diverse perspectives and receive feedback on the resulting HIVE data framework that was developed from the interviews (e.g., Figure 4.9). In these workshops, the data framework was presented to workshop participants who were asked to provide feedback through a set of pre-determined questions and activities. Document analysis

provided further validation against participants' statements and the researcher's own understanding of the DRR and EWS processes in NZ. For example, when participants referred to specific events like the 2017 Edgecumbe flood and the 2010-2011 Canterbury earthquake sequence, further documentation of these events was tracked down to verify the claims made by participants and to fill in any outstanding gaps from those discussions. In another example, documentation from LINZ (e.g., LINZ, 2014a, 2014b) and NEMA (e.g., NEMA, 2020a; NEMA, 2020c) were sought and used to clarify processes of data collection and management.

Member-checking tests the accuracy of research findings by presenting the findings to the participants for feedback (Stake, 1995). Member-checking was done by providing the results chapters to the participants for feedback on how their words were interpreted and presented. The participants were asked to return their feedback within two weeks and were informed that if they did not return feedback by the specified deadline, it was assumed that they had no feedback. Most of the participants who replied were pleased with how their words were interpreted and presented. Seven participants returned feedback which consisted of slight changes to their direct quotes, further elaboration or clarification on their quotes, re-interpretation of their words, and suggestions for terminology and literature references. All these suggestions were adopted into the final doctoral thesis.

Logic was ensured by requesting feedback from the supervisory team throughout the data collection, analysis, and writing phases. This provided supervisors with regular opportunities to identify critical gaps in data collection, analysis, and reporting that required further investigation and/or clarification. Furthermore, when 'new' ideas and concepts were identified from the analysis, the researcher consulted with the literature to provide clarity and additional understanding of these ideas. For example, when the concept of 'Individual and Community leadership' was found to be an action/interaction strategy for overcoming misdirected management priorities and lack of motivation or interest, the literature was consulted to verify this finding with existing theories; it was found that the Policy Capacity Framework provided a fitting explanation for this concept, as reported in Chapter Seven.

Variation was achieved through theoretical sampling and constant comparison such that the data and resulting codes were constantly compared against each other and against new data pieces (Strauss & Corbin, 1998). For example, in the early stages of conducting interviews, participants from meteorological services pointed to EM services for collecting impact data. Thus, EM agencies were targeted for subsequent interview recruitment to verify these suggestions and gain further understanding of EM agencies' data collection practices. EM agencies and meteorological services also pointed to regional councils and hydrologists for knowing more about flood impacts and warnings. As such, regional councils and flood hydrologists were recruited for further verification and validation. Cases that varied from the trend or pattern were picked up and investigated further. For example, while most participants were found to support the concept and goals of IFWs, one participant varied from this trend. This participant offered a different perspective of IFWs and provided their reasons for this variation. This variation was compared against the trend and further explanation was sought from the literature, as reported in Chapter Six.

The researcher strived to maintain sensitivity to the participants and to the data at every point in the research process. In the early stages of the research, when the researcher felt that a particular research solution was being forced without the strong presence of a research problem, the researcher made the critical decision to pivot the focus of the research to address a problem that was being identified by the research participants. Thus, the researcher practiced sensitivity to the participants and their contexts from this pivotal point until conclusion of this research.

Memos were written during initial research problem definition, data collection, and data analysis. These memos played a critical role in the researcher's reflections and explorations of emerging themes, concepts, and relationships.

The presentation of the findings is organised by phenomena and core category in the first order (i.e., research paper topic). In the second order, the findings pertaining to each phenomenon are organised and presented by theme and/or concept. Further context of these themes and concepts is provided in the reporting of the findings. Thus, this ensures that concepts and contextualisation of concepts are presented and discussed in detail.

3.7 Chapter Summary

This chapter described the overall design of the research by first justifying the choice of the Pragmatic paradigm to guide this doctoral research, followed by the methodology, which consists of the research approach, research strategy, data collection methods, and analytical techniques. A qualitative research approach was chosen for this exploratory study, using the Evolved-Straussian Grounded Theory research strategy. The next chapter presents the results of the data collection methods and overarching findings from the analysis.

Position	of the	reader
----------	--------	--------

	Chapter One		
Part 1	Introduction		
ιαιιι	Chapter Two		
	Background and Context		
	Chapter Three		
Part 2	Research Design		
Tartz	Chapter Four		
	Methods Results and Overarching Findings		
	Chapter Five		
	Volunteered Geographic Information for people-		
	centred severe weather early warning: A		
	literature review		
	Chapter Six		
	Identifying the data uses and gaps for severe		
	weather impact forecasts and warnings		
	Chapter Seven		
Part 3	Identifying data sources for severe weather		
	impact forecasts and warnings in Aotearoa New		
	Zealand		
	Chapter Eight		
	Data Governance and Access for Severe		
	Chapter Nine		
	Chapter Nine		
	Support IEWs and HIVE Data Collection Assess		
	and Use		
_	Chapter Ten		
Part 4	Discussion and Conclusion		

Chapter Four: Data Collection, Analysis, and Overarching Findings

This chapter presents the results of the data collection methods, and the overarching findings from the Evolved-Straussian Grounded Theory (ES-GT; Strauss & Corbin, 1990) analytical techniques employed in this PhD study. The purpose of this chapter is to demonstrate the use of the ES-GT data collection methods and analytical techniques in more depth than could be provided in the resulting manuscript chapters (Chapter Six, Chapter Seven, Chapter Eight, and Chapter Nine). The chapter first describes the process of collecting the data by conducting a series of interviews (Section 4.1) and workshops (Section 4.2) followed by presenting the results obtained after analysing the data collected from the interviews and workshops (Section 4.2.3). The chapter then introduces the two phenomena under study in this PhD research along with an elaboration of how they were identified through the axial coding process (Strauss & Corbin, 1990). Finally, the chapter presents the identified core category that resulted from the selective coding process (Corbin & Strauss, 2015). A chapter summary (Section 4.4) concludes this chapter.

4.1 Data Collection: Interviews

In this study thirty-nine (n=39) individuals were interviewed. The interview participants were recruited based on their roles and interactions with severe weather risk communication and hazard, impact, vulnerability, and exposure data, as shown in Table 4.1. Meteorological experts (e.g., Senior, consultant, and communication meteorologists, social scientists, senior policy advisors, and public safety managers) from National Meteorological Services (NMS) both inside and outside of New Zealand (NZ) were recruited. Officials from agencies outside of NZ were sought to understand international perspectives that may help implementation efforts in NZ. NZ-based leaders, chief operators, communications specialists, data specialists, and hydrologists were sought from the NZ Civil Defence and Emergency Management (CDEM) sector to understand local and regional practices in collecting and using hazard, impact, exposure, and vulnerability information. NZ-based experts in data creation, management, standards, and custodianship, and experts in risk and loss modelling were also sought. Furthermore, the theoretical sampling resulted in additional recruitment of one public health worker, one Lifelines group manager, and one agriculture policy coordinator.

The interviews ranged in length from 45 minutes to 1.5 hours. Interviews were audio-recorded and transcribed verbatim for further analysis. Due to the COVID-19 pandemic and the nature of the targeted participants, recruitment was limited towards the final months of data collection as many of the target participants were responding to COVID-19. Interviews were conducted both in person and virtually. Prior to COVID-19, in-person interviews were conducted with participants located in the same vicinity as the researcher (i.e., in Wellington, NZ), and virtual interviews were conducted with participants outside of Wellington using either the telephone or Zoom, depending on the participant's preference.

Category	n	General Roles	Representation
Hydro- meteorological	6	Senior, consultant, and communication meteorologists, social scientists, operational manager, senior policy advisors, and public safety managers	International: Argentina, Australia, Austria, United Kingdom, USA
Experts	6	Meteorologists, operational manager, communication meteorologist, senior policy advisor, Weather analyst	New Zealand: MetService, NIWA, Weather Watch
	1	Regional flood specialist	New Zealand: Greater Wellington Regional Council
Emergency Management and	3	First responder, Senior Hazard Risk Management Advisor	New Zealand National Level: National Emergency Management Agency (NEMA), Fire and Emergency New Zealand (FENZ)
Response Sector	9	Chief operators, communications specialists	New Zealand Regional Level: Auckland, Bay of Plenty, Canterbury, West Coast, Northland
	2	Geographic Information Systems (GIS) specialists	Private Industry: Eagle Technology, Independent
Data Creation, Management, Standards, Custodianship	3	GIS specialists, Statistical Analysis specialist	Public Sector: Land Information New Zealand, Stats NZ/DataVentures, West Coast
	1	GIS specialist	Research: Massey University
Risk and Loss	4	Risk modellers	Research and policy: GNS Science, NIWA
Modelling	1	Loss modeller	Research and policy: Victoria University of Wellington
	1	Public health worker	Practice: Regional hospital
Other	1	Civil Engineer	Practice: Wellington Region Lifelines Group
	1	Agriculture policy coordinator	Practice: Federated Farmers
Total	39		

Table 4.1. Summary of targeted and recruited interview participants.

During the NZ COVID-19 lockdown (Alert Level 4) from 25 March 2020 to 27 April 2020, and during Alert Level 3 (27 April 2020 to 11 May 2020) and Alert Level 2 (11 May 2020 to 8 June 2020; 12 August 2020 to 30 August 2020), all interviews were conducted virtually using either telephone, Microsoft Teams, or Zoom, depending on the participant's preference. During Alert Level 1 (8 June 2020 to 12 August 2020; 20 August to present), interviews were conducted according to the participants' location and preference (i.e., in person or virtually). From these interviews, following ES-GT, a theoretical framework was developed around the data needs, sources, and uses of

impact, vulnerability, and exposure data for IFW systems in New Zealand. This framework was then verified through the workshops.

4.2 Data Collection: Workshops

Three workshops were held in the final stages of the research. EM officials and MetService staff participated in two regional workshops for Auckland and Southland. A third workshop was conducted with researchers and scientists from GNS Science who are involved in hazard and risk management in NZ. Workshop participants were recruited via email using existing connections from previous interviews and the research team's network. The processes and results of the workshops are described next. All three workshops were conducted virtually due to uncertainty around changing COVID-19 restrictions. The virtual whiteboard platform, Mural, was used to facilitate all three workshops and capture participants' response.

4.2.1 Auckland Region Workshop

The first workshop was held with Auckland Council and Civil Defence on 12 October 2020. Auckland is the largest urban centre in NZ, located in the north of the North Island (Figure 2.6). More information on the hydrometeorological hazard profile of Auckland is provided in Appendix G. Auckland Council was chosen for the first workshop due to their experience with notable hydrometeorological events, such as the 2017 New Lynn storms and consequent flood, and the 2018 windstorm (Golubiewski, 2019; Smol, 2018). Furthermore, as a large metropolitan city, Auckland offers an urban context to this PhD study.

Four people attended the Auckland workshop, representing Auckland Civil Defence, the MetService, and Auckland Council from the Planning Department and the Healthy Waters Department. Every participant joined Mural individually while being connected through Microsoft Teams. This allowed for each participant to enter their own virtual sticky notes and contributions into the Mural 'whiteboard'. The participants were asked to assign a unique colour to their sticky notes and keep that colour for the duration of the workshop for ease of tracking responses, as shown in Figure 4.1. With each activity, participants were given time to enter their responses and share them with the group.

Auckland Impact Data & Impact-Based Forecast and



Figure 4.1. Final image of the populated Mural whiteboard from the Auckland Workshop on

I



12 October 2020.

Southland Impact Data & Impact-Based Forecast and



Figure 4.2. Final image of the populated Mural whiteboard from the Southland Workshop on

Ŧ

proactive messaging to help people and spare them having to ask

I Warning PhD Workshop 3 December 2020 Activity 2: Data Requirements & Sources Activity 5: Application 1. What data do we need for impact-based forecasts and warnings? 1. How would you use or apply this 2. What do we have already? framework? 3. What's missing from the list in the blue box in the framework? 4. What are some impediments and facilitators to accessing these data? We do apply Access to 3. The list is missing 4a. Access impediments a form of the 1. We need 2. We have 4b. Access facilitate more framework naturally datasets Hazards We have lots of forecast data but not much on the Exposure impacts data Vulnerability 2. Are there any aspects that you would adapt or change for your use? Aspect that I would How I would change it and Impacts change... why... Always 💵 Break Time (5 mins) 🎑 reasses as new data sets become Activity 3: System & Data Life available Uncertainty 1. Please pick 1-2 diverse datasets/data sources that are important and value - how identify the life track of these data. confident are What is the data used What happens to the data after its initial use? does the data Name the dataset we of a nodel result Straight to the ustomer (enviro data website), Stored for flood clata, recipitation mo from Metservic 3. Do you have any thoughts on Hydro flood historical stats, updating Data 1 hydro duty seam, EMS duty whether you would use impact-based modeling actual live data feeds, models, stored team forecasts and warnings? locally Public-facing: Push out warnings are possible, but We would Data 2 use them as should be used judiciously a baseline 2. How do you understand what impacts are occurring or could occur? I/we understand impacts that are occurring by... I/we understand impacts that could occur by... Why or why not? Can predict what Recon impacts could be See it from by what the data is telling us upstream public, Understanding Why? Why not? the live data contractors, of the hazards social media **Activity 4: Outputs & Outcomes** Wrap up From the datasets/data sources identified in activity 3.1, please Thank you! answer the following quesitons: 1. What are the outputs? Questions, comments, additional 2. How are the outputs produced and disseminated? thoughts? 3. What are the outcomes? Next steps and future rch plans.. Outputs Production Dissemination Outcomes Please Live feeds other ES enviro Modellina. Data 1 from in river website, in house to key thoughts: s.harrison@massey.ac.n change in monitors, return stakeholders (modeling and , risk analysis of period, the data live data) Data 2

3 December 2020.

4.2.2 Southland Region Workshop

The second workshop was held with Southland Council and Civil Defence on 3 December 2020. Southland is the southernmost region of NZ, located on the southwestern portion of NZ's South Island. Southland is considered a remote region, and thus provides a rural context. In February 2020, Southland experienced heavy rainfall, floods, and landslides. Southland Region was selected for the second regional workshop because of this event which resulted in the NZ MetService issuing its first Red Warning since the implementation of the new warning system (see Chapter Two). Furthermore, Southland is the southernmost region of NZ, located on the southwestern portion of NZ's South Island, providing a geographic spread for these regional workshops and a rural context. See Appendix G for more on the hazard profile of Southland.

Five people attended the Southland workshop, representing Southland Regional Council from the Geographical Information Systems (GIS) team, the Hydrology team, Communications and Engagement team, and Emergency Management Southland. All Southland participants coordinated with each other and joined the Microsoft Teams meeting from the same physical meeting room, sharing one keyboard between five participants. This was an unexpected outcome and made it difficult to record each participants' responses onto virtual sticky notes as the keyboard had to be shared until one person was anointed scribe to record the responses. While this resulted in less detailed responses on the Mural (see Figure 4.2), the discussion was found to be richer than that from the completely virtual Auckland workshop.

4.2.3 Community of Research Workshop

A third virtual workshop was conducted with a portion of the risk and hazards research community in New Zealand. Researchers and scientists from GNS Science were invited to participate in a virtual workshop to understand the organisation's current position and plans around IFWs. GNS Science is a Crown Research Institute (CRI) of New Zealand. The agency is heavily involved in the management of natural hazards risks across New Zealand by providing scientific evidence for response, planning, and policy (GNS Science, n.d.; Woods et al., 2017). The workshop was organised by Dr. Sally Potter and Dr. Danielle Charlton from GNS Science, and myself. The workshop design included a data collection activity for this PhD study where the participants were asked to share their feedback on the data process framework that resulted from previous interviews and workshops (see Section 4.3.1 for more). Eleven people from GNS Science attended the virtual community of research workshop

4.3 Data Analysis and Key Findings

Various tools and techniques were used following the ES-GT research strategy to guide the analysis of this PhD study, as outlined in Chapter Three. The following sections provide further details of the analytical process and key findings, including the ensuing codes, categories, phenomena, and core category. Integration of these analytical findings with literature is provided in the resulting manuscripts and chapters (Chapter Six to Chapter Nine).

4.3.1 Memo writing and diagramming

Memo writing and diagramming were two techniques regularly used throughout the data collection and analysis for this study. When the researcher noticed common themes emerging from the interviews, a memo was written to identify the theme and begin to understand the potential significance of the theme. For example, on 4 November 2019, a memo was created regarding the emergence of exposure:

Memo: Emergence of Exposure, Harrison, S., 4 November 2019

Interviewee 12 has a clear bias towards exposure for risk/impact modelling because they are working on a project to develop a population exposure model. However, I think this bias is not undue, and is valid because the literature around impact-based warnings lists exposure as one of the factors for impact-based warnings, and exposure is an important piece of the risk modelling puzzle. Fragility functions or vulnerability curves are based on asset exposure, be it buildings, infrastructure, or people.

From the discussion with my first participant, it has become clear that exposure, particularly exposure of people, is important yet difficult to capture. I think this is an important, and interesting area to explore.

Until this point in the data collection and parallel analysis, exposure and vulnerability were not considered within the scope of this research, with impact data being the sole focus. However, after writing this memo, returning to the literature for further verification, and discussing with supervisors, it was decided to expand the focus of this research onto exposure and vulnerability data. Following this decision, after more data was collected regarding exposure, another memo was written:

Memo: Emergence of Exposure, Harrison, S., 5 January 2021:

Since [4 November 2019], the concept of "dynamic exposure" has come up. I haven't been able to find a definition of it, but my guess is that it's the changing of exposure over space and time. We already know that exposure changes over space and time, so the term "dynamic exposure" might be a bit redundant. But this term seems to be used mostly in the risk modelling space, so perhaps the idea is to bring attention to the need for capturing the dynamic nature of exposure, and how to capture that? This is a challenge that has already been identified in the literature. Some ideas for capturing "dynamic exposure" are using mobile phones for population movement, social media, crowdsourcing. Data Ventures, a branch of Stats NZ, is playing with mobile phone data for COVID-19 analysis. There is potential to use this data for early warning systems as well.

Exposure is a tricky factor for IFWs. It is (arguably) an important factor for defining warning thresholds. But as my MetService forecaster participant described, exposure is different at all scales, and the warnings are not currently issued at all scales. So which scale do we pick for the warning thresholds? And how can we justify excluding people who may be exposed but don't meet the warning criteria, and thus do not receive a warning?

Upon identifying the theme of exposure, the researcher then compared physical and social impacts and exposure, which became another memo:

Memo: Physical vs. Social/Human Impacts and Exposure, Harrison, S., 4 November 2019:

As I discovered in my first interview, there's a clear emphasis on the physical world when talking about risk and impact data and assessments. The focus in New Zealand, to date, has been on capturing the physical impacts of hazards, such as earthquakes. This means a lot of emphasis on capturing the physical damage to buildings, service utility networks, and key infrastructure. There has been minimal attention paid to social impacts, aside from counting deaths and injuries. What about traffic delays? Evacuation routes and delays? Service outages?

When talking about data, for the past two interviews, much of the focus was on assets of buildings, service utilities networks, and key infrastructure. Discussion about population exposure is cropping up, though, and there's a clear emerging shift in the direction of risk and impact modelling towards modelling social impacts. But what are social impacts? It could be service outages and their effect on how people live their day-to-day lives, it could be evacuation routes and behaviours, traffic delays, etc.

What kind of impacts are we interested in for impact-based warnings? Just physical? or social as well?

This led the researcher to attempt to identify sources of social/human impacts and exposure in subsequent interviews, whereby Wellbeing surveys/welfare assessments were identified as a potential source for capturing social/human impacts:

Memo: Physical vs. Social/Human Impacts and Exposure, Harrison, S., 21 February 2020:

Welfare assessments that take place during an event, as mentioned by the participant in Interview 15, are a potential data source for social impacts. However, this data contains a lot of personally identifiable information (PII), that it is next to impossible to obtain this data. The most that could be obtained from this kind of information is a high-level summary of the results of the assessments,

as mentioned by participant from Interview 14 (from MCDEM/NEMA), and participant from Interview 15 (regional CDEM).

In line with these memos, interview questions specifically asked meteorologists and emergency management officials about using exposure and vulnerability for warnings and collecting exposure and vulnerability data.

Following interviews with CDEM Groups and meteorologists, diagrams were drawn to visualise the warning and data collection processes described by interviewees. For example, Figure 4.3 and Figure 4.4 are diagrams that were drawn following interviews with the West Coast CDEM Group and Emergency Management Bay of Plenty, respectively, to visualise and understand the severe weather warning processes within those groups. Similar diagrams were drawn following interviews with the other participating CDEM Groups and meteorological services. These diagrams were compared and digitised into a more generalised diagram of the severe weather warning chain, shown in Figure 4.5.

Diagramming was also performed to represent the data collection and use process in the context of IFWs based on interviews. This resulted in several iterations of an impact data framework. For example, Figure 4.6 shows the first iteration of the framework drawn on 8 October 2019. At this point, hazard, vulnerability, and exposure data were considered out of the scope of this research. Following the emergence of the exposure theme as documented in the memo writing, exposure data sets began to be identified, as shown in Figure 4.7. In October 2020 after a year of data collection via interviews, the framework became more filled in as shown in Figure 4.8.

The diagram in Figure 4.8 was presented to the workshop participants for feedback and validation. Based on feedback from the workshops and additional interviews, the framework was updated again and presented to the GNS workshop participants as shown in Figure 4.9. At this point, uncertainty about forecasting impacts was identified in interviews and workshops as a concern for IFWs. It was drawn in as an increasing wedge to represent the compounding uncertainty that begins from hazard forecast through to the warning. The diagram was also simplified to allow workshop participants to focus on key gaps in the framework which were represented as dashed lines, where the link from risk modelling to warning remained unclear.

Further memo writing and comparisons between the warning chain diagram and the data framework process diagram were conducted to determine how these two frameworks/concepts could be integrated. It was found that the impact data process framework focused heavily on the process for collecting and using impact, vulnerability, and exposure data for IFWs but excluded the initial stages of the process where the hazard is first detected, monitored, and forecasted. However, this hazard forecasting component was captured in the warning chain diagrams. Thus, integrating the warning chain diagram and the data process framework allowed for representation of the hazard forecasting component. This was done using the Warning Value Chain concept presented by Golding et al. (2019) as a guiding framework. These results are presented in Chapter Six (Harrison et al., 2022).



Figure 4.3. Diagram of West Coast warning process following an interview with the West Coast CDEM Groupon 22 April 2020.



Figure 4.4. Diagram of the Bay of Plenty severe weather warning processfollowing an interview with Emergency Management Bay of Plenty.



Figure 4.5. Diagram of severe weather warning chain based on interviews with NZ CDEM Groups and the MetService.



8 October 2019

Figure 4.6. First iteration of the impact data frameworkdrawn on 8 October 2019



Figure 4.7. Second iteration of the impact data framework following the emergence of the exposure theme, drawn on 7 November 2019.



Figure 4.8. Data process framework resulting from the Auckland workshop in October 2020.


and used in the Southland workshop in December 2020.



Figure 4.9. Impact data process framework presented to GNS workshop participants in February 2021.

4.3.2 Coding

The coding process of the ES-GT research strategy guided the analysis for this study. Results of the three coding phases (open, axial, and selective coding) are presented next.

4.3.2.1 Open coding

Open codes were developed directly from the data (i.e., in vivo), and with reference to literature and theories (Vollstedt & Rezat, 2019). For example, during the pilot interviews, open codes were created to classify the different sources of impact data, such as Twitter posts (i.e., Tweets), Facebook posts and Facebook comments, SnapChat posts, etc. Constant comparison took place by comparing the data pieces (e.g., sentences, paragraphs) for similarities and differences; similar parts were assigned the same code and different parts were assigned a different code (Vollstedt & Rezat, 2019).

The open coding process was documented with theoretical memos and in the code descriptions (Corbin & Strauss, 2015). For example, in the following memo trust in the data was identified as a key theme for being able to use data for IFWs:

Memo: High level data needs for IFWs, Harrison S., 6 January 2021:

The first round of interviews ... explored two things. The first was identifying the data needs for IFWs, and the second thing was exploring whether or not VGI (in the form of geo-located social media posts, geo-located crowdsourcing, and participatory mapping) would be useful for addressing these data needs.

During these interviews, a number of needs were identified for IFWs, but they were not specifically linked to particular datasets or data sources. They were more general data needs. The data needs are listed here ...

- the importance of and need for official data
- the need for quality control
- the need for real-time impact data
- the need to be able to trust the data
- the need for up-to-date information

These data needs are at a much higher level than I expected them to be, and they do not delve into the specific needs for exposure and vulnerability data. Perhaps because I did not specifically ask about vulnerability and exposure, although I know that I asked about it with my [redacted] participants. This tended to link more into the "data uses" theme, which draws out what the data are actually used for.

These needs are intrinsically linked to the socio-technical data challenges. Official data must be standardised and interoperable to enable the technical sharing and use of the data. But we also need agencies and organisations to trust each other so that they can build partnerships and share the data with each other. Quality control is supported by standardised and systematic data collection, and validation measures. Real-time impact data can potentially be collected via social media, crowdsourcing, media reports, but these data must be perceived as trustworthy. Trusting these datasets requires a cultural shift in the agencies, as I found in my Masters thesis (Harrison & Johnson, 2019). The need for up-to-date information is closely linked to the need for real-time impact data, and the need for quality control, and for official data. Official data is not very useful if it is not up-to-date or of a high quality. This is a big problem for the census data in Argentina and New Zealand.

From this memo, open codes of trust, official data, unofficial data, standardised data collection, systematic data collection, and real-time data were formed. A full list of open codes is provided in Appendix H.

These open codes were further grouped into categories using constant comparison and memo writing. For example, posts from the various social media platforms identified were grouped into a 'Social Media' category. The 'Social Media' and 'Crowdsourcing' categories were then grouped into 'Alternative or unofficial data sources', while another category was created for 'Official and Trusted Data Sources' to capture the data sources that interviewees deemed as such (e.g., CDEM-led damage surveys and impact assessments, emergency calls, and media reports). This relational process allowed for different dimensions of categories to be described (Vollstedt & Rezat, 2019), such as trustworthiness of the data.

4.3.2.2 Axial coding

The coding paradigm is a tool specific to the ES-GT research strategy intended to guide the researcher in the axial coding process (see Section 3.2.2.3). The coding paradigm provided by Strauss and Corbin (1990) and summarised in Table 3.5 was found to better fit the data than more recent variations of the coding paradigm. For example, this 1990 coding paradigm includes "Causal", "Intervening", and "Contextual Conditions" (Strauss & Corbin, 1990, p. 99), while the most recent variation of the coding paradigm only provides "Conditions" (Corbin & Strauss, 2015, p. 158). Causal, Intervening, and Contextual Conditions were found to better explain the data by allowing for drivers (i.e., causal conditions) for the phenomena (e.g., IFW systems and HIVE data) to be identified, along with challenges in progressing the phenomena (e.g., barriers to IFW implementation, challenges with collecting data). For example, past disaster events such as the 2010-2011 Canterbury earthquake sequence were identified in interviews as a driver (i.e., causal condition) for improving data collection and management within the NZ CDEM Sector. Contextual conditions were determined to be high level influences of overall practice, particularly governance and cultural conditions in NZ. The initial results of this process are presented in Figure 4.10.



Figure 4.10. Axial coding processwhere codes were grouped into categories and related to the coding paradigm dimensions on 16 April 2021 (Image Source: S. Harrison).

Results from the axial coding process were digitised into an Excel spreadsheet where further aggregation and shifting of codes and categories was completed and significant themes were highlighted for reporting. These initial results from 16 April 2021 are provided in Appendix I. Note that further memo writing, results writing, and analysis led to the refinement of these initial results (Corbin & Strauss, 2013), likely causing variation from the results in Appendix I and the results forthwith. Results of the axial coding process are presented next according to each dimension of the coding

paradigm: Phenomena, causal conditions, contextual conditions, intervening conditions, action/interaction strategies, and consequences.

Phenomena

Phenomena are the subjects or objects under study (Strauss & Corbin, 1990). Two overarching phenomena were identified as the outcome of the axial coding process. The first phenomenon to result from the axial coding process is *IFW Systems* (Figure 4.11), and the second is *Hazard, Impact, Vulnerability, and Exposure (HIVE) Data* (Figure 4.12). These two phenomena were then broken down into five categories based on the open code groupings.



Figure 4.11. The open codes and categories that make up the Impact Forecasting and Warning Systems Phenomenon.

Phenomenon 1: Impact Forecasting and Warning (IFW) Systems

Open codes were grouped based on similar characteristics and themes. For example, the codes 'Defining Impact Thresholds' and 'Using impact data to classify hazards', were grouped together because they both relate to the concept of identifying triggers or thresholds for IFWs. The codes for the IFW Implementation category relate to defining and understanding IFW systems and what is needed to implement them. Initially IFW Data Needs were grouped under IFW implementation, but it was elevated to its own category after all codes were consolidated and it was apparent that enough⁸ codes with a common theme exist for it to be a category. The codes in the IFW Data Needs category relate to identifying the different types of data needs for IFWs (e.g., hazard, impact, vulnerability, and exposure data), and justification and uses for these different data types in the IFW systems context.



Figure 4.12. The open codes and categories that make up the HIVE Data phenomenon.

Phenomenon 2: Hazard, Impact, Vulnerability, and Exposure (HIVE) Data

The Hazard, Impact, Vulnerability, and Exposure (HIVE) Data phenomenon consists of three categories: Data Sources and Collection, Data Access and Sharing, and Data Management and Governance (Figure 4.12). During the axial coding process, it was found that enough codes with a common theme existed to form these three categories. These three category topics arose organically by the participants and were discussed at length. Thus, they were deemed to be important enough to be their own categories. Furthermore, these three categories are inherently linked to each other. For example, in the Data Sources and Collection category, two codes emerged

⁸ Note that there is no pre-defined number of codes needed to constitute a category. A category is formed based on commonalities between open codes and concepts. It is the discretion of the Grounded Theory researcher to develop categories and their names based on the data (Corbin & Strauss, 2013).

regarding the existence of many datasets and sources: 'More data is available or exists than we think' and 'Various stakeholders that create, manage, share, access, use data'. These two codes led to the idea that if all of these datasets and sources exist and are collected by various sources, then there must be a way to share these data. As such, the data was then interrogated with questions around data sharing and access, resulting in codes identifying drivers for sharing data, and one particular in vivo code was produced: "Although we all complain about a lack of data, a lot of the time, there is quite a bit of data out there that could be used and isn't being used very well" (Risk Modelling NZ. A; see Table 3.6). This in vivo code resulted in questions around why these data are not being used very well, and this led to identifying challenges with accessing these data (i.e., intervening conditions), as well as issues of Data Management and Governance, where the data are not managed or maintained very well, making it difficult to share and effectively use the data. Questions were asked around the potential causes of data management and governance issues that are inhibiting HIVE data collection and sharing practices in NZ.

Contextual Conditions

Contextual conditions are conditions or characteristics surrounding the phenomena and resulting in action/interaction strategies taken to address the two phenomena (e.g., IFW Systems, and HIVE Data) (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019). Cultural Conditions, Governance, and NZ Warnings were the three categories identified as contextual conditions, as shown in Figure 4.13. These were identified as contextual conditions because they provide a background understanding of the systems in place that influence actions and decisions in practice and operations. For example, when attempting to understand current barriers and inhibitors to data collection, access, and sharing, the contextual conditions of top-level government priorities and investment provided clarity; participants indicated that top-level government priorities, which are embedded in governance structures, influence directions of research and practice (e.g., Smith et al., 2003).



Figure 4.13. Contextual conditions identified from axial coding.

Causal Conditions

Causal conditions are events that influence phenomena or result in the appearance or development of a phenomena. The causal conditions were identified as events that directly drove the need for implementing IFWs (the first phenomenon), and the collection, sharing, and management of HIVE data (the second phenomenon). The four main causal conditions for these phenomena, shown in Figure 4.14, are: (1) Disaster or emergency events, (2) The Need to Improve Warnings, (3) Research, and (4) Technological Advancements. Disaster and emergency events, the need to improve warnings, and research efforts were found to be important drivers for incorporating impact messaging into severe weather warnings, while research, technological advancements, and disaster and emergency events were identified as drivers for collecting, creating, sharing, and managing HIVE data.



Figure 4.14. The causal conditions identified in the axial coding process.

Intervening Conditions

Intervening conditions are unexpected events or factors leading to action/interaction strategies (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019). Intervening conditions were identified as inhibitors and facilitators that influence the actions that are required to implement IFWs (the first phenomenon; Figure 4.15), and to collect, access, and use HIVE data (the second phenomenon; Figure 4.16).

Three categories of intervening conditions resulted from grouping the open codes for the IFW Systems phenomenon, as shown in Figure 4.15: (1) IFW Data Usability and Needs; (2) Challenges with applying the knowledge, data, and information; and (3) Uncertainty.

IFW Data Usability Needs relates to the needs that participants identified for them to use data or sources of data. For example, participants indicated that they need to be able to trust the data and be confident in resulting analyses. They also need the data to be timely, identifying real-time or near real-time data as particularly useful. Challenges were identified with applying the various types and sources of knowledge, data, and information. For example, differences in spatial scale of warnings and available data make it difficult to use the data. Uncertainty was found to be a notable challenge to implementing IFWs. Uncertainty appeared to be present not only in the operational aspects of formulating, issuing, and communicating IFWs, but also in roles and responsibilities, and in the terminology used for IFWs.



Figure 4.15. Intervening conditions identified relating to the IFW Systems phenomenon.

Six categories emerged to represent intervening conditions affecting the second phenomenon, HIVE Data, as shown in Figure 4.16. These six categories are: (1) Data Collection Challenges and Considerations, (2) Data Management Challenges, (3) Social challenges to sharing data, (4) Ethical Considerations, (5) Technical challenges to sharing data, (6) Alternative or unofficial data challenges. Data Collection Challenges and Considerations consists of codes outlining various challenges or issues with collecting HIVE data, such as lack of motivation or interest in doing so or trying to find a balance between collecting too much and enough data to prevent information overload.



Figure 4.16. Intervening conditions identified relating to the HIVE Data phenomenon.

Data Management Challenges involve issues with managing the resulting data, which includes concerns with maintenance, costs, and responsibilities. Both social and technical challenges were identified for sharing data. Social challenges relate to partnerships and trust, while technical challenges relate to processes for integrating datasets/sources. Ethical Considerations were first identified as part of a social challenge to sharing HIVE data, but it was elevated to its own category because of the number of code groupings relating to it. Alternative or Unofficial Data Challenges were also elevated to their own category because the challenges identified were specific to alternative and unofficial data sources and could not be applied to official data sources or datasets. Furthermore, there is a growing interest in using alternative/unofficial data for IFWs and beyond (e.g., Spruce et al., 2020; Spruce et al., 2021), thus it was decided that this topic should have its own category.

In addition to the above identified intervening categories, some of the open codes relating to the contextual condition of Governance were found to influence both IFW Systems and HIVE Data phenomena, and thus could not be associated with just one of these phenomena. As such, they were elevated to their own category of 'Governance Challenges', shown in Figure 4.17. For example, roles and responsibilities are a point of confusion for both the IFW Systems and the HIVE Data phenomena. The NZ MetService and CDEM Groups both questioned who was responsible for conveying impact information and messaging in severe weather warnings, because the MetService is the mandated authority for issuing severe weather warnings, but they do not have the impact knowledge that CDEM Groups possess. With regards to HIVE Data, questions arose around data custodianship and who should take on that role for HIVE data, particularly impact data. These questions of roles and responsibilities link back to governance, as legislation and policies typically identify the roles and responsibilities of each agency.



Figure 4.17. Intervening conditions relating to governance.

Action/Interaction Strategies

The original definition of action/interaction strategies in the ES-GT coding paradigm is 'purposeful and deliberate acts that are taken to address the phenomena' (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019). However, during the axial coding phase for this study, it was found that the action/interaction strategies fit more appropriately with acts taken to address the intervening conditions that arose from the phenomena. For example, action/interaction strategies were identified to address some intervening conditions regarding data collection, such as applying quality control measures to ensure credibility and quality of the data. As such, the definition of action/interaction strategies was expanded for this study to apply to those actions/interactions that were either implemented or suggested to address intervening conditions, as well as action/interaction strategies addressing contextual conditions, such as governance.

Three categories were identified for action/interaction strategies relating to IFW Systems and the associated intervening conditions of this phenomenon. As shown in Figure 4.18, these categories are: (1) Defining and setting thresholds, (2) Strategies for improving warnings, and (3) Strategies for addressing uncertainty. Action/interaction strategies for defining and setting thresholds for IFW systems centred around partnerships, sharing data, and joint decision-making, because an intervening condition of IFW system implementation was the variety of stakeholders that IFW systems are designed for which require different warning thresholds (see Figure 4.15).



Figure 4.18. Action/interaction strategies identified to address intervening conditions and other characteristics of the IFW Systems phenomenon.

Partnerships, sharing data, and joint decision-making were actions identified for addressing this challenge. Several strategies were identified for improving warnings, not all related to IFWs. For example, the NZ MetService's shift from text-based messages to using images in warning messages was a strategy employed for improving warning communication and understanding. One cause of uncertainty was identified as the lack of data, knowledge, and expertise in forecasting and warning for impacts. As such, collecting more of the appropriate data in a form that makes the data usable for impact forecasting was identified as an action/interaction strategy to address uncertainty.

For the HIVE Data phenomenon, action/interaction strategies relating to the intervening conditions were grouped into two themes: (1) Data collection; and (2) Data sharing, management, and Integration. The 'data collection' action/interaction strategies pertains to the two intervening conditions of 'data collection challenges and considerations' and 'alternative or unofficial data challenges previously' previously presented in Figure 4.16. The 'data sharing, management, and integration' action/interaction strategies pertain to the intervening conditions of 'social challenges to data sharing,' 'technical challenges to data sharing,' and 'ethical considerations.' These intervening conditions were grouped into one action/interaction strategy theme because they were found to relate to each other and influence each other. For example, data integration depends on data management practices (Ludäscher et al., 2006).

Three categories were identified for action/interaction strategies pertaining to the 'data collection challenges and considerations' and 'alternative or unofficial data challenges' relating to the HIVE Data phenomenon. These categories, shown in Figure 4.19, are (1) Data collection strategies, (2) Identifying data sources, and (3) Alternative or unofficial data strategies. Data collection strategies centred around improving data quality and implementing measures to ensure that the right balance is achieved for data quality and quantity. Identifying data sources was an action/interaction strategy to address the question of data sources and where existing data are. Strategies for addressing concerns with alternative or unofficial data sources also centred around ensuring data quality and credibility.

Another three categories were identified for action/interaction strategies relating to 'data sharing, management, and integration' in the HIVE Data phenomenon. As shown in Figure 4.20, these are: (1) Building partnerships for better data sharing practices, (2) Data management and integration strategies, and (3) Data standardisation strategies. 'Building partnerships for better data sharing practices' consisted of codes relating to strategies for building and nurturing partnerships, strengthening collaboration, and improving coordination. Strategies for 'data management and integration' identified technical tools to support these actions, such as building data directories, using Geographic Information Systems (GIS) and spatial data, and utilising machine learning and artificial intelligence.



Figure 4.19. Action/interaction strategies identified to address data collection intervening conditions.



Figure 4.20. Action/interaction strategies identified for addressing data sharing, management, and integration intervening conditions.

Finally, two categories were identified for action/interaction strategies relating to governance contextual conditions (e.g., strategies for driving innovation and change) and technological advancement causal conditions (e.g., 'push and pull' system design strategies), as shown in Figure 4.21. 'Strategies for driving innovation and change' were found to be influenced by leadership, political direction, bottom-up organisation, and organisational culture. 'Push and pull system design strategies' involve conducting needs-based research and development, focusing on the users and uses of the data, and being innovative.



Figure 4.21. Action/interaction strategies for addressing governance contextual conditions and technological advancement causal conditions.

Consequences and Outcomes

Consequences and outcomes are the predictable or unpredictable, intended, or unintended, positive or negative outcomes of the action/interaction strategies (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019). The resulting outcome categories were organised by the phenomena and contextual conditions.

Three categories were identified for the outcomes and consequences of implementing IFW systems (i.e., the IFW Phenomenon). As shown in Figure 4.22, these categories are: (1) IFW outcomes, (2) IFW Products, and (3) Consequences of uncertainty. 'IFW outcomes' relates to the outcomes from the impact forecasts and warnings themselves, such as potential or observed changes in warning audience decision-making and behaviour. 'IFW Products' are meteorological products and services that were identified by participants that had some aspect of impact forecasts and/or warnings. 'Consequences of uncertainty' related to reasons for warning services' and agencies' hesitancy to issue IFWs and the perceived increased potential for audiences and stakeholders to misunderstand impact models, forecasts, and warnings.



Figure 4.22. Consequences and outcomes of implementing IFW systems.

Mirroring the action/interaction strategies for the HIVE Data phenomenon, outcomes were grouped into two themes: (1) Data collection, and (2) Data sharing, management, and integration. Three categories were identified for outcomes and consequences relating to HIVE data collection, as shown in Figure 4.23. These are (1) Impact databases, (2) Data uses, and (3) Alternative or unofficial data risks and benefits. 'Impact databases' are products of data collection efforts carried out by agencies and stakeholders identified by participants. 'Data uses' are IFW and non-IFW uses that were identified for the HIVE data. The risks and benefits of collecting and using alternative or unofficial data sources were also identified. Risks in this context differ from intervening conditions because they are an observed outcome of using this data or of people creating or contributing to this data. For example, the code 'people interfering in the response and putting themselves and others at risk' is an observed behaviour of people taking photos and videos during a disaster or emergency event to post on social media or contribute to other crowdsourcing platforms (EM. NZ. Nat. I).



Figure 4.23. Consequences and outcomes of collecting HIVE data.

Two categories were identified for outcomes of 'data sharing, management, and integration,' the second theme of the HIVE Data phenomenon. These are (1) Social data sharing outcomes, and (2) Technical data sharing outcomes, as shown in Figure 4.24. These outcomes resulted from applying a socio-technical (Ghaffarian, 2011) lens to the analysis, where both technical and social aspects of data sharing were identified and explored. Social data sharing outcomes relate to the social outcomes, benefits, and risks of sharing data, such as affecting property values (risk), increasing transparency (benefit), facilitating a community of knowledge (benefit), etc. Technical data sharing outcomes relate to technical products and innovations that resulted from sharing data, such as the Common Alerting Protocol, the Common Operating Picture, online web maps, etc.



Figure 4.24. Consequences and outcomes of data sharing, management, and integration.

Finally, one category of outcomes and consequences relating to governance contextual conditions, specifically 'policy, decision-making, and practice' was identified. The open codes for this are shown in Figure 4.25. These outcomes are not specific to IFW System implementation but apply to overall Disaster Risk Reduction (DRR) policy, practice, and governance. This is because IFW implementation involves many stakeholders beyond the meteorological service (e.g., Hemingway & Gunawan, 2018), and the collection, use, sharing, and governance of HIVE data expands beyond IFW system implementation, and is needed for other aspects of DRR (e.g., UNDRR, 2015b). For example, one overarching theme was found to be that of reactive vs. proactive EM practice, whereby EM practice tends to be more reactive by traditionally focusing on emergency response and recovery (Kox, Lüder, et al., 2018). This reactive practice has resulted in poor data collection and management practices to meet immediate needs, which can then reduce the availability and accessibility of HIVE data for IFW system implementation. Alternatively, proactive EM practice where processes have been pre-established for data collection, access, and sharing in advance of a disaster event can enable more effective and efficient data collection, access, and sharing.



Figure 4.25. Consequences and outcomes related to governance.

4.3.2.3 Selective Coding

During the selective coding phase, decisions were made regarding the inclusion and exclusion of codes and categories based on their explanatory power and based on the objectives of this research (Corbin & Strauss, 2008). For example, cultural conditions were initially identified as a contextual condition to provide contextual understanding of the other categories. While cultural conditions offer an interesting and complex layer to this research, it was determined that further exploration of this category did not align with the research objectives, and as such no further data was collected for this category. Thus, explanatory power and relevance of this category was diminished.

Through selective coding, the core categories were also identified. A core category is that which "incorporates or supersedes other categories in explanatory importance and hence is 'elevated' to the status of an important concept" (Timonen et al., 2018, p. 7). Two core categories were initially identified in this research because of their influence and relation to each category identified in the analysis. The two core categories are: (1) Governance, and (2) Partnerships and Collaboration.

While governance was initially classified as a contextual condition to provide contextual understanding of the other categories, it was found that governance plays a critical role in the implementation of IFWs and in the collection, use, accessibility, and sharing of HIVE data. The underlying influence of each phenomenon and the categories resulting from the axial coding process pointed to some form of governance. For example, in trying to understand the intervening conditions of IFW implementation, governance was identified as a major influencer of role designation for EWSs. The introduction of new or shifting roles and responsibilities for IFW systems is a point of confusion that is interfering with the implementation of IFWs in New Zealand (Potter et al., 2021). With regards to HIVE Data, governance influences the priorities of data collection, and the roles of data custodianship and management. These in turn influence how and what data is collected, used, and shared. However, as with the cultural contextual condition, Governance is a topic area deserving of its own doctoral thesis. Thus, it was determined that there was not enough capacity to explore this as a core category in this PhD study, and it is identified as an area for future research.

Partnerships and Collaboration was identified as the second core category, as it too threads throughout the themes and categories resulting from the analysis. Partnerships and Collaboration appear to be essential for both IFW Systems and HIVE Data. This is shown by the open codes in Table 4.2 and visualised in Figure 4.26. For the IFW System Phenomenon, defining impact thresholds is required for IFWs. Discussions with participants showed that the process of defining impact thresholds requires partnering with agencies that possess the kind of knowledge needed to define these impact thresholds. Such agencies include transportation agencies, flood management agencies, health agencies, insurance agencies, etc. Table 4.2. Open codes about Partnerships and Collaboration relating back to the two phenomena that are the focal points of this study: IFW Systems, and HIVE Data.

Phenomenon	Relevant Partnerships and Collaboration Open Codes
IFW Systems	Nodes\\IBFW Implementation\Impact Thresholds\Defining Impact
	Thresholds\Partnering with agencies for impact knowledge
	Nodes\\Data for IBFWs and General DRR\Data and Information
HIVE Data	Sharing\Drivers for sharing data\Various stakeholders that create,
	coordination
	Nodes\\Data for IBFWs and General DRR\Data and Information
	Sharing Requirements and practices of sharing data Cooperation and collaboration
	Nodes\\Data for IBFWs and General DRR\Data Collection and
	creation\Drivers for IVE Data Collection\Various stakeholders that
	and coordination to establish data collection needs and standards
	Nodes\\Partnerships, Collaboration, and Relationships
	\Agencies running profession development workshops to build
	relationships
	\Cohabitation
Other	\Community of knowledge
	\Informal Partnership
	\MetService-NIWA
	\Regional flood group
	\Science advice groups

For the HIVE Data phenomenon, the concept of Partnerships and Collaboration emerged in two categories: (1) Data and Information Sharing, and (2) Data Collection and Creation. In the first category, 'Data and Information Sharing', the need for cooperation and collaboration was identified as a strategy for facilitating data sharing between the various agencies that possess the data. In the second category, 'Data Collection and Creation', the need for collaboration and coordination was identified for establishing data collection needs and standards across the different agencies that collect the data or need the data. The need for this arises from the idea of agencies collaborating with each other after an event to co-design data collection forms, etc. such that the data is collected to suit various users' needs (Met. Research NZ. J; Risk Modelling NZ. C).

Other codes around partnerships and collaboration were created outside of the direct relation to these two phenomena and are listed in Table 4.2 as 'Other' phenomena which was elevated to the core category of Partnerships and Collaboration. After establishing this category, existing codes were identified relating to this category and relationships were established, as portrayed by the dashed arrows in Figure 4.26.



Figure 4.26. Relationships between the two phenomena under study and the core category. The boxes represent codes developed throughout the analysis. The blue boxes represent the IFW System phenomenon and associated categories and codes. The green boxes represent the HIVE Data phenomenon and associated categories and codes. The grey boxes represent the core category of Partnerships and Collaboration. The solid arrows portray the coding hierarchy. The dashed arrows represent the relationships between the two phenomena and the core category.

The value of forming partnerships and collaboration is further exemplified in the following memo about the New Zealand Geographic Information System for Emergency Management (NZGIS4EM) group, a grassroots community of GIS specialists and EM practitioners in NZ that began working together in the mid-2010s to identify how they could build and use geospatial tools for better emergency response. The following memo identified the ability to work together to drive innovation for GIS in EM as a true strength of the NZGIS4EM group.

Memo: NZGIS4EM, Harrison, S., 13 January 2021:

The NZGIS4EM group is a big driver for technological advancement, for pushing the needle forwards on sharing, accessing, and creating geospatial data and tools for emergency response. This is a great example of how people and groups can work together to get the data to where it needs to go in the format that it needs to be in.

Memo: NZGIS4EM, Harrison, S., 21 January 2021:

Great words that describe NZGIS4EM (from Interviewee 19): relationships and community. These are the core values of NZGIS4EM (from my perspective); or rather, it's the true strength of this group of volunteers for fostering this community and building relationships across organisations, which then drives innovation, knowledge sharing, and data sharing.

4.4 Chapter Summary

This chapter provided an overview of the data collection results and findings from the analytical techniques employed according to the ES-GT research strategy guiding this PhD research. Thirty-nine individuals involved in hydrometeorological warning systems and DRR both within and outside of New Zealand were interviewed for this study, with an additional twenty individuals participating in a series of virtual workshops.

Following the ES-GT research strategy, three stages of coding (open, axial, and selective coding) formed the analytical process, accompanied by memo writing and diagramming. The coding paradigm provided by Strauss and Corbin (1990) aided the axial coding stage of this research. As a result of employing the coding paradigm, two phenomena were identified that made up the main subjects under study for this PhD research. These two phenomena are IFW Systems and HIVE Data. As such, following an exploration of the literature on the potential use of Volunteered Geographic Information (VG) for IFWs presented in Chapter Five, the results of this thesis are organised by phenomenon, as represented in Figure 4.27. Chapter Six presents the results specific to the IFW implementation phenomenon, and Chapter Seven and Chapter Eight present results specific to the HIVE Data phenomenon. Finally, the core category of Partnerships and Collaboration resulting from the selective coding process is defined in Chapter Nine and integrated with the results presented in Chapter Eight.



Figure 4.27. Links between the two phenomena and the resulting core category for this PhD studyand the corresponding results chapters.

Part 1	Chapter One		
	Chapter Two		
	Background and Context		
Part 2	Chapter Three		
	Research Design		
	Chapter Four		
	Methods Results and Overarching Findings		
	Chapter Five		
	Volunteered Geographic Information for people-		
	centred severe weather early warning: A		
	weather impact forecasts and warnings		
	Chapter Seven		
	Identifying data sources for severe weather		
Part 3	impact forecasts and warnings in Aotearoa New		
i dit o	Zealand		
	Chapter Eight		
	A socio-technical analysis of sharing and		
	governing hydrometeorological impact data in		
	Aotearoa New Zealand		
	Chapter Nine		
	Integration of Partnerships and Collaboration to		
	Support IFWs and HIVE Data Collection, Access,		
	and Use		
	Chapter Ter		
Part 4	Chapter Len Discussion and Conclusion		

Position of the reader

Chapter Five: Volunteered Geographic Information for people-centred severe weather early warning: A literature review

This chapter presents the first manuscript for publication: a literature review of the role of Volunteered Geographic Information (VGI) for EWSs. The initial topic of this doctoral project was to explore alternative sources of impact data for IFWs, including crowdsourcing, citizen science, and social media. VGI was chosen as the alternative data type to explore due to its geospatial nature, which was found to be a necessary component of impact data. VGI encompasses crowdsourced data, citizen science data, and social media data that contain geospatial components of the data (i.e., location information). This literature review was performed to explore the role of VGI in EWSs, and to achieve Objective 1.1. of this thesis: To establish VGI as a potential source of impact data for IFWs.

The article presented in this chapter was published in 2020 in the *Australasian Journal of Disaster and Trauma Studies*, as:

Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E. H., & Johnston, D. (2020).
Volunteered Geographic Information for people-centred severe weather early warning: A literature review. *Australasian Journal of Disaster and Trauma Studies*, 24(1). http://trauma.massey.ac.nz/issues/2020-1/AJDTS_24_1_Harrison.pdf
The published form is provided in Appendix J.

5.1 Abstract

Early warning systems (EWSs) can prevent loss of life and reduce the impacts of hazards. Yet, recent severe weather events indicate that many EWSs continue to fail at adequately communicating the risk of the hazard, resulting in significant life and property loss. Given these shortcomings, there has been a shift towards people-centred EWSs to engage with audiences of warnings to understand their needs and capabilities. One example of engaging with warning audiences is through the collection and co-creation of volunteered geographic information (VGI). Much of the research in the past has primarily focused on using VGI in disaster response, with less exploration of the role of VGI for EWSs.

This review uses a scoping methodology to identify and analyse 29 research papers on EWSs for severe weather hazards. Results show that VGI is useful in all components of an EWS, but some platforms are more useful for specific components than are others. Furthermore, the different types of VGI have implications for supporting people-centred EWSs. Future research should explore the characteristics of the VGI produced for these EWS components and determine how VGI can support a new EWS model for which the World Meteorological Organization is advocating: that of impact-based forecasting and warning systems.

5.2 Introduction

Early warning systems (EWSs) can prevent loss of life and reduce the impacts of hazards by providing members of the stakeholders and the public with information about likely, imminent risks on which they can act to prepare themselves and their property. As such, they have been a focus of disaster risk reduction since the Hyogo Framework for Action 2005-2015 through to the current Sendai Framework for Disaster Risk Reduction 2015-2030 (UNDRR, 2015b; UNISDR, 2005). EWSs are described as having four key operational components: Disaster Risk Knowledge; Detection, Monitoring, and Warning Services; Communication and Dissemination Mechanisms; and Preparedness and Response Capacity (see Figure 5.1; Basher, 2006; Golnaraghi, 2012).



Figure 5.1. Four operational components of an Early Warning System. Adapted from Basher (2006); Golnaraghi (2012); WMO (2018).

The first component, Disaster Risk Knowledge, involves systematically collecting and analysing data related to risk, such as the exposure and vulnerability of people and infrastructure to nearby hazards (Ahmed et al., 2012; Basher, 2006; Sai et al., 2018). This involves assessing risk and vulnerability, building evacuation plans, and tailoring warning systems. Detection, Monitoring, and Warning Services make up the second component and are central to EWSs. This component requires reliable technology and involves continuous, automated detection and hazard monitoring (Ahmed et al., 2012; Basher, 2006; Sai et al., 2018). Furthermore, data, forecasts, and warnings should be archived for post-event analysis and for continual system improvements (Ahmed et al., 2012; Basher, 2006; Sai et al., 2006; Sai et al., 2018). Impact data collected during and after a severe weather event would support both of these first two components (Harrison et al., 2015).

The third component of an EWS is Communication and Dissemination, which is needed to reach those at risk. This involves using clear, concise, and understandable messages to enable proper preparedness (Ahmed et al., 2012; Basher, 2006; Sai et al., 2018). Multiple communications channels are necessary to reach as many people as possible (Ahmed et al., 2012; Basher, 2006). The fourth component of an EWS is Preparedness and Early Response Capacity. This involves running education and preparedness programmes to help people "understand their risks, respect the national warning services, and know how to react to warning messages" (WMO, 2018, p. 6). All four components of an EWS play a key role in crisis and risk communication.

EWSs share common characteristics with crisis and emergency risk communication theory. Like EWSs, the goal of crisis and risk communication theory is to provide sufficient and appropriate information to stakeholders that would allow them to "make the best possible decisions about their well-being" in a short period of time under uncertainty (Reynolds & Quinn, 2008, p. 14S). This involves understanding stakeholder (including the public) perceptions of risk and of the effectiveness of response, understanding the needs, capabilities, experiences, and predispositions of the stakeholders, and formulating messages based on these understandings for different audiences throughout the stages of crisis (Morgan et al., 1992; Reynolds & Seeger, 2005; Veil et al., 2008). Crisis and emergency risk communication theory is applied in risk messaging, crisis messaging, and warnings for health and emergency situations including, but not limited to, disease outbreaks, bioterrorism, hurricanes, and tornadoes (Reynolds & Seeger, 2005). The EWS framework presented in Figure 5.1 is thus supported by objectives of crisis and emergency risk communication theory, although the EWS framework does not include an apparent consideration for two- way communication: a key component in crisis and emergency risk communication theory for evaluating the effectiveness of communication (Garcia & Fearnley, 2012; Veil et al., 2008).

Recent severe weather events indicate that many EWSs continue to fail at adequately communicating the risk (and associated impacts) of the hazard, resulting in significant life and property loss due to limited understanding of, and response to, warnings (Ching et al., 2015; Wagenmaker et al., 2011; WMO, 2015). As such, there has been a push for "people-centred" EWSs to bring the "human factor" into consideration when designing and implementing EWSs and issuing warnings.

5.3 People-Centred Early Warning Systems

The broader EWS literature has recognised a communication gap between warning services and warning recipients, resulting in target audiences taking inadequate protective action despite receiving warnings (Anderson-Berry et al., 2018; Basher, 2006; Weyrich et al., 2018). In 2006, Basher introduced the concept of people-centred EWSs to address the "human factor" in EWSs, as he stated "failures in Early Warning Systems typically occur in the communication and preparedness elements" (Basher, 2006, p. 2168). Since then, there has been a shift towards people-centred EWSs which are developed for, and with, the target audiences to identify their needs and capacities and to transfer responsibility back to the audience to take protective

actions (Basher, 2006; Scolobig et al., 2015). The United Nations Office for Disaster Risk Reduction (UNDRR; formerly known as the UNISDR) listed "investing in, developing, maintaining and strengthening people-centred multi-hazard, multisectoral forecasting, and Early Warning Systems" as an objective towards meeting the fourth priority of the Sendai Framework (UNDRR, 2015b, p. 21). This "people-centred" aspect involves incorporating local and indigenous knowledge about hazards, promoting and applying low-cost EWSs that are appropriate to the audience based on their needs and capabilities, and broadening information channels (UNDRR, 2015b; WMO, 2018). According to the Sendai Framework, people-centred EWSs can be developed through engagement with the audiences of warnings (e.g., individuals, communities, sectors: UNDRR, 2015b; WMO, 2018). One such example of engaging with warning audiences and understanding their needs and capabilities is through volunteered geographic information (VGI; WMO, 2017). VGI is information produced by or gathered from the public with associated locational attributes. The locationbased information from VGI allows officials to identify high-risk areas, populations, and infrastructure (Goodchild & Glennon, 2010; Granell & Ostermann, 2016; Haworth, 2018; Roche et al., 2011).

5.4 Volunteered Geographic Information

VGI is valuable to disaster management because disasters are inherently location- and time-dependent and the location information from VGI allows officials to understand where the high-risk areas and populations are (Goodchild, 2007; Goodchild & Glennon, 2010; Granell & Ostermann, 2016; Haworth, 2018; Roche et al., 2011). The broader literature body around VGI, crowdsourcing, citizen science, and social media discusses and debates the relationship of these terms to each other and their associated characteristics and differences. It is argued that VGI overlaps both with citizen science and crowdsourcing (Cooper et al., 2018; Haklay, 2013, 2017). In Haklay's (2013) typology, crowdsourcing is classified as the lowest level of participation in citizen science. Citizen science (including crowdsourcing) is considered VGI when the information produced through the differing levels of participation includes geographic information (Haklay, 2017).

VGI can be collected in various ways, producing different types and formats of data. From reviewing the VGI and disaster risk reduction literature, we identified four types of VGI that are generally produced and/or collected for disaster risk reduction; these are summarised in Table 5.1. Geo-located social media refers to VGI that is posted online by social media users that has associated geographical location information. The term social media recognises online blogs, micro-blogs, online social networking, and forums, which enable sharing of text, audio, photographs, and videos (Alexander, 2014). Facebook, Twitter, Sina Weibo, WeChat, Instagram, and SnapChat are some examples of popular social media platforms. During a severe weather event, authorities can use social media to disseminate alerts and warnings and collect information from members of the public about the event and its impacts (Alexander, 2014; Goodchild, 2007; Harrison & Johnson, 2016; Porto de Albuquerque et al., 2017; Roche et al., 2011; Simon et al., 2015; Slavkovikj et al., 2014).

Table 5.1. Summary of VGI types.

VGI Process	Spatial Data Format	Data type	Data Sources	Disaster Risk Reduction Phase	Analysis/Outcomes
Geo-located social media harvesting	Point data	Impact data, exposure data, vulnerability data, hazard data Photos, videos, text	Facebook, Instagram, Twitter, Snapchat, Flickr, Sina Weibo, etc.	All	Cluster analysis, early detection, situational awareness, post-event damage/impact assessment, response coordination
Crowd-sourcing	Point data	Impact data, exposure data, vulnerability data, hazard data Photos, videos, text	Online reporting forms, mobile application	Readiness, Risk Reduction, During, Response	Cluster analysis, early detection, situational awareness, damage/impact assessment, response coordination
Participatory mapping / Participatory GIS	Point, line, polygon	Impact data, exposure data, vulnerability data, hazard data, expert local knowledge Shapefiles	Community members, community leaders, stakeholders	Readiness, Risk Reduction, Recovery	Hazard and risk assessments/modelling, impact forecasting, customise/personalise warnings systems for the community, identify impact thresholds, inform/improve readiness and reduction efforts based on local knowledge
Local Knowledge	Point, line, polygon, written, audio	Impact data, exposure data, vulnerability data, hazard data, expert local knowledge Shapefiles	Community members, community leaders, stakeholders, experts	Readiness, Risk Reduction, Recovery	Hazard and risk assessments/modelling, impact forecasting, customise/personalise warnings systems for the community, identify impact thresholds, inform/improve readiness and reduction efforts based on local knowledge

For this review, crowdsourcing refers to gathering information from active public participation, namely reports submitted via online forms or mobile applications (Harrison & Johnson, 2016). Crowdsourcing has historically been used in the response to a disaster for building situational awareness, coordinating resources, and aiding response efforts (Harrison & Johnson, 2016; Haworth & Bruce, 2015; Poblet et al., 2014). Within the severe weather context, crowdsourcing was used in the aftermath of Hurricane Katrina to locate missing people and allocate response efforts (Roche et al., 2011). In other examples, crowdsourcing is used for people on the ground to submit reports on flood levels and weather phenomena observations (Harrison & Johnson, 2016; Horita et al., 2018).

Participatory mapping and participatory Geographic Information Systems (participatory GIS) use local spatial knowledge to create spatial data or to verify and update existing data (Peters-Guarin et al., 2012). Participatory mapping generally evolves into participatory GIS when hand-drawn maps or features are digitised and integrated into a GIS for further analysis (Brown & Kyttä, 2014; Forrester & Cinderby, 2011). Participatory mapping is often used to map exposure and vulnerability to hazards in communities to support disaster risk planning (Gaillard & Pangilinan, 2010; Haklay et al., 2014). For weather-related hazards, Haworth et al. (2016) found that participatory mapping enabled local knowledge exchange for community preparedness to bushfire risks.

Local knowledge refers to knowledge possessed by locals about their communities, neighbourhoods, traditions, history, environment, and hazards, among others. Local knowledge has not been clearly defined in the literature. For the purposes of this paper, we consider local knowledge as information gathered in similar participatory mapping and participatory GIS processes but not translated into a map or GIS. Recently, the access to and integration of local knowledge has been recognised for its importance to disaster risk reduction (Anderson-Berry et al., 2018; Gall & Cutter, 2016; Sebastian et al., 2017; UNDRR, 2015b).

Past research has focused heavily on the role of VGI in disaster response, with less exploration in understanding how VGI can inform warnings before or during a severe weather event (Harrison & Johnson, 2016; Haworth & Bruce, 2015; Horita et al., 2013; Klonner et al., 2016). In Klonner et al. (2016) systematic literature review, the authors focused on documenting research on VGI for preparedness and mitigation but did not provide clear findings in the context of warnings for severe weather. Assumpção et al. (2018) identified the role of citizen observations in providing data for flood modelling and forecasting to solve issues of data scarcity, but again with no mention of warnings.

The original conception of VGI began with identifying its value for early detection and warning of hazards, using 'citizens as sensors' (Goodchild, 2007). Since then, some work has emerged exploring VGI for early warnings of various hazards, such as earthquakes, landslides, and tsunami (Carley et al., 2016; Elwood et al., 2012; Goodchild, 2007; Granell & Ostermann, 2016; Harrison & Johnson, 2016). Horita et al. (2016) argued that VGI may help address challenges of assigning proper warning

thresholds by incorporating local knowledge of response capabilities. Meissen and Fuchs-Kittowski (2014a) developed a conceptual framework which demonstrated how crowdsourced data can be fully integrated into an existing EWS as another dataset to augment or enhance the warnings by providing context. However, no further evidence to date indicates the adoption into practice of this framework for any type of EWS. Finally, Marchezini et al. (2018) conducted a literature review of research on citizen science and EWSs and found that more research is needed to identify how citizen science can be "mainstreamed" into EWSs.

Some agencies have started collecting VGI to detect, monitor, and track events and their impacts. In the United Kingdom (UK), the British Geological Survey collects landslide impact data from Twitter including text descriptions, photos, and video footage of the resulting impacts (Pennington et al., 2015). These data are integrated into the National Landslide Database, which is used to create a Hazard Impact Model (Pennington et al., 2015). In Canada, the National Meteorological Service uses hazard information posted by the public on Twitter to detect weather events such as tornadoes and to verify and update current weather watches and warnings (Harrison & Johnson, 2016). However, there is a gap in the literature for fully characterising the role of VGI for severe weather warnings. It is important to fill this gap because information and knowledge possessed by citizens have the potential to uncover "areas of importance or concern" that have yet to be identified in an official capacity (Haworth et al., 2012). VGI offers a way to capture local knowledge about previous severe weather events and their extent, severity, and resulting impacts, as well as information on the local exposure and vulnerability that warning services may not necessarily possess (GFDRR, 2016; Krennert, Pistotnik, et al., 2018; Sai et al., 2018; WMO, 2015, 2017). This paper uses a scoping review method to identify previous research into the use of VGI for severe weather EWSs, to attempt to answer the research question: What are the current and potential uses of VGI for severe weather warnings? The objective of this review is to determine how VGI has been, or could be, used within EWSs for severe weather hazards.

5.5 Method

This literature review uses a scoping method to explore areas of existing research and identify research gaps in VGI for severe weather early warning systems (Arksey & O'Malley, 2005; Paré et al., 2015). Scoping reviews provide a "rigorous and transparent method for mapping areas of research" in a short time (Arksey & O'Malley, 2005, p. 30). The aim is to describe the nature of the current literature on VGI for severe weather EWSs by describing the quality and quantity of the research (Grant & Booth, 2009; Paré et al., 2015). Scoping reviews are recognised for their strength in providing a broad picture of the state of research in a given topic area and are well-cited in the information systems field (Grant & Booth, 2009; Paré et al., 2015; Tan et al., 2017). This scoping review follows the five- step process defined by Arksey and O'Malley (2005): (1) identify the research question, (2) identify relevant studies, (3) select studies, (4) chart the data, and (5) report the results.

The initial literature search involved developing a search string to capture the broad topic area of VGI and social media for warning of severe weather hazards. The search string comprised three joined statements, shown in Table 5.2, to cover warnings and Disaster Risk Knowledge (as per the first component of the EWS framework: Basher, 2006; Golnaraghi, 2012), VGI, and severe weather, which were entered into two academic- focused databases, Scopus and EBSCO Discovery Service, in August 2018. Literature review papers have been published on similar topics in this space that have searched no more than two databases (e.g., Klonner et al., 2016; Tan et al., 2017). Furthermore, Scopus is recognised for indexing a larger number of journals than other databases and is the largest searchable citation and abstract source for various scientific fields (Falagas et al., 2008; Guz & Rushchitsky, 2009). Moreover, when searching the two databases many duplicate results were found between the two databases, ensuring confidence in the coverage.

Topics covered	Search string statement
Warnings and Disaster Risk Knowledge	("risk communication" OR "warning*" OR "impact model*" OR "risk model*" OR "impact warning*" OR "impact*based warning*" OR "impact forecast*" OR "impact*based forecast*" OR "risk*based warning*" OR "risk*based communication")
	AND
A broad definition of VGI to include social media, participatory mapping, local knowledge based on location	("participatory" OR "participatory mapping" OR "VGI" OR "volunteered geographic information" OR "participatory GIS" OR "PPGIS" OR "geographic crowdsourc*" OR "citizen science" OR "crowdsourc*" OR "social media")
	AND
Severe weather hazards as defined under the WWRP HIWeather Implementation Plan (Jones & Golding, 2014)	("weather" OR "storm*" OR "snow*" OR "wind*" OR "tornado*" OR "hurricane*" OR "cyclone*" OR "typhoon*" OR "monsoon*" OR "flood*" OR "mudslide" OR "flash flood*" OR "rain*" OR "wildfire")

Table 5.2. Search string employed in EBSCO Discovery and Scopus databases.

"Participatory GIS" and "participatory mapping" are different types of VGI, and thus were identified as separate search terms. During the process of developing the search string, it was found that additional VGI research was left out of the search due to the specificity of "participatory mapping" and "participatory GIS", thus the search was widened with the term "participatory" to capture more VGI studies. Similarly, "flash flood" and "flood" are likely redundant, however, they were both included to ensure full coverage. The asterisk in the search string acts as a wildcard to search for variations of the root term. The search covered all years from the earliest available until mid-2018 and included only peer-reviewed journals and conference proceedings in English. The search resulted in 1,015 hits from Scopus and 122 from EBSCO. After removing duplicates, 1,027 unique publications were captured.

The following inclusion-exclusion criteria were used to select publications most relevant to this study:

- Publications that specifically focused on severe weather hazards as defined under the World Weather Research Programme's (WWRP) High Impact Weather Implementation Plan (Jones & Golding, 2014, p.; n = 254);
- 2) Studies that explicitly discussed warnings, preparedness, mitigation, impact modelling and
- Studies that focused on VGI, crowdsourcing, citizen science, participatory mapping, local knowledge gathering, or social media data (reducing to n = 42).
- 4) Finally, publications had to be original, complete research papers (n = 29).

After applying the inclusion-exclusion criteria, information from the resulting papers was extracted according to different categories (see Table 5.3). Initially, the severe weather hazard(s) considered in the study were identified, after which the EWS framework was used to classify the papers and determine how VGI is or could be used within the EWS framework (these results are presented later in Figure 3). This classification involved identifying for which EWS component the VGI was used (see Figure 1), followed by the element within the EWS component (i.e., the specific task, tool, or process that the VGI was used for within the EWS component, such as risk mapping, detection, monitoring, forecasting, or warning dissemination). The VGI platform was identified (e.g., participatory mapping, participatory GIS, social media, crowdsourcing, citizen science, local knowledge), as well as the type of data that was collected (Haklay, 2017; Harrison & Johnson, 2016). These categories were chosen to determine the representation of VGI in severe weather EWSs.

5.6 Results

The search of the two databases led to 1,027 unique publications. After applying the inclusion-exclusion criteria, the final number of papers selected for this study was 29. The categories listed in Table 5.3 were used as a structure for analysis and discussion and were chosen based upon the dominance of those themes in the papers.

Category	Description
Hazard	The type of severe weather hazard(s) considered in the study.
Early Warning System	The component from the EWS framework that each study
Component	applies to.
VGI Platform	The source of the VGI data, such as from social media, or from crowdsourcing (i.e., citizen observation), citizen science (i.e., a high level of engagement than crowdsourcing (Haklay, 2013)), participatory mapping, participatory GIS, or local knowledge.
Data Type	The type of data that was collected through the VGI process, such as local knowledge captured through interviews and/or participatory mapping, hazard data from social media or crowdsourcing, etc.

Table 5.3. Categories for literature review.
5.6.1 Hazard Type

The selected articles covered a range of severe weather hazards as defined in the World Weather Research Programme (WWRP) High Impact Weather (HIWeather) implementation plan (Jones & Golding, 2014). Some hazards are represented more than others; of the 29 articles, 16 focused on flood hazards, followed by seven studies that covered general severe weather hazards, two studies that examined rain-induced landslides, two for cyclones, and one each for air quality and urban heat wave.

The 16 flood studies covered a range of elements within the EWS components. These elements were identified by reviewing the selected studies and aligning them with the EWS components. Table 5.4 provides a summary of the selected studies which examined floods. Most studies covered flood detection, monitoring, and forecasting using VGI collected from social media and crowdsourcing. The next most common elements that were covered in the flood studies were vulnerability assessments and risk mapping and modelling, using VGI from participatory GIS, participatory mapping, local knowledge, and social media. Just one study looked at using social media for detection, warning messaging, and for informing preparedness decisions (Allaire, 2016).

The remaining 13 studies covered other hazards, such as general severe weather, cyclones, landslides, air quality, and urban heatwaves. Table 5.5 provides a summary of the selected studies covering these various hazards. The general category refers to studies that did not identify a specific severe weather hazard, but referred only to "severe weather", usually in the context of severe weather warnings (Fdez-Arroyabe et al., 2018; Grasso & Crisci, 2016; Grasso et al., 2017; He et al., 2018; Krennert, Pistotnik, et al., 2018; Longmore et al., 2015; Lu et al., 2018).

In the general category, most of the selected studies looked at detection and forecasting using social media and crowdsourcing, followed by tracking warning dissemination across social media, and one study that used crowdsourcing for both risk and vulnerability assessment and providing warnings. The two cyclone studies each used social media and local knowledge to detect and forecast cyclone damage and to understand local responses to warnings, respectively. The two landslide studies both used VGI for landslide hazard and impact modelling, using crowdsourcing and social media. Finally, both the air quality and urban heatwave studies explored VGI from social media to forecast air quality and detect heatwaves based on individual exposure.

These studies indicate that VGI is used in the mapping, modelling, detection, monitoring, and warning of a number of severe weather hazards but that floods are the most heavily studied, with the widest range of VGI application across all of the elements. How these studies fit within the EWS framework is analysed in the following section.

Table 5.4. Summary of selected studies covering flood hazards.

EWS Component	Element	Purpose of the study	VGI Platform	Data Type	Reference
Disaster Risk Knowledge	Modelling	To integrate local knowledge into GIS outputs for flood risk management using PGIS in order to understand how people cope and adapt	Participatory GIS	Interviews with households in Barangay, Philippines	(Peters-Guarin et al., 2012)
	Modelling	Validating flood models using quantitative and qualitative VGI	Participatory Mapping	Local knowledge from workshop participants and interviewees	(Rollason et al., 2018)
	Risk mapping	To provide an example of how to engage and collaborate with local stakeholders for flood management	Participatory Mapping	Land feature layers, input from locals	(Lavers & Charlesworth, 2018)
	Vulnerability assessment	To present a risk management framework that is based on local knowledge of the vulnerability to water hazards	Local knowledge	Meetings, workshops, interviews with people, media, and public sectors related to risk management	(Arias et al., 2016)
	Vulnerability assessment	To present a new methodology for incorporating stakeholder's participation, local knowledge, and locally spatial characteristics for vulnerability assessments of flood risk	Participatory GIS	Demographic data, infrastructure, hazard data (e.g., average annual rainfall), questionnaire interviews with experts and community members	(Hung & Chen, 2013)
	Vulnerability assessment	To present a new database for collection and assessment of flood damage using a bottom- up approach to gather and identify damage data	Social media	Personal blogs, on-site observations, public administration, social media, online media, local authorities, corporate websites	(Saint-Martin et al., 2018)

EWS	Element	Purpose of the study	VGI Platform	Data Type	Reference
Component					
Detection, Monitoring, Warning Services	Detection	To develop a service-oriented architecture for flood management to capture information real- time about floods	Crowdsourcing	Rainfall, river, news, OpenStreetMap	(Sharma et al., 2016)
	Detection	To develop a methodology for interpreting image tags on social media to allow for early detection of a flood and recording the impacts	Social media	Flickr posts - timestamps and location metadata	(Tkachenko et al., 2017)
	Detection, Forecasting	SWOT analysis of web-based access to data and model simulations, and insight on pEWMS, and conceptual framework for a Nordic pEWMS	Crowdsourcing, Social Media	Denmark: groundwater level observations Iceland: flood photos Finland: mobile phone observations	(Henriksen et al., 2018)
	Detection, Monitoring	To assess social media feasibility for flood detection, monitoring, and forecasting and develop a novel methodology for doing so	Social media	Twitter data	(Rossi et al., 2018)
	Forecasting	To develop a methodology using social media for estimating rainfall runoff estimations and flood forecasting	Social media	Twitter data	(Restrepo- Estrada et al., 2018)
	Forecasting	To present a real-time modelling framework to identify likely flooded areas using social media	Social Media	Twitter data, LiDAR	(Smith et al., 2017)
	Monitoring	To estimate flood severity in an urban coastal setting using crowdsourced data	Crowdsourcing	Crowdsourced street flooding reports	(Sadler et al., 2018)
	Monitoring	To present a conceptual framework for collecting and integrating heterogeneous data from sensor networks and VGI	Crowdsourcing	Flood data from in-situ sensors and volunteers	(Horita et al., 2015)
	Monitoring	To present a new methodology for monitoring flood hazards using remote sensing and VGI	Crowdsourcing, Social Media	Volunteered data (photos, videos, news), Landsat, DEM, meteorological data, river data	(Schnebele & Cervone, 2013)

EWS	Element	Purpose of the study	VGI Platform	Data Type	Reference
Component					
Detection, Monitoring, Warning Services; Communication and Dissemination Mechanism; Preparedness and Early Response Capacity	Warning messaging, preparedness	To test if evidence exists for social media reducing flood losses by informing mitigation decisions before the flood	Social media	Surveys, in-depth interviews with households who experienced flooding in Bangkok, 2011	(Allaire, 2016)

Hazard	EWS Component	Element	Purpose of the study	VGI Platform	Data Type	Reference
General	Disaster Risk Knowledge; Detection, Monitoring, Warning Services	Risk mapping	To present a data infrastructure that can be used to delineate individual vulnerability to meteorological changes	Crowdsourcing	User profiles on a mobile app	(Fdez-Arroyabe et al., 2018)
		Detection	To present an Android-based application for geohazard reduction using crowdsourcing	Crowdsourcing	Crowdsourced information (field data, photos, videos)	(He et al., 2018)
		Detection, Monitoring	To present a conceptual framework for collecting weather photos	Crowdsourcing	User reports, photos, videos	(Longmore et al., 2015)
	Detection, Monitoring, Warning Services	Detection, Monitoring	To evaluate the occurrence of crowdsourcing for severe weather within European NMHSs	Crowdsourcing, Social Media	Surveys with European National Meteorological and Hydrological Services	(Krennert, Pistotnik, et al., 2018)
		Forecasting	To use social media as a new way of forecasting and generating traffic alerts due to weather hazards	Social media	Temporal, spatial, traffic, and meteorological data from Weibo	(Lu et al., 2018)
	Communication and Dissemination Mechanism	Warning dissemination	To study the use of codified hashtags relating to weather warnings in Italy	Social media	Twitter data	(Grasso & Crisci, 2016)
		Warning dissemination	To evaluate the use of a list of predefined codified hashtags for weather warnings in Italy	Social media	Twitter data	(Grasso et al., 2017)
one	Detection, Monitoring, Warning Services	Forecasting	To determine if social media and geo- location information can contribute to a more efficient early warning system and help with disaster assessment	Social media	Twitter data, Hurricane damage loss data	(Wu & Cui, 2018)
Cycl	Preparedness and Early Response Capacity	Response to warnings	To integrate local and scientific meteorological knowledge and actions within coconut farming communities in the Philippines	Local knowledge	Interviews with key stakeholders	(Ton et al., 2017)

Table 5.5. Summary of selected studies covering other severe weather hazards.

Hazard	EWS Component	Element	Purpose of the study	VGI Platform	Data Type	Reference
de		Modelling	To present a crowdsourcing smartphone app for landslide reports which populates a landslide database	Crowdsourcing	Crowdsourced landslide reports from app users	(Choi et al., 2018)
Landsli	Disaster Risk Knowledge	Modelling	To present a national landslide database in the U.K. which is partially populated with social media data to capture the impacts of landslides and for early detection of landslides	Social media	Twitter data	(Pennington et al., 2015)
Air quality	Detection, Monitoring, Warning Services	Forecasting	To explore the use of social media as a real-time data source for forecasting smog-related health hazards	Social media	Social media data and physical sensors data	(Chen et al., 2017)
Urban heat wave	Detection, Monitoring, Warning Services	Detection	To investigate the relationship between heat exposure and tweet volume over time	Social media	Twitter data	(Jung & Uejio, 2017)

5.6.2 Early Warning System Components

The papers were categorised by EWS component, as per Basher's (2006) framework (see Figure 5.1): (1) Disaster Risk Knowledge (n = 8); (2) Detection, Monitoring, and Warning Services (n = 16); (3) Communication and Dissemination Mechanisms (n = 2); and (4) Preparedness and Early Response Capacity (n = 1). Two studies were found to fall into more than one EWS component. The studies were then classified by the specific elements within each component (e.g., hazard mapping, risk mapping, vulnerability assessment, modelling, hazard monitoring, detection, monitoring, warning, messaging, dissemination).

5.6.2.1 Disaster Risk Knowledge

Eight studies fall into the Disaster Risk Knowledge component of the EWS framework. Four of these studies looked at the use of VGI for hazard, risk, or impact modelling for landslides and floods (Choi et al., 2018; Pennington et al., 2015; Peters-Guarin et al., 2012; Rollason et al., 2018). Choi et al. (2018) presented a crowdsourcing-based smartphone application to aggregate landslide reports, which populates a landslide database for further hazard analysis. Similarly, Pennington et al. (2015) presented a landslide database for the UK that is partially populated by reports from Twitter to capture their impacts for further modelling. In the floods space, Peters-Guarin et al. (2012) utilised participatory GIS to integrate local knowledge of coping and adaptation practices into GIS-based flood risk analysis. Alternatively, Rollason et al. (2018) used participatory mapping to validate existing flood models.

The other four studies in the Disaster Risk Knowledge component involved risk mapping and vulnerability assessments, also for floods (Arias et al., 2016; Hung & Chen, 2013; Lavers & Charlesworth, 2018; Saint-Martin et al., 2018). Lavers and Charlesworth (2018) engaged with landowners to capture their knowledge of flood risk to inform flood management. Arias et al. (2016) presented a risk management framework for floods based on local knowledge of the vulnerability to water hazards. Hung and Chen (2013) incorporated stakeholders' participation and local knowledge through participatory GIS for vulnerability assessments of flood risk. Saint-Martin et al. (2018) developed a flood damage database (DamaGIS) to collect and assess flood damage, sourced from corporate websites, personal blogs, local authorities, on-site observations, social media, and online media. Furthermore, Saint-Martin and colleagues argued that social media can extend coverage to areas lacking regular media coverage and reveal damage that might have otherwise gone undetected.

5.6.2.2 Detection, Monitoring, and Warning

Within the Detection, Monitoring, and Warning component, 16 studies were identified. Four studies used VGI for hazard detection. Tkachenko et al. (2017) and Sharma et al. (2016) looked at VGI for detecting floods and capturing impacts from social media and crowdsourced data respectively. Jung and Uejio (2017) tested the effectiveness of measuring heat exposure on social media and consequently detecting urban heatwaves. Similarly, He et al. (2018) developed a crowdsourcing

application to detect various weather hazards and to capture impacts to improve the decision-making of local governments. Henriksen et al. (2018) indicated the role of social media and crowdsourcing for both detection and forecasting of floods, while Rossi et al. (2018) assessed the feasibility of social media for flood detection and monitoring. Longmore et al. (2015) presented a conceptual crowdsourcing framework for collecting photos of severe weather hazards in the United States to improve weather monitoring by the National Weather Service. In Europe, Krennert, Pistotnik, et al. (2018) assessed the occurrence of crowdsourcing (either through specialised applications or social media) by national hydrological and meteorological services to capture severe weather observations and impacts for real-time warning verification and improvement.

VGI for forecasting alone was used for floods, cyclone damage, general severe weather traffic impacts, and air quality. Restrepo-Estrada et al. (2018) developed a methodology using social media for estimating rainfall runoff estimations and flood forecasting, while Smith et al. (2017) presented a real-time modelling framework to identify likely flooded areas using social media. Alternatively, Wu and Cui (2018) found that geo-located social media can help with disaster assessment, and for future forecasting. Lu et al. (2018) explored how social media might be used to forecast and generate traffic alerts due to severe weather. Likewise, Chen et al. (2017) explored social media for smog-related hazards.

Finally, three studies used VGI to monitor floods. Schnebele and Cervone (2013) crowdsourced from social media and other online media to monitor flood hazards and to create hazard maps, finding that the VGI is useful when satellite data is unavailable. Horita et al. (2015) developed a framework to integrate crowdsourced flood observations with official sensor data. The authors found that the VGI made it possible to capture data from areas lacking flood sensors (Horita et al., 2015). Sadler et al. (2018) crowdsourced street flooding reports to estimate flood severity for flood prediction, but the poor temporal and spatial coverage of the crowdsourced reports hindered the performance of the prediction model (Sadler et al., 2018).

5.6.2.3 Communication and Dissemination Mechanisms

Two studies were identified for the third EWS component, Communication and Dissemination Mechanisms. Both studies used VGI to assess warning dissemination via social media (namely Twitter) for general severe weather (Grasso & Crisci, 2016; Grasso et al., 2017). Grasso and Crisci (2016) analysed codified hashtags of regions in Italy impacted by rainfall and found that codified hashtags for different regions effectively enable the sharing of useful information during severe weather events. Additionally, many tweets included geo-location information along with hazard information to update and complement official data. As such, the authors argued that institutions might adopt codified hashtags to improve the performance of systems for disseminating and retrieving information. Grasso et al. (2017) built on this work by adding more regions to their tweet analyses and emphasised the importance of institutions and warning services to promote codified hashtags for warnings to streamline message delivery and reach.

5.6.2.4 Preparedness and Early Response Capacity

For the last component, Preparedness and Early Response Capacity, only one study applied. Ton et al. (2017) collected VGI in the form of local knowledge using interviews and questionnaires with farmers to understand their response to cyclone warnings. In this process, the farmers identified economic, physical, social, and natural impacts of cyclone hazards. The authors found that while farmers forecast weather conditions and impacts based on their local knowledge, their confidence in the leadtime of their forecasts has declined due to changing climate conditions. As such, the authors argued for the integration of local knowledge with scientific forecasts to verify local knowledge-based forecasts and increase confidence.

5.6.2.5 Multiple Components

Two studies were found to fall into more than one EWS component. Allaire (2016) used VGI for Detecting, Monitoring, and Warning, assessing Communication and Dissemination Mechanisms, and for measuring Preparedness and Early Response capacities for flood hazards. Allaire (2016) found that social media was an effective tool for flood monitoring (falling in the Detection, Monitoring, and Warning component), for receiving and spreading flood information (as a Communication and Dissemination Mechanism), and for receiving and spreading preparedness information, leading to reduced impacts (informing Preparedness and Early Response Capacity). Alternatively, Fdez-Arroyabe et al. (2018) developed a mobile application to obtain individual vulnerabilities to meteorological changes (thus informing Disaster Risk Knowledge) and to provide personalised alerts based on the individual vulnerabilities to meteorological conditions (informing Detection, Monitoring, and Warning services).

5.6.3 VGI Platforms and Data Types

In this review, we broadly define VGI to include participatory mapping, participatory GIS, geo-located social media, and location-based local knowledge (Porto de Albuquerque et al., 2016). Figure 5.2 shows the distribution of platforms discussed in each of the selected studies and to which component of the EWS framework they apply. The following section provides definitions of the platforms displayed in Figure 5.2 along with a description of how the VGI is used for severe weather warnings.

Chapter 5: Volunteered Geographic Information for people-centred severe weather early warning



Figure 5.2. Distribution of VGI platforms used for each Early Warning System (EWS) framework component. Note: Two studies fell into multiple components and have been counted for each EWS component that they apply to, which results in a total of 32, rather than 29.

5.6.3.1 Geo-located social media

Geo-located social media refers to VGI that is posted online by users of Facebook, Twitter, Sina Weibo, Flickr, YouTube, Instagram, and SnapChat that has geographical location information associated to it. The heavy representation of social media (15 studies) demonstrates the growing popularity of these platforms as a data source for severe weather events (Tkachenko et al., 2017). The results indicate that social media is a valid tool for measuring the effectiveness of warning dissemination by following Twitter hashtags (Allaire, 2016; Grasso & Crisci, 2016; Grasso et al., 2017; Taylor et al., 2018). The online platforms are also useful for early hazard detection and for estimating event magnitude for early warnings (Chen et al., 2017; Jung & Uejio, 2017; Restrepo-Estrada et al., 2018; Tkachenko et al., 2017). Reasons for collecting social media data were to increase coverage of the dataset(s), the ease of access and quantity of data available, real- time or near-real-time monitoring and collection, and the multi-directional communication during disaster enabled by social media (Allaire, 2016; Chen et al., 2017; Grasso & Crisci, 2016; Grasso et al., 2017; Jung & Uejio, 2017; Pennington et al., 2015; Rossi et al., 2018; Saint-Martin et al., 2018; Smith et al., 2017; Wu & Cui, 2018).

5.6.3.2 Crowdsourcing applications and forms

Eight of the selected studies used crowdsourcing via mobile applications, reporting forms, or other active contributions (e.g., storm spotters). The crowdsourcing applications in the selected studies were used for hazard detection and monitoring and for developing personalised risk knowledge. These applications allow citizens to report the occurrence of hazards such as landslides (Choi et al., 2018; He et al., 2018) and to monitor hazards such as rainfall-induced floods (Horita et al., 2015) and storms (Krennert, Pistotnik, et al., 2018; Longmore et al., 2015). The ability to efficiently collect reports and monitor hazards in real-time, in a standardised format

to ensure quality, and to increase the scale and resolution of hazard-related data were arguments made for using crowdsourcing as opposed to other VGI collection types (Choi et al., 2018; He et al., 2018; Henriksen et al., 2018; Horita et al., 2015; Longmore et al., 2015; Sadler et al., 2018; Sharma et al., 2016).

5.6.3.3 Participatory mapping and participatory GIS

In the selected studies, participatory mapping and participatory GIS were employed for severe weather risk assessments and hazard modelling. Lavers and Charlesworth (2018) engaged UK farmers in participatory mapping to identify flood impacts on their properties and subsequent opportunities for mitigation. Peters-Guarin et al. (2012) had locals in the Philippines map their historical knowledge of recurring floods and impacts for a risk assessment. In Taiwan, Hung and Chen (2013) consulted with locals and stakeholders to verify flood vulnerability maps. Participatory mapping and interviews were utilised by Rollason et al. (2018) to validate flood models using local knowledge and experiences. In all of these studies, the mapped information was entered into a GIS for further mapping and analysis, thus qualifying it as participatory GIS. Reasons for using participatory GIS and participatory mapping over other types of VGI were formally recognising and integrating local knowledge in a systematic way, and supporting local engagement (Hung & Chen, 2013; Lavers & Charlesworth, 2018; Peters-Guarin et al., 2012; Rollason et al., 2018).

5.6.3.4 Local Knowledge

For the purposes of this paper, we consider local knowledge as information gathered in participatory processes containing knowledge of the participants' local area and geography, that may or may not be translated onto a map. Just one selected study included local knowledge. After evaluating local knowledge of cyclone hazards and response capabilities to scientific knowledge, Ton et al. (2017) argued that local knowledge should be integrated with scientific meteorological knowledge for verification and to increase confidence in forecasts. The choice of using local knowledge for this study was to begin a dialogue between the locals and the meteorologists towards building trust (Ton et al., 2017).

5.7 Discussion

The results show that VGI is useful in all components of the early warning system (EWS) framework, but some platforms are more useful for specific components than are others. Furthermore, the different types of VGI have implications for supporting people-centred EWSs, which is a guiding principle for EWSs under the Sendai Framework.

5.7.1 Volunteered Geographic Information in Severe Weather Early Warning Systems

The purpose of this study is to determine the current and potential uses of VGI for severe weather warnings. We used the EWS framework to guide the analysis of the results.

The results from this literature review show that VGI has value in all four components of an EWS for severe weather hazards (Basher, 2006), but some forms of VGI are more useful for specific EWS components than are others (see Figure 5.3). Figure 5.3 is an update of Figure 5.1 based on the findings from this literature review to better represent how the different types of VGI inform or support the EWS components. For example, the majority of included studies used social media and crowdsourcing for hazard detection, monitoring, and early warning, while all of the included participatory mapping and participatory GIS studies used VGI for building disaster risk knowledge.

The selected studies show that social media and crowdsourcing for severe weather are effective for early detection, monitoring, and verifying warnings (e.g., Harrison & Johnson, 2016; Henriksen et al., 2018; Krennert, Pistotnik, et al., 2018). The value of social media and crowdsourcing for EWSs lies in the real-time, or near- real-time, hazard and impact detection, forecasting, and warning verification (Henriksen et al., 2018; Kox, Kempf, et al., 2018; Krennert, Pistotnik, et al., 2018). However, the papers included in this scoping review lack forward-thinking for integrating these tools into official EWSs which is a challenge for warning services and emergency management services (Haworth, 2016; Henriksen et al., 2018; Kox, Lüder, et al., 2018). Despite this challenge, some national hydrological and meteorological services and emergency management agencies in Europe and North America collect information from social media for detection, monitoring, and warning verification (Harrison & Johnson, 2016; Henriksen et al., 2018; Krennert, Pistotnik, et al., 2018; Pennington et al., 2015).

Social media supports multi-directional communication, which allows for both crowdsourcing and broadcasting severe weather information. While most of the selected social media studies demonstrated the value of social media for detection and early warning, two studies also indicated its utility for disseminating warnings and assessing the spread of, and response to, warning messages (Grasso & Crisci, 2016; Grasso et al., 2017). This allows warning services to gauge the reach of their message, understand the responses to their message, and update subsequent messages based on what they see on social media (Harrison & Johnson, 2016).

Before warnings are issued, knowledge of disaster risk is needed to be able to create tailored warnings. Participatory mapping and participatory GIS might be considered a long-term process for building knowledge and datasets for improving disaster risk knowledge as well as validating hazard and risk maps or models. While social media is valuable for real-time detection and communication, the participatory nature of participatory mapping enables more in-depth engagement with locals and communities in other areas of the EWS process to produce new knowledge (Haworth

et al., 2018; Lavers & Charlesworth, 2018; Maskrey et al., 2016; Peters-Guarin et al., 2012; Zolkafli et al., 2017). Integrating local, spatial knowledge about disaster risk into an EWS through participatory mapping and participatory GIS fosters efforts towards people-centred EWSs as it translates local knowledge into usable and useful spatial data for risk analysis and for improved warnings (Basher, 2006; UNDRR, 2015b).



Figure 5.3. Volunteered Geographic Information for people-centred severe weather early warning systems.

These results support the findings from Marchezini et al. (2018), who presented a framework for bridging citizen science into EWSs. Like Marchezini et al. (2018), we found that VGI processes can bridge the gap between EWSs and audiences of warnings by incorporating local knowledge and personal experiences from stakeholders into the EWS components (see also Ton et al., 2017). This creates new data and unearths vulnerabilities at various scales (e.g., from the individual level to the community level; Haworth et al., 2018; Henriksen et al., 2018; Kox, Kempf, et al., 2018; Ton et al., 2017).

5.7.2 Implications for the different types of VGI

The results show that social media is a dominant platform for collecting VGI across severe weather hazards. Given the ease of access to, and the versatility of, social media (Harrison & Johnson, 2016), it is not surprising that social media is the most common platform used across hazards for collecting VGI (Granell & Ostermann, 2016). Social media is also now considered a "go-to" for collecting data because it is where the members of the public already are, thus groups or agencies looking to crowdsource do not have to do the heavy-lifting of creating a new app and attracting new users (Harrison & Johnson, 2016).

The perceived benefits of social media also come with some caveats. The data tend to be biased due to the uneven distribution of the social media user base (Granell & Ostermann, 2016; Harrison & Johnson, 2016). By relying on social media as a data source, those members of the public who are not present on social media are not represented in the data nor in the EWS process (i.e., the digital divide; Allaire, 2016; Harrison & Johnson, 2019). Additionally, tweet or post ambiguity and keyword selection for data-capture hinder data collection and analysis (Chen et al., 2017; Longmore et al., 2015; Tkachenko et al., 2017). Assimilating data of different formats into a database remains a challenge (Horita et al., 2015; Lu et al., 2018).

Capturing enough geo-located social media data is a constant challenge. It is widely known that only a small percentage of tweets contain geo-located information (Steed et al., 2019). Furthermore, the accessibility and availability of geo-located social media data are continuously limited. For example, Facebook does not offer an Application Programming Interface (API) to allow for researchers or media agencies to systematically collect Facebook posts, much less geo-located posts; it only offers an API for marketing and advertising agencies (Dubois et al., 2018; Thakur et al., 2018). In addition, in June 2019 Twitter announced plans to disable the geo-location feature for tweets due to its limited adoption by users and growing privacy concerns; however, the feature will still be available on photos taken within the Twitter mobile application (Benton, 2019; Khalid, 2019). While geo-located information on Instagram appears to be available for the moment (Arapostathis, 2019; Boulton et al., 2016), given the recent trends in the other major social media platforms, the continued availability and accessibility of this data in the future is uncertain.

A specialised crowdsourcing application can help to address some limitations found in social media. Crowdsourcing applications offer quality assurance, noise avoidance, application customisation, and citizen engagement (Choi et al., 2018; Longmore et al., 2015). On the other hand, crowdsourcing applications remain limited in the volume of participation due to public motivation to participate, the digital divide, and privacy concerns (Choi et al., 2018; Fdez-Arroyabe et al., 2018). Bias in reporting is also a concern, as contributors may over-exaggerate their personal experiences (Fdez-Arroyabe et al., 2018). Developing an application has the potential to streamline the integration of crowdsourced data into official processes, yet maintenance costs impede the willingness of officials to do so (Choi et al., 2018).

Capturing and representing local knowledge through participatory mapping and participatory GIS may help in bridging the digital divide, ensuring data quality, and enabling data integration. Participatory mapping and participatory GIS also enable community engagement (Haworth, 2016; Lavers & Charlesworth, 2018; Peters-Guarin et al., 2012). Participatory mapping and participatory GIS can be done using paper-mapping, as was done by Rollason et al. (2018), Lavers and Charlesworth (2018), and Peters-Guarin et al. (2012), or through digital-mapping (Haworth, 2016). In addition to the value of the resulting information and data itself, the process of engaging with and between locals provides another level of value in the social context by strengthening social networks, growing social capital, and increasing civic participation (Haworth, 2016).

Participatory GIS and participatory mapping do not come without their own limitations. For example, participatory GIS appears to be more effective with smallscale local projects. This is because most of the data collected is at a local or small scale, resulting in poor spatial distribution if scaled-up to a larger area. This could lead to underrepresentation and potential biases in the participatory GIS data (Rollason et al., 2018). Nevertheless, the rich quality and the ease of integrating this VGI into official processes may outweigh this limitation if the study is well-designed and the data is used appropriately (Brabham, 2013; Lauriault & Mooney, 2014). Within the EWS context, these perceived benefits further the movement towards people-centred EWSs by incorporating knowledge and information produced by the people into warnings that are ultimately for them (UNDRR, 2015b).

5.8 Conclusion

This paper conducted a scoping literature review and explored 29 journal papers published in academic journals and conference proceedings retrieved from EBSCO Discovery and Scopus. The literature review found that VGI plays various roles for severe weather early warning systems (EWSs). The examples from the selected studies show that VGI furthers the development of people-centered EWSs; it brings people, their knowledge, and their experiences into EWSs. Still, the current research captured in this scoping review lacks forward-thinking for integrating these tools into official EWSs which is a challenge for warning services and emergency management services (Haworth, 2016; Henriksen et al., 2018; Kox, Kempf, et al., 2018).

In the always shifting EWSs landscape, a new type of severe weather EWS is emerging that is causing national meteorological and hydrological services and warning services to re-think their traditional warning practices. The World Meteorological Organization is advocating for the aforementioned services to adopt impact-based forecasts and warning systems (Fleming et al., 2015). Impact-based forecasts and warnings are meant to shift the focus from the physical hazard phenomena to the risk of impacts produced by the hazard, including communicating impacts in warning messages and building new warning thresholds based on risk of impact (Morss et al., 2018; Poolman, 2014; Potter et al., 2018; Rogers et al., 2017; Sai et al., 2018; WMO, 2015). However, warning services have indicated a limited understanding of, and access to, the data required for developing impact-based

forecasting and warning systems (Harrison et al., 2014; Kox, Kempf, et al., 2018; Obermeier & Anderson, 2014).

Future research would benefit from a systematic review of this topic area in the future. Additional research should investigate the data needs for impact-based forecasts and warnings and explore how VGI can help in meeting these data needs while also maintaining a people-centred focus. This would align with the goals of the World Meteorological Organization's High Impact Weather research programme (http://hiweather.net) which aims to improve the effectiveness of weather-related warnings in support of advances in weather prediction and forecasting (Zhang et al., 2019). While this literature review characterised the role of VGI within severe weather EWSs and demonstrated how it supports people-centred EWSs, future research can delve into the nature of the resulting data and how it might support impact-based forecast and warning systems. It should be noted that in spite of the popularity of collecting and using social media data, given the uncertainty of reliable access to social media data in the future (e.g., disestablishing the geolocation function on Twitter), it would be wise to minimise reliance on these platforms and consider additional VGI sources and collection processes to capture the desired information.

	Chapter One
Part 1	Chapter Two
	Background and Context
	Chapter Three
	Research Design
Part 2	Chapter Four
	Methods Results and Overarching Findings
	Chapter Five
	Volunteered Geographic Information for people-
	centred severe weather early warning: A
	literature review
	Chapter Six
	Identifying the data uses and gaps for severe
	weather impact forecasts and warnings
	Chapter Seven
	Identifying data sources for severe weather
Part 3	impact forecasts and warnings in Aotearoa New Zealand
	Chapter Eight
	A socio-technical analysis of sharing and
	A socio-technical analysis of sharing and governing hydrometeorological impact data in
	A socio-technical analysis of sharing and governing hydrometeorological impact data in Aotearoa New Zealand
	A socio-technical analysis of sharing and governing hydrometeorological impact data in Aotearoa New Zealand Chapter Nine
	A socio-technical analysis of sharing and governing hydrometeorological impact data in Aotearoa New Zealand Chapter Nine Integration of Partnerships and Collaboration to
	A socio-technical analysis of sharing and governing hydrometeorological impact data in <u>Aotearoa New Zealand</u> Chapter Nine Integration of Partnerships and Collaboration to Support IFWs and HIVE Data Collection, Access,
	A socio-technical analysis of sharing and governing hydrometeorological impact data in Aotearoa New Zealand Chapter Nine Integration of Partnerships and Collaboration to Support IFWs and HIVE Data Collection, Access, and Use
	A socio-technical analysis of sharing and governing hydrometeorological impact data in Aotearoa New Zealand Chapter Nine Integration of Partnerships and Collaboration to Support IFWs and HIVE Data Collection, Access, and Use
Part 4	A socio-technical analysis of sharing and governing hydrometeorological impact data in Aotearoa New Zealand Chapter Nine Integration of Partnerships and Collaboration to Support IFWs and HIVE Data Collection, Access, and Use Chapter Ten

Position of the reader

This chapter presents the second manuscript prepared for publication that identifies the data uses and gaps for severe weather IFWs. Chapter 5 explored the potential uses of VGI within EWSs and identified that further research is needed to explore additional data sources to support the implementation of IFWs. This was confirmed through conducting a series of preliminary interviews with officials from meteorological agencies both within and outside of NZ, and EM agencies within NZ, where it was also determined that exploring VGI as an alternative source of impact data was not a critical need of the NZ-based participants. Further analysis of the preliminary interview data also identified that more clarity was needed around the implementation processes involved with IFW, namely who needs the data to issue IFWs, how the underpinning data is used, and where the gaps are.

Thus, Chapter Six focuses on research Objectives 2.1: Identify the actors involved in an IFW system and their associated roles; 2.2 Determine how HIVE data are used in an IFW system; and 2.3 Identify further data gaps/needs for implementing IFWs. It presents the results of the research conducted for this thesis as described in Chapters Three and Four relating to these objectives.

The article presented in this chapter was published in 2022 in the *Weather, Climate, and Society Journal*:

Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E. H., & Johnston, D. M. (2022). Identifying the data uses and gaps for severe weather impact forecasts and warnings. *Weather, Climate, and Society*. https://doi.org/10.1175/WCAS-D-21-0093.1

The published form as available in Appendix K © American Meteorological Society. Used with permission.

6.1 Abstract

Impact forecasts and warnings (IFW) are key to resilience for hydrometeorological hazards. Communicating the potential social, economic, and environmental hazard impacts allows individuals and communities to adjust their plans and better prepare for the consequences of the hazard. IFW systems require additional knowledge about impacts and underlying vulnerability and exposure. Lack of data or knowledge about impacts, vulnerability, and exposure has been identified as a challenge for IFW implementation.

In this study, we begin to address this challenge by developing an understanding of the data needs and uses for IFWs. Using the grounded theory method, we conducted a series of interviews with users and creators of hazard, impact, vulnerability, and exposure data (e.g., warning services, forecasters, meteorologists, hydrologists, emergency managers, data specialists, risk modelers) to understand where these data are needed and used in the warning value chain, a concept used to represent and understand the flow of information among actors in the warning chain. In support of existing research, we found a growing need for creating, gathering, and using impact, vulnerability, and exposure data for IFWs.

Furthermore, we identified different approaches for impact forecasting and defining impact thresholds using objective models and subjective impact-oriented discussions depending on the data available. We also provided new insight into a growing need to identify, model, and warn for social and health impacts, which have typically taken a back seat to modeling and forecasting physical and infrastructure impacts. Our findings on the data needs and uses within IFW systems will help guide their development and provide a pathway for identifying specific relevant data sources.

6.2 Introduction

Warning systems are key to resilience for hydrometeorological hazards as they alert people to the risk of potential hazards and encourage protective action to be taken (Basher, 2006). To improve early warning systems, the World Meteorological Organization (WMO) is encouraging nations to adopt Impact Forecasts and Warnings, as they argue that communicating the potential social, economic, and environmental hazard impacts allows individuals and communities to adjust their plans and better manage the potential consequences of the hazard (Harrowsmith et al., 2020; WMO, 2015).

Traditionally, warnings have relied solely on weather-based factors (e.g., wind speeds, snowfall depth) and hazard timing and location, but 'impact-based' warnings also consider exposed and/or vulnerable populations and infrastructure (Harrowsmith et al., 2020; WMO, 2015). Impact Forecasting and Warning (IFW) systems differ from traditional warnings by communicating what the hazard(s) will do, rather than only what they will be. They are built on impact-based rather than hazard-based thresholds (Harrowsmith et al., 2020; WMO, 2015) that may misrepresent the impacts of the hazard(s) (Sai et al., 2018). Furthermore, hazard-based warnings may lack messaging about risk, leading to potential inaction (Sai et al., 2018).

The risk of hydrometeorological hazard impacts depends on the vulnerability of the people, infrastructure, and environment, and the exposure of these 'assets' multiplied by the likelihood of the hazard (Harrowsmith et al., 2020; Poolman, 2014; Tarchiani et al., 2020; WMO, 2015). Accurately communicating the risk of hydrometeorological hazard impacts in IFWs thus requires combining likelihood and hazard severity with exposure and vulnerability (Poolman, 2014; Tarchiani et al., 2020; WMO, 2015).

Notable historic severe weather events revealed communication gaps between warning services and recipients. These have been attributed to various underlying social behaviours such as warning fatigue (Mackie, 2013; Wagenmaker et al., 2011), and understanding warning terminology (Ching et al., 2015). IFW systems may help reduce the effects of these factors and increase warning compliance (Morss et al., 2018; Potter et al., 2018; Weyrich et al., 2018). However, their effectiveness depends on several caveats in design and implementation (see Morss et al., 2018; Potter et al., 2018; Ripberger, Silva, Jenkins-Smith, & James, 2015; Scolobig et al., 2015; Weyrich et al., 2018).

Implementing IFW systems can be costly, requiring careful cost-benefit analysis (Merz et al., 2020). Potter et al. (2021) found that key benefits include a perceived increase in understanding impacts, added awareness of antecedent conditions, possible reductions of 'false alarms', and increased interagency communication. Others found benefits to be enhanced situational awareness (Kox, Kempf, et al., 2018), and improved planning and response efforts (Terti et al., 2015). Alternatively, Potter et al. (2021) identified key challenges to be lack of data, the potential for conflicting messages, and an increased burden on agencies providing information to forecasters.

Verification and conflicting roles and responsibilities are other significant challenges identified by Hemingway and Robbins (2019) and Kaltenberger et al. (2020), respectively. Data challenges can be another roadblock, including their availability, processing capabilities, and management (Hemingway & Robbins, 2019; Potter et al., 2021; Wei et al., 2018).

There is a growing need to address challenges with data availability and access and identify appropriate data sources. To begin addressing these challenges, we must begin to understand the data needs and requirements for implementing IFWs, such that we can identify appropriate datasets and data sources. To achieve this, we first present a review of IFW system elements and data needs, followed by findings from a series of qualitative interviews, the majority of which were producers of severe weather warnings and users of hazard, impact, vulnerability, and exposure data in New Zealand.

6.2.1 Elements of Impact-Based Forecasting and Warning Systems

Traditional (hazard-based) severe weather warning systems rely on hazard forecasting and observations, with warning thresholds based on measurable characteristics of the hazard (e.g., minimum/maximum windspeeds, snow depth) (Harrison et al., 2014; Obermeier & Anderson, 2014). IFW systems introduce the human element to early warning systems.

The terminology of IFWs varies across the literature. In this paper, we will use the definitions in Table 6.1, largely from the WMO guidelines on IFW systems (see WMO, 2015)⁹. Kaltenberger et al. (2020) proposed an additional term of 'impact-oriented warnings' to refer to warnings that are independent of the production process (i.e., the warning thresholds or criteria) used to issue warnings. These may be hazard-based or impact-based and include both a tangible and understandable description of expected impacts and clear advice on what to do. Impact-based warnings and impact warnings differ based on the production process used to issue them. The goal of the WMO Guidelines is to evolve severe weather warning services into the final 'impact forecasting and warning' form, where vulnerability and exposure are integrated into the thresholds. Herein, we refer to all as Impact Forecasts and Warning (IFW) systems to align with the goal of the WMO. However, when the type of warning is specified, we will refer to its proper term according to the WMO definitions.

⁹ Note that an update on the WMO Guidelines on Multi-Hazard Impact-Based Forecast and Warning Services is anticipated to be released soon. A more recent resource providing further guidance on IFW implementation was published by the International Federation of Red Cross and Red Crescent Societies and is cited as (Harrowsmith et al., 2020) throughout this paper for further support.

Warning Type	Definition	Example	Data Considered
Hazard- based warning	These are traditional warnings, with thresholds based on the physical characteristics of the hazard (WMO, 2015).	"Bora winds are expected tonight with wind speeds of 20 metres per second" (WMO, 2015, p. 6).	Hazard only
lmpact- oriented warning	These warnings are independent from the criteria of a warning, which may be hazard-based or impact-based and contain information on impacts and clear protective action advice (Kaltenberger et al., 2020).	"Power and phone distribution networks may be disrupted for relatively long periods. Roofs and chimneys can be damaged" (Météo- France via Kaltenberger et al., 2020, p. 30).	Hazard and impact
lmpact- based warning	These warnings include hazard and vulnerability, and are "designed to express the expected impacts as a result of the weather" (WMO, 2015, p. 6).	"Bora winds are expected tonight which may result in delays or cancellation to ferry services" (WMO, 2015, p. 6).	Hazard and vulnerability
lmpact warning	These warnings include hazard, vulnerability, and exposure, and are designed to provide detailed messages down to the individual, activity, or community level (WMO, 2015).	"Ferry services for the island of Brač will most likely be cancelled tonight due to Bora winds" (WMO, 2015, p. 6).	Hazard, vulnerability, and exposure

Table 6.1. Definition and	examples of the w	arning terms use	ed in this paper.
		5	

IFW Systems require knowledge of hazards and their likelihood, as well as the underlying vulnerability and exposure of the assets (e.g., people, infrastructure) at risk (Poolman, 2014; Tarchiani et al., 2020; WMO, 2015). Thus, the four data types needed for an IFW system are hazard (incorporating likelihood), impact, vulnerability, and exposure; defined in Table 6.2.

These types of data form the conceptual basis of IFWs. The lack of data and knowledge around impacts and risks relating to hydrometeorological hazards amongst meteorologists has been identified as a key barrier to implementing IFWs (Harrison et al., 2014; Obermeier & Anderson, 2014; Potter et al., 2021). In particular, exposure and vulnerability data appear to be largely left out of weather warning systems (Potter et al., 2021). Thus, the WMO highlights the need for better collection and access to vulnerability and exposure information and understanding of how vulnerability and exposure data can be integrated into an IFW system (WMO, 2015). The Warning Value Chain provides a framework for understanding how hazard, impact, vulnerability, and exposure data fit into an IFW system.

Table 6.2. The four types of data needed for IFWs.

Hazard	A hazard is "a hydrometeorological-based, geophysical or human-induced element that poses a level of threat to life, property or the environment", such as rainfall, snowfall, high winds, tornado, hail, flood (WMO, 2015, p. 4). In weather forecasting, likelihood is usually combined with hazard (WMO, 2015).
Impact	Impacts are the negative outcomes of an event (Casteel, 2016), and are defined as "a loss of life and injuries, damage to the environment, infrastructure, and private property, often followed by secondary effects like psychological trauma, or disruption of workflow and traffic" (Kox, Lüder, et al., 2018, p. 116). Direct and indirect impacts may be a function of vulnerability and exposure to the hazard (WMO, 2015). Impact data refers to observed post-event impacts and outputs and results from risk/impact models.
Vulnerability	Vulnerability is "the susceptibility of exposed elements, such as human beings and their livelihoods and property, to suffer adverse effects when affected by a hazard" (WMO, 2015, p. 6). Predispositions, sensitivities, fragilities, weaknesses, deficiencies, or lack of capacities that favour adverse effects on exposed elements can increase vulnerability (WMO, 2015). Vulnerability may also be time- and space-dependent.
Exposure	Exposure refers to people, infrastructure, housing, and other tangible human assets that may be affected in an area where hazards may occur (WMO, 2015). Exposure is time- and space-dependent, and it is possible to be exposed to a hazard but not vulnerable.

6.2.2 The Impact Forecasting and Warning Value Chain

Golding et al. (2019) outlined the Warning Value Chain approach for representing the flow of information amongst actors in the warning chain for designing and operating hydrometeorological warnings. The Value Chain consists of six components for an operational IFW system: (1) Observation, Monitoring, and Detection; (2) Weather Forecasting; (3) Hazard Forecasting; (4) Impact Forecasting; (5) Warning; and (6) Decision/Action, as shown in Figure 6.1.

The first component of the Warning Value Chain is Observation, Monitoring, and Detection. Meteorological monitoring and observation are routine practices (WMO, 2018). Observation, Monitoring, and Detection traditionally use ground-based observations (e.g., from weather stations) and remote sensing tools such as satellite and radar imagery for meteorological phenomena (e.g., Brotzge et al., 2013). Social media and crowdsourcing have also become useful tools for detecting and monitoring hydrometeorological phenomena such as tornados, hail, etc. (Harrison & Johnson, 2016).





Weather forecasting makes heavy use of computer modelling, which has seen technological advances in the last half-century to increase accuracy (Bauer et al., 2015). A shift occurred from deterministic forecasting to probabilistic forecasting, with the advent of approaches such as ensemble prediction systems (GFDRR, 2016; WMO, 2015). In South Africa and the UK, the implementation of an Impact-Based Forecasting and Warning system is solely based on probabilistic forecasting (e.g., Neal et al., 2014; Poolman, 2014). With probabilistic forecasting, warnings can be issued earlier using low probabilities that can increase or decrease as confidence in the likelihood increases (Neal et al., 2014). Adding probabilistic uncertainty estimates in forecasts was found to improve both compliance and decision quality amongst technical forecast users (LeClerc & Joslyn, 2015).

Upon producing weather forecasts, Hazard Forecasting may occur. Weatherinduced hazards include flood and flash flood, storm surge, drought, wildfire, snowfall, ice, extreme temperatures, and damaging winds, etc. Weather forecasts are typically used in conjunction with other data/information about a hazard to produce a hazard forecast. For example, storm surge forecasts in the UK are produced using wind forecasts, sea-level models, and wave/tidal models (Flowerdew et al., 2013).

Hazard forecasting is challenging due to errors in predicting atmospheric conditions (Golding et al., 2019; National Research Council, 2006). For example, forecasting for heavy rainfall in a nearby catchment, rather than the catchment that experienced the downstream flooding (Majumdar et al., 2021).

Impact Forecasting is the next step in the Warning Value Chain to identify and understand the potential impacts of a given hazard. Impact Forecasting involves combining meteorological and hazard information with information about exposed and/or vulnerable assets (e.g., people, property) (Merz et al., 2020). Risk modelling can support impact forecasts by providing quantitative assessments of impacts and loss based on the hazard and asset information using a vulnerability/fragility function (Schmidt et al., 2011). However, the use of risk models by practitioners has been met with key challenges around data availability and investment priorities (Crawford et al., 2018). The data needs for this part of the Warning Value Chain are thus the main topic of this paper.

The Warning may be considered the end-product or service of the warning system. Warnings are intended to communicate information about an oncoming hazard and the associated risk(s) to exposed and vulnerable audiences (WMO, 2018). Design and delivery are two key factors in the success of a warning (Golding et al., 2019). Design involves having organisational and decision-making processes, and operational communication systems and equipment (WMO, 2018). Delivery involves disseminating and communicating the relevant information to the target audiences in terms that are understandable and actionable by the audiences (WMO, 2018).

Upon receiving a warning alert or message, recipients face many decisions. The Decision-making end of the Warning Value Chain has received a plethora of social science research. IFWs emerged from that research as a recommendation for inciting the desired responses to warnings (e.g., Ching et al., 2015; Wagenmaker et al., 2011). Much research has since been done to evaluate IFWs on their effectiveness for recipient decision-making indicating that behaviour to impact-oriented warnings varies across studies and hazards (see Casteel, 2018; Morss et al., 2018; Potter et al., 2018; Ripberger, Silva, Jenkins-Smith, & James, 2015; Taylor et al., 2019; Weyrich et al., 2018; Weyrich et al., 2020a; Weyrich et al., 2020b).

Evidence has shown that including impact information with warnings can improve understanding of and response to warnings (Potter et al., 2018; Ripberger, Silva, Jenkins-Smith, Carlson, et al., 2015; Taylor et al., 2019; Weyrich et al., 2018; Weyrich et al., 2020a, 2020b) by aligning with how people already tend to interpret warnings based on the impacts of the severe weather (Williams et al., 2017) and to contextualise the information (Schroeter et al., 2020). However, caution is advised when formulating the messages as research has found that the terminology used in impact-oriented warnings may induce fear and can cause undesired responses to warnings (Ripberger, Silva, Jenkins-Smith, & James, 2015). For example, impact-oriented warnings can incite people to evacuate from hurricanes (the desired behaviour for those at risk) (Morss et al., 2018; Morss, Demuth, et al., 2016; Ripberger, Silva, Jenkins-Smith, & James, 2015). However, impact-oriented warnings and fear-

based messages also have the potential to increase 'shadow evacuations'¹⁰ during hurricanes (Morss, Demuth, et al., 2016), causing gridlock on motorways, increasing exposure, and thus risk, of even more people (Baker, 1991; Lamb et al., 2012; Yin et al., 2016).

For tornadic hazards, Casteel (2016, 2018) found that impact-oriented warnings led to greater behavioural intentions of sheltering in place. Conversely, Ripberger, Silva, Jenkins-Smith and James (2015) found that impact-oriented messages have the potential to 'backfire' as their results showed a decreased probability of taking shelter and an increased probability of evacuating for events with tags of higher impacts such as 'devastating' or 'incredible'. This behaviour could result in people putting themselves at even more risk in situations where sheltering in place is advised (Ross et al., 2015).

The implications for these effects must be seriously considered when designing an IFW system. Limiting the use of fear-inducing messages can enhance the perceived credibility of the messages and reduce the effects of shadow evacuations (Morss et al., 2018). Including clear prescribed actions can also increase perceptions and appropriate intended responses to impact-oriented warnings (Weyrich et al., 2018).

Partnerships and sharing information amongst the various agencies involved are an important aspect of the Warning Value Chain framework (Golding et al., 2019). It has been argued that poor linkages between warning system components have been major causes for warning systems to fail, resulting in disasters (Garcia & Fearnley, 2012). Thus, linking the components into an integrated early warning system is critical to their effective performance (Garcia & Fearnley, 2012). This requires coordination and collaboration across the various agencies and levels that have the relevant information about hazards, impacts, vulnerability, and exposure (Garcia & Fearnley, 2012; Golnaraghi, 2012).

As outlined in relevant guidelines and resources (e.g., Harrowsmith et al., 2020; WMO, 2015), there is a need for not just hazard data, but also impact, vulnerability, and exposure data. However, it is unclear where each of these data types is used in the Warning Value Chain. Thus, the objective of this research is to identify the uses and gaps for hazard, impact, vulnerability, and exposure data within an IFW system for each component of the Warning Value Chain framework. We adopt a Grounded Theory methodology, described next, to collect and analyse interviews to address the question '*what are the data uses and gaps for impact forecasts and warnings?* This will be explored in a New Zealand context to support the country's efforts towards fulfilling the WMO's objectives for member nations to implement IFWs.

¹⁰ Shadow evacuation occurs when people who are not at risk or are not in the official evacuation zone evacuate unnecessarily (Zeigler, Brunn, & Johnson, 1981).

6.3 Research Method

To address the objective of this research, we adopted a qualitative research approach to explore the issues in depth with participants both within and outside of Aotearoa New Zealand. This is opposed to a quantitative approach, which is more relevant for understanding the breadth of results across a population (Mack et al., 2005). We adopted a Grounded Theory Methodology (GTM) to collect and analyse the data. As an exploratory strategy, GTM is particularly useful for research on areas with little to no prior research and where theory building is needed or desired (Fernandez, 2005; Lehmann, 2010; Seidel & Urquhart, 2016). This study employed the Evolved-Straussian Grounded Theory Methodology (ES-GTM) due to its wide use in relevant fields (Matavire & Brown, 2013; Urquhart et al., 2010), its ease of use (Charmaz, 2006; Kelle, 2005), and its allowance for a priori literature review (Hughes & Jones, 2003; Matavire & Brown, 2013).

A purposive sampling method (Chun Tie et al., 2019) was used to target and recruit participants based on their roles in severe weather risk communication and response, and use of impact, vulnerability, and exposure data. After the initial interviews, theoretical sampling guided the rest of the data collection process for this research, which is "the process of data collection for generating theory whereby the analyst jointly collects, codes, and analyses [their] data and decides what data to collect next, and where to find them, to develop [their] theory as it emerges" (Glaser & Strauss, 1967, p. 45)

Thirty-nine (n=39) experts in weather forecasting, warning, response, risk modelling, and data collection and management were interviewed between November 2018 to May 2021 (described in Table 6.3). These 30-60 minute in-person or virtual semi-structured interviews aimed to address the gaps in the literature regarding IFW data needs and sources. Questions asked for participants' thoughts on (1) IFWs (e.g., what they know about IFWs, perceived challenges and benefits, requirements for implementation); (2) what kind of impact, vulnerability, and/or exposure data they use or need, why, and how; (3) the life path of the data (e.g., how it is obtained, used, stored, what happens to it after its intended use); (4) experienced and/or perceived challenges obtaining data required for IFWs and other uses; and (5) thoughts on collecting and using alternative data sources (e.g., social media and crowdsourcing). Interviews were audio-recorded and transcribed verbatim. In support of achieving the objective of this paper, findings regarding the data uses and gaps for IFWs are reported here, while findings regarding data sources and access will be reported in separate papers currently under preparation (See Harrison et al., 2021).

Category	n	General Roles	Representation
Hydro- meteorological	6	Senior, consultant, and communication meteorologists, social scientists, operational manager, senior policy advisors, and public safety managers	International: Argentina, Australia, Austria, United Kingdom, USA
Experts	6	Meteorologists, operational manager, communication meteorologist, senior policy advisor, Weather analyst	New Zealand: MetService, NIWA, Private Weather Service
	1	Regional flood specialist	New Zealand: Greater Wellington Regional Council
Emergency Management and	3	First responder, Senior Hazard Risk Management Advisor	New Zealand National Level: National Emergency Management Agency (NEMA), Fire and Emergency New Zealand (FENZ)
Response Sector	9	Chief operators, communications specialists	New Zealand Regional Level: Auckland, Bay of Plenty, Canterbury, West Coast, Northland, Wellington
	2	Geographic Information Systems (GIS) specialists	Private Industry: Eagle Technology, Independent
Data Creation, Management, Standards, Custodianship	3	GIS specialists, Statistical Analysis specialist	Public Sector: Land Information New Zealand, Stats NZ/DataVentures, West Coast
	1	GIS specialist	Research: Massey University
Risk and Loss	4	Risk modellers	Research and policy: GNS Science, NIWA
Modelling	1	Loss modeller	Research and policy: Victoria University of Wellington
	1	Respiratory Doctor	Practice: Regional hospital
Other	1	Civil Engineer	Practice: Wellington Region Lifelines Group
	1	Agriculture policy coordinator	Practice: Federated Farmers
Total	39		

Table 6.3. Summary classification of interview participants.

A 'low risk' ethics notification was lodged with the Massey University Human Ethics Committee before data collection in 2018. All participants remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by the area of expertise and/or practice, industry, location, or governance level (Table 6.4). Interviews were audio-recorded and transcribed verbatim. Qualitative analysis (including coding and memo-writing) was conducted using NVivo 12 (Bergin, 2011) following the ES-GTM.

Table 6.4. Participant Codes. All participants remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by the area of expertise and/or practice, industry, location, or governance level.

Interview Code	Position	Classification	Location	Government Level
Agriculture/Rural NZ. A	Agriculture policy coordinator	Agriculture/Rural	NZ	National
Data Management Gov. NZ. Nat. A	Senior Resilience Advisor	Data Management	NZ	National
Data Management Private NZ. B	Geospatial Specialist	Data Management	NZ	
Data Management Research NZ. C	GIS Specialist	Data Management	NZ	
Data Management Private NZ. D	GIS Specialist	Data Management	NZ	
Data Management Gov. NZ. Nat. E	Head of Data	Data Management; Governance	NZ	National
EM. NZ. Reg. A	Director	Emergency Management	NZ	Regional
EM. NZ. Reg. B	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. C	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. D	Principal Science Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. E	Principal Advisor Strategy and Partnerships	Emergency Management	NZ	Regional
EM. NZ. Reg. F	GIS Lead	Emergency Management; Data Management	NZ	Regional
EM. Gov. NZ. Nat. G	Senior Hazard Risk Management Advisor	Emergency Management; Governance	NZ	National
EM. NZ. Reg. H	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Nat. I	First Responder	Emergency Management	NZ	National
EM. Gov. NZ. Nat. J	National Operations Manager	Emergency Management; Governance	NZ	National
EM. NZ. Reg. K	Regional Manager	Emergency Management	NZ	Regional
EM. NZ. Reg. L	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. M	Group Controller	Emergency Management	NZ	Regional
Health NZ. Reg. A	Respiratory Doctor	Public Health	NZ	Regional

Code	Position	Classification	Location	Government Level
Hyd. Gov. NZ. Reg. A	Flood EWS Programme manager	Hydrology; Governance	NZ	Regional
Lifelines NZ. Reg. A	Civil Engineer	Lifelines	NZ	Regional
Loss Modelling Research NZ. A	Economist	Loss Modelling; Research	NZ	
Met. Int. A	Science Manager	Meteorology	Internation al	National
Met. Int. B	National Manager Disaster Mitigation Policy	Meteorology	Internation al	National
Met. Int. C	Senior Policy Officer	Meteorology	Internation al	National
Met. Int. D	Senior Social Scientist	Meteorology	Internation al	National
Met. Int. E	Consultant Meteorologist	Meteorology	Internation al	National
Met. NZ. F	Senior Meteorologist	Meteorology	NZ	National
Met. NZ. G	Communications	Meteorology	NZ	National
Met. NZ. H	Public Relations	Meteorology	NZ	National
Met. Int. I	Division Chief/Meteorologist	Meteorology	Internation al	National
Met. Research NZ. J	Meteorologist	Meteorology; Research	NZ	National
Met. NZ. K	Senior Meteorologist	Meteorology	NZ	National
Met. Private NZ. L	Head Weather Analyst	Meteorology	NZ	
Risk Modelling NZ. A	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. B	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. C	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. D	Risk Modeller	Risk Modelling	NZ	National

6.4 Findings and Discussion

We identified two key themes from the interview data: (1) Data uses for IFWs, and (2) the need for more understanding on different types of impacts, discussed next.

6.4.1 Data Uses in the Impact Forecasting and Warning Value Chain

The actors and data uses for IFWs are discussed in the following sections using the Warning Value Chain as a framework for guiding the discussion. The focus of this paper is on the data uses and gaps of the impact forecasting and warning portion of the Warning Value Chain. As such, a brief overview of results on the data uses and actors in the first three components of the Warning Value Chain (Observation, monitoring, and detection; Weather Forecasting; and Hazard Forecasting) will be provided. Following this, results and discussions will be presented for each of the Impact Forecasting and Impact Warning Components.

6.4.1.1 Weather and Hazard Observation and Forecasting

Extensive research has been conducted for weather and hazard observation and forecasting that documents more comprehensive explanations of the data gaps and uses for weather and hazard observation and forecasting (e.g., Gneiting & Raftery, 2005; Saima et al., 2011).

In NZ, meteorological services (e.g., the NZ MetService), hydrological services (e.g., regional and local council hydrologists in NZ), and hydrometeorological services (e.g., New Zealand's National Institute of Water and Atmospheric Research or NIWA) are the primary actors in this value chain component. The NZ MetService monitors the weather, council hydrologists monitor river networks while simultaneously monitoring the weather themselves, and NIWA collaborates with councils to set up monitoring and modelling scenarios for flood early warning systems. Forecasters also use media and social media to monitor areas where radar coverage is lacking (Met. Int. C; Met. NZ. K; see Table 6.4).

Weather forecasting is primarily done by the meteorological services (public and private), as outlined by an NZ meteorologist: "our responsibility is for the forecast of rain until it hits the ground. After that it's the responsibility of other agencies to deal with what that rain does" (Met. NZ. F). By using Numerical Weather Prediction (NWP) models, weather forecasts can provide meteorologists, hydrologists, and lifelines sectors such as transportation with advance notice, such as 48 hours, 3 days, or 5 days, of a potentially hazardous event, which allows for extra preparation time (Met. Int. I; Met. NZ. K). 'Nowcasts' use observational data (e.g., lightning strikes, and spatiotemporal extrapolation of observed precipitation) or very short-range NWP for near-real-time forecasting out to a few hours (Farnell et al., 2017; Kotsuki et al., 2019; Srivastava & Bhardwaj, 2013). It is important to note the differentiation between the data needs for IFWs in terms of timescale; the needs, type, and availability of data for real-time operations and for developing warning systems can differ. For example, social media and crowdsourcing have value in their real-time contribution to a warning system for hazard and impact detection, forecasting, and warning verification

(Harrison et al., 2020), but may be less useful due to the qualitative nature for defining impact warning thresholds. This should be explored in future research.

In NZ, hazard forecasting can be triggered by an initial discussion between the meteorological service and the council hydrologists¹¹. For example, when a NZ MetService forecaster issues a watch or warning, they typically make a direct phone call to the regional council hydrologist to alert them of a potentially impactful rainfall that could lead to flooding. From there a follow-up conversation can occur where the MetService asks:

'are there particular concerns for your region? Is there anything that we should be on the lookout for?' And it's really those interactions that also feed into the decision for 'should this be red?' because if we talk to them after we issue a Watch, but before initiating the Warning, and we say 'look, we're considering a warning for 300mm', they might go 'oh, that's going to flood our entire region' and that might spark that two-way discussion to say 'well, okay, that's really interesting, how concerned are you, what level of flooding do you anticipate?' Which might lead us down the path of 'this should be red (Met. NZ. K).

From this discussion, a watch or warning decision is made, as well as the warning level. The hydrologist possesses more knowledge of the current conditions (e.g., river levels, soil moisture content) and is thus able to contribute to a more informed warning decision (Met. NZ. A, K).

The NZ MetService primarily uses hazard-based thresholds for their watches and warnings, such as "widespread (over an area of 1000 square-kilometres or more) rainfall greater than 50mm within 6 hours or 100mm within 24 hours" (MetService, 2021b). In some cases, the MetService uses more dynamic thresholds depending on the region, antecedent conditions, and feedback from EM groups and councils. For example, in Auckland, the minimum speed of strong winds was lowered for Northeasterly winds as these winds tend to be more damaging in Auckland than Southwesterlies (Met. NZ. K). Furthermore, senior meteorologists also use their tacit knowledge developed over the years to make a judgement call on issuing hazard forecasts and warnings (Met. NZ. K). The meteorologists also use media reports and social media to detect hazards and impacts and inform their warnings (Met. NZ. K).

Floods were the main hydrometeorological hazard that participants discussed. In NZ, participants described how regional councils typically conduct flood hazard forecasting using meteorological forecast data combined with their hydrological data (e.g., river level and flow gauges) (Met. NZ. A), and sometimes their local and historical knowledge of past impacts (EM NZ. Reg. A, B, C). These hazard forecasts usually do not provide impact information, as one NZ regional council flood warnings specialist

¹¹ Not all hydrological services are provided by councils, this depends on the governance structure of the location under study. In the case of New Zealand, hydrological services are provided by councils as per the Resource Management Act 1991.

stated, "we don't then do any quantitative work to ... determine how much impact that flood financially ... damaged, that type of thing. We've not done that in the past to my knowledge" (Hyd. Gov. NZ. Reg. A).

6.4.1.2 Impact Forecasting

The WMO Guidelines briefly suggest that impact forecasting "could be done in a subjective way working alongside [partners], or in an objective way through developing an impact model using vulnerability and exposure datasets as well as meteorological information" (WMO, 2015, p. 3). Interview findings corroborate this suggestion. Figure 6.2 provides a conceptual visualisation, based on our interviews, of the four data types used as inputs for impact forecasts, which then support either the more objective risk/impact modelling approach, or the more subjective impactoriented discussion approach. These two approaches will be discussed next.



Figure 6.2. Data inputs and uses for Impact Forecasts and the two main approaches. The data inputs are represented in the top-left with arrows pointing to the impact forecast component. Blue is associated with hazard, red with impacts, orange with vulnerability, and yellow with exposure. The two approaches of impact forecasting are listed on the left side of the impact forecast box, and the actors involved are listed on the right. More details on the activities used for each impact forecasting approach are provided in the bottom box. The multi-directional arrow between the two impact forecasting approaches signifies that these two approaches are best used complementarily with each other.

Model-Based Impact Forecasting Approaches

Model-based impact forecasting approaches use computer models, algorithms, and quantitative data to identify potential impacts of varying severity. Specifically, risk models require detailed data about the fragility/vulnerability of assets (including buildings, infrastructure, people, vehicles, etc.), dynamic exposure data, and hazard data¹² (Schmidt et al., 2011). In the UK, under the Natural Hazards Partnership (NHP), several models have been developed, or are under development, for real-time forecasting of various hazards and impacts. Two of these models were discussed in interviews.

The Vehicle OverTurning (VOT) Model is an impact forecasting model for UK highway networks during high wind events (Hemingway & Robbins, 2019). The model uses a "classic risk assessment approach" (Met. Int. A), by defining a hazard footprint (i.e., the wind affecting vehicles), and using vulnerability and exposure indices to calculate the risk of vehicles overturning (Met. Int. A). Vulnerability factors are road altitude, number of lanes, road direction relative to the wind, and road attributes (e.g., tunnel, bridge) (Met. Int. A). The exposure index "is split out by vehicle type because we recognise that different vehicles are more susceptible to being overturned than others" (Met. Int. A; see Hemingway & Robbins, 2019 for more). The UK MetOffice does not traditionally possess knowledge or data on traffic flows (Met. Int. A), thus they partnered with agencies that do, namely transportation agencies in England and Scotland (Met. Int. A).

The second impact forecasting model that is under development with the NHP in the UK is the *Surface Water Flooding Model* (SWFM) and is "very much focused on heavy rainfall leading to surface water and that can lead to property flooding, infrastructure flooding, effects on critical infrastructure" (Met. Int. A). Like the VOT, the UK MetOffice lacked the key data or knowledge for this model, including property types (vulnerability) and people movement (exposure). Thus, partnership again played a key role in acquiring this data:

We needed a lot of information on different property types and we needed information on how people move around the city and things like that to account for the fact that people commute in and out ... And ... within our broader partnership under that Natural Hazards Partnership we have an organisation called the Health and Safety Executive and they do a lot of work in the risk assessment space and have a lot of available data. And that allowed us to really produce that whole second part ... Had we not had that; I think we would've really struggled (Met. Int. A).

¹² For the purposes of this study, hazard data will refer to meteorological and hydrological data. For example, wind speed and rainfall data may be considered meteorological hazard data, while river level and river flow data are considered hydrological hazard data.

The approach of the SWFM is slightly different from the VOT as vulnerability is included but not via a vulnerability index layer; rather via classification of the property itself (e.g., commercial properties vs. residential properties) (Met. Int. A). Output is impact grades ranging from lesser to greater impacts to life and safety (Met. Int. A; see Aldridge et al., 2016; Cole, Moore, Aldridge, et al., 2016; Cole, Moore, Wells, et al., 2016 for more).

Similarly, a *Flood Early Warning System* (FEWS) is under development in collaboration with the Samoan Ministry of Natural Resources and Environment for the Vaisagano River in Apia, Samoa, which incorporates outputs from RiskScape¹³. An impact-forecasting component has been integrated into this FEWS to assess life and vehicle road closure safety needs (Risk Modelling NZ. C, D). After a post-event analysis of flooding in Wellington, NZ, it was suggested that risk modelling would have been useful to identify the risk of flood damage to emergency vehicles that drove through floodwaters and were subsequently damaged (Risk Modelling NZ. C). The data/tools used in this system are near real-time rain intensity forecasts based on available NWP models (e.g., PACIOOS 3km grid), rainfall-runoff predictive relationships, exposure of assets (e.g., roads, buildings), LiDAR topography, flood inundation models for different annual exceedance probability events, and vulnerability functions (Risk Modelling NZ. D). More exploration into characterizing these data sources will be covered in future research.

These impact forecasting systems are still under development and testing and thus are not yet fully operational (Met. Int. A; Risk Modelling NZ. C, D). Consequently, impact forecasting models are not currently the sole tool used for designing and issuing IFWs (Met. Int. A). "Impact-oriented discussions" (Met. Int. A) between different agencies and groups are still the main approach for designing and issuing IFWs, which aligns with the suggestion for a subjective approach in the WMO Guidelines (WMO, 2015), discussed next.

Impact-Oriented Discussions

Impact-oriented discussions typically involve many stakeholders that possess the different types of knowledge and information needed to understand and forecast impacts. In New Zealand, the impact forecasting approach tends to be an impactoriented discussion. For example, the Red Warnings from the MetService's new warning system (see MetService, 2019a) are issued based on impact-oriented discussions with hydrologists and EM groups (Met. NZ. K). A Red Warning does not have "fixed thresholds" (Met. NZ. K); it is only issued if the event is expected to produce significant impacts. The information used to inform a Red Warning decision is gathered from directly communicating with the local EM groups and regional/local councils where the impacts are expected to occur (Met. NZ. K). Impacts reported in the media and on social media platforms then help to verify and update warnings (Met. NZ. K).

¹³ RiskScape is an open-source risk modelling software that was co-developed in New Zealand by GNS Science and NIWA (see Schmidt et al., 2011).

Similar impact-oriented discussions occurred when ex-Tropical Cyclone (TC) Debbie and ex-TC Cook made landfall in New Zealand in 2017 and caused significant flooding in Edgecumbe. After ex-TC Debbie impacted the Bay of Plenty Region in NZ from 3-6 April 2017, causing widespread flooding (Cullen et al., 2017), "[the Bay of Plenty] were suddenly faced with ex-TC Cook coming in after that [on 13 April 2017]" (EM NZ. Reg. A). In preparation of ex-TC Cook, the Bay of Plenty EM group had a team conduct a probability assessment to "look at what ... the potential impact [might] be of this event coming in, given the current situation that we already had been impacted" (EM NZ. Reg. A).

This risk assessment helped identify which neighbourhoods needed to be evacuated in response to the forecast for ex-TC Cook (EM NZ. Reg. A). It relied on discussions and collaboration where decision-makers and stakeholders shared knowledge of recent impacts caused first by ex-TC Debbie that made some areas and assets more vulnerable to damage/flooding from ex-TC Cook (EM NZ. Reg. A). Additional hazard information such as river and stream levels were also shared (EM NZ. Reg. A).

Impact-oriented discussions also make use of decision support systems and integration of spatial data to identify exposed and vulnerable areas and assets. For example, an impact-based decision support system was set up between the National Weather Service and emergency managers in New York to support forecast updates, response decisions, and public and partner messaging (Hosterman et al., 2019). Integration of spatial layers has also been used to analyse social vulnerability to extreme precipitation in Colorado (Wilhelmi & Morss, 2013) and to support flood risk management decision-making in Brazil (Horita et al., 2015). More work is needed to explore these similar efforts for integrating hazard, impact, vulnerability, and exposure (HIVE) data to support IFWs in New Zealand.

These examples demonstrate various approaches and methodologies for conducting impact forecasts. Further examples are provided in a review of Impact Forecasting capabilities worldwide (see Schroeter et al., 2020). The most appropriate approach depends on the goals of the system, the available data, and the type(s) of hazard(s) to be forecasts:

... if you're doing the collaborative approach with very broad scenario-type impacts, then perhaps the update frequency can be less. If you're doing a more robust, sort of bespoke risk algorithm, then I would hope the update frequency would be higher because you are trying to actually capture risk given the current circumstances. And if you're not updating frequently, then you're using a historical dataset to inform the current risk and I don't think that's very accurate (Met. Int. A).
In these examples, all four data types (hazard, vulnerability, exposure, and impacts) come into play for impact forecasting (Figure 6.2). Model-based impact forecasting uses quantitative datasets, typically in spatial format (e.g., GIS shapefiles). Alternatively, impact-oriented discussions utilise the forecasts along with more holistic qualitative information in the forms of tacit knowledge and experience, observations from social media and/or media reports, meteorological observations, hydrological observations, etc. Discussions between stakeholders bearing different knowledge are critical for the subjective approach. While objective model-based approaches may be considered the more "sophisticated" approach (e.g., Schroeter et al., 2021), our participants indicated that both approaches have value, and the most effective approach is to use them together rather than preference one over the other (EM. NZ. Reg. D). This exemplifies the importance of incorporating different scientific epistemologies into transdisciplinary science and research to inform policy and decision-making (see International Science Council, 2021).

6.4.1.3 Impact warning

Findings from the interviews show that impact warnings introduced new needs for setting warning thresholds, discussed below.

Defining Impact Thresholds

Hydrometeorological warnings are based on thresholds traditionally set around physical hazard characteristics, such as 'x' amount of rainfall in 'y' amount of time. Participants indicated that previous event impact data can help determine warning thresholds (Met. Int. A, B, D; EM. NZ. Reg. B, C). For example, investigating the drivers of the resulting impacts from a recent event can help redefine thresholds such that they are specifically impact-related (Met. Int. A). This can involve identifying the rainfall and windspeed leading up to the event that resulted in the impacts to determine which thresholds result in such impacts (Met. Int. A), such as 'x' windspeed will result in 'y' type of damage to 'z' structures. A NZ EM official highlighted the value of cataloguing impact data to prevent "history from repeating itself", by allowing officials to identify trigger points (or thresholds) based on past events (EM. NZ. Reg. B). Like the impact forecasting approaches, countries and regions appear to use different approaches to defining impact-based thresholds, varying in how systematic they are (Figure 6.3).

Tacit knowledge and experience are less systematic ways to identify future event thresholds. NZ EM officials use tacit knowledge to identify different river level impacts (EM. NZ. Reg. B, C). The Northland EM agency works with council hydrologists to informally forecast these flood impacts and inform responders and communities (EM. NZ. Reg. B). Similarly, West Coast EM officials have identified and logged river level thresholds for specific points in the river network and are looking to do this more systematically with GIS.

The UK MetOffice has a slightly more systematic approach to documenting tacit knowledge. They use a series of "impact tables" (Met. Int. A) that were developed in consultation with local EM to "discuss what impacts are observed at different levels" to identify response capacity levels (Met. Int. A). These tables helped the MetOffice understand "how different impacts have occurred related to different hazards" (Met. Int. A). The MetOffice then uses a risk matrix (see WMO, 2015, p. 14), with impacts on one dimension and likelihood on the other to relate the level of impact from the tables to the level of impact in the matrix (Met. Int. A). The MetOffice still relies on contacting EM authorities to verify the most appropriate warning level from the impact matrix (Met. Int. A). This remains a "continual collaborative process" where the warning can be updated based on feedback from the authorities on the ground (Met. Int. A).



Figure 6.3. Data inputs and uses for defining Impact Thresholdsfor the Impact Warning Component. This component uses similar approaches to Forecasting Impacts: the objective Risk/Impact Modelling approach, or the subjective Impact-Oriented Discussion approach, as shown at the bottom of the figure. All four data types feed into both approaches. Asset data is represented in grey as it overlaps with both vulnerability and exposure data. The actors are listed on the side of the Impact Warning box. Dynamic vulnerability (orange) and dynamic exposure (yellow), such as antecedent physical conditions (e.g., drought or over-saturated soil), human response capabilities, or human locations, are needed to update thresholds so that the risk estimates are accurate.

Still, this is less systematic than model-based techniques, and participants envisioned using impact data in risk/impact models to classify what is hazardous. Engineers often use impact data to inform the vulnerability function in risk or impact models by associating levels of damage to hazard intensities for a given asset (Pita & de Schwarzkopf, 2016; Tarbotton et al., 2015). Post-event damage surveys can be used to develop and calibrate vulnerability functions and identify impact thresholds (Met. Int. A, B). For example, an event database with empirical impact data from damage surveys can provide evidence about how hazard intensity will cause damage (Met. Int. B). Eventually, these models can help identify the causes of certain impacts. This includes determining which wind speed or rainfall amount led to high impact events (Met. Int. A). By understanding the mechanisms that led to the impact, forecasters can improve their models and identify the specific impact and warning thresholds (Met. Int. A).

Currently, risk modellers do not appear to be involved in the warning operations of some of the participating meteorological, hydrological, and emergency management agencies, such as in NZ. The VOT developed by the UK MetOffice is undergoing evaluation with operational meteorologists (Hemingway & Robbins, 2019). Furthermore, the FEWS that is currently under development for the Vaisagano River in Apia, Samoa has been developed in collaboration with risk modellers and developers of the RiskScape software and with the Samoan Ministry of Natural Resources and Environment, indicating that risk modelling will play a role in the operational FEWS. There is further interest and future work planned for developing the capabilities of risk model frameworks like RiskScape to conduct rapid impact modelling assessments for warning systems and response and recovery (Risk Modelling NZ. A).

Thresholds based on impact data alone (e.g., observed impacts, post-event damage assessments) are less useful if there is a change in the underlying vulnerability and exposure of assets due to human activity, seasonality, etc., or if the event is more extreme than anticipated due to climate change:

I think there can be provided a bit of a false sense of what's going to happen? Like people think that that's going to be the impact, but there's always a chance that a) the forecast, there might be an area that might get more rain than that, also we're in a climate change, so things are just a little bit more intense now. And there might be another factor that has never been considered, and it could just be something that's just a change in the landscape, that sort of meant that the storm might just have a bit more sting in its tail and might, you know the local experience might, it can tell you about what's happened but it won't tell you about the really extreme event (Met. Research NZ. J).

There is a need for dynamic vulnerability and exposure data to account for other factors that may change impact thresholds, such as antecedent physical conditions (e.g., drought or over-saturated soil), human response capabilities, or human location (Agriculture/Rural NZ. A; Met. NZ. K). These factors can interact with each other to exacerbate impacts. For example, a drought in the South Island of New Zealand caused farmers to be more vulnerable to medium and low impact snow events than normal, due to drought-impacted feed supply:

This year the drought's been so widespread that we're really low on specific types of feed that are really common in a farmland system normally, really easy to source ... So ... that's ... actually put the rural community at a really big risk of not being able to cope for events that otherwise would be quite easy to recover from (Agriculture/Rural NZ. A).

Thus, a low impact snow event can become high impact due to underlying vulnerability conditions caused by a separate event (e.g., drought); discussed further in Section 6.4.2.4. This example also demonstrates the need to consider cascading hazards, as stated by Pascal et al. (2006); Potter et al. (2021); and WMO (2018).

Updating Warnings

Real-time or near real-time hazard and impact data streaming onto meteorologists' screens are highly valuable to participating NMSs, as real-time information allows meteorologists to adjust warnings on the fly (Figure 6.4):

... the most important thing here is that [the European Weather Observer is] real-time data, and we will have this data available on our screens as forecasters wants (sic) it ... So ... in our view it's really important for the forecaster to have real-time data available in order to adjust warnings, for example, if things are more severe out there, ... to upgrade the warning or vice versa (Met. Int. E).

This impact data can show forecasters and responders "what's happening now versus what you predicted is going to happen" (EM. NZ. Reg. C). For example, in New Zealand, one MetService participant described how they can upgrade an Orange Warning to a Red Warning if they observe impacts that they were not expecting, "coupled with a forecast with a lot more to come, and that might then prompt you once you've seen those impacts to then escalate to Red" (Met. NZ. K). Social media, crowdsourcing, and media reports are valuable for gathering real-time impact data, which is useful for updating warnings on the fly (Met. Int. E; Met. NZ. K; Krennert, Pistotnik, et al., 2018), despite the limitations in quality and trustworthiness of social media and crowdsourced data (Harrison & Johnson, 2019).



Figure 6.4. Data inputs and uses for updating warnings.Real-time or near real-time data/information on the hazards, impacts, vulnerability, and exposure are key for updating warnings. Updating warnings starts from the initial impact warning, which leads to actions taken by the warning audience. These actions can determine what human impacts occur (e.g., injuries, fatalities, traffic impacts). The hazard can change in space and time, also causing different impacts. Warnings can be updated with this up-to-date information. Dynamic vulnerability and dynamic exposure can also change as impacts unfold and as people take protective actions.

Verification

To continuously improve their services, warning organisations routinely verify and validate models, forecasts, and warnings. Participating NMS officials described how warnings have traditionally been verified based on the occurrence of the forecasted hazard, with little investigation into what happened 'on the ground' (Met. Int. A, B, D, I; Met. NZ. F). This verification involves meteorological observations to confirm the occurrence and timing of the hazard compared to the forecasts and warnings (Sharpe, 2016; Wilson & Giles, 2013). Participating NMS and EM officials indicated that verifying impacts is the next step towards improving forecasts and warnings, as shown in Figure 6.5. This can help determine if they are under- or overwarned and strive for the "right buoyancy of [the] warnings" (Met. Int. E).



Figure 6.5. Verification of impact warningsrelies on observed hazard and impact data. The impact warnings that were issued lead to certain actions, which then result in impacts (or no impacts). These impacts are then used to verify the warnings that were issued. The feedback loop from learning from verification to adjust procedures and impact thresholds is represented by the arrow pointing back from verification to impact warning.

Along with observational hazard data, impact data from post-event assessments, media reports, social media, and other crowdsourcing or citizen science efforts are used for warning verification (Met. Int. D, E; Met. NZ. F). Post-event assessments offer credible, systematic ways of collecting impact data, useful for post-event analyses (Harrison et al., 2015; Pita & de Schwarzkopf, 2016). For example, NMSs use assessments to determine whether they 'cried wolf' (Met. Int. B). Most participating NMSs use these assessments for post-event analysis and evaluation (Met. Int. B, C, E, I). The NMSs in the USA and Austria use these data, along with crowdsourced storm reports, to build event databases that allow for event comparison, past event learnings, and future preparedness (Met. Int. E, I; Krennert, Kaltenberger, et al., 2018; Krennert, Pistotnik, et al., 2018).

IFW verification challenges exist. Hemingway and Robbins (2019) questioned how verification can be done if the impacts are reduced due to the warnings themselves. These concerns were echoed by Potter et al. (2021), who suggested several approaches for verifying IFWs if and when impact data is not available. These include using predetermined, dynamic hazard-based criteria, or identifying the mitigation decisions that were made in response to the warning (Potter et al., 2021). A NZ meteorologist in our study indicated that verification is not binary (Met. NZ. K). NZ MetService meteorologists use a combination of observational hazard data, impact data from media reports, and experience in a 'pragmatic' approach to assigning a

verification score to the warnings. Thus, in line with Potter et al. (2021), a multi-factored and multi-data approach grounded in Pragmatism and experience presents an alternative to verifying IFWs.

6.4.1.4 Decision/Action

After a forecast or warning is issued, the recipients must decide on appropriate protective actions. Investigating the effectiveness of IFWs for warning recipient decision-making was out of the scope of this research, yet some notable comments were made by participants on this topic that should be acknowledged.

Potter et al. (2021) found that NMSs in both New Zealand and internationally perceived the benefits of IFWs to include a reduction of warning fatigue and cry-wolf syndrome. However, our findings have some differences from Potter et al. (2021). While some NMS officials in our study indicated that IFWs may potentially reduce the effects of warning fatigue and cry-wolf, two NZ participants (EM, meteorologist) had concerns about crying wolf when warning for impacts that did not eventuate (EM. NZ. Reg. E; Met. Research NZ. J). They attribute this to the added uncertainty of forecasting and warning for impacts across space and time for different audiences (Met. Research NZ. J), stating "it's not just one person, it's a population you're forecasting for over an area, and people are going to experience different effects" (Met. Research NZ. J). There are also uncertainties around human behaviour, making it difficult to model and forecast impacts based on human behaviour (Risk Modelling NZ. C). This impact uncertainty compounds the pre-existing uncertainty from weather and hazards forecasting (Met. Int. F; Golding et al., 2019), as outlined by a US-based meteorologist who highlighted the uncertainties of forecasting winter weather hazards:

From just the base physical science aspect, I don't know that ... we're at a level at least with consistent, calibrated probabilistic forecasts that we can predict those kind of events, or ... consistently in a calibrated way predict what the intensity of a tornado is going to be or the size of hail (Met. Int. F).

More information on hazards and impacts such as hail size, tornado intensity, and resulting damage is needed to calibrate models for hazard forecasting (Met. Int. I). For example, reports of hail size from crowdsourcing and citizen science platforms can help calibrate radar algorithms used for forecasting precipitation types (Met. Int. I), which can then inform hazard and impact forecasts (e.g., damaging hail to crops) by overlaying the hazard forecast with asset data (e.g., crop fields, infrastructure). However, more work is still needed in this area (Met. Int. I).

IFWs may help individuals conceptually link hazards to impacts, thus increasing their risk awareness. For example, with regards to severe thunderstorm-induced asthma attacks, one NZ-based respiratory doctor indicated that people who experience severe thunderstorm-induced asthma attacks and have never experienced an asthma attack before may not realise what is happening to them, resulting in a delay of them seeking appropriate care (Health NZ. Reg. A). IFWs containing information on thunderstorm-induced asthma may benefit warning recipients by clearly drawing the

links from the hazard(s) (i.e., the thunderstorm and pollen) to the impacts (i.e., asthma attacks), helping individuals to more quickly identify what may be happening to them and seek proper care.

Studies have found that IFWs may be more effective if they include prescribed actions (Weyrich et al., 2018). Many of our participants from the EM sector reiterated the need to include prescribed actions in the warnings (EM. NZ. Reg. A, E, H). However, one NZ participant from the flood warning space was skeptical of such prescribed actions (Hyd. Gov. NZ. Reg. A), preferring to issue general warnings over specific action-based warnings. This is due to the large size of the communities in their jurisdiction, the high number of stakeholders for which the warning would need to be designed, and concerns with liability and control when introducing prescribed actions (Hyd. Gov. NZ. Reg. A). They questioned whether it would be more effective to issue a general flood warning with the location and duration (Hyd. Gov. NZ. Reg. A). This would shift the responsibility back to property owners and residents where

people then can react to that in the way that they deem appropriate, and that then moves the responsibility back on the property owner, the residents to be able to react. Obviously we're there to help and guide and advise, but we just haven't got the reach and the people so ... you need enough bubbles of control out there that they're able to self-manage and deal with it themselves ... so really it's get the message out as quickly as possible and let them react as they see fit. That's sort of where our, for me anyway, our responsibilities should end, we can't help everyone (Hyd. Gov. NZ. Reg. A).

This participant suggested having multiple 'bubbles of control' across the community through self-management, rather than following prescribed actions that may not be situationally appropriate. They acknowledged this would require heavy early community engagement efforts, so communities and individuals understand their local flood risk and develop their emergency response plans for floods. This raises questions around the needs of the warning recipients and the communities, and the need to identify what they deem the most appropriate approach for warnings, as suggested by Thomalla et al. (2006), and done by Tarchiani et al. (2020) in their design of a flood IFW system in Niger. This would align with the concept of people-centred early warning systems (Basher, 2006). Still, the evidence from the literature suggesting the efficacy and value of including guidance messaging cannot be denied (e.g., Weyrich et al., 2018).

In terms of data and information for the Decision component of the Warning Value Chain, the information produced from the previously described components (Observation, Weather Forecasting, Hazard Forecasting, Impact Forecasting, and Impact Warning) and other social data such as past experience and behaviours of peers influence the recipients' decision-making. Furthermore, outcomes of these decisions produce new impact information resulting from the event (Hemingway & Robbins, 2019). Thus, rather than being an input, impact data can also be an output from the Decision Component.



Figure 6.6. This figure represents the Warning Value Chain and the associated data inputs and outputs for each component, along with the activities and actors for each value chain component. Uncertainty compounds at every stage of the chain and is represented by the compounding uncertainty wedge.

Each component links together to complete the Impact Forecasting and warning Value Chain. Figure 6.6 provides a visual representation of the connected chain, starting with Observation, Monitoring, and Detection, where meteorological services and hydrological services use raw observational hazard data. These observational hazard data are then used to formulate Weather Forecasts in the next component. Weather Forecast data and raw observational data are then used to produce hazard forecasts, in conjunction with impact data if available. These hazard forecasts can then be combined with impact, vulnerability, and exposure data to produce impact forecasts, using models, impact-oriented discussions, or both. After an impact forecast is produced, the meteorological, hydrological, and EM services must decide on the appropriate impact warning level to issue to the appropriate audiences. Hazard, impact, vulnerability, and exposure data are all utilised to define the warning thresholds, update the warnings, and verify the warnings. Once a warning is issued, the warning audiences must decide on the appropriate action to take (if any). These actions (or lack thereof) may then result in forecasted or un-forecasted impacts. Uncertainty begins at the beginning of the value chain with weather forecasts. From there, the uncertainty compounds, as demonstrated by the compounding uncertainty wedge, which can influence warning audiences' decisions to act.

6.4.2 Need for more understanding on different types of impacts

Ranges of impacts occur from hydrometeorological hazards that can be physical (e.g., damage to infrastructure and the built environment), economic (e.g., financial losses), social/societal/cultural (e.g., traffic delays, disruptions to large events, educational services, health services), or environmental (e.g., pollution spills, wildlife endangerment). The impacts can be direct (e.g., injuries, casualties, building damage) or indirect (e.g., job loss, displacement, illness) (Lindell et al., 2006). Discussions with participants revealed a gap in modelling, forecasting, and communicating some indirect and social impacts.

6.4.2.1 Social Impacts

When considering the impacts of severe weather, the initial thinking is towards direct, physical impacts, and financial and economic impacts/losses. Examples of the IFWs provided in the WMO Guidelines appear to have a heavier focus on direct and physical impacts (e.g., wind damage to trees and power lines), with some social impacts considered such as traffic delays. However, all types of possible impacts must be considered. For example, in Argentina a severe weather warning created major impacts due to a soccer match, a culturally significant event:

... here in Argentina, football, soccer ... it's very, very important ... And there were (sic) South American final soccer cup here, and the final soccer cup was between the two most important soccer clubs, and everybody was very crazy. There was a very, very heavy storm and the match had to be postponed. You cannot imagine how was (sic) that day for us, and it was just a soccer game, you know. [High-level authorities were] calling my boss to 'tell me what's going on, what

do we have to do' you know, because people really got very, very angry. They blame us, on the match. So, what can I do? (Met. Int. D).

This example indicates the importance of identifying cultural impacts from hydrometeorological hazards and including this in communications. Further, messaging needs to be more culturally sensitive/aware and empathetic. As highlighted by Ayeb-Karlsson et al. (2019), several cultural and social limitations exist in disaster preparedness, particularly warning response, and there is a need to understand the cultural contexts influencing preparedness. Doing so would further align with the concept of people-centred EWSs (Basher, 2006).

A shift is occurring in the NZ risk modelling space towards social impact assessments in response to needs identified by the EM sector. A NZ-based risk modeller described the importance of considering social impacts beyond just the direct, physical impacts, to provide a fuller picture (Risk Modelling NZ. B). For example, direct impacts such as power and other critical service outages may cause population displacement, which is often not modelled.

Indeed, an official from NZ's National Emergency Management Agency (NEMA) echoed the need for documenting indirect impacts (EM. Gov. NZ. Nat. G). While NEMA collects impact and loss data for the Sendai Framework (UNDRR, 2015b), the face value of the impact/loss data lacks the deeper meaning of the indirect, potentially longer-term, impacts because "recording a road outages as a 1 or a 0, you know it was out or it wasn't out, doesn't give you any scale of duration, the impact, lost productivity, stress on people's lives" (EM. Gov. NZ. Nat. G).

Following flood events in NZ, risk modellers and other researchers have disseminated post-event surveys to the affected people. Challenges arose around collecting and using social impact data for modelling purposes (Risk Modelling NZ. C). A range of physical and non-physical impacts occurred after the initial event, such as injuries (physical), and anxiety and stress (non-physical). While they were able to capture some of this information, data collection was not well planned, thus it did not result in a high-quality dataset useful for modelling (Risk Modelling NZ. C). Conducting well-planned longitudinal surveys in collaboration with more experts such as social scientists were suggested to increase the usefulness of the surveys (Risk Modelling NZ. C). Furthermore, this participant suggested that risk modellers may not be the best suited for this type of social impact modelling as their expertise lies in modelling buildings and infrastructure. This highlights the need for multidisciplinary work for IFWs (Merz et al., 2020).

6.4.2.2 Health Impacts

Health impacts appear to also have gone largely unnoticed in the IFW literature, aside from the health impacts of extreme heat (Harrowsmith et al., 2020; Pascal et al., 2006; WMO, 2015). Recent events have brought the risk of thunderstorm-induced asthma attacks to light in Australia and New Zealand (Sabih et al., 2020; Thien et al., 2018). Before the 2016 Melbourne epidemic thunderstorm asthma event, this risk was not forecasted or warned for. Following this event recommendations included developing an epidemic thunderstorm asthma forecast (Hew et al., 2017; Thien et al., 2018), which is now provided by the Australian Bureau of Meteorology (Met. Int. B; Bannister et al., 2021). A similar event occurred on a smaller scale in Waikato, NZ in 2017 with no advanced warning (Health NZ Reg. A; Sabih et al., 2020). There has thus been a push to capture such severe weather health impacts (Aitsi-Selmi & Murray, 2016), which must then be integrated into IFWs. Health impacts from poor air quality due to pollution (Gao et al., 2015), wildfires (Cisneros & Schweizer, 2018), etc., are also increasingly important and must be considered when designing IFWs.

6.4.2.3 Urban and Rural Impacts

The difference between urban and rural impacts introduces another challenge when communicating impact information, as meteorological hazard impacts differ between these. For example, urban heavy rainfall may create more impacts due to surface flooding and traffic delays (Met. NZ. E). However, equivalent rural heavy rainfall in locations exposed to such rain often (e.g., West Coast, NZ) may experience fewer impacts on the resident population, with minimal flooding (Met. NZ. E), and in some habitually dry regions, the heavy rainfall may be welcomed. In contrast, a tramper/hiker or hunter on the Department of Conservation tracks may need forewarning due to rapid river rising, posing a life safety risk (Met. NZ. E). This conundrum echoes the findings of Potter et al. (2021), who questioned whether IFWs are designed for individuals or society. This key question must be addressed and was raised by participants (Met. NZ. F, K), as exemplified by a NZ-based meteorologist:

... exposure and vulnerability are the two key questions and for some warnings, a short lead time is fine, but for others, it's not, and it very much depends on the user. A farmer for a snow warning is going to need a lot more time to get stock down out of the high country than you as an individual getting a wind warning and needing to go down and tie up the trampoline, right? So if you want to tie down the trampoline, you can do that in a couple hours when you get home, but you know, if you're a farmer, you might need to know that there's snow coming the day after tomorrow to move the stock (Met. NZ. K).

This relates to the need for vulnerability and exposure to be included in IFWs and the identification of appropriate audiences' relevant trigger points (or thresholds) (Zhang et al., 2021). It is not enough to collect only past event impact data and expect the impacts to be the same for future events or different populations. However, as Potter et al. (2021) found, vulnerability and exposure are often overlooked when designing EWSs. This may be due to challenges in capturing the dynamic nature of vulnerability and exposure, discussed next.

6.4.2.4 Dynamic Exposure and Vulnerability

Exposure and vulnerability change over space and time (The World Bank, 2014). Thus, up-to-date datasets for this information are critical for accurately representing the level of risk (Met. Int. A, D; Met. NZ. F, K). If exposure data is not current, the risk assessment will be inaccurate, resulting in improper warning buoyancy (i.e., overwarning or under-warning will occur) (Met. Int. A, B). Thus, a UK-based meteorological risk specialist hopes "we all recognise the importance of updated vulnerability and exposure data is a challenge for participating NMSs, such as the UK MetOffice who are determining how best to maintain up-to-date data and integrate it into an IFW system (Met. Int. A).

In recent years, research has emerged investigating the dynamic nature of exposure and vulnerability (Terti et al., 2015) and methodologies for integrating dynamic exposure and vulnerability into risk analyses (Gallina et al., 2016; Shabou et al., 2017; Wilhelmi & Morss, 2013). Our findings echo those from Merz et al. (2020) regarding the role of dynamic exposure and vulnerability in impact forecasting, and the need to integrate dynamic exposure and vulnerability into models through collaboration across sectors (e.g., social science, engineering, and natural sciences). Other options commonly used include visual overlays for integrated multi-hazard decision support platforms supporting impact-oriented discussions. There is a need to identify the potential sources, creators, and custodians of dynamic exposure and vulnerability data and investigate how it can be accessed and used for IFWs.

6.5 Conclusions

This exploratory study identified the uses and gaps for hazard, impact, exposure, and vulnerability data for implementing severe weather IFW systems within the Warning Value Chain framework presented by Golding et al. (2019). The qualitative nature of data collection and analysis herein limits the generalisability of results beyond the participants. However, the qualitative approach offers an in-depth understanding of a problem not readily available from quantitative approaches (Blumer, 1969; Miles & Huberman, 1994; Patton, 2002).

Our findings support existing research pointing to the need for creating, gathering, and using impact, vulnerability, and exposure data for IFW systems. This study builds on existing research by identifying where each data type (hazard, impact, vulnerability, and exposure) can be used for each Warning Value Chain component and offers examples of alternative approaches for impact forecasting and for defining

impact thresholds. For example, models provide an objective approach while impactoriented discussions provide subjective and flexible approaches. These approaches should be used complementarily.

Findings from this research provide new insight into a growing need to identify, model, and warn for social and health impacts, which have typically taken a back seat to modelling and forecasting for physical and infrastructure impacts. The type of impacts to be forecasted and warned for depends on the intended audiences of the warnings.

While this study identified the uses and gaps for hazard, impact, vulnerability, and exposure data, questions remain around who collects, creates, stores and manages these data (particularly impact, exposure, and vulnerability). Future research should explore mapping the various sources, creators, users, and custodians of the relevant data needed for IFWs.

This study has a heavy focus on hydrometeorological hazards and impacts due to the experiences of most of the research participants. Floods are the most frequent and costly hazard in NZ (Rouse, 2011), and thus were the most common examples that NZ participants drew from during interviews. Further research should verify the findings of this research against other hydrometeorological hazards both within and outside of New Zealand.

	Chapter One			
Part 1	Chapter Two			
	Background and Context			
	Chapter Three			
	Research Design			
Part 2	Chapter Four			
	Methods Results and Overarching Findings			
	Chapter Five			
	Volunteered Geographic Information for people-			
	centred severe weather early warning: A			
	Chapter Six			
	Identifying the data uses and gaps for severe			
	Chapter Soven			
	Identifying data sources for severe weather			
Part 3	impact forecasts and warnings in Aotearoa New			
i dit o	Zealand			
	Chapter Eight			
	A socio-technical analysis of sharing and			
	governing hydrometeorological impact data in			
	Aotearoa New Zealand			
	Chapter Nine			
	Integration of Partnerships and Collaboration to			
	Support IFWs and HIVE Data Collection, Access,			
	and Use			
	Chapter Ten			
Part 4	Discussion and Conclusion			
L				

Position of the reader

This chapter presents the third manuscript of this doctoral study prepared for publication. The focus of this manuscript is to identify existing and potential sources of hazard, impact, vulnerability, and exposure data to support IFWs in New Zealand. After developing an understanding of the flow of information along the IFW Warning Value Chain and identifying gaps in supporting data, as presented in the previous chapter, identifying the sources of data to address those gaps was the next objective. This paper aimed to achieve Objective 3.1 Identify the creators, collectors, and users of HIVE data that is relevant to severe weather IFWs; and Objective 3.2 Identify the inhibitors of and facilitators to collecting and using HIVE data to support the implementation of an IFW system in New Zealand for severe weather hazards.

This manuscript was published in 2021 in the International Journal of Disaster Risk Reduction:

Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E. H., & Johnston, D. (2021).
'Where oh where is the data?': Identifying data sources for hydrometeorological impact forecasts and warnings in Aotearoa New Zealand. *International Journal of Disaster Risk Reduction*. https://doi.org/10.1016/j.ijdrr.2021.102619

The published form is available in Appendix L.

7.1 Abstract

Early Warning Systems are a key component to building preparedness and response capacities to hydrometeorological hazards that continue to affect people worldwide. Notable historic events have revealed gaps in current hazard-based warning systems. Impact Forecasts and Warnings (IFWs) have been proposed to fill these communication gaps by re-centring the warning thresholds and language around the consequences, or impacts, of the hazard(s), rather than just the physical characteristics. However, research has shown that implementing IFWs requires not just hazard data, but also data on impacts, vulnerability, and exposure to understand the risk of impacts.

Using Grounded Theory Methodology, we conducted a series of interviews with users and creators of hazard, impact, vulnerability, and exposure (HIVE) data to identify data sources and understand how these data collected and created to support the implementation of IFWs. We focus the study on the New Zealand context to support the country's efforts towards implementing IFWs.

Our findings indicate that many sources for HIVE data exist that are collected for other uses (such as for response efforts for disaster and emergency events, and for research) and have relevant application for IFWs. Our findings further suggest that priorities, motivation, and interest within organisations influence how well data is collected. Moreover, agencies tend to prefer official data, but official data has limitations that unofficial data may address, such as timeliness. To that end, a tension exists between the timeliness and trustworthiness of data needed for emergency response and warnings.

7.2 Introduction

The last two decades have seen a paradigm shift in disaster risk reduction from reactive post-disaster response and recovery to proactive preparedness and mitigation. Early Warning Systems (EWS) are a key component for better preparedness (UNDRR, 2015b). Past severe weather events exposed major communication gaps between meteorologists and warning services and target audiences, resulting in widespread losses including death, injuries, and damage. For example, following Typhoon Haiyan, which resulted in over 6,293 deaths, 28,689 injuries, and 1,061 missing people in the Philippines (McPherson et al., 2015), it was found that 88% of warning recipients did not understand messages about 'storm surge' and 95% of warning recipients did not evacuate because they did not expect the storm to be so catastrophic (Ching et al., 2015). It was recommended that "warning messages ... should be conveyed in terms understood by the population at risk" (Ching et al., 2015, p. 34).

This communication gap is a result of both technical failings and human behaviour. The World Meteorological Organization (WMO) posited that meteorologists and warning services do not typically consider the warning audiences' current state of vulnerability and exposure at the time of the warning or at the expected time of impact (WMO, 2015). Furthermore, warning audiences fail to understand and respond to the warnings effectively due to ambiguous terminology (Ching et al., 2015), lack of trust in the warning system and service provider (Taylor et al., 2018), and warning fatigue (Mackie, 2013). As such, EWSs continue to evolve in attempts to reduce the effects of these factors.

In the hydrometeorological space, Impact Forecasts and Warning (IFW) systems are an advancement of traditional hazard-based EWSs. IFW systems provide an opportunity to integrate knowledge and understanding of exposure, vulnerability, and impacts into an EWS to build new warning thresholds that better align with the position, needs, and capabilities of target audiences (Harrison et al., 2022). However, challenges have been identified around identifying and accessing the required data sources for IFWs (Hemingway & Robbins, 2019). Harrison et al. (2022) identified the specific needs and uses for hazard, impact, vulnerability, and exposure (HIVE) data in an IFW system. We continue this work by identifying sources for these required datasets. We next present a review of existing HIVE data sources from the literature, identifying data gaps that we explore further through a series of key-informant interviews.

7.3 Impact Forecasting and Warning Data Gaps and Sources

Warning agencies and meteorologists are challenged with accessing appropriate data to support the decision-making required for IFW systems (Harrison et al., 2014). The lack of impact, vulnerability, and exposure data is a major obstacle to implementing IFWs (Potter et al., 2021). Beyond IFWs, these data are also needed for enhancing our understanding of disaster risks and subsequent mitigation and reduction, as per the Sendai Framework for Disaster Risk Reduction (UNDRR, 2015b).

The following literature review identifies some existing sources of HIVE data for IFWs. These data sources and datasets are summarised in Table 7.1 and described further in the following sections.

7.3.1 Hazards

For this study, hazard data refers to meteorological, hydrological, and hydrogeological data. For example, windspeeds and rainfall amounts may be considered meteorological hazard data, while river levels would be hydrological hazard data, and slope data are useful for landslide hazards (i.e., hydrogeological hazards).

Understanding of severe weather hazards primarily comes from hydrometeorological observations and measurements usually gathered quantitatively through specialised equipment, such as rain gauges, river gauges, and remote sensing. This information helps to estimate the timing, direction, and magnitude of the hazard for forecasts (Heinselman et al., 2015).

Forecasts are another form of hazard data. Multiple Numerical Weather Prediction (NWP) models are run using observational data to determine probabilities of future weather patterns based on running multiple simulations (Gneiting & Raftery, 2005). This probabilistic forecasting approach allows for the calculation of likelihoods of the weather phenomena to occur, and thus an overall confidence level of the forecast (Gneiting & Raftery, 2005). More observational data can increase the confidence of weather forecasts; however, this confidence decreases as forecast lead time increases (Hirschberg et al., 2011).

In places where hydrometeorological monitoring network coverage is limited or lacking, crowdsourcing and citizen science projects have been used to fill these gaps (e.g. Fava et al., 2014). Eyewitness accounts and storm spotter reports also contribute to an understanding of severe weather hazards and phenomena (Krennert, Pistotnik, et al., 2018).

7.3.2 Impacts

Impacts are the effects, outcomes, or consequences of hazardous events. During a severe weather event, Kox, Lüder, et al. (2018) identified impact information sources to include media coverage, fire stations upstream of a storm track that have already been affected, emergency calls, scout reports and ground-truthing, and emergency vehicle occupancy, to determine if enough capacities are on hand. Social media (Harrison et al., 2015), crowdsourcing (Sadler et al., 2018), and volunteered geographic information (Harrison et al., 2020) are other near real-time sources of impact information. Near real-time impact data is usually collected by local emergency management (EM) agencies to produce situational reports which contain all available information about the developments and impacts of an event (Kox, Lüder, et al., 2018).

Data Type	Description	Data Sources	Datasets
Hazard	Hydrometeorological observations and measurements to formulate warnings.	Meteorological services, hydrological services, Law enforcement, Storm spotters, Responders, social media (Twitter, Facebook, Seina Webo, Flickr, Instagram, SnapChat), the public, online databases, crowdsourcing and citizen science projects.	Meteorological and hydrological observations (e.g., rain gauges, river gauges), satellite and radar imagery, aerial imagery, Natural Hazards Assessment Network (NATHAN), European Severe Weather Database, US Storm Database, public surveys, historic warning records, eyewitness interviews, public surveys, photos and videos.
Impact	Building situational awareness and informs situational assessments for response planning. After an event, impact/damage assessments are conducted to link radar signatures of upcoming or unfolding hazards with a historical database and formulate impact-based warnings based on the historic impacts of similar events. Feeds into vulnerability models and fragility functions for impact modelling.	Media reports, law enforcement, fire stations upstream already affected, the public, emergency responders, engineers, social media (Twitter, Facebook, Seina Webo, Flickr, Instagram, SnapChat), crowdsourcing applications, Google, hydrological services and flood managers, public health agencies, insurance industry, online databases, flood databases, research institutions, environmental protection agencies, storm spotters, emergency services, personal contacts, risk modelling.	Ground-truthing, in-situ observations, reports and emergency calls, situational reports, eye- witness interviews, photos/videos, commentary, incident reports, situational reports, hotline records, loss claims, Emergency Events Database (EM-DA), Natural Hazards Assessment Network (NATHAN), flood databases, digital mapping (e.g. Humanitarian OpenStreetMap, MissingMaps), Google Alerts, Google Analytics Records, technical reports, European Severe Weather Database, public surveys, eyewitness interviews, risk modelling outputs.
Exposure	Highly specific and small-scale nature, usually at the individual, activity, or community level, (e.g., Ferry routes and the locations of large trees overhanging power lines).	Government departments, land and resource management agencies, flood managers, mapping agencies/organisations, crowdsourcing, OpenStreetMap, Google Maps, infrastructure industry, risk model.	Census data (for population counts and density), transportation routes and schedules, infrastructure databases/datasets, land-use spatial layers, building/asset footprints, road network spatial layers, Digital Elevation Models, time-varying population data, river network spatial data, risk modelling outputs.
Vulnerability	Conveys the vulnerability of people, livelihood, and property and typically includes information about infrastructure, buildings, land-use, census data, ecological data, and economic data; feeds into vulnerability models and fragility functions for impact modelling.	Research Institutions, public health agencies, engineers, risk specialists, insurance industry.	Vulnerability assessments, Pacific Risk Information System (PacRIS), vulnerability functions/fragility curves, vulnerability indices, asset characteristics (e.g., building structure information), public surveys.

Table 7.1. Summary of data sources and datasets for hazard, impact, vulnerability, and exposure data identified from the literature review.

Post-event damage information is collected in the aftermath of an event in the form of damage surveys. Post-event data such as damage surveys allows forecasters to correlate damage levels produced from certain hazards, such as tornadoes, with the storm radar signature (Harrison et al., 2015). If this impact information is stored in a historic database, forecasters can refer to the database to compare past storm radar signatures with current radar signatures and understand the level of damage and impacts the current storm may produce, allowing the forecaster to formulate impact-based warnings (Harrison et al., 2015).

Impact data is also produced from risk/impact models for pre-impact, rapidimpact, and post-impact assessments. Data sources can include hazard, exposure and vulnerability information, geo-located social media data (Smith et al., 2017), satellite data (Cervone et al., 2017), crowdsourced impact reports (Sadler et al., 2018), and normalised damage functions (Crompton & McAneney, 2008).

Challenges still exist with collecting enough systematic, in-situ data (i.e., observed impacts) to build empirical models (Panteli et al., 2017) and to validate models (Hemingway & Robbins, 2019). Furthermore, existing impact databases have been created with varying responsibility for which agency collects information, and the methods and purposes of collection (Robbins & Titley, 2018). This limits the databases' use, particularly for analysis and verification. Classifying impacts appears to be subjectively done (Doocy et al., 2013), proving it difficult to use datasets outside of the original purpose for which they were created.

7.3.3 Vulnerability

Vulnerability information is usually created through conducting vulnerability assessments (Sai et al., 2018). This involves combining information about infrastructure, buildings, land-use, census data, ecological data (Poolman, 2014), and socio-economic data (Terti et al., 2015). These vulnerability assessments are typically presented as spatial maps (McCallum et al., 2016).

Vulnerability assessments have traditionally focused on physical vulnerability, such as buildings and infrastructure, with less focus on social vulnerability (McCallum et al., 2016). Studies that have looked at social vulnerability tended to focus on physical impacts such as loss of life or physical injuries (Enarson, 2007), and less so on nuanced social impacts resulting from less quantifiable social vulnerability factors such as "poor biophysical, social, and/or financial capital" in communities (Terti et al., 2015, p. 1482). This is because these vulnerability factors are easier to quantify (McCallum et al., 2016). To address these gaps, Terti et al. (2015) identified a range of underlying social vulnerability factors resulting from human behaviour and demonstrated the highly dynamic nature of social vulnerability, such as risk governance, land use, and individuals' status, all of which change over space and time.

In New Zealand, Mason et al. (2021) developed social vulnerability indicators for flooding using national Census population data. These indicators are based on social vulnerability dimensions such as exposure, age, health and disability status, financial security, social connectedness, knowledge of natural hazards, housing conditions, food and water security, and decision-making and participation (Mason et al., 2021). Health indicators relating to certain health conditions such as cardiovascular disease, respiratory disease, and mental health issues were not completed due to time constraints (Mason et al., 2021). The results of their study provide an opportunity for decision-makers to consider additional factors beyond economic impacts when planning mitigative actions towards flooding (Mason et al., 2021).

Vulnerability information may be obtained through partnerships with the insurance industry who conducts vulnerability assessments for insurance schemes (Rogers et al., 2017). Additionally, the Pacific Catastrophe Risk Assessment and Financing Initiative (PCRAFI) created the Pacific Risk Information System (PacRIS), which houses historical hazard data, and risk profiles for Pacific Island countries (Rogers et al., 2017).

7.3.4 Exposure

The highly specific and small-scale nature of exposure makes it difficult to systematically collect and incorporate into an IFW system with the current tools and data available (GFDRR, 2016; WMO, 2015). An example of one form of general exposure information is population counts of people living in areas where hazards frequently occur (e.g. population data overlaid onto floodplains data) (Poolman, 2014). Exposure data is typically created by mapping the locations of assets, such as buildings and infrastructure in proximity to a hazard (McCallum et al., 2016).

Ferry routes and the locations of large trees overhanging power lines are other examples of exposure data that would be important to consider during a high wind event (WMO, 2015). Building footprints (i.e., a spatial layer of building polygons) is another example of exposure if the building footprints layer is overlaid with a hazard layer (e.g., Paulik et al., 2020). However, even national- or global-scale exposure datasets can be expensive (McCallum et al., 2016), and remain difficult to create and use due to lack of detailed asset information (Lin et al., 2020).

7.4 Data Characteristics for Early Warning Systems

Data usability in disaster response is determined by many factors. For the purposes of this study, we focused on two factors: reliability and timeliness. These appear to be two of the most important factors for choosing data sources for disaster response (Mansourian et al., 2006). The requirements for reliable and timely data for disaster response can also apply to EWSs, as warnings must be timely and accurate to incite early and appropriate action (WMO, 2018; Zhang et al., 2002).

Data is perceived as more reliable if it comes from a trusted source and/or it can be vetted (Harrison & Johnson, 2019). As such, there is a preference for official sources, such as emergency call centre reports, intel from responders themselves, etc. (Kox, Lüder, et al., 2018). For this study, official data refers to data created by recognised officials involved in local disaster response management practices where the disaster is occurring, such as police and fire services, engineers, helicopter pilots, local and regional councils, as well as meteorological and hydrological agencies, and science agencies. Unofficial data refers to data created by external parties of the disaster management practices, such as social media users, contributors to OpenStreetMap, private external corporations (e.g., Google Maps), and media agencies.

This section identified various sources of HIVE data. These components (hazard, impact, vulnerability, and exposure, or HIVE) form the conceptual basis of severe weather IFWs. Meteorologists do not typically possess knowledge of impacts, vulnerability, and exposure (Kaltenberger et al., 2020). As such, sources of these data need to be identified for IFWs (Harrison et al., 2022; Potter et al., 2021). Furthermore, understanding the characteristics of the available data sources in terms of reliability and timeliness will assist data users in determining appropriate datasets for their purposes.

The objectives of this research are to identify sources for HIVE data, and to understand the inhibitors and facilitators to collecting and using these data, to support the implementation of an IFW system in New Zealand for severe weather hazards. We chose to focus on the New Zealand context to support the country's efforts for fulfilling both the WMO's objectives and Sendai Framework priorities for improved documenting of disaster risk and loss data; a need identified in previous research (Crawford et al., 2018; Potter et al., 2021). We employed Grounded Theory (Strauss & Corbin, 1990) to meet the research objectives, described next.

7.5 Research Methods

We used a qualitative approach to address the research question, specifically employing the Evolved-Straussian Grounded Theory research strategy (ES-GT) for data collection and analysis. Interviews and workshops were the primary data collection methods. From November 2018 to April 2021, the lead author interviewed thirty-nine (n=39) experts in weather forecasting, warning, response, risk modelling, and data collection and management, as shown in Table 7.2. Three virtual workshops were held in New Zealand (NZ). Two of these workshops involved EM practitioners, weather forecasters, communication and data specialists, and hydrologists, from Auckland Region (n=4) and Southland Region (n=5). The third workshop involved a portion of the NZ risk and hazard science community based at GNS Science (n=11). Thus, in total 59 people participated in this research.

Table 7.2. Participant Codes. All participants remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by the area of expertise and/or practice, industry, location, or governance level¹⁴.

Interview Code	Position	Classification	Location	Government Level
Agriculture/Rural NZ. A	Agriculture policy coordinator	Agriculture/Rural	NZ	National
Data Management Gov. NZ. Nat. A	Senior Resilience Advisor	Data Management	NZ	National
Data Management Private NZ. B	Geospatial Specialist	Data Management	NZ	
Data Management Research NZ. C	GIS Specialist	Data Management	NZ	
Data Management Private NZ. D	GIS Specialist	Data Management	NZ	
Data Management Gov. NZ. Nat. E	Head of Data	Data Management; Governance	NZ	National
EM. NZ. Reg. A	Director	Emergency Management	NZ	Regional
EM. NZ. Reg. B	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. C	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. D	Principal Science Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. E	Principal Advisor Strategy and Partnerships	Emergency Management	NZ	Regional
EM. NZ. Reg. F	GIS Lead	Emergency Management; Data Management	NZ	Regional
EM. Gov. NZ. Nat. G	Senior Hazard Risk Management Advisor	Emergency Management; Governance	NZ	National
EM. NZ. Reg. H	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Nat. I	First Responder	Emergency Management	NZ	National
EM. Gov. NZ. Nat. J	National Operations Manager	Emergency Management; Governance	NZ	National

¹⁴ The acronyms and abbreviations in Table 7.2 are as follow: Meteorological (Met.), Emergency Management (EM), New Zealand (NZ), Geographic Information Systems (GIS), Regional (Reg), Government (Gov.), Early Warning System (EWS), International (Int.).

Interview Code	Position	Classification	Location	Government
FM NZ Reg K	Regional Manager	Emergency Management	N7	Regional
EM. NZ. Reg. L	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. M	Group Controller	Emergency Management	NZ	Regional
Health NZ. Reg. A	Respiratory Doctor	Public Health	NZ	Regional
Hyd. Gov. NZ. Reg. A	Flood EWS Programme manager	Hydrology; Governance	NZ	Regional
Lifelines NZ. Reg. A	Civil Engineer	Lifelines	NZ	Regional
Loss Modelling Research NZ. A	Economist	Loss Modelling; Research	NZ	
Met. Int. A	Science Manager	Meteorology	International	National
Met. Int. B	National Manager Disaster Mitigation	Meteorology	International	National
	Policy			
Met. Int. C	Senior Policy Officer	Meteorology	International	National
Met. Int. D	Senior Social Scientist	Meteorology	International	National
Met. Int. E	Consultant Meteorologist	Meteorology	International	National
Met. NZ. F	Senior Meteorologist	Meteorology	NZ	National
Met. NZ. G	Communications	Meteorology	NZ	National
Met. NZ. H	Public Relations	Meteorology	NZ	National
Met. Int. I	Division Chief/Meteorologist	Meteorology	International	National
Met. Research NZ. J	Meteorologist	Meteorology; Research	NZ	National
Met. NZ. K	Senior Meteorologist	Meteorology	NZ	National
Met. Private NZ. L	Head Weather Analyst	Meteorology	NZ	
Risk Modelling NZ. A	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. B	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. C	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. D	Risk Modeller	Risk Modelling	NZ	National

Interview questions and workshop activities focused on IFW data needs and sources. We asked for participants' general thoughts on IFWs; what impact, vulnerability, and/or exposure data they use or need, why, and how; the life path of the data; experienced and/or perceived challenges obtaining data required for IFWs and other uses; and thoughts on collecting and using alternative data sources (e.g. social media and crowdsourcing). Herein we report on data sources, collection, and creation.

This research was conducted under a 'low risk' ethics notification with the Massey University Human Ethics Committee prior to data collection in 2018. All interviewees remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by area of expertise and/or practice, industry, location, or governance level (Table 7.2). Interviews were audio-recorded and transcribed verbatim.

Following ES-GT, we analysed the interview and workshop data using open coding and axial coding, supported by memo-writing and diagramming (Corbin & Strauss, 2015), in Nvivo 12 (Bergin, 2011). The coding paradigm introduced by Strauss and Corbin (1990) supported the axial coding stage whereby the codes created from open coding were related back to the coding paradigm dimensions (Table 7.3) for increased density and precision.

Coding Paradigm Dimension	Description
Causal Conditions	A set of events that influence the phenomena or result in the appearance or development of a phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Phenomena	The subject or object under study (Strauss & Corbin, 1990).
Contextual Conditions	The specific set of conditions and characteristics surrounding the phenomena and resulting in action/interaction strategies taken to address the phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Intervening Conditions	Unexpected events or factors leading to action/interaction strategies (e.g., time, space, culture, socioeconomic status, technological status, history) (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Action/Interaction Strategies	Purposeful and deliberate acts taken to address the phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Consequences	Predictable or unpredictable, intended or unintended outcomes of the action/interaction strategies (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).

Table 7.3. Summary of the coding paradigm dimensions that supported the axial coding analysis in this study.

7.6 Findings and Discussion

This section will first present data sources for HIVE data as identified by the participants, followed by an investigation into the causal conditions, intervening conditions, and subsequent action/interaction strategies for collecting the HIVE data.

7.6.1 Data Sources

The interviews and workshops identified several sources for hazard, impact, vulnerability, and exposure data. These data sources are described next. The following sections are organised by data type (hazard, impact, vulnerability, and exposure). Herein, each section will provide an overview, with accompanying summary tables, of the data sources and their characteristics related to IFWs. The accompanying tables present the data or dataset (e.g., weather stations, radar data), whether the data is official or unofficial, who collects or creates the data (e.g., the creator may be a member of the public by posting a report on social media, and the collector may be an EM agency who collects social media posts for situational awareness), the timescale of the data (e.g., "[Near] real-time" is data collected in realtime or near real-time, such as observational data, social media reports, etc.; "Current" is data that is static in time, was created prior to the event, but has been maintained and kept up-to-date; "Forecasted" is data created from forecasting models; and "Historic" is data created after an event and is not kept up-to-date), the type of hazard and/or impact (e.g., meteorological, hydrological, hydrogeological, social, infrastructural, urban, rural, environmental, health, property, built environment), and uses within and outside of the Impact Forecasting and Warning Value Chain (Chapter Six; Harrison et al., 2022).

7.6.1.1 Hazard Data

Hazard data refers to meteorological, hydrological, and hydrogeological data. Table 7.4 provides a summary of the hazard data sources identified by our participants. With regards to uses in the Warning Value Chain, each hazard data source listed in Table 7.4 was found to be used for Observation, Monitoring, and Detecting; Hazard Forecasting; Impact Forecasting; Impact Warning.

Our findings indicate that hazard data is quite systematically collected, documented, and used in New Zealand for many purposes by a select group of official agencies (see Table 7.4). For example, meteorological data, such as rainfall, windspeed, radar data, and forecast data are mostly produced, used, and housed by the Meteorological Service of New Zealand (herein referred to as the NZ MetService) and the National Institute of Water and Atmospheric Research (NIWA). The NZ MetService is NZ's appointed National Meteorological Service (NMS) (Williamson, 1998), and NIWA is NZ's Crown Research Institute (CRI) ¹⁵ for atmospheric and oceanic science (Steiner et al., 1997). NIWA maintains the national climate database with all the rainfall records and hydrological databases. Fire and Emergency New Zealand

¹⁵ Crown Research Institutes (CRIs) are government owned companies that conduct scientific research in New Zealand (MBIE, 2021).

(FENZ) was also found to collect observational meteorological data for their own operations, which helps them plan responses to wildfires and other weather-related emergencies (EM. NZ. Nat. I).

The NZ MetService has made some of their observation and satellite data freely available (MetService, 2021a; Met. Private NZ. L) for commercial use by private weather forecasting companies, and for stakeholder agencies and researchers to conduct their own analyses, such as risk modelling. Following an impactful event on the Wellington South Coast where large swells and waves damaged houses and evacuations of five properties, arrangements have been made for the NZ MetService and their oceanographic branch (the MetOcean), and NIWA, to provide swell data, wave data, and wave forecast data to the regional EM office for a Significant Wave Warning programme (EM. NZ. Reg. L; WREMO, 2021).

Findings from our interviews indicate that hydrological data is collected, created, and used primarily by local and regional councils in New Zealand. This aligns with the mandated role of local and regional councils under the Resource Management Act 1991, the National Civil Defence Emergency Management Plan 2015, and the Local Government Act 2012, wherein local and regional councils are assigned the responsibility of managing, monitoring, forecasting, and warning for flood hazards, with support from NIWA, the MetService, and EM Groups (Rouse, 2011). Hydrological data includes river height and flow gauges, river camera feeds, river network and watershed data (e.g., spatial shapefiles of rivers, floodplain footprints), overland flow paths, river flow and flood forecast models, etc.

The MetService also forecasts pollen counts for those with allergies. Pollen data is desirable for the public health sector (Health NZ. Reg. A), because pollen was found to play a significant factor in thunderstorm-related asthma attacks (D'Amato et al., 2016). While the MetService provides qualitative pollen forecasts (e.g., "Pollen Levels: Moderate, Type is Plantain"), a public health official from the Waikato Region in NZ expressed a need for quantitative pollen data to model the risk of asthmatic attacks due to pollen and spring thunderstorms (Health NZ Reg. A; Harrison et al., 2022). This need was echoed by a NZ-based pollen scientist for climate change risk assessments (see Sharpe, 2020).

Our interviews identified other creators of relevant hazard data alongside the MetService, NIWA, local/regional councils, and FENZ. Land Information New Zealand (LINZ), the public service department charged with handling geographic information in New Zealand, collects sea level data from float gauges (as do port companies), which is useful for coastal flood hazards (Data Management Gov. NZ. Nat. A); and holds slope data, useful for landslide hazard management and mitigation (Data Management Gov. NZ. Nat. A). LINZ and the Ministry for Environment (MfE) also hold river network data.

Table 7.4. Summary of hazard data sources in New Zealand, identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, type of hazard, and non-IFW uses¹⁶.

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	Type of Hazard	Non-IFW Uses
Weather stations (rain gauges, anemometers,	Official and	MetService, FENZ, NIWA, councils, volunteers	[Near] real- time	Meteorological	Research
etc.)	unofficial				
Radar data	Official	MetService, NIWA, Private industry	[Near] real- time	Meteorological	Research
Satellite imagery & observations	Official	MetService, NIWA, LINZ	[Near] real- time	Meteorological	Research
River height and flow gauges	Official	Regional councils, NIWA	[Near] real- time	Hydrological (e.g. flood)	Research
Float gauges/Sea level data	Official	LINZ, port companies, regional councils, NIWA	[Near] real- time	Hydrological (e.g. flood)	Research
River networks	Official	NIWA, LINZ, Ministry for the Environment, Councils	[Near] real- time	Hydrological (e.g. flood)	Research
Vertical Rain Radar	Official	MetService, Private industry, Healthy Waters Auckland Council	[Near] real- time	Meteorological	Research
Regionwide floodplains footprints/shapefiles	Official	Councils	[Near] real- time	Hydrological (e.g. flood)	Research
Overland flow paths	Official	Councils	Current	Hydrological (e.g. flood)	Research
Coastal inundation maps	Official	Councils	Forecasted	Hydrological (e.g. flood)	Research

¹⁶ The acronyms and abbreviations in Table 7.4 are as follows: Fire and Emergency New Zealand (FENZ), National Institute of Water and Atmospheric Research Ltd (NIWA), Land Information New Zealand (LINZ), New Zealand Meteorological Service (MetService), New Zealand Transport Agency (NZTA), Emergency Management (EM), New Zealand Geographic Information Systems for Emergency Management (NZGIS4EM), MetOcean Solutions (MetOcean).

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	Type of Hazard	Non-IFW Uses
Pollen counts	Official	MetService	Forecasted	Health	Research; Response: Preparedness
Slope	Official	LINZ, Councils	Current	Hydrogeological (e.g. landslide)	Research
Camera feeds	Official	Councils, NZTA, Ski fields, Police	Real-time	Meteorological, Hydrological, Hydrogeological (e.g. Iandslide)	Response; Preparedness; Public Awareness
Social media (Tweets, Facebook post comments)	Unofficial	Creators: public / social media users Collectors: MetService, EM, Researchers, Councils, NZGIS4EM	[Near] real- time	Meteorological, Hydrological, Hydrogeological	Research; Response; Situational Awareness
Crowdsourcing (e.g., volunteer rain gauges, NZ Flood Pics, mPing, WeatheX, European Weather Observer)	Unofficial	Creators: public / social media users Collectors: EM, Researchers, Councils, NZGIS4EM, Volunteers (e.g., NZ Flood Pics)	[Near] real- time	Meteorological, Hydrological, Hydrogeological	Research; Response; Situational Awareness
Forecast data	Official	MetService, MetOcean, NIWA, Councils	Forecasted	Meteorological, Hydrological, Marine/Coastal	Research; Response: Preparedness
National Climate Database	Official	NIWA	Historical	Meteorological, Hydrological	Research

Unofficial data sources were also identified by participants for monitoring hydrometeorological hazards and building situational awareness. For example, the MetService, Civil Defence and Emergency Management (CDEM) Groups, and Local/Regional Councils monitor social media for reports or observations of hydrometeorological hazards (Met. Int. E; EM. NZ. Reg. B, C). From our interviews, we found that none of our participating agencies collect or store social media data for future analysis, citing resource limitations as a key barrier (Met. NZ. F, G, H; EM. NZ. Reg. A, E). Rather, the social media platforms are monitored onscreen for staff to pick up posts or events of interest for further investigation (Met. Int. E; EM. NZ. Reg. B, C). Additionally, one CDEM Group described how they actively request social media users to verify impacts if they can safely do so (EM. NZ. Reg. A). Notable posts may be captured and included in situational reports, but the data usually does not have a use beyond that (EM. NZ. Reg. A, C).

Crowdsourced data was found to be collected through specially designed applications and platforms. In NZ, the West Coast CDEM Group crowdsourced hazard and impact reports using the Esri Story Maps platform (EM. NZ. Reg. C, F). Similarly, the NZ Flood Pics Esri Story Map¹⁷ was set up by a volunteer to collect flood and impact photos during flood events in Auckland and has since been expanded to the rest of NZ (Data Management Private NZ. D). In the USA, the meteorological service collects precipitation data and associated impacts through the mobile application mPing¹⁸ (Met. Int. I). Similarly, in Austria and Australia, the European Weather Observer¹⁹ (Met. Int. E) and WeatheX²⁰ (Met. Int. B, C) applications respectively are used to collect severe weather reports. These applications gather hazard (and some impact) reports from the public, who have not received any training. The Story Map data usually contains photos with an optional description for the CDEM Group to build their situational awareness (EM. NZ. Reg. F). Alternatively, the specialised applications used in the USA, Austria, and Australia collect more structured data using reporting forms (Met. Int. B, C, E, I). The benefits of the crowdsourced data are that it is timely (Met. Int. E), it may fill in sensor gaps (Met. Int. I), and the applications may increase public awareness and engagement in severe weather hazards (Met. Int. B, C, I). However, maintaining engagement is a challenge and agencies may struggle with receiving reports once contributors have lost interest (Met. Int. E). A gamification element is being considered to maintain interest for contributors of the European Weather Observer (Met. Int. E). Other challenges include quality control and getting buy-in from scientists on the validity of crowdsourced data (Met. Int. E, I).

Meteorological services in the USA and Austria regularly collect reports from storm spotters. The storm spotters can receive training from the meteorological services to increase their credibility. In Austria, the meteorological services offer training and certificates with identification numbers for storm spotters to then register and receive a quality control rating for their reports based on their level of training

¹⁷ <u>http://www.nzfloodpics.co.nz/</u>

¹⁸ <u>https://mping.nssl.noaa.gov/</u>

¹⁹ <u>https://www.essl.org/cms/european-severe-weather-database/ewob/</u>

²⁰ <u>https://weathex.app/</u>

(Met. Int. E). The meteorological service can then discern the trustworthiness of incoming spotter reports (Met. Int. E). This data is considered highly credible among scientists (Met. Int. E). Furthermore, engagement and awareness are quite high with the "weather enthusiasts" (Met. Int. E). However, connecting with spotters for training remains a challenge (Met. Int. E). The data is also limited in its application for warnings because it is not real-time, thus it cannot be used for real-time warning verification (Met. Int. E). Lastly, storm spotter reports may overlook meteorological phenomena that produce small impacts, a need identified by the US National Weather Service (Met. Int. I).

As reported in a related study (Harrison et al., 2022), much of these datasets are used for steps in the Value Chain for warnings, particularly hydrometeorological Observation, Monitoring, and Detection, Hazard Forecasting, and in some cases Impact Forecasting and Impact Warning. For example, the MetService uses radar and rainfall data to observe, monitor, and detect rainfall amounts, which is then used by regional and local councils to forecast flood hazards (Met. NZ. G). Flood forecasts may then be used to forecast the impacts of the flood, in a quantitative model-based approach, or in a qualitative discussion-based approach, or both (Harrison et al., 2022). Local/regional councils and CDEM Groups may then use these forecasts to inform their flood warning messages, wherein they might include impact-oriented messages (EM. NZ. Reg. H; Kaltenberger et al., 2020).

7.6.1.2 Impact Data

A plethora of impact data were identified in our interviews that are created, collected, and used by different groups for different purposes, as shown in Table 7.5. Impact data is available in a range of official and unofficial capacities.

Impact data is collected, created, and used in an official capacity by CDEM Groups, local/regional councils, FENZ, the lifelines sector (e.g., transportation, power, water/wastewater/stormwater services), the insurance sector (e.g., the Earthquake Commission (EQC), the Accident Compensation Corporation (ACC), the Insurance Council of NZ), and the research sector (e.g., GNS Science, NIWA). The primary purpose of collecting the impact data is not for severe weather forecasts and warnings, but for response, recovery, mitigation, and planning. Depending on the quality of the data, it could be used for additional purposes such as impact forecasting and warning. For example, emergency call centre (i.e., 111 calls in NZ) reports incite responses from the appropriate authorities (e.g., FENZ, police). Afterwards, weather-related emergency reports can be used for post-event analysis to verify and improve warnings (Met. NZ. G; Harrison & Johnson, 2016).

Insurance claims from property damage and injuries are used for recovery and produce rich datasets for future analysis such as risk/loss modelling (Loss Modelling Research NZ. A), which may inform impact forecasts. Likewise, impact assessments conducted by CDEM Groups are primarily used for building situational awareness to inform response efforts (EM. NZ. Reg. H; NEMA, 2020a) and can also be used for warning verification and warning updates in real-time (EM. NZ. Reg. H).

Table 7.5. Summary of impact data sources identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, uses in the IFW Value Chain, type of impact, and non-IFW uses²¹.

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
Tweets	Unofficial	Public / social media users	MetService, CDEM, Researchers, Councils, NZGIS4EM	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Facebook Post comments	Unofficial	Public / social media users	MetService, CDEM, Researchers, Councils, NZGIS4EM	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
SnapChat Heat Maps	Unofficial	Public / social media users	MetService, CDEM, Researchers, NZGIS4EM	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Crowdsourced Photos via Story Maps (e.g., NZ Flood Pics)	Unofficial	Public / social media users	MetService, CDEM, Researchers, Councils, NZGIS4EM, Volunteers (e.g., NZ Flood Pics)	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness

²¹ The acronyms and abbreviations in Table 7.5 are as follows: Fire and Emergency New Zealand (FENZ), National Institute of Water and Atmospheric Research Ltd (NIWA), New Zealand Meteorological Service (MetService), New Zealand Transport Agency (NZTA), Emergency Management (EM), New Zealand Geographic Information Systems for Emergency Management (NZGIS4EM), Insurance Council of New Zealand (ICNZ), Accident Compensation Corporation (ACC), Institute of Geological and Nuclear Sciences Limited (GNS Science), Statistics New Zealand (Stats NZ), Earthquake Commission (EQC), National Emergency Management Agency (NEMA) Real-time Individual Asset Attribute Collection Tool (RiACT).

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
Crowdsourcing / Public Reporting	Unofficial	Public	Councils, CDEM, Volunteers	Current, maybe near real-time	Observation, Monitoring, and Detecting; Hazard Forecasting; Impact Forecasting; Impact Warning	Built environment, infrastructure	Response; Situational Awareness; Recovery; Planning; Mitigation; Public Awareness / Engagement / Education
Red Cross Chained Crowdsourcing	Unofficial	Public	Red Cross	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Response; Situational Awareness
Emergency call centre reports	Official	Public	Police, FENZ	[Near] real- time & Historic?	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Community volunteer radio calls	Unofficial	Designated community volunteers	EM, Councils	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Response; Situational Awareness
Damage surveys (e.g. aerial surveys, building surveys via Survey123, QuickCapture, RiACT, Kobo 2, etc.)	Official	Councils, Researchers (NIWA, GNS Science, Universities), CDEM	Councils, Researchers (NIWA, GNS Science, Universities), CDEM	Historic	Impact Forecasting; Impact Warning	Urban, Rural, Infrastructural	Research; Response; Situational Awareness; Recovery; Mitigation; Land use planning and development policies

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
Media reports	Unofficial	Media outlets	Councils, Researchers (NIWA, GNS Science), CDEM, MetService	[Near] real- time	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Tacit knowledge, experience, intuition	Official	Councils, CDEM, MetService	Councils, CDEM, MetService	Available in real-time, based on historic knowledge and experience	Impact Forecasting; Impact Warning	Urban, Rural, Environmental	Response; Situational Awareness
Health & Injury Data	Official	District Health Boards, ACC, Stats NZ	District Health Boards, ACC, Stats NZ	Historic	Impact Forecasting; Impact Warning	Health	Research; Response; Recovery; Planning and Mitigation
Wellbeing Surveys	Official	CDEM, Councils	EM, Councils	Historic	Impact Forecasting; Impact Warning	Social, Health, Property	Response; Situational Awareness; Recovery
Post-event interviews and surveys	Official	Researchers (e.g. NIWA, GNS Science, other), EM	Researchers (e.g. NIWA, GNS Science, other), EM	Historic	Impact Forecasting; Impact Warning	Social	Response; Mitigation; Preparation
Insurance claims	Official	Insurance companies, EQC, ICNZ	Insurance companies, EQC, ICNZ	Historic	Impact Forecasting; Impact Warning	Property, Health, Economic	Research; Recovery; Mitigation
"Boots on the ground"	Official	Councils, CDEM	Councils, CDEM	[Near] real- time	Impact Forecasting; Impact Warning	Urban, Rural, Environmental	Response; Situational Awareness

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
Lifelines Sectors (e.g. power companies, NZTA)	Official	Lifelines Services (e.g., NZTA, Transpower, KiwiRail), councils, wastewater services	Lifelines Services (e.g., NZTA, Transpower, KiwiRail), councils, wastewater services	[Near] real- time	Impact Forecasting; Impact Warning	Infrastructural	Response; Situational Awareness; Recovery; Business As Usual; Research
Council Requests For Service (RFS)	Official	Public	Councils	[Near] real- time	Impact Forecasting; Impact Warning	Infrastructural; Property; Urban; Rural; Environmental	Situational Awareness; Response; Recovery
Situational Reports	Official	CDEM, Councils	CDEM, Councils	Current	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Built Environment	Response; Situational Awareness; Recovery; Research
Post-event reports	Official	NEMA, CDEM Groups, Councils	NEMA, CDEM Groups, Councils	Historic	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Built Environment	Recovery; Mitigation and Planning; Research
Operations Reports (Council)	Official	Councils	Councils	Historic	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Response; Situational Awareness; Recovery; Research
Flood event reporting	Official	Councils, Healthy Waters Auckland Council/Councils	Councils, Healthy Waters Auckland Council/Councils	Historic	Observation, Monitoring, and Detecting; Impact Warning	Infrastructural, Urban, Rural Environmental	Response; Situational Awareness; Research; Mitigation; Land use planning and development policies
Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
--	---------------------------	--	---	--------------------	---	---	---
Injuries and fatalities (e.g., cause of death)	Official	ACC, Coronial Services of New Zealand, Stats NZ	ACC, Coronial Services of New Zealand, Stats NZ	Historic	Impact Forecasting; Impact Warning	Health	Response; Situational Awareness; Research; Mitigation
Cultural and heritage / historical impacts	Official	Councils	Councils	Historic	Impact Forecasting; Impact Warning	Social, Cultural	Research; Planning; Mitigation
Impact Model outputs	Official	GNS Science, NIWA, Researchers	GNS Science, NIWA, Researchers	Historic, Rapid	Hazard Forecasting; Impact Forecasting	Social, infrastructure, built environment, hydrological, hydrogeological, urban, rural, Property, Economic	Response; Research; Planning; Mitigation
NEMA National Loss Database	Official	CDEM Groups	NEMA	Historic	Impact Forecasting; Impact Warning	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	Sendai Framework Reporting
NIWA NZ Historic Events Catalogue	Unofficial	Media outlets	NIWA	Historic	Impact Forecasting; Impact Warning	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	Research; Mitigation
Local Council Databases	Official	Councils	Councils	Historic	Impact Forecasting; Impact Warning	Hydrological, Hydrogeological, Property, Urban, Rural, Infrastructural, Built Environment	Response; Situational Awareness; Recovery; Mitigation

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
Storm Data (USA)	Official	Meteorological services, Responders, Emergency Services, Storm Spotters, media outlets	National Centers for Environmental Information	Historic	Impact Forecasting; Impact Warning	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	Research; Mitigation
European Severe Weather Database	Official	Meteorological services, Responders, Emergency Services, Storm Spotters, media outlets	European Severe Storms Laboratory	Historic	Impact Forecasting; Impact Warning	Built Environment ct Meteorological, Research; Hydrological, Mitigation Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	

Impact assessments for CDEM Groups in NZ have evolved with the advancements of GIS-based technology for improved data capture and management (EM. NZ. Reg. C, H, F; Data Management Private NZ. B). An overview of how GIS tools were used for a flood event in NZ's West Coast region was provided by Stowell (2020). Officials used rapid data collection tools such as Survey123²² and QuickCapture²³ to collect impact information. QuickCapture was primarily used for collecting photos from aerial and ground assessments, while Survey123 was used to complete form-based assessments (Stowell, 2020). The value of these tools lies in the seamless integration process of the field data directly into a GIS layer for real-time viewing in an Emergency Operations Centre (EOC) (EM. NZ. Reg. C, H, F; Data Management Private NZ. B). However, the pitfalls of these collection methods are that information overload can occur with applications like QuickCapture if the trained staff take a large volume of photographs, and gaps in training can lead to errors in the data, which may skew datasets (EM. NZ. Reg. F).

Tacit knowledge and experience refer to knowledge held by official staff from past experiences who apply this knowledge when planning for or responding to events. NZ participants described how senior EM staff know from past events that certain river levels and peaks will lead to respective levels or types of impacts (EM. NZ. Reg. A, B). This knowledge is passed on verbally either just before or during an event to help with response planning and coordination (EM. NZ. Reg. A, B). The same was found to be true for the Lifelines sector, who "have a good amount of data but ... more importantly ..., their staff have a very good historical knowledge" (Lifelines NZ. Reg. A). This information is highly trusted as it is based on years of experience. It allows agencies to learn from past events and better prepare for current or future events (EM. NZ. Reg. B).

In many cases this tacit knowledge and experience is not formally documented and there is a risk of losing this knowledge when staff move on (EM. NZ. Reg. A, C; Lifelines NZ. Reg. A). As one participant identified: "[hazard and impact forecasting are] primarily based on history and knowledge. And with that comes the risk ... if you have key people out of the equation or people move on ... in life, then you have the knowledge gap until such time as that's filled" (EM. NZ. Reg. C). Upon identifying this vulnerability, this participant indicated that their agency has begun some historical cataloguing (EM. NZ. Reg. C).

²² Survey123 is a location-based application developed by Esri (Esri, n.d.-b) that is used for completing assessment forms such as impact assessment forms, building assessment forms, and welfare needs assessment forms (EM. NZ. Reg. F).

²³ QuickCapture is another location-based field observation application developed by Esri (Esri, n.d.-a) that is particularly useful for taking photographs of damage/impacts and uploading in real-time to be viewed on a dashboard in the Emergency Operations Centre (EM. NZ. Reg. F).

Unofficial impact data/information also helps EMs "build a picture of what's happening" (EM. NZ. Reg. C). Unofficial data includes social media posts, crowdsourced and citizen science data, media reports, and community volunteer radio calls. Interviewees were found to obtain information from Facebook, Twitter, and SnapChat. In most cases, the interviewees indicated that they do not collect or store social media data, but only monitor it for situational awareness and real-time verification. The value of social media appears to lie in its ability to indicate potential hot spots early on in an event and building situational awareness (Met. Int. E; EM. NZ. Reg. C). Social media sentiment analysis can also help to understand the cultural impacts of severe weather and communicate the warnings better (Met. Int. D). However, interviewees use social media data with caution, indicating that while social media data is timely and quantitatively rich, it is difficult to verify (EM. NZ. Reg. B). Relying solely on social media risks missing or overlooking impacted people who are not on social media, such as areas without power or internet access (EM. NZ. Reg. B). Thus, social media data should be used complementarily with other data (Met. Int. D).

Impact databases were also found to exist or be under development. In NZ, the National Emergency Management Agency (NEMA) is developing a national loss database in fulfilment of the Sendai Framework priorities (UNDRR, 2015b). Data is pulled from EM impact assessments, situational reports, and post event reports (EM. Gov. NZ. Nat. G). NIWA hosts a catalogue of historic severe weather events in NZ using media reports (Met. Research NZ. J), primarily for research purposes. Storm Data in the USA and the European Severe Weather Database contain data sourced from storm spotters and impact assessments from storms in the USA and Europe (Met. Int. E, I). These American and European databases are highly credible as only vetted and trusted information goes in (Met. Int. E, I). Unfortunately, the databases are not updated in real-time due to the rigorous quality control measures and may have gaps for small impact events (Met. Int. E, I).

Most of the data sources described above capture direct and physical impacts such as damages to the built and natural environment and do not capture social human impacts or indirect impacts. This is a major gap identified in a previous study (Harrison et al., 2022). Wellbeing surveys and post-event surveys from research studies capture indirect impacts to people's health and social/cultural wellbeing (EM. NZ. Reg. F, H; Risk Modelling NZ. B, C). Wellbeing surveys (also referred to as Needs Assessments and Welfare surveys) are conducted by CDEM Groups and/or health agencies to understand the needs and impacts of people affected by an emergency (MCDEM, 2015; Nielsen, 2017). Furthermore, District Health Boards, ACC, and Stats NZ collect and house health, injury, and mortality data for their own purposes (EM. Gov. NZ. Nat. G; Loss Modelling Research NZ. A). In the context of severe weather IFWs, these data, along with the Wellbeing surveys conducted by EM/councils, could be useful for understanding indirect and social and cultural impacts to inform impact warning messages (EM. Gov. NZ Nat. G; Loss Modelling Research NZ. A; Harrison et al., 2022).

7.6.1.3 Vulnerability Data

Our participants identified some vulnerability data sources, as shown in Table 7.6. As stated in our literature review, vulnerability data is more difficult to obtain or create, as vulnerability changes over space and time. On their own, many of the identified data sources in Table 7.6 do not provide an indication of vulnerability. Rather, these data sources are inputs into vulnerability assessments and risk models. For example, census data and health data may be used to conduct a social vulnerability assessment, while building damage assessments are inputs into vulnerability functions for risk modelling (Harrison et al., 2022; Tarbotton et al., 2015). The results of the vulnerability assessment or risk model then help analysts identify vulnerable areas:

Everything's built off the models ... we've identified ... issues with care homes and areas, hotspots in the community which could potentially ... be at risk ...We've got them in ... behind the stop banks, bungalows, probably not the best place for them but they're there. So just being aware of those [vulnerable areas] (Hyd. Gov. NZ Reg. A).

Asset information refers to characteristics of the asset for which the vulnerability is being assessed. These can be people, buildings, roads, stop banks, the environment, etc. The asset information of buildings would include data on the age of the building, the construction material, and number of floors (Risk Modelling NZ. B). This information is collected through building assessments by building engineers (Risk Modelling NZ. B). Alternatively, asset information of people consists of demographic data, health data, socioeconomic data, etc. (Risk Modelling NZ; Terti et al., 2015), available from Stats NZ and the Ministry of Health. Asset information about stop banks includes age and condition and is held by local and regional councils (Hyd. Gov. NZ. Reg. A).

An additional challenge with using census data identified by our participants is the spatial scale. The NZ census data is at the meshblock²⁴ scale, but risk modellers want to model at the building scale. Participating risk modellers questioned how they can interpolate the census data with the building level (Risk Modelling NZ. A, B).

Health data is another indicator of human vulnerability. The impacts of severe weather events can exacerbate underlying health conditions. For example, people with asthma may experience exacerbated symptoms while cleaning up damaged and contaminated sites after a flood (Risk Modelling NZ. C), or during thunderstorms (Health NZ. Reg. A; Harrison et al., 2022). In NZ, District Health Boards (Health NZ. Reg. A) and the Ministry of Health house this data. The Ministry of Health conducts an annual national Health Survey of NZ.

²⁴ According to Stats NZ (2021), "[a] meshblock is defined by a geographic area, which can vary in size from part of a city block to a large area of rural land."

Table 7.6. Summary of vulnerability data sources identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, uses in the IFW Value Chain, type of impact, and non-IFW uses²⁵.

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	IFW Uses (from the Value Chain)	Type of Impact	Non-IFW Uses
Vulnerability Assessment and Risk Modelling Outputs	Official	Councils, Researchers (NIWA, GNS Science), CDEM, Hired contractors	Historic	Impact Forecasting; Impact Warning	Property, human, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Planning; Mitigation; Research
Asset Information	Official	Councils, Researchers (NIWA, GNS Science), CDEM, Hired contractors	Historic	Hazard Forecasting; Impact Forecasting; Impact Warning	Property, human, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Planning; Mitigation; Research
Building damage assessments	Official	Councils, Researchers (NIWA, GNS Science), CDEM, Hired contractors	Historic	Hazard Forecasting; Impact Forecasting; Impact Warning	Property	Response; Situational Awareness; Recovery; Research; Mitigation
Census data	Official	Stats NZ	Historic	Hazard Forecasting; Impact Forecasting	Human	Research; Business As Usual (BAU)

²⁵ The acronyms and abbreviations in Table 7.6 are as follows: National Institute of Water and Atmospheric Research Ltd (NIWA), Emergency Management (EM), Institute of Geological and Nuclear Sciences Limited (GNS Science), Statistics New Zealand (Stats NZ), Ministry of Health (MOH), District Health Boards (DHBs).

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	IFW Uses (from the Value Chain)	Type of Impact	Non-IFW Uses
Tacit knowledge, experience, intuition	Official	Councils, CDEM	Available in real- time, based on historic knowledge and experience	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Response; Situational Awareness; Preparedness; Mitigation; BAU
Lab-based experiments	Official	Researchers (NIWA, GNS Science, universities)	Historic	Hazard Forecasting; Impact Forecasting	Built environment	Research
Health Data, New Zealand Health Survey	Official	MOH, DHBs	Historic	Impact Forecasting. Impact Warning	Health	Research; Recovery
Infrastructure	Official	Councils, contractors	Historic	Hazard Forecasting; Impact Forecasting; Impact Warning	Built environment, infrastructure	Response; Situational Awareness; Recovery; Planning; Mitigation
Soil/land stability	Official	GNS Science	Historic?	Observation, Monitoring, and Detecting; Hazard Forecasting; Impact Forecasting	Hydrogeological; Built environment, infrastructure, Environment, Property	Planning; Mitigation; Research

Like impact data, vulnerability data is also present as tacit knowledge and experience. As previously mentioned, infrastructure engineers and other asset managers possess a wealth of knowledge and experience around the performance capacities of a building, levy, or other piece of infrastructure. In the risk modelling space, this tacit knowledge, or expert opinion is a valid resource for building vulnerability functions:

If you are an asset manager and you know about your port or your wharf ... and you just intuitively think about ... if a six-meter wave came in what do you think would happen? ... you can elicit information that way. And there's a lot of vulnerability functions out there that are done that way because you don't need an event to happen. And I guess there is a certain level of knowledge and intuition that can go into these things ... So you can do some kind of estimate (Risk Modelling NZ. A).

CDEM Groups possess tacit knowledge around vulnerable areas, people, and communities within their jurisdiction (Hyd. Gov. NZ. Reg. A; Auckland Workshop), but our participants indicated that they still need to know "who is where and what their mobility/health/access considerations are" (Auckland Workshop). Some of this information is available through pre-existing networks with EM services that manage relationships with vulnerable communities, such as the "Caring for Communities" government work programme that was established in response to COVID-19²⁶, and through Welfare coordination groups (Auckland Workshop). However, there is a risk of missing people who are not "in the system" (Auckland Workshop). Furthermore, privacy concerns exist when considering alternative uses for the data beyond its initial purpose, which is usually to support response and recovery (Auckland Workshop; EM. NZ. Reg. H; EM. Gov. NZ. Nat. G). Yet, even this tacit knowledge can still help EM groups formulate their warning messages (EM. NZ. Reg. H).

All datasets listed in Table 7.6 are produced and housed in an official capacity. Our interviews did not find any instances where crowdsourcing or other unofficial methods were used for producing vulnerability data. Some research has been done to use crowdsourcing applications for capturing vulnerability. For example, in Kazakhstan Fdez-Arroyabe et al. (2018) developed mobile application (OxyAlert) where users can answer questions about their health status relative to their geographic location and atmospheric conditions to characterise individual vulnerability to meteorological changes. Hung and Chen (2013) presented a new participatory mapping methodology for incorporating stakeholder's participation, local knowledge, and locally spatial characteristics for vulnerability assessments of flood risk. These studies demonstrate the untapped opportunity to use crowdsourcing and other engagement activities to produce localised or individualised vulnerability assessments.

²⁶ <u>https://ipanz.org.nz/Article?Action=View&Article_id=150258</u>

7.6.1.4 Exposure Data

The dynamic nature of exposure data also makes it difficult to collect and maintain. However, our participants identified some existing and potential sources of exposure data for different uses and with different time scales, summarised in Table 7.7. Most of the data sources are classified as official, as they are produced by official agencies.

Asset footprints refer to the geographical location of assets, such as buildings, historically and culturally significant sites, etc. These data are officially produced by local/regional councils, LINZ, and FENZ. They are also available on unofficial platforms like Google Maps and OpenStreetMap²⁷. This location-based data helps EMs and researchers identify assets that are potentially exposed to a hazard.

Exposure is typically determined by overlaying the asset information with hazard information (Risk Modelling NZ. C). This is common practice for disaster and climate change mitigation. For example, Paulik et al. (2020) created a national-scale built environment exposure model to extreme sea level rise for NZ by overlaying buildings, infrastructure, and built land area with a Digital Elevation Model and coastal flood maps. Similarly, Bell et al. (2016) compared urban and rural exposure to coastal hazards using demographic data overlaid with building, infrastructure, and land assets. Their results provided counts of people, building, infrastructure, and land assets located in areas exposed to coastal hazards. While these exposure models and their outputs help to locate and quantify exposed people and assets to a given hazard, the results of these models represent one point in time. Thus, they do not accurately represent the dynamic nature of exposure.

Other data sources can capture dynamic exposure, such as live traffic flows from transportation agencies (e.g., the New Zealand Transport Agency) (Data Management Gov. NZ. Nat. A; Met. NZ. K). Cell phone data was also discussed for capturing population shifts (Risk Modelling NZ. A; Data Management Gov. NZ. Nat. E). For example, Data Ventures, a commercial data brokerage branch of Stats NZ, used cell phone data to produce population densities at the Statistical Area 2 level (a higher scale than the Meshblock level) for a given time range²⁸.

 ²⁷ OpenStreetMap is a form of geographic crowdsourcing in which volunteers digitize features of the earth onto an online map of the world. This data is produced under a Creative Commons license, thus making it open-source and freely available for download.
 ²⁸ See <u>https://population-density.dataventures.nz/explorer2/help/index.html</u> for more.

Table 7.7. Summary of exposure data sources identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, uses in the IFW Value Chain, type of impact, and non-IFW uses²⁹.

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	IFW Uses (from the Value Chain)	Type of Impact	Non-IFW Uses
Asset footprints (e.g., building, cultural and historical site locations)	Official and unofficial	Councils, Google Maps, OpenStreetMap, LINZ, FENZ, Heritage NZ, Iwi	Historic	Hazard Forecasting; Impact Forecasting	Property, Cultural, Social	Response; Research; BAU; Mitigation
Infrastructure networks, e.g., transportation networks, traffic flows, power & water supplies	Official	Lifelines Services (e.g., NZTA, Transpower, KiwiRail), local & regional councils	Current	Hazard Forecasting; Impact Forecasting	Infrastructural	Response; Situational Awareness; BAU
Census data	Official	Stats NZ	Historic	Hazard Forecasting; Impact Forecasting	Human	Research; BAU
Population movement via cell phone data	Official	DataVentures / Stats NZ	Historic and [Near] real- time?	Hazard Forecasting; Impact Forecasting; Impact Warning	Human	Research

²⁹ The acronyms and abbreviations in Table 7.7 are as follows: Fire and Emergency New Zealand (FENZ), National Institute of Water and Atmospheric Research Ltd (NIWA), New Zealand Transport Agency (NZTA), Institute of Geological and Nuclear Sciences Limited (GNS Science), Statistics New Zealand (Stats NZ), Ministry of Business, Innovation, and Employment (MBIE).

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	IFW Uses (from the Value Chain)	Type of Impact	Non-IFW Uses
Tacit knowledge, experience, intuition	Official	Councils, CDEM	Available in real-time, based on historic knowledge and experience	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Response; Situational Awareness; Preparedness; Mitigation; BAU
Topographical data, e.g., digital elevation models	Official	LINZ, Landcare Research, Universities, GNS Science, NIWA	Historic	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Rural, Urban, Environmental	Response; Situational Awareness; Research; BAU
Land-use	Official	Councils, LINZ	Historic	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Rural, Urban, Environmental	Response; Situational Awareness; Research; BAU
Community events	Official	MBIE, Stats NZ, Local and regional cultural and tourism agencies (e.g., Auckland Unlimited)	[Near] real-time	Hazard Forecasting; Impact Forecasting; Impact Warning	Cultural, Social	Response; Situational Awareness; BAU

This product became useful in the COVID-19 response where Data Ventures provided near real-time population movements to the National Crisis Management Centre which "provided [a] very close to real time view of whether or not people were following the advice" (Data Management Gov. NZ. Nat. E), which informed advice to the Prime Minister's COVID-19 advisory group "about whether or not we move or shift down levels of lockdown" (Data Management Gov. NZ. Nat. E). Through this, they were also able to determine "whether or not people were moving from one region to another because that's really how you have that wider contagion risk" (Data Management Gov. NZ. Nat. E). This has potential application in an IFW system for identifying the "catchment of individuals" (Data Management Gov. NZ. Nat. E) in near real-time for more contextualised warnings. However, this data alone does not provide an overall risk indication; "you have to put other layers of information to go 'well if it's a ... category 5 [ex-tropical cyclone] then are the buildings of a particular standard and ... how many people are in the areas that those buildings are present in'" (Data Management Gov. NZ. Nat. E). Hence the need for vulnerability information as previously discussed.

Knowledge of dynamic human exposure can also come from knowing about community events (Auckland Workshop). A regional EM official in NZ identified the need for "real-time understanding of what is going on in the community (i.e., sports events, hotel capacity, etc.)" (Auckland Workshop). In this case, the regional council's economic and cultural department possesses this knowledge and information (Auckland Workshop) and should be shared with the EMs and other agencies such as the MetService who would need to be aware of events if severe weather were to occur.

7.6.2 Data Creation, Collection, and Use

Discussions with participants revealed several inhibitors and facilitators to collecting, creating, and using the data. Using the ES-GT coding paradigm, we identified the causal conditions driving data collection, creation, and use, intervening conditions to this phenomenon, and actions and strategies that have been used to address those intervening conditions (Figure 7.1). The causal conditions are presented next.



Figure 7.1. Results of applying the coding paradigmto understand the causal conditions, intervening conditions, and action/interaction strategies for HIVE data creation, collection, and use phenomenon as defined in Table 7.3. Causal conditions are the drivers of the phenomena. Intervening conditions inhibit or facilitate the phenomena. Action/interactions strategies were identified to address the intervening conditions.

7.6.2.1 Causal Conditions

Causal conditions were identified as drivers for agencies collecting and using HIVE data in general for different purposes and via different methods. The main causal conditions identified by participants are: (1) Disaster/emergency events, (2) technological advancements, and (3) Research.

Other factors can also be attributed to HIVE data collection, such as existing policies and plans (e.g., the National Disaster Resilience Strategy) within NZ, and international initiatives such as the Sendai Framework. For the purposes of this study, we have decided to focus these current findings on the causal conditions that participants identified more directly.

Disaster/Emergency Events

Our participants gave examples of how the creation, collection, and use of HIVE data has been driven by past disaster/emergency events. Following the 2010-2011 Canterbury earthquake sequence there was a need for data on the damaged buildings, such as their age, construction material, and level of damage to inform recovery (Risk Modelling NZ. B). At the time, no database existed for building characteristics (e.g., history, age, construction material) (Risk Modelling NZ. B). Consequently, engineers had to "go through one by one to record the location and age ... of the material" before they could conduct the damage assessments (Risk Modelling NZ. B). This experience, where decision-makers were caught scrambling for data, revealed a need for building up databases for future events (Risk Modelling NZ. B; EM. NZ. Reg. H).

This and other major events resulted in the formation of a Technical Advisory Group (TAG) by the Ministry of Civil Defence and Emergency Management (now the National Emergency Management Agency or NEMA), in 2017 (Treadgold et al., 2018). The TAG conducted a Ministerial Review and provided several recommendations for better intelligence gathering (i.e., data collection) (Technical Advisory Group, 2017). A participating regional EM official cited this review as a driver for their council and EM Group attempting to improve their data collection efforts for disaster response (EM. NZ. Reg. H). Additionally, in response to two back-to-back ex-tropical cyclones that resulted in disastrous flooding, the Bay of Plenty region EM Group now updates exposure maps for tropical cyclone hazards annually (EM. NZ. Reg. A).

Technological Advancements

Technological advancements, particularly the implementation of geospatial technologies such as Geographic Information Systems (GIS), cloud-based services, and the proliferation of mobile devices with cameras and an internet connection have also driven efforts for better collection of HIVE data. For example, one participant described Esri's ArcGIS Online product as the "catalyst" for using geospatial technologies for emergency response (EM. NZ. Reg. H). The cloud-based aspect of the ArcGIS Online product supports rapid development and sharing of maps and other geospatial applications, which is seen as the most valuable feature for

emergency response (EM. NZ. Reg. H). As such, "the strength of the NZGIS4EM [New Zealand Geographic for Emergency Management] community really ... is built upon that. And ... there's ... been continual growth with event after event" (EM. NZ. Reg. H). Furthermore, this cloud-based technology used in combination with mobile devices using applications like QuickCapture and Survey123 allowed for the redevelopment of how rapid impact assessments conducted for more efficient and timely data collection. For example, with Survey123, "the fieldworker hits submit, that assessment will instantly show up on the map [in the Emergency Operations Centre]" to display the level of damage and safety of the buildings (EM. NZ. Reg. H).

Research

Research has also driven the collection of HIVE data. For example, NIWA's NZ Historic Events Catalogue³⁰ was initially developed for research interests, where "people realised that it would be really good to have this database for people and researchers, and just sort of give an idea of what historically has happened ... And I think it's been quite useful" (Met. Research NZ. J). Other HIVE data has been collected to support the research and development of risk models (Risk Modelling NZ. B).

7.6.2.2 Intervening Conditions for Data Creation, Collection, and Use

Intervening conditions were found to affect whether HIVE data is collected, and the choice for which data source is used. Herein we focus on three intervening conditions: (1) Priorities, motivation, and interest; (2) Official and unofficial data; (3) Timescale and trustworthiness.

Priorities, motivation, and interest

Conflicting priorities and a lack of motivation or interest in data creation and maintenance was a barrier identified by several participants for gathering HIVE data (EM. Gov. NZ. Nat. G; Data Management Gov. NZ. Nat. A; Data Management Private NZ. B; Data Management Research NZ. C; EM. NZ. Reg. D). Management priorities and the personal interests of key staff within an organisation appear to either inhibit or enable data collection and creation, as one NZ risk modeller summarised: "it just depends on who's here and who's leading the team" (Risk Modelling NZ. A).

The National Emergency Management Agency (NEMA) for New Zealand reports losses and impact to the UNDRR in fulfilment of Sendai Framework priorities. Their goal is to develop a national loss and impact database (EM. Gov. NZ. Nat. G). However, progress on this front is slow as NEMA is continuously busy responding to other events and so is unable to direct resources towards developing a national loss and impact database (EM. Gov. NZ. Nat. G). This may be due to the reactive nature of the EM sector (Kox, Lüder, et al., 2018), where agencies lack the time or resources to establish proper data collection and management practices (EM. Gov. NZ. Nat. G; Loss Modelling Research NZ. A; EM. NZ. Reg. A, B, H; Data Management Private NZ. B; Hyd. Gov. NZ. Reg. A).

³⁰ <u>https://hwe.niwa.co.nz/</u>

Official and unofficial data

The leading mandate of many of the participating agencies is to preserve life and property, be it through designing and issuing warnings (e.g., NMSs, warning services) or coordinating emergency response plans and actions (EM agencies). A key role for these agencies is providing the voice of truth during a severe weather event. These agencies must maintain a high level of credibility and trustworthiness amongst the public and stakeholders to ensure that their messages are heeded (NMS. Int. C, D, E; NMS. NZ). Since these agencies use a plethora of data to communicate critical information and alerts to the public, trust in the supporting information underpinned all discussions with participants.

Official and unofficial datasets and sources were discussed and compared with interviewees. Interviewees showed a preference for official datasets and sources because of their role as an authoritative voice in saying "this is what happened" or may happen (NMS. Int. C). However, collecting and using official data is not always possible. For example, agencies who possess the data may be unable or unwilling to share (NMS. Int. C), or agencies may need real-time data, which is rarely official or trusted (NMS. Int. D). In these cases, agencies may need to turn to unofficial data sources such as crowdsourcing and social media (NMS. Int. C).

Time scale and trustworthiness

Forecasting impacts in real-time or near real-time for early warning is an operational goal (Aldridge et al., 2016; Hemingway & Robbins, 2019). Thus, our participants identified a need for real-time or near real-time data (Met. Int. E; EM. NZ. Reg. H; Risk Modelling NZ. C, D; Data Management Private NZ. B). Potential real-time data sources that were identified are crowdsourcing, social media, and mobile tracking data. Aside from the mobile tracking data these sources are unofficial data sources, resulting in decreased perceptions of trust and credibility in the data (Met. NZ. H; Met. Int. D, I). As such, our results indicate trustworthiness, and the timescale needs appear to be two intervening conditions in the choice and use of a data source. A tension appears to exist between these two conditions, as officials (e.g., warning agencies, emergency managers, responders) have a need for both timely and trustworthy data, but often have to compromise on either factor (Mehta et al., 2017).

Distrust is a critical obstacle to warning adherence (Covello & Sandman, 2001). Thus, it is not surprising that trust in the data and in the sources to support IFWs is a primary driver in deciding which impact, exposure, and vulnerability dataset/source to use. It is important that warning officials perceive the data as trustworthy (Terti et al., 2019), as this will ensure that public trust in the agency and in the warnings is maintained. Different uses of impact, exposure, and vulnerability data require varying levels of trustworthiness.

The timescale needed also determines the most appropriate dataset. We found that some data sources, like crowdsourcing and social media, fill a gap in the need for real-time or near real-time data for verification, situational awareness, and response. This finding supports those of a recent survey of European NMSs (Kaltenberger et al., 2020). However, these data may be less useful for defining impact thresholds and informing impact/risk models due to the perceived limited quality and trustworthiness. For example, social media data appeared to be less trustworthy amongst participants because it is difficult to vet and lacks structure needed for forecasting and modelling (Met. NZ. G, H). However, social media remains useful for building situational awareness quickly and updating alerts (EM NZ Reg. C; Harrison & Johnson, 2016).

7.6.2.3 Action/Interaction Strategies

Action/interaction strategies were found to address the intervening conditions previously identified. The three strategies we focused on here are: (1) Garnering support and buy-in, (2) Individual and community leadership, and (3) Quality control and standardisation.

Garnering support and buy-in

Garnering support and buy-in was found to be an action/interaction strategy for overcoming conflicting management priorities or a lack of motivation and interest. One regional EM official has been pushing for a GIS-based approach for improved intelligence gathering, management, sharing, etc., but has faced resourcing challenges. The first "informal" (EM. NZ. Reg. H) stage of this project "didn't turn out to be sustainable because we had a lot of issues with trying to carve out time from people to actually contribute to it" (EM. NZ. Reg. H). As such, they developed a "more formal" (EM. NZ. Reg. H) strategy to approach decision-makers for support for resource allocation across regional stakeholders/agencies (EM. NZ. Reg. H). Efforts for garnering support are still underway (EM. NZ. Reg. H).

Individual and community leadership

Individual and community leadership was found to be another action/interaction strategy for overcoming misdirected management priorities and lack of motivation or interest. Some regional agencies have improved their own data collection practices based on the leadership of their in-house GIS and EM experts. For example, the West Coast CDEM Group regularly develops innovative ways for using GIS-based technology to carry out impact assessments (EM. NZ. Reg. C, F; see Stowell, 2020). Furthermore, the emergence of the NZGIS4EM³¹ group has been identified by participants as a major driver for technological advancement and for pushing the needle forwards on creating, sharing, and accessing geospatial data and tools for emergency response

³¹ NZ Geographic Information Systems for Emergency Management (NZGIS4EM) is a grassroots community of GIS specialists and EM practitioners in NZ that began working together in the mid-2010's to identify how they could build and use geospatial tools for better emergency response (EM. NZ. Reg. H).

(EM. NZ. Reg. H; Data Management Gov. NZ. Nat. A; Data Management Private NZ. B; EM. NZ. Reg. D).

This innovative GIS-based work is credited to specific individuals within the sector who possess a passion and expertise to drive these efforts (Data Management Private NZ. B; EM. NZ. Reg. B, F). Furthermore, the innovation may also be due to available resources and support from management. Our findings align with the Policy Capacity Framework (Wu et al., 2018) which outlines three levels of capacity development and implementation: individual, organisation, and systemic; and three dimensions: analytical, managerial, and political. The individuals accredited with driving the GIS4EM movement in NZ appear to possess analytical, managerial, and political acumen capacity at the individual level by possessing analytical, technical, communication, and leadership skills to drive technological innovation within their sector. Agencies like the West Coast CDEM Group possess technical and administrative capacity at the organisational level by providing and coordinating the resources needed to allow the individuals to implement their innovative solutions. More investigation is needed to understand the policy capacities at the systemic level and political dimension within the NZ EM and severe weather warning space. However, it appears that the GIS movement within NZ's EM sector has reached central government decision-makers with the recent release of the Impact Assessments Director's Guideline for Civil Defence Emergency Management Groups [DGL 22/20] by NEMA (2020a). This guidance document outlines the role of GIS and supporting spatial tools for undertaking impact assessments.

Quality control and standardisation

Participating agencies emphasised the importance of applying quality control measures to increase perceived trustworthiness of the data. Some quality control measures that came up in the conversations included vetting the source(s) of the data, training storm spotters, cross-validating between sources for accuracy, timing, and location. For example, the NMSs in the USA and Austria train and vet storm spotters so that they can trust incoming ground observations (Met. Int. E, I; Kaltenberger et al., 2020; Krennert, Kaltenberger, et al., 2018). Standardised post-event damage/impact assessments collected by those trained to, and entered into a database for further analysis, may be more suitable for needs that do not require real-time data (Kaltenberger et al., 2020).

7.7 Conclusions and Limitations

Documenting hazard, impact, vulnerability, and exposure data fulfils needs for IFWs, and meets Sendai Framework priorities for improved understanding of disaster risks and subsequent mitigation and reduction. The New Zealand focus of this research further supports an identified need for better risk data for modelling and natural hazard management in New Zealand.

In this exploratory study we identified sources for hazard, impact, vulnerability, and exposure data for implementing severe weather IFW systems. Our findings indicate that many sources for hazard and impact data exist that are collected for other uses (such as for response efforts for disaster and emergency events, and for research) and have relevant application for IFWs. Furthermore, underlying datasets for vulnerability and exposure exist and are available. Technological advancements have also enabled the collection and creation of HIVE data, such as GIS-based tools and mobile devices.

We also identified intervening conditions, and action/interaction strategies for collecting HIVE data, as shown in Figure 7.1. Our findings suggest that priorities, motivation, and interest within organisations influence how well data is collected and used. Furthermore, agencies tend to prefer official data, but official data has limitations that unofficial data can sometimes address, such as timeliness. To that end, a tension exists between the timeliness and trustworthiness of data needed for emergency response and warnings.

To address these intervening conditions, we identified some action/interaction strategies using Grounded Theory. Garnering support and buy-in from decision-makers and upper management within an agency can redirect priorities and increase motivation and interest in collecting HIVE data. Individual and community leadership within the field of practice also provides a bottom-up approach for driving industry priorities and practices for collecting HIVE data. Furthermore, measures for quality control and data standardisation may improve the perceived trustworthiness of the data.

The qualitative nature of data collection and analysis herein limits the generalisability of results beyond the interviewees. However, the qualitative approach offers in-depth understanding of a problem not readily available from quantitative approaches (Blumer, 1969; Miles & Huberman, 1994; Patton, 2002). Furthermore, participant recruitment and data collection methods were affected by the COVID-19 pandemic response, as many individuals and agencies targeted for recruitment were involved in the response. As such, some perspectives may be missing from the qualitative dataset.

Discussions with colleagues in the field pointed towards the value of mātauranga Māori (Māori knowledge) in understanding disaster risk and impacts in New Zealand. Considerations about cultural ownership of such knowledge and its use in an impact forecasting and warning system is thus an important area for future research.

Findings from this research provide insight into the drivers and barriers for collecting hazard, impact, vulnerability, and exposure data in New Zealand. Sources of such data were identified such that practitioners and researchers may seek out these datasets if so desired. Further questions remain around how the data can be accessed and acquired for use in an IFW system. Future research should explore the data acquisition process for these datasets.

	Chapter One				
Part 1	Introduction				
	Chapter I wo				
	Background and Context				
	Chapter Three				
	Research Design				
Part 2	Chapter Four				
	Methods Results and Overarching Findings				
	Methods Results and Overarening Findings				
	Chapter Five				
	Volunteered Geographic Information for people-				
	centred severe weather early warning: A				
	literature review				
	Chapter Six				
	Identifying the data uses and gaps for severe				
	weather impact forecasts and warnings				
	Chapter Seven				
	Identifying data sources for severe weather				
Part 3	impact forecasts and warnings in Aotearoa New Zealand				
	Chapter Eight				
	A socio-technical analysis of sharing and				
	governing hydrometeorological impact data in				
	Aotearoa New Zealand				
	Chapter Nine				
	Integration of Partnerships and Collaboration to				
	Support IFWs and HIVE Data Collection, Access,				
	and Use				
	Chapter Ten				
Part 4	Chapter Ten Discussion and Conclusion				

Position of the reader

This chapter presents the fourth manuscript for this doctoral study prepared for publication. The focus of this paper is on accessing and managing hazard, impact, vulnerability, and exposure data for severe weather hazards. Following discussions with participants, and during the Grounded Theory analysis process (Strauss & Corbin, 1990), data governance, management, access, and sharing emerged as important themes that influence the use of hazard, impact, vulnerability, and exposure (HIVE) data.

From Chapter Seven we now know that many sources of HIVE data exist and have identified the need to investigate governance and accessibility of these data. To that end, this chapter focusses on the following objectives of this doctoral thesis: 4.1 Identify and understand the governance and access and sharing processes of HIVE data for severe weather hazards in New Zealand; 4.2 To support efforts to fulfil the Sendai Framework priorities around disaster data access; and 4.3 To support the implementation of a severe weather IFW system.

This manuscript was published in 2022 in *Progress in Disaster Science:*

Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E.-H., & Johnston, D. (2022).
'Sharing is caring': A socio-technical analysis of the sharing and governing of hydrometeorological hazard, impact, vulnerability, and exposure data in Aotearoa New Zealand. *Progress in Disaster Science*. https://doi.org/10.1016/j.pdisas.2021.100213

The published form is available in Appendix M.

8.1 Abstract

There has been a growing recognition of the need to collect disaster and risk data over the last two decades. Accordingly, better collection and management of disaster data was identified as a priority of the Sendai Framework for Disaster Risk Reduction. The introduction and implementation of Impact Forecasts and Warnings (IFWs) have further highlighted this need to collect and access hazard, impact, vulnerability, and exposure (HIVE) data. However, challenges have been met with reporting and using disaster data, which have resulted in an identified need to establish principles for data collection, recording, reporting, exchange/sharing, and comparability. This introduces the concept of data governance and management for disaster data, particularly with regards to data custodianship, stewardship, and sharing. Using Grounded Theory, a series of interviews were conducted with users and creators of HIVE data to develop further understanding around managing and accessing it for severe weather hazards in New Zealand. A socio-technical lens guided the analysis to identify the organisational and technical intervening conditions and action/interaction strategies for accessing and sharing HIVE data in NZ.

Findings indicated that there is a need to establish data governance principles for HIVE data. An additional need was identified for nurturing partnerships to continue building trust between stakeholders for sharing data. Furthermore, integration challenges continue to interfere with the use of various sources of HIVE data for effective risk and impact assessments for IFWs and beyond. Systematic and standardised data collection approaches using GIS-based tools can support integration. Introduction

8.2 Introduction

There has been a growing need to collect disaster and risk data over the last two decades (e.g., Guha-Sapir & Below, 2000). Accordingly, better collection and management of disaster data was identified as a priority of the Sendai Framework for Disaster Risk Reduction (UNDRR, 2015b). In response, global initiatives now exist with the objective of developing technical guidance for building up disaster and risk data, such as the Integrated Research on Disaster Risk (IRDR) programme and subsequent workshops (Clarke et al., 2018; Fakhruddin et al., 2017). The introduction and implementation of Impact Forecasts and Warnings (IFWs) for hydrometeorological hazards (e.g., WMO, 2015) has further highlighted the need to collect and access relevant disaster data for mitigation and prevention; namely hazard, impact, vulnerability, and exposure (HIVE) data (Harrison et al., 2022; Potter et al., 2021). However, challenges have been met with reporting and using disaster data, such as the fact that many stakeholders are involved in the collection, creation, and use of disaster data, making it difficult to integrate different data sources and perform comparative analyses (Díaz & Ferrer, 2018). These challenges have resulted in an identified need to establish principles and standards for data collection, recording, reporting, exchange/sharing, and comparability (Fakhruddin et al., 2017). This introduces the concept of data governance for disaster data (Clarke et al., 2018; Li et al., 2019; Migliorini et al., 2019), particularly with regards to data custodianship, stewardship, and sharing (Plotkin, 2020; Zaidi, 2012).

Building on recent research where we identified the data needs, uses, and sources of HIVE data for implementing hydrometeorological IFW systems in Aotearoa-New Zealand (see Harrison et al., 2021; Harrison et al., 2022), we aim to develop further understanding around managing and accessing these data sources. We begin by presenting background on data governance and management, and data access and sharing, where data accessibility can affect data sharing (OECD, 2019).

8.2.1 Data Governance

Data governance is an emerging field of research, with no agreed-upon definition (Al-Ruithe et al., 2018). We adopt the definition provided by Benfeldt et al. (2019, p. 299) where "data governance refers to the organisation and implementation of rules and responsibilities, which enforce decision making and accountabilities regarding an organisation's data assets." This is different from data management, which focuses on defining the data element, and how it is stored, structured, and moved (Al-Ruithe et al., 2018). Thus, data governance is argued to be a higher level of "planning and control over data management" (Al-Ruithe et al., 2018, p. 841).

There has been little mention of governance for disaster loss data in the disaster risk reduction (DRR) literature, except when it is pointed to as a need to improve data quality, access, sharing, and interoperability (e.g., Clarke et al., 2018; Li et al., 2019; Migliorini et al., 2019). Clarke et al. (2018, p. 4) proposed that data governance and independence be established for "strengthening and protecting data quality through national statistical offices that are functionally autonomous from other government

agencies." Migliorini et al. (2019) identified a lack of appropriate data governance arrangements to be impeding data access for DRR. Similarly, Li et al. (2019) identified issues around data governance when using population health data for disaster risk research. In NZ, Crawford et al. (2018) identified challenges with risk data collection due to unclear roles and responsibilities for doing so. Beyond these examples, we could not find studies that specifically investigated disaster data governance, despite the apparent need to establish appropriate data governance measures for improved data quality, sharing, and integration.

One element of data governance involves assigning roles and responsibilities, including decision rights and accountabilities, around how the data is managed, secured, validated, and made available (Al-Ruithe et al., 2018; Alhassan et al., 2019; Benfeldt et al., 2019). Two such roles are data stewardship and data custodianship (Zaidi, 2012). Often these roles may be confused or merged, however, they have distinct roles and responsibilities, discussed next.

8.2.1.1 Data stewardship and custodianship

Data stewardship and data custodianship relate to roles for managing data. In the context of health data, data stewardship consists of developing methods for "acquisition, storage, aggregation, deidentification, and procedures for data release and use" (Rosenbaum, 2010, p. 1444). Similarly, the NZ Government defines a data steward as an agency that operates at the systems and strategic level and promotes good practice to manage the data over its lifecycle, including planning and adjusting for technological obsolescence and long-term preservation and access (LINZ, 2014b; Secretary of Cabinet, 2011). Related, but distinct roles, are data custodians. For NZ government data these custodians are agencies that implement the data management practices stipulated by data stewards in daily practice (Secretary of Cabinet, 2011) to ensure the quality and accessibility of the data (LINZ, 2014b).

In NZ, several agencies have a data stewardship and/or custodianship role. Toitū Te Whenua Land Information New Zealand (LINZ; the government agency responsible for managing land titles, geodetic and cadastral survey systems, topographic information, hydrographic information, etc.) provided a Steward and Custodian Framework for New Zealand Fundamental Geospatial Themes and Datasets, to outline the responsibilities and expectations of appointed custodians and stewards of fundamental geospatial data (see LINZ, 2014b). LINZ also provided a partner document outlining the process of identifying and selecting fundamental geospatial data (see LINZ, 2014a). In summary, datasets are proposed to the New Zealand Geospatial Office for evaluation against a set of criteria to classify it as a fundamental geospatial dataset (LINZ, 2014a). Fundamental geospatial datasets are "datasets that provide the minimum core set of nationally-significant data that are critical to the effective running of [NZ], and work together to help support growth in the economy" (LINZ, 2014a, p. 4). Most of these fundamental geospatial datasets are not considered disaster datasets as they do not convey disaster losses, impacts, or risks. However, much of these underlying datasets can inform vulnerability and risk assessments when

used in risk models and overlaid with hazard data to determine exposure (see Harrison et al., 2021).

The fundamental geospatial datasets identified by LINZ are labelled as either Suggested (i.e., the agency suggested by LINZ has not yet agreed to the role and may not have been approached yet), Proposed (i.e., the suggested agency has agreed to the role but the commitment has not yet been formalised), Appointed (i.e., the proposed agency has formally committed to the role), or Not Evaluated (i.e., the datasets were proposed as fundamental datasets but were not yet evaluated to determine their status) (LINZ, 2014a). For example, the Waka Kotahi NZ Transport Agency (NZTA) is the proposed custodian for the leadership and delivery of the national road network, while local government and territorial authorities are the suggested respective leadership and delivery custodians for council roads. Leadership Custodians ensure that appropriate data management policies and standards are developed while Delivery Custodians are "responsible for the continued physical existence, availability, and integrity of the dataset" for as long as is required by the leadership custodian (LINZ, 2014b, p. 25).

Like disaster data governance, there is little mention of the stewardship of disaster data in the literature. This gap has been identified by Fakhruddin, Murray, et al. (2019) who identified the need for building capacity in data collection and stewardship to support reporting to the Sendai Framework, an international accord outlining global priorities for disaster risk reduction (DRR) (UNDRR, 2015b). Another challenge faced in DRR is accessing and sharing the required data (Fakhruddin, Chu, et al., 2019) for emergency/disaster response, risk analysis (e.g., risk modelling and risk assessments), vulnerability assessments, and supporting IFWs (Harrison et al., 2021).

8.2.2 Data Access and Sharing

Many stakeholders are involved with DRR who collect, produce, and manage their datasets. Access to and sharing of these datasets is necessary for DRR (Fakhruddin, Murray, et al., 2019). For example, in places like Europe with many countries sharing borders, it is important to be able to share data across borders (De Groeve, 2015). Within a country, sharing data is important since, in many cases, disaster and risk data are collected by different agencies (Harrison et al., 2021). Data access refers to the retrieval and storage of data provided by the data holder and may be subject to technical, legal, and/or organisation requirements (OECD, 2019). Data sharing is the voluntary provision of data by the data holder, including commercial and non-commercial conditional data sharing agreements (OECD, 2019). Data accessibility is a spectrum, ranging from closed data to open data, and affects data shareability (OECD, 2019). Data sharing is a socio-technical activity as it involves various parties forming data-sharing partnerships and technical systems to support integrating multiple datasets.

8.2.2.1 Organisational Aspects of Data Access and Sharing

Sharing disaster and risk data requires building partnerships between data stakeholders (Fakhruddin, Chu, et al., 2019). The Natural Hazards Partnership (NHP) is an example of a formal partnership developed in the UK between public agencies to improve disaster management across the country, including sharing data between the various agencies (Hemingway & Gunawan, 2018). Building the NHP required extensive time, coordination, communication, and interaction between agencies (Hemingway & Gunawan, 2018).

Multi-organisational collaboration is wrought with challenges (Sonnenwald, 2007). Building trust and increasing the willingness of organisations to participate is the first hurdle to overcome (Sonnenwald, 2007). After this, new challenges include mutually identifying goals and objectives and agreeing on timelines, uses of differing terminology and epistemologies, legal issues around intellectual property, and developing workflows and communication standards (Sonnenwald, 2007). These early stages of scouting and building trust can take years (Sonnenwald, 2007). Sustaining collaboration (e.g., maintaining interest and securing funding sources) remains an ongoing challenge (Hemingway & Gunawan, 2018; Sonnenwald, 2007). In addition to building partnerships to share data, technical solutions are needed to enable data integration.

8.2.2.2 Technical Aspects of Data Access and Sharing

Data integration and interoperability are important, yet challenging, technical factors that support data sharing for DRR (Fakhruddin, Chu, et al., 2019). Disaster and risk data are available in countless formats, making integration difficult (Horita et al., 2017). Thus there is a need to understand the various data sources and how they can be effectively and efficiently used (Horita et al., 2017).

Interoperability is a familiar challenge for both EWSs and data sharing and integration (Botterell, 2006; Horita et al., 2015). Interoperability issues for EWSs emerged as technological advancements led to a plethora of warning delivery mechanisms (Botterell, 2006). The Common Alerting Protocol (CAP) was proposed to set standards for warning design and delivery (Botterell, 2006). The CAP relies on standardising warning data for sharing across platforms (OASIS, 2010; Rebelo Moreira et al., 2018). The introduction of IFWs further adds to the challenge of standardised data exchanges (Kaltenberger et al., 2020; Potter et al. 2020).

In NZ the Canterbury earthquakes from 2010-2011 laid bare gaps in interoperability for data sharing during and after the disaster, and the need for standards-based interoperability for improved information and data management (Treadgold et al., 2018). Furthermore, the NZ Civil Defence and Emergency Management (CDEM) sector is striving towards a Common Operating Picture (COP) in which all agencies and stakeholders involved in an event can access and view the same information (Department of the Prime Minister and Cabinet, 2018). This involves developing standards and capabilities for enabling access to and sharing of datasets (Treadgold et al., 2018).

Harrison et al. (2021) identified and reported the various sources for HIVE data for hydrometeorological hazards in New Zealand (NZ). While the data sources were identified, more understanding is needed around managing and accessing these data both for IFW systems and for general DRR (Harrison et al., 2021).

The objectives of this research are to identify and understand the governance and acquisition process for hazard, impact, vulnerability, and exposure (HIVE) data for hydrometeorological hazards in NZ, to support efforts to fulfil the Sendai Framework priorities around disaster data access and to support the implementation of a hydrometeorological IFW system.

8.3 Research Methods

We used a qualitative approach to address the research question, specifically employing the Evolved-Straussian Grounded Theory (ES-GT) research strategy for data collection and analysis (Corbin & Strauss, 2015). Interviews, workshops, and key documents were the primary data sources. Between November 2018 and April 2021, the lead author interviewed thirty-nine (n=39) experts in weather forecasting, warning, emergency management, risk modelling, and data collection and management (see Harrison et al., 2022 for more information on the participant details). Three virtual workshops were held in NZ. Two of these workshops involved Emergency Management (EM) practitioners, weather forecasters, communication and data specialists, and hydrologists from the Auckland (n=4) and Southland (n=5) Regions. The third workshop involved a portion of the NZ risk and hazard science community based at GNS Science (n=11). Thus, 59 people participated in this research.

Interview questions and workshop activities focused on themes regarding IFW data needs and sources. We asked for participants' general thoughts on IFWs; what impact, vulnerability, and/or exposure data they use or need, why, and how; the life path of the data; experienced and/or perceived challenges obtaining data required for IFWs and other uses; and thoughts on collecting and using alternative data sources (e.g., social media and crowdsourcing). Findings around data governance and acquisition are reported in this paper.

This research was collected under a 'low risk' ethics notification with the Massey University Human Ethics Committee. All interviewees remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by the area of expertise and/or practice, industry, location, or governance level (Table 8.1). Interviews were audio-recorded and transcribed verbatim, and qualitative analysis (including coding and memo-writing) followed the axial coding paradigm according to the ES-GTM (Corbin & Strauss, 2015), using NVivo 12 qualitative analysis software (Bergin, 2011).

Table 8.1. Interviewee Codes.

Interview Code	Position	Classification	Location	Government Level
Agriculture/Rural NZ. A	Agriculture policy coordinator	Agriculture/Rural	NZ	National
Data Management Gov. NZ. Nat. A	Senior Resilience Advisor	Data Management	NZ	National
Data Management Private NZ. B	Geospatial Specialist	Data Management	NZ	Private
Data Management Research NZ. C	GIS Specialist	Data Management	NZ	Private
Data Management Private NZ. D	GIS Specialist	Data Management	NZ	Private
Data Management Gov. NZ. Nat. E	Head of Data	Data Management; Governance	NZ	National
EM. NZ. Reg. A	Director	Emergency Management	NZ	Regional
EM. NZ. Reg. B	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. C	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. D	Principal Science Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. E	Principal Advisor Strategy and Partnerships	Emergency Management	NZ	Regional
EM. NZ. Reg. F	GIS Lead	Emergency Management; Data Management	NZ	Regional
EM. Gov. NZ. Nat. G	Senior Hazard Risk Management Advisor	Emergency Management; Governance	NZ	National
EM. NZ. Reg. H	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Nat. I	First Responder	Emergency Management	NZ	National
EM. Gov. NZ. Nat. J	National Operations Manager	Emergency Management; Governance	NZ	National
EM. NZ. Reg. K	Regional Manager	Emergency Management	NZ	Regional
EM. NZ. Reg. L	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. M	Group Controller	Emergency Management	NZ	Regional
Health NZ. Reg. A	Respiratory Doctor	Public Health	NZ	Regional
Hyd. Gov. NZ. Reg. A	Flood EWS Programme manager	Hydrology; Governance	NZ	Regional

Interview Code	Position	Classification	Location	Government Level
Lifelines NZ. Reg. A	Civil Engineer	Lifelines	NZ	Regional
Loss Modelling Research NZ. A	Economist	Loss Modelling; Research	NZ	Private
Met. Int. A	Science Manager	Meteorology	International	National
Met. Int. B	National Manager Disaster Mitigation	Meteorology	International	National
	Policy			
Met. Int. C	Senior Policy Officer	Meteorology	International	National
Met. Int. D	Senior Social Scientist	Meteorology	International	National
Met. Int. E	Consultant Meteorologist	Meteorology	International	National
Met. NZ. F	Senior Meteorologist	Meteorology	NZ	National
Met. NZ. G	Communications	Meteorology	NZ	National
Met. NZ. H	Public Relations	Meteorology	NZ	National
Met. Int. I	Division Chief/Meteorologist	Meteorology	International	National
Met. Research NZ. J	Meteorologist	Meteorology; Research	NZ	National
Met. NZ. K	Senior Meteorologist	Meteorology	NZ	National
Met. Private NZ. L	Head Weather Analyst	Meteorology	NZ	Private
Risk Modelling NZ. A	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. B	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. C	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. D	Risk Modeller	Risk Modelling	NZ	National

Per ES-GT, analysis of the interviews, workshops, and documents followed the open coding (assigning concepts and categories to an instance of the data by line or by word), axial coding (relating categories to each other with option guidance from the coding paradigm), and selective coding (relating all categories to a core category) stages (Strauss & Corbin, 1990). The coding paradigm introduced by Strauss and Corbin (1990) supported the axial coding stage whereby the codes created from the open coding stage were related to the coding paradigm dimensions (Table 8.2) for increased density and precision.

Additional techniques were used to support the ES-GT analysis, including regular memo-writing, diagramming, and constant comparison (Glaser & Strauss, 1967). When the lead author noticed common themes in the interview and workshop data, a memo was written to identify the theme and discuss its relation to other concepts or themes (i.e. constant comparison) (Corbin & Strauss, 2015; Glaser & Strauss, 1967). Diagramming was used to draw out linkages or relationships between the emerging themes. This was an iterative process that occurred during the data collection and analysis.

Coding Paradigm Dimension	Description
Causal Conditions	A set of events that influence the phenomena or result in the appearance or development of a phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Phenomena	The subject or object under study (Strauss & Corbin, 1990).
Contextual Conditions	The specific set of conditions and characteristics surrounding the phenomena and resulting in action/interaction strategies taken to address the phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Intervening Conditions	Unexpected events or factors leading to action/interaction strategies (e.g., time, space, culture, socioeconomic status, technological status, history) (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Action/Interaction Strategies	Purposeful and deliberate acts taken to address the phenomena (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).
Consequences	Predictable or unpredictable, intended or unintended outcomes of the action/interaction strategies (Strauss & Corbin, 1990; Vollstedt & Rezat, 2019).

Table 8.2. Coding paradigm summary (from Harrison et al., 2022).

From the axial coding process (summarised in Table 8.2), two phenomena related to data management and acquisition of hazard, impact, vulnerability, and exposure data were identified. Phenomena are the subjects or objects under study (Strauss & Corbin, 1990). Impact forecasts and warnings and HIVE data were identified as the overarching phenomena being studied in this research. As data collection and analysis progressed, the two phenomena that became the focal point of this manuscript were identified from the themes that emerged in the interviews, memowriting, diagramming, and constant comparison techniques previously described.

These two phenomena are: (1) The Roles and Responsibilities of Data Custodianship, and (2) Data Access and Sharing. These are discussed next.

8.4 Findings and Discussion

Through ES-GT two key themes (or phenomena) related to data management and acquisition for hazard, impact, vulnerability, and exposure data were identified: (1) The Roles and Responsibilities of Data Custodianship, and (2) Data Access and Sharing. These themes were succinctly summarised by an NZ-based risk modeller, who said:

I think it really comes down to ... sharing and collaboration ... and I think there are good efforts, but ... often things get snagged in ... a privacy or confidentiality or legal issues with the datasets, who owns them, who maintains them, how can you rely on them ... And the amount of work that is required to produce and maintain a reliable dataset is massive, it's just so much work that people don't want to do it. It costs money, it costs time, and ... being a custodian of a dataset is not really an enviable position, necessarily. I think that's probably one of the biggest barriers (Risk Modelling NZ. A). These two phenomena (the roles and responsibilities of data custodianship, and data access and sharing) will each be discussed in turn.

8.4.1 The Roles and Responsibilities of Data Custodianship

Amongst participating agencies and countries, it remains unclear as to who is responsible for maintaining datasets for IFWs:

... it's a little bit about our traditional remits just being on the hazards information and so insurance companies and other areas are associated with damage and loss. And now ... the challenge is accessibility to the data. So, we don't necessarily need to or want to become custodians of new data sets, we're just keen, if someone's got it organised and has it, to be able to ... bring it together with our hazard information and our forecasting capability (Met. Int. B).

The responsibility and cost of collecting and storing HIVE data were concerns for participants. Many of the agencies' remits do not include data custodianship, and it would be a costly undertaking, with the uncertainty of maintaining funding (Met. Int. B; Met. NZ. F).

While the literature has pointed to EM agencies for collecting impact information (e.g., Kox, Lüder, et al., 2018; Potter et al., 2021), we found that the participating NZ-based CDEM Groups indeed collect impact information, but they often do not systematically collect it or store it (EM. NZ. Reg. A, B). These findings corroborate those of Crawford et al. (2018), who found that EM agencies and councils in NZ were not clear on who was responsible for collecting risk data. For example, regarding the NZ National Loss Database under development by the National Emergency Management Agency in NZ (NEMA), our participant indicated that NEMA is not a designated data custodian in NZ, and as such, they have been grappling with learning proper data management protocols (EM. Gov. NZ. Nat. G). This raises

questions around who is most suitable for managing and acting as the steward and/or custodian of the data.

Several data custodians were identified in passing during our interviews, which align with some that have been identified by LINZ (2014a), such as LINZ, the NZTA, and local councils. For example, a risk modeller who was involved in collecting building asset information for the 2011 Canterbury earthquake recovery described how maintenance responsibilities for these data were transferred to LINZ due to privacy concerns:

because there's some personal information in the database ... that we didn't pay attention [to] previously ... but to protect us from now on, we say 'okay this is the data we give [to] LINZ, you guys take the data ... and you ... decide if you want to share the data or not' (Risk Modelling NZ. B).

Similarly, for the development and management of the NZ National Loss Database, NEMA has turned to Statistics New Zealand (typically referred to as Stats NZ, New Zealand's national statistics agency) for guidance on proper data management protocols:

Stats NZ ... have helped us think about this, but it's their business. So, for example, we talked about data management, and they talked about, 'okay if you're holding that data then you need to run an integrity test, that your data hasn't become corrupted, and you do that every [so often]. And you've got enough backups and all that ... but we haven't got an explicit data management policy around ensuring it's not corrupt, its integrity ... change controls, and all that tracking; that's not what we do (EM. Gov. NZ. Nat. G).

In addition to being the national statistics agency for NZ, Stats NZ also became the lead agency for government-held data in 2017 (Stats NZ, 2019). In this leadership role Stats NZ acts as a facilitator to support government agencies in building their capabilities and data management practices (Stats NZ, 2019). Thus, their help in guiding NEMA towards proper data management practices for the National Loss Database aligns with their role as the lead for government data. The Stats NZ website³² provides further guidance on principles for safe and effective use of data and analytics, data stewardship, data standards, open data, etc.

Several of the HIVE datasets that were identified by Harrison et al. (2021) have been classed as fundamental geospatial datasets by LINZ and have an appointed, proposed, or suggested data custodian and steward. Table 8.3 presents the Suggested (S), Proposed (P), or Appointed (A) Stewards and Custodians of the HIVE data sets identified by Harrison et al. (2021) from our interviews as designated by LINZ (2014a) The results in Table 8.3 are based on an analysis of two LINZ documents regarding data custodianship (see LINZ, 2014a, 2014b).

³² <u>https://www.stats.govt.nz/about-us/data-leadership</u>. Accessed 7 July 2021.

Table 8.3. Suggested (S), Proposed (P), or Appointed (A) Stewards and Custodians of the HIVE data sets identified by Harrison et al. (2021) from our interviews as designated by LINZ (2014a)³³. N/A is used for datasets that were not proposed as fundamental datasets (note that any person or agency can propose a fundamental geospatial dataset (LINZ, 2014a)), and thus a steward and custodian was not identified. The number of N/A entries demonstrates a lack of direction for managing impact data.

	Dataset	Status	Steward	Leadership Custodian	Delivery Custodian
	Weather stations (rain gauges,	N/A	N/A	N/A	N/A
	Radar data	N/A	N/A	N/A	N/A
	Satellite imagery & observations	Fundamental, Not evaluated	LINZ (A)	LINZ (S), NZDF (S), TAs (S), Police (S)	LINZ, NZDF, TAs, Police, MfE, Landcare (S)
	River height and flow gauges	Not evaluated	Steward Committee (S) or MPI (S), MfE (S)	MfE (S)	RCs (S)
	Float gauges/Sea level data	N/A	N/A	N/A	N/A
ta	River networks	Fundamental	Steward Committee (S) or MPI (S), MfE (S)	MfE (S), LINZ (S)	NIWA (S)
Da	Vertical Rain Radar	N/A	N/A	N/A	N/A
azard	Regionwide floodplains	Not evaluated	Steward Committee (S) or MPI (S), MfE (S)	MfE (S), LINZ (S)	NIWA (S), GNS (S), RCs (S), TAs (S)
Ĥ	Overland flow paths	N/A	N/A	N/A	N/A
	Coastal inundation Maps / Hazards	Not evaluated	Steward Committee (S) or MPI (S), DoC (S)	MfE (S)	RCs (S), GNS (S)
	Pollen Counts	N/A	N/A	N/A	N/A
	Slope	N/A	N/A	N/A	N/A
	Camera feeds	N/A	N/A	N/A	N/A
	Social media (Tweets, Facebook post comments)	N/A	N/A	N/A	N/A
	Crowdsourcing (e.g., volunteer rain gauges, NZ Flood Pics)	N/A	N/A	N/A	N/A

³³ The acronyms in Table 8.3 are as follows: Land Information New Zealand (LINZ), New Zealand Defence Force (NZDF), Territorial Authorities (TAs), Ministry for the Environment (MfE), National Institute of Water and Atmospheric Research Ltd (NIWA), Ministry of Primary Industries (MPI), Institute of Geological and Nuclear Sciences Limited (GNS Science), Regional Councils (RCs), Department of Conservation (DoC), Ministry of Business, Innovation and Employment (MBIE), Local Government New Zealand (LGNZ), Statistics New Zealand (Stats NZ), New Zealand Transport Agency (NZTA), National Emergency Management Agency (NEMA, formerly MCDEM), Local Government Geospatial Alliance (LGGA).

	Dataset	Status	Steward	Leadership Custodian	Delivery Custodian
	Social media (Tweets, Facebook post comments, SnapChat)	N/A	N/A	N/A	N/A
	Crowdsourced Photos via Story Maps (e.g., NZ Flood Pics)	N/A	N/A	N/A	N/A
	Crowdsourcing / Public Reporting	N/A	N/A	N/A	N/A
	Red Cross Chained Crowdsourcing	N/A	N/A	N/A	N/A
	Emergency call centre reports	N/A	N/A	N/A	N/A
	Community volunteer radio calls	N/A	N/A	N/A	N/A
	Damage surveys	N/A	N/A	N/A	N/A
	Media reports	N/A	N/A	N/A	N/A
ata	Tacit knowledge, experience, intuition	N/A	N/A	N/A	N/A
t D	Health & Injury Data	N/A	N/A	N/A	N/A
Dac	Wellbeing Surveys	N/A	N/A	N/A	N/A
duŋ	Post-event interviews and surveys	N/A	N/A	N/A	N/A
	Insurance claims	N/A	N/A	N/A	N/A
	"Boots on the ground"	N/A	N/A	N/A	N/A
	Lifelines Sectors (e.g., power companies, NZTA)	N/A	N/A	N/A	N/A
	Situational Reports	N/A	N/A	N/A	N/A
	Post-event reports	N/A	N/A	N/A	N/A
	Operations Reports (Council)	N/A	N/A	N/A	N/A
	Flood event reporting	N/A	N/A	N/A	N/A
	Injuries and fatalities (e.g., cause of death)	N/A	N/A	N/A	N/A
	Cultural and heritage / historical impacts	N/A	N/A	N/A	N/A
	Impact Model outputs	N/A	N/A	N/A	N/A
	NZ National Loss Database	N/A	N/A	N/A	N/A

	Dataset	Status	Steward	Leadership Custodian	Delivery Custodian
Vulnerability Data	Vulnerability Assessment and Risk Modelling Outputs	N/A	N/A	N/A	N/A
	Asset information - Building footprints	Fundamental	LINZ (A)	MBIE (S)	TAs (S), NZFS (S), MBIE (S)
	Asset information - Historically and Culturally Significant Sites	Not evaluated	Steward Committee (S) consisting of LINZ (S), LGNZ (S), Emergency Services (S)	LINZ (S), LGNZ (S)	LINZ (S), TAs (S), Private Sector (S)
	Building damage assessments		N/A		
	Census data - meshblocks	Fundamental	Stats NZ	Stats NZ	Stats NZ
	Tacit knowledge, experience, intuition	N/A	N/A	N/A	N/A
	Lab-based experiments	N/A	N/A	N/A	N/A
	Health Data, New Zealand Health Survey	N/A	N/A	N/A	N/A
	Soil/land stability	Fundamental	No Steward	LINZ (S), GNS (S), Landcare (S)	LINZ (S), GNS (S), Landcare (S)
	Infrastructure - NZ Road Network	Not evaluated	Ministry of Transport	NZTA (P)	NZTA (P)
	Infrastructure - Council roads	Not evaluated	Ministry of Transport	LGNZ (S)	TAs (S)
	Infrastructure - Water	Not evaluated	Steward Committee (S) or MPI (S), MfE (S)	MfE (S), LINZ (S)	NIWA (S), LINZ (S), RCs (S), TAs (S)

	Dataset	Status	Steward	Leadership Custodian	Delivery Custodian
Exposure Data	Asset information - Building footprints	Fundamental	LINZ (A)	MBIE (S)	TAs (S), NZFS (S), MBIE (S)
	Asset information - Historically and Culturally Significant Sites	Not evaluated	Steward Committee (S) consisting of LINZ (S), LGNZ (S), Emergency Services (S)	LINZ (S), LGNZ (S)	LINZ (S), TAs (S), Private Sector (S)
	Infrastructure - NZ Road Network	Not evaluated	Ministry of Transport	NZTA (P)	NZTA (P)
	Infrastructure - Council roads	Fundamental	Ministry of Transport	LGNZ (S)	TAs (S)
	Infrastructure - Power - Electricity	Not evaluated	No Steward	Transpower (S)	Transpower, Lines Companies
	Infrastructure - Utility Networks	Not evaluated	No Steward	MCDEM/NEMA (S)	Utility Companies (S)
	Infrastructure - Water	Not evaluated	Steward Committee (S) or MPI (S), MfE (S)	MfE (S), LINZ (S)	NIWA (S), LINZ (S), RCs (S), TAs (S)
	Census data - meshblocks	Fundamental	Stats NZ	Stats NZ	Stats NZ
	Population movement via cell phone data	N/A	N/A	N/A	N/A
	Tacit knowledge, experience, intuition	N/A	N/A	N/A	N/A
	Topographical data, e.g. digital elevation models	Not evaluated	No Steward	MfE (S), LINZ (S)	Landcare (S), LINZ (S)
	Land-use and planning zones	Not evaluated	No Steward	LGGA (S), LGNZ (S)	TAs (S), RCs (S)
	Land-use maps	Fundamental	No Steward	MfE (S)	MfE (S)
	Community events	N/A	N/A	N/A	N/A

The results in Table 8.3 show that stewardship/custodianship for most of the underlying datasets for hazards, vulnerability, and exposure have been identified by LINZ and are either Suggested (S), Proposed (P), or Appointed (A), while there is a clear a gap in the stewardship/custodianship of impact data, as shown by the number of 'N/A' entries for these datasets. This, in conjunction with our interviews, shows the need for establishing data management protocols and practices for impact data, including identifying potential data stewards and custodians for these data, such that the data can be accessed by and shared with other relevant users.

8.4.2 Data Access and Sharing

Data access and sharing was the next key theme or phenomena identified in our data analysis. The ES-GT coding paradigm was applied to understand the causal and intervening conditions, and action/interaction strategies for data access and sharing, as shown in Figure 8.1.



Figure 8.1. Summary of findings relating to the causal conditions, intervening conditions, and action/interaction strategies relating to data access and sharing, identified using the ES-GTM coding paradigm. Blue boxes represent organisation aspects and green boxes represent technical aspects.
As discussed by Harrison et al. (2021)³⁴, disaster/emergency events, technological advancements, and research were identified as the causal conditions to data collection and data access and sharing. For example, sharing data was required for the response to the 2019 Pigeon Valley fires in Nelson, NZ, yet organisational and technical issues impeded the data sharing process (Data Management Gov. NZ. Nat. A). However, as shown in Figure 8.1, intervening conditions can inhibit data sharing and access while several action/interaction strategies have been identified as capable of addressing these intervening conditions. These conditions and strategies will be discussed next.

8.4.2.1 Organisational Aspects of Data Access and Sharing

Trust and distrust, and privacy and security were identified as organisational intervening conditions affecting data sharing. Building partnerships was identified as an organisational action/interaction strategy to address these intervening conditions. These will be discussed next.

Trust and Distrust as Intervening Conditions

Trust is a key element to data use and access. While Harrison et al. (2021) identified the importance of trust in the data for IFWs, here we found the need for trust between agencies to be an important condition for data access and sharing, as summarised by a NZ risk and data specialist:

There's a lot of suspicion between institutions in New Zealand as to what people are doing things for and why and with this comes patch protection. This is an important point and that it inhibits risk awareness, collaboration and our risk management as a country (Data Management Private NZ. D).

The management of the NZ Flood Pics³⁵ crowdsourcing platform and access to its resulting data is an example of an initiative established to avoid suspicion and lack of institutional trust, according to one of our interviewees (Data Management Private NZ. D). They described that NZ Flood Pics was developed independently from any institutions, and there is an aversion to tying the platform to any single institution even though that would make available the resources needed to sustain the platform. This aversion stems from the need for the data to remain open and separate from any "ulterior motives" (Data Management Private NZ. D). They indicated that institutionalising the platform may give the impression that there is an agenda with the use of the data, which could make people wary to contribute to it. At the time of conducting this interview, the costs and resources needed to sustain the platform

³⁴ The results of Harrison et al. (2021) are presented in a similar fashion using the ES-GTM Coding Paradigm to analyse the causal conditions, intervening conditions, and action/interaction strategies pertaining to the Data Collection phenomenon. As such, a similar figure is presented here, but pertaining instead to the Data Access and Sharing phenomenon.

³⁵ <u>www.nzfloodpics.co.nz</u>. Accessed 7 July 2021.

remained such that the participant was seeking collaborating support and buy-in from multiple sectors with the requirement of keeping the data openly accessible. An update has since been provided by this participant that NZ Flood Pics is now in the process of being institutionalised with NIWA with an agreed set of principles for how the data will remain open and accessible for "anybody to add value" (Data Management Private NZ. D).

Institutional involvement and open access to and sharing of citizen science data are growing areas of interest in citizen science research. The type of organisation involved in citizen science projects appears to influence people's willingness to participate and contribute their data (Anhalt-Depies et al., 2019). For example, Martin et al. (2016) found that contributors to a marine citizen science project in Australia showed a very high willingness to share data with research organisations, but less so with private research companies or consultants. Additionally, contributors seem to care about how the data they contribute is shared: Ganzevoort et al. (2017) found that surveyed citizen scientists do not support unconditional data sharing, rather their acceptability of sharing the data with third parties depends on the goals of the data user. Moreover, Groom et al. (2017) argue that the motivations of citizen scientists to contribute their data should align with the accessibility of the data for other uses. Thus, it is important to invest in data policies and transparency efforts to protect the interests of the contributors and ensure their continued engagement (Anhalt-Depies et al., 2019).

Participants indicated that some agencies who collect impact data were also found to be averse to sharing the data due to its sensitive nature and distrust in how such data could be used (Met. Int. D; Risk Modelling NZ. A; Loss Modelling Research NZ. A). As one loss modelling participant from NZ outlined, not all agencies are willing to publicly share their data, such as a public insurance group that formed after the 2010-2011 Canterbury earthquake sequence (Loss Modelling Research NZ. A). As such, there is no guarantee that government/public organisations will want to, or can, make any of their data publicly available (Loss Modelling Research NZ. A).

This could be due to the sensitive nature of the data (Met. Int. D; Risk Modelling NZ. A). Certainly, in NZ there are concerns around how releasing impact information could influence property values (Risk Modelling NZ. A). In Argentina, our participant described how a provincial government would not share their impact data with the National Meteorological Service (NMS) because of the political nature of the data:

I was working with the province of [redacted] ... and we implement[ed] this [data collection] form as a final project, and when I [said] 'okay, I want the data' they [told] me 'okay, we will give you the infrastructure data but no data about deaths' ... And I was like 'hey, no. That's not fair.' Why? Because it's political data! Who live and who die during a storm: that's political data, because it has to do with vulnerability and people here [are] very vulnerable. That's why I say that impact data is political data (Met. Int. D).

This finding aligns with those from Harrison and Johnson (2019), where Canadian emergency managers showed concern about displaying heavily damaged areas online via crowdsourcing platforms for emergency response. However, as shown in the above quote, the political nature by which the impact data in Argentina is perceived adds another layer beyond simply showing concern for the privacy and security of people and assets during an emergency. Viewing impact data as a threat to how government response agencies and their response capabilities are perceived indeed introduces a political element that inhibits the global movement for open and collaborative data sharing for improved DRR.

The consequences of restrictive data access and lack of information sharing for DRR can lead to disastrous consequences, as was seen in the USA following the landfall of Hurricane Katrina in 2005 (Peled, 2011). Yet, our findings show that data creators and users remain reluctant to share critical information. In addition to the concerns around the political nature of the data as described above, this aversion to sharing may be due to inter-agency competition over resources, influence, and autonomy (Peled, 2011), and/or loss of control over the datasets (Barry & Bannister, 2014).

Trust is an essential factor in multi-agency collaboration and data sharing, especially for disaster response (Doyle & Paton, 2017). Our findings further illustrate the importance of trust to facilitate data access and sharing through interagency collaboration for DRR (Kapucu, 2006). The importance and value of opening and sharing disaster-related data cannot be denied, and it is possible to establish open data access and sharing for DRR. For example, local, provincial, and national agencies, and regional and national academic and research institutions in Argentina worked together to establish an open data platform for flood impact reduction, "with a view to socializing knowledge for early alerts" (De Giusti et al., 2016, p. 86). Thus, agencies must continue to build and nurture trust between each other to facilitate data and information sharing.

Traditional media outlets may be one party not tied to the political influence and privacy and security concerns of sharing data and play a major role in disseminating information during a disaster (Greenberg & Scanlon, 2016; Nair, 2010). For example, in Aotearoa-NZ, the MetService usually has "a conversation with media" about their official warning message and provides examples of impacts along with the warning message to add meaning and context to the warning. The media then passes on this information to the public (Met. NZ. G, K).

Media outlets and reports can be a timely data stream for information during a disaster (Nair, 2010). They can even provide new information that might not otherwise be picked up or observed elsewhere. For example,

there might be a report of ... a motorcyclist blown off a piece of road or something. But there might not be any weather observations for miles in any direction that support that. So, then you have to go and look for corroborating evidence somewhere else, and how much do you trust the source? (Met. NZ. K)

However, as this example suggests, while media reports can bring new information to light, the new information still needs to be verified, as it might not always be accurate or true. Media outlets are both heavily reliant and influential on perceptions of trust and distrust during disasters (Miles & Morse, 2007). In another example, a regional EM agency in Aotearoa-NZ received crowdsourced reports on their public online story map of a van "stuck in the centre of the Hokitika River" (EM. NZ. Reg. F). The media picked up this report and publicised that "there's people trapped" (EM. NZ. Reg. F). The van was empty, and the EM agency had to re-upload the report themselves to include a note that the van was empty (EM. NZ. Reg. F).

The literature indicates that agencies elsewhere have reported difficulties with communicating with media outlets e (Anzur, 2000; Nair, 2010). In Aotearoa-NZ, the continuous conversations between the NZ MetService and the media described above exemplify a working relationship that ensures a unified message for informing people. In another example, the media provided "quite active and quite positive support" in the response and evacuation of a large-scale flood in Edgecumbe, Bay of Plenty by providing ample media coverage of the event and of the messages issued by the EM agency (EM. NZ. Reg. A). For media outlets and their resulting reports to be effectively used by their audiences, it is thus critical for them to build trusted relationships with those audiences, including members of the public, hydrometeorological services, and EM agencies. This could include developing a process or protocols for sharing data and information effectively.

Building Partnerships as an Action/Interaction Strategy

While NMSs are not typically responsible for collecting non-meteorological data, EM and flood management agencies, amongst others, collect or produce various types of impact data for their own purposes. Findings from the interviews, in support of existing literature (e.g., Hemingway & Gunawan, 2018; WMO, 2015), suggest that either formal or informal partnerships support data sharing for DRR. Continued collaboration to build and nurture partnerships has a positive relationship with trust (Kapucu, 2006), an important factor for sharing data. The NHP is an example of a formal partnership developed in the UK between public agencies to improve disaster management across the country (Hemingway & Gunawan, 2018). This partnership allowed the UK MetOffice to access useful datasets such as traffic count data from the transportation agency, for their impact models (Met. Int. A; Harrison et al., 2022).

In New Zealand and other participating countries, no such formal partnership was found to exist that is comparable to the NHP in the UK. However, evidence of informal partnerships was found in NZ and Australia. In both NZ and Australia, NMS and EM officials frequently contact each other informally during an event to exchange more targeted forecasting information such as levels of uncertainty, and worst-case scenarios to help with planning responses (Met. Int. B; Met. NZ. F, K; EM. NZ. Reg. A, B, C, D, E, F; Harrison et al., 2022). For example, as reported by Harrison et al. (2022), the NZ MetService works together with local and regional EM groups and hydrologists to determine the most appropriate warning level. These informal partnerships are

particularly useful for information-sharing and decision-making on the fly during an event.

A need remains for more formal partnerships to facilitate data sharing after response events (Risk Modelling NZ. A). However, progress in this space will be slow:

there's lots of talk about data sharing and I think there's a general attitude in New Zealand moving towards greater collaboration, greater sharing. [But] a sort of central repository for any of this stuff within the next decade is absolutely not going to happen, or if ever ... It's hard to get people to work together (Risk Modelling NZ. A).

The participant listed data ownership and proprietary licensing as key limitations to sharing data. Perhaps in the context of sharing HIVE data via a central repository as mentioned by the risk modeller (Risk Modelling NZ. A), progress will indeed be slow as this requires more in-depth understanding and legal groundwork to establish data ownership rights and protection of proprietary data (e.g., Clark & Guiffault, 2018; Risk Modelling NZ. A). However, like the informal partnerships between the MetService and EM groups and council hydrologists to share hydrometeorological information for warnings, another informal partnership has been established to fill a gap in EM practice in NZ, the NZ Geographic Information Systems for Emergency Management (NZGIS4EM) group.

The NZGIS4EM group formed in the mid-2010s to boost the use of geospatial tools, such as Geographic Information Systems (GIS) within the EM sector (EM. NZ. Reg. H; Harrison et al., 2021). NZGIS4EM is a grassroots community of GIS specialists and EM practitioners in NZ that work together to share data and tools during responses and to innovate the use of geospatial technologies for EM (EM. NZ. Reg. H; Data Management Private NZ. B; Harrison et al., 2021). The community was formed from the leadership of one skilful individual who identified the need for geospatial innovation in the NZ EM sector, to enhance practices from relying on paper maps to using modern tools and technology (Data Management Private NZ. B).

The NZGIS4EM is credited with fostering a community of practice within the EM and GIS sectors, where relationships across agencies have been built and strengthened for a more coordinated disaster response effort (EM. NZ. Reg. H; Data Management Private NZ. B). In turn, this has driven innovation, knowledge sharing, and data sharing across NZ (EM. NZ. Reg. H; Data Management Private NZ. B). Direct outcomes of this group's work are seen in the recent *Director's Guideline for Civil Defence Emergency Management Groups [DGL 22/20]* (NEMA, 2020a) and *Technical Standard [TS05/20] National Impact Assessments Data Set and Dictionary* (NEMA, 2020c), published by NEMA. In these documents, the use of GIS is heavily referenced, and NZGIS4EM aided in defining the standards presented in the *Technical Standard [TS05/20] National Impact Assessments Data Set and Dictionary* (NEMA, 2020c; Data Management Private NZ. B).

Our findings suggest that informal partnerships are developed to fulfil an immediate need, such as communicating hydrometeorological information for warning decision-making or building a community of practice for coordinated disaster response. This finding aligns with the ad hoc informal partnerships that formed in the response and recovery to the 2010-2011 Queensland floods, where such partnerships formed between public and private agencies for information sharing (Bajracharya & Hastings, 2015). Furthermore, interorganisational networks facilitated by groups like the NZGIS4EM can be voluntary or mandated (Kapucu & Demiroz, 2017); in the case of the NZGIS4EM group, it is a voluntary effort. Success factors of such networks have been identified as effective communication, trust and social capital, and learning and adaptation (Kapucu & Demiroz, 2017).

Formal partnerships appear to be less common, and not as easy to establish, as indicated previously (Risk Modelling NZ. A), particularly for severe weather hazards. However, formal partnerships have been developed in NZ in the form of science advisory panels for natural hazard perils, such as the New Zealand Volcanic Science Advisory Panel (NEMA, 2020b), and the New Zealand Tsunami Advisory Group. The New Zealand Volcanic Science Advisory Panel works with area-specific volcanic advisory groups to coordinate planning for volcanic activity (NEMA, 2020b). These groups were formed from a specific need that was either identified from local/national or international events (EM. Gov. NZ. Nat. J), to better communicate science advice by bringing the expertise from various scientific agencies together (Doyle et al., 2014). For example, the New Zealand Tsunami Advisory Group was established in response to the 2011 Tohoku and 2004 Indian Ocean earthquakes and tsunami (EM. Gov. NZ. Nat. J). These groups differ from the UK NHP in that practitioners are not formally included in them and some of the groups' roles in response and providing formal advice are unclear. Partnerships and multi-agency collaboration are not new concepts in the EM literature (e.g., Owen et al., 2013). It is promising to see these growing examples of successful partnerships both within and outside of NZ, be it formal and informal. However, more clarity is needed to establish the roles of these NZ-based groups in response and formal advice.

Strategies for building and nurturing partnerships and collaboration include networking (EM. Gov. NZ. Nat. J), professional development (EM. NZ. Reg. K; Doyle & Paton, 2017), cohabitation (Hemingway & Gunawan, 2018), and multi-disciplinary collaboration (Risk Modelling NZ. C; Ge et al., 2021), which may or may not be supported by funding and top-down mandates (EM. Gov. NZ. Nat. J; Peek et al., 2021). Participants indicated that a mix of both bottom-up and top-down approaches to building and nurturing partnerships is applicable in the Aotearoa-New Zealand context (EM. Gov. NZ. Nat. J). For example, cohabitation between agencies like the MetService, CDEM Groups, and local/regional councils can be organised from the bottom-up between the agencies in question. However, some political direction such as a national mandate might be more effective at inciting nation-wide cohabitation practices (EM. Gov. NZ. Nat. J). Further exploration of these strategies will be explored in future research.

Privacy and Security as an Intervening Condition

When it comes to sharing data for emergency response purposes, our participant from Fire and Emergency New Zealand (FENZ) identified the importance of protecting the response data. Much of their response data comes from NZ Police, who have very structured and careful security systems protecting their information (EM. NZ. Nat. I). As such, parties are working with the Privacy Commissioner to show that they can ensure the appropriate controls to protect the information, and in turn, there are also cultural challenges around the views of sharing information (EM. NZ. Nat. I). Our FENZ participant indicated that it is a matter of knowing staff members' behaviours and educating staff on why it is important to keep the data safe, while also safely sharing the information for effective response.

Safe and secure mechanisms for sharing the data are needed but are not always feasible. For example, our participant from Argentina indicated that they set up a Google Form to facilitate standardised data collection and sharing between the NMS and local government agencies, but they "have concerns ... because ... it is sensitive data [and] I don't trust ... sending information through Google" (Met. Int. D). However, Google Forms was the most easily accessible tool that the local government agencies could handle (Met. Int. D). This demonstrates the need to have controls in place to maintain privacy and ensure data security.

8.4.2.2 Technical Aspects of Data Access and Sharing

Challenges around data integration were identified as a technical intervening condition affecting data sharing. Systematic Data Collection and Standardised Data Collection were identified as technical action/interaction strategies to address this intervening condition. These will be discussed next.

Data Integration as an Intervening Condition

Sharing data is only successful when the data can be integrated seamlessly into existing systems and practices (Migliorini et al., 2019). Findings from Harrison et al. (2021) show that a wide variety of HIVE data for severe weather events exists. In many cases, the challenge lies not in identifying the data sources but in integrating all sources and types of data in a meaningful way (NMS. Int. A, E; NMS. NZ; EM. NZ Regional. A, B, E). This is summarised by a NZ risk modeller, "although we all complain about a lack of data, a lot of the time, there is quite a bit of data out there that could be used and isn't being used very well" (Risk Modelling NZ. A). For example, this participant identified struggles with integrating building data, which is typically stored as spatial point data, with meshblock data, which is stored as spatial polygons. They stated that matching these point and polygon data together

can be the bigger barrier than the fact that the data does or doesn't exist. Because ... it turns out it's really not easy to put those points in a polygon because ... there's no basic way of doing that, it's complicated (Risk Modelling NZ. A).

Furthermore, our UK-based participant described how they would like to see

an integration of different approaches [like] the use of satellite, the use of social media, the use of citizen science, the use of media, and the integration of media to help understand better the differences and the causes for different types of impact (NMS. Int. A).

Some data integration methods have been successfully implemented. For example, the NZ MetService receive council rainfall data and share warning data files to integrate into hydrological models. In NZ, the West Coast CDEM Group primarily works online so they can share information in near-real-time with other agencies. But this is not standard practice across the country; gaps still exist in developing formal systems for data sharing and integration. A regional EM official elaborated on their challenges of linking up and integrating various data sources:

I would like to see ... some research in understanding what are the tools and systems that can be used for this? Because on one hand ... if you sit around and everyone goes 'yeah, ... we must share data' and ... if we're going to plan for the future and everyone, different agencies hold different data about the same event, but how we're sharing that data? So, what is the system to do that? (EM. NZ Regional. A).

The Common Operating Picture (COP) is another example of a solution currently under development in NZ to allow cross-agency intelligence and data sharing during a response. The COP, a recommendation made following the 2010-2011 Canterbury earthquakes and other notable events (see Technical Advisory Group, 2017), is based on the idea of "everyone contributing to the scenario so that they've got to share, so that everyone's got access to that information" (EM. NZ. Reg. A); in other words, it is "a graphical visualisation of all the data available to make a decision" (Data Management Private NZ. B).

Phase one of the NZ COP programme began in late 2019 led by NEMA, when agencies involved identified the core national datasets needed for EM, and brainstormed how some of those datasets are shared, particularly geospatial datasets (EM. NZ. Reg. H). This required agencies to collaborate to identify different needs, highlighting the need for building trust and partnerships between agencies to facilitate data sharing.

Other solutions have been proposed and developed for improving data interoperability for integration. Horita et al. (2015) developed a spatial decision support system (DSS) to integrate official and unofficial data for flood risk management in Brazil, showing that integration "provides more complete, accurate and updated information about the situation in the affected areas" (Horita et al., 2015, p. 91).

Impact-based decision support systems (IDSS) offer a way for warning services to make effective decisions. An IDSS is "the provision of relevant information and interpretative services that enable partners to prepare for and respond, as planned, to extreme weather, water, and climate events for the protection of lives and livelihoods" (Uccellini & Ten Hoeve, 2019, p. 1928). In a case study comparison of two historic severe winter weather events in the USA with and without a formal IDSS, the IDSS was found to enable quicker and more complete forecast updates, such that emergency managers could relay information to appropriate agencies to take mitigative actions (Hosterman et al., 2019). The IDSS also allowed for improved crafting of public and partner messaging, and for preparing public officials to decide to shut down infrastructure.

Similarly, an integrated analysis of social vulnerability to extreme precipitation in Colorado was carried out by Wilhelmi and Morss (2013) using Geographic Information Systems (GIS). The process involved integrating radar-derived rainfall data and watershed boundaries with national census data and historical impact data. Challenges were faced with the differences in spatial-temporal scales between meteorological and social datasets, as well as the various formats in which the meteorological data were available (Wilhelmi & Morss, 2013).

Systematic Data Collection as an Action/Interaction Strategy

Systematic data collection is a need highlighted by many participants as an action/interaction strategy for better data sharing and integration. Non-systematic data collection occurs when data is collected in various formats and stored in different places (e.g., Massagrande, 1995). Thus, we refer to systematic data collection as having a system in place to process data from raw state to a usable format and storing the data in one place for easy access.

Our participants indicated that systematic data collection would support evidence-based planning and decision-making for warnings and planning (Met. Int. A; Met. NZ. F; EM. NZ. Reg. B). For example, our participating UK-based NMS official highlighted the need for systematic impact, exposure, and vulnerability data collection to be able to understand the causes of impacts with empirical evidence and to be able to compare different impact models (Met. Int. A). In NZ, a lack of systematic data collection severely impacted the organisation of intelligence and data gathered from event responses. For example, "back in 2011 [Canterbury earthquakes] ... the capture of the data was so unstructured" (Data Management Gov. NZ. Nat. A); photos of buildings and letterboxes were not associated with dates or addresses, nor were they associated with damage and impact assessment forms used to capture information about the person and/or building (Data Management Gov. NZ. Nat. A). Consequently, an individual was "employed ... for 18 months, just to take that photo and that building form and that welfare form and try to bring together a file on that family" (Data Management Gov. NZ. Nat. A). Had the data been collected in a more systematic way, "that information ... would have been more informative at the time of the response and recovery" (Data Management Gov. NZ. Nat. A). Furthermore, systematic data collection would allow the sector to learn from past events and

forecast potential future events by identifying impacts from events with similar meteorological signatures (Met. NZ. F; EM. NZ. Reg. B).

The COP previously mentioned provides a streamlined system for collecting, sharing, and managing impact data. However, much of this data does not have a shelf life beyond its initial purpose of building situational awareness for decision making during a response, because "from a Civil Defence perspective, whether it's local or regional, we don't need to keep the data, really. We'd probably keep it for two years and then say right, we're going to dispose of it because we don't need it" (EM. NZ. Reg. H). This also applies to welfare/wellbeing assessments, which are "very challenging, because there is a lot more private and personally identifiable information collected" (EM. NZ. Reg. H). As such, the CDEM Group will "probably … be even more aggressive about deleting that information once the event is over, to protect the people's privacy" (EM. NZ. Reg. H).

Standardised Data Collection as an Action/Interaction Strategy

Standardised collection of data was another need highlighted by many interviewees. Standardised data collection facilitates systematic data collection by providing a set of standards at which the raw data is collected such that it can be seamlessly integrated into a database with minimal processing (De Groeve, 2015; Ferrer et al., 2018). Standardised data collection would ensure all data is collected consistently for rigour and comparability (De Groeve, 2015; Ferrer et al., 2018). Our interviewees identified a need to be able to compare datasets and warnings across countries and regions, such as in Europe (Met. Int. E). Likewise, data collection practices within NZ, Argentina, and Australia differ regionally, making it difficult for regions to share data. In Australia, EM agencies collect post-event impact assessments, but the Australian NMS official indicated that this data lacks necessary fields for forecasting purposes (Met. Int. E). In Argentina, the NMS is working with EM agencies to build a process for collecting standardised impact data using Google Forms to benefit both agencies; the EM agencies will have trustworthy data, while the meteorological service will have data to support IFW implementation (Met. Int. D).

Our investigation into NZ practice showed that several types of standard data collection forms are available for different purposes, whereby the standards/forms were designed by different agencies for various uses. For example, in 2006, the Hawke's Bay Regional Council commissioned the development of *Templates for Consistent Hazard Event Reporting* (herein referred to as the *Template*) with a focus on capturing the impacts of the hazard for use by EM Groups for research, risk modelling, and developing a hazards and impacts database (EMS Limited, 2006). Standardised data collection provided from this template would better facilitate the organisation and storage of impact and hazard data for further use and analysis. However, it is unclear whether this form is used by CDEM Groups or councils across NZ, as no participants identified it as a resource that they use, yet it has potential for development going forward.

NEMA published an Impact Assessments Director's Guideline for Civil Defence Emergency Management Groups outlining the preparation requirements for conducting effective and efficient rapid impact assessments by response agencies to enable a coordinated approach across multiple agencies (see NEMA, 2020a). This document identifies the agencies responsible for conducting impact assessments for various purposes and at various phases. For example, CDEM Groups or local authorities are responsible for planning rapid impact assessments, while FENZ is "likely to be one of the first responding agencies to conduct a rapid impact assessment" (NEMA, 2020a, p. 15) due to their specialised capabilities for collecting and sharing field data (NEMA, 2020a). NEMA also provided forms for completing an initial situation overview, initial damage assessment, and impact report form. These assessments support response planning and coordination (EM. NZ. Reg. H). Furthermore, a companion document called the Technical Standard [TS05/20] National Impact Assessments Data Set and Dictionary (NEMA, 2020c) provides more technical information to support the consistent collection and recording of impact assessment data for easy cross-agency sharing and integration.

Our participants identified welfare needs assessments as another form of impact assessment (EM. Gov. NZ. Nat. G; EM. NZ. Reg. H; Data Management Gov. NZ. Nat. A; Data Management Private NZ. B), focusing on "understanding the needs of people affected by an emergency" (MCDEM, 2015, p. 1). Guidelines have been published outlining the welfare needs assessment process (see MCDEM, 2015). These guidelines indicate that data is collected using "customised CDEM Group/local authority forms" (MCDEM, 2015, p. 11).

For the built environment, the Ministry for Business, Innovation and Employment (MBIE) published field guides, assessment forms, and other learning resources for rapid building damage assessments for flooding, earthquakes, and geotechnical hazards such as landslips (see MBIE, 2018b). These resources are an outcome of lessons learned from the 2010-2011 Canterbury earthquake sequence, the 2011 Nelson storm, and the 2016 Hurunui/Kaikoūra earthquake, and other international events (MBIE, 2018a). These assessments are primarily used by councils via contracted building engineers to assess the building and land safety and usability immediately after an earthquake or flood (Risk Modelling NZ. B; EM. NZ. Reg. H; MBIE, 2018b), but are not suitable for impact/risk modelling because they are not designed for this purpose and are not sufficiently comprehensive (Risk Modelling NZ. B).

Alternatively, our risk modelling participant indicated that developing a "master building database" containing building attribute data pre- and post-disaster would be "ideal" (Risk Modelling NZ. B). The participant envisions the database holding building data for all purposes, populated before an event, and updated with damage information to the respective buildings using a unique identifier post-disaster (Risk Modelling NZ. B). Efforts on this project are currently underway at GNS Science in NZ (see Lin et al., 2020).

The above assessments can be completed on paper, as has traditionally been done. However, a recent technological overhaul of the NZ CDEM sector (Data Management Private NZ. B) has introduced new tools for streamlining the collection and integration of these datasets (EM. NZ. Reg. F; EM. NZ. Reg. H; Data Management Private NZ. B). Much of the system has become location-based (Data Management Private NZ. B) using GIS technology (NEMA, 2020a). For example, the "welfare system [has] moved from being quite a database and table-based system to a location-based system" (Data Management Private NZ. B), and now utilises tools like Survey123³⁶ for welfare needs assessments (EM. NZ. Reg. H). The West Coast example outlined by Stowell (2020) provides more detail on how GIS has been used to streamline the CDEM impact assessment process.

Outside of impact data and the CDEM sector, other agencies have developed their own applications for conducting standardised information. For example, the Real-time Individual Asset Attribute Collection Tool (RiACT) was developed by risk modellers to capture real-time, geolocated, standardised asset information, such as building attributes (Risk Modelling NZ. B; Lin et al., 2014; Lin et al., 2020). The goal of this tool is to support an exposure data development framework whereby exposure data is systematically collected and stored for improved access, management, and use (Risk Modelling NZ. B; Lin et al., 2020).

While various standardised forms have been developed to capture hazard and impact data for various purposes, it would be beneficial to continue the systematic process of integrating the resulting data into a data repository for ease of access, sharing, and use beyond its initial purpose, depending on licensing and proprietary restrictions (e.g., Clark & Guiffault, 2018). Global efforts in the hydrometeorological hazard space are underway for building data repositories. For example, the HIWeather Value Chain Project under the World Weather Research Programme (WWRP) is currently building a catalogue of hydrometeorological events and their impacts to evaluate the end-to-end warning value chain (HIWeather, 2021). Additionally, the World Meteorological Organization (WMO) has established an initiative for cataloguing hazards and events through its members (Douris & Sopaheluwakan, 2019).

Strategies for building an integrated and multi-disciplinary data collection approach rely, again, on building and nurturing partnerships and collaboration. Collaboration allows data collectors and users from various disciplines to jointly define and scope problems for which the data is being collected, and identify the data types that are of interest to the problem (Ge et al., 2021). For example, Harrison et al. (2022) identified the need for including social scientists when designing post-impact assessments for risk modelling such that the modelling can extend into social and cultural impacts. For the aforementioned HIWeather Value Chain Project, an international multi-disciplinary task team of social scientists, meteorologists, and risk scientists was formed to co-develop the data collection template for the event catalogue (HIWeather, 2021). Consideration of emergent technology would also help

³⁶ <u>https://survey123.arcgis.com</u>. Accessed 7 July 2021.

with identifying efficient data collection and integration tools, such as GIS-based technology (e.g., Survey123), mobile devices, unmanned aerial vehicles (e.g., drones), etc. (Lin et al., 2019; Yu et al., 2018).

8.5 Conclusions and Limitations

This exploratory study built further understanding around the management, acquisition, and sharing of hazard, impact, vulnerability, and exposure (HIVE) data for severe weather hazards in New Zealand, towards supporting Sendai Framework priorities and implementing Impact Forecasts and Warnings (IFWs). While the qualitative nature of data collection and analysis herein limits the generalisability of results beyond the interviewees, this approach offers an in-depth understanding of a problem not readily available from quantitative approaches, which is appropriate for an exploratory study such as this (Blumer, 1969; Miles & Huberman, 1994; Patton, 2002).

We employed a socio-technical lens to our analysis to identify the organisational and technical intervening conditions and action/interaction strategies for accessing and sharing HIVE data in NZ. We found that there is a need for data governance of HIVE data. This involves identifying and appointing stewards and custodians of the relevant datasets such that the datasets can be maintained and available for further use beyond their initial purpose.

We also found a need for building and nurturing stronger partnerships to continue building trust between stakeholders for sharing data. Trust is an important factor for facilitating data sharing between agencies, and for people to share their data with hazards and impact monitoring crowdsourcing/citizen science projects.

Furthermore, integration challenges continue to interfere with the use of various sources of HIVE data for effective risk and impact assessments for IFWs and beyond. Systematic and standardised data collection approaches using GIS-based tools can support integration. Many templates for standardised data collection were found to exist in NZ but the resulting data has not been systematically collected into one place for easier access and use. Depending on licensing and proprietary restrictions, it may be beneficial to aggregate these data into a central repository for continued use.

This research provides empirical evidence supporting the need for establishing roles and practices for governing HIVE data in NZ. This is in support of both meeting the Sendai Framework priorities and implementing an IFW system for severe weather hazards in NZ. Building partnerships remains key to improving data collection, access, and sharing practices for DRR. Further research can investigate how these partnerships can be built and strengthened.

	Position	of the	reader
--	----------	--------	--------

	Chapter One
Part 1	Introduction
Farti	Chapter Two
	Background and Context
	Chapter Three
Part 2	Research Design
Tartz	Chapter Four
	Methods Results and Overarching Findings
	Chapter Five
	Volunteered Geographic Information for people-
	centred severe weather early warning: A
	literature review
	Chapter Six
	Identifying the data uses and gaps for severe
	weather impact forecasts and warnings
	Chapter Seven
	Identifying data sources for severe weather
Part 3	impact forecasts and warnings in Aotearoa New Zealand
	Chapter Eight
	A socio-technical analysis of sharing and
	governing hydrometeorological impact data in
	Aotearoa New Zealand
	Chapter Nine
	Integration of Partnerships and Collaboration to
	Support IFWs and HIVE Data Collection, Access,
	and Use
Part 4	Chapter Ten
i di ci	Discussion and Conclusion

This chapter presents the core category found as result of the selective coding processed employed in this Grounded Theory research and integrates it into the broader context of this doctoral study. This chapter aims to address the question *"How can partnerships and collaboration better facilitate the collection, creation, and access to HIVE data for IFWs?"* by achieving the following objectives of this doctoral thesis: 5.1 To identify the required partnerships and collaborations required for IFW systems; 5.2 To identify existing partnerships and collaboration in NZ that can support IFW systems; 5.3 To outline a path forward for nurturing partnerships and collaboration for IFW systems.

The chapter is presented as follows. First, the core category of Partnerships and Collaboration is re-introduced in Section 9.1 from Chapter Four. This will provide an overview of the properties and dimensions of this core category (Section 9.2). Following this, a discussion of how the Partnerships and Collaboration core category threads through both the IFW System (Section 9.3) and HIVE data (Section 9.4) phenomena will be presented. These phenomena are the two subjects under study in this PhD research (see Chapter Four). A discussion will follow to integrate these findings together and with the literature (Section 9.5). A summary will conclude the chapter (Section 9.6).

9.1 Partnerships and Collaboration Core Category Defined

Following the ES-GT research strategy employed in this PhD study (see Chapter Four), the core category is that which all other categories and concepts relate to, and which "incorporates or supersedes other categories in explanatory importance" (Timonen et al., 2018, p. 7). The core category was identified through the selective coding process described in Chapter Three and Chapter Four. It was determined that the Partnerships and Collaboration category threads throughout the categories and sub-categories of this research and explains much of the conditions and action and interaction strategies of IFW Systems and HIVE data collection, use, access, and management. As such, Partnerships and Collaboration threads throughout the findings of this PhD research are listed in Table 9.1. This aligns with the literature, where partnerships and collaboration have been proven critical both in research and in practice for disaster preparedness, planning, and response (Ge et al., 2021; Johnson, 2021; Morss et al., 2021; Peek et al., 2021).

Based on the two phenomena under study in this doctoral research, and the properties and dimensions presented in the next section (9.2), the core category of Partnerships and Collaboration in this PhD study is defined as

formal and informal, bottom-up or top-down approaches to establishing, building, and/or nurturing working relationships with stakeholders involved in the communication of and response to severe weather hazards, to allow for defining warning thresholds; creating consistent warning messages; sharing knowledge and data of hazards, impacts, vulnerability, and exposure; collecting appropriate and useful data; and managing said data towards the implementation of an impact forecasting and warning system.

The theoretical concepts, and thus the micro-theory, on which the above definition is based are represented visually in Figure 9.1, which places Partnerships and Collaboration in the middle of the figure, supported by its properties and dimensions (discussed next, in Section 9.2). Partnerships and Collaboration was found to play a key role in both the IFW System phenomenon and the HIVE data phenomenon, on which this PhD research is focused. This will be discussed in Sections 9.3 and 9.4, citing examples from the previous chapters and additional evidence in the interview and workshop data. Evidence from interviews is referenced using the interview codes previously presented in Table 3.6, for example 'Met. Research NZ. J' for participant J in the meteorological research field in New Zealand.

Table 9.1. Examples of themes relating to the Partnerships and Collaboration core category from the main results of each chapter.

Chapter	Key findings relating to Partnerships and Collaboration	
Chapter Six	Partnerships and collaboration used throughout the Warning Value	
	Chain:	
	 Hazard forecasting (e.g., initial discussions with MetService and hydrologists), 	
	 Impact forecasting (e.g., sharing data between agencies to support impact modelling in the UK Natural Hazards Partnership (NHP), and conducting impact-oriented discussions across stakeholders such as when the Southland Red Warning was issued, and for forecasting the impacts of ex-TC Debbie and ex- TC Cook, 	
	 Impact warning (e.g., defining impact thresholds like the UK MetOffice co-producing an impact matrix with stakeholders and updating warnings based on feedback form the stakeholders,) Co-design of an EWS with the target audiences. Need for coordinated multi-disciplinary collection of 	
	human/social impact data.	
	 Cross-sector collaboration for integrating dynamic exposure and vulnerability data. 	
Chapter Seven	Partnerships and collaboration are needed to:	
·	 Reduce unwillingness to share data, which may be due to a lack of trust between agencies. 	
	 Garner support and buy-in across regional stakeholders to allocate resources for better data collection. 	
	Support community leadership to drive innovation.	
Chapter Eight	 Partnerships and collaboration help to: Clarify roles and responsibilities, guidance in fulfilling tasks (e.g., Stats NZ helping NEMA manage their loss database). 	
	Facilitate data sharing practices.	
	 Develop integration strategies for seamless data sharing, e.g., developing the Common Operating Picture collaboratively to identify common needs and build trust and partnerships between agencies. 	



Figure 9.1. Integrative diagram of the micro-theory resulting from this doctoral research. The figure centres around core category of Partnerships and Collaboration for implementing an Impact Forecasting and Warning System. The two phenomena studied in this PhD thesis (Impact Forecasting and Warnings, and HIVE Data) are outlined in blue and labeled as such. The properties and their dimensions of the core category are outlined in green and labeled accordingly. The core category is presented in the middle, with arrows linking the properties and dimensions to the two phenomena.

9.2 Properties and Dimensions of the Core Category: Partnerships and Collaboration

The core category of Partnerships and Collaboration for IFW implementation and HIVE data collection, creation, and use consists of several properties and dimensions. Properties are "characteristics that define and describe concepts" (Corbin & Strauss, 2015, p. 220), and dimensions are variations within those properties (Corbin & Strauss, 2015, p. 220). Dimensions are represented as a continuum on which the properties lie, to provide range and specificity (Corbin & Strauss, 2015; Strauss & Corbin, 1990). Three properties were identified to make up the core category: types of partnerships, the directional approach to partnerships and collaboration, and strategies for building partnerships. Dimensions were identified for each of these properties. For example, formal and informal partnerships are the dimensions of the 'types of partnerships' property. These properties and their respective dimensions are summarised in Table 9.2 and will be discussed next, with direct references made to the findings reported in previous chapters and to additional interview and workshop data.

9.2.1 Types of partnerships

The first property of the Partnerships and Collaboration core category is types of partnerships and collaboration, incorporating the dimensions of formal and informal partnerships which represent the dimension of this property.

In the context of this research, a **formal partnership** is defined as a partnership that has been established with clear roles and responsibilities between partners and defined objectives for the partnership. Alternatively, an **informal partnership** is one that has formed rather spontaneously, typically in response to an event, to fill an immediate gap, and the roles and responsibilities are not mandated by an authoritative body or document. The NHP in the UK is an example of a formal partnership developed for providing authoritative and consistent hazard, impact, and risk information to responders and governments (Hemingway & Gunawan, 2018). This partnership consists of 17 UK public service agencies and was a result of a post-event review of the 2007 UK summer floods where the need for a national framework for reducing risks to delivering essential services from natural hazards was identified (Cabinet Office, 2008; Hemingway & Gunawan, 2018).

In NZ, no formal partnership like the UK NHP was identified over the course of this doctoral research. Instead, formal partnerships were found to exist in the form of science advice groups for non-hydrometeorological hazards, such as the New Zealand Volcanic Science Advisory Panel (NEMA, 2020b), and the New Zealand Tsunami Advisory Group, as described in Chapter Eight. These partnerships formed in response to gaps in services and communication that were observed during disasters that occurred both within and outside of NZ (Chapter Eight). The purpose of the NZ Volcanic Science Advisory Panel and associated members have been clearly identified in the Terms of Reference (see NEMA, 2020b), along with the roles and responsibilities for the members. While these groups and panels consist primarily of scientists who

provide science advise to CDEM Groups, it is important to note that other groups must be included for response and operations planning. For example, during the 2012 Tongariro eruption crisis, several other sectors became involved in the response, even though they were not initially identified or included in practice scenarios leading up to the event (Leonard et al., 2014). These were the health, agriculture, and the veterinary sectors (Leonard et al., 2014).

Property	Dimensions	Description
Types of Partnerships	FormalInformal	Formal and informal partnerships exist for IFW implementation and for collecting, using, and managing HIVE data. A formal partnership is one that has been established with clear roles and responsibilities between partners and defined objectives for the partnership. An informal partnership is one that has formed rather spontaneously, typically in response to an event, to fill an immediate gap, and the roles and responsibilities are not mandated by an authoritative body or document.
Directional Approach	Top-downBottom-up	The directional approach to form partnerships and collaborations refers to the motivation or drivers for the partnerships and collaboration. The top-down direction involves political guidance and a mandate to form the partnership and collaboration, while a bottom- up approach results in self-organised partnerships and collaboration.
Strategies for Building Partnerships	 Networking Professional Development Cohabitation Multi-disciplinary collaboration 	Four strategies were identified for strengthening and building relationships that can result in informal and formal partnerships and collaboration and can facilitate either a top- down or bottom-up approach.

Table 9.2. Summary of the properties and dimensions of the Partnerships and Collaboration core category.

No such science advice group has been formed for hydrometeorological hazards in NZ. However, the Resilience to Nature's Challenges Kia manawaroa - Ngā Ākina o Te Ao Tūroa (herein referred to as RNC) research programme offers a potential avenue for starting this conversation. RNC was launched in 2015 and is funded by the Ministry of Business, Innovation and Employment (MBIE, n.d.). The RNC programme is a successor to the former Natural Hazards Research Platform, which was a 10-year research programme that funded natural hazards research in NZ and "helped researchers and end-users work more closely together" (Natural Hazards Research Platform, 2020). A key objective of RNC is to promote innovative and collaborative research to build resilience to the natural hazards in NZ (RNC, 2018). The RNC programme is composed of eight themes: Rural, Urban, Matauranga Māori, Built Environment, Earthquake and Tsunami, Coastal, Volcano, and Weather and Wildfire. One participant in this doctoral research "always thought the [Natural Hazards

Research Platform] was a pretty good vehicle for encouraging collaboration" (Met. Research NZ. J). This participant further hopes "the [RNC] does the same", indicating that the leaders of the Weather and Wildfire theme "try to be as inclusive as possible in terms of people developing their plans" and act as a "coordination point" for researchers to align with them and with each other (Met. Research NZ. J). This is reflective of the identified need to build interdisciplinary research teams for rapid response disaster research (Ge et al., 2021; Peek et al., 2021). The RNC Weather and Wildfire theme, thus, may offer a potential mechanism for building both formal and informal partnerships for hydrometeorological hazard research and mitigation. Still, this may be difficult to implement in practice with no authoritative agency involvement, funding, or mandate (Peek et al., 2021).

While few formal partnerships were found to exist in NZ specifically for hydrometeorological hazards, informal partnerships do exist. One such example is between the MetService and regional hydrologists, as described in Chapter Six (Harrison et al., 2022), and Chapter Eight. This form of partnership and collaboration relies on the quality of the relationships between the MetService forecasters and meteorologists and the regional hydrologists, which can be influenced by staff turnover, training, or lack thereof, etc. This is considered an informal partnership because it has not been mandated and no guidance has been written to define roles and responsibilities (Jung et al., 2018; Olmos-Penuela et al., 2013). It is a practical strategy to support decision-making for forming and issuing hydrometeorological warnings. Closely related to the types of partnerships and collaboration identified here (formal and informal), is the directional approach to forming these partnerships.

9.2.2 Directional approach

The directional approach to form partnerships and collaborations refers to the motivation or drivers for the partnerships and collaboration. Top-down and bottomup are the two directions (i.e., property dimensions) that were identified in the analysis of the interviews and workshops. The top-down approach would involve political guidance and a mandate to form the partnership and collaboration, while a bottomup approach results in self-organised partnerships and collaboration (Kapucu & Demiroz, 2017). Thus, formal partnerships are usually formed using a top-down approach where a governing body has mandated the formation of a partnership, such as the NHP in the UK (Hemingway & Gunawan, 2018), and provided funding (Morss et al., 2021). Alternatively, informal partnerships usually arise from a bottom-up approach where potential partners might self-organise to meet a common need (Kapucu & Demiroz, 2017). The NZGIS4EM previously described in Chapter Eight is an example of a bottom-up approach to forming informal partnerships to facilitate effective collaboration and coordination during disaster response efforts, and eventually building a large network and community of practice for the GIS and EM sector in NZ. As noted in the previous section (9.2.1), a top-down approach in which mandates are established and funding is made available by authoritative, governing bodies, would further support the establishment of top-down partnerships that enable

various groups to effectively collaborate with minimal barriers (Ge et al., 2021; Peek et al., 2021).

9.2.3 Strategies for building and strengthening partnerships and collaboration

Several strategies were identified during the interview and workshop data analysis for building and strengthening partnerships and collaboration for IFW implementation and HIVE data collection, creation, and use. These strategies (i.e., property dimensions) are **networking**, **professional development**, **cohabitation**, **and multi-disciplinary collaboration**.

Networking was identified in the interviews as a strategy for building and strengthening partnerships and collaboration (EM. Gov. NZ. Nat. J). A technical or science advisory group like the New Zealand Volcanic Science Advisory Panel and the New Zealand Tsunami Advisory Group may be useful for hydrometeorological hazards in New Zealand to enable further understanding and consistent communication of the risks and impacts of these hazards (EM. Gov. NZ. Nat. J). This may also facilitate more efficient data access and sharing practices (Ge et al., 2021). According to the research participants, forming such a group in NZ for severe weather and floods requires building and nurturing relationships by attending national conferences such as the National Emergency Management Agency (NEMA) Conference, NZ Meteorological Society Conference, NZ Hydrological Society Conference, etc., where attendees could test ideas amongst each other, garner support, and present it to higher-level decision- and policymakers (EM. Gov. NZ. Nat. J). For example, the MetService might register some of their staff to attend the NEMA conference, and CDEM Groups and councils might register their staff to attend the NZ Meteorological Society Conference and the NZ Hydrological Society Conference. This can be beneficial for making new connections, strengthening existing partnerships, and starting conversations.

Agencies that often work together, such as the MetService, councils (e.g., hydrologists), and CDEM Groups, can also nurture their working relationships by running **professional development** workshops together and hosting these workshops at their own agencies to allow other agencies to build an understanding of how each other operates (EM. NZ. Reg. K). One regional CDEM Group indicated that the NZ MetService used to host such visits with their group, which they found valuable for nurturing their relationship (EM. NZ. Reg. K). Past workshops held by researchers from CRIs were also seen as beneficial for keeping practitioners up to date on current research efforts and innovations (EM. NZ. Reg. K). These practices are akin to running scenario exercises to build relationships in advance of disaster events, for effective decision-making, communication, etc. (Doyle & Paton, 2017). Such exercises have proven highly beneficial for building and nurturing partnerships and collaboration for disaster risk reduction (Doyle et al., 2015; Hosterman et al., 2019; Leonard et al., 2014; Wein et al., 2016).

Cohabitation is a strategy that has been employed by the UK Met Office and Environment Agency to improve flood forecasting and warnings (Flood Forecasting Centre, n.d.; Hemingway & Gunawan, 2018). Cohabitation occurs when experts from different disciplines operate from the same location. When asked if cohabitation was an option in NZ for ease of communicating warnings, our NEMA participant (EM. Gov. NZ. Nat. J) indicated that it could be done in a bottom-up fashion where the MetService and CDEM Group(s) and council(s) make their own cohabitation arrangements. This has been done between the NZ MetService and Auckland council, where a MetService meteorologist sits in the Auckland Emergency Management office. Similarly, a supervisor of this PhD research is co-located with a regional CDEM Group. However, the participant indicated that it might be more effective if some political direction was given from the top, for example, if a bill (i.e., a proposed law) were introduced or an existing bill was appended to say that this should be done (EM. Gov. NZ. Nat. J). Doing so would be difficult due to agencies' internal politics, funding sources, operational practices, and governance structures (EM. Gov. NZ. Nat. J). Thus, a starting point would be to look at the governance arrangements of each agency, for example who governs the MetService, NEMA, CRIs, CDEM Groups, etc. and determine "how you would bring, and not bring, the organisations together, [put] some of their functions together or instruct them" to achieve something together (EM. Gov. NZ. Nat. J).

Collaboration between scientists from **different disciplines** (e.g., risk modellers and social scientists) was identified as a needed strategy for ensuring that data collection is comprehensive and accurate, and so that the risk and impact assessments of hydrometeorological hazards go beyond the built environment and extend into social human impacts (Chapter Six; Harrison et al., 2022). This further reflects proposals for building interdisciplinary teams for disaster response research (Ge et al., 2021) to support integrated transdisciplinary risk assessment and management processes (Johnson, 2021). Platforms like the RNC programme can enable this kind of collaboration by providing a channel for researchers to engage with each other. Doing so enables knowledge co-production to support collaborative, adaptive, and robust policies (Johnson, 2021).

The core category of Partnerships and Collaboration, consisting of the above properties and their respective dimensions, was found to thread through the two phenomena studied in this thesis (IFW Systems and HIVE Data). Examples of how Partnerships and Collaboration relates to IFW Systems and HIVE Data are discussed next.

9.3 Partnerships and collaboration in the Impact Forecasting and Warning Systems Phenomenon

Partnerships and collaboration thread throughout the IFW Systems phenomenon as an integral action/interaction strategy for implementing IFWs. Partnerships and Collaboration enable the important practice of sharing data and knowledge between agencies. Sharing of this data and knowledge through partnerships and collaboration can then facilitate the (re-)defining of impact thresholds for an IFW system and can ensure that messages are consistent across agencies. These actions of sharing data and knowledge, (re-)defining impact thresholds, and producing consistent warning messages emerged from codes relating to the IFW phenomenon as outlined in Chapter Four. Examples of how Partnerships and Collaboration enable these actions in the IFW phenomenon are provided next.

9.3.1 Sharing Data and Knowledge

Various agencies possess the knowledge and data needed for IFWs, as reported in Chapter Seven (Harrison et al., 2021), and listed in Table 7.4, Table 7.5, Table 7.6, Table 7.7. These various agencies and the associated data that they possess are visualised in a Venn diagram in Figure 9.2. This diagram displays the agencies that were identified by participants as those that possess HIVE data for hydrometeorological hazards and impacts in NZ, as reported in Chapter Seven (Harrison et al., 2021). Each leaf of the Venn diagram represents data partners for one of the four data types: hazard, impact, vulnerability, and exposure. Within each leaflet, the agencies identified in Table 7.4, Table 7.5, Table 7.6, and Table 7.7 for collecting or possessing datasets from each data type are placed in the Venn diagram. For example, the MetService was identified as possessing hazard, impact, and vulnerability knowledge and data in various forms: they continuously monitor and collect observational data on meteorological hazards, they monitor severe weather impacts reported in the media, and they possess tacit knowledge of regional vulnerabilities to certain meteorological hazards (e.g., knowing that Auckland is particularly vulnerable to winds coming from a northeast direction). They use this information to inform their decisions to issue a warning and how to emphasise and communicate the threats contained within. In another example, Data Ventures holds exposure information in the form of cell phone location data. This data was used recently, not for severe weather hazards, but to inform decision-making around alert level changes in response to the COVID-19 pandemic (Chapter Seven; Harrison et al., 2021). There is potential for this data to be useful for risk and impact modelling for hydrometeorological hazards (Chapter Seven; Harrison et al., 2021).



Figure 9.2. A Venn Diagram displaying agencies that possess HIVE data in New Zealand.

From the interviews, it appears that at least four groups in NZ possess all four types of data needed for IFWs: hazard, impact, vulnerability, and exposure. As shown in Figure 9.2, based on the results presented in Chapter Seven, these are local and regional councils, university and independent researchers, NIWA, and GNS Science. Local and regional councils possess data on flood hazards, such as river height and flow gauges, sea level data, river network, overland flow paths floodplain data, coastal inundation maps, slope data, and live camera feeds of river heights. In terms of impact data, councils have been found to use social media, crowdsourcing, and community volunteers to collect reports. Councils also have their staff and contractors conduct damage assessments, and often write post-event reports after a significant event for post-event analysis. Councils also possess data or information needed for vulnerability assessments, such as asset information (i.e., characteristics) and building damage assessments. Likewise, for exposure, councils have data on asset location and landuse zoning. Council staff also possess tacit knowledge of hazard, impacts, vulnerability, and exposure, and much of this tacit knowledge is undocumented.

Researchers were also found to possess various forms of HIVE data based on the interviews and workshops. Researchers often use social media and crowdsourcing to collect hazard and impact data for their own purposes. Researchers also collect data from media reports (indirectly), and directly from damage surveys and post-event interviews/surveys. Researchers also produce data from conducting risk and impact models.

From the interviews and workshops, NIWA and GNS Science were the third and fourth agencies found to possess HIVE data in various forms for hydrometeorological hazards. Like the MetService, NIWA collects observational data for hydrometeorological hazards, in addition to possessing data on river networks and sea levels in NZ. NIWA and GNS Science also conduct risk modelling for hydrometeorological and geological hazards using RiskScape, and thus collect and/or possess HIVE data to conduct these assessments. For example, NIWA officials conduct damage surveys following flood events to collect flood damage data for buildings and produce exposure layers based on overlaying spatial asset and hazard layers (Paulik et al., 2020). At GNS Science, efforts are underway to build a Pacific region exposure dataset for risk modelling (Lin et al., 2014; Lin et al., 2020)While one organisation might contain specific types of HIVE data (as shown in Figure 9.2), together they may not be able to be used for IFW. For example, GNS Science has landslide hazard data, but does not possess or collect rainfall data. Thus, the distribution of HIVE data across the various agencies represented in Figure 9.2 demonstrates the need to share data and knowledge between agencies for IFWs. This argument is supported in related IFW reports (e.g., Harrowsmith et al., 2020; WMO, 2015). The UK NHP is a prime example of how HIVE data can be shared between agencies to support IFWs. Interview findings from this thesis research (e.g., Chapter Six; Harrison et al., 2022), and recent literature (e.g., Hemingway & Gunawan, 2018), highlighted how this formal partnership enables the UK MetOffice to obtain key information for their IFWs, such as transportation data from the UK transport authority, to inform their Vehicle OverTurning Model, or population movement (i.e., exposure data) from the UK Health and Safety Executive.

Furthermore, as found in Chapter Six (Harrison et al., 2022), if an agency were to possess all of the required data for IFWs, if they do not have the mandate (i.e., roles and responsibilities) to issue IFWs, they cannot do it. Partnerships and collaboration thus are important for IFWs, to allow agencies who possess the required data for IFWs (such as NIWA, GNS Science, EM agencies, etc.) to work with the mandated warning services (such as the MetService and councils). Still, dynamic exposure and vulnerability, which were identified as important for IFWs in Chapter Six (Harrison et al., 2022), were found not to exist or be readily available in NZ.

9.3.2 Re-defining Warning Thresholds

While no formal partnership like the NHP was found to exist in NZ from the interviews and workshops, informal partnerships were found to exist between the MetService and EM Groups and hydrologists in New Zealand. These partnerships allow the MetService to (a) include CDEM Groups and hydrologists in their decision to issue a hazardous weather forecast, and (b) alert them to the fact that they have issued a hazardous forecast or warning so that they can prepare. They do this by 'phoning up' the hydrologists, and/or EM group to start the conversation(s), as shown in Figure 9.3, and as reported in Chapter Seven (Harrison et al., 2022).

Figure 9.3 provides a conceptual example of a severe weather warning chain in NZ and the partners involved in this chain. This figure is the result of diagramming and memo-writing following interviews with NZ MetService, CDEM Group officials, and hydrological participants (an early version of the diagram was previously presented in Chapter Four. The figure presented here is an updated version based on the iterative ES-GT analysis process). Starting with monitoring hydrometeorological hazards by the MetService and council hydrologists, when а potentially hazardous hydrometeorological phenomenon is identified (such as intense, heavy rainfall), the MetService reaches out to their hydrological and CDEM Group contacts in the region of interest to conduct a risk or impact assessment. This assessment can be discussionbased, model-based, or both (Chapter Six; Harrison et al., 2022). From there, both the MetService and council may choose to issue a watch or warning, which they disseminate to the respective CDEM Group(s), members of the public, lifelines sector, and other stakeholders.

Discussions with participants indicated that a result of risk or impact assessment is the decision to issue a warning (or not), and the level of warning. For example, the MetService could issue an Orange or Red Warning based on feedback from the CDEM Group and hydrologist(s), considering antecedent conditions and long-term forecasted conditions (Harrison et al., 2022). The council would also be responsible for issuing a flood warning if needed (Potter et al., 2021). The MetService also works with regions to adjust warning thresholds. For example, the MetService has worked with Auckland to adjust thresholds for damaging winds, as reported in Chapter Six (Harrison et al., 2022). They also lower thresholds if they know the antecedent conditions that might exacerbate the impacts by talking with the hydrologists and CDEM Groups to be aware of their current exposure, vulnerability, and response capacities (Harrison et al., 2022). This shows the importance of communicating and collaborating with stakeholders to define or redefine thresholds based on impacts for more effective warnings.



Figure 9.3. A conceptual example of a severe weather warning chain in NZ and the partners involved in this chain. The figure identifies the actors (warning audiences and decisionmakers and warning authorities) involved in the warning chain and the actions of said actors. The actions are either hazard-based (such as monitoring and collecting observational data of hazards) or risk- and impact-based (where risk and impacts are considered in addition to the hazard information). The direction of information and communication is represented by arrows. Multi-directional arrows indicate that the actors such as the MetService, hydrologists, and CDEM Groups both contribute to the risk assessments and are informed by the risk assessment outputs for their warning messages.

9.3.3 Consistent Warning Messages

Consistency of warning messages has been deemed critical for effective risk communication (Williams & Eosco, 2021). Partnerships and collaboration are also important for ensuring that the warning messages and information is consistent across agencies and reaches target audiences. For this reason, Auckland Emergency Management works with the MetService to make sure that what they are putting out aligns with the forecasts and warnings from the MetService (Auckland Workshop). In another example, WeatherWatch, a private weather forecasting company in NZ, and the MetService have worked together over the years to strengthen their relationship such that they can ensure that they are putting out messages that are productive for the public to take action, rather than detract from each other (Met. Private NZ. L). The objective is to reach the targeted warning audiences with timely, accurate, and relevant information to aid preparedness and response actions (Basher, 2006). In a negative case³⁷ example, instances of conflicting weather forecasts have also occurred between hydrometeorological agencies in NZ where these conflicting forecasts were posted on social media channels (Gorman, 2016). Having multiple providers of (sometimes conflicting) weather forecasts and warnings in the media (and social media) may cause public confusion, leading to potential (or perceived) mistrust of the MetService and their role as the weather warning authority. Further research should verify these affects.

9.4 Partnerships and collaboration in the HIVE Data Phenomenon

Due to the distribution of HIVE data sources across these agencies and stakeholders, as previously shown in Figure 9.2, partnerships and collaboration are needed to open up access to these datasets and data sources for IFWs in order to reduce instances of repetition for data collection and creation. Within the HIVE Data phenomenon of this doctoral research, the category of partnerships and collaboration was found to affect the actions covered in Chapters Chapter Seven and Chapter Eight: data collection, data custodianship and management, and data access and sharing.

9.4.1 Data Collection

Many agencies collect and create various forms of HIVE data for various uses, as reported in Chapter Seven (Harrison et al., 2021). Interdisciplinary or cross-disciplinary collaboration has become increasingly important for disaster-related data collection (Ge et al., 2021). Findings from this doctoral thesis show the importance of this collaboration for creating templates for collecting data that can be used beyond its initial purpose, or so that the data collection is comprehensive. For example, in Chapter Six Harrison et al. (2022) found that to capture the social impacts of severe weather hazards, risk modellers who typically conduct post-damage assessments to calibrate their models should coordinate and collaborate with social scientists to ensure that the templates capture appropriate characteristics to inform social impact models as well, and to ensure that the data collection method itself is ethically sound.

³⁷ A negative case is a case that does not fit the pattern (Corbin & Strauss, 2015).

Such interdisciplinary collaboration has the potential to incorporate new and shared perspectives to a problem (Ge et al., 2021). Collaborators can thus jointly define and scope problems, and identify the data types that are of interest to these problems (Ge et al., 2021). The overall outcomes of this kind of 'data-driven' collaboration can be the production of "more holistic solutions that grow and evolve from the shared space" (Ge et al., 2021, p. 1146). Successful collaborations typically have the institutional support and have a foundation of long-term collaborations (Ge et al., 2021).

9.4.2 Data Governance, Access, and Sharing

Governance, access, and sharing of HIVE data, was also found to involve aspects of partnerships and collaboration. Data governance protocols help to establish authority and control over data by assigning clear roles and responsibilities (Benfeldt et al., 2019). Data governance also involves examining practices for data collection, management, accessibility, and use (Janssen et al., 2020). As reported in Chapter Eight, data governance became an important theme in discussions with participants about managing HIVE data. Furthermore, partnerships and collaboration were found to be important for agencies to learn from each other for best practices, particularly for agencies who have not traditionally been responsible for collecting such data in the past. For example, NEMA was working with Stats NZ to receive guidance on best practices for managing the national loss database currently in development for reporting under the Sendai Framework (see Chapter Eight). This example supports the notion that data governance depends on collaboration between the organisations and people of which the system comprises (Janssen et al., 2020). This includes establishing trusted frameworks for reliable and secure data sharing between organisations (Janssen et al., 2020).

Partnerships and collaboration were found to be an action/interaction strategy to enable access to and sharing of hydrometeorological HIVE data. Several examples of partnerships and collaboration were found to enable sharing data and knowledge for hydrometeorological hazards. The partnerships between the MetService and CDEM Groups and hydrologists were outlined in the previous section to inform severe weather warnings, as reported in Chapter Six (Harrison et al., 2021). In addition to that, interviews indicated that the MetService shares their data files (see Chapter Eight) with hydrologists to integrate with their flood models. In another example, the partnerships and collaboration formed in the NZGIS4EM community facilitates the sharing of data across agencies (see Chapter Eight).

The NZ MetService and WeatherWatch are also working together to form amicable data access and sharing arrangements (Met. Private NZ. L), however this is not necessarily the case with other agencies due to data sharing restrictions (Met. Private NZ. L). This has introduced a debate around which data should be made 'open' (Met. NZ. K). Global calls have been made to make hydrometeorological data openly accessible, citing various economic benefits (e.g., Rogers & Tsirkunov, 2021). However, a review of open access to weather data in New Zealand conducted by MBIE (MBIE, 2017) found that most data reuse principles are being met by the involved

agencies, but access to observational weather data in NZ is more restricted than in other countries (MBIE, 2017). This is due to the State-Owned Enterprise (SOE) and CRI models under which the agencies operate; these models are based on earning commercial revenue to support data collection and cover operating costs (MBIE, 2017). The review determined that the costs to the NZ taxpayers of increasing open access to raw observation data would outweigh the benefits (MBIE, 2017).

While the MBIE review determined that the current accessibility of weatherrelated data is acceptable when considering the cost-benefit to NZ-taxpayers, this issue highlights the importance of forming functional partnerships such that data sharing agreements can be made between agencies (The World Bank, 2020). The Weather Enterprise in the USA is an example of a public-private-academic partnership formed across the various sectors involved in collecting, creating, using, and communicating weather information (e.g., Government agencies, EM agencies, academia, private agencies, broadcast media, social science) (NWS, n.d.). The Global Weather Enterprise (GWE) is another example of efforts towards increasing the accessibility of weather information (Thorpe & Rogers, 2018). The GWE comprises all the Warning Value Chain components, products, processes, and actors that must come together to provide accurate and reliable weather information (Thorpe & Rogers, 2018). As the GWE is collaborative in nature, partnerships across culturally different sectors are essential to its success (Thorpe & Rogers, 2018).

9.5 Discussion and Integration

The relationship of Partnerships and Collaboration with the two phenomena in this study (IFWs and HIVE Data) is represented in Figure 9.1. Sharing knowledge and data is the interface between these two phenomena, as demonstrated in Figure 9.1, because it fills the knowledge gap identified by meteorologists in practice and in the literature for implementing IFWs (e.g., Potter et al., 2021). Findings in Chapter Six showed the needs of HIVE data throughout the Warning Value Chain for IFWs. Additional findings in Chapter Seven identified existing and potential sources for HIVE data that can support IFWs. However, because the data sources are numerous with many actors involved in collecting and using it, the need for effectively managing, sharing, and accessing the data was identified and further explored in Chapter Eight. Thus, it was found that Partnerships and Collaboration are essential for facilitating effective data sharing practices for IFW implementation.

The findings of the thesis which point to Partnerships and Collaboration as a necessary strategy for implementing IFWs aligns with recommendations in the WMO Guidelines (WMO, 2015) and a more recent guide published by the International Federation of Red Cross and Red Crescent Societies (IFRC) (Harrowsmith et al., 2020). These findings are further exemplified by the UK NHP (Hemingway & Gunawan, 2018). Partnerships for IFW systems allow for partner organisations to understand hazards, identify impacts, and assess user requirements (Harrowsmith et al., 2020). The NHP was lauded for "leading the way in moving from hazard-based to impact-based natural hazard research to better understand and forecast potential impacts" (Hemingway & Gunawan, 2018, p. 508). Such a partnership allows for the inclusion of diverse scientific

expertise which promotes efficient, robust, and "practically relevant" forecasting tools (Hemingway & Gunawan, 2018, p. 508). This doctoral research provides further empirical evidence of the need for building and nurturing partnerships and collaboration both for implementing IFWs and for better management of, and access to, HIVE data for IFWs and DRR in general.

Multi-organisational collaboration has many challenges (Sonnenwald, 2007). The process of building partnerships like the NHP requires extensive time, coordination, communication, and interaction between agencies (Hemingway & Gunawan, 2018). Barriers to these factors include resourcing, such as available funding for billable hours, workloads, and establishing facilitations roles, etc. Building trust and increasing the willingness of organisations to participate are additional hurdles (Pennington et al., 2015; Sonnenwald, 2007). Further challenges include mutually identifying goals and objectives and agreeing on timelines, the use of differing terminology and epistemologies, legal issues around intellectual property, and developing workflows and communication standards, and sustaining the collaboration (e.g., maintaining interest, securing funding sources, etc.) (Pennington et al., 2015; Sonnenwald, 2007). Finally, developing partnerships to share data between agencies for an IFW system is based on the initial assumption that agencies that possess the desired data are both willing and able to share this data.

The strategies and approaches portrayed in Figure 9.1 can be enacted to address some of these challenges and for building and nurturing partnerships and collaboration in NZ towards implementing IFWs and collecting, sharing, and using the required HIVE data for IFWs and other DRR efforts. The findings of this thesis build on existing recommendations from the IFRC guide (Harrowsmith et al., 2020) and provide more tangible and direct strategies. For example, networking by attending conferences was identified as a strategy for scoping initial interest and capabilities in building partnerships and collaboration. The results of the networking exercise may then be brought to policymakers and decisionmakers to implement a more top-down approach for mandating formal partnerships which may then open funding opportunities to support these efforts and provide guidance on legal issues with data sharing. Hosting professional development workshops and visits across agencies and cohabitation support building trust between agencies, mutually identifying goals, clarifying terminologies and epistemologies, and sustaining the collaboration.

9.6 Chapter Summary

This chapter presented the core category resulting from the Evolved-Straussian Grounded Theory analysis, accompanied by a chain of evidence, from Chapter Six to Chapter Eight and additional interview and workshop data, supporting the elevation of this category to core category. Partnerships and Collaboration was identified as the core category due to its explanatory power around IFW Systems and accessing and using the required HIVE Data for IFWs. The core category of Partnerships and Collaboration involves using various approaches to establish partnerships across the various agencies and stakeholders that collect, create, and use HIVE to support the implementation of IFWs.

The concepts and relationships comprising this core category were represented in Figure 9.1. The properties and dimensions of this core category were further elaborated. These properties and dimensions are types of partnerships (e.g., formal, and informal), directional approach (e.g., bottom-up, and top-down), and strategies (e.g., Networking, Professional Development, Cohabitation, and Multi-disciplinary collaboration). These properties and respective dimensions represent approaches and strategies to building and nurturing partnerships and collaboration in support of IFW implementation and HIVE data collection, access, and sharing.

Sharing data and knowledge was identified as the interface between the two phenomena in this study (IFW Systems and HIVE Data), as this allows warning agencies to access and use information that they may not otherwise have but require for an IFW system. This activity of sharing data and knowledge is essentially supported by Partnerships and Collaboration in various forms.

The chapter concluded by integrating the properties and dimensions of the core category with the IFW Systems and HIVE data phenomena, and with the IFW and DRR literature. Challenges with building Partnerships and Collaboration were identified from the literature, and the approaches and strategies identified in this chapter were proposed as potential solutions to these challenges. This meets Objectives 5.1 To identify the required partnerships and collaboration in NZ that can support IFW systems; and 5.3 To outline a path forward for nurturing partnerships and collaboration for IFW systems. The next chapter will provide an overall discussion and conclusion of the thesis.

	Chapter One		
Deut 1	Introduction		
Part I	Chapter Two		
	Background and Context		
	Chapter Three		
	Research Design		
Part 2	Chapter Four		
	Mathada Pasulta and Overarching Findings		
	Methods Results and Overarching Findings		
	Chapter Five		
	Volunteered Geographic Information for people-		
	centred severe weather early warning: A		
	literature review		
	Chapter Six		
	Identifying the data uses and gaps for severe		
	weather impact forecasts and warnings		
	Chapter Seven		
	Identifying data sources for severe weather		
Part 3	impact forecasts and warnings in Aotearoa New		
	Zealand		
	Chapter Fight		
	A socio-technical analysis of sharing and		
	averning hydromotoorological impact data in		
	Actoarca New Zoaland		
	Chapter Nine		
	Chapter Nine		
	Integration of Partnerships and Collaboration to		
	Support IFWs and HIVE Data Collection, Access,		
	and Use		
Part 4	Chapter I en		
i dit i	Discussion and Conclusion		

Position of the reader

Chapter Ten: Discussion and Conclusion

This chapter provides a discussion of, and conclusion for, this doctoral thesis. The first section (10.1) revisits the research questions and outlines how these research questions were addressed in the thesis. The second section (10.2) discusses the research implications, followed by limitations and future research directions in the third section (10.3). The final section (10.4) concludes the thesis.

10.1 Research Overview

This thesis contributes to current efforts towards implementing Impact Forecasting and Warning systems for hydrometeorological hazards. This doctoral research study set out to answer the research question:

Which data are needed to support Impact Forecasting and Warning (IFW) systems for hydrometeorological hazards, and how are they currently collected, stored, and shared in New Zealand?

This question was broken down into five sub-questions. The Evolved-Straussian Grounded Theory research strategy was employed to address these research questions. Each of these research questions and their associated objectives are presented in Table 10.1 and will be addressed in turn. Table 10.1 further summarises the highlights of each chapter in alignment with the research questions and objectives.

Re	search Questions	Objectives	Chapter and Paper Highlights
1.	What are the current and potential uses of Volunteered Geographic Information (VGI) for severe weather warnings?	1.1 To establish VGI as a potential source of impact data for IFWs.	 Chapter Five Various forms VGI, from geo-located social media, crowdsourcing platforms, participatory mapping/participatory GIS, and local knowledge, have value and a role to play in all components of a severe weather EWS. Some forms of VGI are more useful for specific EWS components than are others. VGI processes can bridge the gap between EWSs and audiences of warnings by incorporating local knowledge and personal experiences from stakeholders into the EWS components.
2.	What are the data uses and gaps for impact forecasts and warnings?	 2.1 Identify the actors involved in an IFW system and their associated roles. 2.2 Determine how HIVE data are used in an IFW system. 2.3 Identify further data gaps/needs for implementing IFWs. 	 Chapter Six Supports existing research pointing to the need for creating, gathering, and using impact, vulnerability, and exposure data for IFW systems. Identifies where each data type (hazard, impact, vulnerability, and exposure) can be used for each Warning Value Chain component. Offers examples of alternative approaches for impact forecasting and for defining impact thresholds. Provides new insight into a growing need to identify, model, and warn for social and health impacts.
3.	What are the sources of HIVE data?	 3.1 Identify the creators, collectors, and users of HIVE data that is relevant to severe weather IFWs. 3.2 Identify the inhibitors and facilitators to collecting and using these data, to support the implementation of an IFW system in New Zealand for severe weather hazards. 	 Chapter Seven Identifies various sources for hazard, impact, vulnerability, and exposure data for implementing severe weather IFW systems. Points to technological advancements for enabling collection and creation of HIVE data, such as GIS-based tools and mobile devices. Suggests that priorities, motivation, and interest within organisations influence how well data is collected and used. Identifies a tension between the timeliness and trustworthiness of data needed for emergency response and warnings. Proposes strategies for addressing challenges and barriers for collecting and using HIVE data.

Table 10.1. Alignment of chapters with research questions and objectives.
Re	search Questions	Objectives	Chapter and Paper Highlights	
4. Ho go sh hy ha Ze	How can HIVE data governance, access, and sharing be improved for hydrometeorological hazards in New Zealand?	 4.1 Identify and understand the governance and acquisition process for HIVE data for severe weather hazards in New Zealand. 4.2 To support efforts to fulfil the Sendai Framework priorities around disaster data access. 4.3 To support the implementation of a severe weather IFW system. 	 Chapter Eight Identifies a need for data governance of hazard, impact, vulnerability, and exposure data. Identifies a need for building and nurturing stronger partnerships to continue building trust between stakeholders for sharing data. Proposes systematic and standardised data collection approaches using GIS-based tools to address data integration challenges. 	
5.	How can partnerships and collaboration facilitate better collection, creation, and access to HIVE data for IFWs?	 5.1 To identify the required partnerships and collaborations required for IFW systems. 5.2 To identify existing partnerships and collaboration in NZ that can support IFW systems. 5.3 To outline a path forward for nurturing partnerships and collaboration for IFW systems. 	 Chapter Nine Identifies Partnerships and Collaboration as the core category due to its explanatory power around IFW implementation and accessing and using the required HIVE data for IFWs. Proposes approaches and strategies to building and nurturing partnerships and collaboration in support of IFW implementation and HIVE data collection, access, and sharing (e.g., providing more networking opportunities via conferences, training, workshops; cohabitation; inter-disciplinary collaboration). Identifies sharing data and knowledge as the interface between the two phenomena in this study (IFW Systems and HIVE Data). Integrates the properties and dimensions of the core category with the IFW and HIVE data phenomena, and with the IFW and DRR literature. 	

10.1.1 Research Question 1: What are the current and potential uses of Volunteered Geographic Information (VGI) for severe weather warnings?

Chapter Five answered the first research question "What are the current and potential uses of Volunteered Geographic Information (VGI) for severe weather warnings?" in the form of a scoping literature review. This research question was posed for the original topic of this thesis research which aimed to explore the role of social media and crowdsourcing for collecting impact data to support IFWs. Volunteered Geographic Information (VGI) was chosen as the form of social media and crowdsourcing data to explore because of the location information associated with it, which was proven to be an essential data piece for IFWs. Findings from the scoping literature review methodology employed in this manuscript indicated that:

- VGI is useful in all components of a severe weather EWS, but some platforms are more useful for specific components than are others.
 - Participatory mapping/participatory GIS and crowdsourcing support the disaster risk knowledge component of an EWS by validating risk maps/maps, integrating local and spatial knowledge, and populating hazard and impact databases.
 - Geolocated social media and crowdsourcing support detection, monitoring, and warning services by providing near real-time hazard and impact detection, forecasting, and warning verification, and estimates of hazard and impact severity based on incoming reports through these platforms.
 - Geolocated social media supports the communication and dissemination mechanism component by enabling warning services to assess the spread of, and response to, warning messages, and allowing the warning services to update subsequent warning messages based on social media activity.
 - Local knowledge and geolocated social media support preparedness and early response capacity by providing further understanding of knowledge and perceptions of local hazards risks and impacts, and informing local preparedness efforts with real-time hazard and impact information.
- Furthermore, VGI supporting people-centred EWSs by bringing people and their knowledge and experiences into EWSs.

Following the completion of this phase of the research, interviews with participants both in and outside of New Zealand pointed towards more pressing issues around accessing and using vulnerability and exposure data, not just impact data, for IFW systems. Thus, the focus of this doctoral thesis broadened to the topic of accessing and using hazard, impact, vulnerability, and exposure (HIVE) data for IFWs. This led to the development of the next three research questions, discussed next.

10.1.2 Research Question 2: What are the data uses and gaps for impact forecasts and warnings?

Chapter Six answered the research question "What are the data uses and gaps for impact forecasts and warnings?" using qualitative data collection and analysis methods employed under the Evolved-Straussian Grounded Theory research strategy. Using the Warning Value Chain (Golding et al., 2019) as a framework, the actors involved in IFW systems were identified, along with various uses of HIVE data from the interview and workshop data for the Value Chain components. Findings from this manuscript included:

- Within the Warning Value Chain:
 - Hazard data is used for Hazard, Observation, Monitoring, and Detection; Weather Forecasting; Hazard Forecasting; Impact Forecasting; and Impact Warnings.
 - Impact data is used for Hazard Forecasting; Impact Forecasting; Impact Warning; and Decision/Action.
 - Vulnerability data and Exposure data are used for Impact Forecasting and Impact Warning.
- Within the Impact Forecasting and Impact Warning value chain components, different approaches for Impact Forecasting and defining impact thresholds were identified that use quantitative models and impact-oriented discussions depending on the data available. For example, model-based approaches use algorithms and quantitative data to identify potential impacts. If quantitative data are not available, impact-oriented discussions using tacit knowledge, map/spatial data, etc. are used. Ideally, both approaches should be used complementarily.
- There is a growing need fill the following data gaps for IFWs:
 - We need to identify, model, and warn for social and health impacts, which have typically taken a back seat to modelling and forecasting physical and infrastructure impacts.
 - Dynamic exposure and dynamic vulnerability are needed to account for antecedent conditions, human response capabilities, and human behaviour, which change over space and time, thus pointing to the need for thresholds to be based on these dynamic factors.

These findings on the data needs and uses within IFW systems will help guide their development, fill data gaps, and provide a pathway for identifying specific relevant data sources.

10.1.3 Research Question 3: What are the sources for HIVE data?

The third research question, *"What are the sources for HIVE data?"* was answered in Chapter Seven using the Evolved-Straussian Grounded Theory data collection and analysis methods of the interview and workshop data. Findings indicated that:

- Many sources of HIVE data are collected for other uses (such as for response efforts for disaster and emergency events, and for research) that have relevant application for IFWs.
 - Examples of sources of HIVE data that are not currently used widely for IFW are social media and crowdsourced data, emergency call centre reports, damage surveys, health and injury data, damage assessments, cultural impacts, census data, and health data.
- Priorities, motivation, and interest within organisations influence how well data is collected. Moreover, agencies tend to prefer official data, but official data has limitations that unofficial data (such as social media and crowdsourcing) may address, such as timeliness.
- To that end, an interesting tension was identified between the timeliness and trustworthiness of data needed for emergency response and warnings.

The findings from this manuscript provide a resource for warning services and other stakeholders, who are looking to implement IFWs or otherwise access hydrometeorological HIVE data, to identify data sets and data sources for their needs.

10.1.4 Research Question 4: What is the governance and acquisition process for HIVE data for severe weather hazards in New Zealand?

The fourth research question, *"How can HIVE data governance, access, and sharing be improved for hydrometeorological hazards in New Zealand?"* was answered in Chapter Eight, again using the Evolved-Straussian Grounded Theory data collection and analysis methods for the qualitative interview and workshop data. Findings are summarised below:

- HIVE data is largely managed by the creators and owners of the datasets. For example, the MetService manages its meteorological observational data and forecasts, NIWA manages the national climatological database, CDEM Groups manager their damage assessment data, LINZ manages national geospatial information, Stats NZ manages census data, and NEMA manages the national loss database for Sendai Framework reporting.
 - A need for establishing data governance principles for HIVE data was identified to improve the accessibility to and sharing of HIVE data.
 - A need for nurturing partnerships was identified for building trust between stakeholders such that data and knowledge can be efficiently shared between agencies for IFWs.
- Furthermore, data integration challenges continue to plague the use of various sources of HIVE data for effective risk and impact assessments for IFWs and beyond. Systematic and standardised data collection approaches using GIS-based tools were identified as strategies for supporting integration.

The findings from this manuscript proposed practical strategies for stakeholders to improve the management of, access to, and sharing of HIVE data for IFWs and other DRR efforts.

10.1.5 Research Question 5: How can partnerships and collaboration facilitate better collection, creation, and access to HIVE data for IFWs?

The fifth and final research question, *"How can partnerships and collaboration facilitate better collection, creation, and access to HIVE data for IFWs?"* was answered in Chapter Nine, which integrates the findings from Chapter Five to Chapter Eight into the core category of Partnerships and Collaboration.

- Partnerships and Collaboration was identified as the core category due to its explanatory power around IFW implementation and accessing and using the required HIVE data for IFWs.
- Sharing data and knowledge was identified as the interface between the two phenomena in this study (IFWs and HIVE Data). The partnerships and collaboration to support this facilitate better collection of, creation of, and access to HIVE data as this allows warning agencies to access and use information that they may not otherwise have but require for an IFW system.
- Challenges with building Partnerships and Collaboration were identified from the literature, and the approaches and strategies identified in this chapter were proposed as potential solutions to these challenges.

Chapter Five to Chapter Eight have answered the research questions and addressed the research objectives, as shown in Table 10.1. Together, these chapters answered the overarching question of "*Which data are needed to support Impact Forecasting and Warning (IFW) systems for hydrometeorological hazards, and how are they currently collected, stored, and shared in New Zealand?*"

10.2 Research Implications and Contributions

Impact Forecasts and Warnings (IFWs) have been proposed by the WMO to address communication gaps and challenges with traditional hydrometeorological Early Warning Systems (EWSs) (Harrowsmith et al., 2020; WMO, 2015). IFWs are commonly described as communicating 'what the weather will do, rather than what the weather will be,' such that the warnings might better align with the position, needs, and capabilities of target audiences. However, a key challenge with implementing IFWs that has been identified both in the literature and in practice is the perceived lack of impact, vulnerability, and exposure data. Yet, impact, vulnerability, and exposure are key elements in assessing and understanding the risk posed by hydrometeorological hazards. This doctoral research thus provides four main contributions to the IFW literature and implementation efforts with regards to the access to, and use of, hazard, impact, vulnerability, and exposure (HIVE) data:

- 1) This research reiterates the growing need for gathering and using impact, vulnerability, and exposure data for IFWs. Additionally, this research points to an increasingly important need to broaden the apparent focus on understanding impacts to the built environment to social and health impacts arising from hydrometeorological hazards.
- 2) This research identifies many sources of HIVE data in New Zealand, and points to agencies that possess the data. This provides a pathway for stakeholders to identify data sources and partnerships required for obtaining the required data and knowledge not only for IFWs, but also for general Disaster Risk Reduction (DRR).
- 3) This research reveals a need for more effective management and governance of impact, vulnerability, and exposure data for hydrometeorological hazards in New Zealand. This would support both IFW implementation and DRR in New Zealand.
- 4) Finally, this research produced a micro-theory grounded in the qualitative data highlighting the importance of Partnerships and Collaboration for implementing IFWs. This micro-theory (Figure 9.1) also provides strategies for building and nurturing such partnerships to support stakeholders both in and outside of NZ with these efforts.

These implications are further discussed in the following sections.

10.2.1 Data Uses and Gaps for Impact Forecasts and Warnings

Research on implementing IFWs has highlighted the lack of supplemental data as a key barrier to implementing IFWs (Hemingway & Robbins, 2019; Potter et al., 2021; Wei et al., 2018). Meteorologists and forecasters have indicated that they do not possess knowledge of impacts, exposure, or vulnerability to effectively compose IFWs (Potter et al., 2021). While the literature has identified the need for impact, vulnerability, and exposure data for IFWs, it falls short in specifying where or why these data are needed. This doctoral study contributes to the discourse around the datafocused challenges of implementing IFWs, by first providing an evidence base underpinning these needs. This doctoral study then situates each data type (hazard, impact, vulnerability, and exposure) within the Warning Value Chain components, along with actors or users of these data, and provides additional justification around the need for dynamic exposure and dynamic vulnerability data. As shown in Chapter Six, HIVE data is needed throughout the Warning Value Chain. The Warning Value Chain consists of six components (Golding et al., 2019); the first three components (e.g., Observations, Monitoring, and Detection; Weather Forecasting; and Hazard Forecasting) were found to require only the hazard data that is collected and used by hydrometeorological services. These components have been well documented in the literature (Farnell et al., 2017; Kotsuki et al., 2019; Srivastava & Bhardwaj, 2013), thus findings from Chapter Six coincide with this literature and does not explore these components any further.

Findings in Chapter Six regarding the remaining three components of the Warning Value Chain (e.g., Impact Forecasting, Impact Warning, and Decision/Action) contribute to the literature by providing a broader picture of where impact, vulnerability, and exposure data are needed for IFW systems, and for what purpose(s). For example, within the Impact Forecasting component, all four types of data (hazard,

impact, vulnerability, and exposure) are required to support either a subjective approach to impact forecasting (i.e., impact-oriented discussions) or an objective approach (i.e., risk or impact modelling)³⁸. The actors in this component who possess these data and may be involved in the forecasting process were identified in Chapter Six of this doctoral research as hydrometeorological services, civil defence and emergency management, the lifelines sector, and risk modellers. Within the wider research body of risk and impact modelling (e.g., Deligne et al., 2017; Schmidt et al., 2011), this finding positions the role of risk modelling and impact-oriented discussions and associated actors and datasets within the IFW process. It recognises the value of both subjective and objective approaches and suggests that using these approaches complementarily may be the most effective use.

Within the Impact Warning component of the Warning Value Chain, Chapter Six of this doctoral research outlines the data needs for, actors involved in, and approaches for defining thresholds for impact warnings. Previous research has identified a gap in understanding the decision-making process for issuing IFWs and selecting appropriate thresholds (Harrison et al., 2014; Kox, Kempf, et al., 2018; Potter et al., 2021). This doctoral thesis sheds light on and begins to fill this gap by outlining subjective and objective approaches (like the approaches identified in the Impact Forecasting component) to setting impact thresholds, such as identifying trigger points for damage levels using damage surveys and risk models. Findings from Chapter Five suggest that some of the participatory processes of VGI, such as participatory mapping and participatory GIS may help with setting impact thresholds based on community inputs. Furthermore, crowdsourcing and social media can be used to collect information on damages and other impacts to inform thresholds. This doctoral research further points to the need for dynamic vulnerability and exposure data (Merz et al., 2020); signalling that historic impact data is not enough for defining impact thresholds when population characteristics and movement, land-use practices, and seasons change over space and time.

Within the final component of the Warning Value Chain, Decision/Action, Chapter Six of this doctoral thesis contributes to the literature by offering some contrary findings around the perceived benefits of IFWs to reduce warning fatigue and cry-wolf syndrome (Potter et al., 2021). While most participants indicated the potential for IFWs to reduce the effects of warning fatigue and cry-wolf syndrome, it is important to note that some participants, as reported in Chapter Six, have concerns about crying wolf when warning for impacts that did not eventuate. This finding highlights the compounding uncertainty of forecasting impacts in addition to forecasting hazards and points to the need for more data to verify and/or calibrate models, as well as further behavioural research to understand warning fatigue in the context of IFWs.

³⁸ It is recognised that risk models, NWP models, etc. can involve assumptions and some subjective determination of parameters (e.g., Flage & Askeland, 2020), and thus have a subjective element in their design. However, when comparing risk modelling approaches to the discussion-oriented approaches identified in this doctoral thesis, these model-based approaches are differentiated as more objective in nature than discussion-oriented approaches.

These findings further highlight the importance, and challenge, of evaluating IFWs (Potter et al., 2021; Robbins & Titley, 2018).

Findings of Chapter Six further contribute to the discussion around providing prescribed actions along with IFWs (Weyrich et al., 2018). While many participants from the EM sector reiterated the need to include prescribed actions in the warnings, one participant remained sceptical around the responsibility of providing prescribed actions to the communities within their jurisdiction, which are perceived as large and self-sufficient. This finding raises two important questions around designing and implementing IFWs: (1) what are the needs of warning recipients? and, (2) who is responsible for formulating and providing the prescribed actions? The first question echoes the need to work with communities to determine their specific warning systems needs to ensure the warnings are useful (Tarchiani et al., 2020; Thomalla et al., 2006). Findings from the literature review presented in Chapter Five suggest that Volunteered Geographic Information (VGI) is one option that may facilitate community engagement to identify warning needs for people-centred EWSs. The second question highlights the importance of clearly establishing roles and responsibilities within an IFWs. This work thus requires partnerships and collaboration such that the roles and responsibilities can be defined and allocated accordingly (Becker et al., 2017; Wein et al., 2016).

Finally, the findings from Chapter Six of this thesis highlight the need for understanding and warning for different types of impacts. Findings are consistent with previous research identifying cultural and social limitations in disaster preparedness, particularly warning response, and the need to understand the cultural contexts influencing preparedness (Ayeb-Karlsson et al., 2019). In addition to identifying gaps in modelling and warning for social and cultural impacts, this research identified a gap in NZ of warning for health impacts of certain hydrometeorological hazards, particularly severe thunderstorm-induced asthma. A previously overlooked impact of severe thunderstorms, that of asthma attacks, have caused outbreaks in both Australia and NZ, where hospitals were responding to abnormally high numbers of asthma attacks in a short period of time (Sabih et al., 2020; Thien et al., 2018). Findings from this research echo calls for capturing severe weather health impacts such as asthma attacks (Hew et al., 2017), wildfire smoke inhalation and smog (Chen et al., 2017; Cisneros & Schweizer, 2018), exposure to extreme temperatures and conditions (Astrom et al., 2014; Burgstall et al., 2019; Chen et al., 2015; Zhang et al., 2021), and mental health impacts, to build a wider view and understanding of health impacts beyond injuries and deaths. This should then be integrated into an IFW system to meet the goals of the Sendai Framework (Aitsi-Selmi & Murray, 2016). Understanding these health impacts can be met through increased partnerships with health agencies.

Research has identified confusion around who IFWs are designed for and whether they are designed for individuals or society (Potter et al., 2021). This doctoral research echoes this question, identifying differences in vulnerability and exposure to hydrometeorological hazards between urban and rural settings. This is a critical question that the global community must address.

10.2.2 Sources of hazard, impact, vulnerability, and exposure (HIVE) data in New Zealand

The literature has identified a key challenge for implementing IFWs around identifying and accessing the required data sources for IFWs (Hemingway & Robbins, 2019). While Chapter Six identified the specific needs and uses for hazard, impact, vulnerability, and exposure data in an IFW system and provide the above-mentioned contributions to the literature, Chapter Seven continued addressing this challenge by identifying sources for the required HIVE datasets both for IFWs and for general Disaster Risk Reduction (DRR). A key contribution of this doctoral research lies in identifying specific existing and potential data sources for hazards, impacts, vulnerability, and exposure. In addition to identifying these data sources, this doctoral research also identified both IFW and non-IFW uses for these datasets to allow researchers and practitioners to identify which datasets might be suitable for their needs.

In conjunction with existing literature, this doctoral research further highlights the importance and value of tacit knowledge and experience in defining impact thresholds and informing warning decisions. Tacit knowledge and experience for decision-making in EWS is not a new concept. For example, previous studies have demonstrated the use of tacit knowledge and experience for decision-making in EWSs and IFWs (Doyle, 2011; Fearnley, 2013; Potter, 2014). EM services and councils provide feedback to warning services to adjust warning thresholds according (Chapter Six; Chapter Seven; Frugis & Wasula, 2011; Kox, Lüder, et al., 2018). Forecasters and scientists also rely on experience for critical decisions involving much uncertainty around setting Volcanic Alert Levels (Fearnley, 2013; Potter, 2014), issuing flash flood and other hydrometeorological warnings (Morss et al., 2015), and developing and issuing weather forecasts (Doswell III, 2004). Much of this transferable knowledge and experience is developed through education, training, and experiencing hazardous events (Doswell III, 2004; Doyle, 2011; Fearnley, 2013; Roebber & Bosart, 1996). The findings of this doctoral research in Chapters Six and Seven add to this body of research with the NZ context of hydrometeorological hazards for IFWs, and explicitly recognise tacit knowledge and experience as an important data source for IFWs and DRR. More importantly, this doctoral research highlights the risk of losing undocumented tacit knowledge and experience due to staff turnover, etc. Thus, it is recommended that efforts are made to document this valuable resource, ideally in a format that can be used for data-driven decisionmaking, such as in the form of spatial layers. Regular exercises can also help to boost experience, facilitate knowledge transfer, and build capacity between team members, departments, and across agencies (Bhagavathula et al., 2021; Doyle, 2011; Molka-Danielsen et al., 2018; NEMA, 2009), and may help with defining and reviewing warning thresholds.

In terms of using these data sources, this doctoral research identified a critical tension between timely and trustworthy data, as officials (e.g., warning agencies, emergency managers, responders) need both timely and trustworthy data, but often have to compromise on either factor. This finding echoes the argument made by Mehta et al. (2017) and Potter et al. (2020) on the need to verify data from social media, crowdsourcing, and other publicly generated data. Furthermore, this doctoral research links the importance of being able to trust data with the need for warning services to be perceived as trustworthy (Covello & Sandman, 2001; Terti et al., 2019).

Findings from this doctoral research around strategies for developing innovative ways to collect HIVE data align with the Policy Capacity Framework (Wu et al., 2018) which outlines three levels of capacity development and implementation: individual, organisation, and systemic; and three dimensions: analytical, managerial, and political. This doctoral research found that individuals credited with driving innovative approaches possess analytical, technical, communication, and leadership skills to drive technological innovation within their sector (see Chapter Seven). Additionally, some agencies were found to possess technical and administrative capacity at the organisational level by providing and coordinating the resources needed to allow the individuals to implement their innovative solutions (see Chapter Seven). More investigation is needed to understand the policy capacities at the systemic level and political dimension within the NZ EM and severe weather warning space.

10.2.3 Management and Governance of HIVE Data

This doctoral study built further understanding around the management, acquisition, and sharing of HIVE data for severe weather hazards in New Zealand, towards supporting Sendai Framework priorities and implementing IFWs. The literature has indicated that the Emergency Management sector may possess much of the impact, vulnerability, and exposure data needed for IFWs (e.g., Kox, Lüder, et al., 2018; Potter et al., 2021). Findings from this doctoral research presented in Chapter Seven provided the evidence to support this suggestion and additional issues have been identified pertaining to the management of and access to these data, further explored in Chapter Eight. These findings corroborate those of Crawford et al. (2018), who found that CDEM Groups and Councils in NZ were not clear on who was responsible for collecting and managing risk data. Land Information New Zealand (LINZ) identified data custodians for many fundamental geospatial datasets (LINZ, 2014a), yet findings from this doctoral research show that many risk-related datasets have been overlooked in the LINZ assessment classification of fundamental geospatial datasets. To that end, this doctoral research identified a need for data governance of impact, vulnerability, and exposure data for hydrometeorological hazards. This involves identifying and appointing stewards and custodians of the relevant datasets such that the datasets can be maintained and available for further use beyond their initial purpose.

Accessing and sharing datasets for IFWs and other DRR uses is another challenge that has been identified in the literature (De Groeve, 2015; Fakhruddin, Chu, et al., 2019; Kaltenberger et al., 2020). In Chapter Eight of this doctoral research, socio-technical intervening conditions were found to inhibit these efforts, such as trust/distrust, privacy and security, and integration challenges. Some action/interaction strategies were identified to address these intervening conditions: building partnerships and ensuring privacy and security to increase trust, and implementing systematic and standardised data collection practices. These findings highlight the importance of inter-agency knowledge and data sharing both for IFWs and for general DRR efforts (Doyle & Paton, 2017; Fakhruddin, Chu, et al., 2019; Hemingway & Gunawan, 2018). The consequences of restrictive data access and lack of information sharing for DRR can lead to disastrous consequences, as was seen in the USA following the landfall of Hurricane Katrina in 2005 where agencies made duplicative evacuation plans resulting in a 24-hour delay to the evacuation, and where agencies sent survivors to overcrowded hospitals (Peled, 2011). It is possible to establish open data access and sharing for DRR. Local, provincial, and national agencies, and regional and national academic and research institutions in Argentina collaboratively established an open data platform for flood impact reduction, "with a view to socializing knowledge for early alerts" (De Giusti et al., 2016, p. 86). The findings from this doctoral research point to the need to establish similar efforts for hydrometeorological data in NZ.

10.2.4 The importance of Partnerships and Collaboration for implementing IFWs

Partnerships and Collaboration was identified as the central theme (i.e., core category) to implementing IFWs and collecting, accessing, and using the required data. The relationship of Partnerships and Collaboration with the two phenomena in this study (IFWs and HIVE Data) is represented in Figure 9.1. Sharing knowledge and data is the interface between these two phenomena because it fills the data gap identified by meteorologists in practice and in the literature for implementing IFWs (e.g., Potter et al., 2021). The idea of Partnerships and Collaboration is not new, as it has been identified by the WMO and other agencies as a necessary strategy for implementing IFWs (Harrowsmith et al., 2020; WMO, 2015). Furthermore, the UK NHP has been exemplified for its role in implementing IFWs in the UK (Hemingway & Gunawan, 2018). This study provides further empirical evidence from the NZ context of the need for building and nurturing partnerships and collaboration both for implementing IFWs and for better management of, and access to, HIVE data for IFWs and DRR in general.

Multi-organisational collaboration has many challenges that have been identified in the literature (Pennington et al., 2015; Sonnenwald, 2007). The strategies and approaches proposed in Chapter Nine can be enacted for addressing some of these challenges and for building and nurturing partnerships and collaboration in NZ towards implementing IFWs and collecting, sharing, and using the required HIVE data for IFWs and other DRR efforts. The findings of this thesis build on existing

recommendations from the IFRC guide on implementing IFWs (Harrowsmith et al., 2020) and provide more tangible and direct strategies.

This doctoral thesis contributes to the global understanding of how hydrometeorological and emergency management services can implement IFWs, by advancing the discussion around implementing IFWs as per the WMO's guidelines, and around building up disaster risk data in accordance with the Sendai Framework Priorities. An important outcome of this research is the provision of a pathway for stakeholders to identify data sources and partnerships required for implementing a severe weather IFW system.

10.3 Limitations and Future Research Directions

Several limitations of this doctoral study must be acknowledged. While the qualitative nature of data collection and analysis herein limits the generalisability of results beyond the participants, this approach offers in-depth understanding of a problem not readily available from quantitative approaches, appropriate for an exploratory study such as this (Blumer, 1969; Miles & Huberman, 1994; Patton, 2002). Furthermore, the purpose of theory-building in Grounded Theory is not to generalise, but to generate a theory with the most explanatory power for a particular set of data (Strauss & Corbin, 1998). The results of this doctoral thesis are grounded in the experience and knowledge of the participants of this study in a specific time and place (Alammar, 2018). Thus, this may be considered a local, micro-theory (Figure 9.1), rather than a formal theory (Alammar, 2018). To increase the generalisability of these results beyond the participants and beyond the NZ context, future research can be conducted to test the concepts developed in this research to a study in another area and/or amongst a different set of participants (Chametzky, 2013).

The events of and response to the COVID-19 pandemic interfered with the data collection efforts for this PhD study. As such, several key informants were unable to be recruited and participate in this study due to their involvement in the COVID-19 response, such as EM practitioners well-versed in the collection and management of Wellbeing and Welfare data. Furthermore, interview and workshop methods had to be flexible and adaptable to the uncertain and ever-changing conditions posed by the COVID-19 risks and response. While in-person interviews and workshops were preferred to facilitate high quality data collection, this was not always feasible. The first workshop coincided with COVID-19 alert level changes and associated 'lockdowns' in parts of NZ, and as such had to be made virtual. For consistency, the remaining two workshops were also held virtually. This created an opportunity to experiment with novel workshop data collection methods, such as the online whiteboard platform Mural³⁹. The results of these workshops revealed both strengths and weaknesses of running virtual workshops and provides opportunities for learning from and planning future virtual data collection methods.

³⁹ https://www.mural.co/

Chapter 10: Discussion and Conclusion

The coding process in Grounded Theory research can be described as interpretive and hence subjective in nature (Alammar, 2018). The role of the researcher is to construct concepts and categories that are both grounded in the data and 'feel right' to the researcher (Alammar, 2018). In this process, the researcher relies on their experience and knowledge of the literature and of the field to interpret the data, construct concepts, and identify relationships between the concepts (Glaser & Strauss, 1967). As a geographer with several years of experience working in the GIS field, it can be argued that the interpretation of the data was rather data-heavy featuring elements of GIS. If the data collected for this study were given to a different coder, different codes and categories, and thus, different results, would likely be produced. Broadening the topic of this PhD research when the data and participants were pointing away from the original topic of VGI for collecting impact data towards the new topic of accessing impact, vulnerability, and exposure data exemplifies my ability as a researcher to discover the story that the participants and the data want to tell. Furthermore, I ensured transparency of my methodology and results chapters (Chapters Chapter Three to Chapter Nine) by providing direct quotes from my participants; member checking the interview references, guotes, and interpretations with my participants; and practicing sensitivity towards my participants.

The findings presented in this research are focused on the core category of Partnerships and Collaboration. However, a core category of Governance was also identified (Chapter Four) and is an equally interesting and important component to this research. However, it was deemed beyond the scope of this doctoral research and identified as a research question worthy of its own future research project. For example, future research could explore how governance can support these partnerships and collaboration, and how the governance structure of NZ or other countries influence how IFWs and other EWSs are designed and implemented.

Theoretical sensitivity is a required skill employed by the Grounded Theory researcher (Glaser & Strauss, 1967), and pertains to the researcher's ability to develop meaning from the data and assign relevant concepts (Corbin & Strauss, 2008). The researcher must be able to decide which categories are important and provide explanations on the categories' relationships (Corbin & Strauss, 2008). As a novice researcher in Grounded Theory, my level of theoretical sensitivity might have limited some of the results of this research.

This study focuses on one family of hazards out of many that are present in New Zealand. Throughout the course of this research, participants and colleagues pointed to the need for multi-hazard research and multi-hazard EWSs. The choice to focus on hydrometeorological hazards for this research is justified by my own background and experience in conducting research on hydrometeorological hazards, and because IFWs have predominantly emerged from hydrometeorological hazard research. The funding for this research through the Resilience to Nature's Challenges Weather & Wildfire theme also required a hydrometeorological focus. The PhD topic/scholarship was also developed to be in alignment with the WMO WWRP HIWeather research project, which has a hydrometeorological focus. There is opportunity for exploring

how the findings from this research and the IFW research body can be transferred to other hazards and to multi-hazard approaches.

There is further opportunity to address the issues identified in this research, such as the question of roles and responsibilities and strengthening partnerships for IFWs and HIVE data collection and sharing, through the current review of the CDEM sector in NZ (i.e., the Regulatory Framework Review ('Trifecta') Programme) (NEMA, 2021b). This review involves projects for developing a new Emergency Management Act; reviewing the National CDEM Plan Order 2015 and the accompanying Guide to the National CDEM Plan 2015; and developing the National Disaster Resilience Strategy Roadmap (NEMA, 2021b).

10.4 Conclusion

Three research gaps were identified at the beginning of this doctoral research study. First, the initial gap that this research aimed to fill was that of the role of Volunteered Geographic Information (VGI) within people-centred Early Warning Systems (EWS), intending to support Impact Forecasts and Warnings (IFWs). After conducting a scoping literature review and preliminary interviews with stakeholders of hydrometeorological warning systems both outside of and in New Zealand, the second research gap was identified as a need to understand and characterise the *uses* for hazard, impact, vulnerability, and exposure (HIVE) data within an IFW system. Finally, upon developing this understanding, there was a need to identify the *sources* of HIVE data and understanding how they are *governed, accessed,* and *shared* to support IFWs. This exploratory study employed an Evolved-Straussian Grounded Theory (ES-GT) research strategy to address these gaps.

The overarching findings and results of the ES-GT research strategy were first presented in Chapter Four. Thirty-nine individuals involved in hydrometeorological warning systems and DRR within and outside of NZ were interviewed, with an additional twenty individuals that participated in three virtual workshops. Results of the ES-GT analysis identified two research phenomena underpinning this doctoral research: (1) IFW implementation, and (2) HIVE Data. Following an exploration of the first research question regarding VGI for EWSs, the results of this thesis were organised by phenomenon. Finally, the core category resulting from the ES-GT analysis was identified as Partnerships and Collaboration. The resulting micro-theory of this doctoral thesis thus integrated the core category with the two research phenomena.

The first research gap regarding understanding the role of VGI within peoplecentred EWSs was explored through a scoping literature review (Chapter Five) and preliminary interviews. The scoping literature review identified VGI as a tool for both capturing the necessary data in an EWS and for facilitating engagement to support people-centred EWSs. Interviews following this scoping literature review indicated that there was a perceived greater need to identify the sources of not just impact data, but also vulnerability and exposure data for IFWs. The second research gap was identified. The second research gap concerning understanding and characterising the role of HIVE data within an IFW system was answered by situating the role of HIVE data within the Warning Value Chain (Chapter Six) based on interviewing and conducting workshops with experts, practitioners, and researchers in the hydrometeorological warnings and disaster management space. Chapter Six identified the various actors and users of the data within the Warning Value Chain and explored how HIVE data is used across the value chain. Key findings were that two approaches exist for forecasting impacts (e.g., an objective model-based approach, and a subjective discussion-based approach) and that there is a need to further integrate social impacts of hydrometeorological hazards into IFWs.

Following this, the sources of HIVE data were identified based on interviews and workshops with its creators and users to begin addressing the second research gap (Chapter Seven). Following the ES-GT research strategy, intervening conditions were identified that may interfere with data collection efforts, such as priorities, motivation, interest within organisations, and mistrust in the data. Action/interaction strategies were proposed based on the qualitative data to address these intervening conditions. Examples of these action/interaction strategies include garnering support and buy-in from decision-makers, leadership within the field of practice to drive priorities and establishing measures for quality control and data standardisation.

Upon identifying the HIVE data sources and learning from a participant that "although we all complain about a lack of data, a lot of the time, there is quite a bit of data out there that could be used and isn't being used very well" (Risk Modelling NZ. A) subsequent research explored why these data are not being used very well (Chapter Eight). Data governance and data access and sharing were identified as important themes influencing the use of HIVE data in New Zealand. Intervening conditions that inhibit or facilitate data governance, access, and sharing were identified, along with action/interaction strategies. Intervening conditions include unclear roles and responsibilities for managing and governing the data, distrust between agencies, and data integration challenges. Action/interaction strategies to address these intervening conditions include building partnerships and establishing standards and processes for systematic data collection.

One key action/interaction strategy was found to thread throughout most themes of this doctoral research, tying the research gaps together: Partnerships and Collaboration. This category was explored in Chapter Nine, where it was integrated into the broader results of the thesis and the literature and a micro-theory was created. Sharing data and knowledge was identified as the interface between the two phenomena in this study (IFW implementation and HIVE Data), as it allows warning agencies to access and use information that they may not otherwise have but require for an IFW system. Sharing data and knowledge requires strong partnerships and collaboration. Approaches and strategies for building and nurturing partnerships were proposed to support IFW implementation and HIVE data collection, access, and sharing.

Chapter 10: Discussion and Conclusion

The chapters presented in this thesis have limitations and point towards areas for future research. This doctoral thesis contributes to the academic discussion around the steps required to implement IFW systems and provides practical guidance to warning services and researchers around opening access to and sharing HIVE data for both IFW implementation and DRR efforts. This doctoral research thus supports current and essential efforts to reduce impacts on people and communities from hydrometeorological hazards and to help report this reduction through Sendai Framework initiatives.

Appendix A: Statements of Contribution

DRC 16



STATEMENT OF CONTRIBUTION DOCTORATE WITH PUBLICATIONS/MANUSCRIPTS

We, the candidate and the candidate's Primary Supervisor, certify that all co-authors have consented to their work being included in the thesis and they have accepted the candidate's contribution as indicated below in the *Statement of Originality*.

Name of candidate:		Sara Harrison
Name/title of Primary Supervisor:		Raj Prasanna
In whi	ch chapter is the manuscript /pu	ublished work: Five
Please select one of the following thr		e options:
• The manuscript/published wo		rk is published or in press
	• Please provide the full ref Harrison, S., Potter, S., Prasan Geographic Information for peo Australasian Journal of Disaste	ference of the Research Output: ina, R., Doyle, E. E. H., & Johnston, D. (2020). Volunteered ople-centred severe weather early warning: A literature review. er and Trauma Studies, 24(1).
Ο	The manuscript is currently un	der review for publication – please indicate:
8	• The name of the journal:	
	 The percentage of the mawas contributed by the ca Describe the contribution The candidate conducted the consubsequent revisions based or 	anuscript/published work that 80.00 andidate: a that the candidate has made to the manuscript/published work: data collection and analysis, drafted the manuscript, and made n supervisors' and reviewers' feedback.
It is intended that the manuscript will be published, but it has not yet been s		ript will be published, but it has not yet been submitted to a journal
Candidate's Signature:		Digitally signed by Sara Harrison General Devices 2021.002.3 10:30:45 +12:00
Date:		23-Sep-2021
Primary Supervisor's Signature:		Raj Prasanna +1200
Date:		24-Sep-2021

This form should appear at the end of each thesis chapter/section/appendix submitted as a manuscript/ publication or collected as an appendix at the end of the thesis.

DRC 16



STATEMENT OF CONTRIBUTION DOCTORATE WITH PUBLICATIONS/MANUSCRIPTS

We, the candidate and the candidate's Primary Supervisor, certify that all co-authors have consented to their work being included in the thesis and they have accepted the candidate's contribution as indicated below in the *Statement of Originality*.

Name of candidate:	Sara Harrison
Name/title of Primary Supervisor:	Raj Prasanna
In which chapter is the manuscript /p	ublished work: Six
Please select one of the following thr	ee options:
• The manuscript/published wo	۰ rk is published or in press
• Please provide the full re Harrison, S. E., Potter, S. H., I data uses and gaps for severe Society. https://doi.org/10.117	ference of the Research Output: Prasanna, R., Doyle, E. E. H., & Johnston, D. M. (2022). Identifying the weather impact forecasts and warnings. Weather, Climate, and 5/WCAS-D-21-0093.1
The manuscript is currently u	nder review for publication – please indicate:
The name of the journal:	
 The percentage of the m was contributed by the c Describe the contribution The candidate conducted the subsequent revisions based of 	anuscript/published work that andidate: n that the candidate has made to the manuscript/published work: data collection and analysis, drafted the manuscript, and made n supervisors' and reviewers' feedback. cript will be published, but it has not yet been submitted to a journal
Candidate's Signature:	Sara Harrison
Date:	15-Feb-2022
Primary Supervisor's Signature:	Raj Prasanna Digitally signed by Raj Prasanna Diate: 2022.02.15 08:21:34
Date:	15-Feb-2022

This form should appear at the end of each thesis chapter/section/appendix submitted as a manuscript/ publication or collected as an appendix at the end of the thesis.

DRC 16



STATEMENT OF CONTRIBUTION DOCTORATE WITH PUBLICATIONS/MANUSCRIPTS

We, the candidate and the candidate's Primary Supervisor, certify that all co-authors have consented to their work being included in the thesis and they have accepted the candidate's contribution as indicated below in the *Statement of Originality*.

below in the statement of originality.		
Name of candidate:		Sara Harrison
Name/title of Primary Supervisor:		Raj Prasanna
In whi	In which chapter is the manuscript /published work: Seven	
Please	select one of the following thre	e options:
• The manuscript/published work is published or in press		rk is published or in press
	• Please provide the full ref Harrison, S. E., Potter, S. H., P where is the data?': Identifying Aotearoa New Zealand. Interna	ference of the Research Output: Prasanna, R., Doyle, E. E. H., & Johnston, D. (In Press). 'Where oh data sources for hydrometeorological impact forecasts and warnings in ational Journal of Disaster Risk Reduction.
Ο	The manuscript is currently un	der review for publication – please indicate:
	• The name of the journal:	
	• The percentage of the ma was contributed by the ca	anuscript/published work that 80.00 andidate:
	Describe the contribution	that the candidate has made to the manuscript/published work:
	The candidate conducted the or subsequent revisions based or	data collection and analysis, drafted the manuscript, and made n supervisors' and reviewers' feedback.
0	It is intended that the manusc	ript will be published, but it has not yet been submitted to a journal
Candic	late's Signature:	Digitally signed by Sam Harison Date: 0221.0922.10.2310.3045 +1200
Date:		23-Sep-2021
Primary Supervisor's Signature:		Raj Prasanna Dote: 2021.08.24 09:02.29
Date:		24-Sep-2021

This form should appear at the end of each thesis chapter/section/appendix submitted as a manuscript/ publication or collected as an appendix at the end of the thesis.

DRC 16



STATEMENT OF CONTRIBUTION DOCTORATE WITH PUBLICATIONS/MANUSCRIPTS

We, the candidate and the candidate's Primary Supervisor, certify that all co-authors have consented to their work being included in the thesis and they have accepted the candidate's contribution as indicated below in the *Statement of Originality*.

Name of candidate:		Sara Harrison
Name/title of Primary Supervisor:		Raj Prasanna
In whic	h chapter is the manuscript /pu	ublished work: Eight
Please	select one of the following thre	e options:
$oldsymbol{igo}$	The manuscript/published wo	۰ rk is published or in press
	• Please provide the full ref Harrison, S. E., Potter, S. H., P A socio-technical analysis of th vulnerability, and exposure data https://doi.org/10.1016/i.pdisas	ference of the Research Output: rasanna, R., Doyle, E. EH., & Johnston, D. (2022). 'Sharing is caring': e sharing and governing of hydrometeorological hazard, impact, a in Aotearoa New Zealand. Progress in Disaster Science. .2021.100213
Ο	The manuscript is currently un	der review for publication – please indicate:
	• The name of the journal:	
0	 The percentage of the mawas contributed by the ca Describe the contribution The candidate conducted the caubsequent revisions based or It is intended that the manusci 	anuscript/published work that andidate: I that the candidate has made to the manuscript/published work: data collection and analysis, drafted the manuscript, and made n supervisors' and reviewers' feedback. ript will be published, but it has not yet been submitted to a journal
Candid	ate's Signature:	Sara Harrison Digitally signed by Sara Harrison Date: 2022.02.15 06.01.143
Date:		15-Feb-2022
Primary Supervisor's Signature:		Raj Prasanna Date: 2022.02.15 08:22:39 +1300
Date:		15-Feb-2022

This form should appear at the end of each thesis chapter/section/appendix submitted as a manuscript/ publication or collected as an appendix at the end of the thesis.

Appendix B: Information Systems Research Paradigms

Positivism/Post-positivism

The Positivist/Post-positivist paradigm is associated with a realist ontological approach, meaning that 'reality is real' and independent of the 'knower' or the 'social actor' (Lee, 2004; Scotland, 2012). The Positivism/Post-positivism research paradigm is typically guided by an objectivist epistemology (Scotland, 2012). The early positivist paradigm was guided by the assumption that the researcher and the research subject were independent, however this was modified by the post-positivism paradigm where it is recognised that the researcher's background knowledge of theories and hypotheses can influence what is observed (Mertens, 2015). Thus, in the post-positivist paradigm, researchers must approach the research subject objectively and prevent their biases from influencing the outcomes (Guba & Lincoln, 1994; Mackenzie & Knipe, 2006; Mertens, 2015).

The approach of Positivist/Post-positivist research is described as "experimental and manipulative" (Guba & Lincoln, 1994, p. 110), suggesting that the research typically follows a quantitative approach, however, a qualitative approach can also be used (Mertens, 2015). The research is guided by hypotheses that are empirically tested and verified (Guba & Lincoln, 1994). James Scotland described the Positivist/Post-positivist methodology as consisting of "verifiable evidence sought via direct experience and observation; this often involves empirical testing, random samples, controlled variables (independent, dependent and moderator) and control groups" (Scotland, 2012, p. 10). This kind of research tends to investigate, determine, and predict the causal relationships (i.e. the 'how' and 'why') and theories developed from the data deductively (Creswell, 2014; Scotland, 2012). Examples of data collection methods include experiments, quasi-experiments, tests, surveys, and scales (Mackenzie & Knipe, 2006).

Interpretivism

Seemingly opposite to the Positivism/Post-positivism paradigm, the ontological position of the Interpretivist paradigm is a relativist one, meaning that reality is formed by the social actors and their social structure and culture, and that the world is 'meaningless' without consciousness (Lee, 2004; Lincoln & Guba, 2005; Scotland, 2012). Interpretivism is guided by a transactional and subjectivist epistemology (Guba & Lincoln, 1994; Mertens, 2015). Knowledge of a subject or object is gained through the researcher's transactional interactions with it and findings are created as the research is conducted (Guba & Lincoln, 1994; Mertens, 2015). In this sense, knowledge is subjective. It is based on personal experience and is built through participation (Scotland, 2012).

Appendix B: Information Systems Research Paradigms

The Interpretivist approach is described as "hermeneutical and dialectical" (i.e., the research is built on discussions and interpretation) (Guba & Lincoln, 1994, p. 111). The research approach is typically qualitative (Mertens, 2015). The researcher employs methods that require interaction with the participant(s) to understand their perspective and build context (Scotland, 2012). Theories are drawn from the data through induction, rather than deduction (Morgan, 2007). Example research strategies include Case Study, Phenomenology, Hermeneutics, and Ethnography (Scotland, 2012). Data collection methods include interviews, observations, document reviews, and visual data analysis (Mackenzie & Knipe, 2006; Mertens, 2015).

Critical Theory

The Critical Theory paradigm holds a historical realism ontological approach (Guba & Lincoln, 1994). This means that human experiences and cultures shape reality over time and there are multiple versions of reality based on social positioning (Guba & Lincoln, 1994; Mackenzie & Knipe, 2006; Mertens, 2015). The epistemological approach under this paradigm is similar to that of the Interpretivist, with the added dimension of applying a cultural lens and maintaining awareness of power issues (Mertens, 2015). This means that the researcher is aware of the cultural complexities and power dynamics present in the interaction (Mertens, 2015).

The Critical Theory paradigm is described as "dialogic and dialectical" (Guba & Lincoln, 1994, p. 110). The researcher and subject must develop a transactional dialogue to exchange and develop ideas, and to "transform ignorance and misapprehensions" (Guba & Lincoln, 1994, p. 110). The researchers are described as "pluralistic and evolving in their methodologies" (Mertens, 2015, p. 33). The methodology is typically designed to include diverse, marginalised groups and allow them to participate in the research (Mackenzie & Knipe, 2006; Mertens, 2015).

Pragmatism

The Pragmatic paradigm is a dissident research paradigm as it places less emphasis on ontology and epistemology, and more emphasis on the nature of human experience (Morgan, 2014). As a paradigm, Pragmatism recognises the value of the former paradigms, namely Positivism/Post-positivism and Interpretivism, in conducting social research (Morgan, 2014). Pragmatists see "no problem with asserting both that there is a single 'real world' and that all individuals have their own unique interpretations of that world" (Morgan, 2007, p. 72). Epistemologically, Pragmatists gain knowledge from inquiries and actions (Goldkuhl, 2004; Morgan, 2014). In Göran Goldkuhl's words, "Pragmatism means the recognition of the complete dialectic between knowledge and action: proper action is an action with knowledge; the right knowledge is active knowledge" (Goldkuhl, 2004, p. 24). Within the field of IS, "a Pragmatist is interested in chance and action... the research endeavour is towards knowledge, which makes a positive difference, i.e., knowledge which contributes to improvement of IS practices" (Goldkuhl, 2004, p. 20).

Given that Pragmatists place less emphasis on the ontological and epistemological approaches to research and more on action and experience, Morgan

(2007) argues that methodology is the focus as it connects "issues at the abstract level of epistemology and the mechanical level of actual methods" (p. 68).

Pragmatic research is most commonly associated with a 'mixed-methods' approach to use both quantitative and qualitative research to develop and test theories through abductive reasoning (Johnson & Onwuegbuzie, 2004; Morgan, 2007). The methodology is guided by the research question(s) and the methods are selected based on their potential to provide "useful answers" (Johnson & Onwuegbuzie, 2004, p. 18). Thus, a Pragmatic researcher must recognise that "our values and our politics are always a part of who we are and how we act", and choose what research would have the most impact, and how (Morgan, 2007, p. 70).

Participatory Inquiry

Participatory Inquiry is a fifth research paradigm with application in IS research (Breu & Peppard, 2001). The Participatory Inquiry paradigm was first proposed in 1997 by John Heron and Peter Reason who argued that the dominant paradigms described by Guba and Lincoln (Guba & Lincoln, 1994) did not account for 'experiential knowing'. The ontological approach to the Participatory Inquiry paradigm is that reality is subjective-objective and participative (Heron & Reason, 1997; Lincoln & Guba, 2005). This means that a world exists outside of the observer (it is 'objectively given'), but the observer perceives and understands the world based on their own experiences and interactions with the world (it is 'subjectively represented' in the human mind) (Breu & Peppard, 2001; Heron & Reason, 1997).

Epistemologically, the Participatory Inquiry paradigm involves four forms of 'knowing': experiential, presentational, propositional, and practical (Heron & Reason, 1997). Experiential knowledge refers to knowing through experience; knowing through direct, participative encounters (Heron & Reason, 1997). Presentational knowledge is grounded from experiential knowing which is then represented in concepts, metaphors, and stories (Breu & Peppard, 2001; Heron & Reason, 1997). Propositional knowledge is knowing "that something is the case" (Heron & Reason, 1997), and is expressed in statements and theories that arise from the research conclusions (ibid). Practical knowledge is "knowing how to do something" (Heron & Reason, 1997, p. 281), shown through skills and competencies (ibid). The researcher or 'knower' practices critical subjectivity to maintain an awareness of the four ways of knowing and of how they are interacting (Heron & Reason, 1997). Critical subjectivity ensures the quality of Participatory Inquiry research as the knowers must continue to critically examine their beliefs, assumptions, and theories in the process (Breu & Peppard, 2001).

The methodology is similar to that of the Pragmatic and Critical Theory paradigms; the methodology is designed for action and intervention, with a focus on participation and collaboration between the researcher and participants (Breu & Peppard, 2001; Heron & Reason, 1997; Lincoln & Guba, 2005). This requires that "all involved engage together in democratic dialogue as co-researchers and co-subjects" (Heron & Reason, 1997, p. 283). The central principle to this type of research is that it is done with the people, rather than on them (Heron & Reason, 1997).

Appendix C: Research Strategies

Action Research

Action Research (AR) has been widely discussed in IS literature (e.g., Baskerville & Myers, 2004; livari & Venable, 2009). It can be described as research that is "grounded in practical action, aimed at solving an immediate problem situation while carefully informing theory" (Baskerville, 1999, p. 3). The goal of AR is to simultaneously inform practice and research to create change (Baskerville & Myers, 2004; livari & Venable, 2009). Three key characteristics of AR are:

- 1. The researcher is actively involved, with expected benefits for both researcher and organization.
- 2. The knowledge obtained can be immediately applied. There is not the sense of the detached observer, but that of an active participant wishing to utilize any new knowledge based on an explicit, clear conceptual framework.
- 3. The research is a cyclical process linking theory and practice (Baskerville & Wood-Harper, 1998) (Baskerville & Wood-Harper, 1998, p. 239).

livari and Venable (2009) describe AR as "highly context dependent" (p. 2) because of the heavy involvement of a "client" to identify and address their needs. AR is an iterative process that involves collaboration between the researcher and subjects through five phases: diagnosing, action planning, action taking, evaluating, and specifying learning (Susman & Evered, 1978). However, Goldkuhl (2008) argued that the evaluation and learning phases happen throughout the process and so should be integrated into the first three phases.

For a comprehensive AR study, each phase should be completed (Susman & Evered, 1978). However, varying degrees of intervention and collaboration has led to the classification of different types of AR (Chein et al., 1948; Susman & Evered, 1978). Different data collection methods can be used, such as questionnaires, interviews, observation, etc. (Susman & Evered, 1978). This makes it a suitable research strategy under the Pragmatic paradigm.

Practical Inquiry

Practical Inquiry is rooted in the Pragmatic paradigm, and is focused on generating knowledge for general practice (Goldkuhl, 2007). Practical Inquiry has many similarities to AR, and was proposed by Goldkuhl (2008) due to criticisms around AR and its practical relevance. Thus, the primary difference between AR and practical inquiry is the contribution towards local practice versus general practice. AR, as stated earlier, is highly contextual and thus contributes knowledge for local practice whilst also contributing to the greater research community (Goldkuhl, 2008). Alternatively, practical inquiry is concerned with creating knowledge for general, practical use outside of the local practice, which also contributes to the scientific body of knowledge (Goldkuhl, 2008). AR and practical inquiry are considered the same research only when the resulting knowledge contributes to both local and general practice (Goldkuhl, 2008).

Design Science Research

Design Science Research (DSR) is an established research strategy in IS (Hevner et al., 2004). DSR "creates and evaluates [information technology] artefacts intended to solve identified organizational problems" (Hevner et al., 2004, p. 77). The leading principle of DSR is to produce an artefact to address a problem (Hevner et al., 2004). Artifacts include constructs, models, methods, instantiation, social innovations, or new resource properties (Aken, 2004; Hevner et al., 2004; Järvinen, 2007). The artefact is built by drawing from existing theories and knowledge (Peffers et al., 2007). DSR is different from AR because there is no specific client or collaboration between the researcher and client (livari & Venable, 2009). However, it can also be argued that the artefact is developed and designed for a generalised class of clients (people or organisations) that would or could use the artefact (livari & Venable, 2009).

Case Study

Case Study is the last strategy that may be applicable to this research. It is the most widely used research strategy in IS research as it enables understanding of the interactions between information technology innovations and organisational contexts (Darke et al., 1998). Case Study research is most commonly associated with qualitative research (Harrison et al., 2017), however it can be argued that it is not tied to one fixed research paradigm, but is flexible in ontology, epistemology, and methodology (Luck et al., 2006). The flexibility of Case Study research makes it a suitable strategy under the Pragmatic paradigm, especially for evaluating the consequences of choices and actions (Jacobs, 2010).

Case Study research has no specific requirements guiding it, which allows for the research design to be tailored to the research questions (Meyer, 2001). However, this flexibility also opens the research up to criticism around the methodological choices (Meyer, 2001). Thus, the researcher must be clear and open about their methodological choices (Meyer, 2001). Case studies can be single- or multi-case, with benefits and limitations for each (Darke et al., 1998). A single Case Study allows researchers to conduct in-depth investigations of a phenomena to provide "rich description and understanding" (Darke et al., 1998, p. 277). Multiple cases allow for cross-case analysis and comparison, and for investigating a phenomenon in various settings (Darke et al., 1998). Multi-case studies can strengthen research findings (Darke et al., 1998).

Appendix D: Interview Guides

Note: The interview guides were meant to be a guideline for how interviews were conducted. Due to the iterative nature of the GT research strategy, and the semistructured nature of the interview guides, the follow-up questions were asked during the interviews that were not provided on the script. Time constraints also limited how many questions could be asked, as such questions of higher priority were highlighted in yellow to ensure they were not missed. Furthermore, due to the diverse backgrounds, expertise, and experiences of the participants involved in this study, the interview guides were frequently adjusted to be more applicable to the participant in question, and to explore emergent themes that were identified during the coding process.

Ge	General Prompts:		
1.	Thank you for meeting me		
2.	Ethics & information sheet		
3.	Introduce myself and my project		
4.	Start audio recording		
PH	IASE 1 INTERVIEWS		
Pre	eliminary Interview Guide		
1.	What is your organisation's role during a severe weather event?		
2.	What is your role within your organisation?		
	a. How long have you been in this role?		
3.	What do you know about impact-based forecasting and warnings?		
	a. Does your organisation issue them?		
	b. If not, why not?		
	c. If so,		
	i. Did or does your organisation face any challenges with implementing an IBFW system? If so, what were/are they?		
	ii. What are the data needs for building impact-based forecasts and warnings (IBFW)?		
	1. How has your organisation met these needs?		
	2. What data do you need to inform your IBFW system?		
	3. What are the challenges with obtaining the required data?		
	d. What benefits do you see from impact-based forecasts and warnings?		
	e. What do you think the future of impact-based forecasting and warnings		
	is?		
4.	What data does your organisation collect about severe weather events?		
	a. What are the sources of these data?		
	b. What is the data used for?		
	c. Does your organisation collect and store impact-related data?		
	1. Il yes,		
	2. What is the impact data used for?		

3. What are the challenges in collecting and storing?

4. How could alternative data sources and processes help to
address these challenges? Which ones?
ii. If no,
1. Why not?
iii. What alternative sources of data has your organisation used or
would be interested in using more of?
1. Why or why not?
d. What data do you need to populate such a database?
i. What are you missing?
5. How can alternative data support the implementation, development, or
formulation of impact-based forecasts and warnings?
a. If so, why and how?
b. If not, why not?
6. What research would you like to see in this area?
7. Do you have any other comments or past experiences which you think might be
relevant to tell me?
PHASE 2 INTERVIEWS
National Loss Database Interview Guide
General Information:
1. What is your role within your organisation?
2. Can you (talk to me very briefly about impact databases) explain the purpose of
the national losses/impacts databases in general?
3. What is your involvement in building the national losses/impacts database?
4. Can you tell me more details about the impacts/losses database that you're
developing:
a. Timeline for completion?
D. Designated name of the database?
Furpose and Uses of the database:
5. Why is MCDEW building this database :
a. Interfued uses: Oblis & Objectives:
6 Can you please tell me about who will be contributing to and using this
database and how they will do that?
a Users - who and how?
b Contributors - who and how?
c Any others you would like to engage with but haven't yet?
Database and data characteristics:
7. Can you please describe the characteristics of the data that you will be using?
a. What data will be in the database?
i. Losses? Impacts? Hazards? Other?
ii. Fields and attributes?
iii. Evaluation/assessment?
b. Format?
i. Spatial or non-spatial? Why?
c. Scale?

- i. Spatial scale why?
- ii. Temporal scale why?
- d. Any other aspects of the data that you think are worth highlighting?
- 8. Can you describe the reasons for using the data that you're using, versus other possible sources?
- 9. Are there any other loss databases in NZ that this one will relate to?
- 10. Do you know how key contributors get their data? (sources of impact data)
 - a. Is that enough data? Is there a gap there that needs to be filled?
 - b. Would more data be useful?
 - c. Is the right type of data being used?
 - d. What do you see as potential data sources to fill gaps that currently exist?
 - e. Is crowdsourcing going to be an effective gap-filler? And if not, why not?
 - Challenges & Opportunities:
- 11. How do you think people will use this database?
 - a. I'm aware of the global movement of impact-based warnings.
 - i. Briefly, what do you understand an impact-based warning to be?
 - ii. Has MCDEM thought much about them and who would be responsible for issuing them?

12. What have been some challenges you've faced in building the database so far?

- 13. What are some of the challenges you see going forward?
- 14. What would help you to overcome some of these challenges?
- 15. If you could wave a magic wand, and have a complete database, what would be the value/influence of having it?

Wrap up:

- 16. What would help you make progress on developing, building, designing this database for its intended uses and additional uses?
- 17. Is there anything you would like to add or discuss?

Risk Modelling Interview Guide

General Information:

- 1. What is your role within your organisation?
- 2. Can you briefly share your thoughts on the purpose of the national losses/impacts databases in general?
- 3. What is your involvement in using and collecting impact data?
- 4. Can you tell me about the impacts/losses data that you use and collect? Purpose and Uses of the impact data and risk/impact modelling:
- 5. Why does your organisation collect and use impact data?
 - a. Intended uses? Goals & objectives?
- 6. Can you please tell me about the risk/impact modelling that you conduct?
 - a. What type of data do you need?
 - b. How do you get that data?
 - i. Why do you get the data this way?
 - c. What kind of reformatting of the data do you have to do, if at all?
 - d. How do you manage situations when the data isn't available?

Database and data characteristics:

7 Can you please describe the characteristics of the data that you use?	
i Lossos? Impacts? Hazards? Othor?	
i. Eosses: impacts: hazards: Other:	
ii. Fulluation/assossment?	
ni. Evaluation/assessment:	
i Spatial or non spatial? Why?	
ii Other?	
f Scale?	
i Spatial scale - why?	
ii. Temporal scale - why?	
a Any other aspects of the data that you think are worth highlighting?	
8. Can you describe the reasons for using the data that you're using versus other	
nossible sources?	
9 Is there any other impact or loss data in NZ that would help in your work?	
a Is it enough data? Is there a gap there that needs to be filled?	
b Would more data be useful?	
c. Is the right type of data being used?	
d. What do you see as potential data sources to fill gaps that currently	
exist?	
e. Is crowdsourcing going to be an effective gap-filler? And if not, why not?	
Challenges & Opportunities:	
10. I'm aware of the global movement of impact-based warnings.	
a. Briefly, what do you understand an impact-based warning to be?	
b. How do you think your work, such as risk/impact modelling, can	
contribute to impact-based warnings?	
11. What have been some challenges in collecting and using the impact data so far	
for your purposes (e.g., risk modelling)?	
12. What are some of the challenges you see going forward?	
13. What would help you to overcome some of these challenges?	
14. If you could wave a magic wand, and have a complete database, what would be	
the value/influence of having this database?	
Wrap up:	
15. What would help you make progress on developing, building, designing risk	
models and relevant data for its intended uses and additional uses?	
16. Is there anything you would like to add or discuss?	
Spatial Data Interview Guide 1	
1. How has this project progressed?	
a. Who is involved?	
b. What are the roles?	
2. Impact data - where, why, what, how, and when is it collected?	
a. How is it used?	
b. How is it stored?	
c. What are the sources of impact data?	

d. Why is it needed?

- 3. Mentioned in presentation the importance of data custodianship, what has been your experience with assigning or managing data custodianship in [region]? Who is the custodian? Why?
 - a. What have been some challenges in this regard?
 - b. "Where existing solutions may exist, attempts should be made to leverage of existing national schemas and datasets that may be hosted by another agency, e.g., Land Information New Zealand's Property Data Management Framework (PDMF) and Address Information Management System (AIMS)." (p. 192)
- 4. What have been the biggest challenges and lessons learned?
- 5. Can it be scaled nationally? Why or why not? How?
- 6. How do you see this helping warning systems, particularly impact-based forecasts and warnings?
- 7. How could or does this relate to the national loss database at [agency]?
- 8. What can help progress this project?
- 9. What is your vision or hope for this project?

NZ Severe Weather Catalogue Guide

General Information:

- 1. What is your role within your organisation?
- 2. What is your involvement in using and collecting impact data?
 - a. Why does your organisation collect impact data?
- 3. Can you tell me about your organisation's severe weather database?
- 4. Can you tell me about the standardised collection form for capturing weather impacts?
- 5. Can you describe the reasons for using the data that you're using, versus other possible sources?
- 6. Is there any other impact or loss data in NZ that would help in your work?
 - a. Is it enough data? Is there a gap there that needs to be filled?
 - b. Would more data be useful?
 - c. Is the right type of data being used?
 - d. What do you see as potential data sources to fill gaps that currently exist?
- 7. Briefly, what do you understand an impact-based warning to be?
 - a. How do you think your collection of impact data via the storms database or the standardised collection form can contribute to impact-based warnings?
- 8. What have been some challenges in collecting and using the impact data so far for your purposes?
- 9. What are some of the challenges you see going forward?
- 10. What would help you to overcome some of these challenges?
- 11. If you could wave a magic wand, and have a complete database, what would be the value/influence of having this database?
- 12. Is there anything you would like to add or discuss?

Spatial Data Interview Guide 2 1. Can you tell me a little bit about your organisation? 2. What is your role within your organisation? 3. What do you know about a national loss or impact database for New Zealand? 4. What is [organisation's] involvement in building a national losses/impacts database? 5. Can you tell me what it means to be a data custodian? Data custodianship? What is involved? 6. How does [organisation] gain access and provide access to their datasets? 7. Could you tell me about [organisation's] key data sets for resilience project? a. What is the purpose or objective of this project? What are the data sets? b. What are the data sources? i. Do you know how key contributors get their data? c. How does impact data or the national loss database fit in here, if at all? 8. How do you think people will use this database? a. I'm aware of the global movement of impact-based warnings. i. Briefly, what do you understand an impact-based warning to be? ii. How do you see the [organisaton's] project fitting into IBWs? 9. What have been some challenges you've faced in this project so far? 10. What are some of the challenges you see going forward? 11. What would help you to overcome some of these challenges? 12. If you could wave a magic wand, and have a complete database, what would be the value/influence of having it? Loss Modelling Interview Guide 1. Can you briefly provide an overview of the work that you do with regards to risk, impact, and/or loss modelling? 2. Can you briefly share your thoughts on the purpose of the national losses/impacts databases in general? 3. What does risk, impact, and loss modelling mean to you? 4. What kind of data do you need or use for your work? a. What are the various sources of your data? And resulting products? 5. What do you think is the difference between loss and impact data? 6. Can you tell me about the impacts/losses data that you use? 7. Can you please **describe** the characteristics of the data that you use? i. Losses? Impacts? Hazards? Other? ii. Fields and attributes? iii. Evaluation/assessment? h. Format? i. Spatial or non-spatial? Why? ii. Other? Scale? i. i. Spatial scale - why? ii. Temporal scale - why? Any other aspects of the data that you think are worth highlighting?

- 8. Can you describe the reasons for using the data that you're using, versus other possible sources? 9. Is there any other impact or loss data in NZ that would help in your work? a. Is it enough data? Is there a gap there that needs to be filled? b. Would more data be useful? c. Is the right type of data being used? d. What do you see as potential data sources to fill gaps that currently exist? e. Is crowdsourcing going to be an effective gap-filler? And if not, why not? 10. I'm aware of the global movement of impact-based warnings. a. Briefly, what do you understand an impact-based warning to be? b. How do you think your work, such as risk/impact modelling, can contribute to impact-based warnings? 11. What have been some challenges in collecting and using the impact data so far for your purposes (e.g. risk modelling)? 12. What are some of the challenges you see going forward? 13. What would help you to overcome some of these challenges? 14. If you could wave a magic wand, and have a complete database, what would be the influence of having this database? 15. Is there anything you would like to add or discuss? Response Agency Interview Guide Hazard information: 1. Does FENZ use any hazard forecasts or information, such as severe weather forecasts, felt reports or shake maps for earthquakes, or ashfall prediction maps? a. If yes, what is it used for? Impact information: 2. What kind of impact information do you collect? 3. Where do you collect the information from (e.g., 111 calls, social media, CDEM (follow up question if needed: do you look at impact data collected by other agencies)) a. Aside from 111 calls, do you collect any impact information from the public through crowdsourcing and social media? Why/why not? 4. What is the impact information used for? a. What are other uses of this impact information (thinking of heat mapping done of 111 calls to map tornado damage)? 5. Does the resulting impact data get stored, or shared with any other agencies? Who, and what for?
 - a. Prompt about national loss database if not brought up
 - 6. What are some challenges in collecting, managing, storing, and/or sharing impact information/data?
 - a. What would help you overcome some of these challenges?
 - b. Do you think FENZ would benefit from accessing more impact data, such as from crowdsourcing using apps, or through impact models? How so?

c. What are some of the challenges for FENZ if real-time impact information was collected from the public and made openly available on a web tool?
Lifelines Sector Interview Guide
Current warnings and warning needs:1. Can you tell me about the severe weather warnings that you receive from the MetService?
 a. Do they currently meet the lifelines groups' needs? Why or why not? b. What kind of decisions do lifelines try to make with severe weather warnings?
c. Are these warnings meeting your needs?
2. How might impact-based warnings be useful or not useful for the lifelines group sector? Why or why not?
a. Impact-based forecasting and warning systems require knowledge of exposure and vulnerability of assets to begin designing new trigger points for warnings that are based on impacts - Do you think lifelines groups could help MetService or CDEM set thresholds, such as for wind speeds, at which impacts occur?
i. How so? Why not? ii. What are the potential opportunities and challenges? b. Are lifelines in touch with CDEM?
Impact Data:
3. What kind of impact information do lifelines utilities use and collect?
a. Staff have historical knowledge
4. What is the information used for?
5. What are some challenges in collecting, managing, storing, and/or sharing impact information/data?
a. What would help you overcome some of these challenges?
b. Do you think lifelines utilities would benefit from accessing more impact data, such as from crowdsourcing using apps, or through impact models? How so?
6. Is there any potential for using this information to support the development of
an impact-based forecasting and warning system? Why or why not?
Agriculture Sector Interview Guide
Introduction:
1. Could you please tell me about [organisation] and your role with them? Impact Data:
2. What kind of impact information do [organisation] use and collect?
a. Staff have historical knowledge
3. What is the information used for?
4. What are some challenges in collecting, managing, storing, and/or sharing impact information/data?

b. What would help you overcome some of these challenges?

- c. Do you think [organisation] would benefit from accessing more impact data, such as from crowdsourcing using apps, or through impact models? How so?
- 5. Is there any potential for using this information to support the development of an impact-based forecasting and warning system? Why or why not?
 - share what we have done to connect directly with our farmers during severe weather
 - how we use geospatial technology right through our intel space,
 - barriers with connecting this information to the right agencies and how we are overcoming those

Current warnings and warning needs:

- 1. Can you tell me about the severe weather warnings that you receive from the MetService?
- 2. How might impact-based warnings be useful or not useful for the agricultural sector? Why or why not?
- 3. Impact-based forecasting and warning systems require knowledge of exposure and vulnerability of assets to begin designing new trigger points for warnings that are based on impacts Do you think [organisation] could help MetService or CDEM set thresholds, such as for wind speeds, at which impacts occur?
 - i. How so? Why not?
 - ii. What are the potential opportunities and challenges?

Data Integration and Sharing Interview Guide

- 1. Provide brief overview of reason for interviewing
 - a. There are a lot of different groups/organisations collecting different types of impact data in different ways
 - b. Some of the key challenges with using the impact data
- 2. I'm curious about any efforts in New Zealand to develop standard practices data sharing and integration in NZ, do you know of any?
- 3. What has your experience been in NZ with data sharing, integration, and interoperability? Perhaps in the context of government (policy & planning), and maybe even more specifically for disaster management.
- 4. The future of spatial data infrastructure/data sharing in New Zealand?

Weather Warnings Interview Guide

- 1. Can you please tell me about the MetServices' new warning system?
 - a. What are the thresholds for the new warning system?
- 2. For the Southland event in February this year, what was the decision-making process behind issuing the Red Warning for the Southland event?
- 3. How accurate was the warning in comparison to what occurred on the ground?
 - a. How was the warning verified? (e.g., did any observed impacts help to verify the warning? How so or why not?)
- 4. What kind of feedback (if any) did you receive from Southland with regards to warning for this event?
- 5. What are the key learnings from this event?

6.	How does the MetService's new warning system compare to impact-based
	warnings?
PH	ASE 3 INTERVIEWS
Flo	od Warning Interview Guide 1
1.	Can you please tell me about the [region's] flood warning programme?
2.	What do you think about impact-based warnings?
	a Will the new warning system include impacts? Why or why not?
	b What do you think are the challenges limitations or barriers with impact-
	based warnings?
2	What are your current warning needs?
J.	How does your toom or organisation collect impact data?
4. E	What is the impact data used for?
э. 7	Vinat is the impact data used for :
6. 7	N/h = see a second stored ?
7.	Who can access and use the data?
8.	what are your challenges with impact data?
9.	what kind of information or data do you have on flood exposure and
10.	Are you missing any of this data? If so, what are you missing?
11.	What are your big plans for the [region's] flood warning programme and what
	do you need to get there?
	Warnings & impact-based warnings:
12.	Decision-making for issuing warnings and what kind of content to include in the
	warning, any inclusion of impact information
13.	Perspectives on impact-based warnings – data needs, challenges, opportunities
14.	Current warning needs
	Impact data:
15.	How impact data is collected, used, stored, and accessed
16.	General impact data management practices, impact data sources, sharing, etc.
17.	Impact assessments
4.0	Exposure and vulnerability:
18.	Any information, knowledge, or data on exposure and vulnerability (both
	physical and social) that may support impact-based warnings (e.g., floodplains
	and flood banks, socio-economic status, etc.).
Pul	olic Health Response Interview Guide
1.	Can you please describe your role at [Hospital]?
Ζ.	vvnat is your experience with severe weather warnings in your role?
3.	How does [Hospital] receive and respond to severe weather warnings?
4.	Are there any improvements you'd like to see in the severe weather warnings
_	from a public health perspective?
5.	Can you please tell me about severe thunderstorm induced asthma and the
,	a. Vvas the hospital well-prepared for this event?
6.	Is this the first observance of severe thunderstorm asthma in New Zealand?
_	a. vvhy do you think this hasn't this been observed before or after?
1.	How was the diagnosis of thunderstorm asthma determined?

8. Was CDEM notified?	
9. What would you like to see improved in terms of severe weather communication	
from the public health perspective?	
Risk Modelling Interview Guide 2	
Collecting, using, storing, and sharing impact data for risk and impact modelling	
and other uses	
1. What have been some challenges in collecting and using the impact data so far	
for your purposes?	
Vulnerability and exposure	
2. How does vulnerability and exposure fit into risk and impact modelling?	
3. What are potential data sources of vulnerability and exposure?	
4. Can you please tell me about dynamic exposure ?	
5. What are the challenges with capturing exposure and vulnerability?	
6. What do you think might help to overcome these challenges?	
Defining thresholds for impact-based warnings, and the potential role of	
vulnerability curves for this.	
7. Do you have any thoughts on using vulnerability curves for setting thresholds for	
impact-based warnings?	
8. What are some of the challenges you see going forward?	
9. What would help you to overcome some of these challenges?	
NZ Historic Weather Events Catalogue	
TO. How useful is this catalogue for fisk modelling?	
 If not useful, why not? 11 If you could wave a manipulate database, what would like the second second base and have a complete database, what would like the second s	
the value/influence of having this database?	
Data Science Interview Guided	
1. Can you please tell me about [organisation]?	
2. What is your role at [organisation]?	
3. What role do you think [organisation] has in Disaster Risk Reduction in New	
Zealand?	
a. For example, what has been [organisation's] role during COVID-19?	
b. Can the data from your organisation data be used for risk and impact	
modelling? How so?	
4. How do you think the work at [organisation] can support early warning systems,	
for example, severe weather warnings?	
5. How can agencies (e.g., Civil Defence Groups, councils, etc.) access your	
services and/or products?	
6. What do you think the future is for big data in disaster risk reduction?	
a. What are some perceived challenges and opportunities?	
Risk Modelling and Flood Warning Interview Guide	
1. Can you please tell me about the project?	
2. Is impact information included in the warnings?	
- 4. What have been challenges with the impact forecasting? How have you overcome them?
- 5. How were the warning thresholds defined?
 - What data/information were these based off of?
- 6. From the challenges that you listed in your email, could you please expand on the user needs/requirements challenge and how you addressed that?
 - Likewise for:
 - Assets data
 - Exposure, damage/fragility or loss functions?
- 7. Ongoing investment in maintenance/improvements to data quality?
- 8. What are your thoughts on real-time IBFWs? What do you think the future is for IBFWs, and real-time IBFWs in the Pacific Islands context and the NZ context?
- 9. Where does crowdsourcing fit in?

Weather Service Interview Guide

- 1. Can you please tell me about your organisation?
- 2. Where do you get your data from?
- 3. Do you have any issues with getting data?
- 4. Do you have any challenges with verification?
- 5. The World Meteorological Organisation is pushing for member nations to issue impact-based warnings, which aim to communicate not only what the weather will be, but also what it will do to give extra meaning to the warning audiences. Do you know much about IBWs?
 - a. Do you give advice about impacts?
 - b. Do you gather any information or data on impacts?
 - i. If so, what do you do with that data or information?
- 6. For impact-based warnings, we need to know more about exposure and vulnerability of people and assets. Do you have any data or knowledge on vulnerability and exposure?
 - a. If so, what do you do with it?
- 7. Whose responsibility do you think it is to warn about severe weather in New Zealand and give advice about impacts?
- 8. There's a global movement towards building stronger public, private, and academic partnerships for weather forecasting and warning, what are your thoughts on public, private, and academic partnerships?
- 9. Do you have any other comments or questions?

National Governance Interview Guide

- 1. Do you know anything about Impact-based warnings?
- 2. The WMO is pushing for member nations to implement impact-based warnings for severe weather hazards; and the MetService is aware that they need to integrate impact-oriented or impact-based messaging, and potentially thresholds, into their warnings. But the challenge is that impact-based forecasts and warnings require an understanding of the underlying vulnerability and exposure of people and assets. I've spoken with the MetService and a variety of civil defence groups and council hydrologists and found that the MetService don't have the knowledge or data on exposure and vulnerability, and thus don't

think it's their responsibility to build up that knowledge base and integrate it into their warnings. Meanwhile, CDEM Groups and councils may possess this knowledge and data but have indicated that they're not responsible for issuing severe weather warnings; rather they usually just pass the warnings on from the MetService.

- a. Whose responsibility do you think it would be in NZ to meld vulnerability and exposure together for severe weather impact-based forecasts and warnings?
- 3. As said earlier, I've seen that it's the MetService's responsibility to issue warnings, but I've noticed other agencies doing that as well. What are your thoughts on that?
 - a. What are some potential implications for that?
- 4. A key theme for implementing severe weather impact-based forecasts and warnings is that of building strong working relationships and partnerships with agencies that possess the various knowledge and data needed for the warning system. For example, the Natural Hazards Partnership was established in the UK consisting of 17 public sector organizations and government departments involved in monitoring, forecasting and warning for their special areas of interest (I can pass on a paper about this partnership to you if you haven't seen it yet). This partnership provides a structured forum for exchanging knowledge, information, data, expertise, etc. for improving the delivery of warning services.
 - a. How do you think we can strengthen partnerships and collaboration in NZ towards a similar aim? What is [organisation's] role in that?
- 5. Science advice groups seem to be popping up for various hazards like volcanos, tsunami, and I heard one might be forming for landslips... Do you have any thoughts on a weather and floods one?
- 6. Building onto this idea of partnerships and collaboration, in some countries, like the UK and China, these partnerships have facilitated the cohabitation of agencies, where, for example, meteorologists and hydrologists sit together in the same room, enabling more direct and efficient communication. Do you think that's a possibility in NZ? Why or why not?

If we have time:

- 7. I chatted with someone from [organisation] in late 2018 about the national loss database for Sendai Reporting and the technical ins and outs of the database and the status of it. But I wanted to ask you about the governance perspective of this database, particularly whose responsibility is it to provide data to the database?
- 8. Could you please clarify who is responsible for issuing urban and coastal flood warnings in NZ?

Flood Warning Interview Guide 2

1. I've had a look at the National Civil Defence and Emergency Management Plan and what it says about floods, and it doesn't appear to define the different type of floods nor does it specify warning responsibilities. The Ministry for the Environment website says that "Regional councils issue flood warnings and work with district and city councils to let people know that floods are on the way. If a flood is severe and widespread your regional council may declare an emergency for all or part of the region and coordinate a response with your district/city council." (Ministry for the Environment, 2016, para. 16). The [organisation's] Group Plan aligns with this by outlining the [Council] as the agency responsible for issuing flood warnings, but neither of these sources specify the type of flood.

- a. Could you please walk me through who is responsible for issuing urban floods and coastal floods? Are they treated differently from river floods?
- 2. What constitutes an urban flood?
 - a. What are the thresholds and where is the boundary of urban flood?
- 3. Could you please tell me about the Significant Wave Hazards:
 - a. how has that project been going?
 - b. What have been some challenges with it?
 - c. What challenges still need to be addressed for it?
 - d. How has this new warning system performed over the last year?

Appendix E: Workshop Agenda and Activities

Auckland and Southland Workshops

- 1. Introduction
 - a. Purpose of workshop, ethics, and recording
 - b. Ice breaker and introduction to Mural
 - c. Introduction to the project
- 2. Present the impact framework
- 3. Activity 1: Initial Impressions
 - a. What are your general thoughts on the framework?
 - b. What do you like?
 - c. What do you dislike?
- 4. Activity 2: Data Requirements and Sources
 - a. What data do we need for Impact-based Forecasts and Warnings?
 - b. What do we have already?
 - c. What's missing from the list in the blue box in the framework?
 - d. What are some impediments and facilitators to accessing these data?

	We Need	We have	The list is missing	Access impediments	Access facilitators
Hazards	-put sticky notes here-				
Exposure					
Vulnerability					
Impacts					

- 5. Activity 3: System and Data Life
 - a. Please pick 2-3 diverse datasets/datasources that are important and identify the life track of these data

	Name of the dataset	How is it collected / created?	Where does the data go?	What is the data used for?	What happens to the data after its initial use?
Data 1					
Data 2					

b. How do you understand what impacts are occurring or could occur?

I/we understand impacts that are	I/we understand impacts that could		
occurring by	occur by		

- 6. Activity 4: Outcomes
 - a. Are there any examples of how you use the data identified previously?

7. Activity 5: Application

- a. How would you use or apply this framework?
- b. Are there any aspects that you would adapt or change for your use?

Aspects that I would change	How I would change it and why

- c. Do you have any thoughts on whether you would use impact-based forecasts and warnings?
 - i. Why or why not?

Why?	Why not?

GNS Workshop

1. Present the impact framework



- 2. Please help me strengthen the links along the dotted lines in the framework:
 - a. How do we see risk modelling supporting or informing impact[-based] forecasts and warnings?
 - b. How can risk modelling and impact forecasts be used to help define impact[-based] thresholds for impact[-based] warnings?
 - c. What are the challenges or limitations to using risk modelling for impact[-based] forecasts and warnings?
 - What is needed to overcome or address these challenges/limitations?
- 3. Share your thoughts on sticky notes directly on the framework in the Mural

Appendix F: Human Ethics Notifications, Information Sheets, and Consent Forms

Human Ethics Notification - 4000019769

humanethics@massey.ac.nz <humanethics@massey.ac.nz> To: Sara.Harrison. Cc: humanethics@massey.ac.nz Tue, Jul 3, 2018 at 10:36 AM

c. numaneuncs@massey.a

HoU Review Group A/Pro Ross Flett

Ethics Notification Number: 4000019769 Title: Contextual Interviews with emergency management practitioners and meteorologists

Thank you for your notification which you have assessed as Low Risk.

Your project has been recorded in our system which is reported in the Annual Report of the Massey University Human Ethics Committee.

The low risk notification for this project is valid for a maximum of three years.

Please note that travel undertaken by students must be approved by the supervisor and the relevant Pro Vice-Chancellor and be in accordance with the Policy and Procedures for Course-Related Student Travel Overseas. In addition, the supervisor must advise the University's Insurance Officer.

A reminder to include the following statement on all public documents:

"This project has been evaluated by peer review and judged to be low risk. Consequently it has not been reviewed by one of the University's Human Ethics Committees. The researcher(s) named in this document are responsible for the ethical conduct of this research.

If you have any concerns about the conduct of this research that you want to raise with someone other than the researcher(s), please contact Professor Craig Johnson, Director (Research Ethics), email humanethics@massey.ac.nz. "

Please note that if a sponsoring organisation, funding authority or a journal in which you wish to publish require evidence of committee approval (with an approval number), you will have to complete the application form again answering yes to the publication question to provide more information to go before one of the University's Human Ethics Committees. You should also note that such an approval can only be provided prior to the commencement of the research.

You are reminded that staff researchers and supervisors are fully responsible for ensuring that the information in the low risk notification has met the requirements and guidelines for submission of a low risk notification.

If you wish to print an official copy of this letter, please login to the RIMS system, and under the Reporting section, View Reports you will find a link to run the LR Report.

Yours sincerely

Professor Craig Johnson Chair, Human Ethics Chairs' Committee and Director (Research Ethics)

Human Ethics Notification - 4000021843

humanethics@massey.ac.nz <humanethics@massey.ac.nz> R.A.Flett To: Sara.Harrison R.Prasanna

Fri, Oct 11, 2019 at 2:28 PM

Cc: humanethics@massey.ac.nz

HoU Review Group A/Pro Ross Flett

Ethics Notification Number: 4000021843

Title: Data Needs for Impact-based forecasting and warning systems in New Zealand: Exploring the production, sharing and management of impact data from severe weather events

Thank you for your notification which you have assessed as Low Risk.

Your project has been recorded in our system which is reported in the Annual Report of the Massey University Human Ethics Committee.

The low risk notification for this project is valid for a maximum of three years.

Please note that travel undertaken by students must be approved by the supervisor and the relevant Pro Vice-Chancellor and be in accordance with the Policy and Procedures for Course-Related Student Travel Overseas. In addition, the supervisor must advise the University's Insurance Officer.

A reminder to include the following statement on all public documents:

"This project has been evaluated by peer review and judged to be low risk. Consequently it has not been reviewed by one of the University's Human Ethics Committees. The researcher(s) named in this document are responsible for the ethical conduct of this research.

If you have any concerns about the conduct of this research that you want to raise with someone other than the researcher(s), please contact Professor Craig Johnson, Director (Research Ethics), email humanethics@massey.ac.nz.

Please note that if a sponsoring organisation, funding authority or a journal in which you wish to publish require evidence of committee approval (with an approval number), you will have to complete the application form again answering yes to the publication question to provide more information to go before one of the University's Human Ethics Committees. You should also note that such an approval can only be provided prior to the commencement of the research.

You are reminded that staff researchers and supervisors are fully responsible for ensuring that the information in the low risk notification has met the requirements and guidelines for submission of a low risk notification.

If you wish to print an official copy of this letter, please login to the RIMS system, and under the Reporting section, View Reports you will find a link to run the LR Report.

Yours sincerely

Professor Craig Johnson Chair, Human Ethics Chairs' Committee and Director (Research Ethics)

R.A.Flett

Human Ethics Notification - 4000022328

humanethics@massey.ac.nz <humanethics@massey.ac.nz> To: Sara.Harrison for an and the second s Wed, Jul 22, 2020 at 5:55 PM

HoU Review Group A/Pro Ross Flett

Ethics Notification Number: 4000022328 Title: Impact data for severe weather hazards and events in New Zealand

Thank you for your notification which you have assessed as Low Risk.

Your project has been recorded in our system which is reported in the Annual Report of the Massey University Human Ethics Committee.

The low risk notification for this project is valid for a maximum of three years.

Please note that travel undertaken by students must be approved by the supervisor and the relevant Pro Vice-Chancellor and be in accordance with the Policy and Procedures for Course-Related Student Travel Overseas. In addition, the supervisor must advise the University's Insurance Officer.

A reminder to include the following statement on all public documents:

"This project has been evaluated by peer review and judged to be low risk. Consequently it has not been reviewed by one of the University's Human Ethics Committees. The researcher(s) named in this document are responsible for the ethical conduct of this research.

If you have any concerns about the conduct of this research that you want to raise with someone other than the researcher(s), please contact Professor Craig Johnson, Director (Research Ethics), email humanethics@massey.ac.nz. "

Please note that if a sponsoring organisation, funding authority or a journal in which you wish to publish require evidence of committee approval (with an approval number), you will have to complete the application form again answering yes to the publication question to provide more information to go before one of the University's Human Ethics Committees. You should also note that such an approval can only be provided prior to the commencement of the research.

You are reminded that staff researchers and supervisors are fully responsible for ensuring that the information in the low risk notification has met the requirements and guidelines for submission of a low risk notification.

If you wish to print an official copy of this letter, please login to the RIMS system, and under the Reporting section, View Reports you will find a link to run the LR Report.

Yours sincerely

Professor Craig Johnson Chair, Human Ethics Chairs' Committee and Director (Research Ethics)





Data Needs for Impact-based forecasting and warning systems in New Zealand: Exploring the production, sharing and management of impact data from severe weather events

INFORMATION SHEET

Research Information

This research contributes to Sara Harrison's PhD through Massey University. There is a growing need for the standardised collection and storage of impact data post-disaster. The importance of systematically recording, sharing, and publicly accounting for disaster losses and impacts is threaded throughout the Sendai Framework for Disaster Risk Reduction 2015-2030. Moreover, the World Meteorological Organization (WMO) has pushed for the implementation of impact-based forecasting and warning systems, furthering the need to systematically collect, store, and share impact data. Yet, methods for collecting the required impact data differ by country and region, making standardised collection and sharing difficult.

This exploratory PhD research, supported by the HIWeather Project within the WMO's World Weather Research Programme and funded by the Resilience to Nature's Challenges Phase 2 Weather & Wildfire Theme, aims to map out existing and potential impact data sources from severe weather events in New Zealand. The expected outcome of this research is a framework outlining the process of getting impact data from the source (e.g. the public) to the end-users (e.g. Civil Defence groups, the MetService, impact/risk modellers, etc.) for impact-based forecasts and warnings.

Results may be published in scientific journals and other publications. This research will contribute to the implementation of impact-based forecasts and warnings in New Zealand while also supporting efforts towards meeting the requirements of the Sendai Framework to build a national impacts and losses database.

Invitation to Participate

You are cordially invited to participate in this study through a qualitative interview in which Sara will ask a number of questions around the collection, storage, and use of impact data, as well as the process of implementing impact-based forecasting and warning systems in New Zealand. The purpose of these interviews is to help build an understanding of the current and potential uses and sources of impact data from severe weather events in New Zealand. It is envisaged that the interview will take one hour (depending on your availability).

Participant Identification and Recruitment

- Participants will be selected and recruited based on their position and role in risk communication
 and warning systems for severe weather hazards and in handling impact data including, but not
 limited to, officials from national hydrometeorological services, national and local civil defence both
 within and outside of New Zealand.
- Participant names and contact information will be obtained from publicly available information such as organization web pages, as well as key contacts within Sara's research network.
- There are no known or anticipated risks to you as a participant in this study.

Page 1 of 2

Research Procedure

With your permission, the interview will be audio recorded to facilitate collection of information, and later transcribed for analysis. All information you provide is considered completely confidential. Your name will not appear in any thesis or report resulting from this study, however, with your permission, anonymous quotations may be used. Interview transcripts will be anonymised to protect your confidentiality.

Data Management

At the conclusion of the research, the data will be stored in a secure location at the Joint Centre for Disaster Research, which is part of the School of Psychology at Massey University, Wellington, New Zealand. All data collected will be held in this secure location for five years post research completion. The only persons with access to this data will be the current research team.

Participant's Rights

Whilst your contribution would be most appreciated, you are under no obligation to participate in this research. Following the Massey University Code of Ethical Conduct, if you choose to participate you have the right to:

- decline to answer any particular question;
- withdraw from the study at any time;
- · ask any questions about the study at any time during participation;
- provide information on the understanding that your name will not be used unless you give permission to the researcher;
- be given access to a summary of the project findings when it is concluded;
- ask for the recorder to be turned off at any time during the interview.

Information about the Researchers

Please feel free to contact one or all of the following people if you have questions about the research being undertaken.

Sara Harrison, PhD Candidate, Joint Centre for Disaster Research, Massey University, Wellington, New Zealand.

Email: herricon@massery propphone: +o+ (o+) corror of the content of the content

Supervisors:

Email: 💻

Dr. Sally Potter, Hazard and Risk Management Researcher, GNS Science, Lower Hutt, New Zealand. Email: _______ phone: ______

Dr. Raj Prasanna, Senior Lecturer, Joint Centre for Disaster Research, Massey University, Wellington, New Zealand. Email: Email:

Email: _____phone: _____phone: _____phone: ______phone: _____phone: ______phone: ______phone: ______phone: ______phone: _____phone: ____phone: ___

phone:

Prof. David Johnston, Director/Professor of Disaster Management, Joint Centre for Disaster Research, Massey University, Wellington, New Zealand. Email:

This project has been evaluated by peer review and judged to be low risk. Consequently, it has not been reviewed by one of the University's Human Ethics Committees. The researcher(s) named above are responsible for the ethical conduct of this research.

If you have any concerns about the conduct of this research that you wish to raise with someone other than the researcher(s), please contact Prof Craig Johnson, Director, Research Ethics, telephone 06 356 9099 x 85271, email <u>humanethics@massey.ac.nz</u>.

Page 2 of 2





Data Needs for Impact-based forecasting and warning systems in New Zealand: Exploring the production, sharing and management of impact data from severe weather events

PARTICIPANT CONSENT FORM

I have read, or have had read to me in my first language, and I understand the Information Sheet attached. I have had the details of the study explained to me, any questions I had have been answered to my satisfaction, and I understand that I may ask further questions at any time. I have been given sufficient time to consider whether to participate in this study and I understand participation is voluntary and that I may withdraw from the study at any time.

Declaration by Participant:

- 1. I agree / do not agree (circle) to the interview being audio recorded.
- 2. I wish / do not wish (circle) to have my recordings returned to me.
- I agree / do not agree (circle) to the use of anonymous quotations in any thesis or publication that comes of this research.
- I agree / do not agree (circle) to participate in this study under the conditions set out in the Information Sheet.

I	_ hereby consent to take part in this study.			
Signature:	Date:			
Findings:				
I wish to be sent a copy of the findings from this research - Yes / No (Circle)				
If yes, please include your email here:				





Data Needs for Impact-based forecasting and warning systems in New Zealand: Exploring the production, sharing and management of impact data from severe weather events

INFORMATION SHEET

Research Information

This research contributes to Sara Harrison's PhD through Massey University. There is a growing need for the standardised collection and storage of impact data post-disaster. The importance of systematically recording, sharing, and publicly accounting for disaster losses and impacts is threaded throughout the Sendai Framework for Disaster Risk Reduction 2015-2030. Moreover, the World Meteorological Organization (WMO) has pushed for the implementation of impact-based forecasting and warning systems, furthering the need to systematically collect, store, and share impact data. Yet, methods for collecting the required impact data differ by country and region, making standardised collection and sharing difficult.

This exploratory PhD research, supported by the HIWeather Project within the WMO's World Weather Research Programme and funded by the Resilience to Nature's Challenges Phase 2 Weather & Wildfire Theme, aims to map out existing and potential impact data sources from severe weather events in New Zealand. The expected outcome of this research is a framework outlining the process of getting impact data from the source (e.g. the public) to the end-users (e.g. Civil Defence groups, the MetService, impact/risk modellers, etc.) for impact-based forecasts and warnings.

Results may be published in scientific journals and other publications. This research will contribute to the implementation of impact-based forecasts and warnings in New Zealand while also supporting efforts towards meeting the requirements of the Sendai Framework to build a national impacts and losses database.

Invitation to Participate

You are cordially invited to participate in this study through a workshop in which Sara will ask a number of questions around the collection, storage, and use of impact data, as well as the process of implementing impact-based forecasting and warning systems in New Zealand. The purpose of these workshops is to help build an understanding of the current and potential uses and sources of impact data from severe weather events in New Zealand. It is envisaged that the workshop will take two hours.

Participant Identification and Recruitment

- Participants will be selected and recruited based on their position and role in risk communication
 and warning systems for severe weather hazards and in handling impact data including, but not
 limited to, officials from national and local hydrometeorological services, national and local civil
 defence both within and outside of New Zealand.
- Participant names and contact information will be obtained from publicly available information such as organization web pages, as well as key contacts within Sara's research network.
- · There are no known or anticipated risks to you as a participant in this study.

Page 1 of 2

Research Procedure

With your permission, the workshop will be recorded to facilitate collection of information, and later transcribed for analysis. All information you provide is considered completely confidential. Your name will not appear in any thesis or report resulting from this study, however, with your permission, anonymous quotations may be used. Transcripts will be anonymised to protect your confidentiality.

Data Management

At the conclusion of the research, the data will be stored in a secure location at the Joint Centre for Disaster Research, which is part of the School of Psychology at Massey University, Wellington, New Zealand. All data collected will be held in this secure location for five years post research completion. The only persons with access to this data will be the current research team.

Participant's Rights

Whilst your contribution would be most appreciated, you are under no obligation to participate in this research. Following the Massey University Code of Ethical Conduct, if you choose to participate you have the right to:

- decline to answer any particular question;
- withdraw from the study at any time;
- ask any questions about the study at any time during participation;
- provide information on the understanding that your name will not be used unless you give permission to the researcher;
- be given access to a summary of the project findings when it is concluded;
- · ask for the recorder to be turned off at any time during the interview.

Information about the Researchers

Please feel free to contact one or all of the following people if you have questions about the research being undertaken.

Sara Harrison, PhD Candidate, Joint Centre for Disaster Research, Massey University, Wellington, New Zealand.

Email: phone: +01(ct) == 5700 ovt-00050

Supervisors:

Dr. Raj Prasanna, Senior Lecturer, Joint Centre for Disaster Research, Massey University, Wellington, New Zealand.

Email: phone: The second second

Dr. Emma Hudson-Doyle, Lecturer, Joint Centre for Disaster Research, Massey University, Wellington, New Zealand.

Email: Contraction Concept as no, phone: Children Concept Conc

Prof. David Johnston, Director/Professor of Disaster Management, Joint Centre for Disaster Research, Massey University, Wellington, New Zealand. Email: Emai

This project has been evaluated by peer review and judged to be low risk. Consequently, it has not been reviewed by one of the University's Human Ethics Committees. The researcher(s) named above are responsible for the ethical conduct of this research.

If you have any concerns about the conduct of this research that you wish to raise with someone other than the researcher(s), please contact Prof Craig Johnson, Director, Research Ethics, telephone 06 356 9099 x 85271, email <u>humanethics@massey.ac.nz</u>.

Page 2 of 2





Data Needs for Impact-based forecasting and warning systems in New Zealand: Exploring the production, sharing and management of impact data from severe weather events

PARTICIPANT CONSENT FORM

I have read, or have had read to me in my first language, and I understand the Information Sheet attached. I have had the details of the study explained to me, any questions I had have been answered to my satisfaction, and I understand that I may ask further questions at any time. I have been given sufficient time to consider whether to participate in this study and I understand participation is voluntary and that I may withdraw from the study at any time.

Declaration by Participant:

- 1. I agree / do not agree (circle) to the workshop being recorded.
- 2. I wish / do not wish (circle) to have my recordings returned to me.
- I agree / do not agree (circle) to the use of anonymous quotations in any thesis or publication that comes of this research.
- I agree / do not agree (circle) to participate in this study under the conditions set out in the Information Sheet.

I		hereby consent to take part in this study.

Signature:	Date:
------------	-------

Findings:

I wish to be sent a copy of the findings from this research - Yes / No (Circle)

If yes, please include your email here:

Appendix G: Auckland and Southland Hazard Profiles

Auckland Profile

Auckland Region was selected because of notable historic severe weather events such as the 2017 New Lynn storms and consequent flood, and the 2018 windstorm (Golubiewski, 2019; Smol, 2018). Auckland is a large metropolitan city in the North Island of NZ; thus, it offers an urban context for the case analysis.

In 2018, the population of Auckland was 1.57 million and growing, making up 32% of New Zealand's population (Stats NZ, 2018). Thirty percent of the region is considered urban, with the remaining 70% rural (Auckland Emergency Management, 2016); however, 90% of Auckland's population lives in urban areas (Auckland Emergency Management, 2016). Auckland is the largest urban centre in New Zealand, located on the northern coast of the North Island, and makes up 2% of New Zealand's landmass (Auckland Emergency Management, 2016). Auckland Region is made up of harbours, mountain ranges, islands, lakes, and streams (Auckland Emergency Management, 2016). In addition to Auckland's central urban centre, the region also has isolated communities, such as on islands (Auckland Emergency Management, 2016). Auckland is exposed to several hazards, namely volcanic and meteorological hazards. Auckland lies on an active volcanic field containing at least 50 volcanoes (Auckland Civil Defence, 2016).

Meteorological Hazards

The weather in and around Auckland can change quickly, with weather systems originating from the Tasman Sea during summer and autumn, including subtropical storms and ex-tropical cyclones (Auckland Emergency Management, 2016). Floods are the most common weather hazard in Auckland, followed by damaging winds and rough seas (Auckland Emergency Management, 2016). A 'superstorm' (i.e., a storm system that incorporates numerous hazards) poses a significant risk to Auckland, as a combination of severe winds, heavy rain, and land instability; and flood has the potential to produce power outages, storm surge, and coastal erosion (Auckland Emergency Management, 2016). A regional risk assessment for Auckland identified four meteorological hazards as the highest priority for risk management, out of five: 1) coastal inundation (storm surge due to severe weather), 2) flooding (river and catchment due to heavy rainfall), 3) infectious human disease pandemics, 4) severe winds, and 5) severe storms (i.e., "super storm" (Auckland Emergency Management, 2016, p. 58)).

Warnings

The warning setup in Auckland is like that of the rest of New Zealand, relying on NEMA for national warnings, and the MetService for weather monitoring and issuing subsequent watches, warnings, and alerts (Auckland Emergency Management, 2016). The Auckland Council is the primary source of flood information, and is responsible for monitoring and issuing flood warnings in Auckland Region (Auckland Emergency Management, 2016). The Auckland Council uses a network of remote

stations to monitor rainfall, river flows, and lake levels, which is used for forecasting flood (Auckland Emergency Management, 2016). When the respective levels reach a predetermined threshold (or warning level), notifications are sent to stakeholders (Auckland Emergency Management, 2016).

The MetService is the primary agency responsible for issuing severe weather watches, warnings, and alerts, and the Auckland Civil Defence Group offers support for disseminating the Auckland-oriented alerts, and adds additional information to the warning, such as recommended protective actions (which includes advice that reflects guidelines developed by NEMA) (EM. NZ. Reg. D, E). In a recent interview, officials from the Auckland EM Group indicated that they are moving towards using impact-oriented language in their messages (EM. NZ. Reg. D, E).

Southland Profile

Southland Region was selected for the second regional workshop. In February 2020, Southland experienced heavy rainfall, floods, and landslides. During this event, the MetService issued the first Red Warning since the implementation of its new national severe weather warning system (See Chapter Two Section 2.4). Southland is the southernmost region of NZ, located on the southwestern portion of NZ's South Island. Southland is considered a remote region, as such it provides a rural context.

Southland Region encapsulates 13% of NZ's land area, making it the country's second largest region by area (Emergency Management Southland, 2017). Southland has a "unique and rugged" landscape comprised of mountain ranges, forested wilderness, fiords, and rivers (Emergency Management Southland, 2017). Despite taking up such a large land area, the population of Southland makes up only 2% of NZ's population; having just over 97,400 residents. Unlike Auckland, Southland does not have any significant urban centres, rather it has several smaller cities and numerous rural communities. Invercargill City has over half of Southland Region's population, with 51,696 residents (Emergency Management Southland, 2017). The second largest urban area is Gore, with 7,356 residents (Emergency Management Southland, 2017). The sparse spread of Southland Region's population, along with the remoteness of the settlements and communities in the Region's rugged landscape increase the potential for people to become quickly isolated in emergencies (Emergency Management Southland, 2017). As such, it is of high importance for these communities to be self-reliant and prepared (Emergency Management Southland, 2017). Southland's landscape and geographical location presents many hazards and associated risks; the most "extreme" risk being tsunami (specifically landslide induced tsunami) and earthquake (Emergency Management Southland, 2017, p. 23). Severe weather also presents "very high" risk to the region, and with climate change, it is expected that risks from these hazards will increase (Emergency Management Southland, 2017, p. 23).

Meteorological Hazards

Southland is exposed to weather systems moving into the region from the west and from the south (Emergency Management Southland, 2017). Gale force winds occur frequently, but rarely cause damage (Emergency Management Southland, 2017). Fiordland, an area of Southland with few permanent residents but is a large tourist attraction, receives over 8,000mm of rain per year (Emergency Management Southland, 2017). Alternatively, Southland's lowlands and hills receive 800 to 1,200mm of rain per year (Emergency Management Southland, 2017). Most of Southland's population lives on floodplains, putting these communities at significant risk to flood impacts, despite extensive flood mitigation work (Emergency Management Southland, 2017). This risk became a reality in February 2020, when residents of the low-lying areas of Gore, Wyndham, and Mataura were evacuated during a flood emergency (Quinlivan et al., 2020). This event led to the MetService issuing its first Red Warning since the implementation of the new warning system in May 2019 (MetService, 2019a).

Warnings

The warning system in Southland Region is similar to that of Auckland, whereby NEMA is the authoritative agency for issuing national warning messages (Emergency Management Southland, 2017). Emergency Management Southland (EMS) is the point of contact for receiving and disseminating all warnings to the appropriate stakeholders (Emergency Management Southland, 2017). The Southland Emergency Management Group Plan identifies the MetService as the agency responsible for surveillance, monitoring, assessment, and issuing of alerts for severe weather hazards (Emergency Management Southland, 2017). Environment Southland Regional Council is the agency responsible for issuing flood warnings and operates a Flood Warning Operations Centre (Environment Southland Regional Council, 2021).

Appendix H: Codebook

Name	Files	References
Although we all complain about a lack of data, a lot of the time, there is quite a bit of data out there that could be used and isn't being used very well.	1	1
Community Risk Awareness	1	1
Consuming data vs. holding information	2	2
Cultural Conditions	0	0
Cultural Conditions\New Zealand Practice	5	5
Cultural Conditions\New Zealand Practice\Canterbury Governance Structure	1	1
Cultural Conditions\New Zealand Practice\CIMS Structure	1	3
Cultural Conditions\Reactive vs. Proactive CDEM practices	6	9
Cultural Conditions\Reactive vs. Proactive CDEM practices\Lack of	1	2
motivation to improve, think things are good as they are		
Cultural Conditions\Reactive vs. Proactive CDEM practices\Lack of motivation to improve, think things are good as they are\Economic viability	1	3
Cultural Conditions\Reactive vs. Proactive CDEM practices\Lack of motivation to improve, think things are good as they are\She'll be alright	1	1
Data for IBFWs and General DRR	0	0
Data for IBFWs and General DRR\Data and Information Sharing	1	2
Data for IBFWs and General DRR\Data and Information Sharing\Attitude shift for better data sharing and access	2	4
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data	13	37
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data\Anti-competitive behaviour	1	2
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data\Commercial or proprietary licensing	2	4
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data\Comparing New Zealand to China and Russia	1	1
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data\Data collected by different agencies is managed and stored differently which makes it difficult for integration	1	1
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data\Different countries have different standards and cultures	1	3
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data\Fear of publishing impacts	2	2
Data for IBFWs and General DRR\Data and Information Sharing\Challenges and barriers to sharing data\Fear of publishing impacts\Catastrophising it	1	1

Name	Files	References
Data for IBFWs and General DRR\Data and Information	2	5
Sharing\Challenges and barriers to sharing data\Fear of publishing		
impacts\Property Values		
Data for IBFWs and General DRR\Data and Information	1	2
Sharing\Challenges and barriers to sharing data\Liability		
Data for IBFWs and General DRR\Data and Information	2	2
Sharing\Challenges and barriers to sharing data\Near real-time data		
Data for IBFWs and General DRR\Data and Information	4	10
Sharing\Challenges and barriers to sharing data\Open Data		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Challenges and barriers to sharing data\Open Data\Open		
	1	1
Data for IBFWs and General DKK\Data and Information		1
Sharing (Challenges and barriers to sharing data (Open Data (Open		
Data for private use America-Canada model	2	2
Data for IBFWs and General DKK\Data and Information	2	2
sharing (Challenges and barriers to sharing data (Outdated model of		
Sharing data	1	1
Data for IBFWs and General DRK (Data and Information Sharing) Challenges and barriers to sharing data) Outdated model of		I
sharing (Challenges and Damers to sharing data (Outdated moder of sharing data) Einancial motive vs. for the good of New Zealand		
Sharing data/Financial motive vs. for the good of New Zealand	2	2
Sharing/Challenges and barriers to sharing data/Ownership	2	2
Data for IREWs and Constal DRP/Data and Information	2	5
Sharing Challenges and barriers to sharing data Political	5	5
Data for IBEWs and Constal DPP/Data and Information	1	1
Sharing/Challenges and barriers to sharing data/Political/Impact	'	4
data is political data		
Data for IBEWs and General DRR\Data and Information	2	6
Sharing/Challenges and barriers to sharing data/Political/Impact	-	Ũ
data is political data/Impact data has a lot to do with the organisation		
of the government		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Challenges and barriers to sharing data\Poor data collection		
and management practices		
Data for IBFWs and General DRR\Data and Information	3	3
Sharing\Challenges and barriers to sharing data\Poor data collection		
and management practices\Pessimism or frustration		
Data for IBFWs and General DRR\Data and Information	8	27
Sharing\Challenges and barriers to sharing data\Privacy		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Challenges and barriers to sharing data\Trust		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Challenges and barriers to sharing data\Trust\Reliability		
Data for IBFWs and General DRR\Data and Information	2	2
Sharing\Challenges and barriers to sharing data\Trust\Suspicion		
between institutions in New Zealand inhibits progress in risk		
maangement		
Data for IBFWs and General DRR\Data and Information Sharing\Data	9	22
		04
Data for IBFWs and General DKR\Data and Information Sharing\Data	8	21
Integration/Common Operating Picture	1	
Integration Spatial data infractivity of a and information Sharing Data	1	5

Name	Files	References
Data for IBFWs and General DRR\Data and Information Sharing\Data	1	3
Integration\Spatial data infrastructure\Transition from SDI to		
unstructured or hybrid data sharing and access		
Data for IBFWs and General DRR\Data and Information	2	3
Sharing\Drivers for sharing data		
Data for IBFWs and General DRR\Data and Information	3	6
Sharing\Drivers for sharing data\More data is available or exists than		
we think		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Drivers for sharing data\Realisation that data has value and		
use beyond its initial intended purpose		_
Data for IBFWs and General DRR\Data and Information	3	7
Sharing\Drivers for sharing data\Various stakeholders that create,		
manage, share, access, use data		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Drivers for sharing data\Various stakeholders that create,		
manage, share, access, use data\Different ways of communicating		
the information or data	2	4
Data for IBFWs and General DKK/Data and Information	3	4
Sharing Drivers for sharing data Various stakeholders that create,		
Data for JPE/Ma and Concrete DPP/ Data and Information	1	0
Data for IBFWs and General DKK/Data and Information	6	9
Sharing Drivers for sharing data Various stakeholders that create,		
manage, share, access, use data liveed for a common framework for		
Data for IREWs and Constal DRP/ Data and Information	4	7
Sharing Drivers for sharing data Warieus stakeholders that create	0	/
manage share access use data/Mand for a common framework for		
aggregating all sources of data/Need to get time critical data before		
it is lost		
Data for IBEWs and General DRR\Data and Information	1	1
Sharing Drivers for sharing data Various stakeholders that create	•	•
manage, share, access, use data/Need for a common framework for		
aggregating all sources of data Need to get time critical data before		
it is lost\Sometimes a time delay in collecting the data can help		
Data for IBFWs and General DRR\Data and Information	5	5
Sharing\Drivers for sharing data\Various stakeholders that create,		
manage, share, access, use data\Need for better collaboration and		
coordination		
Data for IBFWs and General DRR\Data and Information	0	0
Sharing\Outcomes of sharing data		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Outcomes of sharing data\Using data from other agencies		
to learn from past events and predict future events		
Data for IBFWs and General DRR\Data and Information	0	0
Sharing\Requirements and practices of sharing data		
Data for IBFWs and General DRR\Data and Information	18	38
Sharing\Requirements and practices of sharing data\Cooperation		
and collaboration		
Data for IBFWs and General DRR\Data and Information	5	10
Sharing\Requirements and practices of sharing data\Custodianship		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing Requirements and practices of sharing data Data directories	_	
Data for IBFWs and General DRR\Data and Information	5	15
Sharing\Requirements and practices of sharing data\Data integration		

Name	Files	References
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Requirements and practices of sharing data\Data		
integration\Common alerting protocol		
Data for IBFWs and General DRR\Data and Information	2	12
Sharing\Requirements and practices of sharing data\Data portal		
Data for IBFWs and General DRR\Data and Information	1	6
Sharing\Requirements and practices of sharing data\Data		
portal\Central repository		
Data for IBFWs and General DRR\Data and Information	4	15
Sharing\Requirements and practices of sharing data\Data standards		
Data for IBFWs and General DRR\Data and Information	1	2
Sharing\Requirements and practices of sharing data\Data		
standards\Challenges		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Requirements and practices of sharing data\Data		
standards\Meta-data		
Data for IBFWs and General DRR\Data and Information	1	4
Sharing\Requirements and practices of sharing data\Data		
standards\Nation-specific		
Data for IBFWs and General DRR\Data and Information	10	20
Sharing\Requirements and practices of sharing data\Data		
standards\Spatial GIS Data		
Data for IBFWs and General DRR\Data and Information	1	1
Sharing\Requirements and practices of sharing data\Interconnected		
systems		
Data for IBFWs and General DRR\Data and Information	5	11
Sharing\Requirements and practices of sharing data\Networking and		
relationships		
Data for IBFWs and General DRR\Data and Information	3	12
Sharing\Requirements and practices of sharing data\Online Web		
Maps		
Data for IBFWs and General DRR\Data and Information	2	2
Sharing\Requirements and practices of sharing data\Training		
Data for IBFWs and General DRR\Data and Information	1	2
Sharing\Requirements and practices of sharing data\Two-way flow of		
information from public-authorities		
Data for IBFWs and General DRR\Data Collection and creation	0	0
Data for IBEWs and General DRR\Data Collection and	8	28
creation/Challenges or Barriers to Data Use Access Creation	U	20
Data for IBEWs and General DRR\Data Collection and	9	24
creation/Challenges or Barriers to Data Use Access Creation/Data	,	
challenges		
Data for IBEWs and General DRR\Data Collection and	1	2
creation/Challenges or Barriers to Data Use, Access, Creation/Data		-
ownership		
Data for IBEWs and General DRR\Data Collection and	1	4
creation/Challenges or Barriers to Data Use, Access, Creation/Privacy		
Data for IBFWs and General DRR\Data Collection and	7	7
creation/Challenges or Barriers to Data Use Access		
Creation/Resource Limitations		
Data for IBFWs and General DRR\Data Collection and	5	6
creation/Challenges or Barriers to Data Use. Access	-	
Creation\Resource Limitations\Finance or funding limitations		

Name	Files	References
Data for IBFWs and General DRR\Data Collection and	4	6
creation\Challenges or Barriers to Data Use, Access,		
Creation\Resource Limitations\Human-Power and Time Limitations		
Data for IBFWs and General DRR\Data Collection and	2	4
creation\Challenges or Barriers to Data Use, Access, Creation\Roles		
and Responsibility		
Data for IBFWs and General DRR\Data Collection and	7	7
creation\Challenges or Barriers to Data Use, Access,		
Creation\Sharing		
Data for IBFWs and General DRR\Data Collection and	3	5
creation\Challenges or Barriers to Data Use, Access,		
Creation\Transparency		
Data for IBFWs and General DRR\Data Collection and creation\Data	11	62
Sources		
Data for IBFWs and General DRR\Data Collection and creation\Data	0	0
Sources\Alternative or unofficial data sources		
Data for IBFWs and General DRR\Data Collection and creation\Data	4	8
Sources\Alternative or unofficial data sources\Citizen science		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Citizen science\Do the		
challenges with citizen science outweigh the benefits of it	2	,
Data for IBFWs and General DRR\Data Collection and creation\Data	3	6
Sources\Alternative or unofficial data sources\Citizen		
science Engagement and contributions	1	0
Data for IBFWs and General DKK\Data Collection and creation\Data	1	2
Sources/Alternative or unofficial data sources/Citizen		
Science (Training	2	E
Data for IBFWs and General DRR (Data Collection and creation (Data	2	5
it a form of citizen science)		
Data for IBEWs and Constal DPP\Data Collection and creation\Data	12	52
Sources\Alternative or unofficial data sources\Crowdsourcing	12	52
Data for IBEWs and General DRR\Data Collection and creation\Data	1	2
Sources\Alternative or unofficial data sources\Crowdsourcing\Digital	'	2
divide and urban bias		
Data for IBEWs and General DRR\Data Collection and creation\Data	1	19
Sources\Alternative or unofficial data sources\Crowdsourcing\NZ		
Flood Pics		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Crowdsourcing\NZ		
Flood Pics\Citizen science for quality control		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	4
Sources\Alternative or unofficial data sources\Crowdsourcing\NZ		
Flood Pics\Desire to keep the project independent from institutions		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Crowdsourcing\NZ		
Flood Pics\Desire to keep the project independent from		
institutions\Public trust in the platform		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Crowdsourcing\NZ		
Flood Pics\Desire to keep the project independent from		
Institutions\What insitutionalising it could mean		
Data for IBFWs and General DRR\Data Collection and creation\Data	2	3
Sources\Alternative or unofficial data		
sources\Crowdsourcing\Trusted crowdsourcing and crowdsourcers		

Name	Files	References
Data for IBFWs and General DRR\Data Collection and creation\Data	3	5
Sources\Alternative or unofficial data sources\Media Reports		
Data for IBFWs and General DRR\Data Collection and creation\Data	18	66
Sources Alternative of unofficial data sources (Social media	0	0
Sources Alternative or unofficial data sources Social modia) Drivers	0	0
for adaption of social modia platforms and data		
Data for IBEWs and Gonoral DRR\Data Collection and croation\Data	1	1
Sources\Alternative or unofficial data sources\Social media\Drivers	1	1
for adoption of social media platforms and data\Age		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Social		
media\Facebook		
Data for IBFWs and General DRR\Data Collection and creation\Data	2	3
Sources\Alternative or unofficial data sources\Social media\For		
evaluation		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	3
Sources\Alternative or unofficial data sources\Social		
media\Instagram		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Social media\One-way		
social media communication		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Social media\People		
interfering in the response and putting themselves and others at risk	-	-
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Social media\Public		
Image, transparency, and trust	4	
Data for IBFWs and General DKR\Data Collection and creation\Data	1	1
Sources Alternative or unofficial data sources Social media Public		
perceptions vs. official perceptions of impacts	2	2
Sources Alternative or unofficial data sources Social modia Quality	3	3
control		
Data for IBEWs and General DRR\Data Collection and creation\Data	3	3
Sources\Alternative or unofficial data sources\Social media\Skewed	Ŭ	5
data		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Social		
media\SnapChat		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	2
Sources\Alternative or unofficial data sources\Social media\Tourism		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Social		
media\Triangulating across different platforms		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	1
Sources\Alternative or unofficial data sources\Social media\Twitter		
Data for IBFWs and General DRR\Data Collection and creation\Data	0	0
Sources\Official and Trusted Data Sources	0	
Data for IBFWs and General DKK/Data Collection and creation/Data	2	2
Data for IREWs and Goneral DPP/Data Collection and creation/Data	1	1
Sources/Official and Trusted Data Sources/Dropps		
Data for IBEWs and General DRR/Data Collection and creation/Data	2	7
Sources/Official and Trusted Data Sources/QuickCapture	_	,

Name	Files	References
Data for IBFWs and General DRR\Data Collection and creation\Data	2	4
	2	2
Sources/Official and Trusted Data Sources/Survey123	2	2
Data for IBFWs and General DRR\Data Collection and creation\Data	10	23
Sources\Official and Trusted Data Sources\Tacit knowledge and		
prior experience		
Data for IBFWs and General DRR\Data Collection and creation\Data	2	2
Sources\Official and Trusted Data Sources\Tacit knowledge and		
prior experience\Experience to inform decisions and to learn from		
previous impacts is a double edged sword		
Data for IBFWs and General DRR\Data Collection and creation\Data	1	2
Sources\Official and Trusted Data Sources\Tacit knowledge and		
prior experience\Risk of losing undocumented tacit knowledge and		
experience		
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection		
Data for IBFWs and General DRR\Data Collection and	0	0
creation\Drivers for IVE Data Collection\Events		
Data for IBFWs and General DRR\Data Collection and	2	5
creation\Drivers for IVE Data Collection\Events\Canterbury		
earthquakes		
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Events\Canterbury		
earthquakes\Post-event Technical Advisory Group Review and		
		0
Data for IBFWs and General DRR\Data Collection and	1	2
creation/Drivers for IVE Data Collection/Events/Canterbury		
earthquakes\Rapid Impact Assessment needs	2	7
Data for IBFWs and General DKR/Data Collection and	2	/
creation/Drivers for IVE Data Collection/Events/Canterbury		
Pate for IPEW/a and Congral DPP/ Data Collection and	1	1
creation Drivers for IVE Data Collection Events Canterbury	1	1
earthquakes\Rapid Impact Assessment peeds\Efficiency		
Data for IBEWs and General DRR\Data Collection and	1	1
creation/Drivers for IVE Data Collection/Events/Canterbury	•	
earthquakes\Rapid Impact Assessment needs\Post-earthquake		
damage assessments		
Data for IBEWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Events\Canterbury		
earthquakes\Rapid Impact Assessment needs\Pre-earthquake data		
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Events\Canterbury		
earthquakes\Rapid Impact Assessment needs\Time constraints		
Data for IBFWs and General DRR\Data Collection and	1	4
creation\Drivers for IVE Data Collection\Events\COVID-19 Response		
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Events\Cyclone Pam		
Data for IBFWs and General DRR\Data Collection and	2	2
creation\Drivers for IVE Data Collection\Events\Edgecumbe Floods		
Data for IBFWs and General DRR\Data Collection and	3	7
creation\Drivers for IVE Data Collection\Events\Kaikoura		
earthquakes		

Name	Files	References
Data for IBFWs and General DRR\Data Collection and	1	3
creation\Drivers for IVE Data Collection\Events\Melbourne		
Thunderstorm-induced asthma		
Data for IBFWs and General DRR\Data Collection and	1	3
creation\Drivers for IVE Data Collection\Events\Pigeon Valley Fire		
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Events\Port Hills fires		
Data for IBFWs and General DRR\Data Collection and	2	5
creation\Drivers for IVE Data Collection\Events\Southland Floods		
Data for IBFWs and General DRR\Data Collection and	1	5
creation\Drivers for IVE Data Collection\Events\Southland		
	4	
Data for IBFWs and General DKR\Data Collection and	1	1
creation/Drivers for IVE Data Collection/Events/Southland		
	1	1
Data for IBFWs and General DKK/Data Collection and	I	I
Creation Drivers for IVE Data Collection Events Southland		
Produstiked Warning Onexpected public perceptions of the red		
Note for IREWs and General DRP\Data Collection and	1	10
creation Drivers for IVE Data Collection Events Waikato	I	10
Thunderstorm-induced asthma		
Data for IBEWs and General DRR\Data Collection and	1	3
creation/Drivers for IVE Data Collection/Events/Waikato		5
Thunderstorm-induced asthma\Over-burdened health system		
Data for IBEWs and General DRR\Data Collection and	1	2
creation\Drivers for IVE Data Collection\Events\West Coast events		_
Data for IBFWs and General DRR\Data Collection and	1	3
creation\Drivers for IVE Data Collection\Events\Whakaari White		
Island Volcanic Eruption		
Data for IBFWs and General DRR\Data Collection and	1	6
creation\Drivers for IVE Data Collection\Modelling Calibration		
Data for IBFWs and General DRR\Data Collection and	2	3
creation\Drivers for IVE Data Collection\Organisational culture can		
drive innovation		
Data for IBFWs and General DRR\Data Collection and	1	2
creation\Drivers for IVE Data Collection\Organisational culture can		
drive innovation\An individual with the required skills and		
knowledge to drive change and innovation		
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Recognising the value of		
Information	1	1
Data for IBFWs and General DKK/Data Collection and	I	I
Creation/Drivers for the Data Collection/Research	2	1
reaction Drivers for IVE Data Collection Technological	2	4
Data for IBEWs and Gonoral DRR\Data Collection and	Δ	5
creation/Drivers for IVF Data Collection/Ton level government	-	
investment and prioritisation		
Data for IBFWs and General DRR\Data Collection and	3	8
creation\Drivers for IVE Data Collection\Various stakeholders that		
create, manage, share, access, use data		

Name	Files	References
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Various stakeholders that		
create, manage, share, access, use data\Different ways of		
communicating the information or data		
Data for IBFWs and General DRR\Data Collection and	3	4
creation\Drivers for IVE Data Collection\Various stakeholders that		
create, manage, share, access, use data\Ethical data collection		
Data for IBFWs and General DRR\Data Collection and	6	9
creation\Drivers for IVE Data Collection\Various stakeholders that		
create, manage, share, access, use data\Need for a common		
framework for aggregating all sources of data	_	
Data for IBFWs and General DRR\Data Collection and	5	6
creation\Drivers for IVE Data Collection\Various stakeholders that		
create, manage, share, access, use data\Need for a common		
framework for aggregating all sources of data\Need to get time		
critical data before it is lost		
Data for IBFWs and General DRR\Data Collection and	1	1
creation\Drivers for IVE Data Collection\Various stakeholders that		
create, manage, share, access, use data\Need for a common		
framework for aggregating all sources of data lived to get time		
critical data before it is lost Sometimes a time delay in collecting the		
data can neip	1	1
Data for IBFWs and General DRK (Data Collection and	4	4
creation/Drivers for IVE Data Collection/Various stakeholders that		
collaboration and coordination to establish data collection needs		
and standards		
Data for IREWs and Constal DRP\Data Collection and	6	11
creation/Systematic Impact Data Collection	0	11
Data for IBEWs and Constal DBP\Data Collection and	1	1
creation/Systematic Impact Data Collection/Best practice guidelines		1
Data for IBEWs and Constal DBP/Data Collection and	3	1
creation/Systematic Impact Data Collection/Best practice	5	4
Data for IBEWs and General DRR\Data Collection and	3	11
creation/Systematic Impact Data Collection/Standardisation		
Data for IBEWs and General DRR\Data Storage and Management	9	23
Data for IDF We and Conseral DDD) Data Storage and	1	1
Management Archiving		I
Data for IREWs and Constal DRD Data Storage and	1	1
Management/Puriness as usual vs. new tools only for emergency use		1
Data for IPEW/a and Conoral DPP/ Data Storage and	1	1
Management/Different ways of storing and managing data		1
Data for IREWs and Constal DRP/Data Storage and	1	1
Management/Need to know what to do with the data		1
Data for IPEW/s and Constal DPP/Data Storage and	2	2
Management)Trust or distruct in how the data is stored and	5	5
management (Trust of distrust in now the data is stored and		
Data for IBEWs and General DRR\Data Uses and Usability	0	0
	0	0
Data for IBEVVS and General DKK\Data Uses and Usability\Data	I	1
	10	25
Data for IBEVVS and General DKK\Data Uses and Usability\IBEW Data	12	35
Usability Needs	1	0
Use hilt Needel Quelity control	4	9

Name	Files	References
Data for IBFWs and General DRR\Data Uses and Usability\IBFW Data Usability Needs\Trust in the data	7	20
Data for IBFWs and General DRR\Data Uses and Usability\IBFW Data Usability Needs\Trust in the data\Confidence in data and subsequent analyses	1	1
Data for IBFWs and General DRR\Data Uses and Usability\IBFW Data Usability Needs\Trust in the data\Importance of official data	1	3
Data for IBFWs and General DRR\Data Uses and Usability\IBFW Data Usability Needs\Up-to-date information	4	5
Data for IBFWs and General DRR\Data Uses and Usability\IBFW Data Usability Needs\Up-to-date information\Real-time impact data for impact-based warnings	4	6
Data for IBFWs and General DRR\Data Uses and Usability\IBFW Data Usability Needs\Up-to-date information\Time-value of data	1	2
Data for IBFWs and General DRR\Data Uses and Usability\Stakeholders and Users of Impact Data	0	0
Data for IBFWs and General DRR\Data Uses and Usability\Stakeholders and Users of Impact Data\NZGIS4EM	4	7
Data for IBFWs and General DRR\Data Uses and Usability\Stakeholders and Users of Impact Data\Pacific Pilot Case Studies	1	2
Data for IBFWs and General DRR\Data Uses and Usability\Stakeholders and Users of Impact Data\Stakeholders	2	2
Data for IBFWs and General DRR\Data Uses and Usability\Stakeholders and Users of Impact Data\Stakeholders\Different needs for different stakeholders	10	31
Generation Change	1	1
Government and data access	1	7
Government-Private System	1	3
IBFW Implementation	0	0
IBFW Implementation\Challenges or Barriers to IBFW	0	0
IBFW Implementation\Challenges or Barriers to IBFW	16	47
IBFW Implementation\Challenges or Barriers to IBFW implementation\Agency remit or responsibility\Data custodianship	2	5
IBFW Implementation\Challenges or Barriers to IBFW	2	2
IBFW Implementation\Challenges or Barriers to IBFW implementation\Agency remit or responsibility\Need national cohesion	2	6
IBFW Implementation\Challenges or Barriers to IBFW implementation\Agency remit or responsibility\Shifting roles or adding onto existing roles	4	5
IBFW Implementation\Challenges or Barriers to IBFW implementation\Communicating to different audiences	6	11
IBFW Implementation\Challenges or Barriers to IBFW implementation\Conflicting messages	1	2
IBFW Implementation\Challenges or Barriers to IBFW	1	2
IBFW Implementation\Challenges or Barriers to IBFW	1	1
IBFW Implementation\Challenges or Barriers to IBFW implementation\Scaling problems\Spatial Scale	5	10

Name	Files	References
IBFW Implementation\Challenges or Barriers to IBFW	1	1
implementation\Scaling problems\Temporal Scale		
IBFW Implementation\Challenges or Barriers to IBFW	1	1
implementation\Scaling problems\Warning spatial scale doesn't		
match spatial scale of impacts		
IBFW Implementation\Challenges or Barriers to IBFW	0	0
implementation\Uncertainty		
IBFW Implementation\Challenges or Barriers to IBFW	0	0
implementation\Uncertainty\Communicating Uncertainty		
IBFW Implementation\Challenges or Barriers to IBFW	0	0
implementation\Uncertainty\Compounding Uncertainty	-	-
IBFW Implementation\Challenges or Barriers to IBFW	0	0
implementation\Uncertainty\Consequences of Uncertainty	_	
IBFW Implementation\Challenges or Barriers to IBFW	0	0
implementation\Uncertainty\Consequences of Uncertainty\Hesitancy		
to include impact messaging and to implement IBFWs		-
IBFW Implementation\Challenges or Barriers to IBFW	0	0
Implementation\Uncertainty\Consequences of		
Uncertainty\Misunderstanding of the models, forecasts, and		
	0	0
IBFW Implementation Challenges or Barriers to IBFW	0	0
Include the interview of the interview o	0	0
implementation/Challenges of Damers to IDFW	0	0
IPE/W Implementation/Challenges or Parriers to IPE/W	0	0
implementation/Uncortainty/Sources of Uncortainty/Lack of	0	0
knowledge and expertise		
IBEW Implementation/Challenges or Barriers to IBEW	0	0
implementation/Uncertainty/Sources of Uncertainty/Uncertainty in	U	0
forecasting the bazard		
IBEW Implementation/Challenges or Barriers to IBEW	0	0
implementation/Uncertainty/Sources of Uncertainty/Uncertainty of	Ũ	0
roles and responsibilities		
IBFW Implementation\Challenges or Barriers to IBFW	0	0
implementation\Uncertainty\Sources of Uncertainty\Uncertainty of		
the impacts		
IBFW Implementation\Drivers for IBFWs	0	0
IBEW Implementation\Drivers for IBEWs\Better communication with	5	9
the public	-	
IBFW Implementation\Drivers for IBFWs\Better communication with	1	1
the public\Boundary warnings vs. free-hand drawn warnings		
IBFW Implementation\Drivers for IBFWs\Better communication with	1	2
the public\CDEM offer value-add to MetService messages by adding		
on impact information		
IBFW Implementation\Drivers for IBFWs\Better communication with	4	5
the public\Giving meaning to the meteorological information		
IBFW Implementation\Drivers for IBFWs\Better communication with	1	1
the public\Shift from hazard focus to human and impacts focus		
IBFW Implementation\Drivers for IBFWs\Better communication with	1	1
the public\Shift from text-based to images in warning messages		
IBFW Implementation\Drivers for IBFWs\International trend of IBFWs	2	2
IBFW Implementation\Drivers for IBFWs\International trend of	1	3
IBFWs\National Review and Recommendations		

Name	Files	References
IBFW Implementation\Drivers for IBFWs\International trend of	4	8
IBFWs\Need to evaluate and communicate current efforts towards		
IBFWs for other countries to learn from and prepare for challenges		
IBFW Implementation\Drivers for IBFWs\International trend of	1	1
IBFWs\Sendai Framework requirements		
IBFW Implementation\Drivers for IBFWs\International trend of	2	4
IBFWs\WMO Guidelines		
IBFW Implementation\Drivers for IBFWs\New MetService Warning	1	5
System		
IBFW Implementation\Drivers for IBFWs\New MetService Warning	1	1
System\Still Threshold-based		
IBFW Implementation\Drivers for IBFWs\New MetService Warning	1	1
System\Stronger media presence and engagement		
IBFW Implementation\Drivers for IBFWs\Vision for impact-based	5	9
warnings		
IBFW Implementation\Drivers for IBFWs\Vision for impact-based	1	2
warnings\Using impact data to classify hazards		
IBFW Implementation\IBFW Data and Information Needs	0	0
IBFW Implementation\IBFW Data and Information Needs\Exposure	19	60
IBFW Implementation\IBFW Data and Information	3	8
		-
IBFW Implementation \IBFW Data and Information	6	11
Needs\Exposure\Dynamic Exposure	-	
IBFW Implementation\IBFW Data and Information	1	1
Needs\Exposure\Dynamic Exposure\Agent-based modelling		
IBFW Implementation \IBFW Data and Information	2	2
Needs\Exposure\Exposure data sources		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Exposure\Exposure data sources\FENZ		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Exposure\Exposure data sources\Flood models		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Exposure\Exposure data sources\KiwiRail		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Exposure\Exposure data sources\LINZ		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Exposure\Exposure data sources\NZTA		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Exposure\Exposure data sources\Ongoing question of data		
sources		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Exposure\Exposure data sources\StatsNz DataVentures	-	-
IBFW Implementation \IBFW Data and Information	3	3
Needs\Exposure\Movement and location of people		
IBFW Implementation \IBFW Data and Information Needs \Hazard	6	13
IBFW Implementation\IBFW Data and Information	1	1
Needs\Hazard\Flood Models		
IBFW Implementation\IBFW Data and Information	3	5
Needs\Hazard\Hazard Data Needs		
IBFW Implementation\IBFW Data and Information	0	0
Needs\Hazard\Hazard Data Sources		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Hazard\Hazard Data Sources\Climate Records		

Appendix H: Codebook

Name	Files	References
IBFW Implementation\IBFW Data and Information	2	4
Needs\Hazard\Hazard Data Uses		
IBFW Implementation\IBFW Data and Information	1	3
Needs\Hazard\Hazard Data Uses\Hazard Risk Communication		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Hazard\Hazard Data Uses\Land use zoning disputes		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Hazard\Hazard Data Uses\Learning from experience		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Hazard\Hazard Data Uses\Model Calibration		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Hazard\Hazard Data Uses\Model Calibration\Damage		
Surveys to correct models and reduce uncertainty		
IBFW Implementation\IBFW Data and Information Needs\Impact	4	4
IBFW Implementation\IBFW Data and Information	5	12
Needs\Impact\Geographic and Location information		
IBFW Implementation\IBFW Data and Information	3	3
Needs\Impact\Impact Data as input vs. output		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data as input vs. output\Input		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Data as input vs. output\Output		
IBFW Implementation\IBFW Data and Information	1	5
Needs\Impact\Impact Data Creation and Management		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Data Creation and Management\Balance		
between too much and enough data		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Data Creation and Management\Challenge		
with capturing less tangible or physical impacts and indirect impacts		
IBFW Implementation\IBFW Data and Information	3	9
Needs\Impact\Impact Data Creation and Management\Costs and		
Resources		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Creation and Management\Council		
documenting impacts on internal assets vs. privately owned assets		
IBFW Implementation\IBFW Data and Information	2	14
Needs\Impact\Impact Data Creation and Management\Data		
availability		
IBFW Implementation\IBFW Data and Information	2	7
Needs\Impact\Impact Data Creation and Management\Level of		
detail		
IBFW Implementation\IBFW Data and Information	2	7
Needs\Impact\Impact Data Creation and Management\Maintenance		
IBFW Implementation\IBFW Data and Information	6	11
Needs\Impact\Impact Data Creation and		
Management\Maintenance\Lack of motivation or interest in data		
creation and maintenance		
IBFW Implementation\IBFW Data and Information	2	3
Needs\Impact\Impact Data Creation and		
Management\Maintenance\Lack of motivation or interest in data		
creation and maintenance\Alternatively, interest and skills as driver		
for better data creation and management practices		

Name	Files	References
IBFW Implementation\IBFW Data and Information	1	4
Needs\Impact\Impact Data Creation and		
Management\Maintenance\Lack of motivation or interest in data		
creation and maintenance\Amount of work required		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Data Creation and		
Management\Maintenance\Who is the most suitable for managing		
data in terms of skills and expertise		
IBFW Implementation\IBFW Data and Information	3	4
Needs\Impact\Impact Data Creation and Management\Needs-based	-	
datasets & database development		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Creation and Management\Priorities		
IBEW Implementation/IBEW Data and Information	4	4
Needs/Impact/Impact Data Creation and		
Management/Priorities/Influence of management on science		
directions		
IBEW Implementation/IBEW Data and Information	1	1
Needs/Impact/Impact Data Creation and	1	
Management/Priorities/Waiting for someone else to take the lead		
IBEW Implementation/IBEW Data and Information	2	3
Needs/Impact/Impact Data Creation and Management/Resources	2	5
IREW Implementation/IREW Data and Information	1	1
Neede/Impact/Impact Data Creation and Management/Struggles	I	
IREW Implementation/IREW Data and Information	4	0
	0	7
Recostingaciting action and management whose		
	1	1
Neede/Impact/Impact Data Creation and Management/W/heas	I	1
	1	1
Needs\Impact\Impact Data Creation and Management\Whose		
	0	24
IBFW Implementation (IBFW Data and Information	8	24
Needs\Impact\Impact Data Sources	0	
IBFW Implementation (IBFW Data and Information	2	5
Needs\Impact\Impact Data Sources\ACC		
IBFW Implementation \IBFW Data and Information	2	8
Needs/Impact/Impact Data Sources/CDEM response records and		
situational reports		-
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Data Sources\GNS Science		
IBFW Implementation\IBFW Data and Information	5	13
Needs\Impact\Impact Data Sources\Insurance		
IBFW Implementation\IBFW Data and Information	2	3
Needs\Impact\Impact Data Sources\Media reports		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Sources\National Archives		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Data Sources\Ongoing question of data		
sources		
IBFW Implementation\IBFW Data and Information	7	22
Needs\Impact\Impact Data Sources\Post-event damage assessment		

Name	Files	References
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Data Sources\Post-event damage		
assessment\Access to sites		
IBFW Implementation\IBFW Data and Information	1	21
Needs\Impact\Impact Data Sources\Post-event damage		
assessment\Capture Tool		
IBFW Implementation\IBFW Data and Information	1	4
Needs\Impact\Impact Data Sources\Post-event damage		
assessment\Capture Tool\Flexibility and customisation		
IBFW Implementation\IBFW Data and Information	1	3
Needs\Impact\Impact Data Sources\Post-event damage		
assessment\CDEM Impact Assessments		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Data Sources\Post-event damage		
assessment\CDEM Impact Assessments\Phase 1 Wide area impact		
assessment		
IBFW Implementation\IBFW Data and Information	2	5
Needs\Impact\Impact Data Sources\Post-event damage		-
assessment\CDEM Impact Assessments\Phase 2 Rapid Damage		
Assessment		
IBFW Implementation\IBFW Data and Information	2	6
Needs\Impact\Impact Data Sources\Post-event damage		
assessment\CDEM Impact Assessments\Phase 3 Detailed Impact		
Assessment		
IBFW Implementation\IBFW Data and Information	1	1
Needs/Impact/Impact Data Sources/Post-event damage		
assessment\Collect only the data that will be used		
IBFW Implementation\IBFW Data and Information	1	2
Needs/Impact/Impact Data Sources/Post-event damage	-	
assessment\Contractors		
IBFW Implementation\IBFW Data and Information	3	9
Needs/Impact/Impact Data Sources/Post-event damage	•	
assessment\Standardisation		
IBFW Implementation\IBFW Data and Information	2	5
Needs/Impact/Impact Data Sources/Post-event damage		
assessment\Standardisation\Government endorsed template		
IBFW Implementation\IBFW Data and Information	1	2
Needs/Impact/Impact Data Sources/Post-event damage		
assessment\Standardisation\National damage assessment dictionary		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Data Sources\Post-event damage	-	
assessment\Training		
IBFW Implementation\IBFW Data and Information	1	1
Needs/Impact/Impact Data Sources/Post-event damage		
assessment/When is a good time to collect the data		
IBEW Implementation\IBEW Data and Information	1	1
Needs/Impact/Impact Data Sources/Public scan		
IBEW Implementation\IBEW Data and Information	2	4
Needs/Impact/Impact Data Sources/Risk Modelling Inputs	-	
IBFW Implementation\IBFW Data and Information	2	6
Needs/Impact/Impact Data Sources/Social media and		
crowdsourcing		
IBFW Implementation\IBFW Data and Information	2	3
Needs\Impact\Impact Data Sources\Stats NZ		-

Appendix H: Codebook

Name	Files	References
IBFW Implementation\IBFW Data and Information	3	4
Needs\Impact\Impact Data Sources\Tacit knowledge		
IBFW Implementation\IBFW Data and Information	15	67
Needs\Impact\Impact Data Uses		
IBFW Implementation\IBFW Data and Information	2	4
Needs\Impact\Impact Data Uses\Business case		
IBFW Implementation\IBFW Data and Information	4	6
Needs\Impact\Impact Data Uses\Decision-making and planning		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Decision-making and		
planning\Research vs. practice		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Data Uses\Decision-making and		
planning\Science-based practice and policy		
IBFW Implementation\IBFW Data and Information	5	8
Needs\Impact\Impact Data Uses\Emergency Management Planning,		
Response and Recovery		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Data Uses\Emergency Management Planning,		
Response and Recovery\Messaging for planning and preparedness	_	-
IBFW Implementation\IBFW Data and Information	5	9
Needs\Impact\Impact Data Uses\Impact data for different uses and		
purposes		-
IBFW Implementation\IBFW Data and Information	3	4
Needs\Impact\Impact Data Uses\Intelligence		
IBFW Implementation \IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Intelligence\Human Intelligence		
IBFW Implementation \IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Intelligence\Open Source		
	1	1
IBEW Implementation (IBEW Data and Information	I	I
Needs\Impact\Impact Data Uses\Intelligence\Operational		
	2	2
IBEW Implementation (IBEW Data and Information	2	2
development policies		
IREW/Implementation/IREW/Data and Information	2	2
Needs/Impact/Impact Data Lices/Landuse planning and	2	2
development policies/Planning		
IBEW Implementation/IBEW Data and Information	2	5
Needs\Impact\Impact Data Lises\Population Exposure Model	2	5
IBEW Implementation/IBEW Data and Information	Δ	12
Needs/Impact/Impact Data Uses/Rapid Impact Assessments	7	12
IBEW Implementation/IBEW Data and Information	1	1
Needs\Impact\Impact Data Uses\Research Analysis		
IBEW Implementation/IBEW Data and Information	1	3
Needs/Impact/Impact Data Uses/Risk Communication	•	Ŭ
IBEW Implementation/IBEW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk Management		
IBEW Implementation\IBEW Data and Information	4	11
Needs\Impact\Impact Data Uses\Risk Modelling		
IBFW Implementation\IBFW Data and Information	1	12
Needs\Impact\Impact Data Uses\Risk Modelling\Asset information		

Name	Files	References
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk Modelling\Asset		
information\Attributes		
IBFW Implementation\IBFW Data and Information	1	6
Needs\Impact\Impact Data Uses\Risk Modelling\Asset		
information\Building data		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Data Uses\Risk Modelling\Computational		
Fluid Dynamics Model		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk Modelling\Computational		
Fluid Dynamics Model/Variability in wind gusts and damage surveys		
IBFW Implementation\IBFW Data and Information	2	4
Needs\Impact\Impact Data Uses\Risk Modelling\Sense-checking		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk Modelling\Tunnel vision		
limiting risk and impact modelling		
IBFW Implementation\IBFW Data and Information	8	17
Needs\Impact\Impact Data Uses\Risk Modelling\Uncertainty		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk		
Modelling\Uncertainty\Damage Surveys to correct models and		
reduce uncertainty		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk Modelling\Uncertainty\How to		
communicate the uncertainty		
IBFW Implementation\IBFW Data and Information	6	30
Needs\Impact\Impact Data Uses\Risk Modelling\Vulnerability		
functions		
IBFW Implementation\IBFW Data and Information	0	0
Needs\Impact\Impact Data Uses\Risk Modelling\Vulnerability		
functions\Types of vulnerability functions		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk Modelling\Vulnerability		
functions\Types of vulnerability functions\Analytical vulnerability		
function	-	-
IBFW Implementation \IBFW Data and Information	2	6
Needs\Impact\Impact Data Uses\Risk Modelling\Vulnerability		
functions/Types of vulnerability functions/Empirical vulnerability		
		2
IBFW Implementation (IBFW Data and Information	2	3
Needs\Impact\Impact Data Uses\Kisk Modelling\Vulnerability		
inclions (Types of vulnerability functions (Expert opinion		
Vulnerability function	1	2
IBEVV Implementation (IBEVV Data and Information	1	2
functional Turpes of unla orse bility functional Judgement based		
IREW Implementation VIREW Data and Information	7	11
Needelympaet/mpaet Data Leas/Piele Medalling///ulperability	/	11
functions/Vorification		
IBEW Implementation/IBEW/ Data and Information	1	1
Needs/Impact/Impact Data Lisos/Pick Medalling///ulperability		1
functions/Verification/Bringing reality to the forecastors and		
modellers		

Name	Files	References
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Risk Modelling\Vulnerability		
functions\Verification\Bringing reality to the forecasters and		
modellers\An expert's surprise at the value of damage assessments		
to drive home the reality of the event		
IBFW Implementation\IBFW Data and Information	2	9
Needs\Impact\Impact Data Uses\Sendai Framework		
IBFW Implementation\IBFW Data and Information	1	9
Needs\Impact\Impact Data Uses\Sendai Framework\DesInventar		
Sendai		
IBFW Implementation\IBFW Data and Information	3	9
Needs\Impact\Impact Data Uses\Warnings		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Data Uses\Warnings\Impact-based warnings		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Data Uses\Warnings\Impact-based		
warnings\Historic Impacts		
IBFW Implementation\IBFW Data and Information	1	6
Needs\Impact\Impact Data Uses\Warnings\Impact-based		
warnings\Identifying triggers or Indicators		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Data Uses\Warnings\Impact-based		
warnings\Question of IBW efficacy		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Impact\Impact Databases		
IBFW Implementation\IBFW Data and Information	1	3
Needs\Impact\Impact Databases\European Severe Weather		
Database		
IBFW Implementation\IBFW Data and Information	7	11
Needs\Impact\Impact Databases\National dataset or database		
IBFW Implementation\IBFW Data and Information	1	3
Needs\Impact\Impact Databases\National dataset or database\Event		
driven vs. not event driven		
IBFW Implementation\IBFW Data and Information	2	4
Needs\Impact\Impact Databases\National dataset or database\How		
to convey the meaning of the data		
IBFW Implementation\IBFW Data and Information	1	4
Needs\Impact\Impact Databases\National dataset or		
database\Inclusion criteria		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Impact Databases\National dataset or		
database\Inclusion criteria\Judgement call for including impact data		
into national loss database		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Impact Databases\National dataset or		
database\Maturing the system, the future of the system		
IBFW Implementation\IBFW Data and Information	1	4
Needs\Impact\Impact Databases\National dataset or		
database\Software		
IBFW Implementation\IBFW Data and Information	1	3
Needs\Impact\Impact Databases\National dataset or database\The		
value of it		
IBFW Implementation\IBFW Data and Information	1	4
Needs\Impact\Impact Databases\National dataset or database\Users		1

Name	Files	References
IBFW Implementation\IBFW Data and Information	1	10
IREW/Implementation/IREW/ Data and Information	1	1
Noods/Impact/Impact Databases/National dataset or	1	1
database/Vision		
IBEW Implementation/IBEW Data and Information	1	1
Neede/Impact/Impact Databases/National dataset or databases/W/by	1	1
NEMA is doing it		
IBEW Implementation/IBEW Data and Information	3	7
Needs\Impact\Impact Databases\NIWA Historic events catalog	5	,
IBEW Implementation/IBEW Data and Information	1	Δ
Needs\Impact\Impact Databases\Storm Data	1	-
IBEW Implementation/IBEW Data and Information	0	0
Needs\Impact\Impact Models	U	0
IBEW Implementation/IBEW Data and Information	1	2
Needs)Impact/Impact Models/Different methodologies depending	1	2
on the bazard and the kinds of impacts you want to model		
IBEW Implementation/IBEW Data and Information	1	1
Neede\Impact\Impact Medals\Different methodologies depending	1	1
on the bazard and the kinds of impacts you want to		
model/Emergency Service Vehicle Damage Models		
IBEW Implementation/IBEW Data and Information	1	1
Needs\Impact\Impact Models\Different methodologies depending	I	1
on the bazard and the kinds of impacts you want to model/Net		
wanting to rely only on models		
IBEW Implementation/IBEW Data and Information	1	3
Needs\Impact\Impact Models\Different methodologies depending	I	5
on the bazard and the kinds of impacts you want to model/UK		
Surface Water Flooding Model		
IBEW Implementation/IBEW Data and Information	1	6
Needs\Impact\Impact Models\Different methodologies depending	•	Ũ
on the hazard and the kinds of impacts you want to model/UK		
Vehicle Overturn Model		
IBFW Implementation\IBFW Data and Information Needs\Impact\Lag	1	2
or delay in incoming impact information and issuing warnings		
IBFW Implementation\IBFW Data and Information	0	0
Needs\Impact\Types of Impacts		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Types of Impacts\Community Impacts		
IBFW Implementation\IBFW Data and Information	3	7
Needs\Impact\Types of Impacts\Direct vs. indirect impacts		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Types of Impacts\Focus on physical impacts		
IBFW Implementation\IBFW Data and Information	5	11
Needs\Impact\Types of Impacts\Human and social Impacts		
IBFW Implementation\IBFW Data and Information	2	3
Needs\Impact\Types of Impacts\Human and social Impacts\Health		
Impacts		
IBFW Implementation\IBFW Data and Information	1	6
Needs\Impact\Types of Impacts\Human and social Impacts\Mortality		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Types of Impacts\Human and social Impacts\Post-		
event surveys		
Appendix H: Codebook

Name	Files	References
IBFW Implementation\IBFW Data and Information	5	19
Needs\Impact\Types of Impacts\Human and social Impacts\Social		
and Cultural Impacts		
IBFW Implementation\IBFW Data and Information	1	3
Needs\Impact\Types of Impacts\Human and social Impacts\Social		
and Cultural Impacts\Cultural Heritage and Impacts		-
IBFW Implementation\IBFW Data and Information	1	3
Needs\Impact\Types of Impacts\Human and social Impacts\Social		
and Cultural Impacts/Cultural Heritage and Impacts/How to measure		
	2	1
Neede/Impact/Types of Impacts/Human and social Impacts/Social	3	4
and Cultural Impacts Physical impacts load to social impacts Social		
IREW Implementation/IREW Data and Information	1	2
Needs/Impact/Types of Impacts/Human and social Impacts/Social	1	2
and Cultural Impacts Physical impacts load to social impacts How to		
capture that properly in the data		
IBEW Implementation/IBEW Data and Information	2	6
Needs\Impact\Types of Impacts\Human and social Impacts\Social	2	U
and Cultural Impacts Social impact post-event assessment		
IBFW Implementation\IBFW Data and Information	4	9
Needs\Impact\Types of Impacts\Human and social Impacts\Social	-	
and Cultural Impacts\Welfare information		
IBFW Implementation\IBFW Data and Information	1	2
Needs\Impact\Types of Impacts\Human and social Impacts\Social		
and Cultural Impacts\Welfare information\Need a process to capture		
it properly		
IBFW Implementation\IBFW Data and Information	7	27
Needs\Impact\Types of Impacts\Physical Damage and Impacts		
IBFW Implementation\IBFW Data and Information	1	3
Needs\Impact\Types of Impacts\Physical Damage and Impacts\Utility		
	2	2
IBEW Implementation (IBEW Data and Information	2	Z
IREW/Implementation/IREW/Data and Information	1	1
Needs/Impact/Why we need more than just historical impact data	I	1
IREW Implementation/IREW Data and Information	1	1
Needs/Impact/Why we need more than just historical impact	1	1
data/Change in landscape (dynamic exposure)		
IBEW Implementation/IBEW Data and Information	1	1
Needs\Impact\Why we need more than just historical impact	•	
data/Climate Change		
IBFW Implementation\IBFW Data and Information	1	1
Needs\Impact\Why we need more than just historical impact		
data\Future events can be more extreme than past events		
IBFW Implementation\IBFW Data and Information	14	26
Needs\Vulnerability		
IBFW Implementation\IBFW Data and Information	5	12
Needs\Vulnerability\Dynamic Vulnerability		
IBFW Implementation\IBFW Data and Information	2	2
Needs\Vulnerability\Underlying health conditions		
IBFW Implementation\IBFW Data and Information	1	1
Needs/Vulnerability/Vulnerability Data Sources	1	1
Neede///ulperability//ulperability/Data and Information		
I INEEUS VUIHELADIILY VUIHELADIILY DALA SOULCES (FIOOD MODELS	1	1

Name	Files	References
IBFW Implementation\Impact Thresholds	3	5
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds	11	26
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds\Partnering with agencies for impact knowledge	15	46
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds\Partnering with agencies for impact knowledge\Difference across stakeholders	3	7
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds\Partnering with agencies for impact knowledge\Feedback or Circular Process	2	2
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds\Partnering with agencies for impact knowledge\Joint decisions	2	2
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds\Partnering with agencies for impact knowledge\Joint or consistent messaging across agencies	8	14
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds\Partnering with agencies for impact knowledge\Sharing Data	3	5
IBFW Implementation\Impact Thresholds\Defining Impact Thresholds\Subscription system	1	2
IBFW Implementation\Impact Thresholds\Many factors to consider when deciding on the warning level	1	1
IBFW Implementation\Impact Thresholds\Thresholds that are too	2	2
IBFW Implementation\Needs of the data to be useful and usable for IBFWs	12	35
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Trust in the data	7	20
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Trust in the data\Confidence in data and subsequent analyses		1
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Trust in the data\Credibility		2
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Trust in the data\Importance of official data	1	3
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Trust in the data\Quality control		9
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Trust in the data\Validation		7
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Up-to-date information	4	5
IBFW Implementation\Needs of the data to be useful and usable for IBFWs\Up-to-date information\Real-time impact data for impact- based warnings	4	6
IBFW Implementation\Warning audience decision-making	6	7
IBFW Implementation\Warning audience decision-making\In favour of more general warnings as opposed to impact-based or action- based warnings	1	2
IBFW Implementation\Warning audience decision-making\In favour of more general warnings as opposed to impact-based or action- based warnings\Community-based plans and actions		5

Name	Files	References
IBFW Implementation\Warning audience decision-making\In favour	1	1
of more general warnings as opposed to impact-based or action-		
based warnings\Need to have enough 'bubbles of control'		
IBFW Implementation\Warning audience decision-making\Lifelines	1	3
Utilities		
IBFW Implementation\Warning audience decision-making\Methods	2	2
of Communicating Impact Information	4	
IBFW Implementation/Warning audience decision-making/Methods	1	2
agoncios for communicating information.		
IBEW Implementation/Warning audience decision making/Matheds	5	7
of Communicating Impact Information/Including prescribed actions	5	,
IBEW Implementation/Warning audience decision-making/Methods	4	8
of Communicating Impact Information/Role of the media		Ŭ
IBFW Implementation\Warning audience decision-making\Methods	1	2
of Communicating Impact Information/Stakeholders working in silos		_
and not referring to each other's information		
IBFW Implementation\Warning audience decision-making\Warning	0	0
Impacts and Outcomes		
IBFW Implementation\Warning audience decision-making\Warning	4	6
Impacts and Outcomes\Awareness		
IBFW Implementation\Warning audience decision-making\Warning	5	6
Impacts and Outcomes\Cry Wolf		
IBFW Implementation\Warning audience decision-making\Warning	1	2
Impacts and Outcomes\Not associating impacts with the hazards	-	
IBFW Implementation\Warning audience decision-making\Warning	2	3
Impacts and Outcomes\Not taking the warnings seriously based on		
	0	4
IBEW Implementation (Warning audience decision-making (Warning	2	4
LIMS and PIMS	1	1
	1	r F
Machine learning and artificial intelligence		5
Need for government leadership		3
Need for government leadership\Political Direction		1
Need for government leadership\Political Direction\Political		1
direction vs. bottom-up		
Needs-based research and development	2	2
Needs-based research and development\User Needs Drive the	6	10
Design of the System		
Partnerships, Collaboration, and Relationships	0	0
Partnerships, Collaboration, and Relationships\Agencies running	1	1
professional development workshops to build relationships		
Partnerships, Collaboration, and Relationships\Cohabitation	1	1
Partnerships, Collaboration, and Relationships\Cohabitation\Need	1	1
for cohabitation		
Partnerships, Collaboration, and Relationships\Cohabitation\NZ	1	1
example of cohabitationa		
Partnerships, Collaboration, and Relationships\Community of	1	3
knowledge		
Partnerships, Collaboration, and Relationships\Informal Partnership	3	4
Partnerships, Collaboration, and Relationships\MetService-NIWA	1	7
Partnerships, Collaboration, and Relationships\MetService-	1	3
NIWA\Government cannibalising itself		

Name	Files	References
Partnerships, Collaboration, and Relationships\Regional flood group	1	1
Partnerships, Collaboration, and Relationships\Science advice groups	1	1
Private Weather forecasting and warning services	1	2
Red Cross Hazards App	3	6
Red Cross Hazards App\Chained Crowdsourcing	1	1
Red Cross Hazards App\Impact alerts	1	1
Resilience	4	4
Resilience\LINZ Resilience Challenge	1	2
Samoa Flood Early Warning Project	1	2
Systems Needs	0	0
Systems Needs\Adaptability and flexibility	1	2
Tax-funded publicly owned data	1	1
Unclear definitions and terminology	10	37
Unclear definitions and terminology\Base data vs. operational data	2	2
Unclear definitions and terminology\Examples of impact-based	4	6
services		
Unclear definitions and terminology\Impact assessment vs. situational awareness	2	2
Unclear definitions and terminology\Impact-based warnings definitions	2	3
Unclear definitions and terminology\Impact-oriented warnings	1	1
Unclear definitions and terminology\Information vs. intelligence	1	1
Unclear definitions and terminology\Loss	1	3
Unclear definitions and terminology\Loss\Monetary Loss	2	3
Unclear definitions and terminology\Multi-disciplinary	3	4
Unclear definitions and terminology\Vulnerability and Exposure	16	30
Unclear definitions and terminology\Vulnerability and	4	9
Exposure\Always changing over time		
Unclear definitions and terminology\Vulnerability and	11	32
Unclear definitions and terminology\Vulnerability and	12	22
Exposure\Vulnerability		
Warning Chain in New Zealand	7	25
Warning Chain in New Zealand\Greater Wellington Region Flood Early Warning Program	2	4
Warning Chain in New Zealand\Greater Wellington Region Flood Early Warning Program\Integration	1	1
Warning Chain in New Zealand\Greater Wellington Region Flood Early Warning Program\Lead Times	2	3
Warning Chain in New Zealand\MetService Convective Storm	1	1
Warning Evaluation	1	7
Who is governing who	2	2

Appendix I: Axial Coding Results

Contextual Conditions		
Cultural	Economic Viability	
Conditions	Nation-specific cultural conditions; NZ Culture and Practice	
	("She'll be right"; think things are good as they are; Lack of	
	motivation to improve	
	LIMS and PIMS	
	Drivers for adoption of social media platforms and data	
Governance	Roles and responsibilities	
Structure	Financial motive vs. for the good of NZ; Top level government	
	investment priorities	
	Influence of management on science	
	Sendai Framework requirements; Why NEMA is doing it	
	(building the national loss database)	
	LINZ Resilience Challenge	
	Canterbury Governance Structure	
	CIMS Structure	
	WMO Guidelines	
NZ Warnings	MetService convective storm warnings	
	Warning chain in New Zealand	
	Greater Wellington Region Flood Early Warning Program	
	Integration	
	Lead times	

Causal Conditions		
Events	Melbourne Thunderstorm-induced asthma; Waikato Thunderstorm-induced asthma	
	Cyclone Pam	
	West Coast events	
	Whakaari White Island Volcanic Eruption	
	COVID-19 response	
	Edgecumbe Floods; Southland Floods	
	Pigeon Valley Fire; Port Hills Fire	
	Kaikoura earthquakes; Canterbury earthquakes	
	Event driven vs. not event driven	
Research &	ArcGIS Online; Survey123; Social media;	
Technological	Crowdsourcing	
Advancements	Risk modelling; risk assessments	
	Developing a historical database for research	
Need to Improve warnings	International trend of IBFWs; National Review and recommendations (Australia); Need to evaluate and communicate current efforts towards IBFWs for other countries to learn from and prepare for challenges	
	Drivers for IBFWs; Shift from hazard focus to human and impacts focus; Not associating impacts with the hazards	
	Better communication with the public; Not taking the warnings seriously based on past experiences; Warning fatigue	
	Boundary warnings vs. free-hand drawn warnings (Is this NZ or UK or USA or multiple?)	
	Warnings; Impact-based warnings; Question of IBW efficacy	

Phenomena	
Data Collection	Data Availability; More data exists than we think
	Various stakeholders that create, manage, share, access, use data; Stakeholders and users of impact data; Stakeholders
	Different ways of communication the information or data
Data Sharing	Drivers for sharing data; Realisation that data has value and use beyond its initial intended purpose; "Although we all complain about a lack of data, a lot of the time, there is quite a bit of data out there that could be used and isn't being used very well."
	Requirements and practices of sharing data
Data Management	Custodianship
	Different ways of storing and managing data
IBFW Implementation	Vision for IBFWs
	Defining Impact Thresholds; Impacts to identify hazards
	Impact-based warnings definitions
	Impact-oriented warnings
	Multi-disciplinary
IBFW Data Needs	Hazard; Hazard data needs
	Impact; Historic Impacts; Why we need more than just historical impact data
	Physical damage and impacts; Utility outages
	Types of impacts; Physical impacts lead to social impacts; Focus on physical impacts; Direct vs. indirect impacts; Urban vs. rural impacts
	Social and cultural impacts; Cultural Heritage and impacts; Human and social impacts; Health Impacts; Mortality; Community Impacts
	Loss; Monetary loss
	How to measure significance; How to capture that properly in the data; How to convey the meaning of the data
	Impact data as input vs. output; Input; Output
	Exposure; Dynamic Exposure; Change in landscape (Dynamic exposure); Movement and location of people
	Vulnerability and Exposure; Always changing over time
	Vulnerability; Dynamic Vulnerability; Underlying health conditions
	Climate change; Future events can be more extreme than past events
	Geographic and location information

Intervening Conditions		
Data Collection Challenges and	Balance between too much and enough data; Need to know what to do with the data	
Considerations	Level of detail	
	Government role vs. sector roles	
	Challenge with capturing less tangible or physical impacts and indirect impacts	
	Council documenting impacts on internal assets vs. privately owned assets	
	Priorities; Lack of motivation or interest in data creation and maintenance; Waiting for someone else to take the lead	
	Amount of work required	
	Resources	
	Scaling problems; Spatial Scale; Temporal Scale	
	Struggles; Access to sites	
	Different countries have different standards and cultures	
	Ongoing question of data sources	
	When is a good time to collect the data?	
	Information vs. intelligence	
Data	Maintenance	
Management	Whose Responsibility; NEMA as a broker; Who is the most	
Challenges	suitable for managing data in terms of skills and expertise	
	Costs	
	Consuming data vs. holding information	
	Business as usual vs. new tools only for emergency use	
Social	Trust; Suspicion between institutions in NZ inhibits progress in	
challenges to	risk management; Trust or distrust in how the data is stored and	
sharing data	managed	
	Pessimism or frustration	
	Ownership (of the data)	
	Political; Impact data is political data; Impact data has a lot to do with the organisation of the government	
	Resource limitations; Finance or funding limitations; Human-	
	power and time limitations	
Ethical	Privacy	
Considerations	Fear of publishing impacts; Catastrophising it (the event by	
	publishing the impacts)	
	Commercial or proprietary licensing;	
Technical	Outdated model of sharing data; Need for better collaboration	
challenges to	and coordination	
sharing data	Reliability of the people producing and maintaining the data	
	Trust in the data; Confidence in the data and subsequent analysis	
	Welfare information (need a process to capture it properly)	
	Need a common framework for aggregating all sources of data:	
	Data collected by different agencies is managed and stored	
	differently which makes it difficult for integration	

Appendix I: Axial Coding Results

	Need to get time critical data before it is lost; Sometimes a delay in collecting the data can help
	Poor data collection and management practices
Uncertainty	Communicating uncertainty
	Compounding Uncertainty
	Sources of Uncertainty
	Uncertainty of the impacts; Uncertainty in forecasting the hazard
	Lack of knowledge and expertise; Lack of data
	Uncertainty of roles and responsibilities
	Unclear definitions and terminology (Base data vs. operational data; impact assessment vs. situational awareness; Information vs. intelligence; impact-based warnings definitions; impact- oriented warnings; vulnerability and exposure)
Governance challenges	Agency remit or responsibility; Shifting roles or adding onto existing roles
	Data custodianship
	Need national cohesion
	Legislation
	Liability
	Different platforms and agencies for communicating information
IBFW Data Usability Needs	Trust in the data; Confidence in the data and subsequent analysis; Importance of official data
	Different needs for different stakeholders
	Up-to-date information; Near real-time data; Real-time impact data for IBFWs; Lag or delay in incoming impact information and issuing warnings
Alternative or	Skewed data; Digital divide and urban bias; Age
unofficial data challenges	Credibility
	Do the challenges with citizen science outweigh the benefits of it?
Challenges with applying the knowledge, data, and	Defining thresholds; Difference across stakeholders; Many factors to consider when deciding on the warning level; Thresholds that are too sensitive or too widescale; Not wanting to rely only on models
information	Warning spatial Scale doesn't match spatial scale of impacts
	Tunnel vision limiting risk and impact modelling
	Evaluation
	Experience to inform decisions and to learn from previous impacts is a double edged sword
	Communication to different audiences

Action/Interaction Strategies		
Data Collection Strategies	Quality Control; Inclusion criteria; Judgement call for including impact data into national loss database	
	Needs-based datasets and database development; Collect only the data that will be used	
	Post-event surveys	
	Training	
	Recognising the value of information	
Building Partnerships for better data sharing practices	Networking and relationships; Attending conferences (e.g., MetService attending NEMA Conference; building the idea of a national severe weather and flood advisory group to fellow conference attendees); Facilitating workshops (e.g., MetService hosted workshops for CDEM people); Cooperation and collaboration; Need for better	
	collaboration and coordination	
	Attitude shift for better data sharing and access	
	Information partnership	
Data Management	Spatial Data infrastructure	
and Integration	Archiving	
Strategies	Transition from SDI to unstructured or hybrid data sharing	
	and access	
	Interconnected systems	
	Data integration	
	Data directories	
	Spatial GIS Data	
	Machine Learning and Artificial Intelligence	
Standardisation	Systematic Impact Data Collection; Training	
Strategies	Data standards; Best practice guidelines; Government endorsed template; National damage assessment dictionary	
	Meta-data	
Hazard Data		
Collection and	Hazard data sources	
Sharing Strategies	Climate Records	
Impact Data	Impact Data sources	
Collection and	ACC	
Sharing Strategies	CDEM response records and situational reports; CDEM	
	Impact Assessments (Phase 1 Wide Area Assessment, Phase	
	2 Rapid Damage Assessment, Phase 3 Detailed Impact	
	Assessment)	
	Public scan	
	GNS Science	
	Insurance	
	Media Reports	
	National Archives	

	Social media and crowdsourcing
	Stats NZ
	Tacit knowledge; Tacit knowledge and prior experience
	Impact data assessments; Social impact post-event assessment; Rapid impact assessments; CDEM Impact
	Assessments; Post-event damage assessment; Contractors
Exposure Data	Exposure Data Sources
Collection and	FENZ
Sharing Strategies	Demographics
	Ongoing question of data sources
	Stats NZ Data Ventures
	Flood models
	KiwiRail; NZTA
Vulnerability Data	Vulnerability Data Sources
Sharing Strategies	Flood models
Alternative or unofficial data	Crowdsourcing; Red Cross Hazards app Chained Crowdsourcing; NZ Flood Pics
sources	Citizen science
	Social media; Instagram; Twitter; Facebook; SnapChat;
	Community Reports
Official or	111 calls
Traditional Data	Drones
sources	Media reports
	QuickCapture; RiACT/Capture tool; Flexibility and customisation
	Satellite Imagery
	Survey123
Alternative or unofficial data strategies	Trusted crowdsourcing and crowdsourcers; Citizen science for quality control; Triangulating across different platforms; Training
Defining and Setting Thresholds	Partnering with agencies for impact knowledge; Sharing data
	Joint decisions; Joint or consistent messaging across
	Different methodologies depending on the hazard and the
	kinds of impacts you want to model
	Identifying triggers or indicators
Strategies to improving	CDEM offer value-add to MetService messages by adding on impact information
warnings	Shift from text-based to images in warning messages
	Stronger media presence and engagement
"Push and pull" system design strategies	Needs-Based research and development; User needs drive the design of the system; System needs; Adaptability and flexibility

	Users (of the national loss/impact database); Uses (of the national loss/impact database)
	Innovation
Strategies for addressing uncertainty	Damage Surveys to correct models and reduce uncertainty
Strategies for driving innovation and change	An individual with the required skills and knowledge can drive change and innovation; Alternatively, Interest and skills as driver for better data creation and management practices; generation change
	Political direction
	Bottom-up organisation
	Organisational culture can drive innovation

Consequences/	/Outcomes/Impacts (+/-)
Social Data Sharing	Property Values; Liability
Outcomes	Transparency
	NZGIS4EM
	Ethical Data collection
	Community of knowledge
Technical Data	Common alerting protocol
Sharing Outcomes	Common Operating Picture
	Open data
	Online web maps
	Standardisation
	Risk of losing undocumented tacit knowledge and
	experience
	Desinventar Sendai
Consequences of Uncertainty	Hesitancy to include impact messaging and to implement IBFWs
	Misunderstanding of the models, forecasts, and warnings
(IBF)Warning	Giving meaning to the meteorological information
Outcomes	Feedback or circular process
	Warning audience decision-making; Lifelines Utilities
	Awareness; Cry wolf; Community risk awareness
	In favour of more general warnings as opposed to impact-
	based or action-based warnings
	Community-based plans and actions
	Need to have enough 'bubbles of control'
	Role of the media
Alternative or	Two-way flow of information from public authorities; one-
Unofficial Data Risks	way social media communication; Engagement and
and Benefits	contributions
	Public trust in the platform; Public impact, transparency,
	and trust

	What institutionalising it (NZ Flood pics) could mean;
	Desire to keep the project independent from institutions (NZ Flood pics)
	People interfering in the response and putting themselves
	Public perceptions vs. official perceptions of impacts
IBFW Products	Pacific Pilot Case Studies
	Emergency Services Vehicle Damage Models
	UK Surface Water Flooding Model
	UK Vehicle Overturn Model
	Flood Early Warning Project
	New MetService Warning System
	Subscription system
	Methods of Communicating impact information
	Including prescribed actions
	Examples of impact-based services
Impact Databases	Storm Data (US)
Impact Databases	Furopean Severe Weather Database
	NZ National Loss Database: Maturing the system the
	future of the system: Software: The value of it: Vision
	NIWA Historic Events catalogue
Policy Decision-	Reactive vs. proactive CDEM practice)
Making, Practice	Tourism
,	Business Case
	Decision-making and planning: Planning: Land use
	planning and development policies
	Rosparch analysis
	Research analysis
	Communication: Emorgoney Management Planning
	Response And Recovery: Messaging for planning and
	preparedness
	Land use zoning disputes
	Learning from experience
	Science-based practice and policy
	Impact data for different uses and purposes
	Intelligence: Human Intelligence: Open Source
	Intelligence; Operational Intelligence
Data uses	For evaluation
	Impact data uses
	Hazard data uses
	Model calibration: Validation
	Population exposure model
	Risk modelling: Risk Modelling inputs: Asset information:
	Building data: Attributes: Vulnerability functions (empirical
	vulnerability function, expert opinion vulnerability
	function, judgement-based vulnerability function)
	Computational fluid dynamics (hazard modelling)

Appendix I: Axial Coding Results
Vulnerability functions; Types of vulnerability functions; Analytical vulnerability function
Verification
Bringing reality to the forecasters and modellers; An expert's surprise at the value of damage assessments to drive home the reality of the event

Appendix J: Chapter Five Publication

Australasian Journal of Disaster and Trauma Studies

Volume 24, Number 1

Volunteered Geographic Information for people-centred severe weather early warning: A literature review

Sara Harrison¹, Sally Potter², Raj Prasanna¹, Emma E. H. Doyle¹, David Johnston¹

- ¹ Joint Centre for Disaster Research, Massey University, New Zealand.
- ² GNS Science, New Zealand.

© The Author(s) 2020. (Copyright notice)

Author correspondence:

Sara Harrison, Joint Centre for Disaster Research, Private Box 756, Wellington 6140 New Zealand. Email: S.Harrison@massey.ac.nz URL: http://trauma.massey.ac.nz/issues/2020-1/AJDTS_24_1_Harrison.pdf

Abstract

Early warning systems (EWSs) can prevent loss of life and reduce the impacts of hazards. Yet, recent severe weather events indicate that many EWSs continue to fail at adequately communicating the risk of the hazard, resulting in significant life and property loss. Given these shortcomings, there has been a shift towards peoplecentred EWSs to engage with audiences of warnings to understand their needs and capabilities. One example of engaging with warning audiences is through the collection and co-creation of volunteered geographic information (VGI). Much of the research in the past has primarily focused on using VGI in disaster response, with less exploration of the role of VGI for EWSs.

This review uses a scoping methodology to identify and analyse 29 research papers on EWSs for severe weather hazards. Results show that VGI is useful in all components of an EWS, but some platforms are more useful for specific components than are others. Furthermore, the different types of VGI have implications for supporting people-centred EWSs. Future research should explore the characteristics of the VGI produced for these EWS components and determine how VGI can support a new EWS model for which the World Meteorological Organization is advocating: that of impact-based forecasting and warning systems. **Keywords:** early warning system, people-centred early warning system, volunteered geographic information, disaster risk reduction, severe weather

Early warning systems (EWSs) can prevent loss of life and reduce the impacts of hazards by providing members of the stakeholders and the public with information about likely, imminent risks on which they can act to prepare themselves and their property. As such, they have been a focus of disaster risk reduction since the Hyogo Framework for Action 2005-2015 through to the current Sendai Framework for Disaster Risk Reduction 2015-2030 (UNISDR, 2005, 2015). EWSs are described as having four key operational components: Disaster Risk Knowledge; Detection, Monitoring, and Warning Services; Communication and Dissemination Mechanisms; and Preparedness and Response Capacity (see Figure 1; Basher, 2006; Golnaraghi, 2012).

The first component, Disaster Risk Knowledge, involves systematically collecting and analysing data related to risk, such as the exposure and vulnerability of people and infrastructure to nearby hazards (Ahmed et al., 2012; Basher, 2006; Sai, Cumiskey, Weerts, & Bhattacharya, 2018). This involves assessing risk and vulnerability, building evacuation plans, and tailoring warning systems. Detection, Monitoring, and Warning Services make up the second component and are central to EWSs. This component requires reliable technology and involves continuous, automated detection and hazard monitoring (Ahmed et al., 2012; Basher, 2006; Sai et al., 2018). Furthermore, data, forecasts, and warnings should be archived for post-event analysis and



Adapted from Basher (2006), Golnaraghi (2012), and WMO (2018).

for continual system improvements (Ahmed et al., 2012; Basher, 2006; Sai et al., 2018). Impact data collected during and after a severe weather event would support both of these first two components (Harrison, Silver, & Doberstein, 2015).

The third component of an EWS is Communication and Dissemination, which is needed to reach those at risk. This involves using clear, concise, and understandable messages to enable proper preparedness (Ahmed et al., 2012; Basher, 2006; Sai et al., 2018). Multiple communications channels are necessary to reach as many people as possible (Ahmed et al., 2012; Basher, 2006). The fourth component of an EWS is Preparedness and Early Response Capacity. This involves running education and preparedness programmes to help people "understand their risks, respect the national warning services, and know how to react to warning messages" (WMO, 2018, p. 6). All four components of an EWS play a key role in crisis and risk communication.

EWSs share common characteristics with crisis and emergency risk communication theory. Like EWSs, the goal of crisis and risk communication theory is to provide sufficient and appropriate information to stakeholders that would allow them to "make the best possible decisions about their well-being" in a short period of time under uncertainty (Reynolds & Quinn, 2008, p. 14S). This involves understanding stakeholder (including the public) perceptions of risk and of the effectiveness of response, understanding the needs, capabilities, experiences, and predispositions of the stakeholders, and formulating messages based on these understandings for different audiences throughout the stages of crisis (Morgan, Fischhoff, Bostrom, Lave, & Atman, 1992; Reynolds & Seeger, 2005; Veil, Reynolds, Sellnow, & Seeger, 2008). Crisis and emergency risk communication theory is applied in risk messaging, crisis messaging, and warnings for health and emergency situations including, but not limited to, disease outbreaks, bioterrorism, hurricanes, and tornadoes (Revnolds & Seeger, 2005). The EWS framework presented in Figure 1 is thus supported by objectives of crisis and emergency risk communication theory, although the EWS framework does not include an apparent consideration for twoway communication: a key component in crisis and emergency risk communication theory for evaluating the effectiveness of communication (Garcia & Fearnley, 2012: Veil et al., 2008).

Recent severe weather events indicate that many EWSs continue to fail at adequately communicating the

Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

risk (and associated impacts) of the hazard, resulting in significant life and property loss due to limited understanding of, and response to, warnings (Ching, Carr de los Reyes, Sucaldito, & Tayag, 2015; Fleming et al., 2015; Wagenmaker et al., 2011). As such, there has been a push for "people-centred" EWSs to bring the "human factor" into consideration when designing and implementing EWSs and issuing warnings.

People-Centred Early Warning Systems

The broader EWS literature has recognised a communication gap between warning services and warning recipients, resulting in target audiences taking inadequate protective action despite receiving warnings (Anderson-Berry et al., 2018; Basher, 2006; Weyrich, Scolobia, Bresch. & Patt. 2018). In 2006, Basher introduced the concept of people-centred EWSs to address the "human factor" in EWSs, as he stated "failures in Early Warning Systems typically occur in the communication and preparedness elements" (Basher, 2006, p. 2168). Since then, there has been a shift towards people-centred EWSs which are developed for, and with, the target audiences to identify their needs and capacities and to transfer responsibility back to the audience to take protective actions (Basher, 2006; Scolobig, Prior, Schröter, Jörin, & Patt, 2015).

The United Nations Office for Disaster Risk Reduction (UNDRR; formerly known as the UNISDR) listed "investing in, developing, maintaining and strengthening people-centred multi-hazard, multi-sectoral forecasting, and Early Warning Systems" as an objective towards meeting the fourth priority of the Sendai Framework (UNISDR, 2015, p. 21). This "people-centred" aspect involves incorporating local and indigenous knowledge about hazards, promoting and applying low-cost EWSs that are appropriate to the audience based on their needs and capabilities, and broadening information channels (UNISDR, 2015; WMO, 2018), According to the Sendai Framework, people-centred EWSs can be developed through engagement with the audiences of warnings (e.g., individuals, communities, sectors: UNISDR, 2015; WMO, 2018).

One such example of engaging with warning audiences and understanding their needs and capabilities is through volunteered geographic information (VGI; WMO, 2017). VGI is information produced by or gathered from the public with associated locational attributes. The location-based information from VGI allows officials to identify high-risk areas, populations, and infrastructure

(Goodchild & Glennon, 2010; Granell & Ostermann, 2016; Haworth, 2018; Roche, Propeck-Zimmermann, & Mericskay, 2011).

Volunteered Geographic Information

VGI is valuable to disaster management because disasters are inherently location- and time-dependent and the location information from VGI allows officials to understand where the high-risk areas and populations are (Goodchild, 2007; Goodchild & Glennon, 2010; Granell & Ostermann, 2016; Haworth, 2018; Roche et al., 2011). The broader literature body around VGI, crowdsourcing, citizen science, and social media discusses and debates the relationship of these terms to each other and their associated characteristics and differences. It is argued that VGI overlaps both with citizen science and crowdsourcing (Cooper, Coetzee, & Kourie, 2018; Haklay, 2013, 2017). In Haklay's (2013) typology, crowdsourcing is classified as the lowest level of participation in citizen science. Citizen science (including crowdsourcing) is considered VGI when the information produced through the differing levels of participation includes geographic information (Haklay, 2017).

VGI can be collected in various ways, producing different types and formats of data. From reviewing the VGI and disaster risk reduction literature, we identified four types

Table 1 Summary of Volunteered Geographic Information types

Harrison et al.

of VGI that are generally produced and/or collected for disaster risk reduction; these are summarised in Table 1. Geo-located social media refers to VGI that is posted online by social media users that has associated geographical location information. The term social media recognises online blogs, micro-blogs, online social networking, and forums, which enable sharing of text, audio, photographs, and videos (Alexander, 2014). Facebook, Twitter, Sina Weibo, WeChat, Instagram, and SnapChat are some examples of popular social media platforms. During a severe weather event, authorities can use social media to disseminate alerts and warnings and collect information from members of the public about the event and its impacts (Alexander, 2014; de Albuquerque et al., 2017; Goodchild, 2007; Harrison & Johnson, 2016; Roche et al., 2011; Simon, Goldberg, & Adini, 2015; Slavkovikj, Verstockt, Van Hoecke, & Van de Walle, 2014).

For this review, crowdsourcing refers to gathering information from active public participation, namely reports submitted via online forms or mobile applications (Harrison & Johnson, 2016). Crowdsourcing has historically been used in the response to a disaster for building situational awareness, coordinating resources, and aiding response efforts (Harrison & Johnson, 2016; Haworth & Bruce, 2015; Poblet, Garcia-Cuesta, Casanovas, 2014). Within the severe weather context,

VGI Process	Spatial Data Format	Data Type	Data Sources	Disaster Risk Reduction Phase	Analysis/Outcomes
Geo-located social media harvesting	Point data	Impact data, exposure data, vulnerability data, hazard data Photos, videos, text	Facebook, Instagram, Twitter, Snapchat, Flickr, Sina Weibo, etc.	All	Cluster analysis, early detection, situational awareness, post-event damage/ impact assessment, response coordination
Crowdsourcing	Point data	Impact data, exposure data, vulnerability data, hazard data Photos, videos, text	Online reporting forms, mobile application	Readiness, Risk Reduction, During, Response	Cluster analysis, early detection, situational awareness, damage/impact assessment, response coordination
Participatory mapping/ Participatory GIS	Point, line, polygon	Impact data, exposure data, vulnerability data, hazard data, expert local knowledge Shapefiles	Community members, community leaders, stakeholders	Readiness, Risk Reduction, Recovery	Hazard and risk assessments/modelling, impact forecasting, customise/personalise warnings systems for the community, identify impact thresholds, inform/improve readiness and reduction efforts based on local knowledge
Local Knowledge	Point, line, polygon, written, audio	Impact data, exposure data, vulnerability data, hazard data, expert local knowledge Shapefiles	Community members, community leaders, stakeholders, experts	Readiness, Risk Reduction, Recovery	Hazard and risk assessments/modelling, impact forecasting, customise/personalise warnings systems for the community, identify impact thresholds, inform/improve readiness and reduction efforts based on local knowledge
	wy w with				
auma.massey.	.ac.nz			- M	211

crowdsourcing was used in the aftermath of Hurricane Katrina to locate missing people and allocate response efforts (Roche et al., 2011). In other examples, crowdsourcing is used for people on the ground to submit reports on flood levels and weather phenomena observations (Harrison & Johnson, 2016; Horita et al., 2018).

Participatory mapping and participatory Geographic Information Systems (participatory GIS) use local spatial knowledge to create spatial data or to verify and update existing data (Peters-Guarin, Mccall, & van Westen, 2012). Participatory mapping generally evolves into participatory GIS when hand-drawn maps or features are digitised and integrated into a GIS for further analysis (Brown & Kyttä, 2014; Forrester & Cinderby, 2011). Participatory mapping is often used to map exposure and vulnerability to hazards in communities to support disaster risk planning (Gaillard & Pangilinan, 2010; Haklay, Antoniou, & Basiouka, 2014). For weather-related hazards, Haworth, Whittaker, and Bruce (2016) found that participatory mapping enabled local knowledge exchange for community preparedness to bushfire risks.

Local knowledge refers to knowledge possessed by locals about their communities, neighbourhoods, traditions, history, environment, and hazards, among others. Local knowledge has not been clearly defined in the literature. For the purposes of this paper, we consider local knowledge as information gathered in similar participatory mapping and participatory GIS processes but not translated into a map or GIS. Recently, the access to and integration of local knowledge has been recognised for its importance to disaster risk reduction (Anderson-Berry et al., 2018; Gall & Cutter, 2016; Sebastian et al., 2017; UNISDR, 2015).

Past research has focused heavily on the role of VGI in disaster response, with less exploration in understanding how VGI can inform warnings before or during a severe weather event (Harrison & Johnson, 2016; Haworth & Bruce, 2015; Horita, Degrossi, Assis, Zipf, & de Albuquerque, 2013; Klonner et al., 2016). In Klonner and colleagues' (2016) systematic literature review, the authors focused on documenting research on VGI for preparedness and mitigation but did not provide clear findings in the context of warnings for severe weather. Assumpção, Popescu, Jonoski, and Solomatine (2018) identified the role of citizen observations in providing data for flood modelling and forecasting to solve issues of data scarcity, but again with no mention of warnings. Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

The original conception of VGI began with identifying its value for early detection and warning of hazards, using "citizens as sensors" (Goodchild, 2007). Since then, some work has emerged exploring VGI for early warnings of various hazards, such as earthquakes, landslides, and tsunami (Carley, Malik, Landwehr, Pfeffer, & Kowalchuck, 2016; Elwood, Goodchild, & Sui, 2012; Goodchild, 2007; Granell & Ostermann, 2016; Harrison & Johnson, 2016). Horita, de Albuquerque, Marchezini, and Mendiondo (2016) argued that VGI may help address challenges of assigning proper warning thresholds by incorporating local knowledge of response capabilities. Meissen and Fuchs-Kittowski (2014) developed a conceptual framework which demonstrated how crowdsourced data can be fully integrated into an existing EWS as another dataset to augment or enhance the warnings by providing context. However, no further evidence to date indicates the adoption into practice of this framework for any type of EWS. Finally, Marchezini and colleagues (2018) conducted a literature review of research on citizen science and EWSs and found that more research is needed to identify how citizen science can be "mainstreamed" into EWSs.

Some agencies have started collecting VGI to detect, monitor, and track events and their impacts. In the United Kingdom (UK), the British Geological Survey collects landslide impact data from Twitter including text descriptions, photos, and video footage of the resulting impacts (Pennington, Freeborough, Dashwood, Dijkstra, & Lawrie, 2015). These data are integrated into the National Landslide Database, which is used to create a Hazard Impact Model (Pennington et al., 2015). In Canada, the National Meteorological Service uses hazard information posted by the public on Twitter to detect weather events such as tornadoes and to verify and update current weather watches and warnings (Harrison & Johnson, 2016). However, there is a gap in the literature for fully characterising the role of VGI for severe weather warnings. It is important to fill this gap because information and knowledge possessed by citizens have the potential to uncover "areas of importance or concern" that have yet to be identified in an official capacity (Haworth, Bruce, & Middleton, 2012, p. 40). VGI offers a way to capture local knowledge about previous severe weather events and their extent. severity, and resulting impacts, as well as information on the local exposure and vulnerability that warning services may not necessarily possess (Fleming et al., 2015; GFDRR, 2016; Krennert, Pistotnik, Kaltenberger, & Csekits, 2018; Sai et al., 2018; WMO, 2017). This

paper uses a scoping review method to identify previous research into the use of VGI for severe weather EWSs, to attempt to answer the research question: *What are the current and potential uses of VGI for severe weather warnings*? The objective of this review is to determine how VGI has been, or could be, used within EWSs for severe weather hazards.

Method

This literature review uses a scoping method to explore areas of existing research and identify research gaps in VGI for severe weather early warning systems (Arksey & O'Malley, 2005; Paré, Trudel, Jaana, & Kitsiou, 2015). Scoping reviews provide a "rigorous and transparent method for mapping areas of research" in a short time (Arksey & O'Malley, 2005, p. 30). The aim is to describe the nature of the current literature on VGI for severe weather EWSs by describing the quality and quantity of the research (Grant & Booth, 2009; Paré et al., 2015). Scoping reviews are recognised for their strength in providing a broad picture of the state of research in a given topic area and are well-cited in the information systems field (Grant & Booth, 2009; Paré et al., 2015; Tan et al., 2017). This scoping review follows the fivestep process defined by Arksey and O'Malley (2005): 1) identify the research question, 2) identify relevant

Table 2

Search string employed in EBSCO Discovery and Scopus databases.

Topics covered	Search string statement
Warnings and Disaster Risk Knowledge	("risk communication" OR "warning*" OR "impact model*" OR "risk model*" OR "impact warning*" OR "impact*based warning*" OR "impact forecast*" OR "impact*based forecast*" OR "risk*based warning*" OR "risk*based communication")
	AND
A broad definition of VGI to include social media, participatory mapping, local knowledge based on location	("participatory" OR "participatory mapping" OR "VGI" OR "volunteered geographic information" OR "participatory GIS" OR "PGIS" OR "geographic crowdsourc*" OR "citizen science" OR "crowdsourc*" OR "social media")
	AND
Severe weather hazards as defined under the WWRP HIWeather Implementation Plan (Jones & Golding, 2014)	("weather" OR "storm*" OR "snow*" OR "wind*" OR "tornado*" OR "hurricane*" OR "cyclone*" OR "typhoon*" OR "monsoon*" OR "flood*" OR "mudslide" OR "flash flood*" OR "rain*" OR "wildfire")
rauma.massey.ac.nz	Washington and All

studies, 3) select studies, 4) chart the data, and 5) report the results.

The initial literature search involved developing a search string to capture the broad topic area of VGI and social media for warning of severe weather hazards. The search string comprised three joined statements, shown in Table 2, to cover warnings and Disaster Risk Knowledge (as per the first component of the EWS framework: Basher, 2006; Golnaraghi, 2012), VGI, and severe weather, which were entered into two academicfocused databases, Scopus and EBSCO Discovery Service, in August 2018. Literature review papers have been published on similar topics in this space that have searched no more than two databases (e.g., Klonner et al., 2016; Tan et al., 2017). Furthermore, Scopus is recognised for indexing a larger number of journals than other databases and is the largest searchable citation and abstract source for various scientific fields (Falagas, Pitsouni, Malietzis, & Pappas, 2008; Guz & Rushchitsky, 2009). Moreover, when searching the two databases many duplicate results were found between the two databases, ensuring confidence in the coverage.

"Participatory GIS" and "participatory mapping" are different types of VGI, and thus were identified as separate search terms. During the process of developing the search string, it was found that additional VGI research was left out of the search due to the specificity of "participatory mapping" and "participatory GIS", thus the search was widened with the term "participatory" to capture more VGI studies. Similarly, "flash flood" and "flood" are likely redundant, however, they were both included to ensure full coverage. The asterisk in the search string acts as a wildcard to search for variations of the root term. The search covered all years from the earliest available until mid-2018 and included only peer-reviewed journals and conference proceedings in English. The search resulted in 1,015 hits from Scopus and 122 from EBSCO. After removing duplicates, 1,027 unique publications were captured.

The following inclusion-exclusion criteria were used to select publications most relevant to this study:

- Publications that specifically focused on severe weather hazards as defined under the World Weather Research Programme's (WWRP) High Impact Weather Implementation Plan (Jones & Golding, 2014; n = 254);
- 2) Studies that explicitly discussed warnings, preparedness, mitigation, impact modeling and

forecasting, or risk mapping (reducing to n = 141); and,

- Studies that focused on VGI, crowdsourcing, citizen science, participatory mapping, local knowledge gathering, or social media data (reducing to n = 42).
- Finally, publications had to be original, complete research papers (n = 29).

After applying the inclusion-exlcusion criteria, information from the resulting papers was extracted according to different categories (see Table 3). Initially, the severe weather hazard(s) considered in the study were identified, after which the EWS framework was used to classify the papers and determine how VGI is or could be used within the EWS framework (these results are presented later in Figure 3). This classification involved identifying for which EWS component the VGI was used (see Figure 1), followed by the element within the EWS component (i.e., the specific task, tool, or process that the VGI was used for within the EWS component, such as risk mapping, detection, monitoring, forecasting, or warning dissemination). The VGI platform was identified (e.g., participatory mapping, participatory GIS, social media, crowdsourcing, citizen science, local knowledge), as well as the type of data that was collected (Haklay, 2017; Harrison & Johnson, 2016). These categories were chosen to determine the representation of VGI in severe weather EWSs.

Results

Table 4

The search of the two databases led to 1,027 unique publications. After applying the inclusion-exclusion criteria, the final number of papers selected for this study was 29. The categories listed in Table 3 were used as a structure for analysis and discussion, and were chosen based upon the dominance of those themes in the papers.

Cummony	of coloctod	ctudioc	oovoring	flood	hazarde
Summary	UI SEIELLEU	suules	covening	11000	nazarus.

Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

lable 3			
Categories	for literature	review	

Table 2

Category	Description
Hazard	The type of severe weather hazard(s) considered in the study.
Early Warning System Component	The component from the EWS framework that each study applies to.
VGI Platform	The source of the VGI data, such as from social media, or from crowdsourcing (i.e., citizen observation), citizen science (i.e., a higher level of engagement than crowdsourcing; Haklay, 2013), participatory mapping, participatory GIS, or local knowledge.
Data Type	The type of data that was collected through the VGI process, such as local knowledge captured through interviews and/or participatory mapping, hazard data from social media or crowdsourcing, etc.

Hazard Type

The selected articles covered a range of severe weather hazards as defined in the World Weather Research Programme (WWRP) High Impact Weather (HIWeather) implementation plan (Jones & Golding, 2014). Some hazards are represented more than others; of the 29 articles, 16 focused on flood hazards, followed by seven studies that covered general severe weather hazards, two studies that examined rain-induced landslides, two for cyclones, and one each for air quality and urban heat wave.

The 16 flood studies covered a range of elements within the EWS components. These elements were identified by reviewing the selected studies and aligning them with the EWS components. Table 4 provides a summary of the selected studies which examined floods. Most studies covered flood detection, monitoring, and forecasting using VGI collected from social media and crowdsourcing. The next most common elements that were covered in the flood studies were vulnerability

EWS Component	Element	Purpose of the study	VGI Platform	Data Type	Reference
Disaster Risk Knowledge	Modelling	To integrate local knowledge into GIS outputs for flood risk management using participatory GIS in order to understand how people cope and adapt	Participatory GIS	Interviews with households in Barangay, Philippines	Peters-Guarin et al., 2012
	Modelling	Validating flood models using quantitative and qualitative VGI	Participatory Mapping	Local knowledge from workshop participants and interviewees	Rollason et al., 2018
	Risk mapping	To provide an example of how to engage and collaborate with local stakeholders for flood management	Participatory Mapping	Land feature layers, input from locals	Lavers et al., 2018
		CAM	Manualta	IL WANT	
	MARAGE	Marrie Marrie Marrie Marriel	111111 tonal And	traur	na.massey.ac.n

Harrison et al.

Table 4 (continued)					
	Vulnerability assessment	To present a risk management framework that is based on local knowledge of the vulnerability to water hazards	Local knowledge	Meetings, workshops, interviews with people, media, and public sectors related to risk management	Arias et al., 2016
	Vulnerability assessment	To present a new methodology for incorporating stakeholder's participation, local knowledge, and locally spatial characteristics for vulnerability assessments of flood risk	Participatory GIS	Demographic data, infrastructure, hazard data (e.g., average annual rainfall), questionnaire interviews with experts and community members	Hung & Chen, 2013
	Vulnerability assessment	To present a new database for collection and assessment of flood damage using a bottom-up approach to gather and identify damage data	Social media	Personal blogs, on-site observations, public administration, social media, online media, local authorities, corporate websites	Saint-Martin et al., 2018
Detection, Monitoring, Warning Services	Detection	To develop a service-oriented architecture for flood management to capture real-time information about floods	Crowdsourcing	Rainfall, river, news, OpenStreetMap	Sharma et al., 2016
	Detection	To develop a methodology for interpreting image tags on social media for early detection of a flood and recording the impacts	Social media	Flickr posts - timestamps and location metadata	Tkachenko et al., 2017
	Detection, Forecasting	SWOT analysis of web-based access to data and model simulations, and insight on pEWMS, and conceptual framework for a Nordic pEWMS	Crowdsourcing, Social Media	Denmark: groundwater level observations lceland: flood photos Finland: mobile phone observations	Henriksen et al., 2018
	Detection, Monitoring	To assess social media feasibility for flood detection, monitoring, and forecasting and develop a novel methodology for doing so	Social media	Twitter data	Rossi et al., 2018
	Forecasting	To develop a methodology using social media for estimating rainfall runoff estimations and flood forecasting	Social media	Twitter data	Restrepo- Estrada et al., 2018
	Forecasting	To present a real-time modelling framework to identify likely flooded areas using social media	Social Media	Twitter data, LiDAR	Smith et al., 2017
	Monitoring	To estimate flood severity in an urban coastal setting using crowdsourced data	Crowdsourcing	Crowdsourced street flooding reports	Sadler et al., 2018
	Monitoring	To present a conceptual framework for collecting and integrating heterogeneous data from sensor networks and VGI	Crowdsourcing	Flood data from in-situ sensors and volunteers	Horita et al., 2015
	Monitoring	To present a new methodology for monitoring flood hazards using remote sensing and VGI	Crowdsourcing, Social Media	Volunteered data (photos, videos, news), Landsat, DEM, meteorological data, river data	Schnebele & Cervone, 2013
Detection, Monitoring, Warning Services; Communication and Dissemination Mechanism;	Warning messaging, preparedness	To test if evidence exists for social media reducing flood losses by informing mitigation decisions before the flood	Social media	Surveys, in-depth interviews with households who experienced flooding in Bangkok, 2011	Allaire, 2016
Preparedness and Early Response Capacity			7	2	
	Mass	A AND MARKEN AND A	G.		9
trauma.massey.ac.	nz				THE

assessments and risk mapping and modelling, using VGI from participatory GIS, participatory mapping, local knowledge, and social media. Just one study looked at using social media for detection, warning messaging, and for informing preparedness decisions (Allaire, 2016).

The remaining 13 studies covered other hazards, such as general severe weather, cyclones, landslides, air quality, and urban heatwaves. Table 5 provides a summary of the selected studies covering these various hazards. The general category refers to studies that did not identify a specific severe weather hazard, but referred only to "severe weather", usually in the context of severe weather warnings (Fdez-Arroyabe, Lecha Estela, & Schimt, 2018; Grasso & Crisci, 2016; Grasso, Crisci, Morabito, Nesi, Pantaleo, et al., 2017; He, Ju, Xu, Li, & Zhao, 2018; Krennert et al., 2018; Longmore et al., 2015; Lu et al., 2018).

In the general category, most of the selected studies looked at detection and forecasting using social media and crowdsourcing, followed by tracking warning dissemination across social media, and one study that used crowdsourcing for both risk and vulnerability assessment and providing warnings. The two cyclone studies each used social media and local knowledge to detect and forecast cyclone damage and to understand local responses to warnings, respectively. The two landslide studies both used VGI for landslide hazard and impact modelling, using crowdsourcing and social media. Finally, both the air quality and urban heatwave studies explored VGI from social media to forecast air quality and detect heatwaves based on individual exposure.

These studies indicate that VGI is used in the mapping, modelling, detection, monitoring, and warning of a number of severe weather hazards but that floods are the most heavily studied, with the widest range of VGI application across all of the elements. How these studies fit within the EWS framework is analysed in the following section.

Early Warning System Components

10

The papers were categorised by EWS component, as per Basher's (2006) framework (see Figure 1): 1) Disaster Risk Knowledge (n = 8); 2) Detection, Monitoring, and Warning Services (n = 16); 3) Communication and Dissemination Mechanisms (n = 2); and 4) Preparedness and Early Response Capacity (n = 1). Two studies were found to fall into more than one EWS component. The studies were then classified by the specific elements Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

within each component (e.g., hazard mapping, risk mapping, vulnerability assessment, modelling, hazard monitoring, detection, monitoring, warning, messaging, dissemination).

Disaster Risk Knowledge. Eight studies fall into the Disaster Risk Knowledge component of the EWS framework. Four of these studies looked at the use of VGI for hazard, risk, or impact modelling for landslides and floods (Choi, Cui, & Zhou, 2018; Pennington et al., 2015: Peters-Guarin et al., 2012; Rollason, Bracken, Hardy, & Large, 2018). Choi and colleagues (2018) presented a crowdsourcing-based smartphone application to aggregate landslide reports, which populates a landslide database for further hazard analysis. Similarly, Pennington and colleagues (2015) presented a landslide database for the UK that is partially populated by reports from Twitter to capture their impacts for further modelling. In the floods space, Peters-Guarin and colleagues (2012) utilised participatory GIS to integrate local knowledge of coping and adaptation practices into GIS-based flood risk analysis. Alternatively, Rollason and colleagues (2018) used participatory mapping to validate existing flood models.

The other four studies in the Disaster Risk Knowledge component involved risk mapping and vulnerability assessments, also for floods (Arias et al., 2016; Hung & Chen, 2013; Lavers & Charlesworth, 2018; Saint-Martin et al., 2018). Lavers and Charlesworth (2018) engaged with landowners to capture their knowledge of flood risk to inform flood management. Arias and colleagues (2016) presented a risk management framework for floods based on local knowledge of the vulnerability to water hazards. Hung and Chen (2013) incorporated stakeholders' participation and local knowledge through participatory GIS for vulnerability assessments of flood risk. Saint-Martin and colleagues (2018) developed a flood damage database (DamaGIS) to collect and assess flood damage, sourced from corporate websites, personal blogs, local authorities, on-site observations, social media, and online media. Furthermore, Saint-Martin and colleagues argued that social media can extend coverage to areas lacking regular media coverage and reveal damage that might have otherwise gone undetected.

Detection, Monitoring, and Warning. Within the Detection, Monitoring, and Warning component, 16 studies were identified. Four studies used VGI for hazard detection. Tkachenko, Jarvis, and Procter (2017) and Sharma and colleagues (2016) looked at VGI for

Harrison et al.

Table 5	
Summary of selected studies covering other severe weather hazar	ds.

228		-				
Hazard	EWS Component	Element	Purpose of the study	VGI Platform	Data Type	Reference
General	Disaster Risk Knowledge; Detection, Monitoring, Warning Services	Risk mapping	To present a data infrastructure that can be used to delineate individual vulnerability to meteorological changes	Crowdsourcing	User profiles on a mobile app	Fdez-Arroyable et al., 2018
	Detection, Monitoring, Warning Services	Detection	To present an Android-based application for geohazard reduction using crowdsourcing	Crowdsourcing	Crowdsourced information (field data, photos, videos)	He et al., 2018
		Detection, Monitoring	To present a conceptual framework for collecting weather photos	Crowdsourcing	User reports, photos, videos	Longmore et al., 2015
		Detection, Monitoring	To evaluate the occurrence of crowdsourcing for severe weather within European NMHSs	Crowdsourcing, Social Media	Surveys with European National Meteorological and Hydrological Services	Krennert et al., 2018
		Forecasting	To use social media as a new way of forecasting and generating traffic alerts due to weather hazards	Social media	Temporal, spatial, traffic, and meteorological data from Weibo	Lu et al., 2018
	Communication and Dissemination Mechanism	Warning dissemination	To study the use of codified hashtags relating to weather warnings in Italy	Social media	Twitter data	Grasso & Crisci, 2016
		Warning dissemination	To evaluate the use of a list of predefined codified hashtags for weather warnings in Italy	Social media	Twitter data	Grasso et al., 2017
Cyclone	Detection, Monitoring, Warning Services	Forecasting	To determine if social media and geo-location information can contribute to a more efficient early warning system and help with disaster assessment	Social media	Twitter data, Hurricane damage loss data	Wu & Cui, 2018
	Preparedness and Early Response Capacity	Response to warnings	To integrate local and scientific meteorological knowledge and actions within coconut farming communities in the Philippines	Local knowledge	Interviews with key stakeholders	Ton et al., 2017
Landslide	Disaster Risk Knowledge	Modelling	To present a crowdsourcing smartphone app for landslide reports which populates a landslide database	Crowdsourcing	Crowdsourced landslide reports from app users	Choi et al., 2018
		Modelling	To present a national landslide database in the UK which is partially populated with social media data to capture the impacts of landslides and for early detection of landslides	Social media	Twitter data	Pennington et al., 2015
Air quality	Detection, Monitoring, Warning Services	Forecasting	To explore the use of social media as a real-time data source for forecasting smog- related health hazards	Social media	Social media data and physical sensors data	Chen et al., 2017
Urban heat wave	Detection, Monitoring, Warning Services	Detection	To investigate the relationship between heat exposure and tweet volume over time	Social media	Twitter data	Jung & Uejio, 2017



detecting floods and capturing impacts from social media and crowdsourced data respectively. Jung and Uejio (2017) tested the effectiveness of measuring heat exposure on social media and consequently detecting urban heatwaves. Similarly, He and colleagues (2018) developed a crowdsourcing application to detect various weather hazards and to capture impacts to improve the decision-making of local governments. Henriksen and colleagues (2018) indicated the role of social media and crowdsourcing for both detection and forecasting of floods, while Rossi and colleagues (2018) assessed the feasibility of social media for flood detection and monitoring. Longmore and colleagues (2015) presented a conceptual crowdsourcing framework for collecting photos of severe weather hazards in the United States to improve weather monitoring by the National Weather Service. In Europe, Krennert and colleagues (2018) assessed the occurrence of crowdsourcing (either through specialised applications or social media) by national hydrological and meteorological services to capture severe weather observations and impacts for real-time warning verification and improvement.

VGI for forecasting alone was used for floods, cyclone damage, general severe weather traffic impacts, and air quality. Restrepo-Estrada and colleagues (2018) developed a methodology using social media for estimating rainfall runoff estimations and flood forecasting, while Smith, Liang, James, Lin, and Qiuhua Liang (2017) presented a real-time modelling framework to identify likely flooded areas using social media. Alternatively, Wu and Cui (2018) found that geolocated social media can help with disaster assessment, and for future forecasting. Lu and colleagues (2018) explored how social media might be used to forecast and generate traffic alerts due to severe weather. Likewise, Chen, Chen, Wu, Hu, and Pan (2017) explored social media for real-time forecasting of smog-related hazards.

Finally, three studies used VGI to monitor floods. Schnebele and Cervone (2013) crowdsourced from social media and other online media to monitor flood hazards and to create hazard maps, finding that the VGI is useful when satellite data is unavailable. Horita, de Albuquerque, Degrossi, Mendiondo, and Ueyama (2015) developed a framework to integrate crowdsourced flood observations with official sensor data. The authors found that the VGI made it possible to capture data from areas lacking flood sensors (Horita et al., 2015). Sadler, Goodall, Morsy, and Spencer (2018) crowdsourced street flooding reports to estimate flood

12

Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

severity for flood prediction, but the poor temporal and spatial coverage of the crowdsourced reports hindered the performance of the prediction model (Sadler et al., 2018).

Communication and Dissemination Mechanisms. Two studies were identified for the third EWS component, Communication and Dissemination Mechanisms, Both studies used VGI to assess warning dissemination via social media (namely Twitter) for general severe weather (Grasso & Crisci, 2016; Grasso, Crisci, Morabito, Nesi, Pantaleo, et al., 2017). Grasso and Crisci (2016) analysed codified hashtags of regions in Italy impacted by rainfall and found that codified hashtags for different regions effectively enable the sharing of useful information during severe weather events. Additionally, many tweets included geo-location information along with hazard information to update and complement official data. As such, the authors argued that institutions might adopt codified hashtags to improve the performance of systems for disseminating and retrieving information. Grasso and colleagues (2017) built on this work by adding more regions to their tweet analyses and emphasised the importance of institutions and warning services to promote codified hashtags for warnings to streamline message delivery and reach.

Preparedness and Early Response Capacity. For the last component, Preparedness and Early Response Capacity, only one study applied. Ton, Gaillard, Cadag, and Naing (2017) collected VGI in the form of local knowledge using interviews and guestionnaires with farmers to understand their response to cyclone warnings. In this process, the farmers identified economic, physical, social, and natural impacts of cyclone hazards. The authors found that while farmers forecast weather conditions and impacts based on their local knowledge, their confidence in the leadtime of their forecasts has declined due to changing climate conditions. As such, the authors argued for the integration of local knowledge with scientific forecasts to verify local knowledge-based forecasts and increase confidence

Multiple components. Two studies were found to fall into more than one EWS component. Allaire (2016) used VGI for Detecting, Monitoring, and Warning, assessing Communication and Dissemination Mechanisms, and for measuring Preparedness and Early Response capacities for flood hazards. Allaire (2016) found that social media was an effective tool for flood monitoring (falling in

the Detection, Monitoring, and Warning component), for receiving and spreading flood information (as a Communication and Dissemination Mechanism), and for receiving and spreading preparedness information, leading to reduced impacts (informing Preparedness and Early Response Capacity). Alternatively, Fdez-Arroyable and colleagues (2018) developed a mobile application to obtain individual vulnerabilities to meteorological changes (thus informing Disaster Risk Knowledge) and to provide personalised alerts based on the individual vulnerabilities to meteorological conditions (informing Detection, Monitoring, and Warning services).

VGI Platforms and Data Types

In this review, we broadly define VGI to include participatory mapping, participatory GIS, geo-located social media, and location-based local knowledge (de Albuquerque, Eckle, Herfort, & Zipf, 2016). Figure 2 shows the distribution of platforms discussed in each of the selected studies and to which component of the EWS framework they apply. The following section provides definitions of the platforms displayed in Figure 2 along with a description of how the VGI is used for severe weather warnings.

Geo-located social media. Geo-located social media refers to VGI that is posted online by users of Facebook, Twitter, Sina Weibo, Flickr, YouTube, Instagram, and SnapChat that has geographical location information associated to it. The heavy representation of social media (15 studies) demonstrates the growing popularity of these platforms as a data source for severe weather events (Tkachenko et al., 2017). The results indicate that social media is a valid tool for measuring the effectiveness of warning dissemination by following Twitter hashtags (Allaire, 2016; Grasso & Crisci, 2016; Grasso, Crisci, Morabito, Nesi, Pantaleo, et al., 2017; Taylor, Kox, & Johnston, 2018). The online platforms are also useful for early hazard detection and for estimating

	00	Ste Bat House	More Comme	station and the second	edonist and sand
Geo-located Social Media	2	9	3	1	15
Crowdsourcing	2	6			8
Crowdsourcing and Geo-located Social Media		3			3
Local Knowledge	1			1	2
Participatory Mapping/PGIS	4				4
Total	9	18	3	2	32

Figure 2. Distibution of VGI platforms used for each early warning system (EWS) framework component. Two studies fell into multiple components and have been counted for each EWS component that they apply to, which results in a total of 32, rather than 29.

trauma.massey.ac.nz

Harrison et al.

event magnitude for early warnings (Chen et al., 2017; Jung & Uejio, 2017; Restrepo-Estrada et al., 2018; Tkachenko et al., 2017). Reasons for collecting social media data were to increase coverage of the dataset(s), the ease of access and quantity of data available, realtime or near-real-time monitoring and collection, and the multi-directional communication during disaster enabled by social media (Allaire, 2016; Chen et al., 2017; Grasso & Crisci, 2016; Grasso, Crisci, Morabito, Nesi, & Pantaleo, 2017; Jung & Uejio, 2017; Pennington et al., 2015; Rossi et al., 2018; Saint-Martin et al., 2018; Smith et al., 2017; Wu & Cui, 2018).

Crowdsourcing applications and forms. Eight of the selected studies used crowdsourcing via mobile applications, reporting forms, or other active contributions (e.g., storm spotters). The crowdsourcing applications in the selected studies were used for hazard detection and monitoring and for developing personalised risk knowledge. These applications allow citizens to report the occurrence of hazards such as landslides (Choi et al., 2018; He et al., 2018) and to monitor hazards such as rainfall-induced floods (Horita et al., 2015) and storms (Krennert et al., 2018; Longmore et al., 2015). The ability to efficiently collect reports and monitor hazards in real-time, in a standardised format to ensure quality, and to increase the scale and resolution of hazard-related data were arguments made for using crowdsourcing as opposed to other VGI collection types (Choi et al., 2018; He et al., 2018; Henriksen et al., 2018; Horita et al., 2015; Longmore et al., 2015; Sadler et al., 2018; Sharma et al., 2016).

Participatory mapping and participatory GIS. In the selected studies, participatory mapping and participatory GIS were employed for severe weather risk assessments and hazard modelling. Lavers and Charlesworth (2018) engaged UK farmers in participatory mapping to identify flood impacts on their properties and subsequent opportunities for mitigation. Peters-Guarin et al. (2012) had locals in the Philippines map their historical knowledge of recurring floods and impacts for a risk assessment. In Taiwan, Hung and Chen (2013) consulted with locals and stakeholders to verify flood vulnerability maps. Participatory mapping and interviews were utilised by Rollason and colleagues (2018) to validate flood models using local knowledge and experiences. In all of these studies, the mapped information was entered into a GIS for further mapping and analysis, thus gualifying it as participatory GIS. Reasons for using participatory GIS

13

and participatory mapping over other types of VGI were formally recognising and integrating local knowledge in a systematic way, and supporting local engagement (Hung & Chen, 2013; Lavers & Charlesworth, 2018; Peters-Guarin et al., 2012; Rollason et al., 2018).

Local knowledge. For the purposes of this paper, we consider local knowledge as information gathered in participatory processes containing knowledge of the participants' local area and geography, that may or may not be translated onto a map. Just one selected study included local knowledge. After evaluating local knowledge of cyclone hazards and response capabilities to scientific knowledge, Ton and colleagues (2017) argued that local knowledge should be integrated with scientific meteorological knowledge for verification and to increase confidence in forecasts. The choice of using local knowledge for this study was to begin a dialogue between the locals and the meteorologists towards building trust (Ton et al., 2017).

Discussion

The results show that VGI is useful in all components of the early warning system (EWS) framework, but some platforms are more useful for specific components than are others. Furthermore, the different types of VGI have implications for supporting people-centred EWSs, which is a guiding principle for EWSs under the Sendai Framework.

Volunteered Geographic Information in Severe Weather Early Warning Systems

The purpose of this study is to determine the current and potential uses of VGI for severe weather warnings. We used the EWS framework to guide the analysis of the results.

The results from this literature review show that VGI has value in all four components of an EWS for severe weather hazards (Basher, 2006), but some forms of VGI are more useful for specific EWS components than are others (see Figure 3). Figure 3 is an update of Figure 1 based on the findings from this literature review to better represent how the different types of VGI inform or support the EWS components. For example, the majority of included studies used social media and crowdsourcing for hazard detection, monitoring, and early warning, while all of the included participatory mapping and participatory GIS studies used VGI for building disaster risk knowledge.

Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

Disaster Risk Knowledge	Detection, Monitoring, and Warning Services • Geo-located social media • Near-real-time hazard and impact detection, forecasting, and warning verification • Crowdsourcling • Near-real-time hazard and/or impact detection, forecasting, and warning verification • Estimate hazard and/or impact severity based on stundardised reports		
 Participatory mapping & participatory GIS Validate hazard and risk maps/models In-depth engagement and co-production of risk knowledge Integrate local, spatial knowledge Crowdsourcine 			
 Populate a hazard and impact database using standardised reports 			
Communication and Dissemination	Preparedness and Early Response		
Mechanism	Capacity		
 Geo-located social media Assess the spread of, and response to, warning messages 	Local Knowledge Understand knowledge and perceptions of local hazard risks and impacts, and preparedness and response capabilities Geo-located social media Inform local preparedness efforts with real- time hazard and impact information		

Figure 3. Volunteered Geographic Information for people-centred severe weather early warning systems.

The selected studies show that social media and crowdsourcing for severe weather are effective for early detection, monitoring, and verifying warnings (e.g., Harrison & Johnson, 2016; Henriksen et al., 2018; Krennert et al., 2018). The value of social media and crowdsourcing for EWSs lies in the real-time, or nearreal-time, hazard and impact detection, forecasting, and warning verification (Henriksen et al., 2018; Kox, Kempf, Lüder, Hagedorn, & Gerhold, 2018; Krennert et al., 2018). However, the papers included in this scoping review lack forward-thinking for integrating these tools into official EWSs which is a challenge for warning services and emergency management services (Haworth, 2016; Henriksen et al., 2018; Kox et al., 2018). Despite this challenge, some national hydrological and meteorological services and emergency management agencies in Europe and North America collect information from social media for detection, monitoring, and warning verification (Harrison & Johnson, 2016; Henriksen et al., 2018; Krennert et al., 2018; Pennington et al., 2015)

Social media supports multi-directional communication, which allows for both crowdsourcing and broadcasting severe weather information. While most of the selected social media studies demonstrated the value of social media for detection and early warning, two studies also indicated its utility for disseminating warnings and assessing the spread of, and response to, warning messages (Grasso & Crisci, 2016; Grasso, Crisci, Morabito, Nesi, Pantaleo, et al., 2017). This allows warning services to gauge the reach of their message, understand the responses to their message, and update

subsequent messages based on what they see on social media (Harrison & Johnson, 2016).

Before warnings are issued, knowledge of disaster risk is needed to be able to create tailored warnings. Participatory mapping and participatory GIS might be considered a long-term process for building knowledge and datasets for improving disaster risk knowledge as well as validating hazard and risk maps or models. While social media is valuable for real-time detection and communication, the participatory nature of participatory mapping enables more in-depth engagement with locals and communities in other areas of the EWS process to produce new knowledge (Haworth, 2018; Lavers & Charlesworth, 2018; Maskrey, Mount, Thorne, & Dryden, 2016; Peters-Guarin et al., 2012; Zolkafli, Brown, & Liu, 2017). Integrating local, spatial knowledge about disaster risk into an EWS through participatory mapping and participatory GIS fosters efforts towards people-centred EWSs as it translates local knowledge into usable and useful spatial data for risk analysis and for improved warnings (Basher, 2006; UNISDR, 2015).

These results support the findings from Marchezini and colleagues (2018), who presented a framework for bridging citizen science into EWSs. Like Marchezini and colleagues (2018), we found that VGI processes can bridge the gap between EWSs and audiences of warnings by incorporating local knowledge and personal experiences from stakeholders into the EWS components (see also Ton et al., 2017). This creates new data and unearths vulnerabilities at various scales (e.g., from the individual level to the community level; Haworth, 2018; Henriksen et al., 2018; Kox et al., 2018; Ton et al., 2017).

Implications for the different types of VGI. The results show that social media is a dominant platform for collecting VGI across severe weather hazards. Given the ease of access to, and the versatility of, social media (Harrison & Johnson, 2016), it is not surprising that social media is the most common platform used across hazards for collecting VGI (Granell & Ostermann, 2016). Social media is also now considered a "go-to" for collecting data because it is where the members of the public already are, thus groups or agencies looking to crowdsource do not have to do the heavy-lifting of creating a new app and attracting new users (Harrison & Johnson, 2016).

The perceived benefits of social media also come with some caveats. The data tend to be biased due to

trauma.massey.ac.nz

Harrison et al.

the uneven distribution of the social media user base (Granell & Ostermann, 2016; Harrison & Johnson, 2019). By relying on social media as a data source, those members of the public who are not present on social media are not represented in the data nor in the EWS process (i.e., the digital divide; Allaire, 2016; Harrison & Johnson, 2019). Additionally, tweet or post ambiguity and keyword selection for data-capture hinder data collection and analysis (Chen et al., 2017; Longmore et al., 2015; Tkachenko et al., 2017). Assimilating data of different formats into a database remains a challenge (Horita et al., 2015; Lu et al., 2018).

Capturing enough geo-located social media data is a constant challenge. It is widely known that only a small percentage of tweets contain geo-located information (Steed et al., 2019), Furthermore, the accessibility and availability of geo-located social media data are continuously limited. For example, Facebook does not offer an Application Programming Interface (API) to allow for researchers or media agencies to systematically collect Facebook posts, much less geo-located posts; it only offers an API for marketing and advertising agencies (Dubois, Zagheni, Garimella, & Weber, 2018; Thakur et al., 2018). In addition, in June 2019 Twitter announced plans to disable the geo-location feature for tweets due to its limited adoption by users and growing privacy concerns; however, the feature will still be available on photos taken within the Twitter mobile application (Benton, 2019; Khalid, 2019). While geo-located information on Instagram appears to be available for the moment (Arapostathis, 2019; Boulton, Shotton, & Williams, 2016), given the recent trends in the other major social media platforms, the continued availability and accessibility of this data in the future is uncertain.

A specialised crowdsourcing application can help to address some limitations found in social media. Crowdsourcing applications offer quality assurance, noise avoidance, application customisation, and citizen engagement (Choi et al., 2018; Longmore et al., 2015). On the other hand, crowdsourcing applications remain limited in the volume of participation due to public motivation to participate, the digital divide, and privacy concerns (Choi et al., 2018; Fdez-Arroyabe et al., 2018). Bias in reporting is also a concern, as contributors may over-exaggerate their personal experiences (Fdez-Arroyabe et al., 2018). Developing an application has the potential to streamline the integration of crowdsourced data into official processes, yet maintenance costs

15

impede the willingness of officials to do so (Choi et al., 2018).

Capturing and representing local knowledge through participatory mapping and participatory GIS may help in bridging the digital divide, ensuring data quality, and enabling data integration. Participatory mapping and participatory GIS also enable community engagement (Haworth et al., 2016; Lavers & Charlesworth, 2018; Peters-Guarin et al., 2012). Participatory mapping and participatory GIS can be done using paper-mapping, as was done by Rollason and colleagues (2018), Lavers and colleagues (2018), and Peters-Guarin and colleagues (2012), or through digital-mapping (Haworth et al., 2016). In addition to the value of the resulting information and data itself, the process of engaging with and between locals provides another level of value in the social context by strengthening social networks. growing social capital, and increasing civic participation (Haworth et al., 2016),

Participatory GIS and participatory mapping do not come without their own limitations. For example, participatory GIS appears to be more effective with small-scale local projects. This is because most of the data collected is at a local or small scale, resulting in poor spatial distribution if scaled-up to a larger area. This could lead to underrepresentation and potential biases in the participatory GIS data (Rollason et al., 2018). Nevertheless, the rich quality and the ease of integrating this VGI into official processes may outweigh this limitation if the study is well-designed and the data is used appropriately (Brabham, 2013; Lauriault & Mooney, 2014). Within the EWS context, these perceived benefits further the movement towards people-centred EWSs by incorporating knowledge and information produced by the people into warnings that are ultimately for them (UNISDR, 2015).

Conclusion

16

This paper conducted a scoping literature review and explored 29 journal papers published in academic journals and conference proceedings retrieved from EBSCO Discovery and Scopus. The literature review found that VGI plays various roles for severe weather early warning systems (EWSs). The examples from the selected studies show that VGI furthers the development of people-centered EWSs; it brings people, their knowledge, and their experiences into EWSs. Still, the current research captured in this scoping review lacks forward-thinking for integrating these tools into official Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

EWSs which is a challenge for warning services and emergency management services (Haworth, 2016; Henriksen et al., 2018; Kox et al., 2018).

In the always shifting EWSs landscape, a new type of severe weather EWS is emerging that is causing national meteorological and hydrological services and warning services to re-think their traditional warning practices. The World Meteorological Organization is advocating for the aforementioned services to adopt impact-based forecasts and warning systems (Fleming et al., 2015). Impact-based forecasts and warnings are meant to shift the focus from the physical hazard phenomena to the risk of impacts produced by the hazard, including communicating impacts in warning messages and building new warning thresholds based on risk of impact (Fleming et al., 2015; Morss, Cuite, Demuth, Hallman, & Shwom, 2018; Poolman, 2014; Potter et al., 2018; Robbins & Titley, 2018; Rogers, Kootval, & Tsirkunov, 2017; Sai et al., 2018). However, warning services have indicated a limited understanding of, and access to, the data required for developing impact-based forecasting and warning systems (Harrison et al., 2014; Kox et al., 2018; Obermeier & Anderson, 2014).

Future research would benefit from a systematic review of this topic area in the future. Additional research should investigate the data needs for impact-based forecasts and warnings and explore how VGI can help in meeting these data needs while also maintaining a people-centred focus. This would align with the goals of the World Meteorological Organization's High Impact Weather research programme (http://hiweather.net) which aims to improve the effectiveness of weather-related warnings in support of advances in weather prediction and forecasting (Zhang et al., 2019). While this literature review characterised the role of VGI within severe weather EWSs and demonstrated how it supports people-centred EWSs, future research can delve into the nature of the resulting data and how it might support impact-based forecast and warning systems. It should be noted that in spite of the popularity of collecting and using social media data, given the uncertainty of reliable access to social media data in the future (e.g., disestablishing the geolocation function on Twitter), it would be wise to minimise reliance on these platforms and consider additional VGI sources and collection processes to capture the desired information.

Acknowledgements

The authors thank the anonymous reviewers whose feedback greatly improved this manuscript. This research was supported by public research funding from the Government of New Zealand through GNS Science.

References

- Alexander, D. E. (2014). Social media in disaster risk reduction and crisis management. *Science and Engineering Ethics*, 20, 717-733. doi: 10.1007/s11948-013-9502-z
- Allaire, M. (2016). Disaster loss and social media: Can online information increase flood resilience? Water Resources Research, 52, 7408–7423. doi: 10.1002/2014WR016618
- Anderson-Berry, L., Achilles, T., Panchuk, S., Mackie, B., Canterford, S., Leck, A., & Bird, D. K. (2018). Sending a message: How significant events have influenced the warnings landscape in Australia. *International Journal* of Disaster Risk Reduction, 30, 5–17. doi: 10.1016/j. iidtr.2018.03.005
- Arapostathis, S. G. (2019). Instagrammers report about the deadly wildfires of East Attica, 2018, Greece: An introductory analytic assessment for disaster management purposes. In Proceedings of the 16th ISCRAM Conference (pp. 473–481). València, Spain. Retrieved from https:// github.com/h4t0n/
- Arias, P. A., Villegas, J. C., Machado, J., Serna, A. M., Vidal, L. M., Vieira, C., ... Mejía, Ó. A. (2016). Reducing social vulnerability to environmental change: Building trust through social collaboration on environmental monitoring. *Weather, Climate, and Society*, 8(1), 57–66. doi: 10.1175/ WCAS-D-15-0049.1
- Arksey, H., & O'Malley, L. (2005). Scoping studies: Towards a methodological framework. *International Journal* of Social Research Methodology, 8(1), 19–32. doi: 10.1080/1364557032000119616
- Assumpção, T. H., Popescu, I., Jonoski, A., & Solomatine, D. P. (2018). Citizen observations contributing to flood modelling: Opportunities and challenges. *Hydrology and Earth System Sciences Discussions*, 22, 1473–1489. doi: 10.5194/hess-2017-456
- Basher, R. (2006). Global early warning systems for natural hazards: Systematic and people-centred. *Philosophical Transactions of the Royel Society*, 364, 2167–2182. doi: 10.1098/rsta.2006.1819
- Benton, J. (2019, June 19). Twitter is removing preciselocation tagging on tweets — a small win for privacy but a small loss for journalists and researchers. *NiemanLab*. Retrieved from http://niemanlab.org
- Boulton, C. A., Shotton, H., & Williams, H. T. P. (2016). Using social media to detect and locate wildfires. In 10th International AAAI Conference on Web and Social Media (ICWSM 16), W8: Social Web for Environmental and Ecological Monitoring. Cologne, Germany. Retrieved from www.ushahidi.com/
- Brabham, D. C. (2013). Using crowdsourcing in government. California, US: IBM Center for The Business of Government.
- Brown, G., & Kyttä, M. (2014). Key issues and research priorities for public participation GIS (PPGIS): A synthesis based on empirical research. *Applied Geography*, 46, 122–136. doi: 10.1016/j.apgeog.2013.11.004

trauma.massey.ac.nz

- Carley, K. M. K. M., Malik, M., Landwehr, P. M. P. M., Pfeffer, J., & Kowalchuck, M. (2016). Crowd sourcing disaster management: The complex nature of Twitter usage in Padang Indonesia. *Safety Science*, *90*, 48–61. doi: 10.1016/j.ssci.2016.04.002
- Chen, J., Chen, H., Wu, Z., Hu, D., & Pan, J. Z. (2017). Forecasting smog-related health hazard based on social media and physical sensor. *Information Systems*, 64, 281–291. doi: 10.1016/j.is.2016.03.011
- Ching, P. K., Carr de los Reyes, V., Sucaldito, M. N., & Tayag, E. (2015). Responding to Typhoon Haiyan in the Philippines. Western Pacific Surveillance and Response Journal, 6(Supplement 1), 1–4.
- Choi, C. E., Cui, Y., & Zhou, G. G. D. (2018). Utilizing crowdsourcing to enhance the mitigation and management of landslides. *Landslides*, *15*, 1889-1899. doi: 10.1007/ s10346-018-1034-9
- Cooper, A. K., Coetzee, S., & Kourie, D. G. (2018). Volunteered geographical information, crowdsourcing, citizen science and neogeography are not the same. *Proceedings of the ICA*, 1, 1–8. doi: 10.5194/ica-proc-1-131-2018
- de Albuquerque, J. P., Eckle, M., Herfort, B., & Zipf, A. (2016). Crowdsourcing geographic information for disaster management and improving urban resilience: An overview of recent developments and lessons learned. In C. Capineri, M. Haklay, H. Huang, V. Antoniou, J. Kettunen, F. Ostermann, & R. Purves (Eds.), European handbook of crowdsourced geographic information (pp. 309–321). London: Ubiquity Press.
- de Albuquerque, J. P., Horita, F. E. A., Degrossi, L. C., Rocha, R. dos S., Camargo de Andrade, S., Restrepo-Estrada, C., & Leyh, W. (2017). Leveraging Volunteered Geographic Information to improve disaster resilience. In A. Ballatore, B. Bennett, C. Claramount, C. J. Davis, S. Di Martino, A. Klippel, ... B. Schoen-Phelan (Eds.), Volunteered Geographic Information and the future of geospatial data (pp. 158–184). Pennsylvania, US: IGI Global.
- Dubois, A., Zagheni, E., Garimella, K., & Weber, I. (2018). Studying migrant assimilation through Facebook interests. In S. Staab, O. Koltsova, & D. Ignatov (Eds.), Social informatics (pp. 51–60). Switzerland: Springer, Cham.
- Elwood, S., Goodchild, M. F., & Sui, D. Z. (2012). Researching Volunteered Geographic Information: Spatial data, geographic research, and new social practice. *Annals* of the Association of American Geographers, 102(3), 571–590. doi: 10.1080/00045608.2011.595657
- Falagas, M. E., Pitsouni, E. I., Malietzis, G. A., & Pappas, G. (2008). Comparison of PubMed, Scopus, Web of Science, and Google Scholar: Strengths and weaknesses. *The FASEB Journal*, 22(2), 338–342. doi: 10.1096/fj.07-9492LSF
- Fdez-Arroyabe, P., Lecha Estela, L., & Schimt, F. (2018). Digital divide, biometeorological data infrastructures and human vulnerability definition. *International Journal of Biometeorology*, 62(5), 733–740. doi: 10.1007/s00484-017-1398-x
- Fleming, G., Rogers, D., Davies, P., Jacks, E., Milton, J. A., Honore, C., ... Goolaup, P. (2015). WMO Guidelines on Multi-Hazard Impact-Based Forecast and Warning Services [PDF file]. Retrieved from www.wmo.int/pages/ prog/www/DPFS/Meetings/ET-OWFPS_Montreal2016/ documents/WMOGuidelinesonMulti-hazardImpactbasedForecastandWarningServices.pdf

17

Harrison et al.

- Forrester, J., & Cinderby, S. (2011). Aguide to using community mapping and participatory-GIS [PDF file]. Retrieved from https://tweedforum.org/wp-content/uploads/2018/09/ Borderlands-Community-Mapping-Guide.pdf
- Gaillard, J. C., & Pangilinan, M. L. C. J. D. (2010). Participatory mapping for raising disaster risk awareness among the youth. *Journal of Contingencies and Crisis Management*, 18(3), 175–179. doi: 10.1111/j.1468-5973.2010.00614.x
- Gall, M., & Cutter, S. L. (2016). Understanding disaster risk through loss data. Washington, D.C.: International Bank for Reconstruction and Development/The World Bank. doi: 10.1016/0301-4207(78)90055-7
- Garcia, C., & Fearnley, C. J. (2012). Evaluating critical links in early warning systems for natural hazards. *Environmental Hazards*, 11(2), 123–137. doi: 10.1080/17477891.2011.609877
- GFDRR. (2016). Implementing multi-hazard impact-based forecast and warning services [PDF file]. Retreived from www.gfdrr.org/sites/default/files/publication/ Report%20on%20Implementing%20Multi%20Hazard%20 Impact%20Based%20Forecast%20and%20Warning%20 Services%2017-02-10%20Final.pdf
- Golnaraghi, M. (Ed.). (2012). Institutional partnerships in multihazard early warning systems. Berlin: Springer.
- Goodchild, M. F. (2007). Citizens as sensors: The world of Volunteered Geography. *GeoJournal*, 69(4), 211–221.
- Goodchild, M. F., & Glennon, J. A. (2010). Crowdsourcing geographic information for disaster response: A research frontier. *International Journal of Digital Earth*, 3(3), 231– 241. doi: 10.1080/17538941003759255
- Granell, C., & Ostermann, F. O. (2016). Beyond data collection: Objectives and methods of research using VGI and geo-social media for disaster management. *Computers, Environment and Urban Systems*, 59(2016), 231–243. doi: 10.1016/j.compenvurbsys.2016.01.006
- Grant, M. J., & Booth, A. (2009). A typology of reviews: An analysis of 14 review types and associated methodologies. *Health Information and Libraries Journal*, 26(2), 91–108. doi: 10.1111/j.1471-1842.2009.00848.x
- Grasso, V., & Crisci, A. (2016). Codified hashtags for weather warning on Twitter: An Italian case study. *PLoS Currents Disasters*, *8*, 1–30. doi: 10.1371/currents.dis.967e71514 ecb92402eca3bdc9b789529
- Grasso, V., Crisci, A., Morabito, M., Nesi, P., & Pantaleo, G. (2017). Public crowdsensing of heat waves by social media data. Advances in Science and Research, 14, 217–226. doi: 10.5194/asr-14-217-2017
- Grasso, V., Crisci, A., Morabito, M., Nesi, P., Pantaleo, G., Zaza, I., & Gozzini, B. (2017). Italian codified hashtags for weather warning on Twitter - who is really using them? *Advances in Science and Research*, 14, 63–69. doi: 10.5194/asr-14-63-2017
- Guz, A. N., & Rushchitsky, J. J. (2009). Scopus: A system for the evaluation of scientific journals. *International Applied Mechanics*, 45(4), 351–362. doi: 10.1007/s10778-009-0189-4
- Haklay, M. (2013). Citizen science and Volunteered Geographic Information: Overview and typology of participation. In D. Sul, S. Elwood, & M. Goodchild (Eds.), *Crowdsourcing geographic knowledge: Volunteered Geographic Information (VGI) in theory and practice* (pp. 105–122). Dordrecht: Springer Netherlands.

18

Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

- Haklay, M. (2017). Volunteered Geographic Information and citizen science. Retrieved from <u>www.opensignal.com/</u>
- Haklay, M., Antoniou, V., & Basiouka, S. (2014). Crowdsourced geographic information use in government. Retrieved from http://discovery.ucl.ac.uk/1433169/
- Harrison, J., Mccoy, C., Bunting-Howarth, K., Sorensen, H., Williams, K., & Ellis, C. (2014). Evaluation of the National Weather Service Impact-based Warning tool [PDF file]. Retrieved from https://publications.aqua.wisc.edu/product/ evaluation-of-the-national-weather-service-impact-basedwarning-tool/
- Harrison, S., & Johnson, P. A. (2016). Crowdsourcing the disaster management cycle. International Journal of Information Systems for Crisis Response and Management, 8(4), 17–40. doi: 10.4018/JISCRAM.2016100102
- Harrison, S., & Johnson, P. A. (2019). Challenges in the adoption of crisis crowdsourcing and social media in Canadian emergency management. *Government Information Quarterly*, 36, 501–509. doi: 10.1016/J. GIQ.2019.04.002
- Harrison, S., Silver, A., & Doberstein, B. (2015). Poststorm damage surveys of tornado hazards in Canada: Implications for mitigation and policy. *International Journal* of Disaster Risk Reduction, 13, 427–440. doi: 10.1016/j. ijdrr.2015.08.005
- Haworth, B. (2016). Emergency management perspectives on volunteered geographic information: Opportunities, challenges and change. Computers, Environment and Urban Systems, 57, 189–198. doi: 10.1016/j. compenvurbsys.2016.02.009
- Haworth, B. (2018). Implications of Volunteered Geographic Information for disaster management and GIScience: A more complex world of volunteered geography. *Annals* of the American Association of Geographers, 108(1), 226–240. doi: 10.1080/24694452.2017.1321979
- Haworth, B., & Bruce, E. (2015). A review of Volunteered Geographic Information for disaster management. Geography Compass, 5(9), 237–250. doi: 10.1111/ gec3.12213
- Haworth, B., Bruce, E., & Middleton, P. (2012). Emerging technologies for risk reduction: assessing the potential use of social media and VGI for increasing community engagement. *Australian Journal of Emergency Management*, 30(3), 36–41.
- Haworth, B., Whittaker, J., & Bruce, E. (2016). Assessing the application and value of participatory mapping for community bushfire preparation. *Applied Geography*, 76, 115–127. doi: 10.1016/j.apgeog.2016.09.019
- He, C., Ju, N., Xu, Q., Li, Y., & Zhao, J. (2018). Designing an Android-based application for geohazard reduction using citizen-based crowdsourcing data. *Mobile Information Systems*, 2018, 1-11.
- Henriksen, H. J., Roberts, M. J., van der Keur, P., Harjanne, A., Egilson, D., & Alfonso, L. (2018). Participatory early warning and monitoring systems: A Nordic framework for web-based flood risk management. *International Journal* of *Disaster Risk Reduction*, 31, 1295–1306. doi: 10.1016/j. ijdrr.2018.01.038
- Horita, F. E. A., de Albuquerque, J. P., Degrossi, L. C., Mendiondo, E. M., & Ueyama, J. (2015). Development of a spatial decision support system for flood risk management in Brazil that combines volunteered geographic information with wireless sensor networks.

mulh

Computers and Geosciences, 80, 84–94. doi: 10.1016/j. cageo.2015.04.001

- Horita, F. E. A., de Albuquerque, J. P., Marchezini, V., & Mendiondo, E. M. (2016). A qualitative analysis of the early warning decision-making process in disaster management. In 13th International Conference on Information Systems for Crisis Response and Management (ISCRAM 2016), (May 2016), 1–8.
- Horita, F. E. A., Degrossi, L. C., Assis, L. F. ., Zipf, A., & de Albuquerque, J. P. (2013). The use of Volunteered Geographic Information and crowdsourcing in disaster management: Asystematic literature review. In *Proceedings* of the Nineteenth Americas Conference on Information Systems (pp. 1–10).
- Horita, F. E. A., Vilela, R. B., Martins, R. G., Bressiani, D. A., Palma, G., & de Albuquerque, J. P. (2018). Determining flooded areas using crowd sensing data and weather radar precipitation: A case study in Brazil. In 15th International Conference on Information Systems for Crisis Response and Management (ISCRAM 2018) (May 2018), 1030– 1039. Retrieved from www.scopus.com/inward/record. uri?eid=2-s2.0-85060678817&partnerID=40&md5=7116 0a6fc9ab22da45c84ad5116adbb9
- Hung, H. C., & Chen, L. Y. (2013). Incorporating stakeholders' knowledge into assessing vulnerability to climatic hazards: Application to the river basin management in Taiwan. *Climatic Change*, 120(1–2), 491–507. doi: 10.1007/ s10584-013-0819-z
- Jones, S., & Golding, B. (2014). HIWeather: A research activity on High Impact Weather within the World Weather Research Programme [PDF file]. Retrieved from www.wmo.int/pages/prog/arep/wwrp/new/documents/ HIWeatherImplementationPlan.pdf
- Jung, J., & Uejio, C. K. (2017). Social media responses to heat waves. International Journal of Biometeorology, 61(7), 1247–1260. doi: 10.1007/s00484-016-1302-0
- Khalid, A. (2019, June 19). Twitter removes precise geotagging option from tweets. *Engadget*. Retrieved from <u>www.engadget.com</u>
- Klonner, C., Marx, S., Usón, T., Porto de Albuquerque, J., Höfle, B., Klonner, C., ... Höfle, B. (2016). Volunteered Geographic Information in natural hazard analysis: A systematic literature review of current approaches with a focus on preparedness and mitigation. *ISPRS International Journal of Geo-Information*, 5(7), 103. doi: 10.3390/ ijql5070103
- Kox, T., Kempf, H., Lüder, C., Hagedorn, R., & Gerhold, L. (2018). Towards user-orientated weather warnings. *International Journal of Disaster Risk Reduction*, 30, 74-80. doi: 10.1016/j.ijdrr.2018.02.033
- Krennert, T., Pistotrik, G., Kaltenberger, R., & Csekits, C. (2018). Crowdsourcing of weather observations at national meteorological and hydrological services in Europe. Advances in Science and Research, 15, 71–76. doi: 10.5194/asr-15-71-2018
- Lauriault, T., & Mooney, P. (2014). Crowdsourcing: A geographic approach to public engagement. Retrieved from https://papers.srn.com/sol3/papers.cfm?abstract_ id=2518233.
- Lavers, T., & Charlesworth, S. (2018). Opportunity mapping of natural flood management measures: A case study from the headwaters of the Warwickshire-Avon. Environmental Science and Pollution Research, 25(20), 19313–19322. doi: 10.1007/s11356-017-0418-z

trauma.massey.ac.nz

- Longmore, S., Miller, S., Bikos, D., Lindsey, D., Szoke, E., Molenar, D., ... Knaff, J. (2015). An automated mobile phone photo relay and display concept applicable to operational severe weather monitoring. *Journal of Atmospheric and Oceanic Technology*, 32(7), 1356–1363. doi: 10.1175/JTECH-D-14-00230.1
- Lu, H., Zhu, Y., Shi, K., Lv, Y., Shi, P., & Niu, Z. (2018). Using adverse weather data in social media to assist with citylevel traffic situation awareness and alerting. *Applied Sciences*, 8(7), 1193-1210. doi: 10.3390/app8071193
- Marchezini, V., Horita, F. E. A., Matsuo, P. M., Trajber, R., Trejo-Rangel, M. A., & Olivato, D. (2018). A review of studies on Participatory Early Warning Systems (P-EWS): Pathways to support citizen science initiatives. *Frontiers* in *Earth Science*, 6, 1–18. doi: 10.3389/feart.2018.00184
- Maskrey, S. A., Mount, N. J., Thorne, C. R., & Dryden, I. (2016). Participatory modelling for stakeholder involvement in the development of flood risk management intervention options. Environmental Modelling and Software, 82, 275–294. doi: 10.1016/j.envsoft.2016.04.027
- Meissen, U., & Fuchs-Kittowski, F. (2014). Crowdsourcing in early warning systems. In Proceedings - 7th International Congress on Environmental Modelling and Software: Bold Visions for Environmental Modelling, iEMSs 2014 (pp. 326–332).
- Morgan, G., Fischhoff, B., Bostrom, A., Lave, L., & Atman, C. J. (1992). Communicating risk to the public. *Environmental Science & Technology*, 26(11), 2048–2056.
- Morss, R. E., Cuite, C. L., Demuth, J. L., Hallman, W. K., & Shwom, R. L. (2018). Is storm surge scary? The influence of hazard, impact, and fear-based messages and individual differences on responses to hurricane risks in the U.S. International Journal of Disaster Risk Reduction, 30, 44– 58. doi: 10.1016/j.ijdrr.2018.01.023
- Obermeier, H., & Anderson, M. R. (2014). Verification and analysis of impact-based tornado warnings in the central region of the National Weather Service. *Electronic Journal* of Severe Storms Meteorology, 10(1), 1–20.
- Paré, G., Trudel, M.-C., Jaana, M., & Kitsiou, S. (2015). Synthesizing information systems knowledge: A typology of literature reviews. *Information & Management*, 52, 183–199. doi: 10.1016/j.im.2014.08.008
- Pennington, C., Freeborough, K., Dashwood, C., Dijkstra, T., & Lawrie, K. (2015). The National Landslide Database of Great Britain: Acquisition, communication and the role of social media. *Geomorphology*, 249, 44–51. doi: 10.1016/j. geomorph.2015.03.013
- Peters-Guarin, G., Mccall, M. K., & van Westen, C. (2012). Coping strategies and risk manageability: Using participatory geographical information systems to represent local knowledge. *Disasters*, 36(1), 1–27. doi: 10.1111/j.1467-7717.2011.01247.x
- Poblet, M., Garcia-Cuesta, E., & Casanovas, P. (2014). Crowdsourcing tools for disaster management: A review of platforms and methods. In Al approaches to the complexity of legal systems (pp. 261–274). Germany: Springer.
- Poolman, E. R. (2014). A probabilistic impact-focussed early warning system for flash floods in support of disaster management in South Africa. South Africa: University of Pretoria.
- Potter, S. H., Kreft, P. V., Milojev, P., Noble, C., Montz, B., Dhellemmes, A., ... Gauden-Ing, S. (2018). The influence of impact-based severe weather warnings on risk perceptions and intended protective actions. *International*

19

Harrison et al.

Journal of Disaster Risk Reduction, 30, 34-43. doi: 10.1016/j.ijdrr.2018.03.031

- Restrepo-Estrada, C., de Andrade, S. C., Abe, N., Fava, M. C., Mendiondo, E. M., & de Albuquerque, J. P. (2018). Geo-social media as a proxy for hydrometeorological data for streamflow estimation and to improve flood monitoring. *Computers and Geosciences*, *111*, 148–158. doi: 10.1016/j.cageo.2017.10.010
- Reynolds, B., & Quinn, S. C. (2008). Effective communication during an influenza pandemic: The value of using a crisis and emergency risk communication framework. *Health Promotion Practice*, 9(4), 13S-17S. doi: 10.1177/1524839908325267
- Reynolds, B., & Seeger, M. W. (2005). Crisis and emergency risk communication as an integrative model. *Journal of Health Communication*, 10(1), 43–55. doi: 10.1080/10810730590904571
- Robbins, J. C., & Titley, H. A. (2018). Evaluating high-impact precipitation forecasts from the Met Office Global Hazard Map using a global impact database. *Meteorological Applications*, 25(4), 548–560. doi: 10.1002/met.1720
- Roche, S., Propeck-Zimmermann, E., & Mericskay, B. (2011). GeoWeb and crisis management: Issues and perspectives of volunteered geographic information. *GeoJournal*, 78(1), 21–40. doi: 10.1007/s10708-011-9423-9
- Rogers, D. P., Kootval, H., & Tsirkunov, V. V. (2017). Early warning, resilient infrastructure and risk transfer. In *Bridging science and policy implication for managing climate extremes* (pp. 65–79). Retrieved from <u>www.gfdrr.</u> org
- Rollason, E., Bracken, L. J., Hardy, R. J., & Large, A. R. G. (2018). The importance of volunteered geographic information for the validation of flood inundation models. *Journal of Hydrology*, 562, 267–280. doi: 10.1016/j. jhydrol.2018.05.002
- Rossi, C., Acerbo, F. S., Ylinen, K., Juga, I., Nurmi, P., Bosca, A., ... Alikadic, A. (2018). Early detection and information extraction for weather-induced floods using social media streams. *International Journal of Disaster Risk Reduction*, 30, 145–157. doi: 10.1016/j.ijdrr.2018.03.002
- Sadler, J. M., Goodall, J. L., Morsy, M. M., & Spencer, K. (2018). Modeling urban coastal flood severity from crowdsourced flood reports using Poisson regression and Random Forest. *Journal of Hydrology*, 559, 43–55. doi: 10.1016/j.jhydrol.2018.01.044
- Sai, F., Cumiskey, L., Weerts, A., & Bhattacharya, B. (2018). Towards impact-based flood forecasting and warning in Bangladesh: A case study at the local level in Sirajganj district. Retrieved from www.nat-hazards-earth-syst-scidiscuss.net/nhess-2018-26/
- Saint-Martin, C., Javelle, P., & Vinet, F. (2018). DamaGIS: A multisource geodatabase for collection of flood-related damage data. *Earth System Science Data*, *10*(2), 1019– 1029. doi: 10.5194/essd-10-1019-2018
- Schnebele, E., & Cervone, G. (2013). Improving remote sensing flood assessment using volunteered geographical data. *Natural Hazards and Earth System Science*, *13*(3), 669–677. doi: 10.5194/nhess-13-669-2013
- Scolobig, A., Prior, T., Schröter, D., Jörin, J., & Patt, A. (2015). Towards people-centred approaches for effective disaster risk management: Balancing rhetoric with reality. *International Journal of Disaster Risk Reduction*, 12, 202–212. doi: 10.1016/j.ijdrr.2015.01.006

20

Australasian Journal of Disaster and Trauma Studies Volume 24, Number 1

- Sebastian, A., Lendering, K., Kothius, B., Brand, N., Jonkman, S. N., van Gelder, P., ... Lhermitte, S. (2017). Hurricane Harvey Report: A fact-finding effort in the direct aftermath of Hurricane Harvey in the Greater Houston Region. Delft: Delft University Publishers.
- Sharma, V. K., Rao, G. S., Amminedu, E., Nagamani, P. V., Shukla, A., Rao, K. R. M., & Bhanumurthy, V. (2016). Event-driven flood management: design and computational modules. *Geo-Spatial Information Science*, 19(1), 39–55. doi: 10.1080/10095020.2016.1151212
- Simon, T., Goldberg, A., & Adini, B. (2015). Socializing in emergencies - A review of the use of social media in emergency situations. *International Journal of Information Management*, 35(5), 609-619. doi: 10.1016/j. ijinfomgt.2015.07.001
- Slavkovikj, V., Verstockt, S., Van Hoecke, S., & Van de Walle, R. (2014). Review of wildfire detection using social media. *Fire Safety Journal*, 68, 109–118. doi: 10.1016/J. FIRESAF.2014.05.021
- Smith, L., Liang, Q., James, P., Lin, W., & Qiuhua Liang, C. (2017). Assessing the utility of social media as a data source for flood risk management using a real-time modelling framework. *Journal of Flood Risk Management*, 10(3), 370–380. doi: 10.1111/jfr3.12154
- Steed, R. J., Fuenzalida, A., Bossu, R., Bondár, I., Heinloo, A., Dupont, A., ... Strollo, A. (2019). Crowdsourcing triggers rapid, reliable earthquake locations. *Science Advances*, 5(4), 1-6. doi: 10.1126/sciadv.aau9824
- Tan, M. L., Prasanna, R., Stock, K., Hudson-Doyle, E., Leonard, G., & Johnston, D. (2017). Mobile applications in crisis informatics literature: A systematic review. *International Journal of Disaster Risk Reduction*, 24, 297–311. doi: 10.1016/j.ijdrr.2017.06.009
- Taylor, A. L., Kox, T., & Johnston, D. (2018). Communicating high impact weather: Improving warnings and decision making processes. *International Journal of Disaster Risk Reduction*, 30, 1–4. doi: 10.1016/j.ijdrr.2018.04.002
- Thakur, G., Sims, K., Mao, H., Piburn, J., Sparks, K., Urban, M., ... Bhaduri, B. (2018). Utilizing geo-located sensors and social media for studying population dynamics and land classification. In S.-L. Shaw & D. Sui (Eds.), *Human dynamics research in smart and connected communities* (pp. 13–40). Switzerland: Springer, Cham.
- Tkachenko, N., Jarvis, S., & Procter, R. (2017). Predicting floods with Flickr tags. *PLoS ONE*, *12*(2), 1-13. doi: 10.1371/journal.pone.0172870
- Ton, K. T., Gaillard, J. C., Cadag, J. R., & Naing, A. (2017). It takes two to tango: Integrating meteorological knowledge and actions for disaster risk reduction. *Climate and Development*, 9(6), 479–492. doi: 10.1080/17565529.2016.1174658
- UNISDR. (2005). Hyogo Framework for Action 2005-2015 [PDF file]. Retrieved from www.unisdr.org/files/1037_ hyogoframeworkforactionenglish.pdf
- UNISDR. (2015). Sendai Framework for Disaster Risk Reduction. doi: A/CONF.224/CRP.1
- Veil, S., Reynolds, B., Sellnow, T. L., & Seeger, M. W. (2008). CERC as a theoretical framework for research and practice. *Health Promotion Practice*, 9(4), 26-34. doi: 10.1177/1524839908322113
- Wagenmaker, R., Weaver, J., Garnet, G., Perry, B., Spinner, J., & Mann, G. (2011). NWS Central Region Service

WWW AM

Harrison et al.

Assessment Joplin, Missouri, Tornado – May 22, 2011. Missouri, US: National Weather Service.

- Weyrich, P., Scolobig, A., Bresch, D. N., & Patt, A. (2018). Effects of impact-based warnings and behavioural recommendations for extreme weather events. *Weather, Climate, and Society, 10,* 781–796. doi: 10.1175/ WCAS-D-18-0038.1
- WMO. (2017). Joint meeting of the Commission For Basic Systems Open Programme Area Group on Public Weather Services Delivery (Opag-Pwsd) Expert Teams on Impact of Multi- Hazard Prediction and Communication (Et/ Impact) and on Services and Products Innovation and Improvement. Switzerland: WMO.
- WMO. (2018). Multi-hazard early warning systems: A checklist. Switzerland: WMO.
- Wu, D., & Cui, Y. (2018). Disaster early warning and damage assessment analysis using social media data and geolocation information. *Decision Support Systems*, *111*, 48–59. doi: 10.1016/j.dss.2018.04.005
- Zhang, Q., Li, L., Ebert, B., Golding, B., Johnston, D., Mills, B., ... Keller, J. (2019). Increasing the value of weatherrelated warnings. *Science Bulletin*, 64(10), 647–649. doi: 10.1016/j.scib.2019.04.003
- Zolkafii, A., Brown, G., & Liu, Y. (2017). An evaluation of Participatory GIS (PGIS) for land use planning in Malaysia. *The Electronic Journal of Information Systems* in Developing Countries, 83, 1–23.



Appendix L: Chapter Seven Publication

International Journal of Disaster Risk Reduction 66 (2021) 102619



journal homepage: www.elsevier.com/locate/ijdrr



'Where oh where is the data?': Identifying data sources for hydrometeorological impact forecasts and warnings in Aotearoa New Zealand

Sara E. Harrison^{a, b,*}, Sally H. Potter^b, Raj Prasanna^a, Emma E.H. Doyle^a, David Johnston^a ^a Massey University, New Zealand ^b GNS Science, New Zealand

ABSTRACT

Keywords: Impact-based warnings Severe weather Risk data Impacts Vulnerability Exposure

ARTICLE INFO

Early Warning Systems are a key component to building preparedness and response capacities to hydrometeo-rological hazards that continue to affect people worldwide. Notable historic events have revealed gaps in current hazard-based warning systems. Impact Forecasts and Warnings (IFWs) have been proposed to fill these communication gaps by re-centring the warning thresholds and language around the consequences, or impacts, of the hazard(s), rather than just the physical characteristics. However, research has shown that implementing IFWs requires not just hazard data, but also data on impacts, vulnerability, and exposure to understand the risk of impacts

Using Grounded Theory Methodology, we conducted a series of interviews with users and creators of hazard, impact, vulnerability, and exposure (HIVE) data to identify data sources and understand how these data are collected and created to support the implementation of IFWs. We focus the study on the New Zealand context to support the country's efforts towards implementing IFWs.

Our findings indicate that many sources for HIVE data exist that are collected for other uses (such as for disaster/emergency response efforts, and for research) and have relevant applications for IFWs. Our findings further suggest that priorities, motivation, and interest within organisations influence how well data is collected. Moreover, agencies tend to prefer official data, but official data has limitations that unofficial data may address, such as timeliness. To that end, a tension exists between the timeliness and trustworthiness of data needed for emergency response and warnings.

1. Introduction

The last two decades have seen a paradigm shift in disaster risk reduction from reactive post-disaster response and recovery to proactive preparedness and mitigation. Early Warning Systems (EWS) are a key component for better preparedness [1]. Past severe weather events exposed major communication gaps between meteorologists and warn-ing services and target audiences, resulting in widespread losses including death, injuries, and damage. For example, following Typhoon Haiyan, which resulted in over 6,293 deaths, 28,689 injuries, and 1,061 missing people in the Philippines [2], it was found that 88% of warning recipients did not understand messages about 'storm surge' and 95% of warning recipients did not evacuate because they did not expect the storm to be so catastrophic [3]. It was recommended that "warning messages ... should be conveyed in terms understood by the population at risk" ([3]; p. 34).

This communication gap is a result of both technical failings and human behaviour. The World Meteorological Organization (WMO) posited that meteorologists and warning services do not typically consider the warning audiences' current state of vulnerability and exposure at the time of the warning or at the expected time of impact [4]. Furthermore, warning audiences fail to understand and respond to the warnings effectively due to ambiguous terminology [3], lack of trust in the warning system and service provider [5], and warning fatigue [6]. As such, EWSs continue to evolve in attempts to reduce the effects of these factors

In the hydrometeorological space, Impact Forecasts and Warning (IFW) systems are an advancement of traditional hazard-based EWSs.

* Corresponding author. Massey University, New Zealand. E-mail address: s.harrison@ma ssey.ac.nz (S.E. Harrison).

tps://doi.org/10.1016/j.ijdrr.2021.102619

Received 1 July 2021; Received in revised form 29 September 2021; Accepted 30 September 2021

Available online 5 October 2021 2212-4209/© 2021 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license

S.E. Harrison et al

IFW systems provide an opportunity to integrate knowledge and understanding of exposure, vulnerability, and impacts into an EWS to build new warning thresholds that better align with the position, needs, and capabilities of target audiences [7]. However, challenges have been identified around identifying and accessing the required data sources for IFWs [8]. [7] identified the specific needs and uses for hazard, impact, vulnerability, and exposure (HIVE) data in an IFW system. We continue this work by identifying sources for these required datasets. We next present a review of existing HIVE data sources from the literature, identifying data gaps that we explore further through a series of key-informant interviews.

1.1. Impact forecasting and warning data gaps and sources

Warning agencies and meteorologists are challenged with accessing appropriate data to support the decision-making required for IFW systems [9]. The lack of impact, vulnerability, and exposure data is a major obstacle to implementing IFWs [10]. Beyond IFWs, these data are also needed for enhancing our understanding of disaster risks and subsequent mitigation and reduction, as per the Sendai Framework for Disaster Risk Reduction [1].

The following literature review identifies some existing sources of HIVE data for IFWs. These data sources and datasets are summarised in Table 1 and described further in the following sections.

1.1.1. Hazards

For this study, hazard data refers to meteorological, hydrological, and hydrogeological data. For example, windspeeds and rainfall amounts may be considered meteorological hazard data, while river levels would be hydrological hazard data, and slope data are useful for landslide hazards (i.e., hydrogeological hazards).

Understanding of severe weather hazards primarily comes from hydrometeorological observations and measurements usually gathered quantitatively through specialised equipment, such as rain gauges, river gauges, and remote sensing. This information helps to estimate the timine, direction, and magnitude of the hazard for forecasts [11].

timing, direction, and magnitude of the hazard for forecasts [11]. Forecasts are another form of hazard data. Multiple Numerical Weather Prediction (NWP) models are run using observational data to determine probabilities of future weather patterns based on running multiple simulations [12]. This probabilistic forecasting approach allows for the calculation of likelihoods of the weather phenomena to occur, and thus an overall confidence level of the forecast [12]. More observational data can increase the confidence of weather forecasts, however, this confidence decreases as forecast lead time increases [13].

In places where hydrometeorological monitoring network coverage is limited or lacking, crowdsourcing and citizen science projects have been used to fill these gaps (e.g. [14]). Eyewitness accounts and storm spotter reports also contribute to an understanding of severe weather hazards and phenomena [15].

1.1.2. Impacts

Impacts are the effects, outcomes, or consequences of hazardous events. During a severe weather event [16], identified impact information sources to include media coverage, fire stations upstream of a storm track that have already been affected, emergency calls, scout reports and ground-truthing, and emergency vehicle occupancy, to determine if enough capacities are on hand. Social media [17], crowdsourcing [18], and volunteered geographic information [19] are other near real-time sources of impact information. Near real-time impact data is usually collected by local emergency management (EM) agencies to produce situational reports which contain all available information about the developments and impacts of an event [16].

Post-event damage information is collected in the aftermath of an event in the form of damage surveys. Post-event data such as damage surveys allows forecasters to correlate damage levels produced from certain hazards, such as tornadoes, with the storm radar signature [17].

2

International Journal of Disaster Risk Reduction 66 (2021) 102619

Table 1

Summary of data sources and datasets for hazard, impact, vulnerability, and exposure data identified from the literature review.

Data Type	Description	Data Sources	Datasets
Hazard	Hydrometeorological observations and measurements to formulate warnings.	Meteorological services, hydrological services, Law enforcement, Storm spotters, Responders, social media (Twitter, Facebook, Seina Webo, Flickr, Instagram, SnapChat), the public, online databases, crowdsourcing and citizen science projects.	Meteorological and hydrological observations (e.g., rain gauges, river gauges), satellite and radar imagery, astellite and radar Hazards Assessment Network (NATHAN), European Severe Weather Database, US Storm Database, Distoric warning record, eyewitness interviews, public surveys, photos and videos.
Impact	Building situational awareness and informs situational assessments for response planning. After an event, impact/ damage assessments are conducted to link radar signatures of upcoming or unfolding hazards with a historical database and formulate impact- based warnings based on the historic impacts of similar events. Feeds into vulnerability functions for impact modelling.	Media reports, law enforcement, fire stations upstream already affected, the public, emergency responders, engineers, social media (Twitter, Facebook, Seina Webo, Filckr, Instagram, SmapChat), crowdsourcing applications, Google, hydrological services and flood managers, public health agencies, insurance industry, online databases, fresearch institutions, environmental protection agencies, storm spotters, emergency services, personal contacts, risk modelling.	and videos. Ground-truthing, in-situ observations, reports and emergency calls, situational reports, eye- witness interviews, photos/videos, commentary, incident reports, situational reports, hotline records, loss claims, Emergency Events Database (EM- DA), Natural Hazards Assessment Network (NATHAN), flood databases, digital mapping (e.g. Humanitarian OpenStreetMap, Google Analytics Records, technical reports, European Severe Weather Database, public surveys, eyewitness interviews, risk modelline
Exposure	Highly specific and small-scale nature, usually at the individual, activity, or community level, (e.g., Ferry routes and the locations of large trees overhanging power lines).	Government departments, land and resource management agencies, flood managers, mapping agencies/ organisations, crowdsourcing, OpenStreetMap,	outputs. Census data (for population counts and density), transportation routes and schedules, infrastructure databases/ databases/ datasets, land-use spatial layers, building/asset

(continued on next page)

S.E. Harrison et al.

Table 1 (continued)

Data Type	Description	Data Sources	Datasets	
		Google Maps, infrastructure industry, risk model.	footprints, road network spatial layers, Digital Elevation Mode time-varying population data river network spatial data, risl modelling outputs.	
Vulnerability	Conveys the vulnerability of people, livelihood, and property and typically includes information about infrastructure, buildings, land-use, census data, ecological data, and economic data; feeds into vulnerability models and fragility functions for impact modelling.	Research Institutions, public health agencies, engineers, risk specialists, insurance industry.	Vulnerability assessments, Pacific Risk Information System (PacRIS) vulnerability functions/ fragility curves, vulnerability indices, asset characteristics (g., building structure information),	

If this impact information is stored in a historic database, forecasters can refer to the database to compare past storm radar signatures with current radar signatures and understand the level of damage and impacts the current storm may produce, allowing the forecaster to formulate impact-based warnings [17].

Impact data is also produced from risk/impact models for preimpact, rapid-impact, and post-impact assessments. Data sources can include hazard, exposure and vulnerability information, geo-located social media data [20], satellite data [21], crowdsourced impact reports [18], and normalised damage functions [22].

Challenges still exist with collecting enough systematic, in-situ data (i.e., observed impacts) to build empirical models [23] and to validate models [8]. Furthermore, existing impact databases have been created with varying responsibilities for which agency collects information, and the methods and purposes of collection [24]. This limits the databases' use, particularly for analysis and verification. The classification of impacts appears to be subjectively done [25], proving it difficult to use datasets outside of the original purpose for which they were created.

1.1.3. Vulnerability

Vulnerability information is usually created through conducting vulnerability assessments [26]. This involves combining information about infrastructure, buildings, land-use, census data, ecological data [27], and socio-economic data [28]. These vulnerability assessments are typically presented as spatial maps [29].

Vulnerability assessments have traditionally focused on physical vulnerability, such as buildings and infrastructure, with less focus on social vulnerability [29]. Studies that have looked at social vulnerability tended to focus on physical impacts such as loss of life or physical injuries [30], and less so on nuanced social impacts resulting from less quantifiable social vulnerability factors such as "poor biophysical, social, and/or financial capital" in communities ([28]; p. 1482). This is because these qualting from human behaviour and demonstrated the highly dynamic nature of social vulnerability, such as risk governance, land use, and individuals' status, all of which change over space and time.

In New Zealand [31], developed social vulnerability indicators for flooding using national Census population data. These indicators are based on social vulnerability dimensions such as exposure, age, health

International Journal of Disaster Risk Reduction 66 (2021) 102619

and disability status, financial security, social connectedness, knowledge of natural hazards, housing conditions, food and water security, and decision-making and participation [31]. Health indicators relating to certain health conditions such as cardiovascular disease, respiratory disease, and mental health issues were not completed due to time constraints [31]. The results of their study provide an opportunity for decision-makers to consider additional factors beyond economic impacts when planning mitigative actions towards flooding [31].

Vulnerability information may be obtained through partnerships with the insurance industry who conducts vulnerability assessments for insurance schemes [32]. Additionally, the Pacific Catastrophe Risk Assessment and Financing Initiative (PCRAFI) created the Pacific Risk Information System (PacRIS), which houses historical hazard data, and risk profiles for Pacific Island countries [32].

1.1.4. Exposure

The highly specific and small-scale nature of exposure makes it difficult to systematically collect and incorporate into an IFW system with the current tools and data available [4,33]. An example of one form of general exposure information is population counts of people living in areas where hazards frequently occur (e.g. population data overlaid onto floodplains data) [27]. Exposure data is typically created by mapping the locations of assets, such as buildings and infrastructure in proximity to a hazard [29].

Ferry routes and the locations of large trees overhanging power lines are other examples of exposure data that would be important to consider during a high wind event [4]. Building footprints (i.e., a spatial layer of building polygons) is another example of exposure if the building footprints layer is overlaid with a hazard layer (e.g., [34]). However, even national- or global-scale exposure datasets can be expensive [29], and remain difficult to create and use due to a lack of detailed asset information [35].

1.2. Data characteristics for Early Warning Systems

Data usability in disaster response is determined by many factors. For the purposes of this study, we focused on two factors: reliability and timeliness. These appear to be two of the most important factors for choosing data sources for disaster response [36]. The requirements for reliable and timely data for disaster response can also apply to EWSs, as warnings must be timely and accurate to incite early and appropriate action [37,38].

Data is perceived as more reliable if it comes from a trusted source and/or it can be vetted [39]. As such, there is a preference for official sources, such as emergency call centre reports, intel from responders themselves, etc. [16]. For this study, official data refers to data created by recognised officials involved in local disaster response management practices where the disaster is occurring, such as police and fire services, engineers, helicopter pilots, local and regional councils, as well as meteorological and hydrological agencies, and science agencies. Unofficial data refers to data created by external parties of the disaster management practices, such as social media users, contributors to OpenStreetMap, private external corporations (e.g., Google Maps), and media agencies.

This section identified various sources of HIVE data. These components (hazard, impact, vulnerability, and exposure, or HIVE) form the conceptual basis of severe weather IFWs. Meteorologists do not typically possess knowledge of impacts, vulnerability, and exposure [40]. As such, sources of these data need to be identified for IFWs [7,10]. Furthermore, understanding the characteristics of the available data sources in terms of reliability and timeliness will assist data users in determining appropriate datasets for their purposes.

The objectives of this research are to identify sources for HIVE data and to understand the inhibitors and facilitators for collecting and using these data, to support the implementation of an IFW system in New Zealand for severe weather hazards. We chose to focus on the New

3
Zealand context to support the country's efforts for fulfilling both the WMO's objectives and Sendai Framework priorities for improved documenting of disaster risk and loss data; a need identified in previous research [10,41]. We employed Grounded Theory [42] to meet the research objectives, described next.

2. Research method

We used a qualitative approach to address the research question, specifically employing the Evolved-Straussian Grounded Theory research strategy (ES-GT) for data collection and analysis. Interviews and workshops were the primary data collection methods. From November 2018 to April 2021, the lead author interviewed thirty-nine (n = 39) experts in weather forecasting, warning, response, risk modelling, and data collection and management, as shown in Table 2. Three virtual workshops involved EM practitioners, weather forecasters, communication and data specialists, and hydrologists, from Auckland Region (n = 4) and Southland Region (n = 5). The third workshop involved a portion of the NZ risk and hazard science community based at GNS Science (n = 11). Thus, in total 59 people participated in this research.

Interview questions and workshop activities focused on IFW data needs and sources. We asked for participants' general thoughts on IFWs; what impact, vulnerability, and/or exposure data they use or need, why, and how; the life path of the data; experienced and/or perceived challenges obtaining data required for IFWs and other uses; and thoughts on collecting and using alternative data sources (e.g. social media and crowdsourcing). Herein we report on data sources, collection, and creation.

This research was conducted under a 'low risk' ethics notification with the Massey University Human Ethics Committee prior to data collection in 2018. All interviewees remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by their area of expertise and/or practice, industry, location, or governance level (Table 2). Interviews were audio-recorded and transcribed verbatim.

(Table 2). Interviews were audio-recorded and transcribed verbatim. Following ES-GT, we analysed the interview and workshop data using open coding and axial coding, supported by memo-writing and diagramming [43], in Nvivo 12 [44]. The coding paradigm introduced by [42] supported the axial coding stage whereby the codes created from open coding were related to the coding paradigm dimensions (Table 3) for increased density and precision.

3. Findings and discussion

This section will first present data sources for HIVE data as identified by the participants, followed by an investigation into the causal conditions, intervening conditions, and subsequent action/interaction strategies for collecting the HIVE data.

3.1. Data sources

The interviews and workshops identified several sources for hazard, impact, vulnerability, and exposure data. These data sources are described next. The following sections are organised by data type (hazard, impact, vulnerability, and exposure). Herein, each section will provide an overview, with accompanying summary tables, of the data sources and their characteristics related to IFWs. The accompanying tables present the data or dataset (e.g., weather stations, radar data), whether the data is official or unofficial, who collects or creates the data (e.g., the creator may be a member of the public by posting a report on social media, and the collector may be an EM agency who collects social media posts for situational awareness), the timescale of the data (e.g., "[Near] real-time" is data collected in real-time or near real-time, such as observational data, social media reports, etc.; "Current" is data that is static in time, was created prior to the event, but has been maintained and kept up-to-date; "Forecasted" is data created from forecasting

International Journal of Disaster Risk Reduction 66 (2021) 102619

models; and "Historic" is data created after an event and is not kept upto-date), the type of hazard and/or impact (e.g., meteorological, hydrological, hydrogeological, social, infrastructural, urban, rural, environmental, health, property, built environment), and uses within and outside of the Impact Forecasting and Warning Value Chain [7].

3.1.1. Hazard data

Hazard data refers to meteorological, hydrological, and hydrogeological data. Table 4 provides a summary of the hazard data sources identified by our participants. With regards to uses in the Warning Value Chain, each hazard data source listed in Table 4 was found to be used for Observation, Monitoring, and Detecting; Hazard Forecasting; Impact Forecasting; Impact Warning.

Our findings indicate that hazard data is quite systematically collected, documented, and used in New Zealand for many purposes by a select group of official agencies (see Table 4). For example, meteorological data, such as rainfall, windspeed, radar data, and forecast data are mostly produced, used, and housed by the Meteorological Service of New Zealand (herein referred to as the NZ MetService) and the National Institute of Water and Atmospheric Research (NIWA). The NZ MetService is NZ's appointed National Meteorological Service (NMS) [46], and NIWA is NZ's Crown Research Institute (CRI)⁶ for atmospheric and oceanic science [48]. NIWA maintains the national climate database with all the rainfall records and hydrological databases. Fire and Emergency New Zealand (FENZ) was also found to collect observational meteorological data for their own operations, which helps them plan responses to wildfires and other weather-related emergencies (EM. NZ. Nat. D.

The NZ MetService has made some of their observation and satellite data freely available [49]; Met. Private NZ. L) for commercial use by private weather forecasting companies, and for stakeholder agencies and researchers to conduct their own analyses, such as risk modelling. Following an impactful event on the Wellington South Coast where large swells and waves damaged houses and evacuations of five properties, arrangements have been made for the NZ MetService and their oceanographic branch (the MetOcean), and NIWA, to provides swell data, wave data, and wave forecast data to the regional EM office for a Significant Wave Warning programme (EM. NZ. Reg. L; [50]).

Findings from our interviews indicate that hydrological data is collected, created, and used primarily by local and regional councils in New Zealand. This aligns with the mandated role of local and regional councils under the Resource Management Act 1991, the National Civil Defence Emergency Management Plan 2015, and the Local Government Act 2012, wherein local and regional councils are assigned the responsibility of managing, monitoring, forecasting, and warning for flood hazards, with support from NIWA, the MetService, and EM Groups [51]. Hydrological data includes river height and flow gauges, river camera feeds, river network and watershed data (e.g., spatial shapefiles of rivers, floodplain footprints), overland flow paths, river flow and flood forecast models, etc.

The MetService also forecasts pollen counts for those with allergies. Pollen data is desirable for the public health sector (Health NZ. Reg. A) because pollen was found to play a significant factor in thunderstormrelated asthma attacks [52]. While the MetService provides qualitative pollen forecasts (e.g., "Pollen Levels: Moderate, Type is Plantain"), a public health official from the Waikato Region in NZ expressed a need for quantitative pollen data to model the risk of asthmatic attacks due to pollen and spring thunderstorms (Health NZ Reg. A; [7]). This need was echoed by a NZ-based pollen scientist for climate change risk assessments (see [53]).

Our interviews identified other creators of relevant hazard data alongside the MetService, NIWA, local/regional councils, and FENZ.

⁶ Crown Research Institutes (CRIs) are government owned companies that conduct scientific research in New Zealand [47].

International Journal of Disaster Risk Reduction 66 (2021) 102619

Table 2

Participant Codes. All participants remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by the area of expertise and/or practice, industry, location, or governance level.¹¹

Interview Code	Position	Classification	Location	Government Level
Agriculture/Rural NZ. A	Agriculture policy coordinator	Agriculture/Rural	NZ	National
Data Management Gov. NZ. Nat. A	Senior Resilience Advisor	Data Management	NZ	National
Data Management Private NZ. B	Geospatial Specialist	Data Management	NZ	
Data Management Research NZ. C	GIS Specialist	Data Management	NZ	
Data Management Private NZ. D	GIS Specialist	Data Management	NZ	
Data Management Gov. NZ. Nat. E	Head of Data	Data Management; Governance	NZ	National
EM. NZ. Reg. A	Director	Emergency Management	NZ	Regional
EM. NZ. Reg. B	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. C	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. D	Principal Science Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. E	Principal Advisor Strategy and Partnerships	Emergency Management	NZ	Regional
EM. NZ. Reg. F	GIS Lead	Emergency Management; Data Management	NZ	Regional
EM. Gov. NZ. Nat. G	Senior Hazard Risk Management Advisor	Emergency Management; Governance	NZ	National
EM. NZ. Reg. H	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Nat. I	First Responder	Emergency Management	NZ	National
EM. Gov. NZ. Nat. J	National Operations Manager	Emergency Management; Governance	NZ	National
EM. NZ. Reg. K	Regional Manager	Emergency Management	NZ	Regional
EM. NZ. Reg. L	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. M	Group Controller	Emergency Management	NZ	Regional
Health NZ. Reg. A	Respiratory Doctor	Public Health	NZ	Regional
Hyd. Gov. NZ. Reg. A	Flood EWS Programme manager	Hydrology; Governance	NZ	Regional
Lifelines NZ. Reg. A	Civil Engineer	Lifelines	NZ	Regional
Loss Modelling Research NZ. A	Economist	Loss Modelling; Research	NZ	
Met. Int. A	Science Manager	Meteorology	International	National
Met. Int. B	National Manager Disaster Mitigation Policy	Meteorology	International	National
Met. Int. C	Senior Policy Officer	Meteorology	International	National
Met. Int. D	Senior Social Scientist	Meteorology	International	National
Met. Int. E	Consultant Meteorologist	Meteorology	International	National
Met. NZ. F	Senior Meteorologist	Meteorology	NZ	National
Met. NZ. G	Communications	Meteorology	NZ	National
Met. NZ. H	Public Relations	Meteorology	NZ	National
Met. Int. I	Division Chief/Meteorologist	Meteorology	International	National
Met. Research NZ. J	Meteorologist	Meteorology; Research	NZ	National
Met. NZ. K	Senior Meteorologist	Meteorology	NZ	National
Met. Private NZ. L	Head Weather Analyst	Meteorology	NZ	
Risk Modelling NZ. A	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. B	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. C	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ. D	Risk Modeller	Risk Modelling	NZ	National

Table 3

Summary of the coding paradigm dimensions that supported the axial coding analysis in this study.

Coding Paradigm Dimension	Description
Causal Conditions	A set of events that influence the phenomena or result in the appearance or development of a phenomenon [42,45].
Phenomena	The subject or object under study [42].
Contextual Conditions	The specific set of conditions and characteristics surrounding the phenomena and resulting in action/ interaction strategies taken to address the phenomena [42, 45].
Intervening Conditions	Unexpected events or factors leading to action/interaction strategies (e.g., time, space, culture, socioeconomic status, technological status, history) [42,45].
Action/Interaction Strategies	Purposeful and deliberate acts taken to address the phenomena [42,45].
Consequences	Predicable or unpredictable, intended or unintended outcomes of the action/interaction strategies [42,45].

Land Information New Zealand (LINZ), the public service department charged with handling geographic information in New Zealand, collects sea level data from float gauges (as do port companies), which is useful for coastal flood hazards (Data Management Gov. NZ. Nat. A); and holds slope data, useful for landslide hazard management and mitigation (Data Management Gov. NZ. Nat. A). LINZ and the Ministry for Environment (MfE) also hold river network data.

Unofficial data sources were also identified by participants for

monitoring hydrometeorological hazards and building situational awareness. For example, the MetService, Civil Defence and Emergency Management (CDEM) Groups, and Local/Regional Councils monitor social media for reports or observations of hydrometeorological hazards (Met. Int. E; EM. NZ. Reg. B, C). From our interviews, we found that none of our participating agencies collect or store social media data for future analysis, citing resource limitations as a key barrier (Met. NZ. F, G, H; EM. NZ. Reg. A, E). Rather, the social media platforms are monitored onscreen for staff to pick up posts or events of interest for further investigation (Met. Int. E; EM. NZ. Reg. B, C). Additionally, one CDEM Group described how they actively request social media users to verify impacts if they can safely do so (EM. NZ. Reg. A). Notable posts may be captured and included in situational reports, but the data usually does not have a use beyond that (EM. NZ. Reg. A, C).

Crowdsourced data was found to be collected through specially designed applications and platforms. In NZ, the West Coast CDEM group crowdsourced hazard and impact reports using the Esri Story Maps platform (EM. NZ. Reg. C, F). Similarly, the NZ Flood Pics Esri Story Map⁷ was set up by a volunteer to collect flood and impact photos during flood events in Auckland and has since been expanded to the rest of NZ (Data Management Private NZ. D). In the USA, the meteorological service collects precipitation data and associated impacts through the mobile application mPing8 (Met. Int. I). Similarly, in Austria and

⁷ http://www.nzfloodpics.co.nz/.
 ⁸ https://mping.nssl.noaa.gov/.

Appendix L: Chapter Seven Publication

S.E. Harrison et al.

Table 4

International Journal of Disaster Risk Reduction 66 (2021) 102619

Summary of hazard data sources in New Zealand, identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, type of hazard, and non-IFW uses.²¹

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	Type of Hazard	Non-IFW Uses
Weather stations (rain gauges, anemometers, etc.)	Official and unofficial	MetService, FENZ, NIWA, councils, volunteers	[Near] real- time	Meteorological	Research
Radar data	Official	MetService, NIWA, Private	[Near] real- time	Meteorological	Research
Satellite imagery & observations	Official	MetService, NIWA, LINZ	[Near] real- time	Meteorological	Research
River height and flow gauges	Official	Regional councils, NIWA	[Near] real- time	Hydrological (e.g. flood)	Research
Float gauges/Sea level data	Official	LINZ, port companies, regional councils, NIWA	[Near] real- time	Hydrological (e.g. flood)	Research
River networks	Official	NIWA, LINZ, Ministry for the Environment, Councils	[Near] real- time	Hydrological (e.g. flood)	Research
Vertical Rain Radar	Official	MetService, Private, Healthy Waters Auckland Council	[Near] real- time	Meteorological	Research
Regionwide floodplains footprints/ shapefiles	Official	Councils	[Near] real- time	Hydrological (e.g. flood)	Research
Overland flow paths	Official	Councils	Current	Hydrological (e.g. flood)	Research
Coastal inundation Maps	Official	Councils	Forecasted	Hydrological (e.g. flood)	Research
Pollen Counts	Official	MetService	Forecasted	Health	Research; Response: Preparedness
Slope	Official	LINZ, Councils	Current	Hydrogeological (e.g. landslide)	Research
Camera feeds	Official	Councils, NZTA, Ski fields, Police	Real-time	Meteorological, Hydrological, Hydrogeological (e.g. landslide)	Response; Preparedness; Public Awareness
Social media (Tweets, Facebook post comments)	Unofficial	Creators: public/social media users Collectors: MetService, EM, Researchers, Councils, NZGIS4EM	[Near] real- time	Meteorological, Hydrological, Hydrogeological	Research; Response; Situational Awareness
Crowdsourcing (e.g., volunteer rain gauges, NZ Flood Pics, mPing, WeatheX, European Weather Observer)	Unofficial	Creators: public/social media users Collectors: EM, Researchers, Councils, NZGIS4EM, Volunteers (e.g., NZ Flood Pics)	[Near] real- time	Meteorological, Hydrological, Hydrogeological	Research; Response; Situational Awareness
Forecast data	Official	MetService, MetOcean, NIWA, Councils	Forecasted	Meteorological, Hydrological, Marine/ Coastal	Research; Response: Preparedness
National Climate Database	Official	NIWA	Historical	Meteorological, Hydrological	Research

Australia, the European Weather Observer⁹ (Met. Int. E) and WeatheX¹⁰ (Met. Int. B, C) applications respectively are used to collect severe weather reports. These applications gather hazard (and some impact) reports from the public, who have not received any training. The Story Map data usually contains photos with an optional description for the CDEM Group to build their situational awareness (EM. NZ. Reg. F). Alternatively, the specialised applications used in the USA, Austrai, and Australia collect more structured data using reporting forms (Met. Int. B, C, E, I). The benefits of the crowdsourced data are that it is timely (Met. Int. E), it may fill in sensor gaps (Met. Int. I), and the applications may increase public awareness and engagement in severe weather hazards (Met. Int. B, C, D. However, maintaining engagement is a challenge and agencies may struggle with receiving reports once contributors have lost interest (Met. Int. E). A gamification element is being considered to maintain interest for contributors of the European Weather Observer (Met. Int. E). Other challenges include quality control and getting buy-in from scientists on the validity of crowdsourced data (Met. Int. E, I).

Meteorological services in the USA and Austria regularly collect reports from storm spotters. The storm spotters can receive training from the meteorological services to increase their credibility. In Austria, the meteorological services offer training and certificates with identification

¹ The acronyms and abbreviations in Table 2 are as follow: Emergency Management (EM), New Zealand (NZ), Geographic Information Systems (GIS), Regional (Reg), Government (Gov.), Early Warning System (EWS), International (Int.).

⁹ https://www.essl.org/cms/european-severe-weather-database/ewob/.
 ¹⁰ https://weathex.app/.

numbers for storm spotters to then register and receive a quality control rating for their reports based on their level of training (Met. Int. E). The meteorological service can then discern the trustworthiness of incoming spotter reports (Met. Int. E). This data is considered highly credible among scientists (Met. Int. E). Furthermore, engagement and awareness are quite high with the "weather enthusiasts" (Met. Int. E). However, connecting with spotters for training remains a challenge (Met. Int. E). The data is also limited in its application for warnings because it is not real-time, thus it cannot be used for real-time warning verification (Met. Int. E). Lastly, storm spotter reports may overlook meteorological phenomena that produce small impacts, a need identified by the US National Weather Service (Met. Int. I).

As reported in a related study [7], much of these datasets are used for steps in the Value Chain for warnings, particularly hydrometeorological Observation, Monitoring, and Detection, Hazard Forecasting, and in some cases Impact Forecasting and Impact Warning. For example, the MetService uses radar and rainfall data to observe, monitor, and detect rainfall amounts, which is then used by regional and local councils to forecast flood hazards (Met. NZ. G). Flood forecasts may then be used to forecast the impacts of the flood, in a quantitative model-based approach, or in a qualitative discussion-based approach, or both [7]. Local/regional councils and CDEM Groups may then use these forecasts to inform their flood warning messages, wherein they might include impact-oriented messages (EM. NZ. Reg. H; [40]).

3.1.2. Impact data

6

A plethora of impact data was identified in our interviews that are created, collected, and used by different groups for different purposes, as

International Journal of Disaster Risk Reduction 66 (2021) 102619

 Table 5

 Summary of impact data sources. identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, uses in the IFW Value Chain, type of impact, and non-IFW uses.³¹

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
Tweets	Unofficial	Public/social media users	MetService, CDEM, Researchers, Councils, NZGIS4EM	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Facebook Post comments	Unofficial	Public/social media users	MetService, CDEM, Researchers, Councils, NZGIS4EM	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
SnapChat Heat Maps	Unofficial	Public/social media users	MetService, CDEM, Researchers, NZGIS4EM	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Crowdsourced Photos via Story Maps (e. g., NZ Flood Pics)	Unofficial	Public/social media users	MetService, CDEM, Researchers, Councils, NZGIS4EM, Volunteers (e.g., NZ Flood Pics)	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Crowdsourcing/ Public Reporting	Unofficial	Public	Councils, CDEM, Volunteers	Current, maybe near real-time	Observation, Monitoring, and Detecting; Hazard Forecasting; Impact Forecasting; Impact Warning	Built environment, infrastructure	Response; Situational Awareness; Recovery; Planning; Mitigation; Public Awareness/ Engagement/ Education
Red Cross Chained Crowdsourcing	Unofficial	Public	Red Cross	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Response; Situational Awareness
Emergency call centre reports	Official	Public	Police, FENZ	[Near] real- time & Historic?	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Community volunteer radio calls	Unofficial	Designated community volunteers	EM, Councils	[Near] real- time	Observation, Monitoring, and Detecting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Response; Situational Awareness
Damage surveys (e.g. aerial surveys, building surveys via Survey123, QuickCapture, RiACT, Kobo 2, etc.)	Official	Councils, Researchers (NIWA, GNS Science, Universities), CDEM	Councils, Researchers (NIWA, GNS Science, Universities), CDEM	Historic	Impact Forecasting; Impact Warning	Urban, Rural, Infrastructural	Research; Response; Situational Awareness; Recovery; Mitigation; Land use planning and development policies
Media reports	Unofficial	Media outlets	Councils, Researchers (NIWA, GNS Science), CDEM, MetService	[Near] real- time	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Research; Response; Situational Awareness
Tacit knowledge, experience, intuition	Official	Councils, CDEM, MetService	Councils, CDEM, MetService	Available in real-time, based on historic knowledge and experience	Impact Forecasting; Impact Warning	Urban, Rural, Environmental	Response; Situational Awareness
Health & Injury Data	Official	District Health Boards, ACC, Stats NZ	District Health Boards, ACC, Stats NZ	Historic	Impact Forecasting; Impact Warning	Health	Research; Response; Recovery; Planning and Mitigation
Wellbeing Surveys	Official	CDEM, Councils	EM, Councils	Historic		Social, Health, Property	Response; Situational

(continued on next page)

International Journal of Disaster Risk Reduction 66 (2021) 102619

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
					Impact Forecasting;		Awareness; Recovery
Post-event interviews and surveys	Official	Researchers (e.g. NIWA, GNS	Researchers (e.g. NIWA, GNS	Historic	Impact Warning Impact Forecasting;	Social	Response; Mitigation;
Insurance claims	Official	Science, other), EM Insurance companies, EOC.	Science, other), EM Insurance companies, EOC,	Historic	Impact Warning Impact Forecasting:	Property, Health, Economic	Preparation Research; Recovery:
Boots on the	Official	ICNZ Councils, CDEM	ICNZ Councils, CDEM	[Near] real-	Impact Warning Impact	Urban, Rural,	Mitigation Response;
ground" ifelines Sectors (e.g.	Official	Lifelines Services (e	Lifelines Services	time	Forecasting; Impact Warning Impact	Environmental	Situational Awareness Response:
power companies, NZTA)	oncia	g., NZTA, Transpower, KiwiRail), councils, wastewater services	(e.g., NZTA, Transpower, KiwiRail), councils, wastewater services	time	Forecasting; Impact Warning		Situational Awareness; Recovery; Busines As Usual; Researc
Council Requests For Service (RFS)	Official	Public	Councils	[Near] real- time	Impact Forecasting; Impact Warning	Infrastructural; Property; Urban; Rural; Environmental	Situational Awareness; Response;
Situational Reports	Official	CDEM, Councils	CDEM, Councils	Current	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Built Environment	Recovery Response; Situational Awareness; Recovery; Researd
Post-event reports	Official	NEMA, CDEM Groups, Councils	NEMA, CDEM Groups, Councils	Historic	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Built Environment	Recovery; Mitigation and Planning; Researc
Operations Reports (Council)	Official	Councils	Councils	Historic	Impact Forecasting; Impact Warning	Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Built Environment	Response; Situational Awareness; Recovery; Researe
lood event reporting	Official	Councils, Healthy Waters Auckland Council/Councils	Councils, Healthy Waters Auckland Council/Councils	Historic	Observation, Monitoring, and Detecting; Impact Warning	Infrastructural, Urban, Rural Environmental	Response; Situational Awareness; Research; Mitigation; Land use planning and development policies
njuries and fatalities (e.g., cause of death)	Official	ACC, Coronial Services of New Zealand, Stats NZ	ACC, Coronial Services of New Zealand, Stats NZ	Historic	Impact Forecasting; Impact Warning	Health	Response; Situational Awareness; Research;
Cultural and heritage/historical	Official	Councils	Councils	Historic	Impact Forecasting;	Social, Cultural	Research; Planning;
impacts impact Model outputs	Official	GNS Science, NIWA, Researchers	GNS Science, NIWA, Researchers	Historic, Rapid	Hazard Forecasting; Impact Forecasting	Social, infrastructure, built environment, hydrological, hydrogeological, urban, rural, Property, Economic	Mitgation Response; Research; Planning; Mitigation
iEMA National Loss Database	Official	CDEM Groups	NEMA	Historic	Impact Forecasting; Impact Warning	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	Sendai Framewor Reporting
IIWA NZ Historic Events Catalogue	Unofficial	Media outlets	NIWA	Historic	Impact Forecasting; Impact Warning	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	Research; Mitigation
Local Council Databases	Official	Councils	Councils	Historic	Impact Forecasting; Impact Warning	Environment Hydrological, Hydrogeological, Property, Urban, Rural,	Response; Situational Awareness;

International Journal of Disaster Risk Reduction 66 (2021) 102619

Data	Official or Unofficial	Data Creators	Data Collectors	Timescale	IFW Value Chain Uses	Type of Impact	Non-IFW Uses
Storm Data (USA)	Official	Meteorological services, Responders, Emergency Services, Storm Spotters, media outlets	National Centers for Environmental Information	Historic	Impact Forecasting; Impact Warning	Infrastructural, Built Environment Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	Recovery; Mitigation Research; Mitigation
European Severe Weather Database	Official	Meteorological services, Responders, Emergency Services, Storm Spotters, media outlets	European Severe Storms Laboratory	Historic	Impact Forecasting; Impact Warning	Meteorological, Hydrological, Infrastructural, Urban, Rural, Environmental, Health, Property, Economic, Built Environment	Research; Mitigation

shown in Table 5. Impact data is available in a range of official and unofficial capacities.

Impact data is collected, created, and used in an official capacity by CDEM Groups, local/regional councils, FENZ, the lifelines sector (e.g., transportation, power, water/wastewater/stormwater services), the insurance sector (e.g., the Earthquake Commission (EQC), the Accident Compensation Corporation (ACC), the Insurance Council of NZ), and the research sector (e.g., GNS Science, NIWA). The primary purpose of collecting the impact data is not for severe weather forecasts and warnings, but for response, recovery, mitigation, and planning. Depending on the quality of the data, it could be used for additional purposes such as impact forecasting and warning. For example, emergency call centre (i.e., 111 calls in NZ) reports incite responses from the appropriate authorities (e.g., FENZ, police). Afterwards, weather-related emergency reports can be used for post-event analysis to verify and improve warnings (Met. NZ, G; [54]).

Insurance claims from property damage and injuries are used for recovery and produce rich datasets for future analysis such as risk/loss modelling (Loss Modelling Research NZ. A), which may inform impact forecasts. Likewise, impact assessments conducted by CDEM Groups are primarily used for building situational awareness to inform response efforts (EM. NZ. Reg. H [55]); and can also be used for warning verification and warning updates in real-time (EM. NZ. Reg. H).

Impact assessments for CDEM Groups in NZ have evolved with the advancements of GIS-based technology for improved data capture and

for Emergency Management (INZGIS4EM), MetUcean Solutions (MetOcean). ³ The acronyms and abbreviations in Table 5 are as follows: Fire and Emergency New Zealand (FENZ), National Institute of Water and Atmospheric Research Ltd (NIWA), New Zealand Meteorological Service (MetService), New Zealand Transport Agency (NZTA), Emergency Management (EM), New Zealand Geographic Information Systems for Emergency Management (NZGI-S4EM), Insurance Council of New Zealand (ICNZ), Accident Compensation Corporation (ACC), Institute of Geological and Nuclear Sciences Limited (GNS Science), Statistics New Zealand (Stats NZ), Earthquake Commission (EQC), National Emergency Management Agency (NEMA) Real-time Individual Asset Attribute Collection Tool (RIACT). management (EM. NZ. Reg. C, H, F; Data Management Private NZ. B). An overview of how GIS tools were used for a flood event in NZ's West Coast region was provided by [56]. Officials used rapid data collection tools such as Survey123¹¹ and QuickCapture¹² to collect impact information. QuickCapture was primarily used for collecting photos from aerial and ground assessments, while Survey123 was used to complete form-based assessments [56]. The value of these tools lies in the seamless integration process of the field data directly into a GIS layer for real-time viewing in an Emergency Operations Centre (ECC) (EM. NZ. Reg. C, H, F; Data Management Private NZ. B). However, the pitfalls of these collection methods are that information overload can occur with applications like QuickCapture if the trained staff take a large volume of photographs, and gaps in training can lead to errors in the data, which may skew datasets (EM. NZ. Reg. F).

Tacit knowledge and experience refer to knowledge held by official staff from past experiences who apply this knowledge when planning for or responding to events. NZ participants described how senior EM staff know from past events that certain river levels and peaks will lead to respective levels or types of impacts (EM. NZ. Reg. A, B). This knowledge is passed on verbally either just before or during an event to help with response planning and coordination (EM. NZ. Reg. A, B). The same was found to be true for the Lifelines sector, who "have a good amount of data but ... more importantly ..., their staff have a very good historical knowledge" (Lifelines NZ. Reg. A). This information is highly trusted as it is based on years of experience. It allows agencies to learn from past events and better prepare for current or future events (EM. NZ. Reg. B).

In many cases, this tacit knowledge and experience is not formally documented and there is a risk of losing this knowledge when staff move on (EM. NZ. Reg. A, C; Lifelines NZ. Reg. A). As one participant identified: "[hazard and impact forecasting are] primarily based on history and knowledge. And with that comes the risk ... if you have key people out of the equation or people move on ... in life, then you have the knowledge gap until such time as that's filled" (EM. NZ. Reg. C). Upon identifying this vulnerability, this participant indicated that their agency has begun some historical cataloguing (EM. NZ. Reg. C).

Unofficial impact data/information also helps EMs "build a picture of

c

² The acronyms and abbreviations in Table 4 are as follows: Fire and Emergency New Zealand (FENZ), National Institute of Water and Atmospheric Research Ltd (NIWA), Land Information New Zealand (LINZ), New Zealand Meteorological Service (MetService), New Zealand Transport Agency (NZTA), Emergency Management (EM), New Zealand Geographic Information Systems for Emergency Management (NZGIS4EM), MetOcean Solutions (MetOcean).

¹¹ Survey123 is a location-based application developed by Esri [57] that is used for completing assessment forms such as impact assessment forms, building assessment forms, and welfare needs assessment forms (EM. NZ. Reg. F).

F). ¹² QuickCapture is another location-based field observation application developed by Esri [58] that is particularly useful for taking photographs of damage/impacts and uploading in real-time to be viewed on a dashboard in the Emergency Operations Centre (EM. NZ. Reg. F).

what's happening" (EM. NZ. Reg. C). Unofficial data includes social media posts, crowdsourced and citizen science data, media reports, and community volunteer radio calls. Interviewees were found to obtain information from Facebook, Twitter, and SnapChat. In most cases, the interviewees indicated that they do not collect or store social media data, but only monitor it for situational awareness and real-time verification. The value of social media appears to lie in its ability to indicate potential hot spots early on in an event and building situational awareness (Met. Int. E; EM. NZ. Reg. C). Social media sentiment analysis can also help to understand the cultural impacts of severe weather and communicate the warnings better (Met. Int. D). However, interviewees use social media data with caution, indicating that while social media data is timely and quantitatively rich, it is difficult to verify, verify (EM. NZ. Reg. B). Relying solely on social media risks missing or overlooking impacted people who are not on social media, such as areas without power or internet access (EM. NZ. Reg. B). Thus, social media data should be used complementarily with other data (Met. Int. D). Impact databases were also found to exist or be under development.

Impact databases were also found to exist or be under development. In NZ, the National Emergency Management Agency (NEMA) is developing a national loss database in fulfilment of the Sendai Framework priorities [1]. Data is pulled from EM impact assessments, situational reports, and post-event reports (EM. Gov. NZ. Nat. G). NIWA hosts a catalogue of historic severe weather events in NZ using media reports (Met. Research NZ. J), primarily for research purposes. Storm Data in the USA and the European Severe Weather Database contain data sourced from storm spotters and impact assessments from storms in the USA and Europe (Met. Int. E, I). These American and European databases are highly credible as only vetted and trusted information goes in (Met. Int. E, I). Unfortunately, the databases are not updated in real-time due to the rigorous quality control measures and may have gaps for small impact events (Met. Int. E, I). International Journal of Disaster Risk Reduction 66 (2021) 102619

impacts such as damages to the built and natural environment and do not capture social human impacts or indirect impacts. This is a major gap identified in a previous study [7]. Wellbeing surveys and post-event surveys from research studies capture indirect impacts on people's health and social/cultural wellbeing (EM. NZ. Reg. F, H; Risk Modelling NZ. B, C). Wellbeing surveys (also referred to as Needs Assessments and Welfare surveys) are conducted by CDEM Groups and/or health agencies to understand the needs and impacts of people affected by an emergency [59,60]. Furthermore, District Health Boards, ACC, and Stats NZ collect and house health, injury, and mortality data for their own purposes (EM. Gov. NZ. Nat. G; Loss Modelling Research NZ. A). In the context of severe weather IFWs, these data, along with the Wellbeing surveys conducted by EM/councils, could be useful for understanding indirect and social and cultural impacts to inform impact warning messages (EM. Gov. NZ Nat. G; Loss Modelling Research NZ. A; [7]).

3.1.3. Vulnerability data

Our participants identified some vulnerability data sources, as shown in Table 6. As stated in our literature review, vulnerability data is more difficult to obtain or create, as vulnerability changes over space and time. On their own, many of the identified data sources in Table 6 do not provide an indication of vulnerability. Rather, these data sources are inputs into vulnerability assessments and risk models. For example, census data and health data may be used to conduct a social vulnerability assessment, while building damage assessments are inputs into vulnerability functions for risk model then help analysts identify vulnerabile areas:

Everything's built off the models \dots we've identified \dots issues with care homes and areas, hotspots in the community which could potentially \dots be at risk \dots We've got them in \dots behind the stop

Table 6

Summary of vulnerability data sources. identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, uses in the IFW Value Chain, type of impact, and non-IFW uses.⁴¹

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	IFW Uses (from the Value Chain)	Type of Impact	Non-IFW Uses
Vulnerability Assessment and Risk Modelling Outputs	Official	Councils, Researchers (NIWA, GNS Science), CDEM, Hired contractors	Historic	Impact Forecasting; Impact Warning	Property, human, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Planning; Mitigation; Research
Asset Information	Official	Councils, Researchers (NIWA, GNS Science), CDEM, Hired contractors	Historic	Hazard Forecasting; Impact Forecasting; Impact Warning	Property, human, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Planning; Mitigation; Research
Building damage assessments	Official	Councils, Researchers (NIWA, GNS Science), CDEM, Hired contractors	Historic	Hazard Forecasting; Impact Forecasting; Impact Warning	Property	Response; Situational Awareness; Recovery; Research; Mitigation
Census data	Official	Stats NZ	Historic	Hazard Forecasting; Impact Forecasting	Human	Research; Business As Usual (BAU)
Tacit knowledge, experience, intuition	Official	Councils, CDEM	Available in real- time, based on historic knowledge and experience	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Response; Situational Awareness; Preparedness; Mitigation; BAU
Lab-based experiments	Official	Researchers (NIWA, GNS Science, universities)	Historic	Hazard Forecasting; Impact Forecasting	Built environment	Research
Health Data, New Zealand Health Survey	Official	MOH, DHBs	Historic	Impact Forecasting. Impact Warning	Health	Research; Recovery
Infrastructure	Official	Councils, Hired contractors	Historic	Hazard Forecasting; Impact Forecasting; Impact Warning	Built environment, infrastructure	Response; Situational Awareness; Recovery; Planning; Mitigation
Soil/land stability	Official	GNS Science	Historic?	Observation, Monitoring, and Detecting; Hazard Forecasting; Impact	Hydrogeological; Built environment, infrastructure, Environment, Property	Planning; Mitigation; Research

banks, bungalows, probably not the best place for them but they're there. So just being aware of those [vulnerable areas] (Hyd. Gov. NZ Reg. A).

Asset information refers to characteristics of the asset for which the vulnerability is being assessed. These can be people, buildings, roads, stop banks, the environment, etc. The asset information of buildings would include data on the age of the building, the construction material, and the number of floors (Risk Modelling NZ. B). This information is collected through building assessments by building engineers (Risk Modelling NZ. B). Alternatively, asset information of people consists of demographic data, health data, socioeconomic data, etc. (Risk Modelling NZ; [28]), available from Stats NZ and the Ministry of Health. Asset information about stop banks includes age and condition and is held by local and regional councils (Hyd. Gov. NZ. Reg. A).

Physical vulnerability of the built environment is captured through building damage assessments, lab-based experiments, infrastructure vulnerability assessments, and the tacit knowledge, experience, and intuition of engineers (Risk Modelling NZ. A). Assessments of building damage and lab-based experiments are typically conducted by engineers hired within or contracted by agencies like local/regional councils, GNS Science, NIWA, and CDEM Groups (Risk Modelling NZ. B). Infrastructure vulnerability assessments are conducted or contracted by the Lifelines Sector (e.g. [621]), and other researchers. While much work has been made in the risk modelling space for the built environment, a gap remains in NZ around the asset information of buildings: "We don't have good building data. We've got [data on] where buildings are, but we don't necessarily know what they're made of and that kind of stuff" (Risk Modelling NZ. A).

Social or human vulnerability is even more difficult to capture and assess and there is a need to capture the dynamic nature of social human vulnerability for IFWs [7]. This need is also echoed for response efforts: "[emergency response agencies are] more focused on the emergency response, so they need to know [who needs] more help, so it's better to break down all the gender ... and age difference (sic) ... " (Risk Modelling NZ. B). The census is a clear choice for obtaining social/human vulnerability data. However, our participants identified surprising challenges with obtaining and using census data, such as in the example below:

We have vulnerability data. The problem with that data is that ... in Argentina we make a national census ... every ten years ... So, the vulnerability information that we have now is old! We have other information but, you know, this strong, national census gives you a lot of information about vulnerability. So that's a problem (Met. Int. D).

In New Zealand, low response rates of the 2018 national census created problems with the dataset [63], which significantly delayed the release of the data and frustrated risk modellers who were waiting on the data to use in their models (Risk Modelling NZ. A).

An additional challenge with using census data identified by our participants is the spatial scale. The NZ census data is at the meshblock 13 scale, but risk modellers want to model at the building scale. Participating risk modellers questioned how they can interpolate the census data with the building level (Risk Modelling NZ. A, B).

Health data is another indicator of human vulnerability. The impacts of severe weather events can exacerbate underlying health conditions.

¹³ According to [64]; "[a] meshblock is defined by a geographic area, which can vary in size from part of a city block to a large area of rural land."

International Journal of Disaster Risk Reduction 66 (2021) 102619

For example, people with asthma may experience exacerbated symptoms while cleaning up damaged and contaminated sites after a flood (Risk Modelling NZ. C), or during thunderstorms (Health NZ. Reg. A; [7]). In NZ, District Health Boards (Health NZ. Reg. A) and the Ministry of Health house this data. The Ministry of Health conducts an annual national Health Survey of NZ.

Like impact data, vulnerability data is also present as tacit knowledge and experience. As previously mentioned, infrastructure engineers and other asset managers possess a wealth of knowledge and experience around the performance capacities of a building, levy, or another piece of infrastructure. In the risk modelling space, this tacit knowledge, or expert opinion is a valid resource for building vulnerability functions:

If you are an asset manager and you know about your port or your wharf ... and you just intuitively think about ... if a six-meter wave came in what do you think would happen? ... you can elicit information that way. And there's a lot of vulnerability functions out there that are done that way because you don't need an event to happen. And I guess there is a certain level of knowledge and intuition that can go into these things ... So you can do some kind of estimate (Risk Modelling NZ. A).

CDEM Groups possess tacit knowledge around vulnerable areas, people, and communities within their jurisdiction (Hyd. Gov. NZ. Reg. A; Auckland Workshop), but our participants indicated that they still need to know "who is where and what their mobility/health/access considerations are" (Auckland Workshop). Some of this information is available through pre-existing networks with EM services that manage relationships with vulnerable communities, such as the "Caring for Communities" government work programme that was established in response to COVID-19,¹⁴ and through Welfare coordination groups (Auckland Workshop). However, there is a risk of missing people who are not "in the system" (Auckland Workshop). Furthermore, privacy concerns exist when considering alternative uses for the data beyond its initial purpose, which is usually to support response and recovery (Auckland Workshop; EM. NZ. Reg. H; EM. Gov. NZ. Nat. G). Yet, even this tacit knowledge can still help EM groups formulate their warning messages (EM. NZ. Reg. H).

All datasets listed in Table 6 are produced and housed in an official capacity. Our interviews did not find any instances where crowdsourcing or other unofficial methods were used for producing vulnerability data. Some research has been done to use crowdsourcing applications for capturing vulnerability. For example, in Kazakhstan [65] developed a mobile application (OxyAlert) where users can answer questions about their health status relative to their geographic location and atmospheric conditions to characterise individual vulnerability to meteorological changes [66], presented a new participatory mapping methodology for incorporating stakeholder's participation, local knowledge, and locally spatial characteristics for vulnerability assessments of flood risk. These studies demonstrate the untapped opportunity to use crowdsourcing and other engagement activities to produce localised or individualised vulnerability assessments.

3.1.4. Exposure data

The dynamic nature of exposure data also makes it difficult to collect and maintain. However, our participants identified some existing and potential sources of exposure data for different uses and with different time scales, summarised in Table 7. Most of the data sources are classified as official, as they are produced by official agencies.

Asset footprints refer to the geographical location of assets, such as buildings, historically and culturally significant sites, etc. These data are

⁴ The acronyms and abbreviations in Table 6 are as follows: National Institute of Water and Atmospheric Research Ltd (NIWA), Emergency Management (EM), Institute of Geological and Nuclear Sciences Limited (GNS Science), Statistics New Zealand (Stats NZ), Ministry of Health (MOH), District Health Boards (DHBs).

¹⁴ https://ipanz.org.nz/Article?Action=View&Article_id=150258.

Appendix L: Chapter Seven Publication

S.E. Harrison et al.

International Journal of Disaster Risk Reduction 66 (2021) 102619

Table 7

Summary of exposure data sources identified in interviews. This table presents the data set, whether it is official or unofficial, the data creators and collectors, timescale, uses in the IFW Value Chain, type of impact, and non-IFW uses.⁵¹

Data	Official or Unofficial	Data Creators and/or Collectors	Timescale	IFW Uses (from the Value Chain)	Type of Impact	Non-IFW Uses
Asset footprints (e.g., building, cultural and historical site locations)	Official and unofficial	Councils, Google Maps, OpenStreetMap, LINZ, FENZ, Heritage NZ, Iwi	Historic	Hazard Forecasting; Impact Forecasting	Property, Cultural, Social	Response; Research; BAU; Mitigation
Infrastructure networks, e.g., transportation networks, traffic flows, power & water supplies	Official	Lifelines Services (e.g., NZTA, Transpower, KiwiRail), local & regional councils	Current	Hazard Forecasting; Impact Forecasting	Infrastructural	Response; Situational Awareness; BAU
Census data	Official	Stats NZ	Historic	Hazard Forecasting; Impact Forecasting	Human	Research; BAU
Population movement via cell phone data	Official	DataVentures/Stats NZ	Historic and [Near] real-time?	Hazard Forecasting; Impact Forecasting; Impact Warning	Human	Research
Tacit knowledge, experience, intuition	Official	Councils, CDEM	Available in real- time, based on historic knowledge and experience	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Social, Infrastructural, Urban, Rural, Environmental, Property, Economic, Built Environment	Response; Situational Awareness; Preparedness; Mitigation; BAU
Topographical data, e.g., digital elevation models	Official	LINZ, Landcare Research, Universities, GNS Science, NIWA	Historic	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Rural, Urban, Environmental	Response; Situational Awareness; Research; BAU
Land-use	Official	Councils, LINZ	Historic	Hazard Forecasting; Impact Forecasting	Meteorological, Hydrological, Hydrogeological, Rural, Urban, Environmental	Response; Situational Awareness; Research; BAU
Community events	Official	MBIE, Stats NZ, Local and regional cultural and tourism agencies (e.g., Auckland Unlimited)	[Near] real-time	Hazard Forecasting; Impact Forecasting; Impact Warning	Cultural, Social	Response; Situational Awareness; BAU

officially produced by local/regional councils, LINZ, and FENZ. They are also available on unofficial platforms like Google Maps and Open-StreetMap.¹⁵ This location-based data helps EMs and researchers identify assets that are potentially exposed to a hazard.

Exposure is typically determined by overlaying the asset information with hazard information (Risk Modelling NZ. C). This is common practice for disaster and climate change mitigation. For example [34], created a national-scale built environment exposure model to extreme sea-level rise for NZ by overlaying buildings, infrastructure, and built land area with a Digital Elevation Model and coastal flood maps. Similarly [67], compared urban and rural exposure to coastal hazards using demographic data overlaid with building, infrastructure, and land assets. Their results provided counts of people, buildings, infrastructure, and land assets located in areas exposed to coastal hazards. While these exposure models and their outputs help to locate and quantify exposed people and assets to a given hazard, the results of these models represent one point in time. Thus, they do not accurately represent the dynamic nature of exposure.

Other data sources can capture dynamic exposure, such as live traffic flows from transportation agencies (e.g., the New Zealand Transport Agency) (Data Management Gov. NZ. Nat. A; Met. NZ. K). Cell phone data was also discussed for capturing population shifts (Risk Modelling NZ. A; Data Management Gov. NZ. Nat. E). For example, Data Ventures,

This product became useful in the COVID-19 response where Data Ventures provided near real-time population movements to the National Crisis Management Centre which "provided [a] very close to real-time view of whether or not people were following the advice" (Data Management Gov. NZ. Nat. E), which informed advice to the Prime Minister's COVID-19 advisory group "about whether or not we move or shift down levels of lockdown" (Data Management Gov. NZ. Nat. E). Through this, they were also able to determine "whether or not people were moving from one region to another because that's really how you have that wider contagion risk" (Data Management Gov. NZ. Nat. E). This has potential application in an IFW system for identifying the "catchment of individuals" (Data Management Gov. NZ. Nat. E) in near real-time for more contextualised warnings. However, this data alone does not provide an overall risk indication; "you have to put other layers of information to go 'well if it's a ... category 5 [ex-tropical cyclone] then are the buildings of a particular standard and ... how many people are in the areas that those buildings are present in'" (Data Management Gov. NZ. Nat. E). Hence the need for vulnerability information as previously discussed

Knowledge of dynamic human exposure can also come from knowing about community events (Auckland Workshop). A regional EM official in NZ identified the need for a "real-time understanding of what is going

¹⁶ See https://population-density.dataventures.nz/explorer2/help/index.html for more.

¹⁵ OpenStreetMap is a form of geographic crowdsourcing in which volunteers digitize features of the earth onto an online map of the world. This data is produced under a Creative Commons license, thus making it open-source and freely available for download.

a commercial data brokerage branch of Stats NZ, used cell phone data to produce population densities at the Statistical Area 2 level (a higher scale than the Meshblock level) for a given time range $^{16}. \label{eq:scale}$

¹²



ternational Journal of Disaster Risk Reduction 66 (2021) 102619



Fig. 1. Results of applying the coding paradigm to understand the causal conditions, intervening conditions, and action/interaction strategies for HIVE data creation, collection, and use phenomenon as defined in Table 3. Causal conditions are the drivers of the phenomena. Intervening conditions are conditions that inhibit or facilitate the phenomena. Action/interactions strategies are strategies that were identified to address the intervening conditions.

on in the community (i.e., sports events, hotel capacity, etc.)" (Auckland Workshop). In this case, the regional council's economic and cultural department possesses this knowledge and information (Auckland Workshop) and should be shared with the EMs and other agencies such as the MetService who would need to be aware of events if severe weather were to occur.

3.2. Data creation, collection, and use

Discussions with participants revealed several inhibitors and facilitators to collecting, creating, and using the data. Using the ES-GT coding paradigm, we identified the causal conditions driving data collection, creation, and use, intervening conditions to this phenomenon, and actions and strategies that have been used to address those intervening conditions (Fig. 1). The causal conditions are presented next.

3.2.1. Causal conditions

Causal conditions were identified as drivers for agencies collecting and using HIVE data in general for different purposes and via different methods. The main causal conditions identified by participants are (1) Disaster/emergency events, (2) technological advancements, and (3) Research.

Other factors can also be attributed to HIVE data collection, such as existing policies and plans (e.g., the National Disaster Resilience Strategy) within NZ, and international initiatives such as the Sendai Framework. For the purposes of this study, we have decided to focus these current findings on the causal conditions that participants identified more directly.

3.2.1.1. Disaster/emergency events. Our participants gave examples of how the creation, collection, and use of HIVE data has been driven by past disaster/emergency events. Following the 2010–2011 Canterbury earthquake sequence there was a need for data on the damaged build-ings, such as their age, construction material, and level of damage to inform recovery (Risk Modelling NZ. B). At the time, no database existed for building characteristics (e.g., history, age, construction material) (Risk Modelling NZ. B). Consequently, engineers had to "go through one by one to record the location and age... of the material" before they could conduct the damage assessments (Risk Modelling NZ. B). This experience where decision-makers were caught scrambling for data revealed a need for building up databases for future events (Risk Modelling NZ. B; EM. NZ. Reg. H).

This and other major events resulted in the formation of a Technical Advisory Group (TAG) by the Ministry of Civil Defence and Emergency Management (now the National Emergency Management Agency or NEMA), in 2017 [58]. The TAG conducted a Ministerial Review and provided several recommendations for better intelligence gathering (i. e., data collection) [59]. A participating regional EM official cited this review as a driver for their council and EM Group attempting to improve their data collection efforts for disaster response (EM. NZ. Reg. H). Additionally, in response to two back-to-back ex-tropical cyclones that resulted in disastrous flooding, the Bay of Plenty region EM Group now updates exposure maps for tropical cyclone hazards annually (EM. NZ. Reg. A).

3.2.1.2. Technological advancements. Technological advancements, particularly the implementation of geospatial technologies such as Geographic Information Systems (GIS), cloud-based services, and the proliferation of mobile devices with cameras and an internet connection have also driven efforts for better collection of HIVE data. For example, one participant described Esri's ArcGIS Online product as the "catalyst" for using geospatial technologies for emergency response (EM. NZ. Reg. H). The cloud-based aspect of the ArcGIS Online product supports rapid development and sharing of maps and other geospatial applications, which is seen as the most valuable feature for emergency response (EM. NZ. Reg. H). As such, "the strength of the NZGIS4EM [New Zealand Geographic for Emergency Management] community really ... is built upon that. And ... there's ... been continual growth with event after event" (EM. NZ. Reg. H). Furthermore, this cloud-based technology used in combination with mobile devices using applications like QuickCapture and Survey123 allowed for the redevelopment of how rapid impact assessments are conducted for more efficient and timely data collection. For example, with Survey123, "the fieldworker hits submit, that assessment will instantly show up on the map [in the Emergency Op-erations Centre]" to display the level of damage and safety of the buildings (EM. NZ. Reg. H).

3.2.1.3. Research. Research has also driven the collection of HIVE data. For example, NIWA's NZ Historic Events Catalogue¹⁷ was initially developed for research interests, where "people realised that it would be really good to have this database for people and researchers, and just sort of give an idea of what historically has happened ... And I think it's been quite useful" (Met. Research NZ. J). Other HIVE data has been collected to support the research and development of risk models (Risk Modelling NZ. B).

17 https://hwe.niwa.co.nz/.

⁵ The acronyms and abbreviations in Table 7 are as follows: Fire and Emergency New Zealand (FENZ), National Institute of Water and Atmospheric Research Ltd (NIWA), New Zealand Transport Agency (NZTA), Institute of Geological and Nuclear Sciences Limited (GNS Science), Statistics New Zealand (Stats NZ), Ministry of Business, Innovation, and Employment (MBIE).

^{3.2.2.} Intervening conditions for data creation, collection, and use Intervening conditions were found to affect whether HIVE data is

collected, and the choice for which data source is used. Herein we focus on three intervening conditions: (1) Priorities, motivation, and interest; (2) Official and unofficial data; (3) Timescale and trustworthiness.

3.2.2.1. Priorities, motivation, and interest. Conflicting priorities and a lack of motivation or interest in data creation and maintenance was a barrier identified by several participants for gathering HIVE data (EM. Gov. NZ. Nat. G; Data Management Gov. NZ. Nat. A; Data Management Private NZ. B; Data Management Research NZ. C; EM. NZ. Reg. D). Management priorities and the personal interests of key staff within an organisation appear to either inhibit or enable data collection and creation, as one NZ risk modeller summarised: "it just depends on who's here and who's leading the team" (Risk Modelling NZ. A).

The National Emergency Management Agency (NEMA) for New Zealand reports losses and impacts to the UNDRR in fulfilment of Sendai Framework priorities. Their goal is to develop a national loss and impact database (EM. Gov. NZ. Nat. G). However, progress on this front is slow as NEMA is continuously busy responding to other events and so is unable to direct resources towards developing a national loss and impact database (EM. Gov. NZ. Nat. G). This may be due to the reactive nature of the EM sector [16], where agencies lack the time or resources to establish proper data collection and management practices (EM. Gov. NZ. Nat. G). Loss Modelling Research NZ. A; EM. NZ. Reg. A, B, H; Data Management Private NZ. B; Hyd. Gov. NZ. Nzt. Ge, A).

3.2.2.2. Official and unofficial data. The leading mandate of many of the participating agencies is to preserve life and property, be it through designing and issuing warnings (e.g., NMSs, warning services) or coordinating emergency response plans and actions (EM agencies). A key role for these agencies is providing the voice of truth during a severe weather event. These agencies must maintain a high level of credibility and trustworthiness amongst the public and stakeholders to ensure that their messages are heeded (NMS. Int. C, D, E; NMS. NZ). Since these agencies use a plethora of data to communicate critical information and alerts to the public, trust in the supporting information underpinned all discussions with participants.

Official and unofficial datasets and sources were discussed and compared with interviewees. Interviewees showed a preference for official datasets and sources because of their role as an authoritative voice in saying "this is what happened" or may happen (NMS. Int. C). However, collecting and using official data is not always possible. For example, agencies who possess the data may be unable or unwilling to share (NMS. Int. C), or agencies may need real-time data, which is rarely official or trusted (NMS. Int. D). In these cases, agencies may need to turn to unofficial data sources such as crowdsourcing and social media (NMS. Int. C).

3.2.2.3. Time scale and trustworthiness. Forecasting impacts in real-time or near real-time for early warning is an operational goal [8,70]. Thus, our participants identified a need for real-time or near real-time data (Met. Int. E; EM. NZ. Reg. H; Risk Modelling NZ. C, D; Data Management Private NZ. B). Potential real-time data sources that were identified are crowdsourcing, social media, and mobile tracking data. Aside from the mobile tracking data, these sources are unofficial data sources, resulting in decreased perceptions of trust and credibility in the data (Met. NZ. H; Met. Int. D, I). As such, our results indicate trustworthiness, and the timescale needs appear to be two intervening conditions in the choice and use of a data source. A tension appears to exist between these two conditions, as officials (e.g., warning agencies, emergency managers, responders) need both timely and trustworthy data, but often have to compromise on either factor [71].

Distrust is a critical obstacle to warning adherence [72]. Thus, it is not surprising that trust in the data and in the sources to support IFWs is a primary driver in deciding which impact, exposure, and vulnerability dataset/source to use. It is important that warning officials perceive the

International Journal of Disaster Risk Reduction 66 (2021) 102619

data as trustworthy [73], as this will ensure that public trust in the agency and in the warnings is maintained. Different uses of impact, exposure, and vulnerability data require varying levels of trustworthiness.

The timescale needed also determines the most appropriate dataset. We found that some data sources, like crowdsourcing and social media, fill a gap in the need for real-time or near real-time data for verification, situational awareness, and response. This finding supports those of a recent survey of European NMSs [40]. However, these data may be less useful for defining impact thresholds and informing impact/risk models due to the perceived limited quality and trustworthiness. For example, social media data appeared to be less trustworthy amongst participants because it is difficult to vet and lacks the structure needed for forecasting and modelling (Met. NZ. G, H). However, social media remains useful for building situational awareness quickly and updating alerts (EM NZ Reg. C; [54]).

3.2.3. Action/interaction strategies

Action/interaction strategies were found to address the intervening conditions previously identified. The three strategies we focused on here are: (1) Garnering support and buy-in, (2) Individual and community leadership, and (3) Quality control and standardisation.

3.2.3.1. Garnering support and buy-in. Garnering support and buy-in was found to be an action/interaction strategy for overcoming conflicting management priorities or a lack of motivation and interest. One regional EM official has been pushing for a GIS-based approach for improved intelligence gathering, management, sharing, etc., but has faced resourcing challenges. The first "informal" (EM. NZ. Reg. H) stage of this project "didn't turn out to be sustainable because we had a lot of issues with trying to carve out time from people to actually contribute to it" (EM. NZ. Reg. H). As such, they developed a "more formal" (EM. NZ. Reg. H) strategy to approach decision-makers for support for resource allocation across regional stakeholders/agencies (EM. NZ. Reg. H).

3.2.3.2. Individual and community leadership. Individual and community leadership was found to be another action/interaction strategy for overcoming misdirected management priorities and lack of motivation or interest. Some regional agencies have improved their own data collection practices based on the leadership of their in-house GIS and EM experts. For example, the West Coast CDEM group regularly develops innovative ways for using GIS-based technology to carry out impact assessments (EM. NZ. Reg. C, F; see [56]). Furthermore, the emergence of the NZGIS4EM¹⁸ group has been identified by participants as a major driver for technological advancement and for pushing the needle forwards on creating, sharing, and accessing geospatial data and tools for emergency response (EM. NZ. Reg. H; Data Management Gov. NZ. Nat. A; Data Management Private NZ. B; EM. NZ. Reg. D).

This innovative GIS-based work is credited to specific individuals within the sector who possess a passion and expertise to drive these efforts (Data Management Private NZ. B; EM. NZ. Reg. B, F). Furthermore, the innovation may also be due to available resources and support from management.

Our findings align with the Policy Capacity Framework [74] which outlines three levels of capacity development and implementation: individual, organisation, and systemic; and three dimensions: analytical, managerial, and political. The individuals accredited with driving the GIS4EM movement in NZ appear to possess analytical, managerial, and political acumen capacity at the individual level by possessing

¹⁸ NZ Geographic Information Systems for Emergency Management (NZGI-S4EM) is a grassroots community of GIS specialists and EM practitioners in NZ that began working together in the mid-2010's to identify how they could build and use geospatial tools for better emergency response (EM. NZ. Reg. H).



analytical, technical, communication, and leadership skills to drive technological innovation within their sector. Agencies like the West Coast CDEM Group possess technical and administrative capacity at the organisational level by providing and coordinating the resources needed to allow the individuals to implement their innovative solutions. More investigation is needed to understand the policy capacities at the systemic level and political dimension within the NZ EM and severe weather warning space. However, it appears that the GIS movement within NZ's EM sector has reached central government decision-makers with the recent release of the Impact Assessments Director's Guideline for *Civil Defence Emergency Management Groups [DGL 22/20]* by [55]. This guidance document outlines the role of GIS and supporting spatial tools for undertaking impact assessments.

3.2.3.3. Quality control and standardisation. Participating agencies emphasised the importance of applying quality control measures to increase the perceived trustworthiness of the data. Some quality control measures that came up in the conversations included vetting the source (s) of the data, training storm spotters, cross-validating between sources for accuracy, timing, and location. For example, the NMSs in the USA and Austria train and vet storm spotters so that they can trust incoming ground observations (Met. Int. E, I; [40,75]). Standardised post-event damage/impact assessments collected by those trained to, and entered into a database for further analysis, may be more suitable for needs that do not require real-time data [40].

4. Conclusions and limitations

Documenting hazard, impact, vulnerability, and exposure data fulfil needs for IFWs and meets Sendai Framework priorities for improved understanding of disaster risks and subsequent mitigation and reduction. The New Zealand focus of this research further supports an identified need for better risk data for modelling and natural hazard management in New Zealand.

In this exploratory study, we identified sources for hazard, impact, vulnerability, and exposure data for implementing severe weather IFW systems. Our findings indicate that many sources for hazard and impact data exist that are collected for other uses (such as for response, and research) and have relevant applications for IFWs. Furthermore, un-derlying datasets for vulnerability and exposure exist and are available. Technological advancements have also enabled the collection and creation of HIVE data, such as GIS-based tools and mobile devices.

We also identified intervening conditions, and action/interaction strategies for collecting HIVE data, as shown in Fig. 1. Our findings suggest that priorities, motivation, and interest within organisations influence how well data is collected and used. Furthermore, agencies tend to prefer official data, but official data has limitations that unofficial data can sometimes address, such as timeliness. To that end, a tension exists between the timeliness and trustworthiness of data needed for emergency response and warnings.

To address these intervening conditions, we identified some action/ interaction strategies using Grounded Theory. Garnering support and buy-in from decision-makers and upper management within an agency can redirect priorities and increase motivation and interest in collecting HIVE data. Individual and community leadership within the field of practice also provides a bottom-up approach for driving industry priorities and practices for collecting HIVE data. Furthermore, measures for quality control and data standardisation may improve the perceived trustworthiness of the data.

The qualitative nature of data collection and analysis herein limits the generalisability of results beyond the interviewees. However, the qualitative approach offers an in-depth understanding of a problem not readily available from quantitative approaches [76-78]. Furthermore, participant recruitment and data collection methods were affected by the COVID-19 pandemic response, as many individuals and agencies

tional Journal of Disaster Risk Reduction 66 (2021) 102619

targeted for recruitment were involved in the response. As such, some perspectives may be missing from the qualitative dataset

Discussions with colleagues in the field pointed towards the value of mātauranga Māori (Māori knowledge) in understanding disaster risk and impacts in New Zealand. Considerations about cultural ownership of such knowledge and its use in an impact forecasting and warning system is thus an important area for future research.

Findings from this research provide insight into the drivers and barriers for collecting hazard, impact, vulnerability, and exposure data in New Zealand. Sources of such data were identified such that practitioners and researchers may seek out these datasets if so desired. Further questions remain around how the data can be accessed and acquired for use in an IFW system. Future research should explore the data acquisition process for these datasets.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

This research was funded by New Zealand's Ministry of Business, Innovation & Employment via the GNS Science Strategic Science Investment Fund and Resilience to Nature's Challenges Kia manawaroa -Ngā Ākina o Te Ao Tūroa. The authors thank the participants of the research for providing their invaluable time and insights, and the New Zealand MetService - Te Ratonga Tirorangi for their insightful conversations and advice over the course of this research. The authors also thank the reviewers whose feedback improved this manuscript.

References

- [1] UNDRR, Sendai Framework for Disaster Risk Reduction 2015 2030, UNDRR,
- 2015-203 uction-2015-2030.
 M. McPherson, M. Counahan, J.L. Hall, Responding to Typhoon haiyan in the Philippines, Western Pac. Surveill. Response J. 6 (2015) 1–4, https://doi.org/
- 10.5365/WPSAR.2015.6.4.HYN_026. [3] P.K. Ching, V. Carr de los Reyes, M.N. Sucaldito, E. Tayag, An assessment of PSAR.2015.6.2.HYN_005.
- P.K. Ching, V. Carr de los Reyes, M.N. Sucaldito, E. Tayag, An assessment of disaster-related mortality post-Haiyan in Tacloban City, Western Pac. Surveill. Response J. 6 (2015) 34–38, https://doi.org/10.5365/WPSAR.2015.6.2.HYN_0
 WMO, WMO guidelines on multi-hazard impact-based forecast and warning services (9798263111500). https://www.wmo.int/pages/prog/www/DPFS/M eetings/ET-OWFPS_Montreal2016/documents/WMOGuidelinesonMulti-hazard vices.pdf, 2015
- Information of the association of the associatio
- 1-4, https://doi.org/10.1016/j.jdfr.2016.04.002.
 B. Mackie, Warning Fatigue: Insights From the Australian Bushfire Context, Doctor of Philosophy in Media and Communication, University of Canterbury, 2013, https://ir.canterbury.ac.nz/handle/10092/9029. [6]
- //ir.canterbury.ac.nz/handle/10092/9029.
 Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E. H., & Johnston, D. M. (submitted for publication). Identifying the Data Uses and Gaps for Severe Weather Impact Forecasts and Warnings. Weather, Climate, and Society.
 R. Hemingway, J. Robbins, Developing a hazard-impact model to support impact-based forecasts and warnings: the Vehicle OverTurning (VOT) Model, Meteorol. Appl. (2019) 1–17, https://doi.org/10.1002/met.1819.
 J. Harrison, C. McCoy, K. Bunting-Howarth, H. Sorensen, K. Williams, C. Ellis, Poularison, 2014
- [9]
- Evaluation of the National Weather Service Impact-Based Warning Tool, 2014.
 [10] S.H. Potter, S.E. Harrison, P. Kreft, The benefits and challenges of implementing impact-based severe weather warning systems: perspectives of weather, flood, and emergency management personnel, Weather, Clim., Soc. 13 (2) (2021) 303–314, https://doi.org/10.1010/j.
- https://doi.org/10.1175/wcas-d-20-0110.1.
 P. Heinselman, D. LaDue, D.M. Kingfield, R. Hoffman, Tornado warning decisions using phased-array radar data, Weather Forecast. 30 (2015) 57–78, https://doi. -14-00042.1
- org/10.1175/WAF-D-14-00042.1.
 [12] T. Gneiting, A.E. Rafrey, Weather forecasting with ensemble methods, Science 310 (5746) (2005) 248–249, https://doi.org/10.1126/science.1115255.
 [13] P.A. Hinschberg, E. Abrams, A. Bleistein, W. Bua, L.D. Monache, T.W. Dulong, J. E. Gaynor, B. Glahn, T.M. Hamill, J.A. Hansen, D.C. Hilderbrand, R.N. Hoffman, B. H. Morrow, B.P. Philips, J. Sokich, N. Stuart, A weather and climate enterprise strategic implementation plan for generating and communicating forecast uncertainty information, Bull. Am. Meteorol. Soc. (2011) 1651–1666, https://doi.org/10.1125/BMM-50-00073.1. rg/10.1175/BAMS-D-10-00

- [14] M.C. Fava, G. Santana, D.A. Bressiani, A. Rosa, F.E.A. Horita, E.M. Mendiondo, Integration Of Information Technology Systems For Flood Forecasting with Hybrid Du Sources 6th International Conference on Flood Management, Sao Paulo, Brazil,
- 2014.
 [15] T. Krennert, G. Pistotnik, R. Kaltenberger, C. Csekits, Crowdsourcing of weather observations at national meteorological and hydrological services in Europe, Adv. Sci. Res. 15 (2018) 71–76, https://doi.org/10.5194/asr-15-71-2018.
 [16] T. Kox, C. Lidder, L. Gerhold, Anticipation and response: emergency services in severe weather situations in Germany, Int. J. Disaster Risk Sci. 9 (2018) 116–128, https://doi.org/10.1007/21252.018.0165.0

- Severe weather situations in certainty, int. J. Disaster Nas. Oct. 7 (2016) 100-100, https://doi.org/10.1007/s13753-0148-0163-z.
 [17] S.E. Harrison, A. Silver, B. Doberstein, Post-storm damage surveys of tornado hazards in Canada: implications for mitigation and policy. Int. J. Disaster Risk Reduct. 13 (2015) 427-440, https://doi.org/10.1016/j.ijdrr.2015.08.005.
 [18] J.M. Sadler, J.L. Goodall, M.M. Morsy, K. Spencer, Modeling urban coastal flood severity from crowd-sourced flood reports using Poisson regression and Random Forest, J. Hydrol. 559 (2018) 43-55, https://doi.org/10.1016/j.ibu-brd 2018.01.044.
- [19] S.E. Harrison, S.H. Potter, R. Prasanna, E.E.H. Doyle, D. Johnston, Volunteered Geographic Information for people-centred severe weather early warning: a literature review, Australas. J. Disaster Trauma Stud. 24 (1) (2020). http://trau
- Internet events, A. F. Sonsker, Markan Stud. 24 (1) (2020). http://doi.org/10.1111/ internet.active.com/acti
- [21] G. Cervone, E. Schnebele, N. Waters, M. Moccaldi, R. Sicignano, Using social media and satellite data for damage assessment in urban areas during emergencies, P. Thakuriah, N. Tilahun, M. Zellner (Eds.), Seeing Cities through Big Data, Springer Geography, 2017, pp. 443–457, https://doi.org/10.1007/978-3-319-40902-3_24.
- 40902-3.24.
 [22] R.P. Crompton, K.J. McAneney, Normalised Australian insured losses from meteorological hazards: 1967-2006, Environmental Science & Policy II, 2008, pp. 371–378, https://doi.org/10.1016/j.envscl.2008.01.005.
 [23] M. Panteli, C. Pickering, S. Wilkinson, R. Dawson, P. Mancarella, Power system resilience to extreme weather: fragility modeling, probabilistic impact assessmen and adaptation measures, IEEE Trans. Power Syst. 32 (5) (2017) 3747–3757, https://doi.org/10.1109/TPWRS.2016.2641463.

- https://doi.org/10.1109/TPWRS.2016.2641463.
 J.C. Robbins, H.A. Titley, Evaluating high-impact precipitation forecasts from the Met Office Global Hazard Map using a global impact database, Meteorol. Appl. (2018) 548–560, https://doi.org/10.1002/met.1720, July 2017.
 S. Doocy, A. Daniels, S. Murray, T.D. Kirsch, The human impact of floods: a historical review of events 1980-2009 and systematic literature review, PLoS Currents (2013), https://doi.org/10.1371/currentsdis. fddeb457904936b07c09dna98ee8171a. APR 2013.
 F. Sai, L. Cumiskey, A. Weerts, B. Bhattacharya, Towards impact-based flood forecasting and warning in Bangladeski: a case study at the local level in Siraiganj district, Nat. Hazards Earth Syst. Sci. Discussions(February) (2018) 1–22, https://doi.org/10.15194/nhess-2016-122.
- [27] E.R. Poolman, A Probabilistic Impact-Focussed Early Warning System for Flash Floods in Support of Disaster Management in South Africa, Doctor of Philosophy in Meteorology, University of Pretoria, Pretoria, 2014, http://hdl.handle.net/2
- [28] G. Terti, I. Ruin, S. Anquetin, J.J. Gourley, Dynamic vulnerability factors for impact-based flash flood prediction, Nat. Hazards 79 (2015) 1481–1497, https:// 10.1007/s11069-015-1910-8
- Colorge 10.1007/S11089-015-1910-8.
 [29] I. McCallum, W. Liu, L. See, R. Mechler, A. Keating, S. Hochrainer-Stigler, J. Mochizuki, S. Fritz, S. Dugar, M. Arestegui, M. Szoenyi, J.-C.L. Bayas, P. Burek, A. French, I. Moorthy, Technologies to support community flood disaster risk reduction, Int. J. Disaster Risk Sci. 7 (2) (2016) 198–204, https://doi.org/10.1007/
- 13753-016-0086-5. E. Enarson, Chapter 13: identifying and addressing social vulnerabilit ... Waugh, K. Tiernev (Eds.). Emergency Management: Principles and i [30] E. Enars
- Enarson, Chapter 13: identifying and addressing social vulnerabilities, in: W. Waugh, K. Tienery (Eds.), Emergency Management: Principles and Practices fo cal Government, second ed., ICMA Press, 2007, pp. 257–278. Mason, K. Lindberg, C. Haenfling, A. Schori, H. Marsters, D. Read, B. Borman, cial vulnerability indicators for flooding in autoearoa New Zealand, Int. J. wiron. Res. Publ. Health 18 (8) (2021), https://doi.org/10.3390/ embl.092020 [31] K. Ma
- J.P.F. Rogers, H. Kootval, V.V. Tsirkunov, Early warning, resilient infrastructure and risk transfer, in: Bridging Science and Policy Implication for Managing Climate Extremes, 2017, pp. 65–79. www.gfdrr.org.
 [33] GFDRR, Implementing Multi-Hazard Impact-Based Forecast and Warning Services,

- 2016.
 R. Paulik, S. Stephens, R. Bell, S. Wadhwa, B. Popovich, National-scale built-environment exposure to 100-year extreme sea levels and sea-level rise, Sustainability 12 (4) (2020), https://doi.org/10.3390/sal2041513.
 S.L. Lin, J. Ungaro, J.T. Nimau, S. Singh, K. Raqisia, Development of pacific exposure dataset for use in catastrophe risk assessment. Natural Hazards, 2020, https://doi.org/10.1007/s1169-020-04290-4.
 A. Mansourian, A. Rajabifard, M.J. Valadan Zoej, I. Williamson, Using SDI and web-based system to facilitate disaster management. Comunt. Geosci. 32 (3)
- eb-based system to facilitate disaster management, Comput. Geosci. 32 (3)
- (2006) 303–315, https://doi.org/10.1016/j.cageo.2005.06.017. WMO, Multi-hazard Early Warning Systems: A Checklist, in: The First Multi-Hazard [37]
- Early Warning Conference, 2018 (Cancun, Mexico).
 D. Zhang, L. Zhou, J.F.J. Nunamaker, A knowledge management framework for the support of decision making in humanitarian assistance/disaster relief, Knowl. Inf. Syst. 4 (2002) 370-385, https://doi.org/10.1007/s101150200012.

International Journal of Disaster Risk Reduction 66 (2021) 102619

- International Journal of Disaster Risk Reduction 66 (2021) 102619
 [39] S.E. Harrison, P.A. Johnson, Challenges in the adoption of crisis crowdsourcing and social media in Canadian emergency management, Govern. Inf. Q. 36 (2019) 501–559, https://doi.org/10.1016/j.giq.2019.04.002.
 [40] R. Kaltenberger, A. Schaffhauser, M. Staudinger, "What the weather will do" results of a survey on impact-oriented and impact-based warnings in European NMHSs, Adv. Sci. Res. 17 (2020) 29–38, https://doi.org/10.1016/j.gid.2019.04.002.
 [41] M.H. Crawford, K. Crowley, S.H. Potter, W.S.A. Saudners, D.M. Johnson, Risk modelling as a tool to support natural hazard risk management in New Zealand local government, Int. J. Disaster Risk Reduct. 28 (January) (2018) 610–619, https://doi.org/10.1016/j.jidrr.2018.61.011.
 [42] A.L. Strauss, J.M. Corbin, Basics of Qualitative Research: Grounded Theory Procedures and Techniques, Sage Publications, 1990.
 [43] J. Corbin, A. Strauss, in: V. Knight, Y. McDuffee, L. Barrett, M. Markanich (Eds.), Basics of Qualitative Research: Techniques and Procedures for Developing Grounded Theory, fourth ed., SAGE Publications Inc, 2015.
 [44] M. Bergin, Nivo 8 and consistency in data analysis: reflecting on the use of a qualitative data analysis program, Nurse Res. 18 (3) (2011) 6–12, https://doi.org/10.07748/nr2011.04.18.3.6c.8457.
 [45] M. Vollstetd, S. Rezat, An introduction to grounded theory with a special focus on axial coding and the coding paradigm, in: G. Kaiser, N. Presmeg (Eds.), Compendium for Early Career Researchers in Mathematics Education, Springer International Publishing, 2019, pp. 81–100, https://doi.org/10.1077/978-3-030-15636-7.4.
 [46] M. Williamson, Meteorological Service Contract, New Zealand Government,
- [46] M. Williamson, Meteorological Service Contract, New Zealand Government, Wellington, New Zealand, 1998. Retrieved from, https://www.beehive.govt.
- [47] MBIE, Crown research Institutes, 10 May from, https://www.mbie.go/ ience-and-technology/science-and-innovation/agencies-policies-and-bit ience-and-technology/science-and-innovation/agencies-policies-and-bit agencies-policies-and-budget-initiat
- research-organisations/cri/, 2021. einer, J.R. Martin, N.D. Gordon, M.A. Grant, Commercialisation in th sion of meteorological services in New Zealand, Meteorol. Appl. 4 ([48] nd. Meteorol. Appl. 4 (1997)

- 147-257.
 [49] McSEvrice, Open access data, Retrieved 22 June from, https://about.metservice.com/our-company/about-this-site/open-access-data/, 2021.
 [50] WREMO, Significant Wave Hazards Project: Gloseout Report, April 2021, W. R. E. M. Office, 2021.
 [51] H. Rouse, Flood risk management research in New Zealand Where are we, and where are we going? (2012/04), GNS Sci. (2011).
 [52] G. D'Amato, C. Vitale, M. D'Amato, L. Cecchi, G. Liccardi, A. Molino, A. Vartella, A. Sanduzzi, C. Maesano, I. Annesi-Maesano, Thunderstorm-related asthma: what happens and why, Clin. Exp. Allergy 46 (3) (2016) 390-396, https://doi.org/10.1111/j.ea.12709.
- [53] M. Sharpe, Hay fever sufferers beware: the great yellow clouds of swirling dust are upon us, Stuff. Retrieved, https://www.stuff.co.nz/national/health/122418563/h at-yellow-clouds-of-swirling-dust-are-upon-2020. (Accessed 10 May 2021).
- 016100102
- IJJSCRIM.201610102.
 ISJ NEMA, Impact assessments Director's guideline for Civil defence emergency management groups [DGL 22/20]. DGL 22/20]. Wellington, NZ: New Zealand government, Retrieved from, https://www.civildefence.govt.nz/edem-sec tor/guidelines/impact-assessments-directors-guideline-and-national-impact nd-dictionary-technical-stand d/. 2020
- [56] J. Stowell, West Coast flooding event, Retrieved 24 April 2021 from, https://stor fe5c8a4138b21f20c215a57f6b. 2020 [57] Esri (n.d.-b), ArcGIS Survey123, from, https://survey123.arcgis.com/. (Accessed
- 20 May 2021)
- 20 May 2021).
 [58] Esri (n.d.-a), ArcGIS QuickCapture, from, https://www.esri.com/en-us/arcgis/products/arcgis-quickcapture/overview. (Accessed 20 May 2021).
 [59] MODEM, Part II section 7 needs assessment welfare services in an emergency Director's guideline [DGL 11/15]. (DGL 11/15). Wellington, NZ: New Zealand government, Retrieved from, https://www.civildefance.govt.nz/assets/Welfare-Services-in-an-Emergency-Directors-Guidelin

- assessment, New Zealand infinites council, http://www.itziteines.org.itz/site/ assets/files/1019/nzlc,nva_2020_full_report.pdf, 2020.
 [63] M. Jack, C. Graziadei, Report of the independent review of New Zealand's 2018
- census. https://www.stats.govt.nz/assets/Uploads/Report-of-the-Independent-Review-of-New-Zealands-2018-Census/Independent-review-report.pdf, 2019.
 [54] Stats Nz, Meshblock 2021 (generalised), from, https://datafinder.stats.govt.nz/1 ayer/105176-meshblock-2021-generalised/#/:--ttext-a%20nmeshblock%20is%20 the%20smallest,large%20area%200f%20rural%20land, 2021. (Accessed 12 May 2021) 2021).
- [65] P. Fdez-Arroyabe, L. Lecha Estela, F. Schimt, Digital divide, biometeorological data infrastructures and human vulnerability definition. Int. J. Biometeorol. 62 (5) (2018) 733-740, https://doi.org/10.1007/s00484-017-1398-x.
 [66] H.-C. Hung, L.-Y. Chen, Incorporating stakeholders' knowledge into assessing vulnerability to climatic hazards: application to the river basin management in

Taiwan, Climatic Change 120 (1-2) (2013) 491-507, https://doi.org/10.1007/

- raiwan, Climatic Change 120 (1-2) (2013) 491-507, https://doi.org/10.1007/ s10584-013-0819-z.
 [67] R. Bell, R. Paulik, S. Wadhwa, Coastal Hazard and Climate-Change Risk Exposure in New Zealand: Comparing Regions and Urban Areas. FIG Working Week 2016 Recovery from Disaster, 2016 (Christehurch, New Zealand).
 [68] G. Treadgold, J. Guan, P. Morton, S. Chambers, *Developing a Regional Approach and Strategy for Geographical Information Systems for Emergency Management* ISCRAM Asia-Pacific, 2018 (Wellington, New Zealand).
 [69] Technical Advisory Group, Ministerial Review Retter Response To Natural Disasters And Other Emergencies, Departmenr of the Prime Minister and Cabinet, 2017. https://dpmc.govt.nz/sites/default/files/2018-01/ministerial-review-better -responses-natural-disaster-other-emergencies.pdi Impact Library for Flood Risk Assessment ELOODrisk 2016 3rd European Conference on Flood Risk Massement ELOODrisk 2016 3rd European Conference on Flood Risk Massement ELOODrisk 2016 3rd European Conference on Flood Risk Massement ELOODrisk 2016 3rd European Conference on Flood Risk Massement ELOODrisk 2016 3rd European Conference on Flood Risk Massement ELOODrisk 2016 3rd European Conference on Flood Risk Massement, 2016 (Lyon, France).
 [7] A.M. Mehra, A. Bruns, J. Newton, Trus, but verify: social media models for disaster management, Disasters 41 (3) (2017) 549-565, https://doi.org/10.1111/ disa.12218.

International Journal of Disaster Risk Reduction 66 (2021) 102619

- International Journal of Disaster Risk Reduction 66 (2021) 102619
 [72] V. Covello, P.M. Sandman, Risk communication: evolution and revolution, in: Solutions to an Environment in Peril, John Hopkins University Press, 2001, pp. 164–178.
 [73] G. Terti, I. Ruin, M. Kalas, I. Láng, A. Cangrös, V. Lorini, ANVCaRE: a role-playing game to investigate crisis decision-making and communication challenges in weather-related hazards, Nat. Hazards Earth Syst. Sci. 19 (2019) 1–43.
 [74] X. Wu, M. Howlet, M. Ramesh, Policy capacity: conceptual framework and essential components, in: Policy Capacity and Governance: Assessing Governmental Competences and Capabilities in Theoreptual framework and essential components, in: Policy Capacity: conceptual framework and sesential components, in: Policy Capacity cancer, P. Caller, M. Stampfl, Trusted Spotter Network Austria a new standard to utilize crowdsourced weather and impact observations, Adv. Sci. Res. 15 (2018) 77–80, https://doi.org/ 10.5194/art.15-77-2018.
 [76] H. Blumer, Symbolic Interactionism: Perspective and Method, first ed., University of California Press, 1969.
 [77] M.B. Miles, A.M. Huberman, Qualitative Data Analysis: an Expanded Sourcebook, second ed., Sage Publications, Inc, 1994.
 [78] M.Q. Patton, Qualitative Research and Evaluation Methods, third ed., Sage Publications, Inc, 2002.

Appendix M: Chapter Eight Publication

Progress in Disaster Science 13 (2022) 100213



'Sharing is caring': A socio-technical analysis of the sharing and governing of hydrometeorological hazard, impact, vulnerability, and exposure data in Aotearoa New Zealand

Sara E. Harrison ^{a,b,*}, Sally H. Potter ^b, Raj Prasanna ^a, Emma E.H. Doyle ^a, David Johnston ^a

^a Joint Centre for Disaster Research, Massey University, Wallace Street, Mount Cook, Wellington 6021, New Zealand ^b GNS Science, 1 Fairway Drive, Avalon, Lower Hutt 5011, New Zealand

ARTICLE	INFO
Article history:	
Received 4 Octobe	er 2021
Received in revise	d form 23 December 2021
Accepted 30 Dece	mber 2021
Available online 0	3 January 2022

Keywords Impact-based warnings Severe weather Risk data Impacts Vulnerability Exposure

ABSTRACT

There has been a growing recognition of the need to collect disaster and risk data over the last two decades. Accordingly, better collection and management of disaster data was identified as a priority of the Sendai Framework for Di-saster Risk Reduction. The introduction and implementation of Impact Forecasts and Warnings (IFWs) have further highlighted this need to collect and access hazard, impact, vulnerability, and exposure (HIVE) data. However, challenges have been met with reporting and using disaster data, which have resulted in an identified need to establish principles for data collection, recording, reporting, exchange/sharing, and comparability. This introduces the concept of data governance and management for disaster data, particularly with regards to data custodianship, stewardship, and sharing.

Using Grounded Theory, a series of interviews were conducted with users and creators of HIVE data to develop further understanding around managing and accessing it for severe weather hazards in New Zealand. A socio-technical lens guided the analysis to identify the organisational and technical intervening conditions and action/interaction strategies for accessing and sharing HIVE data in NZ. Findings indicated that there is a need to establish data governance principles for HIVE data in New Zealand. An ad-

ditional need was identified for nurturing partnerships to continue building trust between stakeholders for sharing data. Furthermore, integration challenges continue to interfere with the use of various sources of HIVE data for effective risk and impact assessments for IFWs and beyond. Systematic and standardised data collection approaches using GIS-based tools can support integration.

1. Introduction

There has been a growing need to collect disaster and risk data over the last two decades (e.g., [32]). Accordingly, better collection and management of disaster data was identified as a priority of the Sendai Framework for Disaster Risk Reduction [81]. In response, global initiatives now exist with the objective of developing technical guidance for building up disaster and risk data, such as the Integrated Research on Disaster Risk (IRDR) programme and subsequent workshops [12,25]. The introduction and implementation of Impact Forecasts and Warnings

(IFWs) for hydrometeorological hazards (e.g., [84,88]) has further highlighted the need to collect and access relevant disaster data for mitigation and prevention; namely hazard, impact, vulnerability, and expo-sure (HIVE) data [35,70]. However, challenges have been met with reporting and using disaster data, such as the fact that many stakeholders are involved in the collection, creation, and use of disaster data, making it difficult to integrate different data sources and perform comparative analyses [18]. These challenges have resulted in an identified need to establish principles and standards for data collection, recording, reporting, exchange/sharing, and comparability [25]. This

Management, view observations and and the second se (E.E.H. Doyle), d.m.johnston@massey.ac.nz (D. Johnston).

http://dx.doi.org/10.1016/j.pdisas.2021.100213

2590-0617/© 2021 Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Abbreviations: IFW, Impact Forecast and Warning; HIVE, Hazard, Impact, Vulnerability, and Exposure; DRR, Disaster Risk Reduction; IRDR, Integrated Research on Disaster Risk; EWS, Early Warning System; NHP, Natural Hazards Partnership; LINZ, Land Information New Zealand; NZTA, New Zealand Transport Agency; NMS, National Meteorological Service; ES-GT, Evolved-Straussian Grounded Theory; CDEM, Civil Defence and Emergency Management; EM, Emergency Management; NZMA, National Emergency Management; Agency; GNS Science, Institute of Geological and Nuclear Science Simited; NWA, National Motelar Science Simited; NWA, National Demergency Management; GG, Geographic Information Systems; COP, Common Operating Picture; CAP, Common Alerting Protocol; FENZ, Fire and Emergency New Zealand; IDSS, Impact-based Decision

introduces the concept of data governance for disaster data [12,45,56], particularly with regards to data custodianship, stewardship, and sharing [69,86].

Building on recent research where we identified the data needs, uses, and sources of HIVE data for implementing hydrometeorological IFW systems in Aotearoa-New Zealand (see [34,35]), we aim to develop further understanding around managing and accessing these data sources. We begin by presenting background on data governance and management, and data access and sharing, where data accessibility can affect data sharing [64].

1.1. Data governance

Data governance is an emerging field of research, with no agreed-upon definition [1]. We adopt the definition provided by Benfeldt et al. [7] where "data governance refers to the organisation and implementation of rules and responsibilities, which enforce decision making and accountabilities regarding an organisation's data assets" (p. 299). This is different from data management, which focuses on defining the data element, and how it is stored, structured, and moved [1]. Thus, data governance is argued to be a higher level of "planning and control over data management" ([1], p. 841).

There has been little mention of governance for disaster loss data in the disaster risk reduction (DRR) literature, except when it is pointed to as a need to improve data quality, access, sharing, and interoperability (e.g., [12,45,56]). Clarke et al. [12] proposed that data governance and in dependence be established for "strengthening and protecting data quality through national statistical offices that are functionally autonomous from other government agencies" (p. 4). Migliorini et al. [56] identified a lack of appropriate data governance arrangements to be impeding data access for DRR. Similarly, Li et al. [45] identified issues around data governance when using population health data for disaster risk research. In Aotearoa-New Zealand, Crawford et al. [14] identified chalenges with risk data collection due to unclear roles and responsibilities for doing so. Beyond these examples, we could not find studies that specifically investigated disaster data governance measures for improved data quality, sharing, and integration.

One element of data governance involves assigning roles and responsibilities, including decision rights and accountabilities, around how the data is managed, secured, validated, and made available [1,2,7]. Two such roles are data stewardship and data custodianship [86]. Often these roles may be confused or merged, however, they have distinct roles and responsibilities, discussed next.

1.1.1. Data stewardship and custodianship

Data stewardship and data custodianship relate to roles for managing data. In the context of health data, data stewardship consists of developing methods for "acquisition, storage, aggregation, deidentification, and procedures for data release and use" ([72], p. 1444). Similarly, the New Zealand (NZ) Government defines a data steward as an agency that operates at the systems and strategic level and promotes good practice to manage the data over its lifecycle, including planning and adjusting for technological obsolescence and long-term preservation and access [50,73]. Related, but distinct roles, are data custodians. For NZ government data these custodians are agencies that implement the data management practices stipulated by data stewards in daily practice [73] to ensure the quality and accessibility of the data [50].

In Aotearoa-New Zealand, several agencies have a data stewardship and/or custodianship role. Toitū Te Whenua Land Information New Zealand (LINZ; the government agency responsible for managing land titles, geodetic and cadastral survey systems, topographic information, hydrographic information, etc.) provided a Steward and Custodian Framework for New Zealand Fundamental Geospatial Themes and Datasets, to outline the responsibilities and expectations of appointed custodians and stewards of fundamental geospatial data (see [50]). LINZ also provided a partner document outlining the process of identifying and selecting fundamental geospatial data (see [49]). In summary, datasets are proposed to the

Progress in Disaster Science 13 (2022) 100213

New Zealand Geospatial Office for evaluation against a set of criteria to classify it as a fundamental geospatial dataset [49]. Fundamental geospatial datasets are "datasets that provide the minimum core set of nationallysignificant data that are critical to the effective running of [Aotearoa-New Zealand], and work together to help support growth in the economy" ([49], p. 4). Most of these fundamental geospatial datasets are not considered disaster datasets as they do not convey disaster losses, impacts, or risks. However, much of these underlying datasets can inform vulnerability and risk assessments when used in risk models and overlaid with hazard data to determine exposure (see [34,35]).

The fundamental geospatial datasets identified by LINZ are labelled as either Suggested (i.e., the agency suggested by LINZ has not yet agreed to the role and may not have been approached yet), Proposed (i.e., the suggested agency has agreed to the role but the commitment has not yet been formalised), Appointed (i.e., the proposed agency has formally committed to the role), or Not Evaluated (i.e., the datasets were proposed as fundamental datasets but were not yet evaluated to determine their status) [49]. For example, the Waka Kotahi New Zealand Transport Agency (NZTA) is the proposed custodian for the leadership and delivery of the national road network, while local government and territorial authorities are the suggested respective leadership and delivery Custodians for council roads. Leadership Custodians ensure that appropriate data management policies and standards are developed while Delivery Custodians are "responsible for the continued physical existence, availability, and integrity of the dataset" for as long as is required by the leadership custodian

Like disaster data governance, there is little mention of the stewardship of disaster data in the literature. This gap has been identified by Fakhruddin et al. [24] who identified the need for building capacity in data collection and stewardship to support reporting to the Sendai Framework, an international accord outlining global priorities for disaster risk reduction (DRR) [81]. Another challenge faced in DRR is accessing and sharing the required data [23] for emergency/disaster response, risk analysis (e.g., risk modelling and risk assessments), vulnerability assessments, and supporting IFWs [14,34,35].

1.2. Data access and sharing

Many stakeholders are involved in DRR who collect, produce, and manage their datasets. Access to and sharing of these datasets is necessary for DRR [24]. For example, in places like Europe with many countries sharing borders, it is important to be able to share data across borders [16]. Within a country, sharing data is important since, in many cases, disaster and risk data are collected by different agencies [34,35]. Data access refers to the retrieval and storage of data provided by the data holder and may be subject to technical, legal, and/or organisation requirements [64]. Data sharing is the voluntary provision of data by the data holder, including commercial and non-commercial conditional data sharing agreements [64]. Data accessibility is a spectrum, ranging from closed data to open data, and affects data shareability [64]. Data sharing is a socio-technical activity as it involves various parties forming data-sharing partnerships and technical systems to support integrating multiple datasets.

1.2.1. Organisational aspects of data access and sharing

Sharing disaster and risk data requires building partnerships between data stakeholders [23]. The Natural Hazards Partnership (NHP) is an example of a formal partnership developed in the UK between public agencies to improve disaster management across the country, including sharing data between the various agencies [36]. Building the NHP required extensive time, coordination, communication, and interaction between agencies [36].

Multi-organisational collaboration is wrought with challenges [74]. Building trust and increasing the willingness of organisations to participate is the first hurdle to overcome [74]. After this, new challenges include mutually identifying goals and objectives and agreeing on timelines, uses of differing terminology and epistemologies, legal issues around intellectual

property, and developing workflows and communication standards [74]. These early stages of scouting and building trust can take years [74]. Sustaining collaboration (e.g., maintaining interest and securing funding sources) remains an ongoing challenge [36,74]. In addition to building partnerships to share data, technical solutions are needed to enable data integration.

1.2.2. Technical aspects of data access and sharing

Data integration and interoperability are important, yet challenging, technical factors that support data sharing for DRR [23]. Disaster and risk data are available in countless formats, making integration difficult [39]. Thus there is a need to understand the various data sources and how they can be effectively and efficiently used [39].

Interoperability is a familiar challenge for both Early Warning Systems (EWSs) and data sharing and integration [10,38]. Interoperability issues for EWSs emerged as technological advancements led to a plethora of warning delivery mechanisms [10]. The Common Alerting Protocol (CAP) was proposed to set standards for warning design and delivery [10]. The CAP relies on standardising warning data for sharing across platforms [63,71]. The introduction of IFWs further adds to the challenge of standardisied data exchanges [41,70].

In Aotearoa-New Zealand the Canterbury earthquakes from 2010 to 2011 laid bare gaps in interoperability for data sharing during and after the disaster, and the need for standards-based interoperability for improved information and data management [79]. Furthermore, the NZ Civil Defence and Emergency Management (CDEM) sector is striving towards a Common Operating Picture (COP) in which all agencies and stakeholders involved in an event can access and view the same information [17]. This involves developing standards and capabilities for enabling access to and sharing of datasets [79].

Harrison et al. [34,35] identified and reported the various sources for HIVE data for hydrometeorological hazards in Aotearoa-New Zealand. While the data sources were identified, more understanding is needed around managing and accessing these data both for IFW systems and for general DRR [34,35].

The objectives of this research are to identify and understand the governance and acquisition process for hazard, impact, vulnerability, and exposure (HIVE) data for hydrometeorological hazards in Aotearoa-New Zealand, to support efforts to fulfil the Sendai Framework priorities around disaster data access and to support the implementation of a hydrometeorological IFW system.

2. Research methods

We used a qualitative approach to address the research question, specifically employing the Evolved-Straussian Grounded Theory (ES-GT) research strategy for data collection and analysis [13]. Interviews, workshops, and key documents were the primary data sources. Between November 2018 and April 2021, the lead author interviewed thirty-nine (n = 39) experts in weather forecasting, warning, emergency management, risk modelling, and data collection and management (see [35] for more information on the participant details). Three virtual workshops were held in Aotearoa-New Zealand. Two of these workshops involved Emergency Management (EM) practitioners, weather forecasters, communication and data specialists, and hydrologists from the Auckland (n = 4) and Southland (n = 5) Regions. The third workshop involved a portion of the NZ risk and hazard science community based at GNS Science (n = 11). Thus, 59 people participated in this research.

Interview questions and workshop activities focused on themes regarding IFW data needs and sources. We asked for participants' general thoughts on IFWs; what impact, vulnerability, and/or exposure data they use or need, why, and how; the life path of the data; experienced and/or perceived challenges obtaining data required for IFWs and other uses; and thoughts on collecting and using alternative data sources (e.g., social media and crowdsourcing). Findings around data governance and acquisition are reported in this paper.

Progress in Disaster Science 13 (2022) 100213

This research was collected under a 'low risk' ethics notification with the Massey University Human Ethics Committee. All interviewees remain anonymous and are assigned an alphabetic code (A, B, C, etc.), being identified only by the area of expertise and/or practice, industry, location, or governance level (Table 1). Interviews were audio-recorded and transcribed verbatim, and qualitative analysis (including coding and memowriting) followed the axial coding paradigm according to the ES-GTM [13], using NVivo 12 qualitative analysis software [8]. Per ES-GT, analysis of the interviews, workshops, and documents

Per ES-GT, analysis of the interviews, workshops, and documents followed the open coding (assigning concepts and categories to an instance of the data by line or by word), axial coding (relating categories to each other with optional guidance from the coding paradigm), and selective coding (relating all categories to a core category) stages [77]. The coding paradigm introduced by Strauss and Corbin [77] supported the axial coding stage whereby the codes created from the open coding stage were related to the coding paradigm dimensions (Table 2) for increased density and precision.

Additional techniques were used to support the ES-GT analysis, including regular memo-writing, diagramming, and constant comparison [29]. When the lead author noticed common themes in the interview and workshop data, a memo was written to identify the theme and discuss its relation to other concepts or themes (i.e. constant comparison) [13,29]. Diagramming was used to draw out linkages or relationships between the emerging themes. This was an iterative process that occurred during the data collection and analysis.

From the axial coding process (summarised in Table 2), two phenomena related to data management and acquisition of hazard, impact, vulnerability, and exposure data were identified. Phenomena are the subjects or objects under study [77]. Impact forecasts and warnings and HIVE data were identified as the overarching phenomena being studied in this research. As data collection and analysis progressed, the two phenomena that became the focal point of this manuscript were identified from the the themes that emerged in the interviews, memo-writing, diagramming, and constant comparison techniques previously described. These two phenomena are: (1) The Roles and Responsibilities of Data Custodianship, and (2) Data Access and Sharing. These are discussed next.

3. Findings and discussion

The two phenomena (The Roles and Responsibilities of Data Custodianship and Data Access and Sharing) were succinctly summarised by an NZbased risk modeller, who said:

I think it really comes down to ... sharing and collaboration ... and I think there are good efforts, but ... often things get snagged in ... a privacy or confidentiality or legal issues with the datasets, who owns them, who maintains them, how can you rely on them ... And the amount of work that is required to produce and maintain a reliable dataset is massive, it's just so much work that people don't want to do it. It costs money, it costs time, and ... being a custodian of a dataset is not really an enviable position, necessarily. I think that's probably one of the biggest barriers (Risk Modelling NZ. A).

These two phenomena (the roles and responsibilities of data custodianship, and data access and sharing) will each be discussed in turn.

3.1. The roles and responsibilities of data custodianship

Amongst participating agencies and countries, it remains unclear as to who is responsible for maintaining datasets for IFWs:

... it's a little bit about our traditional remits just being on the hazards information and so insurance companies and other areas are associated with damage and loss. And now ... the challenge is accessibility to the data. So, we don't necessarily need to or want to become custodians of new data sets, we're just keen, if someone's got it organised and has it, to be able to ... bring it together with our hazard information and our

Progress in Disaster Science 13 (2022) 100213

Table 1

nterviewee codes.				
Interview Code	Position	Classification	Location	Government Level
Agriculture/Rural NZ. A	Agriculture policy coordinator	Agriculture/Rural	NZ	National
Data Management Gov. NZ. Nat. A	Senior Resilience Advisor	Data Management	NZ	National
Data Management Private NZ. B	Geospatial Specialist	Data Management	NZ	Private
Data Management Research NZ. C	GIS Specialist	Data Management	NZ	Private
Data Management Private NZ. D	GIS Specialist	Data Management	NZ	Private
Data Management Gov. NZ. Nat. E	Head of Data	Data Management; Governance	NZ	National
EM. NZ. Reg. A	Director	Emergency Management	NZ	Regional
EM. NZ. Reg. B	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. C	Controller	Emergency Management	NZ	Regional
EM. NZ. Reg. D	Principal Science Advisor	Emergency Management	NZ	Regional
EM. NZ. Reg. E	Principal Advisor Strategy and Partnerships	Emergency Management	NZ	Regional
EM. NZ. Reg. F	GIS Lead	Emergency Management; Data Management	NZ	Regional
EM. Gov. NZ. Nat. G	Senior Hazard Risk Management Advisor	Emergency Management; Governance	NZ	National
EM, NZ, Reg. H	Emergency Management Advisor	Emergency Management	NZ	Regional
EM. NZ. Nat. I	First Responder	Emergency Management	NZ	National
EM. Gov. NZ. Nat. J	National Operations Manager	Emergency Management: Governance	NZ	National
EM. NZ. Reg. K	Regional Manager	Emergency Management	NZ	Regional
EM. NZ. Reg. L	Emergency Management Advisor	Emergency Management	NZ	Regional
EM, NZ, Reg. M	Group Controller	Emergency Management	NZ	Regional
Health NZ, Reg. A	Respiratory Doctor	Public Health	NZ	Regional
Hyd. Gov. NZ. Reg. A	Flood EWS Programme manager	Hydrology; Governance	NZ	Regional
Interview Code	Position	Classification	Location	Government Level
Lifelines NZ. Reg. A	Civil Engineer	Lifelines	NZ	Regional
Loss Modelling Research NZ. A	Economist	Loss Modelling; Research	NZ	Private
Met. Int. A	Science Manager	Meteorology	International	National
Met. Int. B	National Manager Disaster Mitigation Policy	Meteorology	International	National
Met. Int. C	Senior Policy Officer	Meteorology	International	National
Met. Int. D	Senior Social Scientist	Meteorology	International	National
Met. Int. E	Consultant Meteorologist	Meteorology	International	National
Met. NZ. F	Senior Meteorologist	Meteorology	NZ	National
Met. NZ. G	Communications	Meteorology	NZ	National
Met. NZ. H	Public Relations	Meteorology	NZ	National
Met. Int. I	Division Chief/Meteorologist	Meteorology	International	National
Met. Research NZ. J	Meteorologist	Meteorology; Research	NZ	National
Met. NZ. K	Senior Meteorologist	Meteorology	NZ	National
Met. Private NZ. L	Head Weather Analyst	Meteorology	NZ	Private
Risk Modelling NZ. A	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ, B	Risk Modeller	Risk Modelling	NZ	National
Risk Modelling NZ, C	Risk Modeller	Risk Modelling	NZ	National
Rick Modelling NZ D	Risk Modeller	Rick Modelling	NZ	National

forecasting capability (Met. Int. B).

The responsibility and cost of collecting and storing HIVE data were concerns for participants. Many of the agencies' remits do not include data custodianship, and it would be a costly undertaking, with the uncertainty of maintaining funding (Met. Int. B; Met. NZ. F).

While the literature has pointed to Emergency Management (EM) agencies for collecting impact information (e.g., [44,70]), we found that the participating NZ-based EM agencies indeed collect impact information, but

Table 2

Coding paradigm summary(from Harrison et al., In Press).

Coding Paradigm Dimension	Description
Causal Conditions	A set of events that influence the phenomena or result in the appearance or development of a phenomena [77,82].
Phenomena	The subject or object under study [77].
Contextual	The specific set of conditions and characteristics surrounding
Conditions	the phenomena and resulting in action/interaction strategies taken to address the phenomena [77,82].
Intervening Conditions	Unexpected events or factors leading to action/interaction strategies (e.g., time, space, culture, socioeconomic status, technological status, history) [77,82].
Action/Interaction Strategies	Purposeful and deliberate acts taken to address the phenomena [77,82].
Consequences	Predictable or unpredictable, intended or unintended outcomes of the action/interaction strategies [77,82].

they often do not systematically collect it or store it (EM. NZ. Reg. A, B). These findings corroborate those of Crawford et al. [14], who found that EM agencies and councils in Aotearoa-New Zealand were not clear on who was responsible for collecting risk data. For example, regarding the NZ National Loss Database under development by the National Emergency Management Agency in Aotearoa-New Zealand (NEMA), our participant indicated that NEMA is not a designated data custodian in NZ, and as such, they have been grappling with learning proper data management protocols (EM. Gov. NZ. Nat. G). This raises questions around who is most suitable for managing and acting as the steward and/or custodian of the data.

Several data custodians were identified in passing during our interviews, which align with some that have been identified by LINZ [49], such as LINZ, the NZTA, and local councils. For example, a risk modeller who was involved in collecting building asset information for the 2011 Canterbury earthquake recovery described how maintenance responsibilities for these data were transferred to LINZ due to privacy concerns:

because there's some personal information in the database ... that we didn't pay attention [to] previously ... but to protect us from now on, we say 'okay this is the data we give [to] LINZ, you guys take the data ... and you ... decide if you want to share the data or not' (Risk Modelling NZ. B).

Similarly, for the development and management of the NZ National Loss Database, NEMA has turned to Statistics New Zealand (typically referred to as Stats NZ, New Zealand's national statistics agency) for guidance on proper data management protocols:

Stats NZ ... have helped us think about this, but it's their business. So, for example, we talked about data management, and they talked about, 'okay if you're holding that data then you need to run an integrity test, that your data hasn't become corrupted, and you do that every [so of-ten]. And you've got enough backups' and all that ... but we haven't got an explicit data management policy around ensuring it's not corrupt, its integrity ... change controls, and all that tracking; that's not what we do (EM. Gov. NZ. Nat. G).

In addition to being the national statistics agency for Aotearoa-New Zealand, Stats NZ also became the lead agency for government-held data in 2017 [75]. In this leadership role Stats NZ acts as a facilitator to support government agencies in building their capabilities and data management practices [75]. Thus, their help in guiding NEMA towards proper data management practices for the National Loss Database aligns with their role as the lead for government data. The Stats NZ website¹ provides further guidance on principles for safe and effective use of data and analytics, data stewardship, data standards, open data, etc.

Several of the HIVE datasets that were identified by Harrison et al. [34] have been classed as fundamental geospatial datasets by LINZ and have an appointed, proposed, or suggested data custodian and steward. Table 3 presents the Suggested (S), Proposed (P), or Appointed (A) Stewards and Custodians of the HIVE data sets identified by Harrison et al. [34] from our interviews as designated by LINZ [49]. The results in Table 3 are based on an analysis of two LINZ documents regarding data custodianship (see [49,50]).

The results in Table 3 show that stewardship/custodianship for most of the underlying datasets for hazards, vulnerability, and exposure have been identified by LINZ and are either Suggested (S), Proposed (P), or Appointed (A), while there is a clear a gap in the stewardship/custodianship of impact data, as shown by the number of 'N/A' entries for these datasets. This, in conjunction with our interviews, shows the need for establishing data management protocols and practices for impact data, including identifying potential data stewards and custodians for these data, such that the data can be accessed by and shared with other relevant users.

3.2. Data access and sharing

Data access and sharing was the next phenomenon identified in our data analysis. The ES-GT coding paradigm was applied to understand the causal and intervening conditions, and action/interaction strategies for data access and sharing, as shown in Fig. 1. As discussed by Harrison et al. [341],² disaster/emergency events, technological advancements, and research were identified as the causal conditions to data collection and data access and sharing. For example, sharing data was required for the response to the 2019 Pigeon Valley fires in Nelson, NZ, yet organisational and technical issues impeded the data sharing process (Data Management Gov. NZ. Nat. A). However, as shown in Fig. 1, intervening conditions can inhibit data sharing and access while several action/interaction strategies have been identified as capable of addressing these intervening conditions. These conditions and strategies will be discussed next.

3.2.1. Organisational aspects of data access and sharing

Trust and distrust, and privacy and security were identified as organisational intervening conditions affecting data sharing. Building partnerships was identified as an organisational action/interaction strategy to address these intervening conditions. These will be discussed next.

3.2.1.1. Trust and distrust as intervening conditions. Trust is a key element to data use and access. While Harrison et al. [34] identified the importance

¹ https://www.stats.govt.nz/about-us/data-leadership. Accessed 20 December 2021.
² The results of Harrison et al. [34] are presented in a similar fashion using the ES-GTM Coding Paradigm to analyse the causal conditions, intervening conditions, and action/interaction strategies pertaining to the Data Collection phenomenon. As such, a similar figure is presented here, but pertaining instead to the Data Access and Sharing phenomenon. Progress in Disaster Science 13 (2022) 100213

of trust in the data for IFWs, here we found the need for trust between agencies to be an important condition for data access and sharing, as summarised by an NZ risk and data specialist:

There's a lot of suspicion between institutions in New Zealand as to what people are doing things for and why and with this comes patch protection. This is an important point and that it inhibits risk awareness, collaboration and our risk management as a country (Data Management Private NZ, D).

The management of the NZ Flood Pics³ crowdsourcing platform and access to its resulting data is an example of an initiative established to avoid suspicion and lack of institutional trust, according to one of our interviewees (Data Management Private NZ. D). They described that NZ Flood Pics was developed independently from any institutions, and there is an aversion to tying the platform to any single institution even though that would make available the resources needed to sustain the platform. This aversion stems from the need for the data to remain open and separate from any "ulterior motives" (Data Management Private NZ. D). They indicated that institutionalising the platform may give the impression that there is an agenda with the use of the data, which could make people wary to contribute to it. At the time of conducting this interview, the costs and resources needed to sustain the platform remained such that the participant was seeking collaborating support and buy-in from multiple sectors with the requirement of keeping the data openly accessible. An update has since been provided by this participant that NZ Flood Pics is now in the process of being institutionalised with the National Institute for Water and Atmospheric Research (NIWA; a Crown Research Institute in New Zealand) with an agreed set of principles for how the data will remain open and accessible for "anybody to add value" (Data Management Private NZ. D).

Institutional involvement and open access to and sharing of citizen science data are growing areas of interest in citizen science research. The type of organisation involved in citizen science projects appears to influence people's willingness to participate and contribute their data [3]. For example, Martin et al. [51] found that contributors to a marine citizen science project in Australia showed a very high willingness to share data with research organisations, but less so with private research companies or consultants. Additionally, contributors seem to care about how the data they contribute is shared: Ganzevoort et al. [27] found that surveyed citizen scientists do no support unconditional data sharing, rather their acceptability of sharing the data with third parties depends on the goals of the data user. Moreover, Groom et al. [31] argue that the motivations of citizen scientists to contribute their data should align with the accessibility of the data for other uses. Thus, it is important to invest in data policies and transparency efforts to protect the interests of the contributors and ensure their continued engagement [3].

Participants indicated that some agencies who collect impact data were also found to be averse to sharing the data due to its sensitive nature and distrust in how such data could be used (Met. Int. D; Risk Modelling NZ. A; Loss Modelling Research NZ. A). As one loss modelling participant from Aotearoa-New Zealand outlined, not all agencies are willing to publicly share their data, such as a public insurance group that formed after the 2010–2011 Canterbury earthquake sequence (Loss Modelling Research NZ. A). As such, there is no guarantee that government/public organisations will want to, or can, make any of their data publicly available (Loss Modelling Research NZ. A).

This could be due to the sensitive nature of the data (Met. Int. D; Risk Modelling NZ. A). Certainly, in Aotearoa-New Zealand there are concerns around how releasing impact information could influence property values (Risk Modelling NZ. A). In Argentina, our participant described how a provincial government would not share their impact data with the National Meteorological Service (NMS) because of the political nature of the data:

³ www.nzfloodpics.co.nz. Accessed 20 December 2021

Progress in Disaster Science 13 (2022) 100213

Table 3

Table 3 Suggested (S), Proposed (P), or Appointed (A) Stewards and Custodians of the HIVE data sets identified by Harrison et al. [34] from our interviews as designated by LINZ [49]. N/A is used for datasets that were not proposed as fundamental datasets (note that any person or agency can propose a fundamental geospatial dataset [49]), and thus a stew-ard and custodian was not identified. The number of N/A entries demonstrates a lack of direction for managing impact data.

	Dataset	Status	Stewa	ard	Leadership Custodian	Delivery Custodian
Hazard Data	Weather stations (rain gauges, anemometers, etc.)	N/A	N/A		N/A	N/A
	Radar data	N/A	N/A		N/A	N/A
	Satellite imagery & observations	Fundamental, Not evaluated	LINZ	(A)	LINZ (S), NZDF (S), TAs (S), Police (S)	LINZ, NZDF, TAs, Police, MfE, Landcare (S)
	River height and flow gauges	Not evaluated	Stewa	ard Committee (S) or MPI	MfE (S)	RCs (S)
	Float gauges/Sea level data	N/A	N/A	iii: (0)	N/A	N/A
	River networks	Fundamental	Stews	ard Committee (S) or MPI	MfE (S) LINZ (S)	NIWA (S)
	River networks	Fundamentar	(S), N	AfE (S)	MIE (3), LIVE (3)	NIWA (3)
	Vertical Rain Radar	N/A	N/A	10	N/A	N/A
	Regionwide floodplains	Not evaluated	(S), N	ard Committee (S) or MPI AfE (S)	MfE (S), LINZ (S)	NIWA (S), GNS (S), RCs (S), TAs (S)
	Overland flow paths	N/A	N/A		N/A	N/A
	Coastal inundation Maps/Hazards	Not evaluated	Stewa (S), D	ard Committee (S) or MPI loC (S)	MfE (S)	RCs (S), GNS (S)
	Pollen Counts	N/A	N/A		N/A	N/A
	Slope	N/A	N/A		N/A	N/A
	Camera feeds	N/A	N/A		N/A	N/A
	Social media (Tweets, Facebook post comments)	N/A	N/A		N/A	N/A
	Crowdsourcing (e.g., volunteer rain gauges, NZ Flood Pics)	N/A	N/A		N/A	N/A
	Dataset	Status	Stews	ard	Leadership Custodian	Delivery Custodian
Impact	Social media (Tweets, Facebook post	N/A	N/A		N/A	N/A
Data	comments. SnapChat)	14/11			10/11	
Data	Crowdsourced Photos via Story Maps (e.g., NZ Flood Pics)	N/A	N/A		N/A	N/A
	Crowdsourcing/Public Reporting	N/A	N/A		N/A	N/A
	Red Cross Chained Crowdsourcing	N/A	N/A		N/A	N/A
	Emergency call centre reports	N/A	N/A		N/A	N/A
	Community volunteer radio calls	N/A	N/A		N/A	N/A
	Damage surveys	N/A	N/A		N/A	N/A
	Media reports	N/A	N/A		N/A	N/A
	Tacit knowledge, experience, intuition	N/A	N/A		N/A	N/A
	Health & Injury Data	N/A	N/A		N/A	N/A
	Wellbeing Surveys	N/A	N/A		N/A	N/A
	Post-event interviews and surveys	N/A	N/A		N/A	N/A
	Insurance claims	N/A	N/A		N/A	N/A
	"Boots on the ground"	N/A	N/A		N/A	N/A
	Lifelines Sectors (e.g., power companies, NZTA)	N/A	N/A		N/A	N/A
	Situational Reports	N/A	N/A		N/A	N/A
	Post-event reports	N/A	N/A		N/A	N/A
	Operations Reports (Council)	N/A	N/A		N/A	N/A
	Flood event reporting	N/A	N/A		N/A	N/A
	Injuries and fatalities (e.g., cause of death)	N/A	N/A		N/A	N/A
	Cultural and heritage/historical impacts	N/A	N/A		N/A	N/A
	Impact Model outputs	N/A	N/A		N/A	N/A
	NZ National Loss Database	N/A	N/A		N/A	N/A
	Dataset	Statu	ıs S	Steward	Leadership Custodian	Delivery Custodian
Vulnerab Data	ility Vulnerability Assessment and Risk Mod Outputs	elling N/A	1	N/A	N/A	N/A
	Asset information - Building footprints	Fund	amental 1	LINZ (A)	MBIE (S)	TAs (S), NZFS (S), MBIE (S)
	Asset information Historically and Cul	turally Not		Staward Committee (S) con	sisting LINZ (S) LCNZ (S)	LINZ (C) TAS (C) Drivate

Significant Sites	evaluated	of		Sector (S)
		LINZ (S), LGNZ (S), Emergency		
		Services (S)		
Building damage assessments		N/A		
Census data - meshblocks	Fundamental	Stats NZ	Stats NZ	Stats NZ
Tacit knowledge, experience, intuition	N/A	N/A	N/A	N/A
Lab-based experiments	N/A	N/A	N/A	N/A
Health Data, New Zealand Health Survey	N/A	N/A	N/A	N/A
Soil/land stability	Fundamental	No Steward	LINZ (S), GNS (S),	LINZ (S), GNS (S), Landcare
			Landcare (S)	(S)
Infrastructure - NZ Road Network	Not	Ministry of Transport	NZTA (P)	NZTA (P)
	evaluated			
Infrastructure - Council roads	Not	Ministry of Transport	LGNZ (S)	TAs (S)
	evaluated			
Infrastructure - Water	Not	Steward Committee (S) or MPI (S),	MfE (S), LINZ (S)	NIWA (S), LINZ (S), RCs (S),
	evaluated	MfE (S)		TAs (S)

Progress in Disaster Science 13 (2022) 100213

	Dataset	Status	Steward	Leadership Custodian	Delivery Custodian
Exposure	Asset information - Building footprints	Fundamental	LINZ (A)	MBIE (S)	TAs (S), NZFS (S), MBIE (S)
Data	Asset information - Historically and Culturally	Not	Steward Committee (S) consisting	LINZ (S), LGNZ (S)	LINZ (S), TAs (S), Private Sector
	Significant Sites	evaluated	of LINZ (S), LGNZ (S), Emergency Services (S)		(S)
	Infrastructure - NZ Road Network	Not evaluated	Ministry of Transport	NZTA (P)	NZTA (P)
	Infrastructure - Council roads	Fundamental	Ministry of Transport	LGNZ (S)	TAs (S)
	Infrastructure - Power - Electricity	Not evaluated	No Steward	Transpower (S)	Transpower, Lines Companies
	Infrastructure - Utility Networks	Not evaluated	No Steward	MCDEM/NEMA (S)	Utility Companies (S)
	Infrastructure - Water	Not evaluated	Steward Committee (S) or MPI (S), MfE (S)	MfE (S), LINZ (S)	NIWA (S), LINZ (S), RCs (S), TAs (S)
	Census data - meshblocks	Fundamental	Stats NZ	Stats NZ	Stats NZ
	Population movement via cell phone data	N/A	N/A	N/A	N/A
	Tacit knowledge, experience, intuition	N/A	N/A	N/A	N/A
	Topographical data, e.g. digital elevation models	Not evaluated	No Steward	MfE (S), LINZ (S)	Landcare (S), LINZ (S)
	Land-use and planning zones	Not evaluated	No Steward	LGGA (S), LGNZ (S)	TAs (S), RCs (S)
	Land-use maps	Fundamental	No Steward	MfE (S)	MfE (S)
	Community events	N/A	N/A	N/A	N/A

The acronyms in Table 3 are as follows: Land Information New Zealand (LINZ), New Zealand Defence Force (NZDF), Territorial Authorities (TAs), Ministry for the Environment (ME), National Institute of Water and Atmospheric Research Limited (NIWA), Ministry of Primary Industries (MPI), Institute of Geological and Nuclear Sciences Limited (GNS Science), Regional Councils (RCs), Department of Conservation (DoC), Ministry of Business, Innovation and Employment (MBIE), Local Government New Zealand (LGNZ), Statistics New Zealand (Stats NZ), New Zealand Transport Agency (NZTA), National Emergency Management Agency (NEMA, formerly MCDEM), Local Government Geospatial Alliance (LGGA)

I was working with the province of [redacted] ... and we implement[ed] this [data collection] form as a final project, and when I [said] 'okay, I want the data' they [told] me 'okay, we will give you the infrastructure data but no data about deaths' ... And I was like 'hey, no. That's not fair.' Why? Be cause it's political data! Who live and who die during a storm: that's political data, because it has to do with vulnerability and people here [are] very vulnerable. That's why I say that impact data is political data (Met. Int. D).

This finding aligns with those from Harrison and Johnson [33], where Canadian emergency managers showed concern about displaying heavily damaged areas online via crowdsourcing platforms for emergency response. However, as shown in the above quote, the political nature by which the impact data in Argentina is perceived adds another layer beyond simply showing concern for the privacy and security of people and assets during an emergency. Viewing impact data as a threat to how government response agencies and their response capabilities are perceived indeed introduces a political element that inhibits the global movement for open and collaborative data sharing for improved DRR.

The consequences of restrictive data access and lack of information sharing for DRR can lead to disastrous consequences, as was seen in the USA following the landfall of Hurricane Katrina in 2005 [68]. Yet, our findings show that data creators and users remain reluctant to share critical



Fig. 1. Summary of findings relating to the causal conditions, intervening conditions, and action/interaction strategies relating to data access and sharing, identified using the ES-GTM coding paradigm. Blue boxes represent organisation aspects and green boxes represent technical aspects. A socio-technical lens was applied to the analysis to identify both organisational aspects to the intervening conditions and action/interaction strategies and the technical aspects. The following results are separated accordingly, and further partitioned into intervening conditions (i.e., Trust/Distrust, and Privacy and Security as organisational intervening conditions, as a technical intervening condition, and action/interaction strategies (i.e., Building Partnerships as an organisational action/interaction strategy, and Standardised Data Collection as technical action/interaction strategies). (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)

information. In addition to the concerns around the political nature of the data as described above, this aversion to sharing may be due to interagency competition over resources, influence, and autonomy [68], and/or loss of control over the datasets [6].

Trust is an essential factor in multi-agency collaboration and data sharing, especially for disaster response [21]. Our findings further illustrate the importance of trust to facilitate data access and sharing through interagency collaboration for DRR [42]. The importance and value of opening and sharing disaster-related data cannot be denied, and it is possible to establish open data access and sharing for DRR. For example, local, provincial, and national agencies, and regional and national academic and research institutions in Argentina worked together to establish an open data platform for flood impact reduction, "with a view to socializing knowledge for early alerts" ([15], p. 86). Thus, agencies must continue to build and nutrure trust between each other to facilitate data and information sharing.

Traditional media outlets may be one party not tied to the political influence and privacy and security concerns of sharing data, and play a major role in disseminating information during a disaster [30,59]. For example, in Aotearoa-New Zealand, the MetService usually has "a conversation with media" about their official warning message and provides examples of impacts along with the warning message to add meaning and context to the warning. The media then passes on this information to the public (Met. NZ, G, K).

Media outlets and reports can be a timely data stream for information during a disaster [59]. They can even provide new information that might not otherwise be picked up or observed elsewhere. For example,

there might be a report of ... a motorcyclist blown off a piece of road or something. But there might not be any weather observations for miles in any direction that support that. So, then you have to go and look for corroborating evidence somewhere else, and how much do you trust the source? (Met. NZ. K).

However, as this example suggests, while media reports can bring new information to light, the new information still needs to be verified, as it might not always be accurate or true. Media outlets are both heavily reliant and influential on perceptions of trust and distrust during disasters [57]. In another example, a regional EM agency in Aotearoa-NZ received crowdsourced reports on their public online story map of a van "stuck in the centre of the Hokitika River" (EM. NZ. Reg. F). The media picked up this report and publicised that "there's people trapped" (EM. NZ. Reg. F). The van was empty, and the EM agency had to re-upload the report themselves to include a note that the van was empty (EM. NZ. Reg. F).

The literature indicates that agencies elsewhere have reported difficulties with communicating with media outlets e [4,59]. In Aotearoa-New Zealand, the continuous conversations between the NZ MetService and the media described above exemplify a working relationship that ensures a unified message for informing people. In another example, the media provided "quite active and quite positive support" in the response and evacuation of a large-scale flood in Edgecumbe, Bay of Plenty by providing ample media coverage of the event and of the messages issued by the EM agency (EM. NZ. Reg. A). For media outlets and their resulting reports to be effectively used by their audiences, it is thus critical for them to build trusted relationships with those audiences, including members of the public, hydrometeorological services, and EM agencies. This could include developing a process or protocols for sharing data and information effectively.

3.2.1.2. Building partnerships as an action/interaction strategy. While NMSs are not typically responsible for collecting non-meteorological data, EM and flood management agencies, amongst others, collect or produce various types of impact data for their own purposes. Findings from the interviews, in support of existing literature (e.g., [36,84]), suggest that either formal or informal partnerships support data sharing for DRR. Continued collaboration to build and nurture partnerships has a positive relationship with trust [42], an important factor for sharing data. Under the NHP, previously describted in section 1.2.1 [36], the UK MetOffice can access useful

Progress in Disaster Science 13 (2022) 100213

datasets such as traffic count data from the transportation agency, for their impact models (Met. Int. A;).

In Aotearoa-New Zealand and other participating countries, no such formal partnership was found to exist that is comparable to the NHP in the UK. However, evidence of informal partnerships was found in Aotearoa-New Zealand and Australia. In both Aotearoa-New Zealand and Australia, NMS and EM officials frequently contact each other informally during an event to exchange more targeted forecasting information such as levels of uncertainty, and worst-case scenarios to help with planning responses (Met. Int. B; Met. NZ. F, K; EM. NZ. Reg. A, B, C, D, E, F; [35]). For example, as reported by Harrison et al. [35], the NZ MetService works together with local and regional EM groups and hydrologists to determine the most appropriate warning level. These informal partnerships are particularly useful for information-sharing and decision-making on-the-fly during an event.

A need remains for more formal partnerships to facilitate data sharing after response events (Risk Modelling NZ. A). However, progress in this space will be slow:

there's lots of talk about data sharing and I think there's a general attitude in New Zealand moving towards greater collaboration, greater sharing. [But] a sort of central repository for any of this stuff within the next decade is absolutely not going to happen, or if ever ... It's hard to get people to work together (Risk Modelling NZ. A).

The participant listed data ownership and proprietary licensing as key limitations to sharing data. Perhaps in the context of sharing HIVE data via a central repository as mentioned by the risk modeller (Risk Modelling NZ. A), progress will indeed be slow as this requires more in-depth understanding and legal groundwork to establish data ownership rights and protection of proprietary data (e.g., [11]; Risk Modelling NZ. A). However, like the informal partnerships between the MetService, EM groups, and council hydrologists to share hydrometeorological information for warnings, another informal partnership has been established to fill a gap in EM practice in Aotearoa-New Zealand, the NZ Geographic Information Systems for Emergency Management (NZGIS4EM) group.

The NZGIS4EM group formed in the mid-2010s to boost the use of geospatial tools, such as Geographic Information Systems (GIS) within the EM sector (EM. NZ. Reg. H; [34]). NZGIS4EM is a grassroots community of GIS specialists and EM practitioners in Aotearoa-New Zealand that work together to share data and tools during responses and to innovate the use of geospatial technologies for EM (EM. NZ. Reg. H; Data Management Private NZ. B; [34]). The community was formed from the leadership of one skilful individual who identified the need for geospatial innovation in the NZ EM sector, to enhance practices from relying on paper maps to using modern tools and technology (Data Management Private NZ. B).

The NZGIS4EM is credited with fostering a community of practice within the EM and GIS sectors, where relationships across agencies have been built and strengthened for a more coordinated disaster response effort (EM. NZ. Reg. H; Data Management Private NZ. B). In turn, this has driven innovation, knowledge sharing, and data sharing across Aotearoa-New Zealand (EM. NZ. Reg. H; Data Management Private NZ. B). Direct outcomes of this group's work are seen in the recent Director's Guideline for *Civil Defence Emergency Management Forous [DGL.22/20]* [60] and Technical Standard [TS05/20] National Impact Assessments Data Set and Dictionary [62], published by NEMA. In these documents, the use of GIS is heavily referenced, and NZGIS4EM aided in defining the standards presented in the *Technical Standard* [TS05/20] National Impact Assessments Data Set and Dictionary ([62]; Data Management Private NZ. B).

Our findings suggest that informal partnerships are developed to fulfil an immediate need, such as communicating hydrometeorological information for warning decision-making or building a community of practice for coordinated disaster response. This finding aligns with the ad hoc informal partnerships that formed in the response and recovery to the 2010–2011 Queensland floods, where such partnerships formed between public and private agencies for information sharing [5]. Furthermore, interorganisational networks facilitated by groups like the NZGIS4EM can be voluntary or mandated [43]; in

Appendix M: Chapter Eight Publication

S.E. Harrison et al

the case of the NZGIS4EM group, it is a voluntary effort. Success factors of such networks have been identified as effective communication, trust and social capital, and learning and adaptation [43].

Formal partnerships appear to be less common, and not as easy to establish, as indicated previously (Risk Modelling NZ. A), particularly for severe weather hazards. However, formal partnerships have been developed in Aotearoa-New Zealand in the form of science advisory panels for natural hazard perils, such as the New Zealand Volcanic Science Advisory Panel [61], and the New Zealand Tsunami Advisory Group. The New Zealand Volcanic Science Advisory Panel works with area-specific volcanic advisory groups to coordinate planning for volcanic activity [61]. These groups were formed from a specific need that was either identified from local/national or international events (EM. Gov. NZ. Nat. J), to better communicate science advice by bringing the expertise from various scientific agencies to-gether [20]. For example, the New Zealand Tsunami Advisory Group was established in response to the 2011 Töhoku and 2004 Indian Ocean earth quakes and tsunami (EM. Gov. NZ. Nat. J). These groups differ from the UK NHP in that practitioners are not formally included in them and some of the groups' roles in response and providing formal advice are unclear. Partnerships and multi-agency collaboration are not new concepts in the EM literature (e.g., [65]). It is promising to see these growing examples of successful partnerships both within and outside of Aotearoa-New Zealand, be it formal and informal. However, more clarity is needed to establish the roles of these NZ-based groups for response and formal advice.

Strategies for building and nurturing partnerships and collaboration include networking (EM. Gov. NZ. Nat. J), professional development (EM. NZ. Reg. K; [21]), cohabitation [36], and multi-disciplinary collaboration (Risk Modelling NZ. C; [28]), which may or may not be supported by funding and top-down mandates (EM. Gov. NZ. Nat. J; [67]). Participants indicated that a mix of both bottom-up and top-down approaches to building and nurturing partnerships is applicable in the Aotearoa-New Zealand context (EM. Gov. NZ. Nat. J). For example, cohabitation between agencies like the MetService, CDEM Groups, and local/regional councils can be organised from the bottom-up between the agencies in question. However, some political direction such as a national mandate might be more effective at inciting nation-wide cohabitation practices (EM. Gov. NZ. Nat. J). Further exploration of these strategies will be explored in future research.

3.2.1.3. Privacy and security as an intervening condition. When it comes to sharing data for emergency response purposes, our participant from Fire and Emergency New Zealand (FENZ) identified the importance of protecting the response data. Much of their response data comes from NZ Police, who have very structured and careful security systems protecting their information (EM. NZ. Nat. 1). As such, parties are working with the Privacy Commissioner to show that they can ensure the appropriate controls to protect the information, and in turn, there are also cultural challenges around the views of sharing information (EM. NZ. Nat. I). Our FENZ participant indicated that it is a matter of knowing staff members' behaviours and educating staff on why it is important to keep the data safe, while also safely sharing the information for effective response.

Safe and secure mechanisms for sharing the data are needed but are not always feasible. For example, our participant from Argentina indicated that they set up a Google Form to facilitate standardised data collection and sharing between the NMS and local government agencies, but they "have concerns ... because ... it is sensitive data [and] I don't trust ... sending information through Google" (Met. Int. D). However, Google Forms was the most easily accessible tool that the local government agencies could handle (Met. Int. D). This demonstrates the need to have controls in place to maintain privacy and ensure data security.

3.2.2. Technical aspects of data access and sharing

Challenges around data integration were identified as a technical intervening condition affecting data sharing. Systematic Data Collection and Standardised Data Collection were identified as technical action/interaction strategies to address this intervening condition. These will be discussed next.

Progress in Disaster Science 13 (2022) 100213

3.2.2.1. Data integration as an intervening condition. Sharing data is only successful when the data can be integrated seamlessly into existing systems and practices [56]. Findings from Harrison et al. [34] show that a wide variety of HIVE data for severe weather events exists. In many cases, the challenge lies not in identifying the data sources but in integrating all sources and types of data in a meaningful way (NMS. Int. A, E; NMS. NZ; EM. NZ Regional. A, B, E). This is summarised by an NZ risk modeller, "although we all complain about a lack of data, a lot of the time, there is quite a bit of data out there that could be used and isn't being used very well" (Risk Modelling NZ. A). For example, this participant identified struggles with integrating building data, which is stypically stored as spatial point data, with meshblock data, which is stored as spatial polygons. They stated that matching these point and polygon data together

can be the bigger barrier than the fact that the data does or doesn't exist. Because ... it turns out it's really not easy to put those points in a polygon because ... there's no basic way of doing that, it's complicated (Risk Modelling NZ. A).

Furthermore, our UK-based participant described how they would like o see

an integration of different approaches [like] the use of satellite, the use of social media, the use of citizen science, the use of media, and the integration of media to help understand better the differences and the causes for different types of impact (NMS. Int. A).

Some data integration methods have been successfully implemented. For example, the NZ MetService receive council rainfall data and share warning data files to integrate into hydrological models. In Aotearoa-New Zealand, the West Coast EM agency primarily works online so they can share information in near-real-time with other agencies. But this is not standard practice across the country; gaps still exist in developing formal systems for data sharing and integration. A regional EM official elaborated on their challenges of linking up and integrating various data sources:

I would like to see ... some research in understanding what are the tools and systems that can be used for this? Because on one hand ... if you sit around and everyone goes 'yeah, ... we must share data' and ... if we're going to plan for the future and everyone, different agencies hold different data about the same event, but how we're sharing that data? So, what is the system to do that? (EM. NZ Regional. A).

The Common Operating Picture (COP) is a solution currently under development in Aotearoa-New Zealand to allow cross-agency intelligence and data sharing during a response. The COP, a recommendation made following the 2010–2011 Canterbury earthquakes and other notable events (see [78]), is based on the idea of "everyone contributing to the scenario so that they've got to share, so that everyone's got access to that information" (EM. NZ. Reg. A); in other words, it is "a graphical visualisation of all the data available to make a decision" (Data Management Private NZ. B).

Phase one of the NZ COP programme began in late 2019 led by NEMA, when agencies involved identified the core national datasets needed for EM, and brainstormed how some of those datasets are shared, particularly geospatial datasets (EM. NZ. Reg. H). This required agencies to collaborate to identify different needs, highlighting the need for building trust and partnerships between agencies to facilitate data sharing.

Other solutions have been proposed and developed for improving data interoperability for integration. Horita et al. [38] developed a spatial decision support system (DSS) to integrate official and unofficial data for flood risk management in Brazil, showing that integration "provides more complete, accurate and updated information about the situation in the affected areas" ([38], p. 91).

Impact-based decision support systems (IDSS) offer a way for warning services to make effective decisions. An IDSS is "the provision of relevant information and interpretative services that enable partners to prepare for and respond, as planned, to extreme weather, water, and climate events

for the protection of lives and livelihoods" ([80], p. 1928). In a case study comparison of two historic severe winter weather events in the USA with and without a formal IDSS, the IDSS was found to enable quicker and more complete forecast updates, such that emergency managers could relay information to appropriate agencies to take mitigative actions [40]. The IDSS also allowed for improved crafting of public and partner messaging, and for preparing public officials to decide to shut down infrastructure.

Similarly, an integrated analysis of social vulnerability to extreme precipitation in Colorado was carried out by Wilhelmi and Morss [83] using Geographic Information Systems (GIS). The process involved integrating radar-derived rainfall data and watershed boundaries with national census data and historical impact data. Challenges were faced with the differences in spatial-temporal scales between meteorological and social datasets, as well as the various formats in which the meteorological data were available [83].

3.2.2.2. Systematic data collection as an action/interaction strategy. Systematic data collection is a need highlighted by many participants as an action/interaction strategy for better data sharing and integration. Non-systematic data collection occurs when data is collected in various formats and stored in different places (e.g., [52]). Thus, we refer to systematic data collection as having a system in place to process data from raw state to a usable format and storing the data in one place for easy access. Our participants indicated that systematic data collection would sup-

port evidence-based planning and decision-making for warnings and planning (Met. Int. A; Met. NZ. F; EM. NZ. Reg. B). For example, our participating UK-based NMS official highlighted the need for systematic impact, exposure, and vulnerability data collection to be able to understand the causes of impacts with empirical evidence and to be able to compare different impact models (Met. Int. A). In Aotearoa-New Zealand, a lack of systematic data collection severely impacted the organisation of intelligence and data gathered from event responses. For example, "back in 2011 [Canterbury earthquakes] ... the capture of the data was so unstructured" (Data Management Gov. NZ. Nat. A); photos of buildings and letterboxes were not associated with dates or addresses, nor were they associated with damage and impact assessment forms used to capture information about the person and/or building (Data Management Gov. NZ. Nat. A). Consequently, an individual was "employed ... for 18 months, just to take that photo and that building form and that welfare form and try to bring together a file on that family" (Data Management Gov. NZ. Nat. A). Had the data been collected in a more systematic way, "that information ... would have been more infor-mative at the time of the response and recovery" (Data Management Gov. NZ. Nat. A). Furthermore, systematic data collection would allow the sector to learn from past events and forecast potential future events by identifying impacts from events with similar meteorological signatures (Met. NZ. F; EM. NZ. Reg. B).

The COP previously mentioned provides a streamlined system for collecting, sharing, and managing impact data. However, much of this data does not have a shelf life beyond its initial purpose of building situational awareness for decision making during a response, because "from a Civil Defence perspective, whether it's local or regional, we don't need to keep the data, really. We'd probably keep it for two years and then say right, we're going to dispose of it because we don't need it" (EM. NZ. Reg. H). This also applies to welfare/wellbeing assessments, which are "very challenging, because there is a lot more private and personally identifiable information collected" (EM. NZ. Reg. H). As such, the EM agency will "probably ... be even more aggressive about deleting that information once the event is over, to protect the people's privacy" (EM. NZ. Reg. H).

3.2.2.3. Standardised data collection as an action/interaction strategy. Standardised collection of data was another need highlighted by many interviewees. Standardised data collection facilitates systematic data collection by providing a set of standards at which the raw data is collected such that it can be seamlessly integrated into a database with minimal processing [16,26]. Standardised data collection would ensure all data is collected consistently for rigour and comparability [16,26]. Our

Progress in Disaster Science 13 (2022) 100213

interviewees identified a need to be able to compare datasets and warnings across countries and regions, such as in Europe (Met. Int. E). Likewise, data collection practices within Aotearoa-New Zealand, Argentina, and Australia differ regionally, making it difficult for regions to share data. In Australia, EM agencies collect post-event impact assessments, but the Australia, RM agencies collect post-event impact assessments, but the Australian NMS official indicated that this data lacks necessary fields for forecasting purposes (Met. Int. E). In Argentina, the NMS is working with EM agencies to build a process for collecting standardised impact data using Google Forms to benefit both agencies; the EM agencies will have trustworthy data, while the meteorological service will have data to support IFW implementation (Met. Int. D).

Our investigation into Aotearoa-New Zealand practice showed that several types of standard data collection forms are available for different purposes, whereby the standards/forms were designed by different agencies for various uses. For example, in 2006, the Hawke's Bay Regional Council commissioned the development of *Templates for Consistent Hazard Event Reporting* (herein referred to as the *Template*) with a focus on capturing the impacts of the hazard for use by EM agencies for research, risk modelling, and developing a hazards and impacts database [22]. Standardised data collection provided from this template would better facilitate the organisation and storage of impact and hazard data for further use and analysis. However, it is unclear whether this form is used by EM agencies or councils across Aotearoa-New Zealand, as no participants identified it as a resource that they use, yet it has potential for development going forward.

NEMA published an Impact Assessments Director's Guideline for Civil Defence Emergency Management Groups outlining the preparation requirements for conducting effective and efficient rapid impact assessments by response agencies to enable a coordinated approach across multiple agencies (see [60]). This document identifies the agencies responsible for conducting im-pact assessments for various purposes and at various phases. For example, EM agencies or local authorities are responsible for planning rapid impact assessments, while FENZ is "likely to be one of the first responding agencies to conduct a rapid impact assessment" ([60], p. 15) due to their specialised apabilities for collecting and sharing field data [60]. NEMA also provided forms for completing an initial situation overview, initial damage assessment, and impact report form. These assessments support response planning and coordination (EM. NZ. Reg. H). Furthermore, a companion document called the Technical Standard [TS05/20] National Impact Assessments Data Set and Dictionary [62] provides more technical information to support the consistent collection and recording of impact asses sment data for easy cross-agency sharing and integration. Our participants identified welfare needs assessments as another form

Our participants identified welfare needs assessments as another form of impact assessment (EM. Gov. NZ. Nat. G; EM. NZ. Reg. H; Data Management Gov. NZ. Nat. A; Data Management Private NZ. B), focusing on "understanding the needs of people affected by an emergency" ([55], p. 1). Guidelines have been published outlining the welfare needs assessment process (see [55]). These guidelines indicate that data is collected using "customised [EM agency]/local authority forms" ([55], p. 11).

For the built environment, the Ministry for Business, Innovation and Employment (MBIE) published field guides, assessment forms, and other learning resources for rapid building damage assessments for flooding, earthquakes, and geotechnical hazards such as landslips (see [54]). These resources are an outcome of lessons learned from the 2010–2011 Canterbury earthquake sequence, the 2011 Nelson storm, and the 2016 Hurunui/Kaikoūra earthquake, and other international events [53]. These assessments are primarily used by councils via contracted building engineers to assess the building and land safety and usability immediately after an earthquake or flood (Risk Modelling NZ. B; EM. NZ. Reg. H; [54]), but are not suitable for impact/risk modelling because they are not designed for this purpose and are not sufficiently comprehensive (Risk Modelling NZ. B).

Alternatively, our risk modelling participant indicated that developing a "master building database" containing building attribute data pre- and post-disaster would be "ideal" (Risk Modelling NZ. B). The participant envisions the database holding building data for all purposes, populated before an event, and updated with damage information to the respective buildings

Appendix M: Chapter Eight Publication

S.E. Harrison et al

using a unique identifier post-disaster (Risk Modelling NZ. B). Efforts on this project are currently underway at GNS Science in Aotearoa-New Zealand (see [48]).

The above assessments can be completed on paper, as has traditionally been done. However, a recent technological overhaul of the NZ CDEM sector (Data Management Private NZ. B) has introduced new tools for streamlining the collection and integration of these datasets (EM. NZ. Reg. F; EM. NZ. Reg. H; Data Management Private NZ. B). Much of the system has become location-based (Data Management Private NZ. B) using GIS technology [60]. For example, the "welfare system [has] moved from being quite a database and table-based system to a location-based system" (Data Management Private NZ. B), and now utilises tools like Survey123⁴ for welfare needs assessments (EM. NZ. Reg. H). The West Coast example outlined by Stowell [76] provides more detail on how GIS has been used to streamline the CDEM impact assessment process

Outside of impact data and the CDEM sector, other agencies have developed their own applications for conducting standardised information. For example, the Real-time Individual Asset Attribute Collection Tool (RiACT) was developed by risk modellers to capture real-time, geolocated, standardised asset information, such as building attributes (Risk Modelling NZ. B; [47,48]). The goal of this tool is to support an exposure data development framework whereby exposure data is systematically collected and stored for improved access, management, and use (Risk Modelling NZ. B; [47,48]). While various standardised forms have been developed to capture haz-

ard and impact data for various purposes, it would be beneficial to continue the systematic process of integrating the resulting data into a data reposi-tory for ease of access, sharing, and use beyond its initial purpose, depending on licensing and proprietary restrictions (e.g., [11]). Global efforts in the hydrometeorological hazard space are underway for building data repositories. For example, the HIWeather Value Chain Project under the World Weather Research Programme (WWRP) is currently building a catalogue of hydrometeorological events and their impacts to evaluate the end-to-end warning value chain [37]. Additionally, the World Meteorological Organisation (WMO) has established an initiative for cataloguing hazards and events through its members [19].

Strategies for building an integrated and multi-disciplinary data collection approach rely, again, on building and nurturing partnerships and collaboration. Collaboration allows data collectors and users from various disciplines to jointly define and scope problems for which the data is being collected, and identify the data types that are of interest to the problem [28]. For example, Harrison et al. [35] identified the need for including social scientists when designing post-impact assessments for risk modelling such that the modelling can extend into social and cultural impacts. For the aforementioned HIWeather Value Chain Project, an international multi-disciplinary task team of social scientists, meteorologists, and risk scientists was formed to co-develop the data collection template for the event catalogue [37]. Consideration of emergent technology would also help with identifying efficient data collection and integration tools, such as GIS-based technology (e.g., Survey123), mobile devices, unmanned aerial vehicles (e.g., drones), etc. [46,85].

4. Conclusions and limitations

This exploratory study built further understanding around the manage ment, acquisition, and sharing of hazard, impact, vulnerability, and expo-sure (HIVE) data for severe weather hazards in New Zealand, towards supporting Sendai Framework priorities and implementing Impact Forecasts and Warnings (IFWs). While the qualitative nature of data collection and analysis herein limits the generalisability of results beyond the interviewees, this approach offers an in-depth understanding of a problem not readily available from quantitative approaches, which is appropriate for an exploratory study such as this [9,58,66].

We employed a socio-technical lens to our analysis to identify the organisational and technical intervening conditions and action/interaction strategies for accessing and sharing HIVE data in Aotearoa-New Zealand.

4 https://survey123.arcgis.com. Accessed 20 December 2021

Progress in Disaster Science 13 (2022) 100213

We found that there is a need for data governance of HIVE data. This involves identifying and appointing stewards and custodians of the relevant datasets such that the datasets can be maintained and available for further use beyond their initial purpose.

We also found a need for building and nurturing stronger partnerships to continue building trust between stakeholders for sharing data. Trust is an important factor for facilitating data sharing between agencies, and for people to share their data with hazards and impact monitoring crowdsourcing/citizen science projects.

Furthermore, integration challenges continue to interfere with the use of various sources of HIVE data for effective risk and impact assessments for IFWs and beyond. Systematic and standardised data collection approaches using GIS-based tools can support integration. Many templates for standardised data collection were found to exist in Aotearoa-New Zealand but the resulting data has not been systematically collected into one place for easier access and use. Depending on licensing and proprietary restrictions, it may be beneficial to aggregate these data into a central repository for continued use.

This research provides empirical evidence supporting the need for establishing roles and practices for governing HIVE data in Aotearoa-New Zealand. This is in support of both meeting the Sendai Framework priorities and implementing an IFW system for severe weather hazards in Aotearoa-New Zealand. Building partnerships remains key to improving data collec-tion, access, and sharing practices for DRR. Further research can investigate how these partnerships can be built and strengthened.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

This research was funded by New Zealand's Ministry of Business, Innovation & Employment via the GNS Science Strategic Science Investment Fund and Resilience to Nature's Challenges Kia manawaroa - Ngā Ākina o Te Ao Tūroa. The authors thank the participants of the research for providing their invaluable time and insights, and the New Zealand MetService -Te Ratonga Tirorangi for their insightful conversations and advice over the course of this research. Further thanks are given to the reviewers, whose feedback greatly improved the manuscript.

References

- [1] Al-Ruithe M, Benkhelifa E, Hameed K. A systematic literature review of data governar and cloud data governance. Person Ubiq Comput. 2018;23(5-6):839-59. https://doi. org/10.1007/s00779-017-1104-3.
- Albassan I, Sammon D, Daly M. Critical success factors for data governance: A theory building approach. Inform Syst Manag. 2019;36(2):98–110. https://doi.org/10.1080/ 10580530.2019.1589670. [2] Alh
- 10580530.2019.1589670.
 [3] Anhalt-Depies C, Stenglein JL, Zuckerberg B, Townsend PA, Rissman AR. Tradeoffs and tools for data quality, privacy, transparency, and trust in citizen science. Biol Conserv. 2019;238 https://doi.org/10.1016/j.biocon.2019.108195.
 [4] Anzur T. How to talk to the media: televised coverage of public health issues in a disaster. Prehosp Disaster Med. 2000;15(4):70–2. https://doi.org/10.1017/ S1049023X00025310.
- S1049023X00023310.
 Bajracharya B, Hastings P. Public-private partnerships in emergency and disaster man-agement: examples from the Queensland floods 2010-2011. Aust J Emerg Manag. 2015;30(4):30–6.

- 2015;30(4):30-6.
 [6] Barry E, Banrister F, Barriers to open data release: A view from the top. Inform Pol. 2014;19:129-52. https://doi.org/10.3233/IP140327.
 [7] Benfeldt O, Persson JS, Madsen S. Data governance as a collective action problem. Inform Syst Front. 2019;22(2):299-313. https://doi.org/10.1007/s10796-019-09923-2.
 [8] Bergin M. NVivo 8 and consistency in data analysis reflecting on the use of a qualitative data analysis program. Nurse Res. 2011;18(3):6-12. https://doi.org/10.7748/nr2011. 04.18.3.6.c9457.
 [9] Blumer H. Symbolic interactionism: perspective and method. 1st ed. University of California Press; 1969.
 [10] Botterell A. The common alerting protocol: an open standard for alerting, warning and notification. In: Van de Walle In B., Turvi M, editors. 3rd annual internation ISCRAM conference; 2006, May 2006 Newark, NJ, USA.

- [11] Clark N, Guiffault F. Seeing through the clouds: processes and challenges for sharing ospatial data for disaster management in Haiti. Int J Disaster Risk Reduct. 2018;28: i8–70. https://doi.org/10.1016/j.ijdrr.2018.02.019.
- ts.dis.537bd80d1037a2ffde67d66c604d2a78.
- rents.dis.537b48001037a2ffde67d66c604d2a78.
 [13] Corbin J., Strauss A. In: Kinght V, McDuffler V, Barrett L, Markanich M, editors. Basics of qualitative research: techniques and procedures for developing grounded theory. 4th ed. SAGE Publications Inc; 2015.
 [14] Crawford MH, Crowley K, Potter SH, Saunders WSA, Johnston DM. Risk modelling as a tool to support natural hazard risk management in New Zealand local government. Int J Disaster Risk Reduct. 2018;28(January):610–9. https://doi.org/10.1016/j.ijdrr.2018. 011
- [15] De Giusti MR, Villarreal GL, Nusch CJ, Pinto AV, Lira AJ. Open access and open data on natural disasters. IFLA J. 2016;43(1):81–8. https://doi.org/10.1177/0340035216

- Berolantani, and K. S. (2016;43(1):81-8. https://doi.org/10.1177/0340035216
 G78236.
 Groeve T. Guidance for recording and sharing disaster damage and loss data. Publications Office of the European Union; 2015.
 Department of the Prime Minister and Cabinet. Proactive release of government re-sponse to ministerial review: better responses to natural disasters and other emergencies in New Zealand; 2018. Wellington, New Zealand.
 Dizar, R. Ferrer M. Loss database architecture for disaster risk management (EUR 29063 EN). Publications Office of the European Union; 2018.
 Douris J., Sopaheluwakan A. WMO cataloging initiative; 2019. In. Geneva, Switzerland.
 Doyle EEH, McClure J, Paton D, Johnston DM. Uncertainty and decision making: volca-nic crisis scenarios. Int J Disaster Risk Reduct. 2014;10:75–101. https://doi.org/10. 1016/j.jidr.2014.07.006.
 Doyle EEH, Paton D. Decision-making: preventing miscommunication and creating shared meaning between stakeholders. Observing the volcano world; 2017.
- [14] Joyfe Lah, Fakor D. Declarof-making, proceeding, machine indication and creating shared meaning between stakeholders. Observing the volcano world; 2017. p. 549–70. https://doi.org/10.1007/11157_2016_31.
 [22] EMS Limited. Templates for consistent hazard event reporting. NEMA https://www. civildefence.govt.nz/assets/Uploads/publications/NZ-Hazard-Reporting-Template-V1-0-June06.doi: 2006.

- [22] EMS Limited. Templates for consistent hazard event reporting. NEMA https://www.civild6ence.govt.nl/assets/Uploads/publications/NZ-Hazard-Reporting-TemplateV1-0-June06.doc; 2006.
 [23] Fakhruddin B, Chu E, Li G. Next generation disaster data infrastructure A study report of the CODATA task group on linked open data for global disaster risk research; 2019.
 [24] Fakhruddin B, Murray V, Main R. Disaster loss data in monitoring the implementation of the Sendai fmaework. Int Sci Council Integr Res Disaster Risk. 2017:1-4. https://www.preventionweb.net/publications/view/S3050.
 [25] Fakhruddin B, Murray V, Main R. Disaster loss data in monitoring the implementation of the Sendai fmaework. Int Sci Council Integr Res Disaster Risk. 2017:1-4. https://www.preventionweb.net/publications/view/S3050.
 [26] Ferrer M, Do Ó, A., Poljanšek, K., & Vallés, A. C., Disaster damage and loss data for policies; (JRC110366). Publications Office of the European Union; 2018.
 [27] Ganzeroo UW, van den Born RJG, Halfman W, Turnhout S. Sharing biodiversity data: fungs://doi.org/10.1007/s10531-017-1391-z.
 [28] Ge YG, Zobel CW, Murray-Tuite P, Nateghi R, Wang H. Building an interdisciplinary team for disaster response research: a data-driven approach. Risk Anal. 2021;41(7): 1145-51. https://doi.org/10.11017/s131-3280.
 [29] Galaer B, Strauss A. The discovery of grounded theory: strategies for qualitative research. Aldine Publishing Company 1967.
 [30] Greenberg J, Scanlon TJ. Old media, new media, and the complex story of disasters. Oxford forescarch encyclopedia of natural hazard science; 2016 https://doi.org/10.1033/acrefore/978019339407.013.21.
 [31] Groon Q, Weatherdon L, Geijzandorffer IR, Cadotte M. Is citizen science an open science in the case of biodiversity observations? J Appl Ecol. 2017;54(2):612-7. https://doi.org/10.1111/j365.2664.12767.
 [32] Guha-Si, Johnson P. Challenges in the adoption of crisis crowdsourcing and social of three g data?': identifying data sources for hydrometeorological impact forecasts and warnings in Aotearoa New Zealand. Int J Disaster Risk Reduct. 2021. https://doi.org/10.1016/j.
- ijdrr.2021.102619. [35] Harrison SE, Potter SH, Prasanna R, Doyle EEH, Johnston DM. Identifying the data uses
- and gaps for severe weather impact forecasts and warnings. Weather Climat Soc. 2022. https://doi.org/10.1175/WCAS-D-21-0093.1. [36]
- nttps://doi.org/10.1175/WCAS-D-21.0093.1.
 Hemingway R, Gunawan O. The natural hazards partnership: a public-sector collabora-tion across the UK for natural hazard disaster risk reduction. Int J Disaster Risk Reduct. 2018;499–511. https://doi.org/10.1016/j.ijdrr.2017.11.014.
 HWeather: Value chain approaches to evaluate the end-to-end warning chain: Project plan. WMO; 2021. [37]
- [38] Horita FEA, Porto de Albuquerque J, Degrossi LC, Mendiondo EM, Ueyama J. Develop
- [38] Horita FGA, Porto de Albuquerque J, Degrossi LC, Mendlondo EM, Ueyama J. Development of a spatial decision support system for flood risk management in Brazil that combines volunteered geographic information with wireless sensor networks. Comput Geosci. 2015;80:84–94. https://doi.org/10.1016/j.cngeo.2015.04.001.
 [39] Horita FRA, Porto de Albuquerque J, Marchezini V, Mendiondo EM. Bridging the gap between decision-making and emerging big data sources: an application of a modebased framework to disaster management in Brazil. Decision Support Syst. 2017;97: 12–22. https://doi.org/10.1016/j.dss.2017.03.001.
 [40] Hosterman HR, Lazo JK, Sprague-Hilderbrand JM, Adkins JE. Using the National Weather Service's impact-based decision support services to prepare for extreme winter storms. J Emerg Manag. 2019;17(6):455–67. https://doi.org/10.5055/jem. 2019;0439.

Progress in Disaster Science 13 (2022) 100213

- [41] Kaltenberger R, Schaffhauser A, Staudinger M. "What the weather will do" results of a
- [41] Kallenberger K, Schaffnauser A, Stadudinger M. "What the weather Will do results or a survey on impact-oriented and impact-based warnings in European NMHSs. Adv Sci Res. 2020;17:29–38. https://doi.org/10.5194/asr-17-29-2020.
 [42] Kapucu N. Interagency communication networks during emergencies: boundary span-ners in multiagency coordination. Am Rev Public Adm. 2006;36(2):207–25. https:// doi.org/10.1177/0275074005280605.
- (ad. arg/10.117/1/2014/005280605). Kapucu N, Demiroz F. Interogranizational networks in disaster management. In: Jones EC, Faas AJ, editors. Social network analysis of disaster response, recovery, and adapta-tion. Butterworth-Heinemann; 2017. p. 25–39. https://doi.org/10.1016/B978-0-12-895106-200020. [43]
- 305196-2.00003-0

- tion. Butterworth-Heinemann; 2017. p. 25–39. https://doi.org/10.1016/B978-0-12-805166-2.00003-0.
 [44] Kox T, Lüder C, Gerhold L. Anticipation and response: emergency services in severe weather situations in Germany. Int J Disaster Risk Sci. 2018;9:116–28. https://doi.org/10.1007/s13753-018-0163-x.
 [45] Li G, Zhao J, Murray V, Song C, Zhang L. Gap analysis on open data interconnectivity for disaster risk research. Geo-spat Inform Sci. 2019;2:2(1):45–58. https://doi.org/10.1009/s020.2018.1560056.
 [46] Lin S-L, King A, Horspool N, Sadashiva V, Paulik R, Williams S. Development and application of the real-time individual asset attribute collection tool. Front Built Environ. 2019;5: https://doi.org/10.10389/buil.2019.00015.
 [47] Lin SL, King AB, Matcham I. Exposure data development framework: Real-time individual asset attribute collection tool. Front Built Environ. 2019;5: https://doi.org/10.3389/buil.2019.00015.
 [47] Lin SL, King AB, Matcham I. Exposure data development of pacific exposure dataset for use in catastropher risk assessment. Nat Hazards. 2020. https://doi.org/10.1007/s11069-020-04290-4.
 [48] Lin SL, Ware Zealand fundamental geospatial datasets and themes v2.2. Wellington, NZ: LINZ; 2014. Retrieved from. https://www.linz.govt.nz/system./files.force/media/file=attachments/Pundamental%20ataset%20and%20themes.pdf?download = 1. download = 1.
 [49] LINZ. Neurog and eutorian formework for Nav. Zaland fundamental approximation for Nav. Zaland fundamental application for Nav. Scienced and fundamental for Nav. Zaland fundamental application for Nav. Scienced and fundamental for Nav. Zaland fundamental application for Nav. Zaland fundamental application for Nav. Scienced and fundamental fundamen nload = 1.
- download = 1.
 [50] LINZ. Steward and custodian framework for New Zealand fundamental geospatial themes and datasets v2.3. Wellington, NZ: LINZ; 2014. Retrieved from. https:// www.linz.govt.nz/system/files_force/media/pages-attachments/Steward%20and %20custodian%20framework_0.pdf?download = 1&download = 1.
 [51] Martin VY, Christidis I., Ped GT. Public interest in marine citizen science: is there poten-tial for growth? BioScience. 2016;66(8):683–92. https://doi.org/10.1093/biosci/ https://doi.org/10.1093/biosci/
- biw070.
- [52] Massagrande F. Using GIS with non-systematic survey data: The Mediterranean evi-

- biv070.
 [52] Massagrande F. Using GIS with non-systematic survey data: The Mediterranean evidence: a European perspective. In: Lock GR, Stancic G, editors. Archaeology and geographic information systems: A European perspective. CRC Press; 1995. p. 55-66.
 [53] MBE. Field guide: rapid post disaster building usability assessment geotechnical. Wellington, NZ: New Zealand Government; 2018. Retrieved from. https://www.building.govt.nz/managing-buildings/post-emergency-building-assessment/building-usability-assessment-geotechnical.pdf.
 [54] MBE. Fapid building assessment resources. New Zealand Government; 2018. Retrieved 20 May 2021 from. https://www.building.assessment-resources.New Zealand Government; 2018. Retrieved 20 May 2021 from. https://www.building.assessment-resources/.
 [55] MCDEM. Part II section 7 needs Assessnent welfare services in an emergency director's guideline [DGI. 11/15]. (DGI 11/15). Wellington, NZ: New Zealand Government; 2015. Retrieved from. https://www.civildefence.govt.nz/masats/Welfare-Services-in-an-Emergency-10.1108/dpm-0.92019-0291.
 [56] Mgliorini M, Hagen JS, Mihaljevi J, Mysiak J, Rossi J-A, Siegmund A, et al. Data interoperability for disaster risk reduction in Europe. Disaster Prevent Manag. 2019;28(6): 804-16. https://doi.org/10.1016/dpm-0.92019-0291.
 [57] Miles M, Huberman AM. Qualitative data analysis: an expanded sourcebook. 2nd ed. Sage Publications, Inc. 1994.
 [59] Nair P. Role of media in disaster management. Mass Commun. 2010;4(1):36-40.
 [59] Nair P. Role of media in disaster management. Mass Commun. 2010;4(1):36-40.
 [59] Nair P. Role of media in disaster management. Mass Commun. 2010;4(1):36-40.
 [50] NEMA. Impact assessments Director's guideline for civil defence meergency management gave publications, Inc. 1994.
 [50] N. NeMA. Masets Director's guideline analysis: an expanded sourcebook. 2nd ed. Sage Publications, Inc. 1994.
 [50] N. NeW Zealand Volcanic Scie dictionary-technical-standard/. [61] NEMA. New Zealand Volcanic Science Advisory Panel (NZVSAP) terms of reference. Re-
- rieved from https://www.civildefence.govt.nz/assets/Uploads/publications/nzvsap tor-sep-2020.pdf; 2020. [62] NEMA. Technical standard [TS05/20] national impact assessments data set and
- dictionary. (TS05/20). Wellington, NZ: New Zealand Government; 2020. Retrieved from. https://www.civildefence.govt.nz/cdem-sector/guidelines/ impact-assessments-directors-guideline-and-national-impact-assessment-data-setnd-dictionary-technical-standard/.
- [63] OASIS. Common operating protocol version 1.2. http://docs.oasis-open.org/ emergency/cap/v1.2/CAP-v1.2-os.html; 2010.
- emergency/cap/v1.2/CAP-v1.2-co.s.ttml; 2010.
 [64] OECD. Enhancing access to and sharing of data: reconciling risks and benefits for data re-use across societies. OECD Publishing; 2019. https://doi.org/10.1787/276aaca8-en.
 [65] Owen C., Bearman C, Brooks B, Chapman J, Paton D, Hossain L. Developing a research framework for complex multi-team coordination in emergency management. Int J Emerg Manag. 2013;9(1) https://doi.org/10.1504/ijem.2013.05f098.
 [66] Patton MQ. Qualitative research and evaluation methods. 3rd ed. Sage Publications, Inc;
- 2002
- 2002.
 [67] Peek L, Tobin J, van de Lindt JW, Andrews A. Getting interdisciplinary teams into the field: institutional review board preapproval and multi-institution authorization agree-ments for rapid response disaster research. Risk Anal. 2021;41(7):1204–12. https:// doi.org/10.1111/risa.13740.
 [68] Peled A. When transparency and collaboration collide: the USA open data program. J Am Soc Inform Sci Technol. 2011;62(11):2085–94. https://doi.org/10.1002/asi.21622.

Appendix M: Chapter Eight Publication

S.E. Harrison et al

- [69] Plotkin D. Data stewardship: An actionable guide to effective data management and data
- governance, Academic Press; 2020. [70] Potter SH, Harrison SE, Kreft P. The benefits and challenges of implementing impact-Force orr, marson set, keer r the benefits and challenge of implementing implete based severe weather warning systems: perspectives of weather, flood, and emergency management personnel. Weather Climat Soc. 2021;13(2):303–14. https://doi.org/10. 1175/wcas-d-20-0110.1.
- [71] Robelo Moreira JL, Ferreira Pires L, Dockhorn Costa P, van Sinderen M. Semantic inter-operability of early warning systems: a systematic literature review. In: Zelm M, Jaekel F-W, Doumengts G, Wollschlager M, editors. Enterprise interoperability: Smart ser-vices and business impact of Enterprise interoperability. ISTE Ltd and John Wiley & Seer, Inc. 2019.

- F-W, Doumeingts G, Wollschlaeger M, editors. Enterprise interoperability: Smart services and business impact of Enterprise interoperability. ISTE Ltd and John Wiley & Sons, Inc; 2018.
 [72] Rosenbaum S. Data governance and stewardship: designing data stewardship entities and advancing data access. Health Serv Res. 2010;45(5 Pt 2):1442-55. https://doi.org/10.1111/j.1475-6773.2010.01140.x.
 [73] Secretary of Cabinet. Principles for managing data and information held by the New Zealand government, approved by cabinet on 8 august 2011 (CAB Min (11) 29/12 refers). Wellington, New Zealand: 2001. Government Retrieved from. https://www.data.govt.nz/asets/Uploads/cab-min-11-29-12-8-august-2011.pdf.
 [74] Sonnenwald DH, Scientific collaboration. In: Cronin B, editor. Annual review of information science and technology (ARUST). Information Today: 2007. p. 643-81. https://doi.org/10.1011/s.Jtats/2007.1440410121.
 [75] Stats K2. Data leadership/. StatsX2: 2019. Retrieved 7 July 2021 from. https://www.stats.govt.nz/about-us/datal-eadership/.
 [76] Stowell N. U Nest coast Hooding event. Retrieved 24 April 2021 from. https://storymaps.argis.com/stories/c3769/bfeScBat138b2120c215a57f6b; 2020.
 [77] Straus AL, Corbin JM. Bacis: of qualitative research: grounded theory procedures and techniques. Sage Publications; 1990.
 [78] Technical Advisory Group. Ministerial review better responses to natural disasters and other emergencies. Department of the Prime Minister and Cabinet; 2017. https://dpmc.govt.nz/sites/default/files/2018-01/ministerial-review-better-responses-natural-disaster-other-emergencies.pdf.

Progress in Disaster Science 13 (2022) 100213

- [79] Treadgold G, Gunn J, Morton P, Chambers S. Developing a regional approach and strategy for geographical information systems for emergency management ISCRAM Asia-Pacific, Wellington, New Zealand; 2018.
 [80] Uccellini LW, Ten Hoeve JE. Evolving the national weather service to build a weather-ready nation: connecting observations, forecasts, and warnings to decision-makers through impact-based decision support services. Bull Am Meteorol Soc. 2019: 1923–42. https://doi.org/10.1175/BAMS-D-18-0159.1. (October).
 [81] UNDRR. Sendai framework for disaster risk reduction 2015–2030. UNDRR; 2015. https://www.undrr.org/publication/sendai-framework-disaster-risk-reduction-2015-2030.
- 2030
- [82] Vollstedt M, Rezat S. An introduction to grounded theory with a special focus on axial coding and the coding paradigm. In: Kaiser G, Presmey R, editors. Compendium for early career researchers in mathematics education. Springer International Publishing; 2019. p. 81-100. https://doi.org/10.1001/9783-3030-15636-7.4.
 [83] Wilhelmi OV, Morss RE. Integrated analysis of societal vulnerability in an extreme precipitation event: a Fort Collins case study. Environ Sci Policy. 2013;26:49–62. https://doi.org/10.1016/j.envsci.2012.07.005.
 [84] WMO. WMO guidelines on multi-hazard impact-based forecast and warning services (0789263111500). https://www.mon.int/pages/prog/www./DFS/Meetings/ET-OWFPS_Montreal2016/documents/WMOGuidelinesonMulti-hazardImpact-based ForecastandWarningServices.pdf.2013.
 [85] Yu M, Yang C, Li Y. Big data in natural disaster management: a review. Geosciences. 2018;86() https://doi.07.10.3390/geosciences05051065.
 [86] Zaidi J. Data governance demystified lessons from the trenches. Dataversity. 2012. Retrieved 7 July 2021 from. http://www.kataversity.net/data-governance-demystified-lessons-from-the-trenches/.

- [88] WMO. WMO Guidelines on Multi-hazard Impact-based Forecast and Warning Services Part II: Putting Multi-hazard IBFWS into Practice (WMO-No. 1150). World Meteorological Organization; 2021. https://library.wmo.int/doc_num.php?explnum_ id=10936.

References

- 1 News. (2021, 19 June 2021). Auckland tornado 'very significant' for New Zealand meteorologist. *1 News*. https://1news.co.nz/one-news/new-zealand/aucklandtornado-very-significant-new-zealand-meteorologist
- Adam, R. J., Hilton, M. J., Jowett, T., & Stephenson, W. J. (2020). The magnitude and frequency of storm surge in southern New Zealand. New Zealand Journal of Marine and Freshwater Research, 55(2), 336-351. https://doi.org/10.1080/00288330.2020.1764596
- Ahmed, D., Borretti, C., Buan, S., Degrace, J.-N., Diamond, L., Feng, L., Golnaraghi, M., Goto, A., Habib, A., Harada, S., Hasegawa, N., Keeney, H., Jr., Mu, H., Ogawa, S., Puig, M. A., Torres, J. M. R., Shahidullah, M., Steinhorst, G., Tanaka, S., Sasagawa, Y., Tang, X., Vogelgesang, R., Washitake, N., & Zou, Y. (2012). *Institutional Partnerships in Multi-Hazard Early Warning Systems*. Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-642-25373-7
- Aitsi-Selmi, A., & Murray, V. (2016). Protecting the Health and Well-being of Populations from Disasters: Health and Health Care in The Sendai Framework for Disaster Risk Reduction 2015-2030. *Prehospital and Disaster Medicine*, *31*(1), 74-78. https://doi.org/10.1017/S1049023X15005531
- Aken, J. E. v. (2004). Management Research Based on the Paradigm of the Design Sciences: The Quest for Field Tested and Grounded Technological Rules. *Journal of Management Studies*, *41*(2), 0022-2380. https://doi.org/10.1088/1674-1056/18/4/060
- Al-Ruithe, M., Benkhelifa, E., & Hameed, K. (2018). A systematic literature review of data governance and cloud data governance. *Personal and Ubiquitous Computing*, 23(5-6), 839-859. https://doi.org/10.1007/s00779-017-1104-3
- Al Kilani, M., & Kobziev, V. (2016). An Overview of Research Methodology in Information System (IS). *OALib*, *03*(11), 1-9. https://doi.org/10.4236/oalib.1103126
- Alammar, F. (2018). Business Diplomacy in Practice: A Grounded Theory Study in Management Among Professional Diplomats [Doctor of Philosophy, Massey University]. Auckland, New Zealand.
- Aldridge, T., Gunawan, O., Moore, R. J., Cole, S. J., & Price, D. (2016). *A surface water flooding impact library for flood risk assessment* FLOODrisk 2016 - 3rd European Conference on Flood Risk Management, Lyon, France.
- Alexander, D. E. (2014). Social Media in Disaster Risk Reduction and Crisis Management. *Science and Engineering Ethics*, *20*, 717-733. https://doi.org/10.1007/s11948-013-9502-z
- Alhassan, I., Sammon, D., & Daly, M. (2019). Critical Success Factors for Data Governance: A Theory Building Approach. *Information Systems Management*, *36*(2), 98-110. https://doi.org/10.1080/10580530.2019.1589670
- Allaire, M. (2016). Disaster loss and social media: Can online information increase flood resilience? *Water Resources Research, 52*, 7408-7423. https://doi.org/10.1002/2014WR016618.Received
- Anderson-Berry, L., Achilles, T., Panchuk, S., Mackie, B., Canterford, S., Leck, A., & Bird,
 D. K. (2018). Sending a message: How significant events have influenced the warnings landscape in Australia. *International Journal of Disaster Risk Reduction*, 30(March), 5-17. https://doi.org/10.1016/j.ijdrr.2018.03.005
- Anhalt-Depies, C., Stenglein, J. L., Zuckerberg, B., Townsend, P. A., & Rissman, A. R. (2019). Tradeoffs and tools for data quality, privacy, transparency, and trust in

citizen science. *Biological Conservation, 238.* https://doi.org/10.1016/j.biocon.2019.108195

- Ansell, C., & Boin, A. (2017). Taming Deep Uncertainty: The Potential of Pragmatist Principles for Understanding and Improving Strategic Crisis Management. *Administration & Society, 51*(7), 1079-1112. https://doi.org/10.1177/0095399717747655
- Anzur, T. (2000). How to talk to the media: Televised coverage of public health issues in a disaster. *Prehospital and Disaster Medicine*, *15*(4), 70-72. https://doi.org/10.1017/S1049023X00025310
- Arapostathis, S. G. (2019, 2019). Instagrammers report about the deadly wildfires of East Attica, 2018, Greece: An introductory analytic assessment for disaster management purposes. Proceedings of the 16th ISCRAM Conference, València, Spain.
- Arias, P. A., Villegas, J. C., Machado, J., Serna, A. M., Vidal, L. M., Vieira, C., Cadavid, C. A., Vieira, S. C., Ángel, J. E., & Mejía, Ó. A. (2016). Reducing Social Vulnerability to Environmental Change: Building Trust through Social Collaboration on Environmental Monitoring. *Weather, Climate, and Society*, 8(1), 57-66. https://doi.org/10.1175/WCAS-D-15-0049.1
- Arksey, H., & O'Malley, L. (2005). Scoping studies: towards a methodological framework. *International Journal of Social Research Methodology*, 8(1), 19-32. https://doi.org/10.1080/1364557032000119616
- Assumpção, T. H., Popescu, I., Jonoski, A., & Solomatine, D. P. (2018). Citizen observations contributing to flood modelling: opportunities and challenges. *Hydrology and Earth System Sciences Discussions*, 22(July), 1473-1489. https://doi.org/10.5194/hess-2017-456
- Astrom, C., Ebi, K. L., Langner, J., & Forsberg, B. (2014). Developing a heatwave early warning system for Sweden: evaluating sensitivity of different epidemiological modelling approaches to forecast temperatures. *International Journal of Environmental Research and Public Health*, *12*(1), 254-267. https://doi.org/10.3390/ijerph120100254
- Auckland Emergency Management. (2016). *Auckland Civil Defence and Emergency Management Group Plan 2016-2021*.
- Ayeb-Karlsson, S., Kniveton, D., Cannon, T., van der Geest, K., Ahmed, I., Derrington,
 E. M., Florano, E., & Opondo, D. O. (2019). I will not go, I cannot go: cultural and social limitations of disaster preparedness in Asia, Africa, and Oceania. *Disasters*, 43(4), 752-770. https://doi.org/10.1111/disa.12404
- Bajracharya, B., & Hastings, P. (2015). Public-private partnerships in emergency and disaster management: examples from the Queensland floods 2010-2011. *Australian Journal of Emergency Management, 30*(4), 30-36.
- Baker, E. J. (1991). Hurricane Evacuation Behavior. *International Journal of Mass Emergencies and Disasters*, 9(2), 287-310.
- Bannister, T., Ebert, E. E., Williams, T., Douglas, P., Wain, A., Carroll, M., Silver, J., Newbigin, E., Lampugnani, E. R., Hughes, N., Looker, C., Mulvenna, V., Csutoros, D., Jones, P. J., Davies, J. M., Suphioglu, C., Beggs, P. J., Emmerson, K. M., Huete, A., & Nguyen, H. (2021). A Pilot Forecasting System for Epidemic Thunderstorm Asthma in Southeastern Australia. *Bulletin of the American Meteorological Society*, *102*(2), E399-E420. https://doi.org/10.1175/bams-d-19-0140.1
- Barry, E., & Bannister, F. (2014). Barriers to Open Data Release: A View from the Top. *Information Polity*, *19*, 129-152. https://doi.org/10.3233/IP-140327

- Basher, R. (2006). Global early warning systems for natural hazards: systematic and people-centred. *Philosophical Transactions of the Royal Society*, *364*(1845), 2167-2182. https://doi.org/10.1098/rsta.2006.1819
- Baskerville, R. (1999). Investigating Information Systems with Action Research. *Communications of the Association for Information Systems, 2*(19). https://doi.org/10.17705/1CAIS.00219
- Baskerville, R., & Myers, M. D. (2004). Special Issue on Action Research in Information Systems. *MIS Quarterly*, *28*(3), 329-335. https://doi.org/10.2307/25148642
- Baskerville, R., & Wood-Harper, A. T. (1998). Diversity in information systems action research methods. *European Journal of Information Systems*, *7*, 90-107. https://doi.org/10.1057/palgrave.ejis.3000298
- Bauer, P., Thorpe, A., & Brunet, G. (2015). The quiet revolution of numerical weather prediction. *Nature*, *525*, 47-55. https://doi.org/10.1038/nature14956
- Becker, J. S., Leonard, G. S., Potter, S. H., Coomer, M. A., Paton, D., Wright, K. C., & Johnston, D. M. (2017). Organisational Response to the 2007 Ruapehu Crater Lake Dam-Break Lahar in New Zealand: Use of Communication in Creating an Effective Response. In *Observing the Volcano World* (pp. 253-269). https://doi.org/10.1007/11157_2016_38
- Belanger, F. (2012). Theorizing in information systems research using focus groups. *Australasian Journal of Information Systems*, 17(2), 109-135. https://doi.org/10.3127/ajis.v17i2.695
- Bell, R., Lawrence, J., Allan, S., Blackett, P., & Stephens, S. (2017). *Coastal Hazards and Climate Change: Guidance for Local Government*. N. Z. Government.
- Bell, R., Paulik, R., & Wadhwa, S. (2016). Coastal Hazard and Climate-Change Risk Exposure in New Zealand: Comparing Regions and Urban Areas. FIG Working Week 2016 Recovery from Disaster, Christchurch, New Zealand.
- Benfeldt, O., Persson, J. S., & Madsen, S. (2019). Data Governance as a Collective Action Problem. *Information Systems Frontiers*, *22*(2), 299-313. https://doi.org/10.1007/s10796-019-09923-z
- Benton, J. (2019). Twitter is removing precise- location tagging on tweets a small win for privacy but a small loss for journalists and researchers. *NiemanLab*.
- Bergin, M. (2011). NVivo 8 and consistency in data analysis: reflecting on the use of a qualitative data analysis program. *Nurse researcher*, *18*(3), 6-12. https://doi.org/10.7748/nr2011.04.18.3.6.c8457
- Bhagavathula, S., Brundiers, K., Stauffacher, M., & Kay, B. (2021). Fostering collaboration in city governments' sustainability, emergency management and resilience work through competency-based capacity building. *International Journal of Disaster Risk Reduction, 63.* https://doi.org/10.1016/j.ijdrr.2021.102408
- Blumer, H. (1969). *Symbolic interactionism: perspective and method* (First ed.). University of California Press.
- Botterell, A. (2006, May 2006). The Common Alerting Protocol: An Open Standard for Alerting, Warning and Notification. In B. Van de Walle & M. Turof, 3rd Annual Internation ISCRAM Conference, Newark, NJ, USA.
- Boulton, C. A., Shotton, H., & Williams, H. T. P. (2016, 2016). Using Social Media to Detect and Locate Wildfires. 10th International AAAI Conference on Web and Social Media (ICWSM 16), W8: Social Web for Environmental and Ecological Monitoring, Cologne, Germany.
- Brabham, D. C. (2013). Using Crowdsourcing In Government. *IBM Center for The Business of Government*, 1-42.

- Breu, K., & Peppard, J. (2001). The Participatory Paradigm for Applied Information Systems Research. The 9th European Conference on Information Systems, Bled, Slovenia.
- Brown, G., & Kyttä, M. (2014). Key issues and research priorities for public participation GIS (PPGIS): A synthesis based on empirical research. *Applied Geography*, *46*, 122-136. https://doi.org/10.1016/j.apgeog.2013.11.004
- Bryant, A. (2009). Grounded Theory and Pragmatism: The Curious Case of Anselm Strauss. *Forum Qualitative Sozialforschung*, *10*(3), 1-26.
- Burgstall, A., Casanueva, A., Kotlarski, S., & Schwierz, C. (2019). Heat Warnings in Switzerland: Reassessing the Choice of the Current Heat Stress Index. *International Journal of Environmental Research and Public Health*, 16(15), 19-19. https://doi.org/10.3390/ijerph16152684
- Cabinet Office. (2008). *The Pitt Review: Learning lessons from the 2007 floods Full Report.* UK Retrieved from https://webarchive.nationalarchives.gov.uk/20100702215619/http://archive.c abinetoffice.gov.uk/pittreview/thepittreview/final_report.html
- Cameron, G., Pearce, H. G., Moore, J. R., & Bulman, L. S. (2007). *Report to the New Zealand Forest Owners Association on Aspects of Rural Fire Management in New Zealand's Plantation Forests* (Scion Client Report No. 9481). Scion. https://www.researchgate.net/publication/300055396_Report_to_the_New_Z ealand_Forest_Owners_Association_on_aspects_of_rural_fire_management_i n_New_Zealand's_plantation_forests
- Campbell, G. (2021, 29 June 2021). Wellington's south coast braces for 8m swells, residents evacuated. *NZ Herald*. https://www.nzherald.co.nz/nz/wellingtons-south-coast-braces-for-8m-swells-residents-evacuated/RDVRIE3ZRC2T2NSV7JGTTKZZ6M/
- Cardinal, D., Hayward, J., & Jones, G. (2004). *Epistemology: the theory of knowledge*. Trans-Atlantic Publications, Inc.
- Carley, K. M. K. M., Malik, M., Landwehr, P. M. P. M., Pfeffer, J., & Kowalchuck, M. (2016). Crowd sourcing disaster management: The complex nature of Twitter usage in Padang Indonesia. *Safety Science*, 90, 48-61. https://doi.org/10.1016/j.ssci.2016.04.002
- Casteel, M. (2016). Communicating Increased Risk: An Empirical Investigation of the National Weather Service's Impact-Based Warnings. *Weather, Climate, and Society, 8*(3), 219-232. https://doi.org/10.1175/WCAS-D-15-0044.1
- Casteel, M. (2018). An empirical assessment of impact based tornado warnings on shelter in place decisions. *International Journal of Disaster Risk Reduction*, *30*(September), 2212-4209. https://doi.org/10.1016/j.ijdrr.2018.01.036
- Centre for Public Health Research. (2014). *Drought, Climate Change and Health.* https://www.ehinz.ac.nz/assets/Factsheets/Released-2014/EHI-DroughtClimateChangeAndHealth1996-2013-released201407.pdf
- Cervone, G., Schnebele, E., Waters, N., Moccaldi, M., & Sicignano, R. (2017). Using Social Media and Satellite Data for Damage Assessment in Urban Areas During Emergencies. In P. Thakuriah, N. Tilahun, & M. Zellner (Eds.), *Seeing Cities Through Big Data* (pp. 443-457). Springer Geography. https://doi.org/10.1007/978-3-319-40902-3_24
- Chametzky, B. (2013). Generalizability and the Theory of Offsetting the Affective Filter. *The Grounded Theory Review*, *12*(2).
- Charmaz, K. (2006). Constructing Grounded Theory. SAGE Publications Ltd.
- Charmaz, K. (2014). *Constructing grounded theory* (Vol. 2). SAGE Publications.
- Chein, I., Cook, S. W., & Harding, J. (1948). The field of action research. *American Psychologist*, *3*(2), 43-50. https://doi.org/10.1037/h0053515

- Chen, J., Chen, H., Wu, Z., Hu, D., & Pan, J. Z. (2017). Forecasting smog-related health hazard based on social media and physical sensor. *Information Systems, 64*, 281-291. https://doi.org/10.1016/j.is.2016.03.011
- Chen, K., Bi, J., Chen, J., Chen, X., Huang, L., & Zhou, L. (2015). Influence of heat wave definitions to the added effect of heat waves on daily mortality in Nanjing, China. *Science of the Total Environment*, *506-507*, 18-25. https://doi.org/10.1016/j.scitotenv.2014.10.092
- Chen, S.-T., & Yu, P.-S. (2007). Real-time probabilistic forecasting of flood stages. *Journal of Hydrology*, *340*(1-2), 63-77. https://doi.org/10.1016/j.jhydrol.2007.04.008
- Ching, P. K., Carr de los Reyes, V., Sucaldito, M. N., & Tayag, E. (2015). An assessment of disaster-related mortality post-Haiyan in Tacloban City. *Western Pacific Surveillance and Response Journal, 6*, 34-38. https://doi.org/10.5365/WPSAR.2015.6.2.HYN_005
- Choi, C. E., Cui, Y., & Zhou, G. G. D. (2018). Utilizing crowdsourcing to enhance the mitigation and management of landslides. *Landslides*(January), 1-11. https://doi.org/10.1007/s10346-018-1034-9
- Chun Tie, Y., Birks, M., & Francis, K. (2019). Grounded theory research: A design framework for novice researchers. *SAGE Open Medicine*, *7*, 1-8. https://doi.org/10.1177/2050312118822927
- Cisneros, R., & Schweizer, D. W. (2018). The efficacy of news releases, news reports, and public nuisance complaints for determining smoke impacts to air quality from wildland fire. *Air Quality, Atmosphere and Health, 11*(4), 423-429. https://doi.org/10.1007/s11869-018-0545-9
- Clark, N., & Guiffault, F. (2018). Seeing through the clouds: Processes and challenges for sharing geospatial data for disaster management in Haiti. *International Journal of Disaster Risk Reduction*, 28, 258-270. https://doi.org/10.1016/j.ijdrr.2018.02.019
- Clarke, L., Blanchard, K., Maini, R., Radu, A., Eltinay, N., Zaidi, Z., & Murray, V. (2018). Knowing What We Know - Reflections on the Development of Technical Guidance for Loss Data for the Sendai Framework for Disaster Risk Reduction. *PLoS Currents*, *10*.

https://doi.org/10.1371/currents.dis.537bd80d1037a2ffde67d66c604d2a78

- Cole, S. J., Moore, R. J., Aldridge, T. A., Gunawan, O., Balmforth, H., Hunter, N., Mooney, J., Lee, D., Fenwick, K., Price, D., & Demeritt, D. (2016). Natural Hazards Partnership Surface Water Flooding Hazard Impact Model: Phase 2 Final Report. http://www.naturalhazardspartnership.org.uk/wpcontent/uploads/2016/10/NHP-HIM-Surface-Water-Flooding-Phase-2-Final-Report.pdf
- Cole, S. J., Moore, R. J., Wells, S. C., & Mattingley, P. S. (2016). *Real-time forecasts of flood hazard and impact: some UK experiences* FLOODrisk 2016 3rd European Conference on Flood Risk Management, Lyon, France.
- Cools, J., Vanderkimpen, P., El Afandi, G., Abdelkhalek, A., Fockedey, S., El Sammany,
 M., Abdallah, G., El Bihery, M., Bauwens, W., & Huygens, M. (2012). An early
 warning system for flash floods in hyper-arid Egypt. *Natural Hazards and Earth* System Science, 12(2), 443-457. https://doi.org/10.5194/nhess-12-443-2012
- Cooper, A. K., Coetzee, S., & Kourie, D. G. (2018). Volunteered geographical information, crowdsourcing, citizen science and neogeography are not the same. *Proceedings of the ICA*, *1*(August), 1-8. https://doi.org/10.5194/ica-proc-1-131-2018

References

- Corbin, J., & Strauss, A. (2008). *Basics of qualitative research: Techniques and procedures for developing grounded theory* (3rd ed.). SAGE Publications Inc. https://doi.org/10.4135/9781452230153
- Corbin, J., & Strauss, A. (2015). *Basics of Qualitative Research: Techniques and Procedures for Developing Grounded Theory* (V. Knight, Y. McDuffee, L. Barrett, & M. Markanich, Eds. Fourth ed.). SAGE Publications Inc.
- Covello, V., & Sandman, P. M. (2001). Risk communication: Evolution and Revolution. In *Solutions to an Environment in Peril* (pp. 164-178). John Hopkins University Press.
- Cowan, J. (2019). *Fire Investigation Report.* https://www.forestry.org.au/forestry/documents/Pigeon_Valley_Wildfire_Rep ort.pdf
- Cox, L. A., Jr. (2008). What's wrong with risk matrices? *Risk Analysis*, *28*(2), 497-512. https://doi.org/10.1111/j.1539-6924.2008.01030.x
- Crawford, M. H., Crowley, K., Potter, S. H., Saunders, W. S. A., & Johnston, D. M. (2018). Risk modelling as a tool to support natural hazard risk management in New Zealand local government. *International Journal of Disaster Risk Reduction*, *28*(January), 610-619. https://doi.org/10.1016/j.ijdrr.2018.01.011
- Creswell, J. W. (2003). *Research design: qualitative, quantitative, and mixed methods approaches* (2 ed.).
- Creswell, J. W. (2014). *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches* (Fourth ed.).
- Crompton, R. P., & McAneney, K. J. (2008). Normalised Australian insured losses from meteorological hazards: 1967-2006. *Environmental Science & Policy II*, 371-378. https://doi.org/10.1016/j.envsci.2008.01.005
- Crotty, M. (1998). *The Foundations of Social Research*. SAGE Publications.
- Cullen, S. M., Christensen, K., & Price, C. (2017). *Rangitāiki River Scheme Review April 2017 Flood Event*. https://www.boprc.govt.nz/media/773858/rangitaiki-river-scheme-review-april-2017-flood-event-final-report-as-supplied-to-by-of-plenty-regional-council.pdf

Current Results. (2021). Barrie Snowfall Totals & Snowstorm Averages. Retrieved 19 August 2021 from https://www.currentresults.com/Weather/Canada/Ontario/Places/barriesnowfall-totals-snowstorm-averages.php

- D'Amato, G., Vitale, C., D'Amato, M., Cecchi, L., Liccardi, G., Molino, A., Vatrella, A., Sanduzzi, A., Maesano, C., & Annesi-Maesano, I. (2016). Thunderstorm-related asthma: what happens and why. *Clin Exp Allergy*, *46*(3), 390-396. https://doi.org/10.1111/cea.12709
- Darke, P., Shanks, G., & Broadbent, M. (1998). Successfully completing case study research: combining rigour, relevance and pragmatism. *Information Systems Journal*, 8(4), 273-289. https://doi.org/10.1046/j.1365-2575.1998.00040.x
- Day, J. M., Junglas, I., & Silva, L. (2009). Information Flow Impediments in Disaster Relief Supply Chains. *Journal of the Association for Information Systems*, *10*(8), 637-660. https://doi.org/10.17705/1jais.00205
- De Giusti, M. R., Villarreal, G. L., Nusch, C. J., Pinto, A. V., & Lira, A. J. (2016). Open access and open data on natural disasters. *IFLA Journal*, *43*(1), 81-88. https://doi.org/10.1177/0340035216678236
- De Groeve, T. (2015). *Guidance for Recording and Sharing Disaster Damage and Loss Data*. Publications Office of the European Union.
- Deligne, N., Horspool, N., Canessa, S., Matcham, I., Williams, G. T., Wilson, G., & T.M., W. (2017). Evaluating the impacts of volcanic eruptions using RiskScape.

Journal of Applied Volcanology, 6(18), 21. https://doi.org/10.1186/s13617-017-0069-2

- Department of Building and Housing. (2012). *Technical investigation into the collapse of the Stadium Southland roof*. N. Z. Government. https://www.mbie.govt.nz/dmsdocument/100-southland-stadium-technicalinvestigation-pdf
- Department of Internal Affairs. (2012). *Local Government Act 2002 Amendment Act 2012*. Retrieved from http://www.legislation.govt.nz
- Department of the Prime Minister and Cabinet. (2018). *Proactive Release of Government Response to Ministerial Review: Better Responses to Natural Disasters and Other Emergencies in New Zealand*. Wellington, New Zealand
- Díaz, R., & Ferrer, M. (2018). *Loss database architecture for disaster risk management* (EUR 29063 EN). Publications Office of the European Union.
- Doocy, S., Daniels, A., Murray, S., & Kirsch, T. D. (2013). The Human Impact of Floods: A Historical Review of Events 1980-2009 and Systematic Literature Review. *PLoS Currents*(APR 2013).

https://doi.org/10.1371/currents.dis.f4deb457904936b07c09daa98ee8171a

- Doswell III, C. A. (2004). Weather Forecasting by Humans–Heuristics and Decision Making. *Weather and Forecasting*, *19*(6), 1115-1126. https://doi.org/10.1175/WAF-821.1
- Douris, J., & Sopaheluwakan, A. (2019). WMO Cataloging Initiative. In. Geneva, Switzerland.
- Doyle, E. E.-H., McClure, J., Paton, D., & Johnston, D. M. (2014). Uncertainty and decision making: Volcanic crisis scenarios. *International Journal of Disaster Risk Reduction*, *10*, 75-101. https://doi.org/10.1016/j.ijdrr.2014.07.006
- Doyle, E. E.-H., & Paton, D. (2017). Decision-Making: Preventing Miscommunication and Creating Shared Meaning Between Stakeholders. In *Observing the Volcano World* (pp. 549-570). https://doi.org/10.1007/11157_2016_31
- Doyle, E. E.-H., Paton, D., & Johnston, D. M. (2015). Enhancing scientific response in a crisis: evidence-based approaches from emergency management in New Zealand. *Journal of Applied Volcanology, 4*(1). https://doi.org/10.1186/s13617-014-0020-8
- Doyle, E. E. H. (2011). Science Advice for Decision-Making. In D. Paton & J. M. Violanti (Eds.), *Working in High Risk Environments: Developing Sustained Resilience* (Vol. III, pp. 69-92). Charles C. Thomas Publisher.
- DPMC. (2007). *National Hazardscape Report*. Wellington, New Zealand: Officials' Committee for Domestic and External Security Coordination
- Dubois, A., Zagheni, E., Garimella, K., & Weber, I. (2018). Studying Migrant Assimilation Through Facebook Interests. SocInfo 2018, LNCS 11186,
- Dudfield, M., Cameron, G., Carle, J., Ellem, K., & Hill, P. (2020). A comparison of 2019
 Pigeon Valley Forest fire with similar events in the past. New Zealand Journal of Forestry, 67(4), 3-10.
 https://www.forestry.org.au/forestry/documents/Pigeon_Valley_Wildfire_Rep ort.pdf
- Ehnis, C. F. (2018). Social Media within Emergency Management Organisations A case study exploring Social Media utilisation for Emergency and Disaster Management University of Sydney]. https://ses.library.usyd.edu.au/handle/2123/17938?show=full
- Eisenman, D. P., Cordasco, K. M., Asch, S., Golden, J. F., & Glik, D. (2007). Disaster Planning and Risk Communication With Vulnerable Communities: Lessons From Hurricane Katrina. *American Journal of Public Health*, *97*(S1). https://doi.org/10.2105/AJPH.2005.084335

- Elwood, S., Goodchild, M. F., & Sui, D. Z. (2012). Researching Volunteered Geographic Information: Spatial Data, Geographic Research, and New Social Practice. *Annals of the Association of American Geographers*, *102*(3), 571-590. https://doi.org/10.1080/00045608.2011.595657
- Emergency Management Southland. (2017). *Southland Civil Defence Emergency Management Group Plan 2017-2022*. (A313204). Southland, New Zealand: Emergency Management Southland, Retrieved from https://www.es.govt.nz/about-us/meetings?item=id:28zrwy7e21cxbynl4xbc
- Emergency Management Southland. (2020). *Southland Floods Response*. Retrieved 13 August 2021 from https://storymaps.arcgis.com/stories/2e31928535604ed7b1f242ac5e382874
- EMS Limited. (2006). *Templates for Consistent Hazard Event Reporting*. NEMA. https://www.civildefence.govt.nz/assets/Uploads/publications/NZ-Hazard-Reporting-Template-V1-0-June06.doc
- Enarson, E. (2007). Chapter 13: identifying and addressing social vulnerabilities. In W.
 L. Waugh & K. Tierney (Eds.), *Emergency management: principles and practices for local government* (2nd ed., pp. 257-278). ICMA Press.
- Environment Southland Regional Council. (2021). *Flood warning*. Retrieved 12 February from https://www.es.govt.nz/online-services/flood-warning
- Esri. (n.d.-a). *ArcGIS QuickCapture*. Retrieved 20 May 2021 from https://www.esri.com/en-us/arcgis/products/arcgis-quickcapture/overview
- Esri. (n.d.-b). *ArcGIS Survey123*. Retrieved 20 May 2021 from https://survey123.arcgis.com/
- Fakhruddin, B., Chu, E., & Li, G. (2019). *Next Generation Disaster Data Infrastructure A study report of the CODATA Task Group on Linked Open Data for Global Disaster Risk Research*.
- Fakhruddin, B., Murray, V., & Gouvea-Reis, F. (2019). Loss Data in Monitoring.
- Fakhruddin, B., Murray, V., & Maini, R. (2017). *Disaster loss data in monitoring the implementation of the Sendai Framework*. International Science Council & Integrated Research on Disaster Risk. https://www.preventionweb.net/publications/view/53050
- Falagas, M. E., Pitsouni, E. I., Malietzis, G. A., & Pappas, G. (2008). Comparison of PubMed, Scopus, Web of Science, and Google Scholar: strengths and weaknesses. *The FASEB Journal*, 22(2), 338-342. https://doi.org/10.1096/fj.07-9492LSF
- Farnell, C., Rigo, T., & Pineda, N. (2017). Lightning jump as a nowcast predictor: Application to severe weather events in Catalonia. *Atmospheric Research*, *183*, 130-141. https://doi.org/10.1016/j.atmosres.2016.08.021
- Fdez-Arroyabe, P., Lecha Estela, L., & Schimt, F. (2018). Digital divide, biometeorological data infrastructures and human vulnerability definition. *International Journal of Biometeorology*, 62(5), 733-740. https://doi.org/10.1007/s00484-017-1398-x
- Fearnley, C. J. (2013). Assigning a Volcano Alert Level: Negotiating Uncertainty, Risk, and Complexity in Decision-Making Processes. *Environment and Planning A: Economy and Space, 45*(8), 1891-1911. https://doi.org/10.1068/a4542
- Feilzer, M. Y. (2010). Doing mixed methods research pragmatically: Implications for the rediscovery of pragmatism as a research paradigm. *Journal of Mixed Methods Research*, 4, 6-16.
- Feinberg, J. (2009). *Threshold Definition of Early Warning Systems to Natural Hazards* Universitety of Oslo]. Oslo.

- Fernandez, W. D. (2005). The grounded theory method and case study data in IS research: issues and design. In D. N. Hart & S. D. Gregor (Eds.), *Information systems foundations* (pp. 43-59). https://doi.org/10.22459/isf.06.2005.05
- Ferrer, M., Do Ó, A., Poljanšek, K., & Vallés, A. C. (2018). *Disaster damage and loss data for policy: Pre- and post-event damage assessment and collection of data for evidence-based policies* (JRC110366). Publications Office of the European Union.
- Flage, R., & Askeland, T. (2020). Assumptions in quantitative risk assessments: When explicit and when tacit? *Reliability Engineering & System Safety*, *197.* https://doi.org/10.1016/j.ress.2020.106799
- Flood Forecasting Centre. (n.d.). *Welcome to the Flood Forecasting Centre*. Retrieved 22 July 2021 from http://www.ffc-environment-agency.metoffice.gov.uk/
- Flowerdew, J., Mylne, K., Jones, C., & Titley, H. (2013). Extending the forecast range of the UK storm surge ensemble. *Quarterly Journal of the Royal Meteorological Society*, *139*, 187-197. https://doi.org/10.1002/qj.1950
- Forrester, G., & Daly, M. (2021, 28 May 2021). Weather: rare red warning for heavy rain, flooding, snow during weekend as stormy system moves in. *stuff.* https://www.stuff.co.nz/national/weather-news/125273269/weather-rare-red-warning-for-heavy-rain-flooding-snow-during-weekend-as-stormy-system-moves-in
- Forrester, J., & Cinderby, S. (2011). *A Guide to using Community Mapping and Participatory-GIS*. http://www.tweedforum.org/research/Borderlands.Community.Mapping.G

http://www.tweedforum.org/research/Borderlands_Community_Mapping_Guide_.pdf

- Frame, D., Rosier, S., Carey-Smith, T., Harrington, L., Dean, S., & Noy, I. (2018). *Estimating financial costs of climate change in New Zealand*. https://www.treasury.govt.nz/sites/default/files/2018-08/LSF-estimatingfinancial-cost-of-climate-change-in-nz.pdf
- Franks, J. (2021, 21 June 2021). Auckland tornado: Why it happened, and why we weren't warned. *stuff.* https://www.stuff.co.nz/national/125505800/auckland-tornado-why-it-happened-and-why-we-werent-warned
- Frugis, B. J., & Wasula, T. (2011). *Development of Warning Thresholds for One Inch or Greater Hail in the Albany New York County Warning Area*. NOAA. https://repository.library.noaa.gov/view/noaa/6601
- Gabel, J. (2021, 15 July 2021). Weather: Red warning for parts of South Island's West Coast, heavy rain on the way. *NZ Herald*. https://www.nzherald.co.nz/nz/weather-red-warning-for-parts-of-southislands-west-coast-heavy-rain-on-theway/T3K25SOVXZ6Y3W24WUDWSJZ5MA/
- Gaillard, J. C., & Pangilinan, M. L. C. J. D. (2010). Participatory Mapping for Raising Disaster Risk Awareness Among the Youth. *Journal of Contingencies and Crisis Management*, *18*(3), 175-179. https://doi.org/10.1111/j.1468-5973.2010.00614.x
- Gall, M., & Cutter, S. L. (2016). Understanding Disaster Risk Through Loss Data. In L. F.
 P. E. Enterprises (Ed.), *Solving the Puzzle: Innovating to Reduce Risk* (pp. 70-74). International Bank for Reconstruction and Development/The World Bank. https://doi.org/10.1016/0301-4207(78)90055-7
- Gallina, V., Torresan, S., Critto, A., Sperotto, A., Glade, T., & Marcomini, A. (2016). A review of multi-risk methodologies for natural hazards: Consequences and challenges for a climate change impact assessment. *Journal of Environmental Management*, *168*, 123-132. https://doi.org/10.1016/j.jenvman.2015.11.011
- Ganzevoort, W., van den Born, R. J. G., Halffman, W., & Turnhout, S. (2017). Sharing biodiversity data: citizen scientists' concerns and motivations. *Biodiversity and Conservation*, *26*(12), 2821-2837. https://doi.org/10.1007/s10531-017-1391-z
- Gao, M., Guttikunda, S. K. S. K., Carmichael, G. R. G. R., Wang, Y., Liu, Z., Stanier, C. O.
 C. O., Saide, P. E. P. E., & Yu, M. (2015). Health impacts and economic losses assessment of the 2013 severe haze event in Beijing area. *Science of the Total Environment*, *511*(January 2013), 553-561. https://doi.org/10.1016/j.scitotenv.2015.01.005
- Garcia, C., & Fearnley, C. J. (2012). Evaluating critical links in early warning systems for natural hazards. *Environmental Hazards, 11*(2), 123-137. https://doi.org/10.1080/17477891.2011.609877
- Gaztelumendi, S., Egaña, J., Otxoa-de-Alda, K., Hernandez, R., Aranda, J., & Anitua, P. (2012). An overview of a regional meteorology warning system. *Advances in Science and Research*, 8(1), 157-166. https://doi.org/10.5194/asr-8-157-2012
- Ge, Y. G., Zobel, C. W., Murray-Tuite, P., Nateghi, R., & Wang, H. (2021). Building an Interdisciplinary Team for Disaster Response Research: A Data-Driven Approach. *Risk Analysis*, *41*(7), 1145-1151. https://doi.org/10.1111/risa.13280
- GFDRR. (2016). Implementing Multi-Hazard Impact-based Forecast and Warning Services. https://www.gfdrr.org/sites/default/files/publication/Report%20on%20Imple menting%20Multi%20Hazard%20Impact%20Based%20Forecast%20and%20

Warning%20Services%2017-02-10%20Final.pdf Ghaffarian, V. (2011). The new stream of socio-technical approach and main stream

- Ghaffarian, V. (2011). The new stream of socio-technical approach and main stream information systems research. *Procedia Computer Science*, *3*, 1499-1511. https://doi.org/10.1016/j.procs.2011.01.039
- Gibson, B. (2007). Accommodating Critical Theory. In A. Bryant & K. Charmaz (Eds.), *The SAGE Handbook of Grounded Theory* (pp. 436-454). SAGE Publications Ltd.
- Glaser, B. (1992). *Emergence vs. Forcing: Basics of grounded theory analysis.* Sociology Press.
- Glaser, B., & Strauss, A. (1967). *The Discovery of Grounded Theory: Strategies for qualitative research*. Aldine Publishing Company.
- Gneiting, T., & Raftery, A. E. (2005). Weather Forecasting with Ensemble Methods. *Science*, *310*(5746), 248-249. https://doi.org/10.1126/science.1115255
- Goffin, A., & ACC Policy Team. (2014). *Farmers' mental health: A review of the literature*. https://www.acc.co.nz/assets/research/dcaf5b4e0d/farmer-mental-health-review.pdf
- Goldfinger, D. (2020, 28 February 2020). Snow squall warning continues in Barrie, Collingwood areas. *Global News*. https://globalnews.ca/news/6608617/snowsqualls-continue-50-cm-possible-barrie-orillia-midland/
- Golding, B., Mittermaier, M., Ross, C., Ebert, B., Panchuk, S., Scolobig, A., & Johnston, D. M. (2019). *A Value Chain Approach to Optimising Early Warning Systems*. https://www.preventionweb.net/files/65828_f212goldingetalvaluechain.pdf
- Goldkuhl, G. (2004). Meanings of Pragmatism: Ways to conduct information systems research. *Action in Language, Organisations and Information Systems*(June), 13-26.
- Goldkuhl, G. (2007). What Does It Mean To Serve The Citizen In E-Services?-Towards A Practical Theory Founded In Socio-Instrumental Pragmatism. *International Journal of Public Information Systems*, 2007(3), 135-159. www.ijpis.net
- Goldkuhl, G. (2008). Practical Inquiry As Action Research And Beyond. 16th European Conference on Information Systems, Galway, Ireland.

- Golnaraghi, M. (2012). An Overview: Building a Global Knowledge Base of Lessons Learned from Good Practices in Multi-Hazard Early Warning Systems. In M. Golnaraghi (Ed.), *Institutional Partnerships in Multi-Hazard Early Warning Systems: A Compilation of Seven National Good Practices and Guiding Principles* (pp. 1-8). Springer.
- Golubiewski, N. E. (2019). Evaluating Likelihood and Consequence: Understanding the New Lynn storms of 2017.
- Goodchild, M. F. (2007). Citizens as Sensors: The World of Volunteered Geography. *GeoJournal, 69*(4), 211-221. https://doi.org/10.1007/s10708-007-9111-y
- Goodchild, M. F., & Glennon, J. A. (2010). Crowdsourcing geographic information for disaster response: a research frontier. *International Journal of Digital Earth*, *3*(3), 231-241. https://doi.org/10.1080/17538941003759255
- Gorman, P. (2016, 12 October 2016). Editorial: Why do we need both MetService and Niwa to provide weather forecasts? *stuff.* https://www.stuff.co.nz/thepress/opinion/85147912/editorial-why-do-we-need-both-metservice-andniwa-to-provide-weather-forecasts
- Government of Canada. (2018). *Enhanced Fujita scale for wind damage*. Retrieved 10 September 2021 from https://www.canada.ca/en/environment-climatechange/services/seasonal-weather-hazards/enhanced-fujita-scale-winddamage.html
- Granell, C., & Ostermann, F. O. (2016). Beyond data collection: Objectives and methods of research using VGI and geo-social media for disaster management. *Computers, Environment and Urban Systems, 59*(2016), 231-243. https://doi.org/10.1016/j.compenvurbsys.2016.01.006
- Grant, M. J., & Booth, A. (2009). A typology of reviews: An analysis of 14 review types and associated methodologies. *Health Information and Libraries Journal*, *26*(2), 91-108. https://doi.org/10.1111/j.1471-1842.2009.00848.x
- Grasso, V., & Crisci, A. (2016). Codified hashtags for weather warning on twitter: An Italian case study. *PLoS Currents, &*(Disasters), 1-30. https://doi.org/10.1371/currents.dis.967e71514ecb92402eca3bdc9b789529
- Grasso, V., Crisci, A., Morabito, M., Nesi, P., Pantaleo, G., Zaza, I., & Gozzini, B. (2017). Italian codified hashtags for weather warning on Twitter - who is really using them? *Advances in Science and Research*, *14*, 63-69. https://doi.org/10.5194/asr-14-63-2017
- Gray, D. E. (2014). *Doing Research in the Real World*. SAGE Publications Ltd.
- Greenberg, J., & Scanlon, T. J. (2016). Old Media, New Media, and the Complex Story of Disasters. In *Oxford Research Encyclopedia of Natural Hazard Science*. https://doi.org/10.1093/acrefore/9780199389407.013.21
- Gregor, S. (2006). The Nature of Theory in Information Systems. *MIS Quarterly*, *30*(3), 611-642.
- Groom, Q., Weatherdon, L., Geijzendorffer, I. R., & Cadotte, M. (2017). Is citizen science an open science in the case of biodiversity observations? *Journal of Applied Ecology*, *54*(2), 612-617. https://doi.org/10.1111/1365-2664.12767
- Guba, E. G., & Lincoln, Y. S. (1994). Competing Paradigms in Qualitative Research. In N. K. Denzin & Y. S. Lincoln (Eds.), *Handbook of Qualitative Research* (pp. 105-117). Sage.
- Guha-Sapir, D., & Below, R. (2000). *The Quality and Accuracy Disaster Data: A Comparative Analysis of Three Global Data Sets.*
- Guz, A. N., & Rushchitsky, J. J. (2009). Scopus: A system for the evaluation of scientific journals. *International Applied Mechanics*, *45*(4), 351-362. https://doi.org/10.1007/s10778-009-0189-4

- Haklay, M. (2013). Citizen Science and Volunteered Geographic Information: Overview and Typology of Participation. In D. Sui, S. Elwood, & M. Goodchild (Eds.), Crowdsourcing Geographic Knowledge: Volunteered Geographic Information (VGI) in Theory and Practice (pp. 105-122). Springer Netherlands. https://doi.org/10.1007/978-94-007-4587-2
- Haklay, M. (2017). Volunteered Geographic Information and Citizen Science. In R. Kitchin, T. P. Lauriault, & M. Wilson (Eds.), *Understanding Spatial Media* (pp. 127-135). Sage. http://www.opensignal.com/
- Haklay, M., Antoniou, V., & Basiouka, S. (2014). *Crowdsourced Geographic Information Use in Government*. http://discovery.ucl.ac.uk/1433169/
- Harrison, H., Birks, M., Franklin, R., & Mills, J. (2017). Case Study Research: Foundations and Methodological Orientations. *Forum Qualitative Sozialforschung / Forum: Qualitative Social Research*, *18*(1), 1-12. https://doi.org/10.17169/FQS-18.1.2655
- Harrison, J., McCoy, C., Bunting-Howarth, K., Sorensen, H., Williams, K., & Ellis, C. (2014). *Evaluation of the National Weather Service Impact-based Warning Tool.*
- Harrison, S. E., & Johnson, P. A. (2016). Crowdsourcing the Disaster Management Cycle. International Journal of Information Systems for Crisis Response and Management, 8(4), 17-40. https://doi.org/10.4018/ijiscram.2016100102
- Harrison, S. E., & Johnson, P. A. (2019). Challenges in the adoption of crisis crowdsourcing and social media in Canadian emergency management. *Government Information Quarterly*, 36, 501-509. https://doi.org/10.1016/j.giq.2019.04.002
- Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E. H., & Johnston, D. (2020).
 Volunteered Geographic Information for people-centred severe weather early warning: A literature review. *Australasian Journal of Disaster and Trauma Studies*, 24(1). http://trauma.massey.ac.nz/issues/2020-1/AJDTS_24_1_Harrison.pdf
- Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E. H., & Johnston, D. (2021). 'Where oh where is the data?': Identifying data sources for hydrometeorological impact forecasts and warnings in Aotearoa New Zealand. *International Journal of Disaster Risk Reduction*. https://doi.org/10.1016/j.ijdrr.2021.102619
- Harrison, S. E., Potter, S. H., Prasanna, R., Doyle, E. E. H., & Johnston, D. M. (2022). Identifying the data uses and gaps for severe weather impact forecasts and warnings. *Weather, Climate, and Society*. https://doi.org/10.1175/WCAS-D-21-0093.1
- Harrison, S. E., Silver, A., & Doberstein, B. (2015). Post-storm damage surveys of tornado hazards in Canada: Implications for mitigation and policy. *International Journal of Disaster Risk Reduction*, 13, 427-440. https://doi.org/10.1016/j.ijdrr.2015.08.005
- Harrowsmith, M., Nielsen, M., Jaime, C., Coughlan de Perez, E., Uprety, M., Johnson, C., van den Homberg, M., Tijssen, A., Page, E. M., Lux, S., & Comment, T. (2020). *The Future of Forecasts: Impact-based Forecasting for Early Action*. https://reliefweb.int/report/world/future-forecasts-impact-based-forecastingearly-action-0
- Hart, M., & Langer, E. R. (2011). *Mitigating the risk of human caused wildfires: literature review and stakeholder study* (Fire Technology Transfer Note, Issue. Mitigating the risk of human caused wildfires: literature review and stakeholder study
- Haworth, B. (2016). Emergency management perspectives on volunteered geographic information: Opportunities, challenges and change. *Computers,*

Environment and Urban Systems, *57*, 189-198. https://doi.org/10.1016/j.compenvurbsys.2016.02.009

Haworth, B. (2018). Implications of Volunteered Geographic Information for Disaster Management and GIScience: A More Complex World of Volunteered Geography. Annals of the American Association of Geographers, 108(1), 226-240.

https://www.tandfonline.com/doi/pdf/10.1080/24694452.2017.1321979?nee dAccess=true

- Haworth, B., & Bruce, E. (2015). A Review of Volunteered Geographic Information for Disaster Management. *Geography Compass*, *5*(9), 237-250. https://doi.org/10.1111/gec3.12213
- Haworth, B., Bruce, E., & Middleton, P. (2012). Emerging technologies for risk reduction: assessing the potential use of social media and VGI for increasing community engagement. *Australian Journal of Emergency Management*, *30*(3), 36-41.
- Haworth, B., Whittaker, J., & Bruce, E. (2016). Assessing the application and value of participatory mapping for community bushfire preparation. *Applied Geography*, *76*, 115-127. https://doi.org/10.1016/j.apgeog.2016.09.019
- Haworth, B. T., Bruce, E., Whittaker, J., & Read, R. (2018). The Good, the Bad, and the Uncertain: Contributions of Volunteered Geographic Information to Community Disaster Resilience. *Frontiers in Earth Science*. https://doi.org/10.3389/feart.2018.00183
- He, C., Ju, N., Xu, Q., Li, Y., & Zhao, J. (2018). Designing an Android-Based Application for Geohazard Reduction Using Citizen-Based Crowdsourcing Data. *Mobile Information Systems, 2018*, 11-11. https://doi.org/10.1155/2018/1328098
- Heinselman, P., LaDue, D., Kingfield, D. M., & Hoffman, R. (2015). Tornado Warning Decisions Using Phased-Array Radar Data. *Weather and Forecasting*, 30, 57-78. https://doi.org/10.1175/WAF-D-14-00042.1
- Hemingway, R., & Gunawan, O. (2018). The Natural Hazards Partnership: A publicsector collaboration across the UK for natural hazard disaster risk reduction. *International Journal of Disaster Risk Reduction*, 499-511. https://doi.org/10.1016/j.ijdrr.2017.11.014
- Hemingway, R., & Robbins, J. (2019). Developing a hazard-impact model to support impact-based forecasts and warnings: The Vehicle OverTurning (VOT) Model. *Meteorological Applications*(April), 1-17. https://doi.org/10.1002/met.1819
- Hendy, J., Kerr, S., Halliday, A., Owen, S., Ausseil, A.-G., Burton, R., Bell, K., Deans, N., Dickie, B., Hale, J., Hale, S., Kamish, W., Kitson, J., Mullan, B., Rodgers, R., Rosier, S., Storey, B., & Zammit, C. (2018). *Drought and Climate Change Adaptation: Impacts and Projections* (Motu Note #31). D. S. Challenge. https://deepsouthchallenge.co.nz/wp-content/uploads/2021/02/Drought-Dialogue-Report-FINAL.pdf
- Henriksen, H. J., Roberts, M. J., van der Keur, P., Harjanne, A., Egilson, D., & Alfonso, L. (2018). Participatory early warning and monitoring systems: A Nordic framework for web-based flood risk management. *International Journal of Disaster Risk Reduction*, 31, 1295-1306. https://doi.org/10.1016/j.ijdrr.2018.01.038
- Heron, J., & Reason, P. (1997). A Participatory Inquiry Paradigm. *Qualitative Inquiry*, *3*(3), 274-294. https://doi.org/10.1177/107780049700300302
- Hevner, A., March, S. T., Park, J., & Ram, S. (2004). Design Science in Information Systems Research. *MIS Quarterly, 28*(1), 75-75. https://doi.org/10.2307/25148625

- Hew, M., Sutherland, M. F., Thien, F., & O'Hehir, R. (2017). The Melbourne thunderstorm asthma event: can we avert another strike? *Internal Medicine Journal*, *47*(5), 485. https://doi.org/10.1111/imj.13408
- Hirschberg, P. A., Abrams, E., Bleistein, A., Bua, W., Monache, L. D., Dulong, T. W., Gaynor, J. E., Glahn, B., Hamill, T. M., Hansen, J. A., Hilderbrand, D. C., Hoffman, R. N., Morrow, B. H., Philips, B. P., Sokich, J., & Stuart, N. (2011). A Weather and Climate Enterprise Strategic Implementation Plan for Generating and Communicating Forecast Uncertainty Information. *Bulletin of the American Meteorological Society* (December), 1651-1666. https://doi.org/10.1175/BAMS-D-10-00073.1
- HIWeather. (2021). Value Chain Approaches to Evaluate the End-to-End Warning Chain: Project Plan. In: WMO.
- Hofmann, J., & Schüttrumpf, H. (2019). Risk-Based Early Warning System for Pluvial Flash Floods: Approaches and Foundations. *Geosciences*, 9(3). https://doi.org/10.3390/geosciences9030127
- Hong, L., Fu, C., Wu, J., & Frias-Martinez, V. (2018). Information Needs and Communication Gaps between Citizens and Local Governments Online during Natural Disasters. *Information Systems Frontiers*, 1-13. https://doi.org/10.1007/s10796-018-9832-0
- Horita, F. E. A., Degrossi, L. C., Assis, L. F. F., Zipf, A., & Porto de Albuquerque, J. (2013, 2013). The use of Volunteered Geographic Information and Crowdsourcing in Disaster Management: a Systematic Literature Review. Proceedings of the Nineteenth Americas Conference on Information Systems, Chicago, Illinois.
- Horita, F. E. A., Porto de Albuquerque, J., Degrossi, L. C., Mendiondo, E. M., & Ueyama, J. (2015). Development of a spatial decision support system for flood risk management in Brazil that combines volunteered geographic information with wireless sensor networks. *Computers & Geosciences*, *80*, 84-94. https://doi.org/10.1016/j.cageo.2015.04.001
- Horita, F. E. A., Porto de Albuquerque, J., Marchezini, V., & Mendiondo, E. M. (2016, 2016). A qualitative analysis of the early warning decision-making process in disaster management. 13th International Conference on Information Systems for Crisis Response and Management (ISCRAM 2016), Rio de Janeiro, Brazil.
- Horita, F. E. A., Porto de Albuquerque, J., Marchezini, V., & Mendiondo, E. M. (2017).
 Bridging the gap between decision-making and emerging big data sources: An application of a model-based framework to disaster management in Brazil.
 Decision Support Systems, 97, 12-22. https://doi.org/10.1016/j.dss.2017.03.001
- Horita, F. E. A., Vilela, R. B., Martins, R. G., Bressiani, D. A., Palma, G., & Porto de Albuquerque, J. (2018). Determining flooded areas using crowd sensing data and weather radar precipitation: A case study in Brazil. *15th International Conference on Information Systems for Crisis Response and Management, ISCRAM 2018*, *2018-May*(May), 1030-1039. https://www.scopus.com/inward/record.uri?eid=2-s2.0-

85060678817&partnerID=40&md5=71160a6fc9ab22da45c84ad5116adbb9

- Hosterman, H. R., Lazo, J. K., Sprague-Hilderbrand, J. M., & Adkins, J. E. (2019). Using the National Weather Service's impact-based decision support services to prepare for extreme winter storms. *Journal of emergency management*, *17*(6), 455-467. https://doi.org/10.5055/jem.2019.0439
- Huggins, T. J., Langer, E. R. L., McLennan, J., Johnston, D. M., & Yang, L. (2020). The
many-headed beast of wildfire risks in Aotearoa-New Zealand. Australasian
Journal of Emergency Management.

https://knowledge.aidr.org.au/resources/ajem-july-2020-the-many-headed-beast-of-wildfire-risks-in-aotearoa-new-zealand/

- Hughes, J., & Jones, S. (2003). Reflections on the use of Grounded Theory in Interpretive Information Systems Research. European Conference on Information Systems (ECIS), Naples, Italy.
- Hung, H.-C., & Chen, L.-Y. (2013). Incorporating stakeholders' knowledge into assessing vulnerability to climatic hazards: application to the river basin management in Taiwan. *Climatic Change*, *120*(1-2), 491-507. https://doi.org/10.1007/s10584-013-0819-z
- Hunt, E. (2021, 26 February 2021). Words matter: how New Zealand's clear messaging helped beat Covid. *The Guardian.* https://www.theguardian.com/world/2021/feb/26/words-matter-how-newzealands-clear-messaging-helped-beat-covid
- ICNZ. (2021). *Cost of natural disasters.* Insurance Council of New Zealand Te Kāhui Inihua o Aotearoa. Retrieved 30 August 2021 from https://www.icnz.org.nz/natural-disasters/cost-of-natural-disasters
- livari, J., & Venable, J. R. (2009). Action research and design science research-Seemingly similar but decisively dissimilar. European Conference on Information Systems (ECIS), Verona, Italy.
- International Science Council. (2021). Unleashing Science: Delivering Missions for Sustainability. https://council.science/science-funding/missions-forsustainability/
- Jacobs, D. C. (2010). Pragmatism. In A. J. Mills, G. Durepos, & E. Wiebe (Eds.), *Encyclopedia of Case Study Research* (pp. 724-725). SAGE Publications, Inc. https://doi.org/10.4135/9781412957397
- Janssen, M., Brous, P., Estevez, E., Barbosa, L. S., & Janowski, T. (2020). Data governance: Organizing data for trustworthy Artificial Intelligence. *Government* Information Quarterly, 37(3). https://doi.org/10.1016/j.gig.2020.101493
- Järvinen, P. (2007). Action Research is Similar to Design Science. *Quality & Quantity*, 41, 37-54. https://doi.org/10.1007/s11135-005-5427-1
- Johnson, D. R. (2021). Integrated Risk Assessment and Management Methods Are Necessary for Effective Implementation of Natural Hazards Policy. *Risk Analysis*, *41*(7), 1240-1247. https://doi.org/10.1111/risa.13268
- Johnson, R. B., & Onwuegbuzie, A. J. (2004). Mixed Methods Research: A Research Paradigm Whose Time Has Come. *Educational Researcher*, *33*(7), 14-26. https://journals.sagepub.com/doi/pdf/10.3102/0013189X033007014
- Jones, S., & Golding, B. (2014). *HIWeather: A research activity on High Impact Weather within the World Weather Research Programme.* https://www.wmo.int/pages/prog/arep/wwrp/new/documents/HIWeatherIm plementationPlan.pdf
- Joslyn, S., & LeClerc, J. (2013). Decisions With Uncertainty: The Glass Half Full. *Current Directions in Psychological Science*, *22*(4), 308-315. https://doi.org/10.1177/0963721413481473
- Jung, J., & Uejio, C. K. (2017). Social media responses to heat waves. *International Journal of Biometeorology*, *61*(7), 1247-1260. https://doi.org/10.1007/s00484-016-1302-0
- Jung, K., Song, M., & Park, H. J. (2018). The Dynamics of an Interorganizational Emergency Management Network: Interdependent and Independent Risk Hypotheses. *Public Administration Review*, *79*(2), 225-235. https://doi.org/10.1111/puar.12993

- Kaltenberger, R., Schaffhauser, A., & Staudinger, M. (2020). "What the weather will do" - results of a survey on impact-oriented and impact-based warnings in European NMHSs. *Advances in Science and Research*, *17*, 29-38. https://doi.org/10.5194/asr-17-29-2020
- Kantha, L. (2006). Time to Replace the SaffirSimpson Hurricane Scale? *Eos Transactions American Geophysical Union*, *87*(1), 3-6. https://doi.org/10.1029/2006EO010003
- Kapucu, N. (2006). Interagency Communication Networks During Emergencies: Boundary Spanners in Multiagency Coordination. *American Review of Public Administration*, 36(2), 207-225. https://doi.org/10.1177/0275074005280605
- Kapucu, N., & Demiroz, F. (2017). Interorganizational Networks in Disaster Management. In E. C. Jones & A. J. Faas (Eds.), *Social Network Analysis of Disaster Response, Recovery, and Adaptation* (pp. 25-39). Butterworth-Heinemann. https://doi.org/10.1016/B978-0-12-805196-2.00003-0
- Kelle, U. (2005). "Emergence" vs. "forcing" of empirical data? A crucial problem of "Grounded Theory" reconsidered. Forum Qualitative Sozialforschung / Forum: Qualitative Social Research, 6(2), 133-156. http://nbnresolving.de/urn:nbn:de:0114-fqs0502275
- Kelly, S., & Smith, W. (2012). Marginality, adaptation and farming in the New Zealand high country. *Revue de géographie alpine / Journal of Alpine Research*(100-1). https://doi.org/10.4000/rga.1711
- Kennedy, B. (2018). Deduction, Induction, and Adbuction. In U. Flick (Ed.), *The SAGE Handbook of Qualitative Data Collection*. SAGE. https://doi.org/10.4135/9781526416070.n4
- Khalid, A. (2019). Twitter removes precise geo-tagging option from tweets. *Engadget*. https://www.engadget.com/2019-06-19-twitter-removes-precise-geotagging.html#:~:text=Twitter%20has%20announced%20it%20will,%22simplif y%22%20the%20tweeting%20experience.
- Khan, S., Mishra, J. L., Lin, K. h. E., & Doyle, E. E. H. (2017). Rethinking communication in risk interpretation and action. *Natural Hazards*, *88*(3), 1709-1726. https://doi.org/10.1007/s11069-017-2942-z
- Klein, H. K., & Myers, M. D. (1999). A Set of Principles for Conducting and Evaluating Interpretive Field Studies in Information Systems. *MIS Quarterly, 23*(1), 67-93. https://doi.org/10.2307/249410
- Klonner, C., Marx, S., Usón, T., Porto de Albuquerque, J., Höfle, B., Klonner, C., Marx, S., Usón, T., de Albuquerque, J. P., & Höfle, B. (2016). Volunteered Geographic Information in Natural Hazard Analysis: A Systematic Literature Review of Current Approaches with a Focus on Preparedness and Mitigation. *ISPRS International Journal of Geo-Information*, 5(7), 103-103. https://doi.org/10.3390/ijgi5070103
- Kothari, C. R. (2004). *Research Methodology: Methods and Techniques* (Second ed.). New Age International Publishers.
- Kotsuki, S., Kurosawa, K., Otsuka, S., Terasaki, K., & Miyoshi, T. (2019). Global Precipitation Forecasts by Merging Extrapolation-Based Nowcast and Numerical Weather Prediction with Locally Optimized Weights. *Weather and Forecasting*, *34*(3), 701-714. https://doi.org/10.1175/waf-d-18-0164.1
- Kox, T., Gerhold, L., & Ulbrich, U. (2015). Perception and use of uncertainty in severe weather warnings by emergency services in Germany. *Atmospheric Research*, 158-159, 292-301. https://doi.org/10.1016/j.atmosres.2014.02.024
- Kox, T., Kempf, H., Lüder, C., Hagedorn, R., & Gerhold, L. (2018). Towards userorientated weather warnings. *International Journal of Disaster Risk Reduction*, *30*(February), 74-80. https://doi.org/10.1016/j.ijdrr.2018.02.033

- Kox, T., & Lüder, C. (2021). Impacts as Triggers for Weather-Related Decision Making: Observations at the Berlin Fire Brigade Control and Dispatch Center. *International Journal of Disaster Risk Science*. https://doi.org/10.1007/s13753-021-00356-4
- Kox, T., Lüder, C., & Gerhold, L. (2018). Anticipation and Response: Emergency Services in Severe Weather Situations in Germany. *International Journal of Disaster Risk Science*, 9, 116-128. https://doi.org/10.1007/s13753-018-0163-z
- Krennert, T., Kaltenberger, R., Pistotnik, G., Holzer, A. M., Zeiler, F., & Stampfl, M. (2018). Trusted Spotter Network Austria - a new standard to utilize crowdsourced weather and impact observations. *Advances in Science and Research*, 15, 77-80. https://doi.org/10.5194/asr-15-77-2018
- Krennert, T., Pistotnik, G., Kaltenberger, R., & Csekits, C. (2018). Crowdsourcing of weather observations at national meteorological and hydrological services in Europe. Advances in Science and Research, 15, 71-76. https://doi.org/10.5194/asr-15-71-2018
- Lamb, S., Walton, D., Mora, K., & Thomas, J. (2012). Effect of authoritative information and message characteristics on evacuation and shadow evacuation in a simulated flood event. *Natural Hazards Review*, 13(4), 272-282. https://doi.org/10.1061/(ASCE)NH.1527-6996.0000070
- Langer, E. R., & Wegner, S. (2018). Wildfire risk awareness, perception and preparedness in the urban fringe in Aotearoa/New Zealand: Public responses to the 2017 Port Hills wildfire. *Australasian Journal of Disaster and Trauma Studies* 22, 75-84. https://www.massey.ac.nz/~trauma/issues/2018-2/AJDTS_22_2_Langer.pdf
- Lauriault, T., & Mooney, P. (2014). *Crowdsourcing: A Geographic Approach to Public Engagement.* SSRN.

https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2518233

- Lavers, T., & Charlesworth, S. (2018). Opportunity mapping of natural flood management measures: a case study from the headwaters of the Warwickshire-Avon. *Environmental Science and Pollution Research*, *25*(20), 19313-19322. https://doi.org/10.1007/s11356-017-0418-z
- LeClerc, J., & Joslyn, S. (2015). The cry wolf effect and weather-related decision making. *Risk Analysis*, *35*(3), 385-395. https://doi.org/10.1111/risa.12336
- Lee, A. S. (2004). Thinking about Social Theory and Philosophy for Information Systems. In J. Mingers & L. P. Willcocks (Eds.), *Social Theory and Philosophy for Information Systems* (pp. 1-26). John Wiley and Sons Ltd.
- Lehmann, H. (2010). Research Method: Grounded Theory for Descriptive and Exploratory Case Studies. In *The Dynamics of International Information Systems*. Springer Science+Business Media, LLC. https://doi.org/10.1007/978-1-4419-5750-4_5
- Leonard, G. S., Stewart, C., Wilson, T. M., Procter, J. N., Scott, B. J., Keys, H. J., Jolly, G. E., Wardman, J. B., Cronin, S. J., & McBride, S. K. (2014). Integrating multidisciplinary science, modelling and impact data into evolving, syn-event volcanic hazard mapping and communication: A case study from the 2012 Tongariro eruption crisis, New Zealand. *Journal of Volcanology and Geothermal Research, 286, 208-232.* https://doi.org/10.1016/j.jvolgeores.2014.08.018
- Li, G., Zhao, J., Murray, V., Song, C., & Zhang, L. (2019). Gap analysis on open data interconnectivity for disaster risk research. *Geo-spatial Information Science*, *22*(1), 45-58. https://doi.org/10.1080/10095020.2018.1560056
- Lin, S.-L., King, A., Horspool, N., Sadashiva, V., Paulik, R., & Williams, S. (2019). Development and Application of the Real-Time Individual Asset Attribute

Collection Tool. *Frontiers in Built Environment, 5.* https://doi.org/10.3389/fbuil.2019.00015

- Lin, S. L., King, A. B., & Matcham, I. (2014). *Exposure data development framework: Real-time Individual Asset Attribute Collection Tool (RiACT), inventory repository & asset repository web portal* 2014 NZSEE Conference, Auckland, NZ.
- Lin, S. L., Ungaro, J., Nimau, J. T., Singh, S., & Raqisia, K. (2020). Development of pacific exposure dataset for use in catastrophe risk assessment. *Natural Hazards*. https://doi.org/10.1007/s11069-020-04290-4
- Lincoln, Y. S., & Guba, E. G. (2005). Paradigmatic Controversies, Contradictions, and Emerging Confluences. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage handbook of qualitative research* (pp. 191-215). SAGE Publications Ltd.
- Lindell, M. K., Prater, C. S., & Perry, R. W. (2006). Hazard, Vulnerability, and Risk Analysis. In *Fundamentals of Emergency Management* (pp. 153-191). FEMA Emergency Management Institute.
- LINZ. (2014a). New Zealand fundamental geospatial datasets and themes v2.2. Wellington, NZ: LINZ Retrieved from https://www.linz.govt.nz/system/files_force/media/fileattachments/Fundamental%20datasets%20and%20themes.pdf?download=1
- LINZ. (2014b). Steward and custodian framework for New Zealand fundamental geospatial themes and datasets v2.3. Wellington, NZ: LINZ Retrieved from https://www.linz.govt.nz/system/files_force/media/pages-attachments/Steward%20and%20custodian%20framework_0.pdf?download =1&download=1
- Longmore, S., Miller, S., Bikos, D., Lindsey, D., Szoke, E., Molenar, D., Hillger, D., Brummer, R., & Knaff, J. (2015). An automated mobile phone photo relay and display concept applicable to operational severe weather monitoring. *Journal* of Atmospheric and Oceanic Technology, 32(7), 1356-1363. https://doi.org/10.1175/JTECH-D-14-00230.1
- Losee, J. E., & Joslyn, S. (2018). The need to trust: How features of the forecasted weather influence forecast trust. *International Journal of Disaster Risk Reduction*, *30*(February), 95-104. https://doi.org/10.1016/j.ijdrr.2018.02.032
- Losee, J. E., Naufel, K. Z., Locker, L., & Webster, G. D. (2017). Weather Warning Uncertainty: High Severity Influences Judgment Bias. *Weather, Climate, and Society*, 9(3), 441-454. https://doi.org/10.1175/wcas-d-16-0071.1
- Loukis, E., & Charalabidis, Y. (2015). Active and Passive Crowdsourcing in Government. In M. Janssen, M. A. Wimmer, & A. Deljoo (Eds.), *Policy Practice and Digital Science* (pp. 261-289). Springer International Publishing. https://doi.org/10.1007/978-3-319-12784-2_12
- Lu, H., Zhu, Y., Shi, K., Lv, Y., Shi, P., & Niu, Z. (2018). Using Adverse Weather Data in Social Media to Assist with City-Level Traffic Situation Awareness and Alerting. *Applied Sciences*, 8(7), 1193-1193. https://doi.org/10.3390/app8071193
- Luck, L., Jackson, D., & Usher, K. (2006). Case study: A bridge across the paradigms. *Nursing Inquiry*, *13*(2), 103-109. https://doi.org/10.1111/j.1440-1800.2006.00309.x
- Ludäscher, B., Lin, K., Bowers, S., Jaeger-Frank, E., Brodaric, B., & Baru, C. (2006). Managing Scientific Data: From Data Integration to Scientific Workflows. In K. Sinha (Ed.), *Geoinformatics: Data to Knowledge* (pp. 282). Geological Society of America.
- Mack, N., Woodsong, C., MacQueen, K. M., Guest, G., & Namey, E. (2005). *Qualitative research methods: A data collector's field guide*. Family Health International.

- Mackenzie, N., & Knipe, S. (2006). Research dilemmas: Paradigms, methods and methodology. *Issues in Educational Research*, *16*, 10-10. http://www.iier.org.au/iier16/mackenzie.html
- Mackie, B. (2013). *Warning Fatigue: Insights from the Australian Bushfire Context* [Doctor of Philosophy in Media and Communication, University of Canterbury]. https://ir.canterbury.ac.nz/handle/10092/9029
- Mackie, B. (2014). Warning Fatigue is Not a Myth: Understanding Why People Do or Don't Respond to Warnings. *Fire Note*. http://www.bushfirecrc.com/sites/default/files/managed/resource/fire_note_ 122_low_res.pdf
- Majumdar, S. J., Sun, J., Golding, B., Joe, P., Dudhia, J., Caumont, O., Chandra Gouda, K., Steinle, P., Vincendon, B., Wang, J., & Yussouf, N. (2021). Multiscale Forecasting of High-Impact Weather: Current Status and Future Challenges. *Bulletin of the American Meteorological Society*, *102*(3), E635-E659. https://doi.org/10.1175/bams-d-20-0111.1
- Mansourian, A., Rajabifard, A., Valadan Zoej, M. J., & Williamson, I. (2006). Using SDI and web-based system to facilitate disaster management. *Computers & Geosciences*, *32*(3), 303-315. https://doi.org/10.1016/j.cageo.2005.06.017
- Maps Open Source. (n.d.). *Maps of New Zealand*. Retrieved 13 August 2021 from http://www.mapsopensource.com/new-zealand
- Marchezini, V., Horita, F. E. A., Matsuo, P. M., Trajber, R., Trejo-Rangel, M. A., & Olivato, D. (2018). A Review of Studies on Participatory Early Warning Systems (P-EWS): Pathways to Support Citizen Science Initiatives. *Frontiers in Earth Science*, 6(184), 1-18. https://doi.org/10.3389/feart.2018.00184
- Martin, V. Y., Christidis, L., & Pecl, G. T. (2016). Public Interest in Marine Citizen Science: Is there Potential for Growth? *BioScience*, *66*(8), 683-692. https://doi.org/10.1093/biosci/biw070
- Martins, L., & Silva, V. (2020). Development of a fragility and vulnerability model for global seismic risk analyses. *Bulletin of Earthquake Engineering*. https://doi.org/10.1007/s10518-020-00885-1
- Maskrey, S. A., Mount, N. J., Thorne, C. R., & Dryden, I. (2016). Participatory modelling for stakeholder involvement in the development of flood risk management intervention options. *Environmental Modelling and Software*, *82*, 275-294. https://doi.org/10.1016/j.envsoft.2016.04.027
- Mason, J. B., & Senkbeil, J. C. (2015). A Tornado Watch Scale to Improve Public Response*. *Weather, Climate, and Society, 7*, 146-158. https://journals.ametsoc.org/doi/pdf/10.1175/WCAS-D-14-00035.1
- Mason, K., Lindberg, K., Haenfling, C., Schori, A., Marsters, H., Read, D., & Borman, B. (2021). Social Vulnerability Indicators for Flooding in Aotearoa New Zealand. *International Journal of Environmental Research and Public Health*, *18*(8). https://doi.org/10.3390/ijerph18083952
- Massagrande, F. (1995). Using GIS with non-systematic survey data: the Mediterranean evidence: A European Perspective. In G. R. Lock & G. Stancic (Eds.), *Archaeology And Geographic Information Systems: A European Perspective* (pp. 55-66). CRC Press.
- Massey, C., Potter, S., Leonard, G. S., Strawbridge, G., Rosser, B. J., & de Vilder, S. (2019). *Awatarariki catchment debris flow early warning system framework* (GNS Science Report 2019/77).
- Massey University. (n.d.). *Doctoral Thesis With Publication Guidelines*. Retrieved 8 July 2021 from

https://www.massey.ac.nz/massey/fms/Research/Graduate%20Research%20 School/Documents/Administration%20forms/Doctoral%20Thesis%20with%2 0Publications%20-

%20Guildelines%202020%20Amended.pdf?E248979156FD8F340815D246E 5FD6479

- Matavire, R., & Brown, I. (2013). Profiling grounded theory approaches in information systems research. European Journal of Information Systems, 22(1), 119-129. https://doi.org/10.1057/ejis.2011.35
- MBIE. (2017). Weather Permitting: Review of open access to weather data in New Zealand. https://www.mbie.govt.nz/assets/5b3b826f79/weather-permittingreview.pdf
- MBIE. (2018a). Field Guide: Rapid Post Disaster Building Usability Assessment -Geotechnical. Wellington, NZ: New Zealand Government Retrieved from https://www.building.govt.nz/assets/Uploads/managing-buildings/postemergency-building-assessment/building-usability-assessmentgeotechnical.pdf
- MBIE. (2018b). Rapid building assessment resources. New Zealand Government. Retrieved 20 May 2021 from https://www.building.govt.nz/managingbuildings/managing-buildings-in-an-emergency/rapid-building-assessmentresources/
- MBIE. (2021). Crown Research Institutes. Retrieved May from 10 https://www.mbie.govt.nz/science-and-technology/science-andinnovation/agencies-policies-and-budget-initiatives/researchorganisations/cri/
- MBIE. (n.d.). Resilience to Nature's challenges | Kia manawaroa Ngā Ākina o Te Ao Tūroa. Retrieved 21 July 2021 from https://www.mbie.govt.nz/science-andtechnology/science-and-innovation/funding-information-andopportunities/investment-funds/national-science-challenges/the-11challenges/resilience-to-natures-challenges/
- McCallum, I., Liu, W., See, L., Mechler, R., Keating, A., Hochrainer-Stigler, S., Mochizuki, J., Fritz, S., Dugar, S., Arestegui, M., Szoenyi, M., Bayas, J.-C. L., Burek, P., French, A., & Moorthy, I. (2016). Technologies to Support Community Flood Disaster Risk Reduction. International Journal of Disaster Risk Science, 7(2), 198-204. https://doi.org/10.1007/s13753-016-0086-5
- MCDEM. (2010). Part B: Hazard-specific information Thunderstorms (Working from same consistent messages the page for CDEM, Issue. https://www.civildefence.govt.nz/assets/Uploads/publications/consistentmessages-part-B-thunderstorms.pdf
- MCDEM. (2015). Part II Section 7 Needs Assessment Welfare Services in an Emergency Director's Guideline [DGL 11/15]. (DGL 11/15). Wellington, NZ: New Zealand Government Retrieved from https://www.civildefence.govt.nz/assets/Welfare-Services-in-an-Emergency/Welfare-Services-in-an-Emergency-Directors-Guideline.pdf
- Civil Defence Emergency Management Amendment Act 2016, (2016). https://www.legislation.govt.nz/act/public/2016/0088/latest/DLM6648809.ht ml?src=qs
- McPherson, M., Counahan, M., & Hall, J. L. (2015). Responding to Typhoon Haiyan in the Philippines. Western Pacific Surveillance and Response Journal, 6, 1-4. https://doi.org/10.5365/WPSAR.2015.6.4.HYN_026
- McSaveney, M. J., Beetham, R. D., & Leonard, G. S. (2005). The 18 May 2005 debris flow disaster at Matata: Causes and mitigation suggestions (Client Report 2005/71). Science. G.

https://static.geonet.org.nz/info/reports/landslide/CR_2005-071.pdf

- Mehta, A. M., Bruns, A., & Newton, J. (2017). Trust, but verify: social media models for disaster management. *Disasters*, *41*(3), 549-565. https://doi.org/10.1111/disa.12218
- Meissen, U., & Fuchs-Kittowski, F. (2014a, 2014). Crowdsourcing in early warning systems. 7th International Congress on Environmental Modelling and Software: Bold Visions for Environmental Modeling, iEMSs 2014, San Diego, USA.
- Meissen, U., & Fuchs-Kittowski, F. (2014b, 2014). *Towards a reference architecture of crowdsourcing integration in early warning systems* 11th International Conference on Information Systems for Crisis Response and Management, Pennsylvania, USA.
- Mertens, D. M. (2015). An Introduction to Research. In *Research and Evaluation in Education and Psychology* (pp. 1-45). SAGE Publications, Inc. https://www.sagepub.com/sites/default/files/upm-binaries/29985_Chapter1.pdf
- Merz, B., Kuhlicke, C., Kunz, M., Pittore, M., Babeyko, A., Bresch, D. N., Domeisen, D. I. V., Feser, F., Koszalka, I., Kreibich, H., Pantillon, F., Parolai, S., Pinto, J. G., Punge, H. J., Rivalta, E., Schröter, K., Strehlow, K., Weisse, R., & Wurpts, A. (2020). Impact Forecasting to Support Emergency Management of Natural Hazards. *Reviews of Geophysics*, 1-52. https://doi.org/10.1029/2020RG000704
- MetService. (2019a, 14 May 2019). *Changes to the official Severe Weather Warnings* https://about.metservice.com/assets/Watches-and-Warnings/PR-Changes-tothe-official-Severe-Weather-Warnings-140519.pdf
- MetService. (2019b). Severe Weather Warnings and Watches. Retrieved 2 July 2021 from https://about.metservice.com/our-company/national-weatherservices/changes-to-warnings-and-watches/
- MetService. (2021a). *Open Access Data.* Retrieved 22 June from https://about.metservice.com/our-company/about-this-site/open-access-data/
- MetService. (2021b). *Warnings*. Retrieved 9 June from https://www.metservice.com/warnings/home#:~:text=MetService%20will%20 issue%20a%20Heavy,period%20of%2014s%20or%20more
- MetService. (2021c). *Weather Warning Criteria*. Retrieved 2 July 2021 from https://www.metservice.com/warnings/home
- Meyer, C. B. (2001). A Case in Case Study Methodology. *Field Methods*, *13*(4), 329-352. https://journals.sagepub.com/doi/pdf/10.1177/1525822X0101300402
- Migliorini, M., Hagen, J. S., Mihaljević, J., Mysiak, J., Rossi, J.-L., Siegmund, A., Meliksetian, K., & Guha Sapir, D. (2019). Data interoperability for disaster risk reduction in Europe. *Disaster Prevention and Management: An International Journal*, 28(6), 804-816. https://doi.org/10.1108/dpm-09-2019-0291
- Miles, B., & Morse, S. (2007). The role of news media in natural disaster risk and recovery. *Ecological Economics*, *63*(2-3), 365-373. https://doi.org/10.1016/j.ecolecon.2006.08.007
- Miles, M. B., & Huberman, A. M. (1994). *Qualitative data analysis: an expanded sourcebook* (2nd ed.). Sage Publications, Inc.
- Mileti, D. S. (1999). Public response to disaster warnings. In. Boulder, Colorado.
- Mileti, D. S., & Sorensen, J. H. (1990). *Communication of Emergency Public Warnings:* A Social Science Perspective and State-of-the-Art Assessment.
- Resource Management Act 1991, (1991).

- Ministry for the Environment. (2016). *Flooding roles and responsibilities*. Retrieved 16 March from https://www.mfe.govt.nz/more/natural-hazardmanagement/flood-risk-management/roles-and-responsibilites
- Ministry for the Environment, & The Flood Risk Management and River Control Review Steering Group. (2008). *Meeting the Challenges of Future Flooding in New Zealand*. https://environment.govt.nz/publications/meeting-the-challengesof-future-flooding-in-new-zealand/2-flood-risk-management-in-newzealand/#footnote-2

Meteorological Services Act, (1990).

- Molka-Danielsen, J., Prasolova-Førland, E., Fominykh, M., & Lamb, K. (2018). Use of a Collaborative Virtual Reality Simulation for Multi-Professional Training in Emergency Management Communications 2018 IEEE International Conference on Teaching, Assessment, and Learning for Engineering (TALE),
- Morgan, D. L. (2007). Paradigms Lost and Pragmatism Regained: Methodological Implications of Combining Qualitative and Quantitative Methods. *Journal of Mixed Methods Research*, 1(1), 48-76. https://doi.org/10.1177/2345678906292462
- Morgan, D. L. (2014). Pragmatism as a Paradigm for Social Research. *Qualitative Inquiry*, *20*(8), 1045-1053. https://doi.org/10.1177/1077800413513733
- Morgan, G., Fischhoff, B., Bostrom, A., Lave, L., & Atman, C. J. (1992). Communicating Risk to the Public. *Environ. Sci. Technol*, *26*(11), 2048-2056. https://pubs.acs.org/sharingguidelines
- Morss, R. E., Cuite, C. L., Demuth, J. L., Hallman, W. K., & Shwom, R. L. (2018). Is storm surge scary? The influence of hazard, impact, and fear-based messages and individual differences on responses to hurricane risks in the U.S. *International Journal of Disaster Risk Reduction*, 30(January), 1-15. https://doi.org/10.1016/j.ijdrr.2018.01.023
- Morss, R. E., Demuth, J. L., Bostrom, A., Lazo, J. K., & Lazrus, H. (2015). Flash Flood Risks and Warning Decisions: A Mental Models Study of Forecasters, Public Officials, and Media Broadcasters in Boulder, Colorado. *Risk Analysis*, *35*(11), 2009-2028. https://doi.org/10.1111/risa.12403
- Morss, R. E., Demuth, J. L., Lazo, J. K., Dickinson, K., Lazrus, H., & Morrow, B. H. (2016). Understanding Public Hurricane Evacuation Decisions and Responses to Forecast and Warning Messages*. *Weather and Forecasting*, *31*(2), 395-417. https://doi.org/10.1175/WAF-D-15-0066.1
- Morss, R. E., Lazrus, H., & Demuth, J. L. (2021). The "Inter" Within Interdisciplinary Research: Strategies for Building Integration Across Fields. *Risk Analysis*, *41*(7), 1152-1161. https://doi.org/10.1111/risa.13246
- Morss, R. E., Mulder, K. J., Lazo, J. K., & Demuth, J. L. (2016). How do people perceive, understand, and anticipate responding to flash flood risks and warnings? Results from a public survey in Boulder, Colorado, USA. *Journal of Hydrology*, *541*, 649-664. https://doi.org/10.1016/j.jhydrol.2015.11.047
- Morton, J. (2021). Wave warning: Flooding, damage, possible for Wellington's south coast. *NZ Herald*. https://www.nzherald.co.nz/nz/wave-warning-floodingdamage-possible-for-wellingtons-southcoast/R37FFTFQ6TWEVD6J6RQ423M4IQ/
- Mullan, B., Porteous, A., Wratt, D., & Hollis, M. (2005). *Changes in drought risk with climate change* (NIWA Project MFE05305). https://docs.niwa.co.nz/library/public/WLG2005-23execsum.pdf
- Myers, M. D., & Avison, D. E. (2002). An Introduction to Qualitative Research in Information Systems In: Qualitative Research in Information Systems. https://doi.org/10.4135/9781849209687

- Nair, P. (2010). Role of Media in Disaster Management. *Mass Communicator, 4*(1), 36-40.
- National Research Council. (2006). *Completing the Forecast: Characterizing and Communicating Uncertainty for Better Decisions Using Weather and Climate Forecasts*. National Academies Press. https://doi.org/10.17226/11699
- Natural Hazards Research Platform. (2020). *NHRP About Us*. Retrieved 21 July 2021 from https://www.naturalhazards.org.nz/NHRP/About-Us
- Neal, R. A., Boyle, P., Grahame, N., Mylne, K., & Sharpe, M. (2014). Ensemble based first guess support towards a risk-based severe weather warning service. *Meteorological Applications*, 21(3), 563-577. https://doi.org/10.1002/met.1377
- Neal, T. (2020, 29 December 2020). Tasman growers and farmers brace for lasting impact of hailstorm. *Radio New Zealand*. https://www.rnz.co.nz/news/country/433749/tasman-growers-and-farmersbrace-for-lasting-impact-of-hailstorm
- NEMA. (2009). CDEM Exercises: Director's Guideline for Civil Defence Emergency Management Groups. M. o. C. D. a. E. Management. https://www.civildefence.govt.nz/assets/Uploads/publications/dgl-10-09cdem-exercises.pdf
- NEMA. (2019). National Disaster Resilience Strategy Rautaki ā-Motu Manawaroa Aituā. In. New Zealand: New Zealand Government.
- NEMA. (2020a). Impact Assessments Director's Guideline for Civil Defence Emergency Management Groups [DGL 22/20]. (DGL 22/20). Wellington, NZ: New Zealand Government Retrieved from https://www.civildefence.govt.nz/cdem-sector/guidelines/impactassessments-directors-guideline-and-national-impact-assessment-data-setand-dictionary-technical-standard/
- NEMA. (2020b). New Zealand Volcanic Science Advisory Panel (NZVSAP) Terms of Reference. Retrieved from https://www.civildefence.govt.nz/assets/Uploads/publications/nzvsap-torsep-2020.pdf
- NEMA. (2020c). Technical Standard [TS05/20] National Impact Assessments Data Set and Dictionary. (TS05/20). Wellington, NZ: New Zealand Government Retrieved from https://www.civildefence.govt.nz/cdemsector/guidelines/impact-assessments-directors-guideline-and-nationalimpact-assessment-data-set-and-dictionary-technical-standard/
- NEMA. (2021a). CDEM Groups. National Emergency Management Agency. Retrieved 4 July 2021 from https://www.civildefence.govt.nz/cdem-sector/cdemgroups/
- NEMA. (2021b). *Regulatory Framework Review ("Trifecta") Programme*. NEMA. Retrieved 7 September 2021 from https://www.civildefence.govt.nz/cdemsector/regulatory-framework-review-trifecta-programme/
- New Zealand Government. (2015a). *The Guide to the National Civil Defence Emergency Management Plan 2015*. Wellington, NZ: Department of the Prime Minister and Cabinet Retrieved from https://www.civildefence.govt.nz/assets/guide-to-the-national-cdemplan/Guide-to-the-National-CDEM-Plan-2015.pdf
- National Civil Defence Emergency Management Plan Order 2015, (2015b). https://www.civildefence.govt.nz/cdem-sector/plans-and-strategies/nationalcivil-defence-emergency-management-plan-and-guide/
- Nielsen. (2017). *Wellbeing Survey.* https://www.cph.co.nz/wpcontent/uploads/CantyWellbeingSurveyJun2017.pdf

- Noll, B. (2018). *Monthly Climate Summary*. https://niwa.co.nz/sites/niwa.co.nz/files/Climate_Summary_February_2018_Fi nal.pdf
- NWS. (n.d.). *The Weather Enterprise Working Together to Meet the Needs of Society.* Retrieved 6 September 2021 from https://www.weather.gov/about/weatherenterprise
- NZ Mountain Safety Council. (2021). Avalanche Incidents in New Zealand. Retrieved 13 August 2021 from

https://storymaps.arcgis.com/stories/ffb513674dca40389fdbf40ab48ad058

- OASIS. (2010). *Common Operating Protocol Version 1.2*. http://docs.oasisopen.org/emergency/cap/v1.2/CAP-v1.2-os.html
- Obermeier, H., & Anderson, M. R. (2014). Verification and Analysis of Impact-Based Tornado Warnings in the Central Region of the National Weather Service. *Electronic Journal of Severe Storms Meteorology*, *10*(1), 1-20. https://ejssm.org/archives/2015/vol-10-1-2015/
- OECD. (2019). Enhancing Access to and Sharing of Data: Reconciling Risks And Benefits For Data Re-Use Across Societies. OECD Publishing. https://doi.org/10.1787/276aaca8-en
- Olmos-Penuela, J., Molas-Gallart, J., & Castro-Martinez, E. (2013). Informal collaborations between social sciences and humanities researchers and non-academic partners. *Science and Public Policy*, *41*(4), 493-506. https://doi.org/10.1093/scipol/sct075
- Otto, P., Mehta, A., & Liu, B. (2017). Mind The Gap: Towards And Beyond Impact Messaging To Enhance Tropical Cyclone Risk Communication. *Tropical Cyclone Research and Review*, 7(2), 140-151. https://doi.org/10.6057/2018TCRR02.05
- Owen, C., Bearman, C., Brooks, B., Chapman, J., Paton, D., & Hossain, L. (2013). Developing a research framework for complex multi-team coordination in emergency management. *International Journal of Emergency Management*, 9(1). https://doi.org/10.1504/ijem.2013.054098
- Pak, K., Joslyn, S., Jones, D., Pyles, J., & Hunt, E. (2007). The Effect of Probabilistic Information on Threshold Forecasts. *Weather and Forecasting*, 22(4), 804-812. https://doi.org/10.1175/waf1020.1
- Panteli, M., Pickering, C., Wilkinson, S., Dawson, R., & Mancarella, P. (2017). Power System Resilience to Extreme Weather: Fragility Modeling, Probabilistic Impact Assessment, and Adaptation Measures. *IEEE Transactions on Power Systems*, 32(5), 3747-3757. https://doi.org/10.1109/TPWRS.2016.2641463
- Paré, G., Trudel, M.-C., Jaana, M., & Kitsiou, S. (2015). Synthesizing information systems knowledge: A typology of literature reviews. *Information & Management*, 52, 183-199. https://doi.org/10.1016/j.im.2014.08.008
- Pascal, M., Laaidi, K., Ledrans, M., Baffert, E., Caserio-Schönemann, C., Le Tertre, A., Manach, J., Medina, S., Rudant, J., & Empereur-Bissonnet, P. (2006). France's heat health watch warning system. *International Journal of Biometeorology*, 50(3), 144-153. https://doi.org/10.1007/s00484-005-0003-x
- Paton, D., & Johnston, D. M. (2006). *Disaster resilience: an integrated approach* (D. Paton, Ed.). Charles C Thomas.
- Paton, D., Johnston, D. M., Houghton, B., Flin, R., Ronan, K. R., & Scott, B. J. (1999). Managing natural hazard consequences: information management and decision making. *Journal of the American Society of Professional Emergency Managers*, 6, 37-48.
- Patton, M. Q. (2002). *Qualitative research and evaluation methods* (3rd ed.). Sage Publications, Inc.

- Patton, M. Q. (2015). *Qualitative research & evaluation methods: integrating theory and practice* (4th ed.). SAGE.
- Paulik, R., Stephens, S., Bell, R., Wadhwa, S., & Popovich, B. (2020). National-Scale Built-Environment Exposure to 100-Year Extreme Sea Levels and Sea-Level Rise. *Sustainability*, *12*(4). https://doi.org/10.3390/su12041513
- Pearce, H. G. (2018). The 2017 Port Hills wildfires a window into New Zealand's fire future? *Australasian Journal of Disaster and Trauma Studies, 22*(December), 35-50. https://www.massey.ac.nz/~trauma/issues/2018-2/AJDTS 22 2 Pearce.pdf
- Pearce, H. G., Cameron, G., Anderson, S. A. J., & Dudfield, M. (2008). An overview of fire management in New Zealand forestry. *New Zealand Journal of Forestry*, 53(3), 7-11. http://nzjf.org.nz/free_issues/NZJF53_3_2008/BC1A4529-8B01-4cdf-BC78-DC52A31A157C.pdf
- Pearce, H. G., & Clifford, V. R. (2008). Fire weather and climate of New Zealand. *New Zealand Journal of Forestry*, *53*(3), 13-18. http://nzjf.org.nz/free_issues/NZJF53_3_2008/7689F14A-524F-4196-8A71-76F4796DBBBD.pdf
- Peek, L., Tobin, J., van de Lindt, J. W., & Andrews, A. (2021). Getting Interdisciplinary Teams into the Field: Institutional Review Board Preapproval and Multi-Institution Authorization Agreements for Rapid Response Disaster Research. *Risk Analysis*, 41(7), 1204-1212. https://doi.org/10.1111/risa.13740
- Peffers, K., Tuunanen, T., Rothenberger, M. A., & Chatterjee, S. (2007). A design science research methodology for information systems research. *Journal of Management Information Systems*, *24*(3), 45-77. https://doi.org/10.2753/MIS0742-1222240302
- Peled, A. (2011). When transparency and collaboration collide: The USA Open Data program. *Journal of the American Society for Information Science and Technology*, *62*(11), 2085-2094. https://doi.org/10.1002/asi.21622
- Pennington, C., Freeborough, K., Dashwood, C., Dijkstra, T., & Lawrie, K. (2015). The National Landslide Database of Great Britain: Acquisition, communication and the role of social media. *Geomorphology*, 249, 44-51. https://doi.org/10.1016/j.geomorph.2015.03.013
- Peters-Guarin, G., McCall, M. K., & van Westen, C. (2012). Coping strategies and risk manageability: Using participatory geographical information systems to represent local knowledge. *Disasters*, *36*(1), 1-27. https://doi.org/10.1111/j.1467-7717.2011.01247.x
- Pita, G. L., & de Schwarzkopf, M. L. A. (2016). Urban downburst vulnerability and damage assessment from a case study in Argentina. *Natural Hazards, 83*(1), 445-463. https://doi.org/10.1007/s11069-016-2323-z
- Plotkin, D. (2020). *Data Stewardship: An Actionable Guide to Effective Data Management and Data Governance*. Academic Press,.
- Poblet, M., Garcia-Cuesta, E., Casanovas, P., García-cuesta, E., & Casanovas, P. (2014). Crowdsourcing Tools for Disaster Management: A Review of Platforms and Methods. In *Al Approaches to the Complexity of Legal Systems* (Vol. 8929, pp. 261-274). Springer Berlin Heidelberg. https://doi.org/10.5751/ES-05263-170431
- Poolman, E. R. (2014). A probabilistic impact-focussed early warning system for flash floods in support of disaster management in South Africa [Doctor of Philosophy in Meteorology, University of Pretoria]. Pretoria. http://hdl.handle.net/2263/44331

- Poortinga, W., & Pidgeon, N. F. (2003). Exploring the dimensionality of trust in risk regulation. *Risk Analysis*, *23*(5), 961-972. https://doi.org/10.1111/1539-6924.00373
- Porto de Albuquerque, J., Eckle, M., Herfort, B., & Zipf, A. (2016). Crowdsourcing geographic information for disaster management and improving urban resilience: an overview of recent developments and lessons learned. *European Handbook of Crowdsourced Geographic Information*(August), 309-321. https://doi.org/10.5334/bax.w
- Porto de Albuquerque, J., Horita, F. E. A., Degrossi, L. C., Rocha, R. d. S., Camargo de Andrade, S., Restrepo-Estrada, C., & Leyh, W. (2017). Leveraging Volunteered Geographic Information to Improve Disaster Resilience. In A. Ballatore, B. Bennett, C. Claramount, C. Davis, Jr., S. Di Martino, A. Klippel, H. Martin, & B. Schoen-Phelan (Eds.), *Volunteered Geographic Information and the Future of Geospatial Data* (pp. 158-184). https://doi.org/10.4018/978-1-5225-2446-5.ch009
- Potter, S. H. (2014). Communicating the status of volcanic activity in New Zealand, with specific application to caldera unrest [Doctoral, Massey University]. Wellington, New Zealand. http://hdl.handle.net/10179/5654
- Potter, S. H., Harrison, S. E., & Kreft, P. (2021). The benefits and challenges of implementing impact-based severe weather warning systems: Perspectives of weather, flood, and emergency management personnel. *Weather, Climate, and Society*, 13(2), 303-314. https://doi.org/10.1175/wcas-d-20-0110.1
- Potter, S. H., Harrison, S. E., Thomas, K.-L., Leonard, G., Skilton, J., Apiti, M., Balfour, N. J., Harvey, S., Coomer, M. A., & Kilgour, G. N. (2020). *Connecting real-time hazard and impact data with users: Proof of concept of a visualisation tool.*
- Potter, S. H., Jolly, G. E., Neall, V. E., Johnston, D. M., & Scott, B. J. (2014). Communicating the status of volcanic activity: Revising New Zealand's volcanic alert level system. *Journal of Applied Volcanology*, *3*. https://doi.org/10.1186/s13617-014-0013-7
- Potter, S. H., Kreft, P. V., Milojev, P., Noble, C., Montz, B., Dhellemmes, A., Woods, R. J., & Gauden-Ing, S. (2018). The influence of impact-based severe weather warnings on risk perceptions and intended protective actions. *International Journal of Disaster Risk Reduction*, 30(March), 34-43. https://doi.org/10.1016/j.ijdrr.2018.03.031
- Prasanna, R. K. R. (2010). *Information Systems for Supporting Fire Emergency Response* [Doctoral, Loughborough University]. Loughborough, UK. https://repository.lboro.ac.uk/articles/thesis/Information_Systems_for_Supporting_Fire_Emergency_Response/9496778
- Pulwarty, R. S., & Sivakumar, M. V. K. (2014). Information systems in a changing climate: Early warnings and drought risk management. *Weather and Climate Extremes*, *3*, 14-21. https://doi.org/10.1016/j.wace.2014.03.005
- Quinlivan, M., Lewis, L., Wilton, P., Kronast, H., Sadler, R., McRoberts, M., Speedy, J., Henry, H., & Bronwlie, K. (2021). *As it happened: Latest on Canterbury flooding* - *Monday, May 31*. Newshub. Retrieved 13 August 2021 from https://www.newshub.co.nz/home/new-zealand/2021/05/weather-liveupdates-latest-on-canterbury-flooding-monday-may-31.html
- Quinlivan, M., Speedy, J., Fyfe, J., Sadler, R., & Goosselink, D. (2020). Southland flooding: As it happened Evacuations underway for Gore, Mataura and Wyndham residents. Retrieved 12 February from https://www.newshub.co.nz/home/new-zealand/2020/02/southland-flooding-gore-mataura-residents-told-to-evacuate-immediately.html

- Rebelo Moreira, J. L., Ferreira Pires, L., Dockhorn Costa, P., & van Sinderen, M. (2018).
 Semantic Interoperability of Early Warning Systems: a Systematic Literature Review. In M. Zelm, F.-W. Jaekel, G. Doumeingts, & M. Wollschlaeger (Eds.), *Enterprise Interoperability: Smart Services and Business Impact of Enterprise Interoperability*. ISTE Ltd and John Wiley & Sons, Inc.
- Restrepo-Estrada, C., de Andrade, S. C., Abe, N., Fava, M. C., Mendiondo, E. M., & de Albuquerque, J. P. (2018). Geo-social media as a proxy for hydrometeorological data for streamflow estimation and to improve flood monitoring. *Computers and Geosciences*, *111*(October), 148-158. https://doi.org/10.1016/j.cageo.2017.10.010
- Reynolds, B., & Quinn, S. C. (2008). Effective Communication During an Influenza Pandemic: The Value of Using a Crisis and Emergency Risk Communication Framework. *Health Promotion Practice*, 9(4), 13S-17S. https://doi.org/10.1177/1524839908325267
- Reynolds, B., & Seeger, M. W. (2005). Crisis and emergency risk communication as an integrative model. *Journal of Health Communication*, *10*(1), 43-55. https://doi.org/10.1080/10810730590904571
- Rieger, K. L. (2019). Discriminating among grounded theory approaches. *Nursing Inquiry*, *26*(1), 1-12. https://doi.org/10.1111/nin.12261
- Ripberger, J. T., Silva, C. L., Jenkins-Smith, H. C., Carlson, D. E., James, M., & Herron, K. G. (2015). False Alarms and Missed Events: The Impact and Origins of Perceived Inaccuracy in Tornado Warning Systems. *Risk Analysis*, *35*(1), 44-56. https://doi.org/10.1111/risa.12262
- Ripberger, J. T., Silva, C. L., Jenkins-Smith, H. C., & James, M. (2015). The influence of consequence-based messages on public responses to tornado warnings. *Bulletin of the American Meteorological Society*, 96(4), 577-590. https://doi.org/10.1175/BAMS-D-13-00213.1
- RNC. (2018). *The Resilience Challenge*. Retrieved 21 July 2021 from https://resiliencechallenge.nz/
- RNZ. (2021, 17 July 2021). Wild weather: Homes evacuated, roads closed across top of South Island. *Radio New Zealand*. https://www.rnz.co.nz/news/national/447062/wild-weather-homesevacuated-roads-closed-across-top-of-south-island
- Robbins, J. C., & Titley, H. A. (2018). Evaluating high-impact precipitation forecasts from the Met Office Global Hazard Map using a global impact database. *Meteorological Applications*(July 2017), 548-560. https://doi.org/10.1002/met.1720
- Roche, S., Propeck-Zimmermann, E., & Mericskay, B. (2011). GeoWeb and crisis management: issues and perspectives of volunteered geographic information. *GeoJournal, 78*(1), 21-40. https://doi.org/10.1007/s10708-011-9423-9
- Roebber, P. J., & Bosart, L. F. (1996). The Contributions of Education and Experience to Forecast Skill. *Weather and Forecasting*, *11*(1), 21-40. https://doi.org/10.1175/1520-0434(1996)011<0021:TCOEAE>2.0.CO;2
- Rogers, D., & Tsirkunov, V. V. (2021). *Open Data: A Path to Climate Resilience and Economic Development in South Asia?* (157170). T. W. B. Group. https://documents.worldbank.org/en/publication/documents-reports/documentdetail/342901614809536537/open-data-a-path-to-climate-resilience-and-economic-development-in-south-asia
- Rogers, D. P., Kootval, H., & Tsirkunov, V. V. (2017). Early Warning, Resilient Infrastructure and Risk Transfer. In *Bridging Science and Policy Implication for Managing Climate Extremes* (pp. 65-79). www.gfdrr.org

Rogers, D. P., & Tsirkunov, V. V. (2013). Weather and Climate Resilience Effective Preparedness through National Meteorological and Hydrological Services (81113).

http://documents.worldbank.org/curated/en/308581468322487484/pdf/811 130PUB0Clim00Box0379830B0PUBLIC0.pdf

- Rollason, E., Bracken, L. J., Hardy, R. J., & Large, A. R. G. (2018). The importance of volunteered geographic information for the validation of flood inundation models. *Journal of Hydrology*, *562*, 267-280. https://doi.org/10.1016/j.jhydrol.2018.05.002
- Rosenbaum, S. (2010). Data governance and stewardship: designing data stewardship entities and advancing data access. *Health Services Research*, *45*(5 Pt 2), 1442-1455. https://doi.org/10.1111/j.1475-6773.2010.01140.x
- Ross, J., LaDue, D., & Correia, J. (2015, 7 January 2015). *Motivators and Important Factors Influencing Tornado Decisions in Oklahoma During May 2013* 10th Symposium on Societal Applications: Policy, Research and Practice, Pheonix, AZ, USA.
- Rossi, C., Acerbo, F. S., Ylinen, K., Juga, I., Nurmi, P., Bosca, A., Tarasconi, F., Cristoforetti, M., & Alikadic, A. (2018). Early detection and information extraction for weather-induced floods using social media streams. *International Journal of Disaster Risk Reduction*, *30*(March), 145-157. https://doi.org/10.1016/j.ijdrr.2018.03.002
- Rouse, H. (2011). *Flood risk management research in New Zealand Where are we, and where are we going?* (2012/04). GNS Science.
- Sabih, A., Russell, C., & Chang, C. L. (2020). Thunderstorm-related asthma can occur in New Zealand. *Respirology Case Reports*, *8*(7). https://doi.org/10.1002/rcr2.655
- Sadler, J. M., Goodall, J. L., Morsy, M. M., & Spencer, K. (2018). Modeling urban coastal flood severity from crowd-sourced flood reports using Poisson regression and Random Forest. *Journal of Hydrology*, *559*, 43-55. https://doi.org/10.1016/j.jhydrol.2018.01.044
- Safa, M., & Birendra, K. C. (2015). Damage reduction methods of centre pivot irrigation systems in windstorms. *International Journal of Advances in Science Engineering and Technology*, *3*(4), 141-145. http://researcharchive.lincoln.ac.nz/handle/10182/8456
- Sai, F., Cumiskey, L., Weerts, A., & Bhattacharya, B. (2018). Towards impact-based flood forecasting and warning in Bangladesh: a case study at the local level in Sirajganj district. *Natural Hazards and Earth System Sciences Discussions*(February), 1-22. https://doi.org/10.5194/nhess-2016-122
- Saima, H., Jaafar, J., Belhaouari, S., & Jillani, T. A. (2011). *Intelligent methods for weather forecasting: A review* 2011 National Postgraduate Conference, Perak, Malaysia.
- Saint-Martin, C., Javelle, P., & Vinet, F. (2018). DamaGIS: A multisource geodatabase for collection of flood-related damage data. *Earth System Science Data*, 10(2), 1019-1029. https://doi.org/10.5194/essd-10-1019-2018
- Schmidt, J., Matcham, I., Reese, S., King, A., Bell, R., Henderson, R., Smart, G., Cousins, J., Smith, W., & Heron, D. (2011). Quantitative multi-risk analysis for natural hazards: a framework for multi-risk modelling. *Natural Hazards*, 58(3), 1169-1192. https://doi.org/10.1007/s11069-011-9721-z
- Schnebele, E., & Cervone, G. (2013). Improving remote sensing flood assessment using volunteered geographical data. *Natural Hazards and Earth System Science*, 13(3), 669-677. https://doi.org/10.5194/nhess-13-669-2013

- Schroeter, S., Richter, H., Arthur, C., Wilke, D., Dunford, M., Wehner, M., & Ebert, E. (2021). Forecasting the impacts of severe weather. *Australian Journal of Emergency Management*, 10.47389/36(No 1), 76-83. https://doi.org/10.47389/36.1.76
- Scolobig, A., Prior, T., Schröter, D., Jörin, J., & Patt, A. (2015). Towards people-centred approaches for effective disaster risk management: Balancing rhetoric with reality. *International Journal of Disaster Risk Reduction*, *12*, 202-212. https://doi.org/10.1016/j.ijdrr.2015.01.006
- Scotland, J. (2012). Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms. *English Language Teaching*, 5(9), 9-16. https://doi.org/10.5539/elt.v5n9p9
- Sebastian, A., Lendering, K. T., Kothius, B., Brand, N., Jonkman, S. N. S. N., van Gelder, P. H. A. J. M., Godfroji, M., Kolen, B., Comes, T., Lhermitte, S. L. M., Kothuis, B. L. M., Brand, A. D., Jonkman, S. N. S. N., van Gelder, P. H. A. J. M., Godfroij, M., Kolen, B., Comes, M., Lhermitte, S. L. M., Meesters, K. J. M. G., van de Walle, B. A., Ebrahimi Fard, A., Cunningham, S., Khakzad Rostami, N., & Nespeca, V. (2017). *Hurricane Harvey Report: A fact-finding effort in the direct aftermath of Hurricane Harvey in the Greater Houston Region.* https://repository.tudelft.nl/islandora/object/uuid:54c24519-c366-4f2f-a3b9-0807db26f69c
- Secretary of Cabinet. (2011). Principles for Managing Data and Information held by the New Zealand Government, approved by Cabinet on 8 August 2011 (CAB Min (11) 29/12 refers). Wellington, New Zealand: New Zealand Government Retrieved from https://www.data.govt.nz/assets/Uploads/cab-min-11-29-12-8-august-2011.pdf
- Seidel, S., & Urquhart, C. (2016). On emergence and forcing in information systems grounded theory studies: The case of strauss and corbin. In L. P. Willcocks, C. Sauer, & M. C. Lacity (Eds.), *Enacting Research Methods in Information Systems: Volume 1* (Vol. 1, pp. 157-209). Palgrave Macmillan. https://doi.org/10.1007/978-3-319-29266-3_8
- Shabou, S., Ruin, I., Lutoff, C., Debionne, S., Anquetin, S., Creutin, J.-D., & Beaufils, X. (2017). MobRISK: a model for assessing the exposure of road users to flash flood events. *Natural Hazards and Earth System Sciences*, *17*(9), 1631-1651. https://doi.org/10.5194/nhess-17-1631-2017
- Sharma, V. K., Rao, G. S., Amminedu, E., Nagamani, P. V., Shukla, A., Rao, K. R. M., & Bhanumurthy, V. (2016). Event-driven flood management: design and computational modules. *Geo-spatial Information Science*, 19(1), 39-55. https://doi.org/10.1080/10095020.2016.1151212
- Sharpe, M. (2020). *Hay fever sufferers beware: The great yellow clouds of swirling dust are upon us.* Stuff. Retrieved 10 May 2021 from https://www.stuff.co.nz/national/health/122418563/hay-fever-sufferersbeware-the-great-yellow-clouds-of-swirling-dust-are-upon-us
- Sharpe, M. A. (2016). A flexible approach to the objective verification of warnings. *Meteorological Applications*, *23*(1), 65-75. https://doi.org/10.1002/met.1530
- Sheppard, B., Janoske, M., & Liu, B. (2012). Understanding Risk Communication Theory: A Guide for Emergency Managers and Communicators. www.start.umd.edu.
- Simon, T., Goldberg, A., & Adini, B. (2015). Socializing in emergencies-A review of the use of social media in emergency situations. *International Journal of Information Management*, 35, 609-619. https://doi.org/10.1016/j.ijinfomgt.2015.07.001

- Slavkovikj, V., Verstockt, S., Van Hoecke, S., & Van de Walle, R. (2014). Review of wildfire detection using social media. *Fire Safety Journal, 68*, 109-118. https://doi.org/10.1016/J.FIRESAF.2014.05.021
- SMC. (2020). Severe weather in Fiordland Expert Reaction. Science Media Centre. Retrieved 13 August 2021 from https://www.sciencemediacentre.co.nz/2020/02/03/severe-weather-infiordland-expert-reaction/
- Smith, L., Liang, Q., James, P., Lin, W., & Qiuhua Liang, C. (2017). Assessing the utility of social media as a data source for flood risk management using a real-time modelling framework. *Journal of Flood Risk Management*, 10(3), 370-380. https://doi.org/10.1111/jfr3.12154
- Smith, R. J., Muir, R. D. J., Walpole, M. J., Balmford, A., & Leader-Williams, N. (2003). Governance and the loss of biodiversity. *Nature*, *426*, 67-70. https://doi.org/10.1038/nature02025
- Smol, D. (2018). *Review of the Response to the Auckland Storm of 10 April 2018: A report to Auckland Coordinating Executives Group.*
- Sonnenwald, D. H. (2007). Scientific Collaboration. In B. Cronin (Ed.), *Annual Review* of Information Science and Technology (ARIST) (Vol. 41, pp. 643-681). Information Today. https://doi.org/10.1002/aris.2007.1440410121
- Srivastava, K., & Bhardwaj, R. (2013). Real-time nowcast of a cloudburst and a thunderstorm event with assimilation of Doppler weather radar data. *Natural Hazards*, *70*(2), 1357-1383. https://doi.org/10.1007/s11069-013-0878-5
- Stake, R. E. (1995). *The Art of Case Study Research*. SAGE Publications, Inc.
- Stats NZ. (2018). *2018 Census Place Summaries: Auckland Region*. Retrieved 12 February 2021 from https://www.stats.govt.nz/tools/2018-census-placesummaries/auckland-region
- Stats NZ. (2019). *Data Leadership*. StatsNZ. Retrieved 7 July 2021 from https://www.stats.govt.nz/about-us/data-leadership/
- Stats NZ. (2021). *Meshblock 2021 (generalised)*. Retrieved 12 May 2021 from https://datafinder.stats.govt.nz/layer/105176-meshblock-2021-generalised/#:~:text=A%20meshblock%20is%20the%20smallest,large%20ar ea%20of%20rural%20land.
- Steed, R. J., Fuenzalida, A., Bossu, R., Bondár, I., Heinloo, A., Dupont, A., Saul, J., & Strollo, A. (2019). Crowdsourcing triggers rapid, reliable earthquake locations. *Science Advances*, 5(4), eaau9824-eaau9824. https://doi.org/10.1126/sciadv.aau9824
- Steiner, T., Martin, J. R., Gordon, N. D., & Grant, M. A. (1997). Commercialisation in the provision of meteorological services in New Zealand. *Meteorological Applications*, 4, 247-257.
- Stepek, A., Wijnant, I. L., van der Schrier, G., van den Besselaar, E. J. M., & Klein Tank, A. M. G. (2012). Severe wind gust thresholds for Meteoalarm derived from uniform return periods in ECA&D. *Natural Hazards and Earth System Sciences*, 12(6), 1969-1981. https://doi.org/10.5194/nhess-12-1969-2012
- Stowell, J. (2020). *West Coast Flooding Event*. Retrieved 24 April 2021 from https://storymaps.arcgis.com/stories/e3769bfe5c8a4138b21f20c215a57f6b
- Strauss, A., & Corbin, J. (1998). *Basics of qualitative research: Techniques and procedures for developing grounded theory* (2nd ed.). SAGE Publications, Inc.
- Strauss, A. L. (1987). *Qualitative Analysis for Social Scientists*. Cambridge University Press.
- Strauss, A. L., & Corbin, J. M. (1990). *Basics of qualitative research: grounded theory procedures and techniques.* Sage Publications.

- stuff. (2019, 20 November 2019). Giant hail stones hammer Timaru as storm moves up the country. *stuff.* https://www.stuff.co.nz/timaruherald/news/117566456/hail-storm-hammers-timaru
- Susman, G. I., & Evered, R. D. (1978). An Assessment of the Scientific Merits of Action Research. *Administrative Science Quarterly*, *23*(4), 582-603. https://www.uio.no/studier/emner/matnat/ifi/nedlagteemner/INF9930/v12/undervisningsmateriale/Susman_Evered.pdf
- Tan, M. L., Prasanna, R., Stock, K., Hudson-Doyle, E., Leonard, G., & Johnston, D. (2017). Mobile applications in crisis informatics literature: A systematic review. *International Journal of Disaster Risk Reduction*, 24, 297-311. https://doi.org/10.1016/j.ijdrr.2017.06.009
- Tan, Q. (2021). Red alert warning for South Island: 'Danger to life' from severe gales. *NZ Herald*. https://www.nzherald.co.nz/nz/red-alert-warning-for-south-islanddanger-to-life-from-severe-gales/B6VHNNKDRMNXGFKLALUOWOQ7WQ/
- Tarbotton, C., Dall, F., Dominey-Howes, D., & Goff, J. (2015). The use of empirical vulnerability functions to assess the response of buildings to tsunami impact: Comparative review and summary of best practice. *Earth Science Reviews*, 142, 120-134. https://doi.org/10.1016/j.earscirev.2015.01.002
- Tarchiani, V., Massazza, G., Rosso, M., Tiepolo, M., Pezzoli, A., Housseini Ibrahim, M., Katiellou, G. L., Tamagnone, P., De Filippis, T., Rocchi, L., Marchi, V., & Rapisardi, E. (2020). Community and Impact Based Early Warning System for Flood Risk Preparedness: The Experience of the Sirba River in Niger. Sustainability, 12(5). https://doi.org/10.3390/su12051802
- Taylor, A. L., Kause, A., Summers, B., & Harrowsmith, M. (2019). Preparing for Doris: Exploring public responses to impact-based weather warnings in the United Kingdom. *Weather, Climate, and Society, 11*(4), 713-729. https://doi.org/10.1175/WCAS-D-18-0132.1
- Taylor, A. L., Kox, T., & Johnston, D. (2018). Communicating high impact weather: Improving warnings and decision making processes. *International Journal of Disaster Risk Reduction*, *30*, 1-4. https://doi.org/10.1016/j.ijdrr.2018.04.002
- Technical Advisory Group. (2017). *Ministerial Review Better Responses to Natural Disasters and Other Emergencies*. Department of the Prime Minister and Cabinet. https://dpmc.govt.nz/sites/default/files/2018-01/ministerial-review-better-responses-natural-disaster-other-emergencies.pdf
- Terti, G., Ruin, I., Anquetin, S., & Gourley, J. J. (2015). Dynamic vulnerability factors for impact-based flash flood prediction. *Natural Hazards*, 79, 1481-1497. https://doi.org/10.1007/s11069-015-1910-8
- Terti, G., Ruin, I., Kalas, M., Láng, I., Cangròs, A., & Lorini, V. (2019). ANYCaRE: A roleplaying game to investigate crisis decision-making and communication challenges in weather-related hazards. *Natural Hazards and Earth System Sciences, 19*, 1-43.
- Thakur, G., Sims, K., Mao, H., Piburn, J., Sparks, K., Urban, M., Stewart, R., Weber, E., & Bhaduri, B. (2018). Utilizing Geo-located Sensors and Social Media for Studying Population Dynamics and Land Classification. In S. L. Shaw & D. Sui (Eds.), *Human Dynamics Research in Smart and Connected Communities* (pp. 13-40). Springer, Cham. https://doi.org/10.1007/978-3-319-73247-3_2
- The World Bank. (2014). Understanding Risk in an Evolving World Emerging Best Practices in Natural Disaster Risk Assessment. https://www.gfdrr.org/sites/default/files/publication/Understanding_Risk-Web_Version-rev_1.8.0.pdf

References

- The World Bank. (2020). The Power of Partnership: Public and Private Engagement in
HydrometHydrometServices.T.W.Bank.https://openknowledge.worldbank.org/handle/10986/34615
- Theis, S. E., Hense, A., & Damrath, U. (2005). Probabilistic precipitation forecasts from a deterministic model: a pragmatic approach. *Meteorological Applications*, *12*(03). https://doi.org/10.1017/s1350482705001763
- Thien, F., Beggs, P. J., Csutoros, D., Darvall, J., Hew, M., Davies, J. M., Bardin, P. G., Bannister, T., Barnes, S., Bellomo, R., Byrne, T., Casamento, A., Conron, M., Cross, A., Crosswell, A., Douglass, J. A., Durie, M., Dyett, J., Ebert, E., Erbas, B., French, C., Gelbart, B., Gillman, A., Harun, N.-S., Huete, A., Irving, L., Karalapillai, D., Ku, D., Lachapelle, P., Langton, D., Lee, J., Looker, C., MacIsaac, C., McCaffrey, J., McDonald, C. F., McGain, F., Newbigin, E., O'Hehir, R., Pilcher, D., Prasad, S., Rangamuwa, K., Ruane, L., Sarode, V., Silver, J. D., Southcott, A. M., Subramaniam, A., Suphioglu, C., Susanto, N. H., Sutherland, M. F., Taori, G., Taylor, P., Torre, P., Vetro, J., Wigmore, G., Young, A. C., & Guest, C. (2018). The Melbourne epidemic thunderstorm asthma event 2016: an investigation of environmental triggers, effect on health services, and patient risk factors. *The Lancet Planetary Health, 2*(6), e255-e263. https://doi.org/10.1016/s2542-5196(18)30120-7
- Thomalla, F., Downing, T., Spanger-Siegfried, E., Han, G., & Rockström, J. (2006). Reducing hazard vulnerability: towards a common approach between disaster risk reduction and climate adaptation. *Disasters*, *30*(1), 39-48. https://doi.org/10.1111/j.1467-9523.2006.00305.x
- Thorpe, A., & Rogers, D. (2018). The Future of the Global Weather Enterprise: Opportunities and Risks. *Bulletin of the American Meteorological Society*, 99(10), 2003-2008. https://doi.org/10.1175/bams-d-17-0194.1
- Timonen, V., Foley, G., & Conlon, C. (2018). Challenges when using grounded theory: A pragmatic introduction to doing GT research. *International Journal of Qualitative Methods*, 17(1), 1-10. https://doi.org/10.1177/1609406918758086
- Tiranti, D., & Rabuffetti, D. (2010). Estimation of rainfall thresholds triggering shallow landslides for an operational warning system implementation. *Landslides*, 7(4), 471-481. https://doi.org/10.1007/s10346-010-0198-8
- Tkachenko, N., Jarvis, S., & Procter, R. (2017). Predicting floods with flickr tags. *PLoS* ONE, 12(2). https://doi.org/10.1371/journal.pone.0172870
- Ton, K. T., Gaillard, J. C., Cadag, J. R., & Naing, A. (2017). It takes two to tango: integrating meteorological knowledge and actions for disaster risk reduction. *Climate and Development*, 9(6), 479-492. https://doi.org/10.1080/17565529.2016.1174658
- Tonkin & Taylor Ltd. (2013). *Quantitative Landslide and Debris Flow Hazard Assessment Matatā Escarpment* (T&T Ref: 29115). https://www.whakatane.govt.nz/sites/www.whakatane.govt.nz/files/document s/about-council/council-projects/debris-flow-and-landslidehazards/Matata%20Quantitative%20Landslide%20Hazard%20Assessment%2 0-%20Web.pdf
- Tonkin & Taylor Ltd. (2015). *Supplementary Risk Assessment Debris Flow Hazard Matata, Bay of Plenty* (T&T Ref: 29115.2000). https://www.whakatane.govt.nz/sites/www.whakatane.govt.nz/files/document s/appendix_4_supplementary_risk_assessment_for_matata_and_peer_review .pdf
- Treadgold, G., Gunn, J., Morton, P., & Chambers, S. (2018). *Developing a regional* approach and strategy for geographical information systems for emergency management ISCRAM Asia-Pacific, Wellington, New Zealand.

- Turner, R., Paulik, R., Smart, G., Bind, J., Gray, S., Yang, E., Asora, L., Leiof, M., & Flay, R. (2013). Damage surveys following the Hobsonville Tornado in Auckland and Tropical Cyclone Evan in Samoa 16th Australasian Wind Engineering Society Workshop, Brisbane, Australia.
- Uccellini, L. W., & Ten Hoeve, J. E. (2019). Evolving the National Weather Service to Build a Weather-Ready Nation: Connecting Observations, Forecasts, and Warnings to Decision-Makers through Impact-Based Decision Support Services. *Bulletin of the American Meteorological Society*(October), 1923-1942. https://doi.org/10.1175/BAMS-D-18-0159.1
- UN Secretary General. (1989). International Decade for Natural Disaster Reduction: Report of the Secretary-General. https://digitallibrary.un.org/record/49133
- UNDRR. (2015a). From a Reactive to Proactive then People Centered Approach to DRR. UNDRR. https://www.undrr.org/publication/reactive-proactive-thenpeople-centered-approach-disaster-riskreduction#:~:text=From%20a%20reactive%20to%20proactive%20then%20p

eople%20centered%20approach%20to%20Disaster%20Risk%20Reduction,-Documents%20and%20publications&text=the%20growing%20participation %20of%20children,cutting%20nature%20of%20the%20HFA

- UNDRR. (2015b). Sendai Framework for Disaster Risk Reduction 2015 2030. UNDRR. https://www.undrr.org/publication/sendai-framework-disaster-risk-reduction-2015-2030
- UNDRR. (2016). *Report of the open-ended intergovernmental expert working group on indicators and terminology relating to disaster risk reduction*. U. N. G. Assembly.

https://www.preventionweb.net/files/50683_oiewgreportenglish.pdf

- UNDRR. (2020). Disaster Risk Reduction in New Zealand. In. Bangkok, Thailand: United Nations Office for Disaster Risk Reduction (UNDRR), Regional Office for Asia and the Pacific.
- UNDRR. (2021a). *Datasets: Understanding Disaster Risk Data*. Retrieved 1 July from https://www.preventionweb.net/risk/datasets
- UNDRR. (2021b). *Early warning system*. Retrieved 9 August 2021 from https://www.undrr.org/terminology/early-warning-system
- UNISDR. (2005). *Hyogo Framework for Action 2005-2015*. https://www.unisdr.org/2005/wcdr/intergover/official-doc/L-docs/Hyogoframework-for-action-english.pdf
- Urquhart, C., & Fernández, W. (2016). Using Grounded Theory Method in Information Systems: The Researcher as Blank Slate and Other Myths. In L. P. Willcocks (Ed.), *Enacting Research Methods in Information Systems: Volume 1* (Vol. 1, pp. 224-236). Association for Information Technology Trust.
- Urquhart, C., Lehmann, H., & Myers, M. D. (2010). Putting the 'theory' back into grounded theory: Guidelines for grounded theory studies in information systems. *Information Systems Journal*, 20(4), 357-381. https://doi.org/10.1111/j.1365-2575.2009.00328.x
- Veil, S., Reynolds, B., Timothy Sellnow, M. L., & Seeger, M. W. (2008). CERC as a Theoretical Framework for Research and Practice. *Health Promotion Practice*, 9(4), 26S-34S. https://doi.org/10.1177/1524839908322113
- Vollstedt, M., & Rezat, S. (2019). An Introduction to Grounded Theory with a Special Focus on Axial Coding and the Coding Paradigm. In G. Kaiser & N. Presmeg (Eds.), *Compendium for Early Career Researchers in Mathematics Education* (pp. 81-100). Springer International Publishing. https://doi.org/10.1007/978-3-030-15636-7_4

- Wagenmaker, R., Weaver, J., Garnet, G., Perry, B., Spinner, J., & Mann, G. (2011). *NWS Central Region Service Assessment Joplin, Missouri, Tornado - May 22, 2011.* https://www.weather.gov/media/publications/assessments/Joplin_tornado.p df
- Walrond, C. (n.d.). New Zealand's Isolated Position. In *Te Ara The Encyclopedia of New Zealand*. New Zealand: New Zealand Government.
- Weber, R. (1987). Toward a theory of artifacts: A paradigmatic base for information systems research. *Journal of Information Systems 1*(2), 3-19.
- Weber, R. (2003). Editor's Comments: Theoretically Speaking. *MIS Quarterly*, *27*(3), iiixii. https://doi.org/10.2307/30036536
- Wei, L., Li, J., & Yang, X. (2018). Experiments on Impact-based Forecasting and Riskbased Warning of Typhoon in China. *Tropical Cyclone Research and Review*, 31-36. https://doi.org/10.6057/2018TCRR01.04
- Wein, A., Potter, S., Johal, S., Doyle, E. E., & Becker, J. (2016). Communicating with the public during an earthquake sequence: Improving communication of geoscience by coordinating roles. *Seismological Research Letters*, 87(1).
- Weyrich, P., Scolobig, A., Bresch, D. N., & Patt, A. (2018). Effects of impact-based warnings and behavioural recommendations for extreme weather events. *Weather, Climate, and Society*, *10*, 781-796. https://doi.org/10.1175/WCAS-D-18-0038.1
- Weyrich, P., Scolobig, A., Walther, F., & Patt, A. (2020a). Do intentions indicate actual behaviour? A comparison between scenario-based experiments and real-time observations of warning response. *Journal of Contingencies and Crisis Management*, 28(3), 240-250. https://doi.org/10.1111/1468-5973.12318
- Weyrich, P., Scolobig, A., Walther, F., & Patt, A. (2020b). Responses to severe weather warnings and affective decision-making. *Natural Hazards and Earth System Sciences*, 20(10), 2811-2821. https://doi.org/10.5194/nhess-20-2811-2020
- Wiesche, M., Jurisch, M. C., Yetton, P. W., & Krcmar, H. (2017). Grounded Theory Methodology in Information Systems Research. *MIS Quarterly*, *41*(3), 685-701.
- Wilhelmi, O. V., & Morss, R. E. (2013). Integrated analysis of societal vulnerability in an extreme precipitation event: A Fort Collins case study. *Environmental Science* & Policy, 26, 49-62. https://doi.org/10.1016/j.envsci.2012.07.005
- Williams, C. A., & Eosco, G. M. (2021). Is a Consistent Message Achievable?: Defining "Message Consistency" for Weather Enterprise Researchers and Practitioners. Bulletin of the American Meteorological Society, 102(2), E279-E295. https://doi.org/10.1175/bams-d-18-0250.1
- Williamson, M. (1998). *Meteorological Service Contract*. Wellington, New Zealand: New Zealand Government, Retrieved from https://www.beehive.govt.nz/feature/meteorological-service-contract
- Wilson, L. J., & Giles, A. (2013). A new index for the verification of accuracy and timeliness of weather warnings. *Meteorological Applications*, *20*(2), 206-216. https://doi.org/10.1002/met.1404
- WMO. (2015). WMO Guidelines on Multi-Hazard Impact-Based Forecast and Warning Services (9789263111500). https://www.wmo.int/pages/prog/www/DPFS/Meetings/ET-OWFPS_Montreal2016/documents/WMOGuidelinesonMulti-hazardImpactbasedForecastandWarningServices.pdf
- WMO. (2017). Joint Meeting Of The Commission For Basic Systems Open Programme Area Group On Public Weather Services Delivery (OPAG-PSWD) Expert Teams On Impact Of Multi- Hazard Prediction And Communication (ET/IMPACT) And On Services and Products Innovation and Improvement (ET/SPII).

- WMO. (2018). Multi-hazard Early Warning Systems: A checklist. The First Multi-hazard Early Warning Conference, Cancun, Mexico.
- WMO. (2021). State of the Global Climate (WMO-No. 1264). https://library.wmo.int/index.php?lvl=notice_display&id=21880#.YHg0ABMz ZR0
- WREMO. (2020). Report on Significant Wave Hazards on Wellington South Coast: 15 April 2020.

https://www.owhirobay.org.nz/uploads/1/3/1/2/131212593/south_coast_swe ll_report.pdf

- WREMO. (2021). *Significant Wave Hazards Project: Closeout Report, April 2021*. W. R. E. M. Office.
- Wu, D., & Cui, Y. (2018). Disaster early warning and damage assessment analysis using social media data and geo-location information. *Decision Support Systems*, *111*(April), 48-59. https://doi.org/10.1016/j.dss.2018.04.005
- Wu, X., Howlett, M., & Ramesh, M. (2018). Policy Capacity: Conceptual Framework and Essential Components. In *Policy Capacity and Governance: Assessing Governmental Competences and Capabilities in Theory and Practice* (pp. 1-28). Springer Nature. https://doi.org/10.1007/978-3-319-54675-9
- Yates, D., & Paquette, S. (2011). Emergency knowledge management and social media technologies: A case study of the 2010 Haitian earthquake. *International Journal of Information Management*, 31(1), 6-13. https://doi.org/10.1016/j.ijinfomgt.2010.10.001
- Yin, W., Murray-Tuite, P., Ukkusuri, S. V., Gladwin, H., & Via, C. E. (2016). Modeling shadow evacuation for hurricanes with random-parameter logit model. *Transportation Research Record*, 2599, 43-51. https://doi.org/10.3141/2599-06
- Yu, M., Yang, C., & Li, Y. (2018). Big Data in Natural Disaster Management: A Review. *Geosciences*, 8(5). https://doi.org/10.3390/geosciences8050165
- Zaidi, J. (2012). *Data Governance Demystified Lessons from the Trenches*. Dataversity. Retrieved 7 July 2021 from http://www.dataversity.net/datagovernance-demystified-lessons-from-the-trenches/
- Zeigler, D. J., Brunn, S. D., & Johnson, J. H. J. (1981). Evacuation from a Nuclear Technological Disaster. *Geographical Review*, *71*(1), 1-16. https://doi.org/10.2307/214548
- Zhang, D., Zhou, L., & Nunamaker, J. F. J. (2002). A Knowledge Management Framework for the Support of Decision Making in Humanitarian Assistance/Disaster Relief. *Knowledge and Information Systems*, *4*, 370-385. https://doi.org/10.1007/s101150200012
- Zhang, Q., Li, L., Ebert, B., Golding, B., Johnston, D., Mills, B., Panchuk, S., Potter, S., Riemer, M., Sun, J., Taylor, A., Jones, S., Ruti, P., & Keller, J. (2019). Increasing the value of weather-related warnings. *Science Bulletin*, *64*(10), 647-649. https://doi.org/10.1016/j.scib.2019.04.003
- Zhang, Q., Ng, C.-P., Dai, K., Xu, J., Tang, J., Sun, J., & Mu, M. (2021). Lessons Learned from the Tragedy during the 100 km Ultramarathon Race in Baiyin, Gansu Province on 22 May 2021. *Advances in Atmospheric Sciences*. https://doi.org/10.1007/s00376-021-1246-0
- Zolkafli, A., Brown, G., & Liu, Y. (2017). An Evaluation of Participatory GIS (PGIS) for Land Use Planning in Malaysia. *The Electronic Journal of Information Systems in Developing Countries*, *83*, 1-23. www.ejisdc.org